JAKKS PACIFIC INC Form 10-Q August 14, 2009

UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

Washington, D.C. 20549
FORM 10-Q
(Mark one) x QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934
For the quarterly period ended June 30, 2009
OR
" TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934
For the transition period from to
Commission file number: 0-28104
JAKKS Pacific, Inc.
(Exact Name of Registrant as Specified in Its Charter)
Delaware 95-4527222
(State or Other Jurisdiction of Incorporation or Organization) (I.R.S. Employer Identification No.)
22619 Pacific Coast Highway
Malibu, California 90265
(Address of Principal Executive Offices) (Zip Code)
Registrant's telephone number, including area code: (310) 456-7799

Indicate by check mark whether the registrant: (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes x No "

Indicate by check mark whether the registrant has submitted electronically and posted on its corporate Web site, if any, every Interactive Data File required to be submitted and posted pursuant to Rule 405 of Regulation S-T during the preceding 12 months (or for such shorter period that the registrant was required to submit and post such files). Yes o No "

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, or a non-accelerated filer, or a smaller reporting company. See the definitions of "large accelerated filer," "accelerated filer" and "smaller reporting company" in Rule 12b-2 of the Exchange Act. (Check one):

Non-accelerated filer

Large accelerated filer Accelerated filer (Do not check if a smaller Smaller reporting company reporting company) ...

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act). Yes "No x

The number of shares outstanding of the issuer's common stock is 27,928,231 (as of August 13, 2009).

JAKKS PACIFIC, INC. AND SUBSIDIARIES

INDEX TO QUARTERLY REPORT ON FORM 10-Q Quarter Ended June 30, 2009

ITEMS IN FORM 10-Q

Page Part I FINANCIAL INFORMATION Item 1. **Financial Statements** Condensed Consolidated Balance Sheets - December 31, 2008 and June 30, 2009 2 (unaudited) Condensed Consolidated Statements of Operations for the Three and Six Months Ended June 30, 2008 and 2009 (unaudited) 3 Condensed Consolidated Statements of Cash Flows for the Six Months Ended June 30, 2008 and 2009 (unaudited) 4 Notes to Condensed Consolidated Financial Statements (unaudited) 5 Item 2. Management's Discussion and Analysis of Financial Condition and Results of **Operations** 20 30 Item 3. **Ouantitative and Oualitative Disclosures About Market Risk** Controls and Procedures Item 4. 30 Part II OTHER INFORMATION Item 1. 31 **Legal Proceedings** Item 1A. Risk Factors 34 Item 2. Unregistered Sales of Equity Securities and Use of Proceeds None Item 3. **Defaults Upon Senior Securities** None Item 4. Submission of Matters to a Vote of Security Holders None Item 5. Other Information None Item 6. **Exhibits** 41 Signatures 42 Exhibit 31.1 Exhibit 31.2 Exhibit 31.3 Exhibit 32.1 Exhibit 32.2 Exhibit 32.3

DISCLOSURE REGARDING FORWARD-LOOKING STATEMENTS

This report includes "forward-looking statements" within the meaning of Section 27A of the Securities Act of 1933 and Section 21E of the Securities Exchange Act of 1934. For example, statements included in this report regarding our financial position, business strategy and other plans and objectives for future operations, and assumptions and predictions about future product demand, supply, manufacturing, costs, marketing and pricing factors are all forward-looking statements. When we use words like "intend," "anticipate," "believe," "estimate," "plan", "expect" or words of similar import, we are making forward-looking statements. We believe that the assumptions and expectations reflected in such forward-looking statements are reasonable and are based on information available to us on the date hereof, but we cannot assure you that these assumptions and expectations will prove to have been correct or that we will take any

action that we may presently be planning. We are not undertaking to publicly update or revise any forward-looking statement if we obtain new information or upon the occurrence of future events or otherwise.

JAKKS PACIFIC, INC. AND SUBSIDIARIES

CONDENSED CONSOLIDATED BALANCE SHEETS

(In thousands, except share amounts)

TS ets sish		December 31, 2008 (*)	June 30, 2009 (Unaudited)	
tes shit	ETS			
\$ 169,520 \$ 195 net of for e \$2,005 147,587 87,944 ensessurrent 29,670 22,288 come 17,993 at assets 475,197 d ture and 12,390 cooling 63,075 nts 5,947 81,412 ullated a and n 52,914 d net 28,498 in video venture 53,184 et 427,693 i, net 10,491 and 33,061	ets			
195 net of for e \$2,005 2	ash			
net of for e \$2,005 147,587 87,944 enses trrent 29,670 22,288 come 17,993 at assets 475,197 d ture and 12,390 cooling 63,075 nts 5,947 81,412 ulated at and n 52,914 d net 28,498 in video venture 53,184 eet 427,693 t, net 10,491 and 33,061		\$ 169,520	\$	
net of for e \$2,005 147,587 87,944 enses trrent 29,670 22,288 come 17,993 at assets 475,197 d ture and 12,390 cooling 63,075 nts 5,947 81,412 ulated at and n 52,914 d net 28,498 in video venture 53,184 eet 427,693 t, net 10,491 and 33,061				
for e \$2,005 147,587 87,944 enses urrent 29,670 22,288 eome 17,993 at assets 475,197 d ture and 12,390 eooling 63,075 ents 5,947 81,412 ulated a and n 52,914 d enet 28,498 in video venture 53,184 ect 427,693 i, net 10,491 and 33,061		195		
147,587 87,944 enses arrent 29,670 22,288 come 17,993 at assets 475,197 d ture and 12,390 cooling 63,075 nts 5,947 81,412 ulated n and n 52,914 d net 28,498 in video venture 53,184 let 427,693 i, net 10,491 and 33,061	net of for e			
87,944 enses irrent 29,670 22,288 come 17,993 at assets 475,197 d ture and 12,390 cooling 63,075 nts 5,947 81,412 ulated n and n 52,914 d net 28,498 in video venture 53,184 et 427,693 t, net 10,491 and 33,061	\$2,005			
87,944 enses irrent 29,670 22,288 come 17,993 at assets 475,197 d ture and 12,390 cooling 63,075 nts 5,947 81,412 ulated n and n 52,914 d net 28,498 in video venture 53,184 et 427,693 t, net 10,491 and 33,061		1 47 507		
29,670 22,288 come 17,993 at assets 475,197 d 12,390 cooling 63,075 ats 5,947 81,412 ulated a and an and an and an 52,914 d d net 28,498 in video eventure 53,184 etet 427,693 is, net 10,491 and 33,061	1			
29,670 22,288 come 17,993 at assets 475,197 d ture and 12,390 cooling 63,075 and n 52,914 d net 28,498 in video venture 53,184 et 427,693 is, net 10,491 and 33,061	oncoc	67,944		
29,670 22,288 come 17,993 at assets 475,197 d ture and 12,390 cooling 63,075 ants 5,947 81,412 ulated a and an 52,914 d net 28,498 in video venture 53,184 eet 427,693 is, net 10,491 and 33,061				
22,288 come 17,993 at assets 475,197 d ture and 12,390 cooling 63,075 atts 5,947 81,412 ulated and n 52,914 d net 28,498 in video venture 53,184 etet 427,693 is, net 10,491 and 33,061	arrent	29,670		
17,993 at assets 475,197 d ture and 12,390 cooling 63,075 ats 5,947 81,412 ulated an and an 52,914 d anet 28,498 in video venture 53,184 set 427,693 s, net 10,491 and 33,061		27,070		
17,993 at assets 475,197 d ture and 12,390 cooling 63,075 ats 5,947 81,412 ulated an and an 52,914 d anet 28,498 in video venture 53,184 set 427,693 s, net 10,491 and 33,061		22,288		
ture and 12,390 tooling 63,075 nts 5,947 81,412 ulated n and n 52,914 d net 28,498 in video venture 53,184 tet 427,693 5, net 10,491 and 33,061	come			
ture and 12,390 cooling 63,075 nts 5,947 81,412 ulated n and n 52,914 d nnet 28,498 in video venture 53,184 let 427,693 s, net 10,491 and 33,061				
ture and 12,390 cooling 63,075 nts 5,947 81,412 ulated n and n 52,914 d net 28,498 in video venture 53,184 let 427,693 s, net 10,491 and 33,061	nt assets	475,197		
12,390 tooling 63,075 Ints 5,947 81,412 ulated In and In 52,914 Int 28,498 In video Inventure 53,184 Inter 427,693 Is, net 10,491 Inter 33,061	d			
12,390 tooling 63,075 Ints 5,947 81,412 ulated In and In 52,914 Int 28,498 In video Inventure 53,184 Inter 427,693 Is, net 10,491 Inter 33,061	4			
tooling 63,075 Ints 5,947 81,412 Ulated 1 and 1	ture and			
states 5,947 81,412 ulated n and n 52,914 d net 28,498 in video venture 53,184 et 427,693 is, net 10,491 and 33,061	cooling			
81,412 ulated n and n	loomig	03,073		
81,412 ulated n and n	nts	5.947		
ulated n and n				
d net 28,498 in video venture 53,184 net 427,693 s, net 10,491 and 33,061		,		
net 28,498 in video venture 53,184 net 427,693 s, net 10,491 and 33,061	n	52,914		
in video venture 53,184 set 427,693 s, net 10,491 and 33,061				
venture 53,184 net 427,693 s, net 10,491 and 33,061				
net 427,693 s, net 10,491 and 33,061				
s, net 10,491 and 33,061				
and 33,061				
33,061		10,491		
	anu	33.061		
		\$ 1,028,124		

ES AND DLDERS' TY							
oilities							
ayable	\$	57,432	\$				
penses		61,780					
sales							
		23,317					
se		417					
es		7,190					
e senior		_	-				
nt	1	50,136					
aoma	1	50,150					
come		26,237					
payable		4,686					

Can I revoke my proxy or change my voting in

Yes. You may revoke your proxy and/or change your vote at any time before your proxy is voted at the special meeting common shareholder of record, you ca

sending a duly signed revocation to KFN at 555 California Street, 50th Floor, San Francisco, CA, 94104, Att Secretary, that bears a date later than the date of the proxy you want to revoke and is received prior to the spec

submitting a valid, later-dated proxy by mail, telephone or Internet that is received prior to the special

attending the special meeting and voting by ballot in person (your attendance at the special meeting will not, by in any proxy that you have previous any proxy that you have previous any proxy that you have previous attendance at the special meeting will not, by it is a special meeting and voting by ballot in person (your attendance at the special meeting will not, by it is a special meeting and voting by ballot in person (your attendance at the special meeting will not, by it is a special meeting and voting by ballot in person (your attendance at the special meeting will not, by it is a special meeting and voting by ballot in person (your attendance at the special meeting will not, by it is a special meeting will not be a special meeting will no

If you hold your KFN common shares through a broker or other nominee, you must follow the directions you your broker or other nominee in order to revoke your proxy or change your voting

What happens if I sell my common shares after the record date but before the spec

The record date for the special meeting (the close of business on February 26, 2014) is earlier than the date of the special the date that the merger is expected to be completed. If you sell or otherwise transfer your KFN common shares after the but before the date of the special meeting, you will retain your right to vote at the special meeting (unless otherwise agr you and the transferee). However, you will not have the right to receive the merger consideration to be received by KF shareholders in the merger. In order to receive the merger consideration, you must hold your common shares through comp

What do I do if I receive more than one set of voting

You may receive more than one set of voting materials for the special meeting, including multiple copies statement/prospectus, proxy cards and/or voting instruction forms. This can occur if you hold your common shares in m

brokerage account, if you hold common shares directly as a record holder and also in "street name," or otherwise through and in certain other circumstances. If you receive more than one set of voting materials, we encourage you to vote and/o set separately in order to ensure that all of your common shares.

Am I entitled to appraisal rights if I vote against the adoption of the merger

No. Appraisal rights, which generally confer on holders of securities who do not vote in favor of or consent to a merge demand payment of fair value for their securities as determined by a court in a judicial proceeding instead of a consideration offered to such holders in connection with the merger, are not available in connection with the merge Delaware Limited Liability Company Act or under KFN's Second Amended and Restated Operating Agreement, as amend referred to in this proxy statement/prospectus as KFN's operating

Table

Is completion of the merger subject to any

Yes. In addition to the adoption of the merger agreement by KFN common shareholders, completion of the merger require of the necessary governmental clearances and the satisfaction or, to the extent permitted by applicable law, waiver conditions specified in the merger agreement. For a more complete summary of the conditions that must be satisfied (or, permitted by applicable law, waived) prior to completion of the merger, see "The Merger Agreement Conditions to Conthe Merger" beginning on page 134 of this proxy statemen

When do you expect to complete

KFN and KKR are working towards completing the merger promptly. KFN and KKR currently expect to complete the first half of 2014, subject to receipt of KFN shareholder approval, regulatory approvals and clearances and other usual ar closing conditions. However, no assurance can be given as to when, or if, the merge

What are the expected U.S. federal income tax consequences to a KFN common shareholder as a result of the to contemplated by the merger

The merger will be a taxable transaction for U.S. federal income tax purposes. If you are a U.S. holder of KFN common U.S. federal income tax purposes, your receipt of KKR common units and cash in lieu of fractional units in exchange for common shares in the merger generally will cause you to recognize gain or loss measured by the difference, if any, betwee of (A) the fair market value of any KKR common units received, (B) the amount of cash received and (C) your shour nonrecourse debt immediately prior to the merger and (ii) your adjusted tax basis in your KFN common shares. Any such recognized generally will be treated as capital gain or loss and will be long-term capital gain or loss if your holding pe KFN common shares exceeds one year. However, a portion of any such gain will be treated as ordinary income attributable to your allocable share of unrealized gain or loss in KFN's assets as described in Section 751 of the U.S. Inter Code, which is referred to in this proxy statement/prospectus as the Code. If you are a non-U.S. holder of KFN common portion of any gain recognized by you in the merger (which will be calculated in the same manner described above for a may be treated for U.S. federal income tax purposes as effectively connected income, and hence you may be subject to income tax on such portion. All holders of KFN common shares should consult their own tax advisor for a full understar the merger will affect their taxes. See "Material U.S. Federal Income Tax Consequences of the Merger" beginning on pag proxy statement/prospectus for further discussion of the U.S. federal income tax consequences of the Merger.

What are the expected U.S. federal income tax consequences for a KFN common shareholder of the owners common units after the merger is

Subject to the requirements below, KKR will be treated, for U.S. federal income tax purposes, as a partnership association or a publicly traded partnership taxable as a corporation. As a result, a U.S. KKR common unitholder will U.S. federal, state, local and possibly, in some cases, foreign income taxation on its allocable share of KKR's items of it loss, deduction and credit (including its allocable share of those items of any entity in which KKR invests that is partnership or is otherwise subject to tax on a flow-through basis) for each of KKR's taxable years ending with unitholder's taxable year, regardless of whether or when such unitholder receives cash distributions. KKR will be treated federal income tax purposes, as a partnership described above so long as 90% of KKR's gross

Table

each taxable year constitutes qualifying income as defined in Section 7704 of the Code and KKR is not required to a investment company under the U.S. Investment Company Act of 1940, as amended, which is referred to statement/prospectus as the Investment Company Act, on a continuing basis, assuming there is no characteristics.

KKR expects that it will be engaged in a U.S. trade or business for U.S. federal income tax purposes, including investments in U.S. real property holding corporations, real estate assets and natural resource and oil and gas investments case some portion of KKR's income would be treated as effectively connected income with respect to non-U.S. holders, or extent KKR's income is treated as ECI, non-U.S. KKR common unitholders generally would be subject to withholding allocable share of such income, would be required to file a U.S. federal income tax return for such year reporting their all of income effectively connected with such trade or business and any other income treated as ECI, and would be subject to income tax at regular U.S. tax rates on any such income (state and local income taxes and filings may also apply in Non-U.S. KKR common unitholders that are corporations may also be subject to a 30% branch profits tax (potentially rean applicable treaty) on their actual or deemed distributions of such income. In addition, distributions to non-U.S. K unitholders that are attributable to profits on the sale of a U.S. real property interest may also be subject to 30% withholding non-U.S. KKR common unitholders may be subject to 30% withholding on allocations of KKR's income that are U.S. so determinable annual or periodic income under the Code, unless an exemption from or a reduced rate of such withhold (under an applicable treaty of the Code) and certain tax status information

All holders of KFN common shares should consult their own tax advisor for a full understanding of the tax consequences of KKR common units after the merger is completed. See "Material U.S. Federal Tax Consequences of KKR C Ownership" beginning on page 184 of this proxy statement/prospectus for further discussion of the U.S. federa consequences of the ownership of KKR co

What do I need

Carefully read and consider the information contained in and incorporated by reference into this proxy statement/prospects its annexes. Then, please vote your KFN common shares, which you

submitting your proxy by telephone or via the Internet by following the instructions included on your

completing, dating, signing and returning the enclosed proxy card in the accompanying postage-paid

attending the special meeting and voting by ball

If you hold KFN common shares through a broker or other nominee, please instruct your broker or nominee KFN common shares by following the instructions that the broker or nominee provides to you with the

Should I send in my share certif

No. KFN shareholders should not send in their share certificates at this time. After completion of the merger, KKR's exemple and instructions for exchanging your KFN common shares for the merger consideration common units you receive in the merger will be issued in bool

Whom should I call with

KFN shareholders should call Innisfree M&A Incorporated, KFN's proxy solicitor, toll-free at (888) 750-5834 (banks and collect at (212) 750-5833) with any questions about the merger or the special meeting, or to obtain additional copies statement/prospectus, proxy cards or voting instru

<u>Table</u>

SUMMARY

This summary highlights selected information from this proxy statement/prospectus. You are urged to read carefully the statement/prospectus and the other documents referred to in this proxy statement/prospectus because the information in this sec provide all the information that might be important to you with respect to the merger agreement, the merger and the other no considered at the KFN special meeting. See "Where You Can Find More Information" beginning on page 234 of this proxy statemen

The Parties (Se

KKR Financial Ho

KFN is a Delaware limited liability company whose common shares are publicly traded on the NYSE under the symbol "KFI specialty finance company with expertise in a range of asset classes. The principal executive offices of KFN are located at 555 California 94104, and its telephone number is (41:

v

KKR is a Delaware limited partnership whose common units are publicly traded on the NYSE under the symbol "KKR." KK investment firm with \$94.3 billion in assets under management as of December 31, 2013. The principal executive offices of KKR a 9 West 57th Street, Suite 4200, New York, New York 10019, and its telephone number is (212) 750-8300. KKR and its subsidiaries to in this proxy statement/prospectus as the

KKR Fund H

Fund Holdings is an exempted limited partnership formed under the laws of the Cayman Islands and is a subsidiary of KKR. The executive offices of Fund Holdings are located c/o KKR 9 West 57th Street, Suite 4200, New York, New York 10019, and its telephone the subsidiary of KKR.

Copal Merg

Merger Sub is a Delaware limited liability company and is a direct, wholly owned subsidiary of Fund Holdings. The princip offices of Merger Sub are located at 9 West 57th Street, Suite 4200, New York, New York 10019, and its telephone number is (21)

The Merger (Se

KFN, KKR, Fund Holdings and Merger Sub have entered into the merger agreement. Subject to the terms and conditions of agreement and in accordance with Delaware law, Merger Sub will be merged with and into KFN, with KFN continuing as the survey Upon completion of the merger, KFN will be a direct subsidiary of Fund Holdings, and KFN common shares will no longer be put KFN's preferred shares will remain outstanding and listed on the NYSE after completion of

Merger Consideration (Se

The merger agreement provides that, at the effective time of the merger, each KFN common share issued and outstanding immediate to the effective time will be converted into the right to receive 0.51 KKR common units. Each KFN common share that is held by Fu or any subsidiary of Fund Holdings immediately prior to the effective time of the merger will be cancelled without any conversion or consideration in res

Table

KKR will not issue any fractional KKR common units in the merger. Instead, the total number of KKR common units the common shareholder will receive in the merger will be rounded down to the nearest whole number, and each KFN common shareholder cash, without interest, for any fractional KKR common unit that such person would otherwise receive in

Treatment of Equity Awards (Se

Options. Each KFN option to purchase a KFN common share that is outstanding and unexercised immediately prior to the e of the merger will be cancelled and converted into the right to receive an amount in cash equal to the excess of the cash value common units over the exercise price per KFN common share subject to

Restricted Shares. Each restricted KFN common share that is outstanding immediately prior to the effective time of the m converted into 0.51 restricted KKR common units having the same terms and conditions as applied to such restricted KFN common immediately prior to the effective time of the magnetic field of the magnetic field

Phantom Shares. Each outstanding phantom share under KFN's Non-Employee Directors' Deferred Compensation and Share will be converted into a phantom share in respect of 0.51 KKR common units and will otherwise remain subject to the terms

KFN Special Shareholder Meeting; Shareholders Entitled to Vote; Vote Required (S

Meeting. The KFN special meeting will be held at The Bently Reserve, Gateway Room, 301 Battery Street, San Francisc 94111, on April 30, 2014, at 9:00 a.m., local time. At the special meeting, KFN common shareholders will be asked to vote on t

Proposal 1: Adoption of the Merger Agreement. The affirmative vote of holders of at least a majority of the KFN common shares entitled to vote thereon, including a majority of the outstanding KFN common shares en thereon held by shareholders other than KKR and its affiliates. Accordingly, abstentions and unvoted shares same effect as votes "AGAINST" the adoption of the merger

Proposal 2: Adjournment of the KFN Special Meeting (if necessary). The affirmative vote of holders of at least of the outstanding KFN common shares entitled to vote thereon present in person or represented by proxy meeting. Accordingly, an abstention will have the same effect as a vote "AGAINST" the proposal, although an unwill have no effect on the proposal assuming that a quorum is present at the spe

Record Date. Only KFN common shareholders of record as of the close of business on February 26, 2014 will be entitled to record and to vote at the special meeting. As of the close of business on the record date of February 26, 2014, there were 204,824,159 K shares outstanding and entitled to vote at the special meeting, including 4,658,021 shares held by KKR and its affiliates. Each hold common share is entitled to one vote for each common share owned as of the

Required Vote. To adopt the merger agreement, holders of at least a majority of the outstanding KFN common shares en thereon, including a majority of the outstanding KFN common shares entitled to vote thereon held by shareholders other than affiliates, must vote in favor of the adoption of the merger agreement. The merger cannot be completed unless KFN common stadept the merger agreement. Because approval is based on the affirmative vote of at least a majority of the outstanding KI shares, a KFN common shareholder's failure to vote, an abstention from voting or the failure of a KFN common shareholder his or

Table

"street name" through a broker or other nominee to give voting instructions to such broker or other nominee will have the as a vote "AGAINST" the adoption of the merger

To approve the adjournment of the KFN special meeting, if necessary to solicit additional proxies if there are not sufficient very the merger agreement at the time of the special meeting, the affirmative vote of holders of at least a majority of the outstanding K shares entitled to vote thereon present in person or represented by proxy at the special meeting is required. Because approval of this based on the voting power present with respect to such proposal, abstentions will have the same effect as a vote "AGAINST" the parallel failures to be present to vote and failures of KFN common shareholders who hold their shares in "street name" through brokers or other nominees will have no effect on the vote held on such proposal provided that

Share Ownership of KFN's Directors and Executive Officers. As of the close of business on the record date for the special me directors and executive officers beneficially owned and had the right to vote 1,662,070 common shares at the special meeting, approximately 0.8% of the KFN common shares entitled to vote at the special meeting.

It is expected that KFN's directors and executive officers will vote their shares "FOR" the adoption of the merger agreement and proposal to adjourn the special meeting, if necessary, although none of them has entered into any agreement requiring the

Share Ownership of Affiliates of KKR. As of the close of business on the record date for the special meeting, affiliates of KKR owned and had the right to vote 4,658,021 common shares at the special meeting, representing approximately 2.3% of the KFN common shares at the special meeting, representing approximately 2.3% of the KFN common shares at the special meeting, representing approximately 2.3% of the KFN common shares at the special meeting, representing approximately 2.3% of the KFN common shares at the special meeting, representing approximately 2.3% of the KFN common shares at the special meeting, representing approximately 2.3% of the KFN common shares at the special meeting, representing approximately 2.3% of the KFN common shares at the special meeting, representing approximately 2.3% of the KFN common shares at the special meeting, representing approximately 2.3% of the KFN common shares at the special meeting, representing approximately 2.3% of the KFN common shares at the special meeting, representing approximately 2.3% of the KFN common shares at the special meeting, representing approximately 2.3% of the KFN common shares at the special meeting approximately 2.3% of the KFN common shares at the special meeting approximately 2.3% of the KFN common shares at the special meeting approximately 2.3% of the KFN common shares at the special meeting approximately 2.3% of the KFN common shares at the special meeting approximately 2.3% of the KFN common shares at the special meeting approximately 2.3% of the KFN common shares at the special meeting approximately 2.3% of the KFN common shares at the special meeting approximately 2.3% of the KFN common shares at the special meeting approximately 2.3% of the KFN common shares at the special meeting approximately 2.3% of the KFN common shares at the special meeting approximately 2.3% of the KFN common shares at the special meeting approximately 2.3% of the KFN common shares at the special meeting approximately 2.3% of the KFN common shares at the special meet

It is expected that the affiliates of KKR will vote their shares "FOR" the adoption of the merger agreement and "FOR" the adjourn the special meeting, if necessary, although none of them has entered into any agreement requiring them to do so. Any sucl vote will not, however, affect the required approval of the proposal to adopt the merger agreement by the affirmative vote of a major common shares entitled to vote thereon other than KFN common shares held by KKR and

Recommendation of the KFN Board of Directors and Reasons for the Merger; Fairness of the Merger (S

The KFN board of directors, upon the unanimous recommendation of a transaction committee consisting solely of independent recommends that KFN shareholders vote "FOR" the adoption of the merge

In the course of reaching its decision to approve the merger agreement and the transactions contemplated by the merger agreement board of directors considered a number of factors in its deliberations. For a more complete discussion of these factors, Factors Recommendation of the KFN Board of Directors and Reasons for the Merger; Fairness of the Merger" beginning on page 38

Opinion of the Financial Advisor to the KFN Transaction Committee (

Sandler O'Neill & Partners, L.P., referred to in this proxy statement/prospectus as Sandler O'Neill, acted as financial a transaction committee in connection with the proposed transaction and participated in certain of the negotiations leading to the exe merger agreement. At the December 13, 2013 meeting of the transaction committee, Sandler O'Neill delivered to the transaction coral opinion, which was subsequently confirmed in writing on December 16, 2013, that, as of December 16, 2013, the merger consi fair to the holders of KFN common s

Table

financial point of view. The full text of Sandler O'Neill's opinion is attached as Annex B to this proxy statement/prospectus. outlines the procedures followed, assumptions made, matters considered and qualifications and limitations on the review une Sandler O'Neill in rendering its opinion. The description of the opinion set forth below is qualified in its entirety by reference text of the opinion. Holders of KFN common shares are urged to read the entire opinion carefully in connection with their confidence of the proposition of the prop

KKR Unitholder Approval is Not Required (Se

KKR unitholders are not required to adopt the merger agreement or approve the merger or the issuance of KKR common units i

Ownership of KKR after

Based on the number of outstanding KFN common shares (including restricted shares) outstanding as of March 18, 2014, KK issue approximately 104.5 million KKR common units to KFN common shareholders pursuant to the merger agreement. The actu KKR common units to be issued pursuant to the merger agreement will be determined at the completion of the merger based on t ratio of 0.51 and the number of KFN common shares (including restricted shares) outstanding

As of March 18, 2014, KKR Holdings L.P., which is referred to in this proxy statement/prospectus as KKR Holdings, owns partnership interests of each of KKR Management Holdings L.P. and Fund Holdings (the holding companies of the KKR business referred to, together, in this proxy statement/prospectus as the KKR Group Partnerships. These partnership interests are referred to statement/prospectus as the KKR Group Partnership units, and may be collectively exchanged, on a quarterly basis, for KKR commone-for-one basis, subject to customary conversion rate adjustments for splits, unit distributions and recl

As of March 18, 2014, KKR had 300,354,288 of its common units outstanding, which excludes KKR common units beneficial KKR Holdings through its ownership of KKR Group Partnership units, KKR common units available for future issuance KKR & Co. L.P. 2010 Equity Incentive Plan and KKR common units available for future issuance in connection with KKR's acquision the number of KKR common units outstanding as of March 18, 2014 referenced in the immediately preceding sentence, it is anti immediately after the completion of the merger, former KFN common shareholders will own approximately 25.8% of the outst common units. However, assuming all of the KKR Group Partnership units held by KKR Holdings as of March 18, 2014 were extended that, immediately after the completion of the merger, former KFN common will own approximately 13.1% of the outstanding KKR common units. See "Organizational Chart" for a simplified diagram short organization organization.

Holders of KKR common units do not elect KKR's managing partner or its board of directors and, unlike the holders of KF shares with respect to KFN, have only limited voting rights on matters affecting KKR's business and therefore limited ability decisions regarding KKR's business, which is run by its managing partner. See "Comparison of KKR Common Units and KFN Com beginning on page 212 of this proxy statemen

Interests of Directors and Executive Officers of KFN in the Merger (

KFN's executive officers and directors have interests in the merger that are different from, or in addition to, their interests shareholders of KFN. The members of the KFN board of directors were aware of and considered these interests, among other evaluating and negotiating the merger agreement and the merger, and in recommending to KF

Table

shareholders that the merger agreement be adopted. KFN's executive officers (including any executive officers who are members board of directors) did not participate in the vote by the KFN board of directors relating to the merger. The merger agreement pro conversion of KFN restricted common shares and KFN phantom shares into awards in respect of KKR common units, with the nur common shares underlying such converted awards to be adjusted on the same basis as KFN common shares. In addition, KFN officers and directors are entitled to continued exculpation, indemnification, expense advancement rights and insurance continued indemnification agreements and the merger agreement. These different interests are described under "Special Factors Interests of Executive Officers of KFN in the Merger" beginning on page 98 of this proxy statement

Risks Relating to the Merger and Ownership of KKR Common Units (S

KFN common shareholders should consider carefully all the risk factors, together with all of the other information included or by reference, in this proxy statement/prospectus before deciding how to vote. Risks relating to the merger and the ownership of K units are described in the section titled "Risk Factors" beginning on page 104 of this proxy statement/prospectus. Some of these risks are not limited to, those described in the section titled "Risk Factors" beginning on page 104 of this proxy statement/prospectus.

Because the exchange ratio is fixed, KFN common shareholders cannot be sure of the market value of the K units they will receive as merger consideration relative to the value of the KFN common shares the

KFN and KKR may be unable to obtain the regulatory clearances and approvals required to complete the merg required to comply with material restrictions or satisfy material

The merger agreement contains provisions that limit KFN's ability to pursue alternatives to the merger and circumstances, could require KFN to pay a termination payment of \$26,250,000 to Merger Sub or

Executive officers and directors of KFN have certain interests that are different from those of KFN common generally. See "Special Factors Interests of Directors and Executive Officers of KFN in the Merger" beginning this proxy statement.

KFN common shareholders will have a reduced ownership interest and will not have a voting interest in most the merger and will exercise materially less or no influence over n

KKR common units to be received by KFN common shareholders as a result of the merger have materially di than KFN com

KFN common shareholders are expected to recognize taxable income or gain for U.S. federal income tax purpos

KKR is an affiliate of KFN's external manager, which provides executive officers and other serv

Material U.S. Federal Income Tax Consequences of the Merger (Se

The merger will be a taxable transaction for U.S. federal income tax purposes. If you are a U.S. holder of KFN common shafederal income tax purposes, your receipt of KKR common units and cash in lieu of fractional units in exchange for your KFN common the merger generally will cause you to recognize gain or loss measured by the difference, if any, between (i) the sum of (a) the fair of any KKR common units received, (b) the amount of cash received and (c) your share of KFN's nonrecourse debt immediately merger and (ii) y

Table

tax basis in your KFN common shares. Any such gain or loss recognized generally will be treated as capital gain or loss and will capital gain or loss if your holding period for your KFN common shares exceeds one year. However, a portion of any such gain will ordinary income to the extent attributable to your allocable share of unrealized gain or loss in KFN's assets to the extent described in of the Code. If you are a Non-U.S. holder of KFN common shares, a portion of any gain recognized by you in the merger (v calculated in the same manner described above for a U.S. holder) may be treated for U.S. federal income tax purposes as effective income, and hence you may be subject to U.S. federal income tax on such portion. All holders of KFN common shares should cons tax advisor for a full understanding of how the merger will affect their taxes. See "Material U.S. Federal Income Tax Consequences" beginning on page 172 of this proxy statement/prospectus for further discussion of the U.S. federal income tax consequences are the properties of the U.S. federal income tax consequences.

Accounting Treatment of the Merger (Se

The merger will be accounted for by KKR using the acquisition method of accounting. Under this method of accounting, the put will be allocated to the fair value of the net assets acquired at the date of completion of the merger. The excess purchase price over to of the net assets acquired will be recorded

Listing of KKR Common Units; Delisting and Deregistration of KFN Common Shares (Se

KKR common units are currently listed on the NYSE under the ticker symbol "KKR." It is a condition to closing of the monotone KKR common units to be issued in the merger to KFN common shareholders be approved for listing on the NYSE, subject to office

KFN's common shares are currently listed on the NYSE under the ticker symbol "KFN." If the merger is completed, KFN's common shares are currently listed on the NYSE under the ticker symbol "KFN." If the merger is completed, KFN's common shares are currently listed on the NYSE and will be deregistered under the Example 1.

No Appraisal Rights (

Under Delaware law and pursuant to KFN's operating agreement, KFN common shareholders will not have appraisal rights i

Conditions to Consummation of the Merger (Se

KKR and KFN currently expect to complete the merger in the first half of 2014, subject to receipt of required KFN sharehol and regulatory approvals and clearances and subject to the satisfaction or waiver of the other conditions to the merger desc

As more fully described in this proxy statement/prospectus, each party's obligation to complete the merger depends on conditions being satisfied or, to the extent permitted by applicable law, waived, including the

the merger agreement must have been adopted by the affirmative vote of the holders of at least a majority of the KFN common shares entitled to vote thereon on the record date, including the holders of a majority of the outst common shares other than those KFN common shares held by KKR or any affili-

the waiting period applicable to the merger under the Hart-Scott-Rodino Antitrust Improvements Act of 1976, which is referred to in this proxy statement/prospectus as the HSR Act, must have been terminated or expression consents required under any other antitrust law must have been obtained or any applicable waiting period them have been terminate.

Table

there must be no outstanding judgment, injunction, order or decree of a competent U.S. federal or state g authority prohibiting or enjoining the completion of the merger or the other transactions contemplated b

the registration statement of which this proxy statement/prospectus forms a part must have been declared eff SEC and must not be subject to a stop order or similar restraining order by

the KKR common units to be issued in the merger must have been approved for listing on the NYSE, subjective notices

The obligations of each of KKR, Fund Holdings and Merger Sub to effect the merger are subject to the satisfaction or value following additional following ad

the representations and warranties of KFN in the merger agreement being true and correct both when made and the date of the closing of the merger, subject to certain standards, including materiality and material a qualifications, as described under "The Merger Agreement Conditions to Consummation of the Merger page 134 of this proxy statement

KFN having performed in all material respects all obligations required to be performed by it under the merger or before

there not having occurred any events that, individually or in the aggregate, constitute a material adverse effect w

KFN since the date of the merger

the receipt of an officer's certificate executed by an executive officer of KFN certifying that the three precedir have be

the receipt of a payoff letter reasonably acceptable to KKR with respect to the termination of KFN's existing credit facility and any amounts outstanding thereunder (under which \$75.0 million of borrowings were outstanding the second control of the second control

December 31

receipt of a statement in accordance with Treasury Regulation Section 1.1445-11T(d)(2) certifying that 50% of value of the gross assets of KFN does not consist of U.S. real property interests, or that 90% or more of the value assets of KFN does not consist of U.S. real property interests plus cash or cash

The obligations of KFN to effect the merger are subject to the satisfaction or waiver of the following additional

the representations and warranties of KKR in the merger agreement being true and correct both when made and the date of the closing of the merger, subject to certain standards, including materiality and material adqualifications, as described under "The Merger Agreement Conditions to Consummation of the Merger page 134 of this proxy statement."

each of KKR, Fund Holdings and Merger Sub having performed in all material respects, all obligations reperformed by it under the merger agreement at or before

there not having occurred any events that, individually or in the aggregate, constitute a material adverse effect will KKR since the date of the merger agr

the receipt of an officer's certificate executed by an executive officer of KKR certifying that the three precedir

Table

Regulatory Approvals and Clearances Required for the Merger (Se

Consummation of the merger is subject to the expiration or termination of the waiting period under the HSR Act applicable to On February 20, 2014, KKR and KFN each filed a notification and report form under the HSR Act with the Antitrust Division of the of Justice, which is referred to in this proxy statement/prospectus as the Antitrust Division, and the Federal Trade Commission, which to in this proxy statement/prospectus as the FTC. On March 6, 2014, the FTC granted early termination of the waiting period under the See "Special Factors" Regulatory Approvals and Clearances Required for the Merger beginning on page 100 of this proxy statement.

No Solicitation by KFN of Alternative Proposals (So

Under the merger agreement, KFN has agreed that it will not, and it will use reasonable best efforts to cause its and its directors, officers, employees, agents, investment bankers, attorneys, accountants and other representatives not to, directly or indirect otherwise permitted by the merge

initiate or solicit or knowingly encourage any inquiries with respect to, or the making of, an acquisition

engage in any negotiations concerning, or provide any confidential information or data to any person racquisition acquisition or data to any person racquisition acquisition a

approve or recommend, or propose publicly to approve or recommend, any acquisition

approve or recommend, or propose publicly to approve or recommend, or execute or enter into, any let agreement in principle, merger agreement, acquisition agreement, option agreement or other similar agreeme any acquisition

propose publicly or agree to do any of the foregoing relating to any acquisit

In addition, the merger agreement requires KFN and its subsidiaries to (1) cease and cause to be terminated any existidiscussions or negotiations with any person conducted prior to the date of the merger agreement with respect to an acquisition (2) request that each third party that executed a confidentiality agreement that relates to an acquisition proposal before the date of agreement return or destroy all confidential information furnished to the third party by KFN or on its behalf before the date of

Notwithstanding these restrictions, the merger agreement provides that, under specified circumstances at any time before KFN vote in favor of the adoption of the merger agreement, if KFN receives a written unsolicited bona fide acquisition proposal after the merger agreement that the board of directors of KFN has determined in good faith, after consultation with its outside legal counsel advisors (1) constitutes a superior proposal (as described below) or (2) could reasonably be expected to result in a superior proposal

furnish nonpublic information to a third party that makes an acquisition proposal, if, before furnishing the informations an executed confidentiality agreement with provisions no less restrictive to the third party with respect disclosure of nonpublic information than the confidentiality agreement in effect between KFN at

engage in discussions or negotiations with the third party with respect to the acquisit

KFN has also agreed in the merger agreement that it will promptly orally notify KKR of any request for information or a proposals or offers relating to an acquisition proposal indicating, in connection with the notice, the name of the person making inqu

<u>Table</u>

or offer and the material terms and conditions of any proposals or offers, and that it will provide to KKR written notice of any inquired or offer within 24 hours of the request or inquiry, proposal or offer and copies of any written or electronic correspondence to or from making an acquisition proposal. KFN is required to keep KKR informed orally, as soon as reasonably practicable, of the status of any proposal, including with respect to the status and terms of any proposal or offer and whether any proposal or offer has been verificated, and to provide to KKR written notice of any withdrawal or rejection and copies of any written proposals or requests for within 24 hours. KFN has also agreed to provide any information to KKR (not previously provided to KKR) that it provides to an pursuant to these provisions at substantially the same time it provides the information to the

Change in KFN Board Recommendation (Se

The merger agreement provides that the KFN board of directors will not, directly or indirectly, withdraw, modify or qualify, adverse to KKR, the KFN board of directors' recommendation that KFN's common shareholders adopt the merger agreement or ap or recommend, or publicly propose to approve, adopt or recommend, any alternative acquisiti

Notwithstanding the above, subject to certain procedural requirements and limitations as provided for in the merger as described under "The Merger Agreement Change in KFN Board Recommendation" beginning on page 138 of this proxy statement KFN receives a written unsolicited bona fide acquisition proposal or in response to an intervening event, the KFN board of directors change of recommendation under certain ci

Termination of the Merger Agreement (Se

KKR or KFN may terminate the merger agreement at any time prior to the closing, whether before or after KFN common share approved the merge

by mutual writ

if there is any law or regulation that makes completion of the merger illegal or otherwise prohibited, or if an injunction, order or decree of a competent U.S. federal or state governmental authority enjoining the parties from the merger is entered and has become final and nor

if the merger is not completed on or before Septemb

if KFN common shareholders do not adopt the merger agreement at the special meeting (including any ad postponement of the special

if the other party has materially breached any of its representations, warranties, covenants or agreements con merger agreement, or if any fact, circumstance, event, change, occurrence or effect has occurred, which be circumstance, event, change, occurrence or effect would result in the failure of certain closing conditions to be sa prior to September 16, 2014, and the breach or fact, circumstance, event, change, occurrence or effect is not capacured or is not cured by the earlier of (1) 30 business days after written notice is received by the party alleged to or with respect to which a fact, circumstance, event, change, occurrence or effect is alleged to have (2) September 16, 2014, and the breached any of its representations, warranties, covenants or agreements con merger agreements or effect has occurred, which be circumstance, event, change, occurrence or effect is not capacity.

In addition, KKR may terminate the merger agreement if, at any time before KFN common shareholders have adopte

a change of recommendation has

Table

an alternative acquisition proposal in respect of KFN is publicly announced or disclosed (or any person publicle an intention (whether or not conditional) to make an acquisition proposal) after the date of the merger agreement board of directors of KFN fails to affirm the recommendation in favor of the adoption of the merger agreement time periods required under the merger

In addition, KFN may terminate the merger agreement at any time before KFN common shareholders have adopted the merger agreement order to enter into a definitive written agreement with respect to a superior proposal provided that KFN has complied in all material is obligations under described under "The Merger Agreement No Solicitation by KFN of Alternative Proposals" and Agreement Change in KFN Board Recommendation" beginning on pages 139 and 140, respectively, of this proxy statement/prospectively applicable termination payment described the applicable termination payment described to the state of the state

Expenses and Termination Payments Relating to the Merger (Se

Generally, all fees and expenses incurred in connection with the merger will be the obligation of the respective party incurring sexpenses, except that (1) expenses incurred in connection with filing, printing and mailing of the registration statement of which statement/prospectus forms a part and this proxy statement/prospectus (including filing fees) will be shared equally by Fund Holdin and (2) KKR will be responsible for all filing fees under the HSR Act and other a

Following termination of the merger agreement under specified circumstances, KFN may be required to pay Merger Sub or (unless the payment obligation is waived by Merger Sub) a termination payment of \$26,250,000 or to reimburse KKR for its measurement of \$26,250,000 or to reimburse KKR for its measurement of \$26,250,000 or to reimburse KKR for its measurement of \$26,250,000 or to reimburse KKR for its measurement of \$26,250,000 or to reimburse KKR for its measurement of \$26,250,000 or to reimburse KKR for its measurement of \$26,250,000 or to reimburse KKR for its measurement of \$26,250,000 or to reimburse KKR for its measurement of \$26,250,000 or to reimburse KKR for its measurement of \$26,250,000 or to reimburse KKR for its measurement of \$26,250,000 or to reimburse KKR for its measurement of \$26,250,000 or to reimburse KKR for its measurement of \$26,250,000 or to reimburse KKR for its measurement of \$26,250,000 or to reimburse KKR for its measurement of \$26,250,000 or to reimburse KKR for its measurement of \$26,250,000 or to reimburse KKR for its measurement of \$26,250,000 or to reimburse KKR for its measurement of \$26,250,000 or to reimburse KKR for its measurement of \$26,250,000 or to reimburse for its measu

The parties agreed that the amount of KFN's third-party expenses accrued in the fourth quarter of 2013 in connection with the c by KFN of the KKR acquisition proposal would reduce the amount of management fees paid by KFN to a subsidiary of KFN management agreement in an amount equal to such third-party ex

Comparison of KKR Common Units and KFN Common Shares (Se

KFN common shareholders receiving KKR common units in the merger will have materially different rights once they becon KKR's common units due to differences between the governing documents of KFN and the governing documents of KKR. These didescribed in more detail under "Comparison of KKR Common Units and KFN Common Shares" beginning

Litigation Relating to the Merger (Se

Fifteen putative stockholder class action lawsuits, referred to in this proxy statement/prospectus as the merger lawsuits, were KFN and certain other defendants in connection with KFN's entering into the merger agreement. Five of the merger lawsuits wer Superior Court of California, County of San Francisco; one of the merger lawsuits was filed in the United States District Court for the Northern California; and nine of the merger lawsuits were filed in the Court of Chancery of the State of Delaware. The five California actions have been consolidated. The plaintiff in the federal action has moved for an order scheduling a preliminary injunction authorizing expedited discovery, but later withdrew his motion for expedited discovery; the defendants have not been served if action. Two of the Delaware court actions were voluntarily dismissed, and the remaining seven Delaware court actions were consolidated Delaware action.

<u>Table</u>

merger lawsuits was filed on behalf of a putative class of KFN common shareholders against KFN, the individual members of KF directors, KKR, Fund Holdings, and Merger Sub. The merger lawsuits allege variously that the members of the KFN board of direct their fiduciary duties owed to KFN common shareholders by approving the proposed merger for inadequate consideration; at transaction in order to obtain benefits not equally shared by other KFN common shareholders; entering into the merger agreement preclusive deal protection devices; failing to take steps to maximize the value to be paid to the KFN common shareholders; a disclose material information necessary for KFN common shareholders to make a fully informed decision about the proposed merger lawsuits also seek to state claims against KFN, KKR, Fund Holdings, and Merger Sub for aiding and abetting these alleger fiduciary duties. In addition, certain of the complaints allege that KKR controlled KFN by means of a management agreement between KKR Financial Advisors LLC, and that, as a consequence, KKR breached fiduciary duties it owed to KFN common shareholder KFN to approve the merger agreement. The relief sought by the plaintiffs in the merger lawsuits includes, among other things, decinjunctive relief concerning the alleged breaches of fiduciary duties, injunctive relief prohibiting the consummation of the acquisition an accounting by defendants, damages and attorneys' fees and costs, and

		1	a	U.

Organizat

The following diagram depicts a simplified organizational structure of KKR following

KKR Management LLC serves as the general partner of KKR, which is governed by a board of directors consisting of independent directors. KKR Management LLC does not hold any economic interests in KKR and is owned by senior KK

KKR Holdings is the holding vehicle through which KKR principals and other persons indirectly own their interests in KK through ownership of KKR Group Partnership units. As of March 18, 2014, KKR Holdings and KKR held 56.7% respectively, of the KKR Group Partnership units. KKR Group Partnership units held by KKR Holdings represent interes business that are not attributable to holders of KKR common units. KKR Group Partnership units that are held by KKR exchangeable for KKR common units on a one-for-one basis, subject to customary conversion rate adjustments for

distributions and reclassifications and compliance with applicable vesting and transfer restrictions. As limited partner in KKR Group Partnership units are non-voting and do not entitle KKR Holdings to participate in the management of KKR's

Table

KKR Holdings holds special non-economic voting units in KKR that entitle it to cast, with respect to those limited matters submitted to a vote of KKR's unitholders, a number of votes equal to the number of KKR Group Partnership units that

Because the income of KKR Management Holdings L.P. is likely to be primarily non-qualifying income for purposes of the income exception to the publicly traded partnership rules, KKR formed KKR Management Holdings Corp., which is subject as a corporation for U.S. federal income tax purposes, to hold its KKR Group Partnership units in KKR Management Holdings L.P. will be subject to corporate rate. KKR Management Holdings L.P., which is treated as a partnership for U.S. federal income tax purposes, we hold interests in KKR's fee generating businesses and other assets that may not generate qualifying income for pur qualifying income exception to the publicly traded partnership rules. Fund Holdings, which is also treated as a partner federal income tax purposes, was formed to hold interests in KKR's businesses and assets that will generate qualifying purposes of the qualifying income exception to the publicly traded partnership.

40% of the carried interest earned in relation to KKR's investment funds and carry paying co-investment vehicles is allocate pool from which carried interest is allocated to KKR principals, other professionals and selected other individuals who voperations, thereby reducing the amount of carried interest allocable to KKR Holdings and holders of KKR co

Selected Historical Consolidated Financial D

The following tables set forth KKR's selected historical consolidated and combined financial data as of and for the December 31, 2013, 2012, 2011, 2010 and 2009. The historical consolidated financial data as of December 31, 2013 and 2012 and for years ended December 31, 2013, 2012 and 2011 are derived from KKR's audited consolidated financial statements contained in K Report on Form 10-K for the year ended December 31, 2013, which has been incorporated by reference in this proxy statement/pro historical consolidated and combined financial data as of December 31, 2011, 2010 and 2009 and for the years ended December 2009 are derived from KKR's audited consolidated and combined financial statements that are not included or incorporated by refer proxy statement/prospectus. The following information should be read in conjunction with "Management's Discussion and Analysis Condition and Results of Operations" and the consolidated financial statements and notes thereto set forth in KKR's Annu Form 10-K for the year ended December 31, 2013 incorporated by reference in this proxy statement/prospectus. See "Where You Ca Information" beginning on page 234 of this proxy statement

		Year	En	ded Decembe	er 31	,		
2013		2012		2011		2010		2009(
	(\$	in thousands	s, ex	cept unit and	l per	unit data)		
\$ 762,546	\$	568,442	\$	723,620	\$	435,386	\$	33
1,767,138		1,598,788		1,214,005		1,762,663		1,19
8,896,746		9,101,995		1,456,116		9,179,108		7,75
7,892,154		8,071,649		965,731		7,851,831		6,88
37,926		43,405		89,245		75,360		3
7,854,228		8,028,244		876,486		7,776,471		6,85
62,255		34,963		4,318				
7,100,747		7,432,445		870,247		7,443,293		6,00
\$ 691,226	\$	560,836	\$	1,921	\$	333,178	\$	84
	\$ 762,546 1,767,138 8,896,746 7,892,154 37,926 7,854,228 62,255 7,100,747	\$ 762,546 \$ 1,767,138 8,896,746 7,892,154 37,926 7,854,228 62,255 7,100,747	2013 2012 (\$ in thousands) \$ 762,546 \$ 568,442 1,767,138 1,598,788 8,896,746 9,101,995 7,892,154 8,071,649 37,926 43,405 7,854,228 8,028,244 62,255 34,963 7,100,747 7,432,445	2013 2012 (\$ in thousands, ex \$ 762,546 \$ 568,442 \$ 1,767,138 1,598,788 8,896,746 9,101,995 7,892,154 8,071,649 37,926 43,405 7,854,228 8,028,244 62,255 34,963 7,100,747 7,432,445	2013 2012 2011 (\$ in thousands, except unit and the second	2013 2012 2011 (\$ in thousands, except unit and per \$ 762,546 \$ 568,442 \$ 723,620 \$ 1,767,138 1,598,788 1,214,005 8,896,746 9,101,995 1,456,116 7,892,154 8,071,649 965,731 37,926 43,405 89,245 7,854,228 8,028,244 876,486 62,255 34,963 4,318 7,100,747 7,432,445 870,247	\$ 762,546 \$ 568,442 \$ 723,620 \$ 435,386 1,767,138 1,598,788 1,214,005 1,762,663 8,896,746 9,101,995 1,456,116 9,179,108 7,892,154 8,071,649 965,731 7,851,831 37,926 43,405 89,245 75,360 7,854,228 8,028,244 876,486 7,776,471 62,255 34,963 4,318 7,100,747 7,432,445 870,247 7,443,293	2013 2012 2011 2010 (\$ in thousands, except unit and per unit data) \$ 762,546 \$ 568,442 \$ 723,620 \$ 435,386 \$ 1,767,138 1,598,788 1,214,005 1,762,663 8,896,746 9,101,995 1,456,116 9,179,108 7,892,154 8,071,649 965,731 7,851,831 37,926 43,405 89,245 75,360 7,854,228 8,028,244 876,486 7,776,471 62,255 34,963 4,318 7,100,747 7,432,445 870,247 7,443,293

<u>Table</u>

	Year Ended December 31,									Octob 2009 thi Decemb
		2013		2012		2011		2010	-	200
				(\$ in thousand	s. e	except unit and	pe	r unit data)		
Net Loss Attributable to KKR & Co. L.P.				(, , , , , , , , , , , , , , , , , , ,	, -		1.		\$	('
Distributions Declared per KKR & Co. L.P.	¢	1.40	Φ	1 22	¢.	0.74	ď	0.60	φ	
Common Unit Net Income (Loss) Attributable to	\$	1.40	Þ	1.22	Þ	0.74	Э	0.60	Э	
KKR & Co. L.P. Per Common Unit										
Basic	\$	2.51	\$	2.35	\$	0.01	\$	1.62	\$	
Diluted	\$	2.30	\$	2.21	\$	0.01	\$	1.62	\$	
Weighted Average Common Units Outstanding										
Basic		274,910,628		238,503,257		220,235,469		206,031,682		204,9
Diluted		300,254,090		254,093,160		222,519,174		206,039,244		204,9
Statement of Financial Condition Data (period										
end):										
Total Assets	\$	51,427,201	\$	44,426,353	\$	40,377,645	\$, ,	\$	30,22
Total Liabilities	\$	4,842,383	\$	3,020,899	\$	2,692,995	\$	2,391,115	\$	2,8
Redeemable Noncontrolling Interests	\$	627,807	\$	462,564	\$	275,507	\$		\$	
Noncontrolling Interests	\$	43,235,001	\$	38,938,531	\$	36,080,445	\$	34,673,549	\$	26,3
Total KKR & Co. L.P. Partners' Capital(3)	\$	2,722,010	\$	2,004,359	\$	1,328,698	\$	1,326,493	\$	1,0

The financial information reported for periods prior to October 1, 2009 does not give effect to the acquisition of all of the liabilities of KKR & Co. (Guernsey) L.P., formerly known as KKR Private Equity Investors L.P., by affiliates of KKR of 2009 and the related reorganization of KKR's business into a holding company structure in connection with such acquisition of such assets and liabilities and the related reorganization of KKR's business into a holding company structure to in this proxy statement/prospectus as the KPE

Subsequent to the KPE Transaction, net income (loss) attributable to KKR reflects only those amounts that are allocal interest in the business. Net income (loss) that is allocable to KKR Holdings' interest in the business is reflected in net in attributable to noncontroll

Total KKR partners' capital reflects only the portion of equity attributable to KKR. KKR Holdings' interest in the Partnerships is reflected as noncontrolling interests and is not included in the total KKR part

Selected Historical Consolidated Financial I

The following historical consolidated financial data as of December 31, 2013 and 2012 and for each of the years ended December 2012 and 2011 are derived from KFN's audited consolidated financial statements contained in KFN's Annual Report on Form 10-K ended December 31, 2013, which has been incorporated by reference in this proxy statement/prospectus. The following historical financial data as of December 31, 2011, 2010 and 2009 and for the years ended December 31, 2010 and 2009 are derived from K consolidated financial statements that are not included or incorporated by reference in this proxy statement/prospectus. You she following data in conjunction with "Management's Discussion and Analysis of Financial Condition and Results of Operation consolidated financial statements and the related notes thereto set forth in KFN's Annual Report on Form 10-K for the year ended December 31, 2013 incorporated by reference in this proxy statement/prospectus. See "Where You Can Find More Information" beginning on page proxy statement

	Year Ended December 31,									
		2013 2012				2011 2010				2009
				(\$ in thous	and	ls, except per	s, except per share data)			
Consolidated Statements of										
Operations Data										
Total revenues	\$	545,906	\$	555,473	\$	542,021	\$	505,359	\$	572,725
Total investment costs and expenses		305,705		318,375		215,162		188,952		329,169
Total other income (loss)		150,925		205,822		93,447		143,352		(96,275)
Total other expenses		97,429		98,157		94,223		87,993		70,061
Income before income taxes		293,697		344,763		326,083		371,766		77,220
Income tax expense (benefit)		467		(3,467)		8,011		702		284
Net income		293,230		348,230		318,072		371,064		76,936
Preferred share distributions		27,411								
Net income available to common										
shareholders		265,819		348,230		318,072		371,064		76,936
Distributions declared per common										
share	\$	0.90	\$	0.86	\$	0.67	\$	0.43	\$	0.05
Consolidated Balance Sheet Data										
Cash and cash equivalents	\$	157,167	\$	237,606	\$	392,154	\$	313,829	\$	97,086
Restricted cash and cash equivalents		350,385		896,396		399,620		571,425		342,706
Securities		573,312		533,520		922,603		932,823		803,258
Corporate loans, net		6,466,720		5,947,857		6,443,399		6,321,444		6,543,643
Residential mortgage loans										2,097,699
Equity investments, at estimated fair										
value		181,212		161,621		189,845		99,955		120,269
Oil and gas properties, net		400,369		289,929		138,525		33,797		
Interests in joint ventures and										
partnerships		436,241		149,534						
Total assets		8,717,198		8,358,879		8,647,228		8,418,412		10,300,005
Total borrowings		6,020,465		6,338,407		6,778,208		6,642,455		8,970,591
Total liabilities		6,189,464		6,519,757		6,971,396		6,775,364		9,133,347
Total shareholders' equity		2,527,734		1,839,122		1,675,832		1,643,048		1,166,658
Book value per common share	\$	10.58	\$	10.31	\$	9.41	\$	9.24	\$	7.37

Ratio of Earnings to Fixed Charges and Ratio of Earnings to Fixed Charges and Preferred Share Distribut

The following table presents KFN's ratio of earnings to fixed charges and ratio of earnings to fixed charges and preferred share for the fiscal periods indicated. For purposes of determining the ratio of earnings to fixed charges, earnings are defined as earnings be taxes and fixed charges. Fixed charges consist of interesting the ratio of earnings to fixed charges.

		Year Ended December 31,				
		2013	2012			
	(5	in thousands	, exce	ept ratios)		
Earnings:						
Income before income taxes	\$	293,697	\$	348,230		
Add: Fixed charges from below		190,159		216,608		
Total earnings before income taxes and fixed charges	\$	483,856	\$	564,838		
Fixed charges:						
Interest expense	\$	190,159	\$	216,608		
Total fixed charges	\$	190,159	\$	216,608		
Ratio of earnings to fixed charges		2.5x		2.6x		
Total fixed charges	\$	190,159	\$	216,608		
Preferred share distributions		27,411				
Total fixed charges and preferred share distributions	\$	217,570	\$	216,608		
Ratio of earnings to fixed charges and preferred share distributions		2.2x		2.6x		

Unaudited Comparative Per Unit

The following table summarizes unaudited per common unit/share data for (i) KKR and KFN on a historical basis for the December 31, 2013 and 2012, (ii) KKR on a pro forma combined basis giving effect to the proposed transactions (collectively refer proxy statement/prospectus as the pro forma events) and (iii) KFN on a pro forma equivalent basis based on the exchange ratio of common units for each KFN common share. It has been assumed for purposes of the unaudited pro forma condensed combininformation provided below that the pro forma events occurred on January 1, 2012 for earnings per common unit purposes and on E 2013 for book value per common unit purposes. The historical earnings per common unit/share information should be read in conjuction the historical consolidated financial statements and notes thereto of KKR and KFN incorporated by reference in statement/prospectus. See "Where You Can Find More Information" on page 234 of this proxy statement/prospectus. The unaudite combined earnings per share information is derived from, and should be read in conjunction with, the section entitled "Unaudite Condensed Combined Financial Statements" and related notes included in this proxy statement/prospectus beginning on page 151. The information is presented for illustrative purposes only and is not necessarily indicative of the operating results or financial position have occurred if the proxy of the operating results or financial position have occurred if the proxy of the operating results or financial position have occurred if the proxy of the operating results or financial position have occurred if the proxy of the operating results or financial position have occurred if the proxy of the operating results or financial position have occurred if the proxy of the operating results or financial position have occurred if the proxy of the operating results or financial position have occurred if the proxy of the operating results or financial position have occurred in the proxy of the operating results of the proxy of t

<u>Table</u>

events had occurred as of the beginning of the periods presented, nor is it necessarily indicative of the future operating results position of the combin-

	As of / For the Year Ended December 31, 2013	As of / For the Year Ended December 31, 2012							
KKR Historical per Common Unit Data:									
Net income (loss) attributable to KKR & Co. L.P. Basic	\$ 2.51	\$ 2.35							
Net income (loss) attributable to KKR & Co. L.P. Diluted	2.30	2.21							
Book Value(1)	9.45	7.91							
KFN Historical per Common Share Data:									
Net income (loss) Basic	1.31	1.95							
Net income (loss) Diluted	1.31	1.87							
Book Value(1)	10.58	10.31							
Unaudited Pro Forma Combined per KKR Common Unit Data:									
Net income (loss) attributable to KKR & Co. L.P. Basic(2)	2.48	2.49							
Net income (loss) attributable to KKR & Co. L.P. Diluted(2)	2.32	2.38							
Book Value(1)	13.70	N/A							
Unaudited Pro forma Equivalent per KFN Common Share Data:									
Net income (loss) Basic(3)	1.26	1.27							
Net income (loss) Diluted(3)	1.18	1.21							
Book Value(1)(3)	6.99	N/A							

Historical book value per common unit/share is computed by dividing total partners' capital/common shareholders' number of KKR common units or KFN common shares, as applicable, outstanding as of December 31, 2013 and 201 combined book value per common unit is computed by dividing pro forma partners' capital by the pro forma number of K units that would have been outstanding as of December 31, 2013. Pro forma book value per common unit/share as of December 31, 2012 is not meaningful as the estimated pro forma adjustments were calculated as of December 31, 2013.

The pro forma net income (loss) per common unit of the combined company are calculated by dividing the pro forma income the pro forma weighted average number of common units

The pro forma equivalent share amounts were calculated by multiplying the pro forma combined per share amounts by t ratio in the merger (0.51 KKR common units for each KFN common share). This information shows how each KFN common would have participated in the combined company's net income (loss) and book value if the pro forma events had occurrence.

Comparative Unit Prices and D

KKR common units are currently listed on the NYSE under the ticker symbol "KKR." KFN common shares are currently NYSE under the ticker symbol "KFN." The table below sets forth, for the calendar quarters indicated, the high and low sale pricommon unit on the NYSE and per KFN common share on the NYSE. The table also shows the amount of cash distributions declar common units and KFN common shares, respectively, in the calendar quarters indicated. The amount of cash distributions declar common units and KFN common shares in any calendar quarter shown in the table below relate to the earnings of KKR and KFN, for the immediately preceding cale

	KK	CR Common		KFN Common Shares						
	High	Low	Cash Distributions	High	Low	Cash Distributions				
2014 (through March 20, 2014)	Ingi	Low	Distributions	IIIgii	Low	Distributions				
First quarter	\$ 26.50	\$ 22.75	\$ 0.48	\$ 13.38	\$ 11.51	\$ 0.22				
2013										
Fourth quarter	25.87	19.68	0.23	12.39	8.91	0.22				
Third quarter	21.78	18.74	0.42	11.31	10.02	0.21				
Second quarter	21.60	17.27	0.27	11.30	10.05	0.21				
First quarter	20.00	15.38	0.70	11.93	10.74	0.26				
2012										
Fourth quarter	15.49	13.35	0.24	10.89	9.30	0.21				
Third quarter	15.68	12.74	0.13	10.36	8.53	0.21				
Second quarter	15.50	11.03	0.15	9.47	7.95	0.18				
First quarter	15.20	12.74	0.32	9.54	8.70	0.26				
2011										
Fourth quarter	14.58	8.95	0.10	8.85	6.68	0.18				
Third quarter	16.70	10.07	0.11	9.92	7.36	0.18				
Second quarter	19.16	15.10	0.21	10.35	9.43	0.16				
First quarter	18.29	14.09	0.29	10.60	8.80	0.15				

The following table presents per unit closing prices for KKR common units and KFN common shares on December 16, 2 trading day before the public announcement of the merger agreement, and on March 20, 2014, the last practicable trading day before this proxy statement/prospectus. This table also presents the equivalent market value per KFN common shares on such dates. The market value per KFN common share has been determined by multiplying the closing prices of KKR common units on those exchange ratio of 0.51 of a KKR common with the common share the common share as the common units on those exchange ratio of 0.51 of a KKR common units on the common share the common units and KFN common shares on December 16, 22 trading the common shares on Such dates. The common shares on Such dates are common shares on Such dates. The common shares on Such dates are common shares on Such dates are common shares on Such dates.

					Equ	ivalent Market
	K	KR		KFN	Va	lue per KFN
	Comn	on Units	Com	mon Shares	Co	mmon Share
December 16, 2013	\$	25.08	\$	9.45	\$	12.79
March 20, 2014	\$	23.46	\$	11.86	\$	11.96

Although the exchange ratio is fixed, the market prices of KKR common units and KFN common shares will fluctuate consummation of the merger and the market value of the merger consideration ultimately received by KFN common shareholders with the closing price of KKR common units on the day the merger is consummated. Thus, KFN common shareholders will not know market value of the merger consideration they will receive until the closing or

<u>Table</u>

SPECIAL FACTORS

This section of the proxy statement/prospectus describes the material aspects of the merger and certain special factors co merger of which you should be aware. This section may not contain all of the information that is important to you. You should careful entire proxy statement/prospectus and the documents incorporated herein by reference, including the full text of the merger agreement attached as Annex A), for a more complete understanding of the merger. In addition, important business and financial information at KKR and KFN is included in or incorporated into this proxy statement/prospectus by reference. See "Where You Can Find More and beginning on page 234 of this proxy statement."

Effect of

Subject to the terms and conditions of the merger agreement and in accordance with Delaware law, the merger agreement promerger of Merger Sub with and into KFN. KFN will survive the merger and the separate existence of Merger Sub will cease. For merger, KFN will be a direct subsidiary of Fund Holdings. After the completion of the merger, the certificate of formation of operating agreement of KFN as in effect immediately prior to the effective time will be the certificate of formation and operating at the surviving entity, in each case until amended in accordance with applicable law. The officers of KFN immediately prior to the effective time will be the officers of the surviving entity, and the members of the board of directors of the surviving entity individuals designed by KKR prior to the closing date, until their respective successors are duly elected, appointed or qualified earlier death, resignation or removal in accordance with the operating agreement of the surviving entity and ap

The merger agreement provides that, at the effective time of the merger, each KFN common share issued and outstanding immeto the effective time of the merger will be converted into the right to receive 0.51 KKR common units. Each KFN common share the Fund Holdings or any subsidiary of Fund Holdings immediately prior to the effective time of the merger will be cancelled conversion or payment of consideration in res

Because the exchange ratio was fixed at the time the merger agreement was executed and because the market value of KKR co and KFN common shares will fluctuate prior to the consummation of the merger, KFN common shareholders cannot be sure of the merger consideration they will receive relative to the value of the KFN common shares they are exchanging. For example, dec market value of KKR common units will negatively affect the value of the merger consideration that they receive, and increases it value of KFN common shares may mean that the merger consideration that they receive will be worth less than the market value common shares such shareholders are exchanging. See "Risk Factors Risk Factors Relating to the Merger Because the exchange representation because the market price of KKR common units will fluctuate prior to the consummation of the merger, KFN common shareholders are exchanged to the KKR common units they will receive as merger consideration relative to the value of KFN common exchanged beginning on page 104 of this proxy statements.

KKR will not issue any fractional KKR common units in the merger. Instead, each holder of KFN common shares that a pursuant to the merger agreement who otherwise would have received a fraction of a KKR common unit will be entitled to received agent appointed by KKR pursuant to the merger agreement, a cash payment in lieu of such fractional KKR common units such holder's proportionate interest in the proceeds from the sale by the exchange agent of the number of excess KKR common units by the aggregate amount of fractional KKR common units.

Table

Each KFN option or similar right to purchase KFN common shares that was granted under a KFN equity incentive plate outstanding and unexercised immediately prior to the effective time (whether or not then vested or exercisable), as of the effective merger, by virtue of the occurrence of the consummation of the merger and without any action on the part of the holder of such KFN be cancelled, and the holder of such KFN option will be entitled to receive an amount in cash equal to the excess, if any, of (1) 0.51 r the average closing price of a KKR common unit on the NYSE over the ten trading day period ending on the trading day immediate the closing date of the merger over (2) the exercise price per KFN common share subject to such KFN option as in effect immediate the effective time of the merger. In the event that any option to purchase KFN common shares has an exercise price per KFN common is greater than 0.51 multiplied by the average closing price of a KKR common unit on the NYSE over the ten trading day period e trading day immediately preceding the closing date of the merger, such option will be cancelled without payment in respect there effective time of the merger, without liability to KKR, KFN or any of the

Each restricted KFN common share that was granted under a KFN equity incentive plan and that is outstanding immediately effective time of the merger (whether or not then vested), other than vested and unvested KFN common shares held by KF Advisors LLC, as of the effective time of the merger, by virtue of the occurrence of the consummation of the merger and without at the part of the holder of such KFN restricted common share, will be converted into 0.51 restricted KKR common units having the and conditions, including applicable vesting requirements, as applied to such restricted KFN common share immediately prior to time of the merger, with fractional KKR common units to be aggregated and rounded to the nearest whole unit, and KKR will assume equity incentive plan and any award agreement issued thereunder pursuant to which any such restricted KFN common share has been order to provide for the foregoing. Each restricted KFN common share held by KFN Financial Advisors as of the effective time of will automatically be cancelled.

Each KFN phantom share that was granted under KFN's Non-Employee Directors' Deferred Compensation and Share Award converted into a phantom share in respect of 0.51 KKR common units and will otherwise remain subject to the term

The table below sets forth the direct and indirect interests of the KKR Participants and KKR Holdings in the net book value and of KFN prior to and immediately after the merger, based upon the net book value of KFN at December 31, 2013 and net income of year ended December 31, 2013 and net income of the control of the c

Prior to the Merger(1)

	Interest in KFN's Net Book \$ (in	-	Interest in KFN's Net Ear \$ (in	_	Interest in KFN's Net Earnings Net of Preferred Share Distributions \$ (in		
Name	thousands)	%	thousands)	%	thousands)	%	
KKR(1)	824	0.03	96	0.03	87	0.03	
Fund Holdings	0	0	0	0	0	0	
Merger Sub	0	0	0	0	0	0	
KKR Management LLC	0	0	0	0	0	0	
KKR Group Holdings L.P.	824	0.03	96	0.03	87	0.03	
KKR Fund Holdings GP							
Limited	0	0	0	0	0	0	
KKR Group Limited	0	0	0	0	0	0	
KKR Holdings(1)	1,079	0.04	125	0.04	113	0.04	
		29					

Immediately after the Merger(2)

	Interest in KFN's Net Book	Value	Interest in KFN's Net Ear		Interest in KFN's Net Ear. Net of Preferred Sh Distribution	nings are
	\$ (in		\$ (in		\$ (in	
Name	thousands)	%	thousands)	%	thousands)	%
KKR(2)	1,282,009	50.7	148,720	50.7	134,817	50.7
Fund Holdings(3)	1,282,009	50.7	148,720	50.7	134,817	50.7
Merger Sub	N/A	N/A	N/A	N/A	N/A	N/A
KKR Management LLC	0	0	0	0	0	0
KKR Group Holdings L.P.	1,282,009	50.7	148,720	50.7	134,817	50.7
KKR Fund Holdings GP						
Limited	0	0	0	0	0	0
KKR Group Limited	0	0	0	0	0	0
KKR Holdings(2)	1,245,725	49.3	144,510	49.3	131,002	49.3

As of March 18, 2014, KKR Holdings and KKR have an indirect interest in approximately 56.7% and 43.3%, respect 65,011 unvested restricted KFN common shares and 89,150 vested KFN common shares held by KKR Financial Advisors is an indirect wholly-owned subsidiary of KKR Management Holdings L.P., one of the KKR Group

Based on the number of KKR Group Partnership units held as of March 18, 2014 and giving effect to only the antic common units to be issued in connection with the merger, KKR Holdings and KKR would hold approximately 49.3% respectively, of the KKR Group Partnership.

Following the consummation of the merger, all of the outstanding KFN common shares will be held by Fund Holding KKR Group I

Background of

The KFN board of directors regularly reviews and discusses at board meetings KFN's performance, risks, opportunities and strategies board of directors and management team review and evaluate various options as part of KFN's ongoing efforts to strengthen its enhance shareholder value, taking into account economic, regulatory, competitive and other conditions and KFN's status as a management team review and evaluate various options as part of KFN's ongoing efforts to strengthen its enhance shareholder value, taking into account economic, regulatory, competitive and other conditions and KFN's status as a management team review and evaluate various options as part of KFN's ongoing efforts to strengthen its enhance shareholder value, taking into account economic, regulatory, competitive and other conditions and KFN's status as a management team.

KFN is externally managed and advised by KKR Financial Advisors LLC, which is referred to in this proxy statement/pros manager, pursuant to a management agreement between KFN and the manager, which has been in place since before the KFN offering. The manager is a subsidiary of KKR Asset Management LLC, and an indirect subsidiary of Kohlberg Kravis Roberts & Co is an indirect subsidiary of KKR. Pursuant to the management agreement, the manager is responsible for KFN's operations and services relating to the management of KFN's assets, liabilities and operations, and provides KFN with its management te appropriate support personnel. All of KFN's executive officers are employees of one or more subsidiaries of KKR. The manager of the direction of the KFN board of directors, and is required to manage KFN's business affairs in conformity with the investment grave approved by a majority of KFN's independent directors. Due to the relationship created by the management agreement, members board of directors and the KFN management team routinely interact with representatives of KKR in the course of their normal of

Table

In October 2013, Henry Kravis and George Roberts, the co-chief executive officers of KKR's managing partner, informed Pauchairman of the board of directors of KFN, through Craig Farr, the CEO of KFN, that KKR was considering making an offer for a of KFN by KKR, but did not make any specific proposal or discuss any transaction terms. Mr. Hazen informed KKR through Mr would report the matter to the KFN board of directors and that pending instruction from the board neither KKR nor the manager sho discussions with representatives of KFN regarding any potential transaction unless and until the board of KFN authorized the

On October 22, 2013, the KFN board of directors held a regularly scheduled meeting in New York. At this meeting, Mr. Haz the other members of the KFN board of directors of the statements made by KKR regarding the possibility of making an offer to a The board discussed the possibility of such an offer, including that the confidentiality restrictions in the management agreement cour require that KKR obtain authorization from KFN prior to completing the analysis necessary to make any acquisition proposal. After so that the board could become fully informed regarding its options to maximize value for KFN's common shareholders, the board germission to use the information it had about KFN, as well as to access additional information about KFN, to make an acquisition poard also resolved that if there were an acquisition proposal made by KKR, it would be reviewed by a transaction committee of the would make a recommendation to the full board of directors and which would have the full authority of the board to negotiate the acquisition proposal with KKR, reject any such proposal, consider whether KFN should pursue any alternative transaction professional advisors that such transaction committee may deem necessary or appropriate. The board also discussed retaining potent advisors. Scott Nuttall and Mr. Farr, the two members of the board of directors who were employed by KKR, were not present for the Mr. Nuttall and Mr. Farr did not attend any of the KFN board of directors meetings at which a potential transaction with KKR was

Following the October 22, 2013 board meeting, Mr. Hazen instructed Wachtell, Lipton, Rosen & Katz, which is referred to statement/prospectus as Wachtell Lipton, to inform David Sorkin, the general counsel of KKR's managing partner, that the K directors had agreed to permit KKR to use the information it had about KFN, as well as to access additional information about KF proposal to a

Additionally, prior to the formation of the transaction committee, Mr. Hazen discussed with Mr. Farr the possibility of a eliminating the termination fee in the management agreement if there were a change of control of KFN. Mr. Farr discussed with Mr. Robert Lewin, the head of corporate development and treasurer of the KKR Group, and was informed that KKR was unwilling eliminate the termination fee if there were a change of control of the KKR was unwilling that the termination fee if there were a change of control of the KKR was unwilling that the termination fee if there were a change of control of the KKR was unwilling that the termination fee if there were a change of control of the KKR was unwilling that the termination fee if there were a change of control of the KKR was unwilling that the termination fee if there were a change of control of the KKR was unwilling that the termination fee if there were a change of control of the KKR was unwilling that the termination fee if there were a change of control of the KKR was unwilling that the termination fee if there were a change of control of the KKR was unwilling that the termination fee if there were a change of control of the KKR was unwilling that the termination fee if there were a change of control of the KKR was unwilling that the termination fee if there were a change of control of the KKR was unwilling the termination fee if the termination f

On October 30, 2013, KKR submitted a letter to KFN making a proposal to acquire KFN for consideration consisting enticommon units, at an exchange ratio of 0.46 KKR common units per KFN common share, in a taxable transaction. The Octoconditioned KKR's offer on the approval of the transaction by a committee of independent directors of KFN and the approval of a unaffiliated KFN s

On October 31, 2013, the KFN board of directors held a telephonic meeting attended by the directors who were not employ Pursuant to the resolutions adopted on October 22, 2013, the board designated a transaction committee of independent directors with KKR and instructed them to select a time to meet, retain advisors and review the proposal. The transaction committee we comprised of Tracy Collins, Robert Edwards, Vincent Paul Finigan, Ross Kari, Deborah McAneny and

<u>Table</u>

On November 6, 2013, the transaction committee of the KFN board of directors held a telephonic meeting with Robert Edwa chair, with Sandler O'Neill as financial advisors and Wachtell Lipton as legal advisors. The transaction committee selected Sandler O as financial advisors upon the collective recommendation of the members of the transaction committee, who were familiar with Sa from Sandler O'Neill's prior engagement on behalf of the independent members of the KFN board of directors (other than Mr. Mr. Ryles who had not yet been appointed to the KFN board of directors at the time of such prior engagement), due to their of transactions of this type involving financial companies and the absence of any conflicts with respect to either KKR or KFN. At the transaction committee discussed the terms of the acquisition proposal made by KKR, and determined to engage in further discussion of the proposal prior to responding to KKR. The transaction committee also reviewed with its advisors the existing terms of the agreement between KFN and the manager. The transaction committee and its advisors discussed the fact that the management agreement not be terminated unless KFN either had "cause" to do so (limited to circumstances involving the manager's continued material management agreement, fraud or gross negligence or a bankruptcy or change of control of the manager), or by the vote of a major shareholders or independent directors finding that either the manager's performance under the management agreement was unsat way that has been materially detrimental to KFN or that the fees payable to the manager are not fair (and, in such case, the man given the opportunity to agree to a reduced fee prior to any termination). The transaction committee and its advisors also discussed in connection with any termination of the management agreement other than for "cause," KFN would be required to pay a termination manager equal to approximately four times the sum of the average annual base management fee and the average annual incentive case calculated for the two twelve-month periods immediately prior to the termination. The transaction committee and its advise whether a potential buyer for KFN would be interested in buying KFN subject to the termination provisions and other terms of the agreement. The transaction committee with its advisors concluded that the foregoing made it highly unlikely that an acquisition of K of interest to any person other than KKR. The transaction committee also considered that any acquisition agreement between KI would likely include provisions allowing an interested third party to submit a superior proposal. On consideration of the precedin transaction committee determined not to solicit alternative

On November 12, 2013, the transaction committee of the KFN board of directors met. At the request of the transaction representatives of Wachtell Lipton and Sandler O'Neill were in attendance. At the meeting, the transaction committee discussed the acquisition proposal made by KKR with its advisors and determined to inform KKR that the transaction committee believed exchange ratio of 0.46 KKR common units per KFN common share did not offer sufficient value to KFN's shareholders, that the committee was not interested in pursuing a transaction at this value, and that the transaction committee's preference in any transact merger consideration consisting entirely of cash. The transaction committee's preference for cash merger consideration was duinherent uncertainty in the value of merger consideration consisting of equity securities of another entity and the fact that an acquisity KKR would be taxable to KFN common shareholders regardless of the form of merger consideration paid, but would not provide cash that could be used to pay such taxes if the merger consideration consisted entirely of KKR co

On or about November 13, 2013, each of Mr. Sorkin and Mr. Lewin were separately informed of the transaction committee's de and Mr. Sorkin and Mr. Lewin each responded that KKR would consider the transaction committee's response. On November Mr. Sorkin informed Wachtell Lipton and Mr. Lewin informed a representative of Sandler O'Neill that KKR was unwilling to consider the transaction committee's response.

Table

On November 19, 2013, a representative of Sandler O'Neill informed Mr. Lewin that a diligence session scheduled for that of cancelled unless KKR increased the proposed exchange ratio. That same day, Mr. Lewin contacted representatives of Sandle communicate a revised proposal for KKR to acquire KFN for consideration consisting entirely of KKR common units, at an exch 0.48 KKR common units per KFN common share, in a taxable transaction. Mr. Lewin informed the Sandler O'Neill representative was unwilling to consider a transaction in which any portion of the consideration in

On November 19, 2013, at the request of the transaction committee, representatives of Sandler O'Neill met with representatives including Mr. Sorkin, Mr. Lewin and William Janetschek, the chief financial officer of KKR's managing partner, and representatives Sachs in order to discuss the potential transaction. At this meeting, the representatives of KKR presented the Sandler O'Neill representatives at the potential transaction, which included the expectation that a transaction would increase the potential transaction, which included the expectation that a transaction would increase the potential transaction, which included the expectation that a transaction would increase the potential transaction, and allow KKR to redeploy KFN's capital over time into higher returning

On November 21, 2013, the transaction committee of the KFN board of directors held a telephonic meeting. At the representatives of Wachtell Lipton and Sandler O'Neill were in attendance. Mr. Hazen, who was not a metransaction committee, also attended a portion of the meeting, in order to update the members of the transaction committee on KF performance, including KFN operating as a stand-alone entity and the potential ability of KFN to pursue a share repurchase program that a transaction with KKR was not agreed. Following Mr. Hazen's departure from the meeting, the transaction committee discussed the revised acquisition proposal made by KKR with its advisors, including a comparison of the terms of such proposal to alternative share repurchase by KFN. The transaction committee determined to inform KKR that the transaction committee believed that the exchange ratio of 0.48 KKR common units per KFN common share still did not offer sufficient value to KFN's common sharehold the transaction committee continued to view a transaction including cash consideration as preferable to an all-equity

On or about November 22, 2013, a representative of Sandler O'Neill informed Mr. Lewin of the transaction committee's de Mr. Lewin informed the representative of the transaction committee that KKR would consider the transaction committee

On November 24, 2013, at the request of the transaction committee, Sandler O'Neill sent a presentation relating to the value representatives of KKR. The presentation was also sent to the members of the transaction committee. The presentation description reasons for an increase in the then-offered merger consideration, including the current relationship between the trading prices of K units and KFN common shares as compared to historical levels, factors that could currently be negatively affecting the trading price common shares, the expected future trading price of KFN common shares based on comparable companies and historical trading KFN's ability to potentially conduct a share repurchase, premiums paid in comparable transactions involving affiliated entities and the KKR of entering into the proposed merger as compared to potential other capital-raising alternatives that might be available.

On November 26, 2013, Mr. Kravis contacted a representative of Sandler O'Neill to communicate a revised proposal for KK KFN for consideration consisting entirely of KKR common units, at an exchange ratio of 0.50 KKR common units per KFN common taxable transaction. Mr. Kravis informed the representative of Sandler O'Neill that KKR remained unwilling to consider a transact the consideration included any portion of cash, and that the current proposed exchange ratio represented KKR's best and final off

Table

On November 27, 2013, the transaction committee of the KFN board of directors held a telephonic meeting. At the retransaction committee, representatives of Wachtell Lipton and Sandler O'Neill were in attendance. The transaction committee discussions the terms of the revised acquisition proposal made by KKR, including KKR's statement that the proposal represented KKR final offer to acquire KFN, and determined to meet again to further consider the appropriate response to KKR. The transaction through a representative of one of its advisors, requested that the transaction committee be given the opportunity to meet with members senior management in order to gather further information regarding KKR's business and the value of the KKR common units that we to be paid as merger consideration in connection with the potential

During this period, the parties and their advisors engaged in ongoing due diligence investigations of the business and prospec

On December 6, 2013, the transaction committee of the KFN board of directors held a telephonic meeting. At the request of the committee, representatives of Wachtell Lipton and Sandler O'Neill were in attendance. The representatives of Sandler O'Neill discuss members of the transaction committee KKR's responses to its diligence questions regarding KKR's business. In addition, a member O'Neill's equity research group attended the meeting in order to discuss KKR's business with the members of the transaction committee discussed the possibility that R. Glenn Hubbard, KFN's lead independent directorate the views of the transaction committee and the KFN board of directors on the proposed transaction, and specifically reiterate for an increase in the proposed merger consideration, and concluded that Mr. Hubbard should be asked to reinforce KFN's position in at a meeting scheduled with Mr. Kravis and Mr. Roberts on December 1.

Also on December 6, 2013, at the request of Wachtell Lipton, Simpson Thacher & Bartlett LLP, which is referred to statement/prospectus as Simpson Thacher, legal advisor to KKR, sent a draft merger agreement to Wac

On December 9, 2013, at the request of the transaction committee, Mr. Hubbard met with Mr. Kravis and Mr. Roberts in order t proposed transaction. At this meeting, Mr. Hubbard requested that KKR increase the proposed consideration above the most recer exchange ratio of 0.50 KKR common units per KFN common share. Mr. Kravis and Mr. Roberts informed Mr. Hubbard that KKR w

On December 10, 2013, both the KFN board of directors (other than the members of the board of directors that were employ and the transaction committee held meetings in New York. At the request of the transaction committee, representatives of Wachtel Sandler O'Neill were in attendance. During this meeting, the members of the board of directors were informed of the results of the conversations among Mr. Hubbard, Mr. Kravis and Mr. Roberts. At the request of the transaction committee, Mr. Kravis, Mr. Sorkin, Mr. Lewin and Mr. Janetschek, attended a portion of the board of directors meeting in order to discuss KKR's busin members of the board. In addition, at this meeting, Mr. Kravis and Mr. Roberts confirmed that in the event the KFN board of directors not to pursue the potential transaction with KKR, KKR did not anticipate any changes in KFN's relationship with KFN's managemanagement team or the KFN operating plan, and KKR would remain committed to facilitating KFN's success as a standalone Mr. Kravis, Mr. Roberts and other representatives of KKR left the meeting, the members of the transaction committee instructed representative to KKR that the transaction committee continued to believe the proposed consideration sufficient value to KFN's shareholders, and that the transaction committee would not be willing to recommend that the KFN board approve any transaction with KKR in which the merger consideration consisted of KKR common units without an increase in merger consideration with KKR in which the merger consideration consisted of KKR common units without an increase in merger consideration with KKR in which the merger consideration consisted of KKR common units without an increase in merger consideration with KKR in which the merger consideration consisted of KKR common units without an increase in merger consideration with KKR in which the merger consideration consisted of KKR common units without an increase in merger consideration with KKR.

Table

Following the meeting, representatives of Sandler O'Neill communicated the transaction committee's position to Mr. Lewin, a that KKR increase the proposed merger consideration. Subsequently, Mr. Kravis and Mr. Roberts informed Sandler O'Neill representation to 0.51 KKR common units per KFN common share, in a taxable transconsideration consisting entirely of KKR common units. Mr. Kravis and Mr. Roberts further stated that KKR would not under any confurther increase its offer. The Sandler O'Neill representatives communicated the revised proposal to Mr. Edwards and Mr. Edwards of schedule a meeting of the transaction committee to consider

Later in the day on December 10, 2013, Mr. Hazen, at Mr. Edwards' request, contacted Mr. Roberts to request that KKR proposed exchange ratio to 0.52 KKR common units per KFN common share, and was informed by Mr. Roberts that the current exchange ratio of 0.51 KKR common units per KFN common share represented KKR's best and final offer to a

On December 13, 2013, the KFN board of directors (other than the members of the board of directors who were employed by telephonic meeting in order to discuss the proposed transaction. At the request of the board, representatives of Wachtell Lipton O'Neill were in attendance. The representatives of Wachtell Lipton reviewed with the members of the board the material terms of th merger agreement with respect to the proposed transaction, including the required approval of KFN common shareholders (i approval of the holders of a majority of the outstanding KFN common shares, other than those KFN common shares held by KKR or of KKR) and the other closing conditions, the restrictions on the solicitation of alternative transactions by KFN and on changes in the recommendation, the merger consideration and treatment of outstanding KFN equity awards, the tax treatment of the merger, termi of the parties and potential termination payments payable by KFN, and restrictions on KFN's operations pending the closing of the meeting of the KFN board of directors was adjourned, and a meeting of the transaction committee was convened. At this meeting, re of Sandler O'Neill rendered an oral opinion to the transaction committee, which was subsequently confirmed by delivery of a wr dated December 16, 2013, to the effect that, as of such date, and subject to the assumptions, matters considered and limitations and c described in such opinion, the consideration to be exchanged in the proposed merger was fair to the holders of KFN common sl financial point of view. See " Opinion of the Financial Advisor to the KFN Transaction Committee" beginning on page 4: statement/prospectus. Following discussion, the members of the transaction committee unanimously voted to recommend that the K directors approve the merger agreement. Immediately following the conclusion of the transaction committee meeting, the board reconvened in order to consider the proposed transaction. At this meeting, the members of the board were informed of the recommen transaction committee. Following discussion, the members of the board, by the unanimous vote of all directors (other than the me board of directors who were employed by KKR and were not present at the meeting), approved and declared advisable the merger as

On December 15, 2013, the board of directors of KKR's managing partner held a meeting to discuss the potential acquisition of followed previous meetings of the board of directors of KKR's managing partner held on October 11, 2013, October 30, 2013, Nover and November 25, 2013 also in regard to such potential acquisition. At the December 15, 2013 meeting, representatives of KKR provided an update on the current status of the proposed transaction with KFN, including that the KFN board of directors approved at the proposed exchange ratio of 0.51 KKR common units per KFN common share, and also provided a presentation regarding, things, KFN's business and the potential financial implications of an acquisition of KFN, including a valuation analysis of KFN, a KFN's CLO portfolio, including the risks associated with such CLOs, and an accretion/dilution analysis of the

Table

financial effect of the proposed transaction on KKR. Also at this meeting, Mr. Sorkin advised the members of the board of directors in connection with the proposed transaction under Delaware law and under KKR's partnership agreement. Representatives. Thacher then reviewed with the board of directors of KKR's managing partner the findings of Simpson Thacher's legal due diligent. KFN and the material terms of the latest draft merger agreement with respect to the proposed transaction, including the merger consist treatment of outstanding KFN equity awards, the closing conditions, the termination rights of the parties and termination payment KFN, and restrictions on KKR's operations pending the closing of the merger. Representatives of Goldman, Sachs & Co., the financial KKR that is referred to in this proxy statement/prospectus as Goldman Sachs, were also present at this meeting and discussed with directors Goldman Sachs' financial analysis of the proposed transaction, which is summarized in the section entitled. Opinion of Advisor to KKR' beginning on page 61 of this proxy statement/prospectus. At this meeting, Mr. Kravis and Mr. Roberts provided approval under KKR's managing partner's limited liability company agreement of the issuance of KKR common units in connect proposed.

Also on December 15, 2013, a meeting of the conflicts committee of the board of directors of KKR's managing partner was followed previous meetings of the conflicts committee held on November 25, 2013, November 26, 2013 and December 9, 2013 to proposed transaction. During the December 15, 2013 meeting, representatives of Cravath Swaine & Moore LLP, legal counsel to committee that is referred to in this proxy statement/prospectus as Cravath, discussed the proposed acquisition of KFN with committee and reviewed the approval process with respect to the proposed transaction. The conflicts committee discussed the analysis presented by the representatives of KKR management and Goldman Sachs to the board of directors of KKR's management.

On December 16, 2013, the conflicts committee of the board of directors of KKR's managing partner held a meeti representatives of Lazard Frères & Co. LLC, the financial advisor to the conflicts committee that is referred to in this proxy statemer as Lazard, and Cravath were in attendance. At this meeting, representatives of Lazard rendered an oral opinion, subsequently writing, to the independent directors constituting the conflicts committee of the board of directors of KKR's managing partner, the date, and based upon and subject to the assumptions, procedures, factors, qualifications and limitations set forth therein, the expursuant to the merger agreement was fair, from a financial point of view, to KKR. See "Opinion of the Financial Advisor to Committee of KKR's Management was fair, from a financial point of view, to KKR.

Also on December 16, 2013, a meeting of the board of directors of KKR's managing partner was held. At the request of directors, representatives of Simpson Thacher, Cravath, Goldman Sachs, Lazard and KKR management were in attendance. At the Mr. Sorkin advised the board of directors that, subject to receipt of requisite board approval, KKR was in a position to execut agreement with KFN and announce the transaction. At this meeting, representatives of Goldman Sachs rendered an oral opinion to directors of KKR's managing partner, which was subsequently confirmed by delivery of a written opinion dated December 16, 2013, that, as of such date, and based upon and subject to the factors and assumptions set forth in such opinion, the exchange ratio pure merger agreement was fair from a financial point of view to KKR. See "Opinion of the Financial Advisor to KKR" beginning on proxy statement/prospectus. Following the receipt of the oral opinion of Goldman Sachs, by unanimous vote, the board of director managing partner approved and declared advisable the merger agreement and the merger. Immediately following the adjournment of the board of directors of KKR's managing partner, a meeting of the conflicts committee of the board of directors of KKR's managing partner, a meeting, the conflicts committee determined that the proposed acquisition of KFP accordance with the merger agreement

Table

reasonable to KKR and that such determination shall constitute a special approval as contemplated by KKR's partnershi

On December 16, 2013, following the foregoing events, the merger agreement was executed by KKR, Fund Holdings, Me KFN, and the parties issued a joint press release announcing the

Certain Relationships between KK

KKR Financial Advisors LLC, an indirect subsidiary of KKR and referred to in this proxy statement/prospectus as the manager, manages KFN's day to day operations (subject to the direction and oversight of the KFN board of directors) pursuant to a agreement between KFN and the manager, which is referred to in this proxy statement/prospectus as the management agreement. Pu management agreement, the manager provides KFN with its management team, along with appropriate support personnel. A executive officers and Scott Nuttall and Craig Farr, two of KFN's directors, are employees of KKR or one or more of its

The management agreement contains certain provisions requiring KFN to indemnify the manager with respect to all losses arising from acts not constituting bad faith, willful misconduct, or gross

For the years ended December 31, 2013, December 31, 2012 and December 31, 2011, KFN incurred \$39.1 million, \$28.2 \$26.3 million, respectively, in base management fees under the management agreement. A portion of the CLO management fees re affiliate of the manager for certain of KFN's CLOs were credited to KFN via an offset of \$10.0 million to the \$39.1 million in base fees incurred for the year ended December 31, 2013. Furthermore, during 2013, KFN invested in a transaction that generated placem to a minority-owned affiliate of KKR, and the manager agreed to reduce KFN's base management fee payable to the manager for the separate placement fees that were earned by KKR as a result of this minority-ownership. Separately, certain third party expenses accrue the fourth quarter of 2013 in connection with the merger were used to reduce KFN's base management fees payable to the manager is equal to such third-party expenses. In addition, KFN incurred share based compensation expenses of \$3.5 million, \$2.3 million and related to grants of restricted KFN common shares granted to the manager for the years ended December 31, 2013, December 31, 2011,

For the years ended December 31, 2013, December 31, 2012 and December 31, 2011, the manager earned \$22.7 million, \$37.6 \$34.2 million, respectively, of incentive fees under the management

Certain general and administrative expenses are incurred by the manager on KFN's behalf that are reimbursable to the manage the management agreement. For the years ended December 31, 2013, December 31, 2012 and December 31, 2011, KFN reimbursed for \$9.8 million, \$10.2 million and \$8.2 million, respectively, for su

The management agreement expires on December 31 of each year, but is automatically renewed for a one year term on each I unless terminated upon the affirmative vote of at least two thirds of KFN's independent directors, or by a vote of the holders of KFN's outstanding common shares, based upon (1) unsatisfactory performance by the manager that is materially detrimental to determination that the management fee payable to the manager is not fair, subject to the manager's right to prevent such a termi clause (2) by accepting a mutually acceptable reduction of management fees. The manager is required to be provided with 180 days' notice of any such termination and will be paid a termination fee by KFN equal to four times the sum of the average annual base man and the average annual services.

Table

for the two 12-month periods immediately preceding the date of termination, calculated as of the end of the most recently comquarter prior to the date of

In addition, KFN holds corporate loans, debt securities and other investments of entities that are affiliates of KKR. As of D 2013, the aggregate par amount of these affiliated instruments totaled \$2.1 billion, or approximately 27% of KFN's total investment and consisted of 28 issuers. The total \$2.1 billion in affiliated investments was comprised of \$1.9 billion of corporate loans, \$52 corporate debt securities and \$84.5 million of equity investments, at estimate

KFN holds interests in certain joint ventures and partnerships alongside affiliates of KKR. As of December 31, 2013, the against amount of these interests in joint ventures and partnerships totaled \$4

Recommendation of the KFN Board of Directors and Reasons for the Merger; Fairness of

After consideration, the transaction committee of the KFN board of directors and the KFN board of directors has determined that is fair to and in the best interests of KFN and the unaffiliated shareholders of KFN. In reaching its decision to recommend that KF directors approve the merger agreement and recommend that it be adopted by KFN's shareholders, the transaction committee of the of directors consulted with its legal and financial advisors and considered a number of factors, including, but not limited to, the

its knowledge of KFN's business, financial condition, results of operations, industry, competitors and p standalone company, including the anticipated negative impact on results of operations of amortizing CLOs a

the transaction committee's belief that the price of KFN common shares reflected these challenges, which had be previously in connection with KFN's prior earnings ann

its knowledge of KKR's business and investments, financial condition, results of operations and prospects account the results of the transaction committee's due diligence review of KKR and knowledge of KKR's manage the existing relationships between KFR

the financial terms of the merger, including the fact that, based on the closing price on the NYSE of KFN comm KKR common units on December 13, 2013, the last trading day prior to the execution of the merger agreemen consideration represented an approximate 34% premium over the closing price of KFN common shares as of s represented a significant premium above the 30 day volume-weighted average closing price based on the 30 period ending on December 13, 2013, the last trading day prior to the execution of the merger agreemen consideration represented an approximate 34% premium over the closing price of KFN common shares as of s represented a significant premium above the 30 day volume-weighted average closing price based on the 30 day volume-weighted average cl

the nature of KFN's business, which is generally valued at or near book value, and the substantial premium to be of September 30, 2013) of 16% implied by the exchange ratio based on the closing prices of KKR common ur common shares on Decemb

the fact that the transaction committee believed that, taking into account relative risk and potential upside, the memory provide an attractive alternative to KFN shareholders given alternatives reasonably available to KFN (included continued operation of KFN as a standalone entity and the anticipated dividend payments to KFN common share the transaction committee believed would be available in the future if KFN's operations continued as anticipated to be equal to \$0.88 annually through 2017, based on KFN's current distribution levels) and (ii) the repurchase of KFN's outstanding common shares with an estimated total repurchase price of \$200 million (based on the arm estimated to be reasonably available to KFN for such purpose), each of which the transaction

Table

believed would ultimately result in less value being created for KFN common shareholders than the prop

the fact that the transaction committee believed that, taking into account relative risk and potential upside, the noffer attractive value to KFN common shareholders relative to the value that could be realized through a liquidati assets and the distribution of the proceeds to KFN's creditors and equityholders, which the transaction commi would yield proceeds of less than KFN's book value, due in part to the negative effects that a liquidation we certain of KFN's assets such as KFN's CLO portfolio which would be anticipated to yield a lower amount distribution to KFN's common shareholders in a liquidation than could be produced over time, due to fa anticipated negative effects of sales of the underlying assets at a time when prices may not be favorable, liquidation the loss of ongoing interest.

KKR's stated intention to pass through 100% of its realized earnings from KFN's existing portfolio as a distriholders of KKR common units, while maintaining its existing policy of passing through 40% of its net realizinvestment income from KKR's existing portfolio, following the closing of the merger, and the attractive distributions and yield profile that would therefore be available to former KFN common shareholders following

the belief of the transaction committee that the consummation of the merger could lead to an increase in the transaction on the stability of KKR's balar quality of KKR's distributable earnings, thereby leading to an increase in the value of the KKR common units to KFN's common shareholders as merger co

the fact that KFN's common shareholders would have the benefit of a more diversified asset base as unitholders of would be available to them as shareholders of KFN, which would benefit KFN's common shareholders by prowith increased protection against declines in a single asset class or type of investment and through KKI increased flexibility to deploy KFN's capital (including capital that is expected to run off from KFN's existing time) in a manner designed to produce the highest returns in a variety of market

KKR's enhanced ability to raise funds and capital compared to KFN, and the corresponding benefits to KF shareholders as KKR common unitholders of a combined company with significantly enhanced growth of

KKR's ability to mitigate existing tax inefficiencies to KFN common shareholders produced by KFN's currer portfolio, corporate structure and distribut

the fact that the exchange ratio is fixed and therefore the value of the merger consideration payable to KFN's will increase in the event that the trading price of KKR common units increases prior

the financial analyses presented by Sandler O'Neill to the transaction committee described under "Opin O'Neill & Partners, L.P." and the oral opinion of Sandler O'Neill, confirmed by delivery of a written of December 16, 2013, to the effect that, as of such date, and subject to the assumptions, matters considered and lingualifications described in such opinion, the consideration to be exchanged in the proposed merger was fair to the KFN common shares, from a financial point.

the transaction committee's belief that it was unlikely that any bidder for KFN, other than KKR, would pursue as of KFN because of, among other reasons, KKR's existing role as manager of KFN, including its selection and of KFN's assets, the substantial premium to book value being offered by KKR for assets that are generally value

Table

book value and the provisions in the management agreement, including KFN's contractual obligation to pay a KKR a fee in the event KFN terminates the managemen

the fact that exchanging KFN common shares for KKR common units would offer KFN's common shareholders enhanced liquidity due to the trading volume and size of the public float of KKR as compared to the trading volume and size of the public float of KKR as compared to the trading volume and size of the public float of KKR as compared to the trading volume and size of the public float of KKR as compared to the trading volume and size of the public float of KKR as compared to the trading volume and size of the public float of KKR as compared to the trading volume and size of the public float of KKR as compared to the trading volume and size of the public float of KKR as compared to the trading volume and size of the public float of KKR as compared to the trading volume and size of the public float of KKR as compared to the trading volume and size of the public float of KKR as compared to the trading volume and size of the public float of KKR as compared to the trading volume and size of the public float of KKR as compared to the trading volume and size of the public float of KKR as compared to the trading volume and size of the public float of KKR as compared to the trading volume and size of the public float of KKR as compared to the trading volume and size of the public float of KKR as compared to the trading volume and size of the public float of KKR as compared to the trading volume and size of the public float of KKR as compared to the trading volume and size of the public float of KKR as compared to the trading volume and size of the public float of KKR as compared to the trading volume and size of the public float of KKR as compared to the trading volume and size of the trading volume and size of the public float of KKR as compared to the trading volume and size of the public float of KKR as compared to the trading volume and size of the public float of KKR as compared to the trading volume and size of the trading v

the transaction committee's belief that the proposed merger could be completed in a timely fashion and without delays due to required regulatory

the fact that the existing terms of the management agreement, pursuant to which KFN's business is manage KKR employees, would lead to minimal disruption in the continued management of KFN's existing assets an portfolio following the merger and would allow for the completion of the integration of KKR and KFN in efficient

the other terms and conditions of the merge

In addition, the transaction committee of the KFN board of directors also considered a number of factors relating to the safeguards involved in the negotiation of the merger, including those discussed below, each of which it believed supported and provided assurance of the fairness of the merger to the unaffiliated common sharehold.

the creation of a transaction committee of the KFN board of directors composed of directors not affiliated with F process followed by the transaction

the fact that the consummation of the merger is conditioned on obtaining the approval of both a majority of the KFN common shares and a majority of the outstanding KFN common shares held by shareholders other than

the fact that, subject to compliance with certain terms and conditions, including the termination fee of \$26,250, KFN is permitted to terminate the merger agreement in order to accept a superior proposal and enter into agreement in connectio

the fact that the transaction committee of the KFN board of directors made its evaluation of the merger agreed merger based upon the factors discussed in this proxy statement/prospectus, independent of management in knowledge of the interests of management in

the fact that the transaction committee of the KFN board of directors retained Sandler O'Neill to provide an o effect that, as of the date of such opinion, and subject to the assumptions, matters considered and lin qualifications described in such opinion, the consideration to be exchanged in the proposed merger was fair to the KFN common shares, from a financial po

the fact that the transaction committee of the KFN board of directors met numerous times during the course of with KKR to discuss the status of the negotiations with KKR, to review the terms of the proposed merger agree consider the options reasonably available to

the fact that the transaction committee and the KFN board of directors were aware of the existing relationships be and KFN and could take such relationships into account when considering whether to recommend the propose

on the contemplated ter

Table

The transaction committee of the KFN board of directors also considered a variety of risks and other potentially negative factor the merger agreement and the merger, including the

the fact that the exchange ratio is fixed and therefore the value of the merger consideration payable to K shareholders will decrease in the event that the trading price of KKR common units decreases prior

the possibility that the merger may not be completed, or that completion may be unduly delayed, for reason

the risks and costs to KFN if the transactions are not completed, including the potential effect on the trading pr common shares, the potential diversion of management attention, and the potential effect on KFN's business

the fact that KFN's directors and executive officers may ultimately have interests in the transactions that may from, or in addition to, those of other KFN sl

the restrictions imposed by the merger agreement on the conduct of KFN's business prior to the consummation o

the fact that the merger agreement contains provisions that limit KFN's ability to pursue alternatives to

the risk that governmental entities may oppose or refuse to approve the transactions or impose conditions on KFN prior to approving the t

the fact that at the time of the approval and execution of the merger agreement, KKR common units were trad near their 52-week high, while KFN common shares were trading at a price near their 52-week high.

the management relationship between KKR and KFN, including the fact that KFN's management team is proved by KKR and it

the fact that the transaction will be taxable to KFN common sl

the terms of the KKR common units to be issued as merger consideration and their differences from the terms common shares, including the inability to vote to elect the members of the board of directors of KKR's genera the significant control which may be exercised over any vote of KKR's common units by certain controlling pers

the risk of not realizing the anticipated benefits o

In reaching its decision to approve the merger agreement and recommend that it be adopted by KFN's common shareholders, the of directors considered a number of factors, including, but not limited to, the

the unanimous recommendation of the transaction committee of the KFN board of directors which was directors not affiliated

the fact that the transaction is subject to the approval of a majority of KFN stockholders not affiliated

the factors considered by the transaction committee, including the positive factors and potential benefits of the the risks and other potentially negative factors concerning the merger, as describe

the fact that the merger consideration and the other terms of the merger agreement resulted from negotiations t the active participation of the transaction

Table

In reaching its determination and making its recommendations, the transaction committee of the KFN board of directors did not specific going concern value of KFN and did not believe that there is a single method for determining going concern value. It transaction committee of the KFN board of directors believed that each of Sandler O'Neill's valuation methodologies described in the Financial Advisor to the KFN Transaction Committee" represented a valuation of KFN as it continues to operate its busines extent, such analyses could be collectively characterized as forms of going concern valuations and the transaction committee considerable these analyses in reaching its determination and making its recommendation. Further, the transaction committee adopted the conclusions of Sandler O'Neill. As stated above, the transaction committee of the KFN board of directors believed that it was another bidder would pursue an acquisition of KFN, and as a result did not seek to contact potential alternative bidders. As state transaction committee believed that the liquidation value to KFN's unaffiliated shareholders of KFN's assets would likely be lower than the value offered in the proposed merger, although no separate appraisal of was conducted because the transaction committee considered KFN to be a viable going concern and because KKR plans to continuation of KFN's existing businesses following the merger. The transaction committee was not aware of any firm offer made to KFN by any person during the past two years for the merger or consolidation of KFN into another company, the sale or transfer of all or any sulperson during the past two years for the merger or consolidation of KFN into another company, the sale or transfer of all or any sulperson during the past two years for the merger or consolidation of KFN into another company, the sale or transfer of all or any sulperson during the past two years for the merger or consolidation of KFN into another company, and no such fit considered as a factor in the transaction com

In light of the procedural safeguards described above, including the creation of a transaction committee comprised solely of affiliated with KKR and the retention by such committee of an independent financial advisor and independent legal counsel, the KFN committee did not consider it necessary to retain an unaffiliated representative to act solely on behalf of KFN shareholders (other any affiliate of KKR) for purposes of negotiating the terms of the merger agreement or preparing a report concerning the fairness of the merger agreement are agreement and agreement agreement and agreement agreement and agreement and agreement agreem

The KFN board of directors recommends that you vote "FOR" the adoption of the merger agreement and approval of the merge any adjournment of the special meeting, if necessary, to solicit additional proxies if there are not sufficient votes to adopt the merge at the time of the special meeting.

This discussion of the information and factors considered by each of the transaction committee of the KFN board of directors and/or the KFN board of directors, but is not intended to be exhaustive and may not include all of the factors considered by the committee of the KFN board of directors and/or the KFN board of directors, or any individual. Neither the transaction committee board of directors nor the KFN board of directors undertook to make any specific determination as to whether any particular factor, of any particular factor, was favorable or unfavorable to its ultimate determination, and did not quantify or assign any relative or spector the various factors that it considered in making its ultimate decision. Rather, each of the transaction committee of the KFN board and the KFN board of directors conducted an overall analysis of the factors described above. In addition, individual members of transaction committee of the KFN board of directors and the KFN board of directors may have given different weight to different weig

Opinion of the Financial Advisor to the KFN Transaction

By letter dated December 10, 2013, KFN retained Sandler O'Neill to act as financial advisor to the transaction committee of the of directors, referred in this proxy statement/pro

Table

the transaction committee, in connection with a possible business combination transaction. Sandler O'Neill is a nationally recognized banking firm whose principal business specialty is financial institutions. In the ordinary course of its investment banking business O'Neill is regularly engaged in the valuation of financial institutions and their securities in connection with mergers and acquisition corporate transactions. The transaction committee selected Sandler O'Neill to act as the transaction committee's advisor in connection based on its qualifications, expertise, reputation and experience in mergers and acquisitions involved.

Sandler O'Neill acted as financial advisor to the transaction committee in connection with the proposed transaction and pacertain of the negotiations leading to the execution of the merger agreement. At the December 13, 2013 meeting of the transaction Sandler O'Neill delivered to the transaction committee its oral opinion, which was subsequently confirmed in writing on December 16, 2013, the merger consideration was fair to the holders of KFN common shares from a financial point of view of Sandler O'Neill's opinion is attached as Annex B to this proxy statement/prospectus. The opinion outlines the followed, assumptions made, matters considered and qualifications and limitations on the review undertaken by Sandle rendering its opinion. The description of the opinion set forth below is qualified in its entirety by reference to the full text of Holders of KFN common shares are urged to read the entire opinion carefully in connection with their consideration of the

Sandler O'Neill's opinion speaks only as of the date of the opinion. The opinion was directed to KFN's transaction communicated only to the fairness of the merger consideration to the holders of KFN common shares from a financial point of view address the underlying business decision of KFN to engage in the merger or any other aspect of the merger and is not a record to any holder of KFN common shares as to how such holder of KFN common shares should vote at the special meeting with the merger or any other matter. Sandler O'Neill did not express any opinion as to the fairness of the amount or nature of the common be received in connection with the merger by KFN's officers, directors, or employees, or any class of such persons, relative to consideration to be received in the merger by any other shareholds.

In connection with rendering its opinion on December 16, 2013, Sandler O'Neill reviewed and considered, among

the merger

certain publicly available financial statements and other historical financial information of KFN that Sandler O'I

certain publicly available financial statements and other historical financial information of KKR that Sandler O'l

publicly available mean analyst earnings estimates for KFN for the years ending December 31, 2013 and December and publicly available mean analyst growth estimates for the years thereafter and in each case as discussed we managem

publicly available mean analyst economic net income estimates for KKR for the years ending December December 31, 2014 and publicly available mean analyst growth estimates for the years thereafter and in discussed with the senior management.

the pro forma financial impact of the merger on KKR, based on assumptions relating to transaction expens accounting adjustments and cost savings determined by the senior management of KFR

the publicly reported historical price and trading activity for KFN's and KKR's common shares or units, as the including a comparison of certain financial and

Table

information for KFN and KKR and similar publicly available information for certain other companies similar to and KKR, the securities of which are publicly available information for CFN and KKR and similar publicly available information for certain other companies similar to and KKR, the securities of which are publicly available information for CFN and KKR and similar publicly available information for certain other companies similar to a similar publicly available information for certain other companies similar to a similar publicly available information for certain other companies similar to a similar publicly available information for certain other companies similar to a similar publicly available information for certain other companies similar to a similar publicly available information for certain other companies similar to a similar publicly available information for certain other companies similar to a similar publicly available information for certain other companies of the securities of the secu

the financial terms of certain recent business combinations involving companies in the financial services indust buyer and the target had commercial relationships similar to those between KFN and KKR, to the extent public

certain of the terms of the management agreement, pursuant to which KKR Financial Advisors LLC, an indirect s KKR, ma

the current market environment generally and the financial services environment in pa

such other information, financial studies, analyses and investigations and financial, economic and market criter
O'Neill conside

Sandler O'Neill also discussed with certain members of the senior management of KFN the business, financial condition operations and prospects of KFN and held similar discussions with the senior management of KKR regarding the business, financial conditions are prospected from the prospection of KFN and held similar discussions with the senior management of KKR regarding the business, financial conditions are prospected from the prospection of KFN and held similar discussions with the senior management of KKR regarding the business, financial conditions are prospected from the prospected from the prospection of KFN and held similar discussions with the senior management of KKR regarding the business, financial conditions are prospected from the prospected

In performing its reviews and analyses and in rendering its opinion, Sandler O'Neill relied upon the accuracy and completeness financial and other information that was available to Sandler O'Neill from public sources, that was provided to Sandler O'Neill by Kenney or their respective representatives or that was otherwise reviewed by Sandler O'Neill and Sandler O'Neill assumed such a completeness for purposes of rendering its opinion. Sandler O'Neill further relied on the assurances of the senior management of and KKR that they were not aware of any facts or circumstances that would make any of such information inaccurate or misleading respect. Sandler O'Neill was not asked to undertake, and did not undertake, an independent verification of any of such information of Neill did not make an independent evaluation or appraisal of the specific assets, the collateral securing assets or the liabilities (continued to the specific assets) of KFN or KKR or any of their respective

Sandler O'Neill used mean publicly available earnings estimates for KFN and KKR and estimated long-term growth rates research estimates. Sandler O'Neill also received and used in its analyses certain projections of transaction costs, expected cost saving synergies which were provided by KKR's and KFN's respective management teams. With respect to those projections, estimates and the respective managements of KFN and KKR confirmed to us that those projections, estimates and judgments reflected the beavailable estimates and judgments of those respective managements of the future financial performance of KFN and KKR, resp. Sandler O'Neill assumed that such performance would be achieved. Sandler O'Neill expresses no opinion as to such estimates or the on which they are based. Sandler O'Neill assumed that there has been no material change in the respective assets, financial condition operations, business or prospects of KFN and KKR since the date of the most recent financial data made available to Sandler O'Neill also assumed in all respects material to its analysis that KFN and KKR would remain as a going concern for all the periods really analyses. Sandler O'Neill expresses no opinion as to any of the legal, accounting and tax matters relating to the merger and any other contemplated in connection.

Sandler O'Neill's opinion was necessarily based on financial, economic, regulatory, market and other conditions as in effect information made available to Sandler O'Neill as of, the date of its opinion. Events occurring after the date thereof could materia opinion. Sandler O'Neill has not undertaken to update, revise, reaffirm or withdraw its opinion or otherwise comment upon events octhe date of its opinion. Sandler O'Neill expressed no opinion as to the trading values at which the common shares or units, as the case of KFN or KKR

Table

any time or what the value of KKR units will be once it is actually received by the holders of KFN con

In rendering its December 16, 2013 opinion, Sandler O'Neill performed a variety of financial analyses. The following is a surmaterial analyses performed by Sandler O'Neill, but it is not a complete description of all the analyses underlying Sandler O'Neill's summary includes information presented in tabular format. In order to fully understand the financial analyses, these tables in together with the accompanying text. The tables alone do not constitute a complete description of the financial and information has not been prepared in accordance with U.S. GAAP and is unaudited. The preparation of a fairness opinion process involving subjective judgments as to the most appropriate and relevant methods of financial analysis and the applicant methods to the particular circumstances. The process, therefore, is not necessarily susceptible to a partial analysis or summary Sandler O'Neill believes that its analyses must be considered as a whole and that selecting portions of the factors and analyses to be without considering all factors and analyses, or attempting to ascribe relative weights to some or all such factors and analyses, conformated in Sandler O'Neill's comparated below is identical to KFN or KKR and no transaction is identical to the merger. Accordingly, an analysis of comparable of transactions involves complex considerations and judgments concerning differences in financial and operating characteristics of the and other factors that could affect the public trading values or merger transaction values, as the case may be, of KFN and I companies to which they are bein

In performing its analyses, Sandler O'Neill also made numerous assumptions with respect to industry performance, business at conditions and various other matters, many of which cannot be predicted and are beyond the control of KFN, KKR and Sandler analysis performed by Sandler O'Neill is not necessarily indicative of actual values or future results, both of which may be significated less favorable than suggested by such analyses. Sandler O'Neill prepared its analyses solely for purposes of rendering its opinion a such analyses to the transaction committee at its December 13, 2013 meeting. Estimates of the values of companies do not appraisals or necessarily reflect the prices at which companies or their securities may actually be sold. Such estimates are inherent uncertainty and actual values may be materially different. Accordingly, Sandler O'Neill's analyses do not necessarily reflect the values of the prices at which KFN's common shares may be sold at any time. The analyses of Sandler O'Neill and its cannon and number of factors taken into consideration by KFN's board of directors in making its determination to approve of KFN's board of merger agreement and the analyses described below should not be viewed as determinative of the decision of KFN's board of management with respect to the fairness of management with respect to the fairness of management with respect to the fairness of the decision of KFN's board of the decision of KFN's board of management with respect to the fairness of management with respect to the fairness of the decision of KFN's board of the decision of KFN's bo

In arriving at its opinion Sandler O'Neill did not attribute any particular weight to any analysis or factor that it considered. Ra qualitative judgments as to the significance and relevance of each analysis and factor. Sandler O'Neill did not form an opinion as to individual analysis or factor (positive or negative) considered in isolation supported or failed to support its opinions; rather Sandler O'Neill did not form an opinion as to individual analysis or factor (positive or negative) considered in isolation supported or failed to support its opinions; rather Sandler O'Neill did not form an opinion as to individual analysis or factor (positive or negative) considered in isolation supported or failed to support its opinions; rather Sandler O'Neill did not form an opinion as to individual analysis or factor (positive or negative) considered in isolation supported or failed to support its opinions; rather Sandler O'Neill did not form an opinion as to individual analysis or factor (positive or negative) considered in isolation supported or failed to support its opinions; rather Sandler O'Neill did not form an opinion as to individual analysis or factor (positive or negative) considered in isolation supported or failed to support its opinions; rather Sandler O'Neill did not form an opinion as to individual analysis or factor (positive or negative) considered in isolation supported or failed to support its opinions; rather Sandler O'Neill did not form an opinion as to individual analysis or factor (positive or negative) considered in isolation supported or failed to support its opinions; rather Sandler O'Neill did not form an opinion as to individual analysis or factor (positive or negative) considered in isolation supported or failed to support its opinions; rather Sandler O'Neill did not form an opinion as to individual analysis or factor (positive or negative) considered in isolation supported or failed to support its opinions; rather Sandler O'Neill did not form an opinion as to individual analysis or factor

Transacti

Sandler O'Neill reviewed the financial terms of the proposed transaction. As described in the merger agreement, KFN sharehold right to receive consideration consisting of 0.51 nonassessable KKR common units in exchange for each KFN common share.

KKR's closing price of \$24.25 as of December 12, 2013, Sandler O'Neill calculated a merger of the common share.

Table

value of \$12.37 per KFN common share. Based upon 204,824,159 common shares outstanding and using KKR's closing price of December 12, 2013, Sandler O'Neill calculated an aggregate merger consideration value of \$2,533 million. Based upon financial in of the period ended September 30, 2013, Sandler O'Neill calculated the following transitions are supported by the period ended September 30, 2013, Sandler O'Neill calculated the following transitions are supported by the period ended September 30, 2013, Sandler O'Neill calculated the following transitions are supported by the period ended September 30, 2013, Sandler O'Neill calculated the following transitions are supported by the period ended September 30, 2013, Sandler O'Neill calculated the following transitions are supported by the period ended September 30, 2013, Sandler O'Neill calculated the following transitions are supported by the period ended September 30, 2013, Sandler O'Neill calculated the following transitions are supported by the period ended September 30, 2013, Sandler O'Neill calculated the following transitions are supported by the period ended September 30, 2013, Sandler O'Neill calculated the following transitions are supported by the period ended September 30, 2013, Sandler O'Neill calculated the following transitions are supported by the period ended September 30, 2013, Sandler O'Neill calculated the following transitions are supported by the period ended September 30, 2013, Sandler O'Neill calculated the following transitions are supported by the period ended September 30, 2013, Sandler O'Neill calculated the following transitions are supported by the supporte

Transaction Value / Book Value Per Share:

119%

Historical Stock Trad

Sandler O'Neill reviewed the one-year and three-year share trading activity in KFN common shares and noted that the merger of based on the closing price of KKR's common units on December 12, 2013, was in excess of the highest closing share price for both and three-year period. Additionally, Sandler O'Neill observed that the merger consideration reflected a premium to KFN's closing share price for both December 12, 20

Premium to KFN Stock Price (December 12, 2013):

34.6%

Sandler O'Neill also noted that the proposed merger consideration exceeded the highest one-year trading price of KFN's comm \$11.67, as well as the one-year volume weighted average of \$10.57 and the one-year low of \$9.11. Sandler O'Neill did not analyze to KFN's closing share price based on KKR's common unit price on any date other than December 1.

Comparable Compa

Sandler O'Neill used publicly available information to compare selected financial information for KFN and a group of financia selected by Sandler O'Neill based on Sandler O'Neill's professional judgment and experience. The peer group consisted of NA NYSE-traded business development companies and senior / floating fund business development

The following financial institutions were selected for the

Business Development Companies:

Apollo Investment Corporation
Ares Capital Corporation
Blackrock Kelso Capital Corp.

New Mountain Finance Corporation
PennantPark Investment Corporation
Prospect Capital Corporation

Fifth Street Finance Corporation

Golub Capital BDC, Inc.

Hercules Technology Growth Capital, Inc.

Solar Capital Ltd.

TCP Capital Corporation

THL Credit, Inc.

Main Street Capital Corporation

Medley Capital Corporation

Ticc Capital Corporation

Triangle Capital Corporation

Investment Companies Focused on Senior and Floating Rate Loans:

Fifth Street Senior Floating Rate Corp

PennantPark Floating Rate Capital

Oxford Lane Capital Corporation Solar Senior Capital Ltd.

The analysis compared publicly available financial information for KFN and the mean and median financial and market trading peer group as of or for the period ended Septem

with pricing data as of December 12, 2013. The table below sets forth the data for KFN and the mean and median data for the

	KKR Financial (12/12/13 pricing)		BDC Median (12/12/13 pricing)		BDC Mean (12/12/13 pricing)		Inv. Co Senior / Floating Loans Median (12/12/13 pricing)		So FI	v. Co enior / oating coans Mean 2/12/13 ricing)
Market Cap (\$ in millions)	\$	\$ 1,882		789	\$	1,292	\$	167	\$	159
Total Assets (\$ in millions)	\$	8,472	\$	1,173	\$	1,837	\$	216	\$	215
LTM ROE		13.5%	,	10.5%		11.6%)	9.0%		7.5%
Price / NAV		0.88x		1.08x		1.18x		1.00x		0.98x
Price / 2013E EPS		7.2x		11.0x		11.8x		14.5x		20.8x
Price / 2014E EPS		8.8x		10.4x		10.8x		12.2x		12.3x
Price / 2015E EPS		8.0x		9.7x		10.0x		12.6x		12.6x
Current Dividend Yield(1)		9.6%	'n	9.0%		9.1%	,	7.8%)	8.9%

Dividend yield computed as most recent quarterly dividend annualized divided by closing pr

Sandler O'Neill used publicly available information to compare selected financial information for KKR and a group of financial selected by Sandler O'Neill based on Sandler O'Neill's professional judgment and experience. The peer group consisted of N NYSE-traded alternative ass

The following companies were selected for the

Apollo Global Investment, LLC

The Blackstone Group

Oaktree Capital Group, LLC

The Carlyle Group, L.P.

Och-Ziff Capital Management Group

The analysis compared publicly available financial information for KKR and the mean and median financial and market trading peer group as of or for the period ended September 30, 2013 with pricing data as of December 12, 2013. The table below sets forth KKR and the mean and median data for the

	 KR 3 pricing)	Alternative Asset Managers Median (12/13/13 pricing)	Alternative Asset Managers Mean (12/13/13 pricing)
Closing Price as % of 52 Week High	98%	93%	93%
2015E Dividend Yield	6.3%	7.3%	7.6%
LTM AUM Growth	36%	15%	12%
Average Daily Trading Volume (\$ in millions)	\$ 42.1	\$ 12.1	\$ 29.2
Price / 2012 Adjusted Economic Net Income(1)	8.4x	13.8x	13.4x
Price / 2013E Economic Net Income	9.9x	10.5x	10.4x
Price / 2014E Economic Net Income	9.8x	10.6x	10.5x
Price / 2015E Economic Net Income	8.9x	9.7x	9.8x
Implied ENI Growth 2012 Adjusted-2013	(15)%	28%	29%
Implied ENI Growth 2013-2014	0%	(1)%	(0)%

Excludes any one-time, non-red

<u>Table</u>

Stock Price I

Sandler O'Neill reviewed the publicly reported trading prices of KFN's common shares for the one-year and three-year p
December 12, 2013. Sandler O'Neill then compared the relationship between the movements in the price of KFN's common share
movements in the prices of the peer groups referenced above and the S&

One-Year Comparative Stock Performance

	Beginning Value December 12, 2012	Ending Value December 12, 2013		
KFN	100%	88%		
BDC Peers	100%	105%		
Senior / Floating Fund Peers	100%	102%		
S&P 500 Index	100%	124%		

Three-Year Comparative Stock Performance

	Beginning Value	Ending Value
	December 12, 2010	December 12, 2013
KFN	100%	101%
BDC Peers	100%	99%
Senior / Floating Fund Peers	100%	89%
S&P 500 Index	100%	143%

Sandler O'Neill reviewed the publicly reported trading prices of KKR's common units for the one-year and three-year p December 12, 2013. Sandler O'Neill then compared the relationship between the movements in the price of KKR's common unit movements in the prices of the peer group referenced above and the S&

One-Year Comparative Stock Performance

	Beginning Value December 12, 2012	Ending Value December 12, 2013
KKR	100%	172%
Alternative Asset Managers	100%	176%
S&P 500 Index	100%	124%

Three-Year Comparative Stock Performance

	Beginning Value December 12, 2010	Ending Value December 12, 2013
KKR	100%	181%
Alternative Asset Managers	100%	171%
S&P 500 Index	100%	143%

Research Analyst Estimates and I

Sandler O'Neill reviewed analyst estimated earnings per share, or EPS, for KFN for 2013 and 2014 along with analyst estimated targets. The mean and median for 2013 and 2014 EPS were based on reports from five research analysts. The mean and median target for KFN were based on reports from five research

Table

Summary of KFN Anal

		El	PS	Future Price				
	2	2013 2014			Target			
Mean	\$	1.27	\$	1.05	\$	11.70		
Median	\$	1.26	\$	1.07	\$	12.00		

Sandler O'Neill reviewed analyst estimated economic net income for KKR for 2013 and 2014 along with analyst estimated targets. The mean and median for 2013 and 2014 economic net income were based on reports from twelve research analysts. The target was based on reports from eleven research analysts. In some cases future price target information was recommendation.

Sandler O'Neill also noted the merger consideration exceeded the mean and median 12 month analyst target prices for KFN con and exceeded all but one of the individual analyst 12 month target prices for KFN con

Summary of KKR Anal

		Inco	ome	Future Price			
	2	2013		2014	Target		
Mean	\$	2.46	\$	2.47	\$	25.91	
Median	\$	2.49	\$	2.41	\$	26.00	

KFN Net Present Va

Sandler O'Neill performed an analysis that estimated the net present value per KFN common share under various circumstar O'Neill assumed that KFN performed in accordance with the publicly available mean analyst estimated earnings per share for the December 31, 2013 and December 31, 2014 and an estimated long-term growth rate of 10% as provided by mean analyst research the years thereafter, and also included annual cash dividend payments, confirmed by senior manager

To approximate the terminal value of KFN common shares at December 31, 2017, Sandler O'Neill applied dividend yields in 7.0% to 12.0%. The terminal values were then discounted to present values using different discount rates ranging from 7.0% to 12.0% reflect different assumptions regarding required rates of return of holders or prospective buyers of KFN's common shares at December 31, 2017, Sandler O'Neill applied dividend yields in 7.0% to 12.0%.

As illustrated in the following tables, the analysis indicates an imputed range of values per KFN common share of \$8.21 to applying dividend yields to the assumed quarterly dividend of \$0.22 per KFN common share and \$6.21 to \$11.70 when applying tangible book value to the assumed tangible book value, calculated on the basis of analyst

	Dividend Yields											
Discount Rate	7	7.0%	:	8.0%	9	9.0%	1	0.0%	13	1.0%	12	2.0%
7.00%	\$	13.45	\$	12.17	\$	11.17	\$	10.37	\$	9.72	\$	9.17
7.84%	\$	13.18	\$	11.92	\$	10.95	\$	10.17	\$	9.53	\$	9.00
8.00%	\$	13.13	\$	11.88	\$	10.91	\$	10.13	\$	9.50	\$	8.97
9.00%	\$	12.81	\$	11.60	\$	10.66	\$	9.90	\$	9.28	\$	8.77
10.00%	\$	12.51	\$	11.33	\$	10.41	\$	9.68	\$	9.08	\$	8.58
11.00%	\$	12.22	\$	11.07	\$	10.18	\$	9.46	\$	8.88	\$	8.39
12.00%	\$	11.94	\$	10.82	\$	9.95	\$	9.25	\$	8.69	\$	8.21
										49		

	Tangible Book Value Multiples											
Discount Rate	8	80%	9	00%	0% 100%		110%		120%		1	30%
7.00%	\$	7.20	\$	8.10	\$	9.00	\$	9.90	\$	10.80	\$	11.70
7.84%	\$	6.98	\$	7.85	\$	8.72	\$	9.59	\$	10.46	\$	11.34
8.00%	\$	6.93	\$	7.80	\$	8.67	\$	9.54	\$	10.40	\$	11.27
9.00%	\$	6.68	\$	7.52	\$	8.35	\$	9.19	\$	10.03	\$	10.86
10.00%	\$	6.44	\$	7.25	\$	8.05	\$	8.86	\$	9.67	\$	10.47
11.00%	\$	6.21	\$	6.99	\$	7.77	\$	8.55	\$	9.32	\$	10.10

Sandler O'Neill also considered and discussed with the KFN transaction committee how this analysis would be affected by poss in the underlying assumptions, including variations with respect to net income and dividends. To illustrate this impact, San performed a sensitivity analysis assuming KFN's net income varied from 25% above analyst projections to 25% below analyst programming that KFN's quarterly dividends paid varied from \$0.165 per KFN common share to \$0.275 per KFN common share. The analysis resulted in the following range of per share values for KFN common shares, using the same dividend yields of 7.0% to discount range.

	Dividend Yields											
Annual Variance	7	7.0%	1	8.0%	9	0.0%	1	0.0%	1	1.0%	1	2.0%
-25.00%	\$	9.88	\$	8.94	\$	8.21	\$	7.63	\$	7.15	\$	6.75
-20.00%	\$	10.54	\$	9.54	\$	8.76	\$	8.14	\$	7.63	\$	7.20
-15.00%	\$	11.20	\$	10.14	\$	9.31	\$	8.64	\$	8.10	\$	7.65
-10.00%	\$	11.86	\$	10.73	\$	9.85	\$	9.15	\$	8.58	\$	8.10
-5.00%	\$	12.52	\$	11.33	\$	10.40	\$	9.66	\$	9.06	\$	8.55
0.00%	\$	13.18	\$	11.92	\$	10.95	\$	10.17	\$	9.53	\$	9.00
5.00%	\$	13.84	\$	12.52	\$	11.50	\$	10.68	\$	10.01	\$	9.45
10.00%	\$	14.49	\$	13.12	\$	12.04	\$	11.19	\$	10.49	\$	9.90
15.00%	\$	15.15	\$	13.71	\$	12.59	\$	11.70	\$	10.96	\$	10.35
20.00%	\$	15.81	\$	14.31	\$	13.14	\$	12.20	\$	11.44	\$	10.80
25.00%	\$	16.47	\$	14.90	\$	13.69	\$	12.71	\$	11.91	\$	11.25

The following table describes a discount rate calculation for KFN. The discount rate equals the risk free rate plus the product beta and equity ri

Risk Free Rate	2.88% 10 Year UST Yield
Two Year Beta	0.87 Per Bloomberg
Equity Risk Premium	5.70% Ibbotson 60 year market analysis
Discount Rate	7.84%

During the December 13, 2013 meeting of the KFN transaction committee, Sandler O'Neill noted that the net present value widely used valuation methodology, but the results of such methodology are highly dependent upon the numerous assumptions made, and the results thereof are not necessarily indicative of actual values or future results. Sandler O'Neill noted the merger consi within the range of the values indicated in the tables presented in the above net present value analysis of KFN con

KKR Net Present Va

Sandler O'Neill also performed an analysis that estimated the net present value per KKR common unit under various circumstar O'Neill assumed that KKR performed in accordance with publicly available mean analyst estimated economic net income per unit

Table

December 31, 2013, December 31, 2014 and December 31, 2015 and an estimated long-term earnings growth rate of 6.9% as provious analyst estimates for the years thereafter, and also included mean analyst estimates with respect to annual distribution.

To approximate the terminal value of KKR common units at December 31, 2017, Sandler O'Neill applied economic net incorranging from 8.0x to 13.0x, which Sandler O'Neill determined to be a standard deviation both up and down with the calculated distinct the mid-point of that range. The terminal values were then discounted to present values using different discount rates ranging for 13.0%, which Sandler O'Neill determined to be a standard deviation both up and down with the calculated discount rate at the midrange, and were chosen to reflect different assumptions regarding required rates of return of holders or prospective buyers of KK

As illustrated in the following tables, the analysis indicates an imputed range of values per KKR common unit of \$22.41 to applying economic net income multiples to applicable analyst projection

	Economic Net Income Per Unit Multiples											
Discount Rate		8.0x		9.0x		10.0x		11.0x		12.0x		13.0x
8.00%	\$	25.28	\$	27.74	\$	30.21	\$	32.68	\$	35.14	\$	37.61
9.00%	\$	24.66	\$	27.06	\$	29.46	\$	31.86	\$	34.26	\$	36.66
9.94%	\$	24.10	\$	26.44	\$	28.78	\$	31.12	\$	33.45	\$	35.79
10.00%	\$	24.07	\$	26.40	\$	28.74	\$	31.07	\$	33.41	\$	35.74
11.00%	\$	23.50	\$	25.77	\$	28.04	\$	30.31	\$	32.58	\$	34.86
12.00%	\$	22.94	\$	25.15	\$	27.37	\$	29.58	\$	31.79	\$	34.00
13.00%	\$	22.41	\$	24.56	\$	26.72	\$	28.87	\$	31.02	\$	33.18

Sandler O'Neill also considered and discussed with the transaction committee how this analysis would be affected by possible the underlying assumptions, including variations with respect to net income. To illustrate this impact, Sandler O'Neill performed analysis assuming KKR's economic net income and distributions / unit varied from 25% above analyst projections to 25% by projections. This sensitivity analysis resulted in the following range of per unit values for KKR common units, using the same price multiples of 8.0x to 13.0x and a discount range of the common units, using the same price multiples of 8.0x to 13.0x and a discount range of the common units, using the same price multiples of 8.0x to 13.0x and a discount range of the common units, using the same price multiples of 8.0x to 13.0x and a discount range of the common units, using the same price multiples of 8.0x to 13.0x and a discount range of the common units, using the same price multiples of 8.0x to 13.0x and a discount range of the common units, using the same price multiples of 8.0x to 13.0x and a discount range of the common units, using the same price multiples of 8.0x to 13.0x and a discount range of the common units, using the same price multiples of 8.0x to 13.0x and a discount range of the common units, using the same price multiples of 8.0x to 13.0x and a discount range of the common units, using the same price multiples of 8.0x to 13.0x and a discount range of the common units, using the same price multiples of 8.0x to 13.0x and a discount range of the common units, using the same price multiples of 8.0x to 13.0x and a discount range of the common units, using the same price multiples of 8.0x to 13.0x and a discount range of the common units, using the same price multiples of 8.0x to 13.0x and a discount range of the common units, using the same price multiples of 8.0x to 13.0x and a discount range of the common units, using the same price multiples of 8.0x to 13.0x and a discount range of the common units, using the same price multipl

	Economic Net Income Multiples											
Annual Variance		8.0x		9.0x		10.0x		11.0x	1	12.0x		13.0x
-25.00%	\$	18.08	\$	19.83	\$	21.58	\$	23.34	\$	25.09	\$	26.84
-20.00%	\$	19.28	\$	21.15	\$	23.02	\$	24.89	\$	26.76	\$	28.63
-15.00%	\$	20.49	\$	22.47	\$	24.46	\$	26.45	\$	28.44	\$	30.42
-10.00%	\$	21.69	\$	23.80	\$	25.90	\$	28.00	\$	30.11	\$	32.21
-5.00%	\$	22.90	\$	25.12	\$	27.34	\$	29.56	\$	31.78	\$	34.00
0.00%	\$	24.10	\$	26.44	\$	28.78	\$	31.12	\$	33.45	\$	35.79
5.00%	\$	25.31	\$	27.76	\$	30.22	\$	32.67	\$	35.13	\$	37.58
10.00%	\$	26.51	\$	29.08	\$	31.66	\$	34.23	\$	36.80	\$	39.37
15.00%	\$	27.72	\$	30.41	\$	33.09	\$	35.78	\$	38.47	\$	41.16
20.00%	\$	28.92	\$	31.73	\$	34.53	\$	37.34	\$	40.15	\$	42.95
25.00%	\$	30.13	\$	33.05	\$	35.97	\$	38.90	\$	41.82	\$	44.74
									51			

Table

The following table describes a discount rate calculation for KKR. The discount rate equals the risk free rate plus the product beta and equity ri

Risk Free Rate	2.88% 10 Year UST Yield
Two Year Beta	1.24 Per Bloomberg
Equity Risk Premium	5.70% Ibbotson 60 year market analysis
Discount Rate	9 94%

At the December 13, 2013 meeting of the transaction committee, Sandler O'Neill noted that the net present value analysis is a valuation methodology, but the results of such methodology are highly dependent upon the numerous assumptions that must be methodology results thereof are not necessarily indicative of actual values or fi

Sandler O'Neill noted that the trading price of KKR common units at the time of the presentation was within the range indicated in the tables presented in the above net present value analysis of KKR co

Analysis of Selected Merger T

Sandler O'Neill reviewed two groups of comparable mergers and acquisitions. The first group consisted of mergers and accompanies that were considered vehicles that were holders of financial assets and were managed by another party. The second group mergers and acquisitions of companies in the financial services industry where the buyer and the target had commercial relationsh those between KFN and KKR. In each group, these were the only transactions meeting the criteria described above and no transection excluded from either of the comparable merger and acquisitions.

The first group of mergers and acquisitions included eleven transactions announced between January 30, 2011 and Octo selected based on Sandler O'Neill's professional judgment and experience. The group was composed of the following

American Realty Capital Properties, Inc. / Cole Real Estate Inve Parkway Properties, Inc. / Thomas Properties Mid-America Apartment Communities, Inc. / Colonial Pro American Realty Capital Properties, Inc. / Ca Annaly Capital Management, Inc. / CreXus Inve Realty Income Corporation / American Realty Capit HarbourVest Partners / Conv Ventas, Inc. / Cogdell HarbourVest Partners / Absolute P. Ventas, Inc. / Nationwide Health Pr

The second group of mergers and acquisitions included seven transactions announced between November 20, 2006 and Novem selected based on Sandler O'Neill's professional judgment and experience. In order to achieve a sufficient number of representative deemed adequate by Sandler O'Neill, Sandler O'Neill expanded the range of dates used in selecting the second

52

AMB Property Corporation

<u>Table</u>

Alfa Corp. /

compared to that used in selecting the first group. The group was composed of the following

Annaly Capital Management, Inc. / CreXus Investigation of Leucadia National Corp. / Jefferies Sovereign Bancorp, Inc. / Banco SUnionBanCal Corp. / Bank of Tokyo-Mitsubi

Nationwide Financial Services, Inc. / Nationwide Mutual I

TD Banknorth, Inc. / Toronto-Do

Sandler O'Neill then reviewed the following multiples for each of the transactions: transaction price to book value and transaction target company's stock price days before transaction announcement. As illustrated in the following table, Sandler O'Neill compared merger multiples to the median multiples of the comparable

		First	Second
		Group	Group
	KKR/	Transaction	Transaction
	KFN	Medians	Medians
Transaction Value / Book Value Per Share:	119%	103%	NM
Premium to KFN Stock Price (Dec. 12, 2013):	34.6%	8.4%	23.7%

Share Repurch

Sandler O'Neill performed an analysis that estimated the earnings per share and tangible book value accretion to KFN common KFN were to repurchase \$200 million of its common shares at a range of prices between \$9.19 to \$10.00 per share. The analysis is such share repurchase could be projected to result in 2015 earnings per share accretion to KFN common shares of between 3.84 2015. Additionally, Sandler O'Neill noted that based on the assumptions above, such share repurchase could be projected to result book value per share accretion to KFN common shares of between 0.4% to 1.4% as of Septemb

Pro Fo

Sandler O'Neill analyzed certain potential pro forma effects of the merger, assuming the following: (i) the merger closes on Janu (ii) per share merger consideration value of \$12.37, based on KKR's closing unit price on December 12, 2013 of \$24.25; (iii) KI achieve cost savings of approximately \$4.5 million and such savings are 100% realized in 2014; (iv) KFN's performance is co publicly available mean analyst estimated earnings per share for the year ending December 31, 2014 and an estimated long-term graph 10% for the years thereafter; (v) KKR's performance is consistent with publicly available mean analyst economic net income esting years ending December 31, 2014 and December 31, 2015 and an estimated long-term growth rate of 6.9% for the years thereafter. Indicated that (i) for the year ending December 31, 2014, the merger (excluding transaction expenses) would be dilutive to KKI economic net income per unit, (ii) for the year ending December 31, 2015, the merger would be accretive to KKR's projected earnings, and (iii) as of September 30, 2013, the merger would be accretive to KKR's book value per unit. The actual results accombined company, however, may vary from projected results and the variations may

<u>Table</u>

The table below shows Sandler O'Neill's projected accretion/dilution percentages for both KFN and KKR as of closing and fo

	Closing	Year Ending 12/31/2014	Yea 12/
KKR Economic Net Income Accretion / (Dilution) excluding transaction expenses		(1.9)%	
KFN EPS Accretion / (Dilution) excluding transaction expenses (assumes 100% stock			
allocation)		17.7%	
KKR Total Distributable Earnings Accretion / (Dilution) excluding transaction expenses		1.0%	
KKR Distributions / Unit Accretion / (Dilution)		6.9%	
KFN Income Distribution Accretion / (Dilution)		(15.7)%	
KKR BVPS Accretion / (Dilution)	17.5%		
KFN BVPS Accretion / (Dilution)	(42.1)%		

Preliminary Presentations by San

In addition to its December 13, 2013 fairness opinion presentation described above, Sandler O'Neill also made prelim presentations to the transaction committee of the KFN board of directors on the following dates, which are referred to statement/prospectus as the preliminary Sandler p

Novemb Novemb Novemb

Decen

Decen

Decem

Noven

The preliminary Sandler presentations consisted of various summary data and analyses that Sandler O'Neill utilized in for preliminary perspective on the merger, were for discussion purposes only, and did not present any findings or make any recomme constitute an opinion, or a part of any opinion, of Sandler O'Neill with respect to the fairness of the merger consideration. The only in which Sandler O'Neill presented its findings with respect to the fairness of the consideration to be received by KFN shareholders O'Neill's December 13, 2013 presentation described above. The preliminary Sandler presentations contained substantially similar described above in connection with the delivery of Sandler O'Neill's fairness opinion. The preliminary Sandler presentations primary the continued negotiations surrounding the proposed exchange ratio in the proposed

The November 6, 2013 preliminary Sandler presentation contained an overview of the proposed structure and terms of preliminary Sandler valuation analyses, a summary analysis of the proposed merger consideration, a review of the financial term previous business combinations, and analyst estimated future earnings per share, price targets and recommendation

Table

The November 12, 2013 preliminary Sandler presentation contained updated Sandler valuation analyses, an update on the proposed merger consideration and the resulting implied premiums, an accretion/dilution analysis of KKR and case studies of publishess of business of the proposed merger consideration and the resulting implied premiums, an accretion/dilution analysis of KKR and case studies of publishess of the proposed merger consideration and the resulting implied premiums, an accretion/dilution analysis of KKR and case studies of publishess of the proposed merger consideration and the resulting implied premiums, an accretion/dilution analysis of KKR and case studies of publishess of the proposed merger consideration and the resulting implied premiums, an accretion/dilution analysis of KKR and case studies of publishess of the proposed merger consideration and the resulting implied premiums, an accretion/dilution analysis of KKR and case studies of publishess of the proposed merger consideration and the resulting implied premiums.

The November 21, 2013 preliminary Sandler presentation contained updated Sandler valuation analyses reflecting the revision merger consideration and a summary description of Sandler O'Neill's meeting with KKR and its financial advisor regarding rationale for the

The November 24, 2013 preliminary Sandler presentation included a summary of the rationale to be expressed to KKR and advisor by representatives of Sandler O'Neill for an increase in the value of the proposed merger consideration, and was also representati

The November 27, 2013 preliminary Sandler presentation contained updated Sandler valuation analyses reflecting the revision merger of

The December 5, 2013 preliminary Sandler presentation summarized KKR's responses to certain questions regarding KKR's but by Sandler O'Neill during a conversation with members of KKR's management held on December 4, 2013. These included summar responses to certain questions posed by Sandler O'Neill regarding KKR's investing and financial performance and its management's easuch performance, KKR's fundraising activities, the potential evolution of KKR's businesses and the composition of KKR's management and the implications of such evolution, KKR's strategy and management plans, and KKR's fund investor base. In presentation summarized KKR's responses to certain questions regarding its plans following the completion of the transaction, incluanticipated plan to distribute 100% of KFN's cash earnings, KKR's anticipated plan to initially maintain KFN as a separate legal KKR's investment expectations with respect to KFN's and KKR's balance sheet. The presentation also summarized KKR's expendent and anticipated reactions to the announcement of the management of the summarized to the announcement of the management of the summarized to the announcement of the management of the summarized to the announcement of the summarized to the announcement of the summarized to the announcement of the summarized to the summariz

The December 6, 2013 preliminary Sandler presentations contained a pro forma trading analysis of KKR, a summary of Sand equity research coverage of KKR and of KKR's platform, managed assets and financial and stock price performance, and Sandler O'l research analyst's views on the pote

The December 10, 2013 preliminary Sandler presentation contained updated Sandler valuation analyses reflecting the revision merger of

Specifically, the presentations contained the follow

proposed structure and terms of the transaction including the then-current proposed merger consideration and the proposed merger consideration on holders of KFN common shares, calculated on November 21, 2013 at proposed exchange ratio of 0.48 KKR common units per KFN common share, which would imply per share con \$10.91 based on the trading price of KKR common units of \$22.73 on November 20, 2013. The aggregate transaction as of November 21, 2013 would have been \$2,235 million. Previously, similar calculations were per November 6, 2013, at which time the proposed exchange ratio was 0.46 KKR common units per KFN common trading price of KKR common units was \$22.87 on November 4, 2013, and the implied per share consideration. The final transaction proposal, which contemplated an exchange ratio of 0.51 KKR common units per KFN common units pe

Table

comparable public company analyses using the same methodology as that described on pages 46 and 47 above information as of the date on which the analyses were

precedent transaction analyses using the same methodology as that described on pages 52 and 53 above information as of the date on which the analyses were

pro forma financial analyses using the same methodology as described on page 53 and 54 above, and related in of the date on which the analyses were

discounted cash flow analyses, using the same methodology as described on pages 49 and 50 above, and related as of the date on which the analyses wer

historical stock price performance analysis of KFN and KKR as described on page 48 above, and related information the date on which the analysis were

analyst price targets as

accretion/dilution analyses in which Sandler O'Neill considered the standalone net income of each of KKR armade adjustments for the proposed merger (including expense synergies, potential tax consequences and possible business mix, among other pro forma adjustments, based on information provided by each of KKF.

In the November 12, 2013 preliminary presentation, Sandler O'Neill noted that based on information Sandler O'Neill, the proposed acquisition of KFN could be projected to result in a range of earnin accretion/(dilution) to KKR of (4.6)% to 8.5% in 2015 under the strategic plan cases reflected in the K projections and various mixtures of stock and cash merger consideration. Additionally, Sandler O'Neill range of maximum prices that KKR could pay to KFN common shareholders and not suffer earnin dilution under the strategic plan cases reflected in the KFN financial projections and various mixtures cash merger consideration. The analysis indicated a range of \$6.47 to \$24.00 in per share merger could be paid without earnings per share dilutions.

In the November 21, 2013 preliminary presentation, Sandler O'Neill noted that based on information Sandler O'Neill, the proposed acquisition of KFN could be projected to result in a range of earning accretion/(dilution) to KKR of (5.1)% to 1.5% in 2015 under the strategic plan cases reflected in the K projections and assuming merger consideration consisting entirely of KKR common units. Sandler noted that the proposed acquisition of KFN could be projected to result in a range of earning accretion/(dilution) to KFN common shareholders of (8.6)% to 68.6% in 2015 under the strateg reflected in the KFN financial projections and assuming merger consideration consisting entirely of K units. Additionally, Sandler O'Neill calculated a range of maximum prices that KKR could pay to K shareholders and not suffer earnings per share dilution under the strategic plan cases reflected in the K projections and assuming merger consideration consisting entirely of KKR common units. The analysis range of \$6.27 to \$12.27 in merger consideration could be paid without earnings per share dilu Sandler O'Neill also noted that the proposed acquisition of KFN could be projected to result in a range distribution per share accretion to KKR of 0.2 to 12.1% in 2015 under the strategic plan cases reflected financial projections and assuming merger consideration consisting entirely of KKR common u O'Neill also noted that the proposed acquisition of KFN could be projected to result in a ran distribution per share accretion/(dilution) to KFN common shareholders of (37.8)% to 9.7% in 20 strategic plan cases reflected in the KFN financial projections and assuming merger consideration

entirely of KKR co

Table

In the November 27, 2013 preliminary presentation, Sandler O'Neill noted that based on information Sandler O'Neill, the proposed acquisition of KFN could be projected to result in a range of earning accretion/(dilution) to KKR of (5.6)% to 1.0% in 2015 under the strategic plan cases reflected in the K projections and assuming merger consideration consisting entirely of KKR common units. Sandler noted that the proposed acquisition of KFN could be projected to result in a range of earning accretion/(dilution) to KFN common shareholders of (5.2)% to 74.8% in 2015 under the strategoreflected in the KFN financial projections and assuming merger consideration consisting entirely of K units. Sandler O'Neill also noted that the proposed acquisition of KFN could be projected to result income distribution per share accretion/(dilution) to KKR of (0.3)% to 11.5% in 2015 under the strategoreflected in the KFN financial projections and assuming merger consideration consisting entirely of K units. Sandler O'Neill also noted that the proposed acquisition of KFN could be projected to result income distribution per share dilution to KFN common shareholders of (13.9)% to (3.7)% in 20 strategic plan cases reflected in the KFN financial projections and assuming merger consideration entirely of KKR consideration consisting entirely entirely entirely entirely entirely entirely entirely ent

These numbers are in comparison to the December 13, 2013 presentation of Sandler O'Neill in w O'Neill noted that based on information provided to Sandler O'Neill, the proposed acquisition of K projected to result in earnings dilution to KKR of (1.9)% in 2014 and 2015. Sandler O'Neill also n proposed acquisition of KFN could be projected to result in earnings accretion/(dilution) to K shareholders of 17.7% in 2014 and 17.3% in 2015. Sandler O'Neill noted that the proposed acquisic could be projected to result in distribution accretion to KKR of 6.9% in 2014 and 6.7% in 2015. Sandler O'Neill noted that the proposed acquisition of KFN could be projected to result in income distribution dilucommon shareholders of (15.7)% in 2014 and (6.0)

market reviews and potential capital raising alternatives in which Sandler O'Neill reviewed, among other things price of KFN common shares and selected peer group companies, as well as activity in the equity capital market its observations, as well as wider macroeconomic trends, Sandler O'Neill compared for the KFN transaction of equity offering by KKR with the proposed merger with KKR, including an implied price per share analysis

In the November 12, 2013 preliminary presentation Sandler O'Neill noted that a \$2.2 billion equity r (which amount was equal to the then-current proposed aggregate merger consideration value) at a offering prices between \$20.00 to \$21.50 could be projected to result in an implied price range per K share of \$11.3

In the November 21, 2013 preliminary presentation Sandler O'Neill noted that a \$2.2347 billion ec KKR (which amount was equal to the then-current proposed aggregate merger consideration value) net offering prices between \$20.00 to \$21.50 could be projected to result in an implied price ration common share of \$11.0

Has/Gets Analysis based on the publicly available analyst earnings estimates for KFN and KKR and the various scenarios provided by management of KFN, in which Sandler O'Neill conducted a "Has-Gets" analysis of the inproposed transaction on the shareholders of KFN. A "Has-Gets" analysis is a comparison of what shareholders of KFN and the proposed merger in terms of one or more financial ratios (and

Table

not have absent consummation of such merger) to what such shareholders will have upon consummation o

In its November 21, 2013 preliminary presentation Sandler O'Neill analyzed KFN's 2015 earnings p stand-alone basis and then pro forma for the transaction with KKR. Sandler O'Neill also evaluated the related to 2015 Income Distribution, Long-Term Analyst Growth Rate, One Year Daily Trading Volu Year Total Return to Shareholders. Sandler O'Neill's analyses did not include the effect of any synergi of the merger in performing this analysis. The analyses indicated the 2015 Earnings Per Share (strategic plan cases reflected in the KFN financial projections and current consensus mean analysis would increase/(decrease) by a range of \$(0.13) and \$0.51, KKR's One Year Daily Trading Volume w. \$31.5 million, and KKR's One Year Total Return to Shareholders was 77.9% compared to KFN's 5.1 Income Distribution (based on the strategic plan cases reflected in the KFN financial projections consensus mean analysts' estimates) would increase/(decrease) by a range of \$0.06 and \$(0.49) and the Analyst growth rate of KKR was 10.0% compared to KFN's reference in the KFN is reference.

imputed valuation analyses of KFN common shares in which Sandler O'Neill analyzed imputed valuation per coin the preliminary presentations as described below and based on the strategic plan cases reflected in the K projections to Sand

in the November 6, 2013 preliminary presentation, Sandler O'Neill's analysis indicated the ranges based on the median values for each of the strategic plan cases reflected in the KFN financial project following for three comparisons.

\$7.33 to \$14.47 based on a group of business developmen

\$8.44 to \$16.66 based on a group of senior / floating fund business development con

\$8.40 to \$16.57 based on KFN's historical average three year valuation based on discontinuous (calculated assuming 90% of earnings per share are paid out as

in the November 6, 2013 preliminary presentation, Sandler O'Neill's analysis indicated the ranges based on the mean values for each of the strategic plan cases reflected in the KFN financial project following for three comparts

\$7.32 to \$14.44 based on a group of business developmen

\$7.47 to \$14.74 based on a group of senior / floating fund business development con

\$8.62 to \$17.01 based on KFN's historical average three year valuation based on di (calculated assuming 90% of earnings per share are paid out as

in the November 21, 2013 preliminary presentation, Sandler O'Neill's analysis indicated the ranges based on the median values for each of the strategic plan cases reflected in the KFN financial project following for three comparts

\$8.13 to \$15.04 based on a group of business developmen

<u>Table</u>

\$9.02 to \$16.79 based on a group of senior / floating fund business development con

\$8.38 to \$16.53 based on KFN's historical average three year valuation based on di (calculated assuming 90% of earnings per share are paid out as

in the November 21, 2013 preliminary presentation, Sandler O'Neill's analysis indicated the ranges based on the mean values for each of the strategic plan cases reflected in the KFN financial project following for three comparisons.

\$7.93 to \$14.64 based on a group of business developmen

\$8.12 to \$15.01 based on a group of senior / floating fund business development con

\$8.57 to \$16.91 based on KFN's historical average three year valuation based on di (calculated assuming 90% of earnings per share are paid out as

imputed valuation analyses of KKR common units pro forma for a merger with KFN in which, in the Novem preliminary presentation, Sandler O'Neill's analysis indicated the ranges listed below, assuming KKR woul accordance with mean consensus analysts' estimates and KFN would perform in accordance with either mean analysts' estimates or one of three strategic plan cases reflected in the KFN financial projections, for valuation on the following criteria as compared to publicly available trading information and analyst estimates of the response of KF

\$22.46 to \$24.03 based on KKR maintaining its current price to estimated 2015 earning

\$24.55 to \$26.26 based on KKR trading at the current mean price to estimated 2015 earnings multip

\$23.72 to \$26.54 based on KKR's mean consensus analysts' expected div

Share repurchase analysis in which Sandler O'Neill performed an analysis that estimated the earnings per share book value accretion to KFN common shares if KFN were to repurchase various aggregate dollar amounts of contact a range of prices. The analysis indicated that such share repurchase could be projected to result in various learnings per share accretion to KFN. Additionally, Sandler O'Neill estimated the tangible book value accretion to KFN as of September 30, 2013 under these same a

In the November 21, 2013 preliminary presentation Sandler O'Neill noted that based on a share r \$100 million at a range of prices between \$9.75 to \$10.91 per KFN common share that such share repu be projected to result in 2014 earnings per share accretion/(dilution) to KFN common shares of be 1.6% and 2015 earnings per share accretion/(dilution) to KFN common shares of between (0.5 Additionally, Sandler O'Neill noted that based on the assumptions above that such share repurch projected to result in tangible book value per share accretion/(dilution) to KFN common shares of between to \$0.04 as of Septemb

In the November 21, 2013 preliminary presentation Sandler O'Neill noted that based on a share r \$200 million at a range of prices between \$9.75 to \$10.91 per KFN common share that such share repu

be projected to result in 2014 earnings per share accretion to KFN common shares of between 2.2% 2015

Table

share accretion/(dilution) to KFN common shares of between (1.0)% to 5.6%. Additionally, Sandler C that based on the assumptions above that such share repurchase could be projected to result in tangible per share accretion/(dilution) to KFN common shares of between \$(0.05) to \$0.07 as of September 1.00 per share accretion/(dilution) to KFN common shares of between \$(0.05) to \$0.07 as of September 1.00 per share accretion/(dilution) to KFN common shares of between \$(0.05) to \$0.07 as of September 1.00 per share accretion/(dilution) to KFN common shares of between \$(0.05) to \$0.07 as of September 1.00 per share accretion/(dilution) to KFN common shares of between \$(0.05) to \$0.07 as of September 1.00 per share accretion/(dilution) to KFN common shares of between \$(0.05) to \$0.07 as of September 1.00 per share accretion/(dilution) to KFN common shares of between \$(0.05) to \$0.07 as of September 1.00 per share accretion/(dilution) to KFN common shares of between \$(0.05) to \$0.07 as of September 1.00 per share accretion/(dilution) to \$0.07 as of September 1.00 per share accretion/(dilution) to \$0.07 as of September 1.00 per share accretion/(dilution) to \$0.07 as of September 1.00 per share accretion/(dilution) to \$0.07 as of September 1.00 per share accretion/(dilution) to \$0.07 as of September 1.00 per share accretion/(dilution) to \$0.07 as of September 1.00 per share accretion/(dilution) to \$0.07 as of September 1.00 per share accretion/(dilution) to \$0.07 as of September 1.00 per share accretion/(dilution) to \$0.07 as of September 1.00 per share accretion/(dilution) to \$0.07 as of September 1.00 per share accretion/(dilution) to \$0.00 per share accretion/(dilu

These numbers are in comparison to the December 13, 2013 presentation Sandler O'Neill where it w based on a share repurchase of \$200 million at a range of prices between \$9.19 to 10.00 per KFN contact that such share repurchase could be projected to result in 2015 earnings per share accretion to KFN confibetween 3.8% to 4.8%. Additionally, Sandler O'Neill noted that based on the assumptions above the repurchase could be projected to result in tangible book value per share accretion to KFN combetween 0.4% to 1.4% as of Septemb

Each of the analyses performed in these preliminary presentations was subject to further updating and subject to the final analyse to the transaction committee on December 13, 2013 by Sandler O'Neill. Each of these analyses was necessarily based on economic market and other conditions as in effect on, and the information made available to Sandler O'Neill as of, the dates on which San performed such analyses. Accordingly, the results of the financial analyses may have differed due to changes in those condition information, and not all of the written and oral presentations contained all of the financial analyses

The procedures followed by Sandler O'Neill in preparing the material analyses in the preliminary Sandler presentations were similar to the procedures used by Sandler O'Neill to prepare the corresponding analyses in its December 13, 2013 fairness opinion of Copies of these written materials have been filed as exhibits to the Schedule 13E-3 filed with the SEC in connection with the merge made available for inspection and copying at the principal offices of KFN during its regular business hours by any interested he common shares. Copies may be obtained by requesting them in writing from KFN at the address provided in the section titled "Wh Find More Information" beginning on page 234 of this proxy statemen

M

Sandler O'Neill acted as the financial advisor to the transaction committee in connection with the merger and will recapproximately \$17,000,000 in connection with the merger, of which \$500,000 was received upon execution of the engagement letter was received upon delivery of the fairness opinion, \$500,000 will be paid on the date on which KFN first mails the proxy statement this proxy statement/prospectus to KFN shareholders, and approximately \$15,000,000 of which is contingent on the consumr merger. KFN has also agreed to reimburse Sandler O'Neill's reasonable out-of-pocket expenses incurred in connection with its eng to indemnify Sandler O'Neill and its affiliates and their respective partners, directors, officers, employee and agents against certain of liabilities, including liabilities under the second

Sandler O'Neill employs research analysts who cover KKR, but Sandler O'Neill research analysts do not cover KFN and Sandle not issued research reports with res

In the ordinary course of its respective broker and dealer businesses, Sandler O'Neill may purchase securities from and sell KFN and KKR and their respective affiliates. Sandler O'Neill may also actively trade the debt and/or equity securities of KFN or I respective affiliates for their own accounts and for the accounts of their customers and, accordingly may at any time hold a long or s in such securities. Sandler O'Neill has provided investment banking services to, and received fees for such services from, KFN, mos connection with KFN's purchase of debt securities in 2010. Within the past two years, other than its role as financial advisor to the committee of the KFN board of directors in connection with the merger, Sandler has not provided any services to KFN or a subsidiaries. In addition, within the past two years, and the provided and the past two years, the provided any services to KFN or a subsidiaries. In addition, within the past two years, the provided any services to KFN or a subsidiaries.

Table

Sandler O'Neill has also not provided any services to KKR or, to Sandler O'Neill's knowledge, any portfolio comp.

Purpose and Reasons of the KKR Participants for

KKR, Fund Holdings, Merger Sub, KKR Group Holdings L.P., KKR Fund Holdings GP Limited, KKR Group Limit Management LLC, which are referred to collectively in this proxy statement/prospectus as the KKR Participants, are making the included in this section solely for the purpose of complying with the disclosure requirements of Rule 13e-3 and related rules under the Act. However, the making of such statements is not an admission by the KKR Participants that KFN is "controlled" by KKR or an KKR such that KFN should be deemed to be an "affiliate" for purposes of Rule 13e-3 in connection with

Fund Holdings is the sole member of Merger Sub, KKR Group Holdings L.P. and KKR Fund Holdings GP Limited are the gen of Fund Holdings, KKR Group Holdings L.P. is the sole shareholder of KKR Fund Holdings GP Limited, KKR Group Limited i partner of KKR Group Holdings L.P., KKR is the sole shareholder of KKR Group Limited and KKR Management LLC is the generator of KKR Group Holdings L.P., KKR is the sole shareholder of KKR Group Limited and KKR Management LLC is the generator of KKR Group Holdings L.P., KKR is the sole shareholder of KKR Group Limited and KKR Management LLC is the generator of KKR Group Holdings L.P., KKR is the sole shareholder of KKR Group Limited and KKR Management LLC is the generator of KKR Group Holdings L.P., KKR is the sole shareholder of KKR Group Limited and KKR Management LLC is the generator of KKR Group Holdings L.P., KKR is the sole shareholder of KKR Group Limited and KKR Management LLC is the generator of KKR Group Holdings L.P., KKR is the sole shareholder of KKR Group Limited and KKR Management LLC is the generator of KKR Group Limited and KKR Management LLC is the generator of KKR Group Limited and KKR Management LLC is the generator of KKR Group Limited and KKR Management LLC is the generator of KKR Group Limited and KKR Management LLC is the generator of KKR Group Limited and KKR Management LLC is the generator of KKR Group Limited and KKR Management LLC is the generator of KKR Group Limited and KKR Management LLC is the generator of KKR Group Limited and KKR Management LLC is the generator of KKR Group Limited and KKR Management LLC is the generator of KKR Group Limited and KKR Management LLC is the generator of KKR Group Limited and KKR Management LLC is the generator of KKR Group Limited and KKR Management LLC is the generator of KKR Group LLC is the generator of KKR Group

For the KKR Participants, the purpose of the merger is to enable Fund Holdings to acquire all of the outstanding KFN common as a result, for Fund Holdings and its indirect owners, including KKR, to bear the rewards and risk of such ownership of KFN common shares cease to be put

The KKR Participants believe that the transaction structure of the merger is preferable to other structures because it will Holdings to acquire all of the outstanding KFN common shares at one time, while allowing the unaffiliated KFN common sharticipate and share in the potential future profits of KKR, including future profits related to KKR's indirect ownership interest in I

The KKR Participants' reasons for entering into the merger at this time include that KKR is continually exploring potential tran would enhance unitholder value relative to the status of

KKR believes KFN is a specialty finance business with a complementary and known portfolio of assets and capit

KKR expects to acquire additional balance sheet scale to support its growth initiatives, including (1) the further KKR's investment management strategies, (2) increased exposure to capital market transactions that KKR so accelerated growth of KKR's new businesses, and (4) the pursuit of inorganic growth opport

KKR believes that the merger would (1) accelerate KKR's balance sheet objectives, including the diversification balance sheet holdings in addition to increasing their liquidity and yield, (2) increase the size and recurring common to KKR common unitholders, and (3) involve minimal integration risk, because KFN's asset

In addition, KKR's desire to acquire additional balance sheet scale and to diversify its balance sheet holdings, along with the noted above, is more readily achievable at this time given the market's current valuation of KKR's present and potential future earn.

Opinion of the Financial Adv

Goldman Sachs rendered its opinion to the board of directors of KKR Management LLC, which is referred to statement/prospectus as KKR's managing partner, that, as of D

Table

2013 and based upon and subject to the factors and assumptions set forth therein, the exchange ratio pursuant to the merger agreement from a financial point of v

The full text of the written opinion of Goldman Sachs, dated December 16, 2013, which sets forth assumptions made, followed, matters considered and limitations on the review undertaken in connection with the opinion, is attached as An proxy statement/prospectus. Goldman Sachs provided its opinion for the information and assistance of the board of directo managing partner in connection with its consideration of the

In connection with rendering the opinion described above and performing its related financial analyses, Goldman Sachs review

the merger

KKR's Registration Statement on Form S-1, including the prospectus contained therein dated September 16, 20: the initial public offering of KKR con

annual reports to unitholders or shareholders and Annual Reports on Form 10-K of KKR for the three December 31, 2012 and of KFN for the five years ended Decemb

certain interim reports to unitholders or shareholders and Quarterly Reports on Form 10-Q of KK

certain other communications from KKR and KFN to their respective unitholders or sl

certain publicly available research analyst reports for KK

certain financial analyses and forecasts for KFN prepared by its m

certain internal financial analysis and forecasts for KKR prepared by its management for the fourth quarter of financial analyses and forecasts for KKR prepared by research analysts and certain financial analyses and forec prepared by the management of KKR, in each case, as approved for Goldman Sachs' use by KKR, which are beginning on page 95 and are referred to in this proxy statement/prospectus as the Fo

certain cost savings and operating synergies projected by the management of KKR to result from the transaction, for Goldman Sachs' use by KKR, which are summarized beginning on page 93 and are referred to statement/prospectus as the

Goldman Sachs also held discussions with members of the senior managements of KKR and KFN regarding their assess strategic rationale for, and the potential benefits of, the transaction and the past and current business operations, financial condition prospects of their respective companies; reviewed the reported price and trading activity for KKR common units and KFN common compared certain financial and stock market information for KKR and KFN with similar information for certain other companies the which are publicly traded; reviewed the financial terms of certain recent business combinations in the permanent capital industry industries; and performed such other studies and analyses, and considered such other factors, as it deemed

For purposes of rendering the opinion described above, Goldman Sachs, with KKR's consent, relied upon and assumed the a completeness of all of the financial, legal, regulatory, tax, accounting and other information provided to, discussed with or Goldman Sachs, without assuming any responsibility for independent verification thereof. In that regard, Goldman Sachs assumed consent that the Forecasts and the Synergies are reasonable and re

Table

currently available estimates and judgments of the management of KKR. Goldman Sachs did not make an independent evaluation certain valuations of the CLO assets of KFN) or appraisal of the assets and liabilities (including any contingent, derive off-balance-sheet assets and liabilities) of KKR or KFN or any of their respective subsidiaries and Goldman Sachs was not furnish such evaluation or appraisal. Goldman Sachs assumed that all governmental, regulatory or other consents and approvals nece consummation of the transaction will be obtained without any adverse effect on KKR or KFN or on the expected benefits of the transaction will be consummated on the terms seemerger agreement, without the waiver or modification of any term or condition the effect of which would be in any way mean

Goldman Sachs' opinion does not address the underlying business decision of KKR to engage in the transaction or the relative transaction as compared to any strategic alternatives that may be available to KKR; nor does it address any legal, regulatory, tax of matters. Goldman Sachs' opinion addresses only the fairness from a financial point of view, as of the date of the opinion, of the expursuant to the merger agreement. Goldman Sachs' opinion does not express any view on, and does not address, any other term or merger agreement or the transaction or any term or aspect of any other agreement or instrument contemplated by the merger entered into or amended in connection with the transaction, including, without limitation, the fairness of the transaction to, or any of received in connection therewith by, the holders of any class of securities, creditors, or other constituencies of KKR; nor as to the fa amount or nature of any compensation to be paid or payable to any of the officers, directors or employees of KKR or KFN, or any persons, in connection with the transaction, whether relative to the exchange ratio pursuant to the merger agreement or otherwi Sachs does not express any opinion as to the prices at which KKR common units will trade at any time or as to the impact of the tr the solvency or viability of KKR or KFN or the ability of KKR or KFN to pay its obligations when they come due. Goldman Sachs' necessarily based on economic, monetary market and other conditions, as in effect on, and the information made available to it as of the opinion and Goldman Sachs assumed no responsibility for updating, revising or reaffirming its opinion based on ci developments or events occurring after the date of its opinion. Goldman Sachs' advisory services and its opinion were pro information and assistance of the board of directors of KKR's managing partner, in connection with its consideration of the transaction Sachs' opinion was approved by a fairness committee of Gol

The following is a summary of the material financial analyses delivered by Goldman Sachs to the board of directors of KKF partner in connection with rendering the opinion described above. The following summary, however, does not purport to b description of the financial analyses performed by Goldman Sachs, nor does the order of analyses described represent relative in weight given to those analyses by Goldman Sachs. Some of the summaries of the financial analyses include information present format. The tables must be read together with the full text of each summary and are alone not a complete description of Goldman Sac analyses. Except as otherwise noted, the following quantitative information, to the extent that it is based on market data, is based on as it existed on or before December 13, 2013 (the last trading day prior to the date of the merger agreement), and is not necessarily current market.

Stand-alone Financial Ana

Historical Stock Trac

Goldman Sachs reviewed the historical trading prices and volumes for KFN common shares for the five-year period ended D 2013. Goldman Sachs also reviewed the historical ratio of the daily closing market price of KFN common shares to the daily closing market price of KFN common shares t

Table

units for the three-year period ended December 13, 2013 and the historical ratio of the daily closing market price of KFN common book value per KFN common share for the five-year period ended December 13, 2013 and the historical ratio of the daily closing market price of KFN common share for the five-year period ended December 13, 2013 and the historical ratio of the daily closing market price of KFN common book value per KFN common share for the five-year period ended December 13, 2013 and the historical ratio of the daily closing market price of KFN common book value per KFN common share for the five-year period ended December 13, 2013 and the historical ratio of the daily closing market price of KFN common book value per KFN common share for the five-year period ended December 13, 2013 and the historical ratio of the daily closing market price of KFN common share for the five-year period ended December 13, 2013 and the historical ratio of the daily closing market price of KFN common share for the five-year period ended December 14, 2013 and 2013

Goldman Sachs analyzed the \$12.74 implied value of the consideration, based on the closing market price of KKR common December 13, 2013, to be paid per KFN common share pursuant to the merger agreement in relation to the current (last trading day announcement of the merger), 20 trading day volume weighted average, 30 trading day average and 52 week high closing market price of KKR common day announcement of the merger), 20 trading day volume weighted average, 30 trading day average and 52 week high closing market price of KKR common day are trading day average and 52 week high closing market price of KKR common day are trading day average and 52 week high closing market price of KKR common day are trading day average and 52 week high closing market price of KKR common day are trading day average and 52 week high closing market price of KKR common day are trading day average and 52 week high closing market price of KKR common day are trading day average and 52 week high closing market price of KKR common day average and 52 week high closing market price of KKR common day average and 52 week high closing market price of KKR common day average and 52 week high closing market price of KKR common day average and 52 week high closing market price of KKR common day average and 52 week high closing market price of KKR common day average and 52 week high closing market price of KKR common day average and 52 week high closing market price of KKR common day are traded and the common day are traded and traded and the common day are traded and traded an

This analysis indicated that the \$12.74 implied value of the consideration to be paid per KFN common share pursuant t agreement

a premium of 33.7% to the closing market price of KFN common shares on Decemb

a premium of 34.3% to the volume weighted average closing market price of KFN common shares over the 20 period ended Decemb

a premium of 33.9% to the average closing market price of KFN common shares over the 30 trading day

December 13

a premium of 9.6% to the high closing market price of KFN common shares over the 52 week period ended D

Selected Compar

Goldman Sachs reviewed and compared certain financial and stock market information, ratios and public market multiple corresponding financial and stock market information, ratios and public market multiples for the following publicly traded external business development companies, which are referred to in this proxy statement/prospectus as the KFN selected

Ares Capital

Prospect Capital

Apollo Investment

Fifth Street Fi

Solar

PennantPark Investment

Golub Capita

Blackrock Kelso Capital

	Edgar Filing: JAKKS PACIFIC INC - Form 10-Q
New Mountain Finance	
Medley Capital	
THI	
TICC	
MVC	
Fidus Investment	
GSV	
Solar Senio	
	64

Table

PennantPark Floating Rate

Gladstone Investment

Gladstone Capital

Stellus Capital Investment

NGP Capital Resource

Horizon Technology Finance

Although none of the KFN selected companies is directly comparable to KFN, the companies included were chosen because publicly traded companies with operations that for purposes of analysis may be considered similar to certain operations.

The financial and stock market information, ratios and public market multiples for KFN and the KFN selected companies w information Goldman Sachs obtained from publicly available historical data and Institutional Brokers' Estimate System, or "IBES The multiples and ratios were calculated using the applicable closing market prices as of December 13, 2013. The multiples and ratio and each of the KFN selected companies were based on the most recent publicly available information. With respect to KFN selected companies, Goldman Sach

the closing market price as a percentage of the high closing market price over the 52 week period ended Decemb

the ratio of the closing market price to the book value per share as of the latest fiscal

the annualized di

The results of these analyses are summarize

	Median of KFN	
	Selected Companies	KFN
% of 52 week high	93.1%	79.9%
Price / Book Value	1.02x	0.91x
Annualized Dividend Yield	9.2%	9.2%

Illustrative Dividend Disco

Goldman Sachs performed an illustrative dividend discount analysis on KFN on a stand-alone basis using KFN's projected of forth in the Forecasts to calculate a range of implied values per KFN common share. Goldman Sachs calculated illustrative termin KFN in the year 2018 by applying illustrative price-to-book value multiples ranging from 0.80x to 1.25x to KFN's projected book year 2018 per the Forecasts. These illustrative price-to-book multiples were derived by Goldman Sachs utilizing its professional ju experience, taking into account current and historical trading data and the current price-to-book multiple for KFN and the K companies. Goldman Sachs then discounted KFN's estimated dividends for the fourth quarter of 2013 and each of the calendar through 2018 and illustrative terminal values to derive illustrative present values of equity as of September 30, 2013 by assume convention. Goldman Sachs used a range of discount rates from 8.0% to 11.0%, representing estimates of KFN's cost of equity. Go then divided such illustrative present values by the number of KFN common shares on a fully diluted basis as of September 30, 2013 the illustrative per-share equity values. This resulted in a range of implied values of KFN common shares from \$9.13 to \$13.

Stand-alone Financial Ana

Illustrative Dividend Disco

Goldman Sachs performed an illustrative dividend discount analysis on KKR on a stand-alone basis using KKR's project distributed earnings set forth in the Forecasts to calculate a range of implied values per KKR common unit. Goldman Sachs als illustrative terminal values for KKR in the year 2015 by applying illustrative last-twelve-months distributed earnings exit multiples in 16.0x to 22.0x to KKR's projected last-twelve-months distributed earnings in the year 2015 per the Forecasts. Thes last-twelve-months distributed earnings exit multiples were derived by Goldman Sachs utilizing its professional judgment and experint account current and historical trading data and the last-twelve-months distributed earnings multiple for KKR and selected compactive exhibited similar business characteristics to KKR, which are referred to in this proxy statement/prospectus as the KKR selected Goldman Sachs then discounted KKR's estimated after-tax distributed earnings from the fourth quarter of 2013 and each of the cacculate and 2015 and illustrative terminal values to illustrative present values of equity as of September 30, 2013 by assume convention. Goldman Sachs used a range of discount rates from 11.5% to 13.0% representing estimates of KKR's cost of equity. Go then divided such illustrative present values of equity by the number of KKR common units on a fully diluted basis as of September calculate the illustrative per-share equity values. This resulted in a range of implied equity values of KKR common units from \$21.

Illustrative Sum-of-the-P

Goldman Sachs performed an illustrative sum-of-the-parts analysis to calculate a range of illustrative equity values of KKR on a basis by adding illustrative values of (1) KKR's book value (for purposes of this section of this proxy statement/prospectus, su defined in the 10-Q for the most recent fiscal quarter prior to announcement of the transaction, which differs from KKR's total par used elsewhere in this proxy statement/prospectus on a GAAP basis primarily as a result of the exclusion of ownership interests at KKR Holdings), (2) KKR's future fee-related earnings and (3) KKR's future net carried interest. Goldman Sachs calculated the illust of KKR's book value by applying a book value multiple of 1.0x to KKR's book value as of September 30, 2013. Goldman Sach illustrative values of KKR's future fee-related earnings by applying illustrative fee-related earnings multiples ranging from 15.0 KKR's projected after-tax fee-related earnings for the year 2014 per the Forecasts. Goldman Sachs calculated illustrative values of I net carried interest by applying illustrative net carried interest multiples ranging from 5.0x to 9.0x to KKR's projected after-tax interest for the year 2014 per the Forecasts. The illustrative book value multiple, fee-related earnings multiples and net carried interest were derived by Goldman Sachs utilizing its professional judgment and experience, taking into account current and historical trace. Wall Street analysts' estimates of book multiples, fee-related earnings multiples and net carried interest multiples, respectively, for the for KKR and the KKR selected companies. This resulted in a range of implied values of KKR common units from \$22.65 to \$27.

Pro Forma Financial Analysis of the Combin

Contribu

Goldman Sachs calculated the illustrative pro forma relative contributions of KKR and KFN to the combined company we certain financial metrics using the Forecasts for KKR and KFN for 2014 and 2015. Associated pro forma equity ownership was deri relative contributions based on the total number of KKR common units and KKR Group Partnership units outstanding on a fully dil of September 30, 2013 and KKR common units to be issued in connection with the merger based on KFN common shares outstanding diluted basis as of Section 1.

2013, and the exchange ratio of 0.51 KKR common units for each KFN common share. The following table sets forth the results of

Pro Forma Contribution Analysis (\$ in millions) Income Statement (After-tax)

	KKR		KF	N		
	\$	%	\$	%	,	Fotal
2014						
Fee Related Earnings	\$ 358	100.0%	\$ 0	0.0%	\$	358
Incentive Earnings	728	100.0	0	0.0		728
Investment Earnings	696	76.0	220	24.0		916
Economic Net Income	\$ 1,782	89.0%	\$ 220	11.0%	\$	2,002
Distributed Earnings	\$ 944	85.6%	\$ 159	14.4%	\$	1,103
2015						
Fee Related Earnings	\$ 402	100.0%	\$ 0	0.0%	\$	402
Incentive Earnings	842	100.0	0	0.0		842
Investment Earnings	747	74.8	252	25.2		999
Economic Net Income	\$ 1,990	88.8%	\$ 252	11.2%	\$	2,242
Distributed Earnings	\$ 1,072	85.1%	\$ 188	14.9%	\$	1,260

Balance Sheet

	KKI	R	K	FN	
	\$	%	\$	%	Total
30-Sep-2013					
Book Value	\$ 7,208	77.2%	\$ 2,13	3 22.8%	\$ 9,341

	KKR Common		KFN Common	n	
	Unitholders		Shareholders		
Pro Forma Ownership	(in millions)	%	(in millions)	%	Total
30-Sep-2013					
Common Units/Shares Outstanding	715.8	87.2%	105.1	12.8%	820.9

Accretion/(Diluta

Goldman Sachs performed an accretion/dilution analysis of the illustrative pro forma financial effect of the proposed transacti estimated after-tax distributed earnings using the Forecasts for KKR and KFN for the years 2014 and 2015 and taking into account the and assuming a transaction date of December 31, 2013 and transaction fees of approximately \$40 million per KKR management. indicated that the transaction would have an accretive illustrative pro forma financial effect on the after-tax distributed earnings of years 2014 and 2015 in the amounts of \$0.11 (or 7.8%) and \$0.11 (or 7.6%),

Illustrative Sum-of-the-P

Goldman Sachs performed a sum-of-the-parts analysis to calculate a range of illustrative equity values of the combined comp forma basis, based upon the Forecasts and Synergies, by adding illustrative values of (1) the combined company's book value, (2) to company's future fee-related earnings and (3) the combined company's future net carried interest. Goldman Sachs calculated the illus of the combined company's book value by applying a book value multiple of 1.0x to the combined company's book value as of Sc 2013. Go

calculated illustrative values of the combined company's future fee-related earnings by applying illustrative fee-related earning from 15.0x to 17.0x to the combined company's projected after-tax fee-related earnings for the year 2014 per the Forecasts account the Synergies. Goldman Sachs calculated illustrative values of the combined company's future net carried interest illustrative net carried interest multiples ranging from 5.0x to 9.0x to the combined company's projected after-tax net carried interest 2014 per the Forecasts, taking into account the Synergies. The illustrative book value multiple, illustrative fee-related earnings in illustrative net carried interest multiples for the combined company were derived by Goldman Sachs utilizing its professional juty experience, taking into account current and historical trading data and Wall Street analysts' estimates of book value multiples earnings multiples and net carried interest multiples for the year 2014 for KKR and the KKR selected companies. This resulted implied values of common units of the combined company from \$21.68 to \$26

Analysis at Va

Based on the Forecasts for KKR and KFN in the years 2014 and 2015, the Synergies, and an illustrative range of pro for common units of the combined company ranging from \$22.00 per share to \$28.00 per share, Goldman Sachs calculated (i) the in and 2015E after-tax economic net income multiple based on projected pro forma after-tax economic net income for the combined 2014 and 2015, respectively, and (ii) the implied 2014E and 2015E after-tax distributed earnings multiple based on projected pro for distributed earnings for the combined company for 2014 and 2015, respectively. The following table sets forth the results of

Pro Forma Analysis at Various Prices (\$ millions, except per share data)

Pro forma KKR common unit price	\$ 22.00	\$ 24.00	\$ 26.00	\$ 28.00
Premium to current unit price (\$24.98)	(11.9)%	(3.9)%	4.1%	12.1%

Pro Forma Accretion / (Dilution) KKR Metric						
After-tax economic net income (\$)						
2014E	\$	2.43	9.1x	9.9x	10.7x	11.5x
2015E		2.68	8.2	9.0	9.7	10.4
After-tax distributed earnings (\$)						
2014E	\$	1.46	15.1x	16.4x	17.8x	19.2x
20155	Φ	1.60	12.5	14.7	15.0	17.1
2015E	\$	1.63	13.5	14.7	15.9	17.1

Illustrative Dividend Disco

Goldman Sachs performed an illustrative dividend discount analysis of the combined company on a pro forma basis using KKI after-tax distributed earnings and KFN's projected dividends set forth in the Forecasts and taking into account the Synergies to calc of implied values per common unit of the combined company. Goldman Sachs calculated illustrative terminal values for the combine in the year 2015 by applying illustrative last-twelve-months distributed earnings exit multiples ranging from 16.0x to 22.0x to to company's projected last-twelve-months distributed earnings in the year 2015 per the Forecasts. These illustrative last-twelve-months earnings exit multiples were derived by Goldman Sachs utilizing its professional judgment and experience, taking into account historical trading data and the last-twelve-months distributed earnings multiple for KKR and the KKR selected companies. Goldman

<u>Table</u>

discounted the combined company's estimated after-tax distributed earnings from the fourth quarter of 2013 and each of the calenda and 2015 and illustrative terminal values to illustrative present values of equity as of September 30, 2013 by assuming mid-year Goldman Sachs used a range of discount rates from 11.5% to 13.0%, representing estimates of the combined company's cost of equity Sachs then divided such illustrative present values of equity by the number of common units of the combined company outstandid diluted basis as of September 30, 2013 (based on the total number of KKR common units and KKR Group Partnership units outs fully diluted basis as of September 30, 2013 and KKR common units to be issued in connection with the merger based on KFN contoutstanding on a fully diluted basis as of September 30, 2013, and the exchange ratio of 0.51 KKR common units for each KFN contoutstanding on a fully diluted basis as of September 30, 2013, and the exchange ratio of 0.51 KKR common units for each KFN contoutstanding on a fully diluted basis as of September 30, 2013, and the exchange ratio of 0.51 KKR common units for each KFN contoutstanding on a fully diluted basis as of September 30, 2013, and the exchange ratio of 0.51 KKR common units for each KFN contoutstanding on a fully diluted basis as of September 30, 2013, and the exchange ratio of 0.51 KKR common units for each KFN contoutstanding on a fully diluted basis as of September 30, 2013, and the exchange ratio of 0.51 KKR common units for each KFN contoutstanding on a fully diluted basis as of September 30, 2013, and the exchange ratio of 0.51 KKR common units for each KFN contoutstanding on a fully diluted basis as of September 30, 2013, and the exchange ratio of 0.51 KKR common units for each KFN contoutstanding on a fully diluted basis as of September 30, 2013, and the exchange ratio of 0.51 KKR common units for each KFN contoutstanding on a fully diluted basis as of September 30, 2013, and the exchange ratio of 0.51 KKR common units for each KFN cont

Illustrative Dividend Discount-Based Contribu

Goldman Sachs then calculated the implied pro forma ownership of KKR common unitholders in the combined company be illustrative present values of equity of KFN on a stand-alone basis to illustrative present values of equity of the combined company forma basis, based on the Forecasts and taking into account the Synergies. Goldman Sachs performed illustrative dividend discount KFN and the combined company as described above under "Stand-alone Financial Analysis of KFN Illustrative Dividend Discount "Pro Forma Financial Analysis of the Combined Company Illustrative Dividend Discount Analysis," respectively, except that used a discount rate of 12.4% for the combined company and 9.0% for KFN, which represents an illustrative cost of equity for the combined company and KFN, respectively. Goldman Sachs then subtracted the illustrative present values of equity of KFN on a basis from the illustrative present values of equity of the combined company on a pro forma basis to obtain illustrative present values KKR on a stand-alone basis. This resulted in a range of implied pro forma ownership of KKR common unitholders in the combined from 85.5% to 91.8%, as compared to the pro forma ownership of KKR common unitholders in the combined company of 87.2% total number of KKR common units and KKR Group Partnership units outstanding on a fully diluted basis as of September 30, 20 common units to be issued in connection with the merger based on KFN common shares outstanding on a fully diluted basis as of September 30, 20 common units to be issued in connection with the merger based on KFN common shares outstanding on a fully diluted basis as of September 30, 20 common units to be issued in connection with the merger based on KFN common shares outstanding on a fully diluted basis as of September 30, 20 common units for each KFN common units for each KFN

The preparation of a fairness opinion is a complex process and is not necessarily susceptible to partial analysis or summary Selecting portions of the analyses or of the summary set forth above, without considering the analyses as a whole, could create at view of the processes underlying Goldman Sachs' opinion. In arriving at its fairness determination, Goldman Sachs considered the of its analyses and did not attribute any particular weight to any factor or analysis considered by it. Rather, Goldman Sachs determination as to fairness on the basis of its experience and professional judgment after considering the results of all of its company or transaction used in the above analyses as a comparison is directly comparable to KKR or KFN or the contemplated

Goldman Sachs prepared these analyses for purposes of Goldman Sachs' providing its opinion to the board of directors of KKF partner as to the fairness from a financial point of view of the exchange ratio pursuant to the merger agreement These analyses do n be appraisals nor do they necessarily reflect the prices at which businesses or securities actually may be sold. Analyses based upon future results are not necessarily indicative of actual future results, which may be significantly more or less favorable than suggest analyses. Because these analyses are inherently subject to uncertainty, being based upon numerous factors or event

Table

control of the parties or their respective advisors, none of KKR, KFN, Goldman Sachs or any other person assumes responsibility if if are materially different from the

The exchange ratio was determined through negotiations between KKR and KFN and was approved by the board of director managing partner. Goldman Sachs provided advice to KKR during these negotiations. Goldman Sachs did not, however, recommend exchange ratio to KKR or the board of directors of KKR's managing partner or that any specific exchange ratio constituted the only exchange ratio for the

As described above, Goldman Sachs' opinion to the board of directors of KKR's managing partner was one of many factor consideration by the board of directors of KKR's managing partner in making its determination to approve the merger agreement. The summary does not purport to be a complete description of the analyses performed by Goldman Sachs in connection with its optimized in its entirety by reference to the written opinion of Goldman Sachs attached as Annex C to this proxy statement.

Goldman Sachs and its affiliates are engaged in advisory, underwriting and financing, principal investing, sales and tradi investment management and other financial and non-financial activities and services for various persons and entities. In the ordinary these activities, Goldman Sachs and its affiliates and employees, and funds or other entities in which they invest or have other econo or with which they co-invest, may at any time purchase, sell, hold or vote long or short positions and investments in securities, deriv. commodities, currencies, credit default swaps and other financial instruments of KKR and its portfolio companies, KFN and respective affiliates and third parties, or any currency or commodity that may be involved in the transaction for the accounts of Go and its affiliates and employees and their customers. Goldman Sachs acted as financial advisor to KKR in connection with, and particularly actions and their customers. certain of the negotiations leading to, the transaction. Goldman Sachs expects to receive fees for its services in connection with the all of which are contingent upon consummation of the transaction, and KKR has agreed to reimburse Goldman Sachs' expenses indemnify Goldman Sachs against certain liabilities that may arise, out of Goldman Sachs' engagement. Goldman Sachs also has pro investment banking services to KKR and its affiliates and portfolio companies from time to time for which the Investment Banking Goldman Sachs has received, and may receive, compensation, including having acted as joint book-running manager with respe offering of 30,000,000 shares of common stock of Dollar General Corporation, which is referred to in this proxy statement/prospec General, a portfolio company of KKR, in June 2012; as joint book-running manager with respect to a public offering of 36,000, common stock of Dollar General in September 2012; as joint book-running manager with respect to an offering of 6.5% senior no (aggregate principal amount \$825,000,000) and 6.5% senior subordinated notes due 2020 (aggregate principal amount \$800,000). Biomet, Inc., a portfolio company of KKR, in September 2012; as joint book-running manager with respect to a public offering o shares of common stock of Nielsen Holdings N.V., which is referred to in this proxy statement/prospectus as Nielsen, a portfolio KKR, in February 2013; as joint book-running manager with respect to a public offering of 30,000,000 shares of common sto General in March 2013; as joint book-running manager with respect to a public offering of Dollar General's 1.875% senior no (aggregate principal amount \$400,000,000) and Dollar General's 3.250% senior notes due 2023 (aggregate principal amount \$900 April 2013; and as joint book-running manager with respect to a public offering of 35,000,000 shares of common stock of Nielsen i During the two year period ended December 16, 2013, the Investment Banking Division of Goldman Sachs has received companies to the companies of the sachest period ended December 16, 2013, the Investment Banking Division of Goldman Sachs has received companies to the companies of the companies o financial advisory and/or underwriting services provided directly to KKR and/or to its affiliates and portfolio companies (which companies that are not controlled by KKR) of approximately \$144 million. During the two year period ended December 16, 20 Sachs has not been engaged by KFN or its subsidiaries to provide financial advisory or underwriting services for which the Investn Division of Goldman Sachs has received co

Table

Goldman Sachs may also in the future provide investment banking services to KKR, KFN and their respective affiliates and portfolion for which the Investment Banking Division of Goldman Sachs may receive compensation. Goldman Sachs may also in the future investment banking services to KKR, KFN and their respective affiliates and portfolion companies for which the Investment Banking Goldman Sachs may receive compensation. Affiliates of Goldman Sachs also may have co-invested with KKR and its affiliates from and may have invested in limited partnership units of affiliates of KKR from time to time and may do so

On October 7, 2013, Mr. Lewin contacted Goldman Sachs to solicit Goldman Sachs' interest in acting as financial advisc connection with a potential transaction with KFN. KKR approached Goldman Sachs because Goldman Sachs is an internationally investment banking firm that has substantial experience in transactions similar to the transaction and because, as described about Sachs had been retained by KKR and its affiliates on several prior occasions, including in connection with the KPE Transaction and high degree of familiarity with KKR's structure and its business. At KKR's request, Goldman Sachs advised to date of Goldman Sachs' inquiry, based on Goldman Sachs' internal records, no senior member of its investment banking tear participate as an advisor to KKR in connection with the potential transaction with KFN owned any direct interests in KKR or aggregate value in excess of \$10,000. In addition, Goldman Sachs advised that, as of the date of Goldman Sachs' inquiry, each member who is permitted to maintain accounts outside of Goldman Sachs has represented to Goldman Sachs that such person does direct holdings in KKR or KFN in any such outside accounts. Following discussions with Goldman Sachs, the management of K partner considered the qualifications, experience and reputation of Goldman Sachs, and determined to retain Goldman Sachs as KK advisor in connection with the potential transaction with KFN if an acceptable engagement letter could be negotiated with Gol Goldman Sachs was formally retained on December of the goldman Sachs was formally retained on December of the goldman Sachs was formally retained on December of the goldman Sachs was formally retained on December of the goldman Sachs was formally retained on December of the goldman Sachs was formally retained on December of the goldman Sachs was formally retained on December of the goldman Sachs was formally retained on December of the goldman Sachs was formally retained on December of the goldman Sachs was formally retained on December of the goldman Sa

Pursuant to the terms of an engagement letter dated December 13, 2013 among KKR, KKR's managing partner, Koh Roberts & Co. L.P. and Goldman Sachs, KKR has agreed to pay Goldman Sachs a transaction fee of \$5 million, all of which is partner consummation of the transaction. In addition, KKR has agreed to reimburse Goldman Sachs for its expenses, including attorn disbursements, and to indemnify Goldman Sachs and related persons against various liabilities, including certain liabilities under

Other Presentations by Gol

In addition to the presentation delivered to the board of directors of KKR's managing partner on December 16, 2013, as desc. Goldman Sachs also delivered presentations to the board of directors of KKR's managing partner or the management of KKI October 17, 2013, October 29, 2013, November 6, 2013, November 25, 2013, December 10, 2013 and December 15, 2013. Co presentations have been filed as exhibits to the Schedule 13E-3 filed with the SEC in connection with the transaction and will be ma for inspection and copying at the principal executive offices of KKR during its regular business hours by any interested holder of K units. Copies may be obtained by requesting them in writing from KKR at the address provided in the section titled "Where You Ca Information" beginning on page 234 of this proxy statement/prospectus. None of these presentations by Goldman Sachs, alone constitute, or form the basis of, an opinion of Goldman Sachs with respect to the exchange ratio pursuant to the merge

October 17, 2013

Selected Compar

Goldman Sachs reviewed and compared certain financial and stock market information, ratios and public market multiple corresponding financial and stock market informatio

Table

public market multiples for the following publicly traded commercial specialty finance companies, which are referred to statement/prospectus as the CSC selected

Ares Capital

Prospect Capital

Apollo Investment

Fifth Street F

CI

NewStar F

Although none of the CSC selected companies is directly comparable to KFN, the companies included were chosen becapublicly traded companies with operations that for purposes of analysis may be considered similar to certain operations.

The financial and stock market information, ratios and public market multiples for KFN and the CSC selected companies w information Goldman Sachs obtained from publicly available historical data and IBES estimates. The multiples and ratios were calc the applicable closing market prices as of October 16, 2013. The multiples and ratios of KFN and each of the CSC selected combased on the most recent publicly available information. With respect to KFN and the CSC selected companies, Goldman Sach

the closing market price as a percentage of the high closing market price over the 52 week period ended Octob

the ratio of the closing market price to the book value per share as of the latest fiscal

the annualized di

The results of these analyses are summarize

	Medi	an	of	CS	C	
S	elected	10	on	าทล	nie	

	ociccica Companies	171.11
% of 52 week high	92.7%	91.7%
Price/Book Value	1.08x	1.02x
Annualized Dividend Yield	9.2%	7.9%

Goldman Sachs also reviewed and compared certain financial and stock market information, ratios and public market multiple corresponding financial and stock market information, ratios and public market multiples for the following publicly traded resident real estate investment trusts, which are referred to in this proxy statement/prospectus as the REIT selected

Annaly Capital Mana

American Capital A

Edgar Filing: JAKKS PACIFIC INC - Form 10-Q Hatteras Fin New Residential Inves ARMOUR Residentia CYS Inves Capstead Mortgage Anworth Mortgage Asset Western Asset Mortgage Capital Two Harbors Inve 72

<u>Table</u>

Chimera Investment

MFA Fi

Invesco Mortgage

Newcastle Inves

PennyMac Mortgage Inves

Redwoo

American Capital Mortgage Inves

Dynex

Apollo Residential M

AG Mortgage Investme

New York Mortgag

Although none of the REIT selected companies is directly comparable to KFN, the companies included were chosen becapublicly traded companies with operations that for purposes of analysis may be considered similar to certain operations.

The financial and stock market information, ratios and public market multiples for KFN and the REIT selected companies we information Goldman Sachs obtained from publicly available historical data. The multiples and ratios were calculated using the closing market prices as of October 16, 2013. The multiples and ratios of KFN and each of the REIT selected companies were based recent publicly available information. With respect to KFN and the REIT selected companies, Goldman Sach

the closing market price as a percentage of the high closing market price over the 52 week period ended Octob

the ratio of the closing market price to the book value per share as of the latest fiscal

the annualized di

The results of these analyses are summarize

	Median of REIT	
	Selected Companies	KFN
% of 52 week high	72.4%	91.7%
Price/Book Value	0.89x	1.02x
Annualized Dividend Yield	12.2%	7.9%

Goldman Sachs also reviewed and compared certain financial and stock market information, ratios and public market multiples corresponding financial and stock market information, ratios and public market multiples for the following publicly traded alter managers, which are referred to in this proxy statement/prospectus as the AAM selected

The Blackstone

Apollo Global Manag

The Carlyle

Oaktree Capital

Och-Ziff Capital Management

Fortress Investment

Sprott Asset Mai

<u>Table</u>

Par

Ashmor

GAM 1

Ma

Polar Capital I

Gottex Fund Management Hold

Although none of the AAM selected companies is directly comparable to KKR, the companies included were chosen becapublicly traded companies with operations that for purposes of analysis may be considered similar to certain operations.

The financial and stock market information, ratios and public market multiples for KKR and the AAM selected companies we information Goldman Sachs obtained from publicly available historical data. The multiples and ratios were calculated using the closing market prices as of October 16, 2013. The multiples and ratios of KKR and each of the AAM selected companies were based recent publicly available information. With respect to KKR and the AAM selected companies, Goldman Sachs

the closing market price as a percentage of the high closing market price over the 52 week period ended Octob

price as a multiple of estimated 2013 earnings per share and estimated 2014 earnings per share for each of the A

the annualized di

The results of these analyses are summarize

	Median of AAM	
	Selected Companies	KKR
% of 52 week high	94.2%	100%
2013 Calendarized P/E Multiple	13.5x	9.9x
2014 Calendarized P/E Multiple	12.2x	9.2x
Annualized Dividend Yield	4.6%	5.7%

Selected Transacti

Goldman Sachs also analyzed certain publicly available information relating to the following selected transactions in the RE development company, and publicly traded private equity vehicle spaces

Date Announced	Target	Acquiror
September 2013	Thomas Properties Group, Inc.	Parkway Properties, Inc.
June 2013	Colonial Properties Trust	Mid-America Apartment Communities, Inc.
May 2013	CapLease, Inc.	American Realty Capital Properties, Inc.
November 2012	CerXus Investment Corp.	Annaly Capital Management, Inc.
September 2012	American Realty Capital Trust, Inc.	Realty Income Corporation
July 2012	Conversus Capital, L.P.	HarbourVest Partners, LLC

December 2011 Cogdell Spencer Inc. Ventas, Inc.

April 2011 Absolute Private Equity Ltd. HarbourVest Partners, LLC

February 2011 Nationwide Health Properties, Inc. Ventas, Inc.

January 2011 ProLogis, Inc. AMB Property Corporation

For each of the selected transactions, Goldman Sachs calculated and compared the implied offer premium to the target comp market price one trading day prior to announcement of the transaction, which ranged from -0.2% to 19.7% with a median of 6.7%. We have the compared transaction of the transacti

Table

that participated in the selected transactions are directly comparable to KFN, the companies that participated in the selected transcompanies with operations that, for the purposes of analysis, may be considered similar to certain of KFN's results, market size, proposed or structure.

Accretion/(Diluta

Goldman Sachs also performed an accretion/dilution analysis of the illustrative pro forma financial effect of a hypothetical a KFN by KKR at illustrative premia to KFN's closing market price as of October 16, 2013 ranging from 0% to 20%, or an implied pt of \$10.65 to \$12.78, on KKR's estimated after-tax distributed earnings using publicly available KFN and KKR financial statements a 2013 and IBES estimates, and assuming no synergies, 100% stock consideration, a transaction date of December 31, 2013 and transa approximately \$30 million to \$35 million. The analysis indicated that the transaction would have a dilutive illustrative pro forma fin on the after-tax distributed earnings of KKR for the year 2014 ranging from \$0.04 (or 2.8%) to \$0.08 (or 5.1%) and a dilutive illustrative professional forma financial effect on the after-tax distributed earnings of KKR for the year 2015 ranging from \$0.06 (or 3.2%) to \$0.06.

The October 17, 2013 presentation also contained an analysis of potential balance sheet adjustments, market performa shareholder analysis and analyses of the pro forma impact of the acquisition on KKR's credit rating and on various finance.

October 29, 2013, November 6, 2013, November 25, 2013, December 10, 2013 H

The October 29, 2013 presentation contained an analysis of KKR raising equity, in both a single offering and in multiple o acquiring assets in the open market as an alternative to an acquisition of KFN. The analysis examined the potential incremental total earnings per issued KKR common unit, incremental economic net income per issued KKR common unit and total dividends per K unit, relative to an acquisition of KFN at a 0.46 exchange ratio of KKR common units per KFN common units per KFN

The November 6, 2013 presentation contained a qualitative discussion of capital markets-related considerations in connection balance sheet capital from selected alternative sources including: the issuance of equity in multiple tranches; the issuance of debt; approach of acquiring assets with capital raised from a line of credit and repaying with the proceeds of equity issuance.

The November 25, 2013 presentation contained a comparison of the investor bases comprising KFN and KKR and a discussion reactions of KFN shareholders and KKR common unitholders to the contemplated

The December 10, 2013 presentation contained an illustrative sensitivity analyses with respect to the value of CLOs and optim in each case based on underlying portfolio a

The financial analyses and other information in these presentations were based on market, economic and other condition respective dates as well as other information that was available at those times. Accordingly, the results of the financial analysi information differed due to changes in those conditions. Among other things, multiples attributable to selected companies charcompanies' stock prices changed, and implied transaction multiples changed as KKR's and KFN's financial results (as well as the changed. Finally, Goldman Sachs continued to refine various aspects of its financial analyses with respect to KKR and KF

Opinion of the Financial Advisor to the Conflicts Committee of KKR's Manag

On December 16, 2013, Lazard rendered its oral opinion, subsequently confirmed in writing, to the independent directors conconflicts committee of the board of directors of KKF

Table

partner, that, as of such date, and based upon and subject to the assumptions, procedures, factors, qualifications and limitations set the exchange ratio was fair, from a financial point of vi

The full text of Lazard's written opinion, dated December 16, 2013, which sets forth the assumptions made, procedur factors considered and qualifications and limitations on the review undertaken by Lazard in connection with its opinion, is this proxy statement/prospectus as Annex D and is incorporated into this proxy statement/prospectus by reference. The december 15 proxy statement/prospectus is qualified in its entirety by reference to the full text of Lazard's opinion attached as Annex D. You are encouraged to read Lazard's opinion and this section carefully and in the

Lazard's opinion was directed to the independent directors constituting the conflicts committee of the board of director managing partner in connection with their evaluation of the transactions contemplated by the merger agreement and only act fairness, from a financial point of view, of the exchange ratio to KKR. Lazard's opinion was not intended to, and does not recommendation to any shareholder as to how such shareholder should vote or act with respect to the merger or any mat thereto. Lazard's opinion was necessarily based on economic, monetary, market and other conditions as in effect on, and the made available to it as of, the date of Lazard's opinion. Lazard did not express any opinion as to the prices at which KKR co or KFN common shares may trade at any time subsequent to the announcement of the merger. Lazard was retained only to opinion and was not involved in the negotiation of the terms of the merger. Lazard's opinion did not address the relative merger as compared to any other transaction or business strategy in which KKR might engage or the merits of the underly by KKR to engage in

The following is a summary of Lazard's opinion. You are encouraged to read Lazard's written opinion carefully in

In connection with its opin

reviewed the financial terms and conditions of the merger

reviewed certain publicly available historical business and financial information relating to KFI

reviewed various financial forecasts and other data provided to Lazard by KFN and KKR relating to the business certain limited financial forecasts and various other data provided to Lazard by KKR relating to the business also reviewed certain publicly available financial forecasts relating to the business of KFN

held discussions with members of the senior managements of KFN and KKR's managing partner with a businesses and prospects of KFN and KKR, respectively, and reviewed certain synergies and other benefits a the management of KKR's managing partner to be realized from

reviewed public information with respect to certain other companies in lines of business Lazard believed to relevant in evaluating the businesses of KFN and KKR, r

reviewed the financial terms of certain business combinations involving companies in lines of business Lazard b generally relevant in evaluating the businesses of KFN and KKR, r

reviewed historical prices and trading volumes of KFN common shares and KKR con

reviewed the potential pro forma financial impact of the merger on KKR based on the financial forecasts refer relating to KFN and

conducted such other financial studies, analyses and investigations as Lazard deemed

<u>Table</u>

Lazard assumed and relied upon the accuracy and completeness of the foregoing information, without independent verifical information. Lazard did not conduct any independent valuation or appraisal of any of the assets or liabilities (contingent or otherwise KKR or concerning the solvency or fair value of KFN or KKR, and Lazard was not furnished with any such valuation or appraisal. It of KKR's managing partner advised Lazard that (i) financial forecasts prepared by KKR with respect to the future financial performance reflected the best currently available estimates and judgments of the anticipated future financial performance of KFN and (ii) public forecasts with respect to KKR reflected the best currently available estimates and judgments of the anticipated future financial performance of KFN. Accordingly, with the consent of the independent directors constituting the conflicts committee of the board of directors analysis of KFN and KKR, Lazard utilized such financial forecasts. Lazard assumed, with of the independent directors constituting the conflicts committee of the board of directors of KKR's managing partner, that the financial in Lazard's analyses, including those related to synergies and other benefits, had been reasonably prepared on bases reflect currently available estimates and judgments as to the future financial performance of KFN and KKR,

In rendering its opinion, Lazard assumed, with the consent of the independent directors constituting the conflicts committee of directors of KKR's managing partner, that the merger will be consummated on the terms described in the merger agreement, without or modification of any material terms or conditions. Lazard also assumed, with the consent of the independent directors constituting committee of the board of directors of KKR's managing partner, that obtaining the necessary governmental, regulatory or third part and consents for the merger will not have an adverse effect on KKR, KFN or the merger. Lazard did not express any opinion as other consequences that might result from the merger, nor did Lazard's opinion address any legal, tax, regulatory or accounting resulting the conflicts committee of the board of directors of KKR's managements of the merger and described in its opinion. Lazard expressed no view or opinion as to any teaspects (other than the exchange ratio to the extent expressly specified in its opinion) of the merger, including, without limitation structure of the merger or any agreements or arrangements entered into in connection with, or contemplated by, the merger. In add expressed no view or opinion as to the fairness of the amount or nature of, or any other aspects relating to, the compensation to directors or employees of any parties to the merger, or class of such persons, relative to the exchange ratio of directors or employees of any parties to the merger, or class of such persons, relative to the exchange ratio of the merger.

Table

The following is a brief summary of the material financial analyses and reviews that Lazard deemed appropriate in con rendering its opinion. The brief summary of Lazard's analyses and reviews provided below is not a complete description of the reviews underlying Lazard's opinion. The preparation of a fairness opinion is a complex process involving various determinations a appropriate and relevant methods of analysis and review and the application of those methods to particular circumstances, and, the readily susceptible to summary description. Considering selected portions of the analyses and reviews or the summary set forth be considering the analyses and reviews as a whole, could create an incomplete or misleading view of the analyses and reviews underly

In arriving at its opinion, Lazard considered the results of all of its analyses and reviews and did not attribute any particular v factor, analysis or review considered by it; rather, Lazard made its determination as to fairness on the basis of its experience and judgment after considering the results of all of its analyses

For purposes of its analyses and reviews, Lazard considered industry performance, general business, economic, market a conditions and other matters, many of which are beyond the control of KKR and KFN. No company, business or transaction use analyses and reviews as a comparison is identical to KKR, KFN, or the transactions contemplated by the merger agreement, and an at the results of those analyses and reviews is not entirely mathematical. Rather, the analyses and reviews involve complex consict judgments concerning financial and operating characteristics and other factors that could affect the acquisition, public trading or oth the companies, businesses or transactions used in Lazard's analyses and reviews. The estimates contained in Lazard's analyses and the ranges of valuations resulting from any particular analysis or review are not necessarily indicative of actual values or predict results or values, which may be significantly more or less favorable than those suggested by Lazard's analyses and reviews. In additional reviews relating to the value of companies, businesses or securities do not purport to be appraisals or to reflect the pricompanies, businesses or securities actually may be sold. Accordingly, the estimates used in, and the results derived from, Lazard's reviews are inherently subject to substantial

The summary of the analyses and reviews provided below includes information presented in tabular format. In order to full, Lazard's analyses and reviews, the tables must be read together with the full text of each summary. The tables alone do not constitut description of Lazard's analyses and reviews. Considering the data in the tables below without considering the full description of the reviews, including the methodologies and assumptions underlying the analyses and reviews, could create a misleading or incompared to the control of the reviews, and reviews, and reviews, could create a misleading or incompared to the control of the reviews, and reviews, could create a misleading or incompared to the control of the reviews, and reviews, could create a misleading or incompared to the control of the control

Except as otherwise noted, the following quantitative information, to the extent that it is based on market data, is base

KFN Stand-Alo

KFN Selected Comparable Compar

Lazard reviewed and analyzed the following public companies that it viewed as reasonably comparable to KFN: (i) selected managed business development companies, or "BDCs," (ii) selected internally managed BDCs, (iii) selected commercial mortgage investment trusts, or "REITs," and (iv) selected residential mortgage REITs. Although none of the selected companies is directly companies included are publicly traded companies with operations and/or other criteria, such as lines of business, mark risks and size and scale

Edgar Filing: JAKKS PACIFIC INC - Form 10-Q
<u>Table</u>
which for purposes of analysis Lazard considered relevant. Specifically, Lazard compared KFN to the following public
Externally mans
Ares Capital
Prospect Capital
Apollo Investment
Fifth Street Fi
Solar
PennantPark Investment
BlackRock Kelso Capital
MVC (
Gladstone Investment
Gladstone Capital
Internally mana
Onex
American C
Main Street Capital
Triangle Capital
MCG Capital

Commercial mortg

Starwood Propert Newcastle Inves Colony Fi iStar Fi Blackstone Mortgag Resource C RAIT Fin Apollo Commercial Real Estate F Ares Commercial Real Estate Arbor Realt Resource A BRT Residential Mortg Annaly Capital Mana 79

<u>Table</u>

American Capital A

Two Harbors Inves

Chimera Investment

Invesco Mortgage

MFA Fi

Hatteras Fin New Residential Inves PennyMac Mortgage Inves Redwoo ARMOUR Residentia CYS Inves Capstead Mortgage American Capital Mortgage Inves Anworth Mortgage Asset Apollo Residential Mo New York Mortgag Dynex Arlington Asset Inve In performing these analyses, Lazard reviewed and analyzed certain publicly available financial information, implied multiple trading data relating to the selected comparable companies and compared such information to the corresponding information for la 105

calculated and compared, among other things, (i) the ratio of each company's December 13, 2013 share price to its calendar year 20 estimated GAAP earnings per share, or "P/E," (ii) the ratio of each company's December 13, 2013 share price to its latest available value or net asset value per share, as applicable, or "P/BV," and (iii) each company's estimated dividend yield for the calendar ye 2014. The calendar year 2013 and 2014 estimated GAAP earnings per share and estimated dividend yield for the selected companies Lazard in its analysis were based on the median of FactSet consensus analyst estimates. The latest available GAAP book value per selected companies and used by Lazard in its analysis was based on company filings and publicly available information. Lazard als corresponding financial multiples and ratios for KFN, on the basis of KFN projections based on the median of FactSet consensus estimates.

referred to in to in this proxy statement/prospectus as KFN consensus estimates. The metrics observed by Lazard are summ

	P/E		Dividend Yield		
	2013E	2014E	P/BV	2013E	2014E
Externally Managed BDCs					
High	12.0x	16.9x	1.12x	12.1%	11.9%
Mean	10.3x	11.0x	0.98x	9.3%	9.3%
Median	10.7x	10.6x	1.02x	9.2%	9.3%
Low	7.8x	8.8x	0.80x	4.1%	5.9%
Internally Managed BDCs					
High	18.2x	14.1x	1.75x	11.2%	11.2%
Mean	14.2x	12.4x	1.22x	6.9%	6.8%
Median	14.2x	13.1x	1.13x	8.0%	7.8%
Low	10.3x	9.5x	0.75x	0.3%	0.3%
Commercial Mortgage REITs					
High	28.8x	13.7x	1.91x	13.7%	13.7%
Mean	12.5x	9.0x	1.16x	8.1%	8.9%
Median	13.2x	10.9x	1.07x	7.8%	8.6%
Low	8.1x	8.6x	0.68x	2.3%	7.0%
Residential Mortgage REITs					
High	11.9x	11.4x	1.27x	23.1%	16.4%
Mean	8.0x	8.2x	0.90x	14.4%	12.3%
Median	7.7x	8.0x	0.85x	14.7%	12.0%
Low	4.8x	6.0x	0.71x	6.1%	6.8%
KFN Consensus Estimates	7.6x	9.0x	0.91x	9.2%	9.2%

Lazard primarily focused on externally managed BDCs, given their greater operational comparability to KFN, relative managed BDCs and commercial and residential REITs also considered in this analysis. Based on this analysis of the relevant m comparable companies and KFN and on Lazard's professional judgment, Lazard selected reference ranges of (i) 7.6x to 10.7x and 9 for P/E of the selected companies for the calendar years 2013 and 2014, respectively, (ii) 0.91x to 1.02x for P/BV of the selected con (iii) 8.0% to 10.0% for estimated dividend yield of the selected companies for each of the calendar years 2013 and 2014. Lazard ranges, respectively, to (1) estimated GAAP earnings of KFN for the calendar years 2013 and 2014, as included in the KKR-Projections (as defined below in the section of this proxy statement/prospectus titled "KKR-Prepared KFN Projections" beginning (2) GAAP book value of KFN as of September 30, 2013, and (3) estimated dividend yield of KFN for the calendar years 2013 included in the KKR-Prepared KFN Projections. A summary of certain of the information included in the KKR-Prepared KFN Projections of this proxy statement/prospectus titled "KKR-Prepared KFN Projections" beginning on page 93 statement/prospectus. Based on the average of the results of the foregoing analysis, Lazard calculated a range of equity values per short of \$9.23 to \$11.42. Based on the KKR common unit price of \$24.98 as of December 13, 2013, Lazard noted that such value rang range of implied exchange ratios of 0.3

KFN Selected Precedent Transacti

Lazard reviewed and analyzed certain publicly available financial information of target companies in selected preceden acquisition transactions involving companies it viewed as relevant. In performing these analyses, Lazard analyzed certain financial and transactions

relating to the target companies involved in the selected transactions and compared such information to the corresponding informat

Lazard noted that there are limited or no recent target companies that are directly comparable to KFN. Although none of precedent transactions or the companies party to such transactions is directly comparable to the transactions contemplated b agreement or to KFN, all of the transactions were chosen because they involve transactions that, for purposes of analysis, may b relevant in evaluating the transactions contemplated by the merger agreement and/or involve targets that, for purposes of anal considered relevant. Lazard reviewed the following transactions involving (i) REITs and (ii) BDCs and real est

REITs

Date Announced	Acquiror	Target	Price/BV
	American Realty Capital		
October 22, 2013	Properties	Cole Real Estate	1.23x
June 3, 2013	Mid-America Apartment	Colonial Properties Trust	0.92x
	American Realty Capital		
May 28, 2013	Properties	CapLease Inc.	1.17x
		American Realty Capital	
September 6, 2012	Realty Income Corp.	Trust	1.13x
December 24, 2011	Ventas Inc.	Cogdell Spencer Inc.	0.80x
		Nationwide Health	
February 27, 2011	Ventas Inc.	Properties	1.56x

BDCs and Real Estate Property

			Price/CY		
Date Announced	Acquiror	Target	EPS	Price/BV	
	Annaly Capital	CreXus Investment			
January 30, 2013	Management Inc.	Corp.	12.7x	1.09x	
October 26, 2009	Ares Capital Corporation	Allied Capital Corp.	8.9x	0.46x	

KFN Dividend Disco

Based on the KKR-Prepared KFN Projections, Lazard performed a dividend discount analysis of KFN to calculate the estim value of the projected dividend distributions by KFN in the period from the fourth quarter of 2013 through 2018. Lazard also estimated terminal values for KFN by applying a perpetual growth rate range of 0.0% to 2.0%. The projected dividend distributions values were discounted to present value using discount rates ranging from 8.5% to 9.5%. The discount rates applicable to KFN among other things, on Lazard's judgment of the estimated range of discount rates based on an analysis of publicly traded BDCs. The discount rates applicable to \$14.21. Based

Table

common unit price of \$24.98 as of December 13, 2013, Lazard noted that such value range indicated a range of implied exchange ra

Ot

The analyses and data relating to KFN described below were presented to the independent directors constituting the conflicts of the board of directors of KKR's managing partner merely for information

Other Analyses: KFN Historical Trac

Lazard reviewed historical data with regard to the closing prices of KFN common shares for the 52-week period to a December 13, 2013. During this period, the closing price of common shares of KFN ranged from a low of \$9.11 to a high of \$11. which, based on the KKR common unit price of \$24.98 as of December 13, 2013, indicates a range of implied exchange ratios of 0.

Other Analyses: KFN Analyst Price Tar

Lazard reviewed five Wall Street research equity analyst per share target prices for KFN as of December 13, 2013, which targe released on October 24, 2013. The range of these target prices was \$10.50 to \$12.50, which, based on the KKR common unit price of December 13, 2013, indicates a range of implied exchange ratios of 0.4

KKR Stand-Ala

KKR Blended Earnings Selected Comparable Compar

Lazard reviewed and analyzed selected public companies that it viewed as reasonably comparable to KKR. Although none of companies is directly comparable to KKR, the companies included are publicly traded companies with operations and/or other cri lines of business, markets, business risks and size and scale of business, which for purposes of analysis Lazard considered relevant. Lazard compared KKR to the following public

The Blackstone

Apollo Global Manag

The Carly

Och-Ziff Capital Managemen

Fortress Investment

Oaktree Capital

In performing these analyses, Lazard reviewed and analyzed certain publicly available financial information, implied multiple trading data relating to the selected comparable companies and compared such information to the corresponding information for I calculated and compared, among other things, the ratio of each company's December 13, 2013 share or unit price to its (i) calendar y 2014 per share or unit estimated economic net income, or "ENI," (ii) calendar year 2013 and 2014 per share estimated distri earnings, or "DCE," and (iii) calendar year 2013 and 2014 estimated adjusted net income, or "ANI," in each case as applicable. The 2013 and 2014 estimated ENI, DCE and ANI per share or unit for the selected companies and used by Lazard in its analysis were median of publicly available equity analyst estimates or FactSet consensus analyst estimates, if compiled by FactSet. Lazard also c ratio of KKR's December 13, 2013 unit price to its calendar year 2013 and 2014 per u

Table

ENI and DCE, on the basis of KKR projections based on the median of FactSet consensus analyst estimates, which are referred to statement/prospectus as KKR consensus estimates. The metrics observed by Lazard are summarized in the following

	P/E	NI*	P/DC	E**	P/AN	I***
	2013E	2014E	2013E	2014E	2013E	2014E
Selected Companies						
High	13.3x	11.9x	19.4x	14.0x	9.3x	11.5x
Mean	10.8x	10.7x	12.1x	11.3x		
Median	11.6x	10.7x	10.5x	11.0x		
Low	6.8x	9.6x	7.1x	10.0x		
KKR Consensus Estimates	10.2x	10.1x	14.1x	13.1x		

P/ENI multiple calculated for The Blackstone Group L.P., Apollo Global Management LLC, The Carlyle Group LP Capital 0

P/DCE multiple calculated for The Blackstone Group L.P., Apollo Global Management LLC, The Carlyle Group LP, Och Management Group LLC, Fortress Investment Group LLC and Oaktree Capital (

P/ANI multiple calculated for Oaktree Capital

Based on an analysis of the relevant metrics for each of the comparable companies and KKR and on Lazard's professional judgic selected reference ranges of (i) 10.2x to 10.8x and 10.1x to 10.7x for share price to estimated ENI per share of the selected companies for the calendar years 2013 and 2014, respectively, and (ii) 10.2x to 14.1x and 10.8x to 13.1x for share price to estimated DCE per share of companies for the calendar years 2013 and 2014, respectively. Lazard applied such ranges, respectively, to (1) estimated ENI of calendar years 2013 and 2014, as included in the KKR Forecasts (as defined below in the section of this proxy statement/prospectus Forecasts" beginning on page 95), along with certain internal financial forecasts for KKR provided and prepared by its manage fourth quarter of 2013 (such estimates are collectively referred to in this proxy statement/prospectus as the "KKR management esti (2) estimated DCE of KKR for the calendar years 2013 and 2014, as included in the KKR management estimates. Based on the aresults of the foregoing analysis, Lazard calculated ranges of equity values per common unit for KKR of \$25.09 to \$26.57, based analysis, and \$20.52 to \$26.59, based on the DCE analysis, as compared to the \$24.98 price per KKR common unit as of December 10.00 to 10.0

KKR Sum-of-the-P

Lazard performed a sum-of-the-parts analysis of KKR based on separate valuations of the components of KKR's estimated 20 asset value of KKR's general partner investments, and KKR's net cash. In particular, Lazard calculated an implied equity value range the sum of the following

Fee-Related ENI. Lazard calculated 2014 P/E multiples for a group of traditional asset managers that were Lazard. The P/E multiples were calculated based on each asset manager's share price as of December 13, 2013 a 2014 GAAP earnings per share based on the median of FactSet consensus analyst estimates. Lazard observe median and mean P/E multiples of 16.9x and 16.8x, respectively. Based on an analysis of the relevant median trade professional judgment, Lazard selected a P/E multiple reference range of 15.0x to 17.0x and applied estimated 2014 fee-related ENI, as included in the KKR management estimates (as def

Table

Incentive Fee ENI. For each of The Blackstone Group L.P., Apollo Global Management LLC, The Carlyle G Oaktree Capital Group, LLC, Lazard calculated ratios of implied capitalized value of incentive fee ENI as of D 2013 to estimated 2014 incentive fee ENI. The estimated 2014 incentive fee ENI used in the analysis was derive from publicly available equity analyst estimates. Based on an analysis of the relevant metrics and on Lazard's judgment (i) under a first approach, which is referred to in this proxy statement/prospectus as the earnings multip Lazard selected a multiple reference range of 8.9x to 9.3x and applied it to KKR's estimated 2014 incentive included in the KKR management estimates (as defined above), and (ii) under a second approach, which is refer proxy statement prospectus as the capital and earnings multiples approach, Lazard selected (1) a multiple reference 7.1x to 8.1x and applied it to KKR's estimated 2014 incentive fee ENI, as included in the KKR management (2) a multiple reference range of 0.8x to 1.0x and applied it to the book value of KKR's net accrued carry as of Science of the capital and earnings multiple reference range of 0.8x to 1.0x and applied it to the book value of KKR's net accrued carry as of Science of the capital and earnings multiple reference range of 0.8x to 1.0x and applied it to the book value of KKR's net accrued carry as of Science of the capital and earnings multiple reference range of 0.8x to 1.0x and applied it to the book value of KKR's net accrued carry as of Science of the capital and earnings multiple reference range of 0.8x to 1.0x and applied it to the book value of KKR's net accrued carry as of Science of the capital and earnings multiple reference range of 0.8x to 1.0x and applied it to the book value of KKR's net accrued carry as of Science of the capital and earnings multiple reference range of 0.8x to 1.0x and applied it to the book value of KKR's net accrued carry as of Science of the capital accrued to the capital accrued to the capital accrued to the c

GP Investments NAV. Lazard applied a multiple reference range of 0.8x to 1.0x to the net asset value, as of Sc 2013, of KKR's general partner investment portfolio. The reference range was selected based on Lazard's judgment and on an analysis, for a group of listed private equity closed-end funds selected by Lazard, of the fund's December 13, 2013 share price to its latest available net

Net Cash. Lazard considered the value of KKR's net cash as of September 1.

The foregoing analysis resulted in a range of equity values per common unit for KKR, calculated as the average of the obtained by applying, respectively, the earnings multiple approach and the capital and earnings multiples approach, of \$22.98 compared to the \$24.98 price per KKR common unit as of December 1.00 per Exercise 1.00 per Exercise 2.00 per E

KKR Dividend Disco

Based on the KKR management estimates (as defined above), Lazard performed a dividend discount analysis of KKR to estimated present value of the projected dividend distributions by KKR in the period from the fourth quarter of 2013 through 2015 calculated estimated terminal values for KKR, using two alternative methodologies: (i) by applying multiples of 0.9x to the net a KKR's general partner investment portfolio as of September 30, 2013; of 1.0x to KKR's net cash as of September 30, 2013; of 18.0 KKR's estimated 2015 fee-related ENI, as included in the KKR management estimates (as defined above); and of 8.0x to 10. estimated 2015 incentive fee ENI, as included in the KKR management estimates (as defined above), which is referred to statement/prospectus as the sum-of-the-parts methodology, and (ii) by applying multiples of 10.0x to 14.0x to KKR's estimated 2 included in the KKR management estimates (as defined above), which is referred to in this proxy statement/prospectus a methodology. The projected dividend distributions and terminal values were discounted to present value using discount rates ranging to 13.0%. The discount rates applicable to KKR were based, among other things, on Lazard's judgment of the estimated range average cost of capital based on an analysis of publicly traded alternative asset managers. This analysis resulted in ranges of equi common unit for KKR of \$24.99 to \$28.40, using the sum-of-the-parts methodology, and \$19.91 to \$27.65, using the P/DCE met compared to the \$24.98 price per KKR common unit as of December 20.00 to 10.00 to 10.0

<u>Table</u>

Ot

The analyses and data relating to KKR described below were presented to the independent directors constituting the conflicts of the board of directors of KKR's managing partner merely for information

Other Analyses: KKR Historical Trac

Lazard reviewed historical data with regard to the closing prices of KKR common units for the 52-week period to a December 13, 2013. During this period, the closing price of common units of KKR ranged from a low of \$13.85 to a high of \$24.98 unit, as compared to the \$24.98 price per KKR common unit as of December 13, 2013.

Other Analyses: KKR Analyst Price Tar

Lazard reviewed 12 Wall Street research equity analyst per common unit target prices for KKR as of December 13, 2013, prices were released between October 24, 2013 and December 9, 2013. The range of these target prices was \$23.00 to \$28.00, as con \$24.98 price per KKR common unit as of December 13, 2013, prices were released between October 24, 2013 and December 9, 2013. The range of these target prices was \$23.00 to \$28.00, as con

Pro Forma Analyses of the Combin

In addition, Lazard considered a number of pro forma analyses, including a contribution analysis and pro forma valuation anal using similar methodologies to those described above under "KKR Stand-Alone Analyses." The analyses and data relating to company described below were presented to the independent directors constituting the conflicts committee of the board of director managing partner merely for information

Pro Forma Contribu

Based on the KKR-Prepared KFN Projections and the KKR management estimates (as defined above), Lazard calculated the eforma relative contributions of KKR and KFN to the combined company with respect to certain financial metrics and, be contributions, derived the implied exchange ratio for each of such metrics. The following table sets forth the results of the contributions of the contributions of the contributions of the contributions and the KKR management estimates (as defined above), Lazard calculated the eformal relative contributions of KKR and KFN to the combined company with respect to certain financial metrics and, be contributions, derived the implied exchange ratio for each of such metrics.

		Aggreg	ate			P	er Share	Implied Exch.
	KKR	KFN	%KKR	%KFN	KKR		KFN	Ratio
Market								
Capitalization	\$ 17,880	\$ 1,952	90%	10% 5	\$ 24.98	\$	9.53	038x
Total Assets	\$ 8,515	\$ 3,141	73%	27%	\$ 11.90	\$	15.34	1.29x
Book Value*	\$ 7,208	\$ 2,133	77%	23%	\$ 10.07	\$	10.42	1.03x
2013E Earnings								
ENI /	\$ 1,757		87%	13% 5	\$ 2.46	\$	1.25	0.51x
DCE / Net Income	\$ 1,293	\$ 256	83%	17% 5	\$ 1.81	\$	1.25	0.69x
2014E Earnings								
ENI /	\$ 1,782		89%	11% 5	\$ 2.47	\$	1.07	0.43x
DCE / Net Income	\$ 1,283	\$ 220	85%	15% 5	\$ 1.78	\$	1.07	0.60x
Dividends (2014E)	\$ 944	\$ 180	84%	16% 5	\$ 1.36	\$	0.88	0.65x
Transaction			87%	13%				0.51x

Reported non-G

<u>Table</u>

Pro Forma Estimated Financials of the Combin

Based on the KKR-Prepared KFN Projections and the KKR management estimates (as defined above), Lazard calculated, things, pro forma estimated ENI, DCE, fee-related ENI and incentive fee ENI for the combined company for the calendar years 2014 well as net accrued carry book value as of September 30, 2013, general partner investment portfolio net asset value as of September and net cash as of September 30, 2013 for the combined company (which are collectively referred to as "pro forma financials"). The financials reflect the following assumptions: (i) pro forma estimated general and administrative expense savings of approximately (ii) pro forma estimated benefits of approximately \$10 million resulting from the assumed elimination of the requirement for KKR maintain separate liquidity reserves, (iii) the elimination of KFN management fees through consolidation under KKR ownership increase of an almost equivalent amount of KFN investment income and (iv) KKR's intention to distribute 100% of KFN's realized Lazard noted that the pro forma financials were expected to be dilutive to ENI (by 2% in 2014E) and accretive to DCE (by 39).

Pro Forma Blended Earnings Selected Comparable Companies Analysis of the Combin

Based on the analysis of the selected comparable companies described above under "KKR Stand-Alone Analyses KKR BI Selected Comparable Companies Analysis" and on Lazard's professional judgment, Lazard selected and applied multiples ranging for 10.7x to the pro forma estimated ENI for the combined company for the calendar year 2014, as included in the pro forma financial above). Based on this analysis, Lazard calculated a range of implied equity values per common unit for the combined company \$26.25. By comparing these values to stand-alone implied equity values per common unit for KKR of \$25.22 to \$26.70, as deri comparable companies analysis with respect to estimated 2014 ENI of KKR, as described above under "KKR Stand-Alone Blended Earnings Selected Comparable Companies Analysis," Lazard noted that, on this basis, the transaction would have an illustrate effect on KKR's equity value per common unit of approximately approximately and the selected Companies Analysis, "Lazard noted that, on this basis, the transaction would have an illustrate of the companies analysis with respect to the companies analysis," Lazard noted that, on this basis, the transaction would have an illustrate of the companies analysis with respect to the companies analysis," Lazard noted that, on this basis, the transaction would have an illustrate of the companies analysis with respect to the companies analysis, and the companies analysis with respect to estimate the compani

Based on the analysis of the selected comparable companies described above under "KKR Stand-Alone Analyses KKR BI Selected Comparable Companies Analysis" and on Lazard's professional judgment, Lazard selected and applied multiples ranging from the professional professional professional judgment, Lazard selected and applied multiples ranging from the professional prof

Pro Forma Sum-of-the-Parts Analysis of the Combin

Based on the analyses described above under "KKR Stand-Alone Analyses KKR Sum-of-the-Parts Analysis" and on Lazar judgment, Lazard applied (where applicable) to the combined company's pro forma estimated 2014 fee related ENI and incentive fee the combined company's pro forma net accrued carry book value as of September 30, 2013, general partner investment portfolio ne as of September 30, 2013, and net cash as of September 30, 2013, each as included in the pro forma financials (as defined above), applied to the corresponding components for KKR in the stand-alone sum-of-the-parts analysis of KKR, except that Lazard applied reference range of 0.90x to 1.00x, instead of a multiple reference range of 0.80x to 1.00x, to the combined company's pro forma ge investment portfolions.

Table

value as of September 30, 2013, to account for the increased liquidity and diversity of the combined company's underlying assets. La implied equity value ranges of the combined company as the sum of the results of the foregoing calculations. This analysis resulted equity values per common unit for the combined company, calculated as the average of the equity values obtained by applying, respectively earnings multiple approach and the capital and earnings multiples approach (each as defined above), of \$22.30 to \$24.72. By company values to stand-alone implied equity values per common unit for KKR of \$22.98 to \$26.24, as derived from the stand-alone sum analysis with respect to KKR, as described above under "KKR Stand-Alone Analyses KKR Sum-of-the-Parts Analysis," Lazar this basis, the transaction would have an illustrative dilutive effect on KKR's equity value per common unit of approximately

Pro Forma Dividend Discount Analysis of the Combin

Based on the proforma financials (as defined above), Lazard performed a dividend discount analysis of the combined company the estimated present value of the projected pro forma dividend distributions by the combined company in the period from the four 2013 through 2015. Lazard also calculated estimated terminal values for the combined company, using two alternative methodo applying multiples of 0.95x to the net asset value of the combined company's pro forma general partner investment portfolio as of So 2013; of 1.00x to the combined company's net cash as of September 30, 2013; of 18.0x to 20.0x to the combined company's pro form 2015 fee-related ENI; and of 8.0x to 10.0x to the combined company's pro forma estimated 2015 incentive fee ENI, which is refer proxy statement/prospectus as the pro forma sum-of-the-parts methodology, and (ii) by applying multiples of 10.0x to 14.0x to t company's pro forma estimated 2015 DCE, which is referred to in this proxy statement/prospectus as the pro forma P/DCE method projected dividend distributions and terminal values were discounted to present value using discount rates ranging from 11.0% to discount rates applicable to the combined company were based, among other things, on Lazard's judgment of the estimated range average cost of capital based on an analysis of publicly traded alternative asset managers. This analysis resulted in ranges of in values per common unit for the combined company of \$23.93 to \$26.83, using the pro forma sum-of-the-parts methodology, a \$28.25, using the pro forma P/DCE methodology. By comparing these values to stand-alone implied equity values per common uni \$24.99 to \$28.40 and \$19.91 to \$27.65, as derived from the stand-alone dividend discount analysis of KKR using the sum methodology and the P/DCE methodology (each as defined above), respectively, as described above under "KKR Stand-Alone Dividend Discount Analysis," Lazard noted that, on these bases, the transaction would have an illustrative dilutive effect on KKR's of approximately 4% to 6%, using the pro forma sum-of-the-parts methodology and an illustrative accretive effect on KKR's equ common unit of approximately 2%, using the pro forma P/DCE n

KFN ROE Expansion Sensiti

Lazard calculated the effect on the combined company's equity value, as resulting from the foregoing pro forma analyses of the company, of a potential illustrative 2% to 4% increase in KFN's return on capital (which is defined as net income return on average value and which we refer to as ROE) following the transaction, as a result of assumed enhanced deployment options. Lazard or results of this analysis to the results of the stand-alone analyses of KKR, as described above under "KKR Stand-Alone Analyses Earnings Selected Comparable Companies Analysis" (except that Lazard considered the selected comparable companies analysis with KKR's ENI and DCE for 2014 only), "KKR Stand-Alone Analyses KKR Sum-of-the-Parts Analysis" and "KKR Stand-Alone Dividend Discourties analysis of the selected companies analysis with the selected companies analysis of the selected companies analysis with the selected companies analysis."

Table

and calculated applicable ranges of equity value accretion (or dilution). The results of this analysis are summarized in the followed

	Standalone Equity Value per Common Unit			V	ro Forn alue per Unit + 4 Incr	Co %]	mmon ROE	Equity V 2% R Incre	OE	retion/(Dilution) 4% ROE Increase		
		Low]	High		Low		High	Low	High	Low	High
Blended ENI Comparable Companies												
Analysis(a)(c)	\$	25.22	\$	26.70	\$	24.79	\$	27.39	0%	0%	3%	3
Blended DCE Comparable Companies												
Analysis(b)(c)	\$	21.01	\$	25.43	\$	21.40	\$	27.26	5%	4%	7%	7
Sum-of-the-Parts Analysis(d)	\$	22.98	\$	26.24	\$	22.30	\$	25.15	(1%)	(4%)	(1%)) (4
Dividend Discount												
Analysis Sum-of-the-Parts Method(d)	\$	24.99	\$	28.40	\$	23.93	\$	27.26	(3%)	(4%)	(3%)) (4
Dividend Discount Analysis P/DCE												
Method(c)	\$	19.91	\$	27.65	\$	20.38	\$	29.65	5%	5%	8%	7

Stand-alone equity value per common unit of KKR based on comparable companies analysis with respect to estimated EN

Stand-alone equity value per common unit of KKR based on comparable companies analysis with respect to estimated I

ROE increase captured as capitalization of incremental return of

ROE increase captured as a 0.05x increase of the P/BV multiple applied to the general partner investment portfolio ne

M

Lazard prepared these analyses solely for purposes of, and the analyses were delivered to the independent directors constituting committee of the board of directors of KKR's managing partner in connection with, the provision of Lazard's opinion to the independ constituting the conflicts committee of the board of directors of KKR's managing partner as to the fairness from a financial point of exchange ratio to KKR. These analyses do not purport to be appraisals nor do they necessarily reflect or purport to reflect the price businesses or securities actually may be sold or the prices at which any securities have traded or may trade at any time in the future based upon forecasts of future results are not necessarily indicative of actual future results, which may be significantly more or let than suggested by these analyses. Because these analyses are inherently subject to uncertainty, being based upon numerous fact beyond the control of the parties or their respective advisors, neither Lazard nor any other person assumes responsibility if future materially different from the

Lazard, as part of its investment banking business, is continually engaged in the valuation of businesses and their securities in with mergers and acquisitions, negotiated underwritings, secondary distributions of listed and unlisted securities, private placement buyouts, and valuations for estate, corporate and other purposes. In the ordinary course of their respective businesses, Lazard Holdings LLC (an entity indirectly owned in large part by managing directors of Lazard) and their respective affiliates may a securities of KKR, KFN and certain of their respective affiliates for their own accounts and for the accounts of their customers and, may at any time hold a long or short position in such securities, and may a

<u>Table</u>

hold securities on behalf of KKR, KFN and certain of their respective affiliates. Lazard in the past has provided, currently is providing future may provide certain investment banking services to KKR and certain of its affiliates, for which Lazard has received and compensation, including, in the past two years, advising KKR or an affiliate thereof with respect to its proposed acquisition of SB Group and advising a portfolio company of KKR or an affiliate thereof with respect to a potential transaction, and having advised a KKR in connection with its provision of debt financing to Uralita, Capmark Financial Group in connection with its restructuring transactions, and KKR and other affiliates or portfolio companies with respect to transactions that were not consummated. During period ended December 16, 2013, the investment banking division of Lazard has earned compensation for financial advisory servi directly to KKR and/or to its affiliates and controlled portfolio companies of approximately \$9.1 million, in addition to Lazard's c for its services as financial advisor to the independent directors constituting the conflicts committee of the board of directors of KKF partner in connection with the merger. During the two year period ended December 16, 2013, Lazard has not been engaged be subsidiaries to provide financial advisory or underwriting services for which the Investment Banking Division of Lazard compensation. The issuance of Lazard's opinion was approved by the Opinion Committee

In connection with Lazard's services as financial advisor to the independent directors constituting the conflicts committee of directors of KKR's managing partner and pursuant to the terms of the Lazard engagement letter dated December 11, 2013, KKR a Lazard an aggregate fee equal to \$1,500,000 upon delivery of the opinion. In addition, KKR also agreed to reimburse Lazard for incurred in connection with the engagement and to indemnify Lazard and certain related parties against certain liabilities to circumstances that may arise out of the rendering of the connection with the engagement and to indemnify Lazard and certain related parties against certain liabilities to circumstances that may arise out of the rendering of the connection with the engagement and to indemnify Lazard and certain related parties against certain liabilities to circumstances that may arise out of the rendering of the connection with the engagement and to indemnify Lazard and certain related parties against certain liabilities to circumstances that may arise out of the rendering of the connection with the engagement and to indemnify Lazard and certain related parties against certain liabilities to circumstances that may arise out of the rendering of the connection with the engagement and the connection with the connection with the engagement and the

The type and amount of consideration payable pursuant to the merger agreement was determined through negotiations betwee KFN, rather than by any financial advisor, and was approved by the board of directors of KKR's managing partner. Lazard did not any specific merger consideration to the board of directors of KKR's managing partner, to the independent directors constituting committee of the board of directors of KKR's managing partner, or to KKR or that any given merger consideration constituting appropriate consideration for the merger. The decision to enter into the merger agreement was solely that of the board of director managing partner. As described above, the opinion of Lazard was one of many factors taken into consideration by the independent constituting the conflicts committee thereof in their evaluation of the merger in accordance with the merger agreement. Constantly analyses described above should not be viewed as determinative of the opinion of the board of directors of KKR's managing partner.

Lazard is an internationally recognized investment banking firm providing a broad range of financial advisory and other sindependent directors constituting the conflicts committee of the board of directors of KKR's managing partner selected Lazard a advisor because of its qualifications, expertise and reputation in investment banking and mergers and acquisitions, as well as its fan the business of KKR. The conflicts committee of the board of directors of KKR's managing partner identified two investment banks high quality, familiar with the industry of KKR and KFN and unlikely to have conflicts with respect to a transaction between KKR potential financial advisors to the committee. Certain members of the conflicts committee interviewed the two investment banks a with each bank its experience, expertise and qualifications in investment banking and mergers and acquisitions and its familia businesses of KKR and KFN. The members of the conflicts committee also confirmed with each bank the absence of any pending c in the conflicts committee's view, would have impaired each

Table

ability to potentially participate as advisor to the conflicts committee in a transaction between KKR and KFN. After meeting with referom both financial advisor candidates, the members of the conflicts committee considered the discussion with each candidate, all experience, expertise, qualifications and reputation of each candidate, and determined to retain Lazard as its financial advisor.

A copy of the presentation delivered by Lazard to the conflicts committee of the board of directors of KKR's managing December 16, 2013 in connection the rendering of Lazard's oral opinion has been filed as an exhibit to the Transaction Schedule 13E-3 filed with the SEC in connection with the transaction and will be made available for inspection and copying at executive offices of KKR during its regular business hours by any interested holder of KKR common units. Copies may be requesting them in writing from KKR at the address provided in the section titled "Where You Can Find More Information" page 234 of this proxy statemen

Position of the KKR Participants as to the Fairness of

Under SEC rules, the KKR Participants are required to express their belief as to the fairness of the merger to the unaffilia shareholders of KFN. The KKR Participants are making the statements included in this section solely for the purpose of comply requirements of Rule 13e-3 and the related rules under the Exchange Act. However, the making of such statements is not an adm KKR Participants that KFN is "controlled" by KKR or any affiliate of KKR such that KFN should be deemed to be an "affiliate" for Rule 13e-3 in connection with

KKR, Fund Holdings and Merger Sub attempted to negotiate a transaction that would be most favorable to them and did not transaction with a goal of obtaining terms that were fair to the common shareholders of KFN and did not undertake any independent of the fairness of the merger to the unaffiliated common shareholders of KFN or engage a financial advisor for such purpose. However, the procedural safeguards implemented during the negotiation of the merger and the other factors considered by, and the analysis, discussions of, KFN's board of directors and the transaction committee of the KFN board of directors described in the second procedural safeguards in the KFN Board of Directors and Reasons for the Merger; Fairness of the Merger" beginning on page 38 statement/prospectus, which analysis, discussion and resulting conclusions the KKR Participants expressly adopt as their or Participants believe that the merger is substantively and procedurally fair to KFN's unaffiliated common states.

The foregoing discussion of the factors considered by the KKR Participants is not intended to be exhaustive but is believed to material factors considered by the KKR Participants in making a determination regarding the fairness of the merger for the purpose of with the requirements of Rule 13e-3 and the related rules under the Exchange Act. The KKR Participants did not find it practicable not, quantify or otherwise attach relative weights to the foregoing factors in reaching their position as to the fairness of the merger KKR Participants made their fairness determination after considering all of the factor

KFN Projected Financial

KFN does not, as a matter of course, generally publish its business plans and strategies or make external disclosures of it financial position or results of operations. KFN management prepared certain nonpublic financial projections prior to the evaluation transaction, and KKR had access to such projections. While Sandler O'Neill also had access to such financial projections, the projections of used by Sandler O'Neill in performing the analysis in

<u>Table</u>

of their fairness opinion. Such financial projections did not result from a full financial forecast model, but were instead derived from mix of portfolio strategies and a projected rate of return for each such portfolio strategy. For these reasons, among others, publi analyst estimates were deemed by KFN management to be the best available information. A summary of the nonpublic financia prepared by KFN management is provided below and such information is referred to in this proxy statement/prospectus as the K projections. KFN common shareholders are cautioned not to place undue reliance on the KFN financial projections. The K projections are not being included in this proxy statement/prospectus for the purpose of influencing your decision whether to adoption of the merger agreement and should not be regarded as an indication that any of KFN, KKR or their respective affilia officers, directors, partners or representatives or any recipient of this information considered, or now considers, it to be necessarily actual future results, and the KFN financial projections should not be relied on as such. None of KKR, KFN or their respect advisors, officers, employees, directors or representatives can give you any assurance that actual results will not differ from the projection. and none undertakes any obligation, except as required by law, to update or otherwise revise or reconcile these internal K projections to reflect circumstances existing, or changes in assumptions or outlook occurring, after the date the internal K projections were generated or to reflect the occurrence of future events even in the event that any or all of the assumptions un projections are shown to be in error. KFN does not intend to update or otherwise revise the KFN financial projections. None of affiliates, advisors, officers, employees, directors or representatives has made or makes any representation to any shareholder or (including to KKR) regarding KFN's ultimate performance compared to the information contained in the KFN financial projection projected results will

	2	2013E	2	014E	2	015E
Total CLO Income (in millions)	\$	204.3	\$	180.4	\$	167.6
Total Investment Income (Including CLO Income) (in millions)	\$	300.9	\$	327.9	\$	317.2

	2014E		20	015E
Earnings Per Share (Liquidity Forecast Case)	\$	0.94	\$	0.87
Return on Equity (Liquidity Forecast Case)		9.0%		8.3%

	Higl	h Case	Lov	w Case
2014 Earnings Per Share (Management Case)	\$	1.10	\$	1.00

	Downside Case			Base Case	A	ggressive Case
2015 Net Income	\$	152,000,000	\$	250,200,000	\$	299,600,000
2015 Earnings Per Share		\$.74		\$1.22		\$1.46
2015 Return on Equity(1)		6%)	10%	,	12%
2015 Return on Equity(2)		7.1%		11.7%		14%

Calculated based on KFN's total outstanding equity, including preferred equity, and as provided by KFN manageme

Calculated by Sandler O'Neill based on KFN's total outstanding common equity, excluding pref

While presented with numeric specificity, the KFN financial projections reflect numerous estimates and assumptions management with respect to industry performance and competition, general business, economic, market and financial conditions, and additional matters specific to KFN's business, all of which are difficult to predict and many of which are be

<u>Table</u>

control. Other important factors that may affect actual results and cause the KFN financial projections not to be achieved included limited to, risks and uncertainties relating to KFN's business (including its ability to achieve strategic goals, objectives and applicable periods), industry performance, the regulatory environment, general business and economic conditions and other matter under the sections entitled "Cautionary Statement Regarding Forward-Looking Statements" beginning on page 125 statement/prospectus and "Risk Factors" beginning on page 104 of this proxy statement/prospectus. As a result, there can be no assure KFN financial projections will be realized or that actual results will not be materially different than estimated therein, and it is likely results.

The KFN financial projections were not prepared with a view toward public disclosure or with a view toward complying with t guidelines of the SEC regarding projections or the guidelines established by the American Institute of Certified Public Acc preparation and presentation of financial projections but, in the view of KFN's management, were prepared on a reasonable basis, a the assumptions upon which such financial projections are based, reflect the best currently available estimates and judgme management and presents, to the best of management's knowledge and belief, the expected course of action and expected fut performance of KFN. However, this information is not fact and should not be relied upon as being necessarily indicative of future readers of this proxy statement/prospectus are cautioned not to place undue reliance on the prospective financial information. The K projections were prepared by, and are the responsibility of KFN. Neither KFN's nor KKR's independent registered public account any other independent accountants, have compiled, examined or performed any procedures with respect to the projected financial datherein, nor have they expressed any opinion or any other form of assurance on such information or its achievability, and assume nor for, and disclaim any association with, the projected financial data. The reports of the independent registered public account incorporated by reference in this proxy statement/prospectus relate to KFN's and KKR's historical annual financial statements. Tho not extend to the KFN financial projections and should not be received.

KKR-Prepared KFN

KKR does not, as a matter of course, prepare financial projections with respect to KFN. However, in connection with the regarding the proposed transaction with KFN, management of KKR's general partner prepared certain unaudited financial projection KFN's forecasted operating results for fiscal years 2013 through 2018 based on internal KKR modeling derived in part from the K projections. Management of KKR's general partner also provided to Goldman Sachs and Lazard certain estimated potential cost other synergies to be realized if the potential transaction with KFN were to be consummated. Such unaudited financial projections a synergies prepared by management of KKR's general partner are referred to in this proxy statement/prospectus as the KKR-Projections. KKR provided the KKR-Prepared KFN Projections to Goldman Sachs and Lazard for purposes of the financial opinions of Goldman Sachs and Lazard as summarized above, and also to the board of directors of KKR's general partner. A sur KKR-Prepared KFN Projections is provided below. KFN common shareholders are cautioned not to place undue reliance on the KI KFN Projections. The KKR-Prepared KFN Projections are not being included in this proxy statement/prospectus for the purpose of your decision whether to vote for the adoption of the merger agreement and should not be regarded as an indication that any of K their respective affiliates, advisors, officers, directors, partners or representatives or any recipient of this information considers, it to be necessarily predictive of actual future results, and the KKR-Prepared KFN Projections should not be relied on as statement respective affiliates, advisors, officers, employees, directors or representatives can give you any assurance that a will not differ from the projected results, and none undertakes any obligation, except as require

Table

update or otherwise revise or reconcile the KKR-Prepared KFN Projections to reflect circumstances existing, or changes in as outlook occurring, after the date the KKR-Prepared KFN Projections were prepared or to reflect the occurrence of future events. Nor its affiliates, advisors, officers, employees, directors or representatives has made or makes any representation to any KFN common or other person regarding KFN's ultimate performance compared to the information contained in the KKR-Prepared KFN Projection results contained therein will

Unaudited Financia

(\$ in millions, except per KFN common share)	Er	ear ding 1/2013	En	ear ding 1/2014	E	Year nding 31/2015	Er	ear nding 51/2016	En	ear ding 1/2017	En	ear ding 1/2018
Total Investment Income	\$	418	\$	353	\$	387	\$	393	\$	401	\$	421
Net Interest Expense	\$	(74)	\$	(73)	\$	(73)	\$	(75)	\$	(74)	\$	(72)
Net Investment Income	\$	344	\$	280	\$	314	\$	317	\$	327	\$	349
Non-Investment Expenses	\$	(88)	\$	(60)	\$	(62)	\$	(63)	\$	(66)	\$	(72)
Net Income	\$	256	\$	220	\$	252	\$	254	\$	261	\$	276
Earnings Per Share	\$	1.25	\$	1.07	\$	1.23	\$	1.24	\$	1.28	\$	1.35

(\$ in millions, except per KFN common share)	E	Year nding 31/2013	E	Year Inding 31/2014	E	Year Ending	E	Year nding 31/2016	E	Year nding 31/2017	E	Year nding 31/2018
Common Shareholders' Equity	\$	2,166	\$	2,204	\$	2,239	\$	2,268	\$	2,333	\$	2,408
Book Value per KFN Common Share	\$	10.58	\$	10.76	\$	10.93	\$	11.08	\$	11.39	\$	11.76

Potent

	Year	Ending	Year	Ending
(\$ in millions)		1/2014	12/3	1/2015
Net Tax Impact	\$	(6)	\$	(5)
KFN Incremental Earnings(1)	\$	10	\$	8
KFN General and Administrative Expense Savings	\$	5	\$	5

Following an acquisition by KKR, KFN would no longer need to maintain separate liquidity reserves given that KFN v point, be able to access KKR's liquidity reserves. In such a scenario, the amounts currently carried as cash on KFN's be could be deployed by KFN into higher yielding

While presented with numeric specificity, the KKR-Prepared KFN Projections reflect numerous estimates and assumption respect to industry performance and competition, general business, economic, market and financial conditions, interest rates an matters specific to KFN's business, all of which are difficult to predict and many of which are beyond KFN's control. Other importan may affect actual results and cause the KKR-Prepared KFN Projections not to be achieved include, but are not limited to, risks and relating to KFN's business, industry performance, the regulatory environment, general business and economic conditions and described under the sections entitled "Cautionary Statement Regarding Forward-Looking Statements" beginning on page 125 statement/prospectus and "Risk Factors" beginning on page 104 of this proxy statement/prospectus. As a result, there can be no assur KKR-Prepared KFN Projections will be realized or that actual results will not be materially different than estimated therein, and it actual results

<u>Table</u>

The KKR-Prepared KFN Projections were not prepared with a view toward public disclosure or with a view toward comply published guidelines of the SEC regarding projections or the guidelines established by the American Institute of Certified Public Acc preparation and presentation of financial projections but, in the view of the management of KKR's general partner, reflected the estimates and judgments of the management of KKR's general partner with respect to the expected future financial performa However, this information is not fact and should not be relied upon as being necessarily indicative of future results, and readers statement/prospectus are cautioned not to place undue reliance on the KKR-Prepared KFN Projections. The KKR-Prepared KFN were prepared by, and are the responsibility of, KKR. Neither KFN's nor KKR's independent registered public accounting firm, reindependent accountants, have compiled, examined or performed any procedures with respect to the projected financial data contains on have they expressed any opinion or any other form of assurance on such information or its achievability, and assume no responsant disclaim any association with, the projected financial data. The reports of the independent registered public accounting firms incorreference in this proxy statement/prospectus relate to KFN's and KKR's historical annual financial statements. Those reports do not a KKR-Prepared KFN Projections and should not be reference.

KK

KKR does not, as a matter of course, generally publish its business plans and strategies or make external disclosures of it financial position or results of operations. However, the management of KKR's general partner, for purposes of the financial opinions of Goldman Sachs and Lazard as summarized above, advised Goldman Sachs and Lazard that the publicly available for respect to KKR reflected the best available estimates and judgments of the anticipated future financial performance of KKR. S available forecasts that were provided to Goldman Sachs and Lazard are referred to in this proxy statement/prospectus as the KKR addition, KKR provided Goldman Sachs and Lazard with a set of projected pro forma numbers for KKR following the acquisi which is referred to in this proxy statement/prospectus as the KKR Pro Forma Forecasts. KKR calculated the KKR Pro Forma Fo the KKR-Prepared KFN Projections, the KKR Forecasts and the potential synergies described above as the inputs. Summaries Forecasts and the KKR Pro Forma Forecasts are provided below. KFN common shareholders are cautioned not to place undue re KKR Forecasts or the KKR Pro Forma Forecasts. The KKR Forecasts and the KKR Pro Forma Forecasts are not being included statement/prospectus for the purpose of influencing your decision whether to vote for the adoption of the merger agreement and s regarded as an indication that any of KFN, KKR or their respective affiliates, advisors, officers, directors, partners or represent recipient of this information considered, or now considers, it to be necessarily predictive of actual future results, and neither the KI nor the KKR Pro Forma Forecasts should be relied on as such. None of KKR, KFN or their respective affiliates, advisors, officers directors or representatives can give you any assurance that actual results will not differ from the projected results, and none un obligation, except as required by law, to update or otherwise revise or reconcile the KKR Forecasts or the KKR Pro Forma Forecast circumstances existing, or changes in assumptions or outlook occurring, after the date the KKR Forecasts and the KKR Pro Fore were prepared or to reflect the occurrence of future events. None of KKR or its affiliates, advisors, officers, employees representatives has made or makes any representation to any KFN common shareholder or other person regarding KKR's ultimate compared to the information contained in the KKR Forecasts and the KKR Pro Forma Forecasts or that the results contained th

<u>Table</u>

KF

(\$ in millions, except per KKR common unit)	Ending 1/2014(1)	ar Ending 31/2015(1)
Economic Net Income (ENI)	\$ 1,992	\$ 2,208
After Tax Economic Net Income	\$ 1,782	\$ 1,990
After Tax Economic Net Income Per KKR Common Unit	\$ 2.47	\$ 2.72
Total Distributable Earnings	\$ 1,393	\$ 1,565
Distribution Per KKR Common Unit	\$ 1.36	\$ 1.52
Breakdown of Economic Net Income Pre-Tax		
Fee Related Earnings	\$ 490	\$ 553
Net Carried Interest	\$ 750	\$ 856
Other Investment Income	\$ 753	\$ 799

Represents the average estimates of 12 equity research analysts who actively cover KKR and had updated their research KKR reported its Q3 2013 earnings. Included firms were Bank of America/Merrill Lynch, BMO, Citi, Credit Suiss Goldman Sachs, KBW, Morgan Stanley, Oppenheimer, Sandler O'Neill, Sterne Agee and Wells Fargo. BMO, KBW, Oppe Sandler O'Neill only provided estimates for 2014. Taxes are allocated to ENI earnings streams based upon the tax assumpt in the research analysts' models to arrive at the ENI provision for income taxes. To the extent the tax provision calc analyst's model is not derived from the separate ENI earnings streams, the tax provision is assumed to be allocated to earnings, which historically has been the largest component of taxable income for KKR. Reductions in taxes at management fee refunds were applied to fee related earnings. Taxes allocated to total distributable earnings are based on to income tax provision used in the analyst's distribution calculation, grossed up by the percentage of distributed earnings

KKR Pro Forn

(\$ in millions, except per KKR common unit)	Ending 31/2014	Year Ending 12/31/2015		
Economic Net Income (ENI)	\$ 2,226	\$	2,466	
After Tax Economic Net Income	\$ 2,010	\$	2,243	
After Tax Economic Net Income Per KKR Common Unit	\$ 2.43	\$	2.68	
Total Distributable Earnings	\$ 1,627	\$	1,823	
After Tax Total Distributable Earnings	\$ 1,510	\$	1,675	
After Tax Total Distributable Earnings Per KKR Common Unit	\$ 1.82	\$	2.00	

While presented with numeric specificity, the KKR Forecasts and the KKR Pro Forma Forecasts reflect numerous e assumptions made with respect to industry performance and competition, general business, economic, market and financial condit rates and additional matters specific to KKR's business, all of which are difficult to predict and many of which are beyond KKR's comportant factors that may affect actual results and cause the KKR Forecasts or the KKR Pro Forma Forecasts not to be achieved incomo not limited to, risks and uncertainties relating to KKR's business, industry performance, the regulatory environment, general economic conditions and other matters described under the sections entitled "Cautionary Statement Regarding Forward-Looking beginning on page 125 of this province."

<u>Table</u>

prospectus and "Risk Factors" beginning on page 104 of this proxy statement/prospectus. As a result, there can be no assurance t Forecasts or the KKR Pro Forma Forecasts will be realized or that actual results will not be materially different than estimated their likely that actual results.

The KKR Forecasts and the KKR Pro Forma Forecasts were not prepared with a view toward public disclosure or with a complying with the published guidelines of the SEC regarding projections or the guidelines established by the American Institute Public Accountants for preparation and presentation of financial projections but, in the view of the management of KKR's ger reflected the best available estimates and judgments of the management of KKR's general partner with respect to the expected fur performance of KKR. However, this information is not fact and should not be relied upon as being necessarily indicative of future readers of this proxy statement/prospectus are cautioned not to place undue reliance on the KKR Forecasts or the KKR Pro Forma Forecasts were prepared by, and are the responsibility of, KKR. Neither KFN's nor KKR's registered public accounting firm, nor any other independent accountants, have compiled, examined or performed any procedures we the projected financial data contained herein, nor have they expressed any opinion or any other form of assurance on such infor achievability, and assume no responsibility for, and disclaim any association with, the projected financial data. The reports of the registered public accounting firms incorporated by reference in this proxy statement/prospectus relate to KFN's and KKR's hist financial statements. Those reports do not extend to the KKR Forecasts and should not be responsible.

Fees an

The following is an estimate of the fees and expenses incurred or to be incurred by KFN in connection with the property

Description	 nount nillions)
Financial, legal, accounting and tax advisory fees and expenses	\$ 22.2
Proxy solicitation, printing and mailing costs	0.3
Filing fees	0.2
Miscellaneous	0.2
Total	\$ 22.9

The following is an estimate of the fees and expenses incurred or to be incurred by the KKR Participants in connection with

	Amount		
Description	(in millions)		
Financial, legal, accounting and tax advisory fees and expenses	\$	12.0	
Printing and mailing costs		0.3	
Filing fees (including filings under the HSR Act)		0.5	
Miscellaneous		0.2	
Total	\$	13.0	

Subject to the termination payment that is payable under certain circumstances, the merger agreement generally provides expenses incurred in connection with the merger agreement and the transactions contemplated by the merger agreement will be paid incurring such costs and expenses, except that (1) expenses incurred in connection with filing, printing and mailing of the registrati of which this proxy statement/prospectus forms a part at

Table

statement/prospectus (including filing fees) will be shared equally by Fund Holdings and KFN and (2) KKR will be responsible for a under the HSR Act and other antitrust laws. See "The Merger Agreement Termination Payment and Expenses" beginning on payment and Expenses proxy statemen

Interests of Directors and Executive Officers of KFN in

In considering the recommendation of the KFN board of directors that KFN common shareholders vote to adopt the merge KFN common shareholders should be aware that aside from their interests as common shareholders of KFN, KFN's executive directors have interests in the merger that are different from, or in addition to, those of other common shareholders of KFN go members of the KFN board of directors were aware of and considered these interests, among other matters, in evaluating and ne merger agreement and the merger, and in recommending to the common shareholders of KFN that the merger agreement be ad interests are described in more detail below, and certain of them are quantified in the table below. KFN's executive officers (in executive officers who are members of the KFN board of directors) did not participate in the vote by the KFN board of directors re

Treatment of Equity-Be

0 1 0

Each restricted KFN common share that was granted under a KFN equity incentive plan and that is outstanding immediately effective time of the merger, including those held by KFN executive officers and directors, as of the effective time, by virtue of the of the consummation of the merger and without any action on the part of the holder of such KFN restricted share, will be conver restricted KKR common units having the same terms and conditions, including applicable vesting requirements, as applied to su KFN common share immediately prior to the effective time, with fractional KKR common units to be aggregated and rounded t whole unit, and KKR will assume the related equity incentive plan and any award agreement issued thereunder pursuant to who restricted KFN common share has been granted in order to provide for the foregoing. The table below sets forth the unvested rescommon shares held by the KFN directors and execu

Pha

Each KFN phantom share that was granted under KFN's Non-Employee Directors' Deferred Compensation and Share Award converted into a phantom share in respect of 0.51 KKR common units and will otherwise remain subject to the terms of the plan. The sets forth the vested and restricted KFN phantom shares held by the KFN directors. The KFN executive officers do not participate in as such do not hold any KFN phantom shares held by the KFN directors.

Outstanding KFN Equity-Based Awards Held by KFN Directors and Execu

Set forth below are the number of outstanding KFN restricted common shares, phantom shares and restricted phantom shares of the KFN executive officers and directors that are outstanding as of March 18, 2014 and will be converted into an award in restrommon units in connection with the merger. Pursuant to the merger agreement, KFN may provide for the accelerated vesting of awards held by its directors and executive officers in connection with the merger. If the restricted shares and restricted phantom share accelerated (upon the cessation of the directors' service with KFN due to the merger), there would be a benefit to the applicable indicates to the value of the accelerated restricted awards (shown in the table below assuming a price per KFN common share of \$12.20 closing price of KFN control of the cessation of the directors' service with KFN due to the merger.

Table

over the first five business days following the announcement of the merger). The phantom shares are fully vested and non-forfeita date hereof. The numbers below with respect to phantom shares do not include additional phantom shares to be credited with distributions declared in the 2014 cr

Director/Executive Officer	Restricted Shares	Restricted Shares (\$)	Phantom Shares	Restricted Phantom Shares	Restricted Phantom Shares (\$)
Craig Farr	0	\$	0	0	\$
Michael R. McFerran	41,972	512,058.40	0	0	
Nicole J. Macarchuk	26,264	320,420.80	0	0	
Tracy L.Collins	14,538	177,363.60	0	0	
Robert L. Edwards	15,429	188,233.80	0	0	
Vincent Paul Finigan	6,548	79,885.60	81,145	7,990	97,478.00
Paul M. Hazen	5,057	61,695.40	6,348	19,864	242,340.80
R. Glenn Hubbard	0		219,900	20,768	253,369.60
Ross J. Kari	14,538	177,363.60	0	0	
Ely L. Licht	14,538	177,363.60	0	0	
Deborah H. McAneny	0		55,453	14,538	177,363.60
Scott C. Nuttall	0		0	0	
Scott A. Ryles	14,538	177,363.60	0	0	
Willy R. Strothotte	0		185,790	14,538	177,363.60
Total:	153,422	\$ 1,871,748.41	547,736	77,698	\$ 947,915.60

The KFN executive officers are employed by KFN's manager and do not receive compensation from KFN or any of its subserving as KFN's executive officers, and the manager cannot identify the portion of the compensation awarded the KFN executive of manager that relates solely to services for KFN. KFN and, to KFN's knowledge, KKR and the manager are not party to any arrang the KFN executive officers that provide for termination pay or benefits in connection with the KFN executive officers' service to K its subsidiaries that are based on or otherwise relate to the proposed merger of KFN. As a result, the say-on-golden-parachute provided Frank Wall Street Reform and Consumer Protection Act are

Indemnific

The merger agreement provides that, following the completion of the merger, KKR and the surviving entity in the merger will KFN's obligations to indemnify the current and former executive officers and directors of KFN and all of its subsidiaries for omissions by such indemnified parties that occurred prior to the merger. In addition, the merger agreement provides that KKR may effective time of the merger, purchase a "tail" insurance policy covering claims for at least six years following the effective time of with respect to directors' and officers' liability insurance with coverage and amounts and terms and conditions no less favorable to be persons than KFN's existing coverage. If KKR elects not to purchase such a "tail" insurance policy, KFN is permitted to purchase policy so long as the cost of such a "tail" policy does not exceed 300% of the current annual premium paid by KFN for its directors' liability insurance. See "The Merger Agreement Indemnification; Directors' and Officers' Insurance" beginning on page 147 statemen

No Appr

Under Delaware law and pursuant to KFN's operating agreement, KFN common shareholders will not have appraisal rights i

<u>Table</u>

Plans for KFN after

Following the consummation of the merger, it is anticipated that KFN will continue to conduct operations substantially as they being conducted, except that KFN common shares will cease to be publicly traded and KFN will be a wholly-owned subside Holdings. As of the date of this proxy statement/prospectus, KKR has no current plans or proposals or negotiations which would result in an extraordinary transaction involving KFN's business or management, such as a merger, reorganization, liquidation, relocations, or sale or transfer of a material amount of assets, or the incurrence of any indebtedness. Following the merger, KKF partner will continuously evaluate and review KFN's business and operations and may propose or develop new plans and propose considers to be in the best interests of KKR and its unitholders, including engaging in acquisitions of new businesses or assets, diexisting businesses or assets, the movement of businesses or assets within the KKR corporate structure, the alteration of the mix of a KFN, or any of the types of extraordinary transactions described that the conduction of the mix of a KFN, or any of the types of extraordinary transactions described the conduction of the mix of a KFN, or any of the types of extraordinary transactions described the conduction of the mix of a KFN.

Board of Directors and Management of the General Partner of KKR after

Neither the board of directors nor the management of the general partner of KKR is expected to change in connection with

Accounting Treatment of

The merger will be accounted for by KKR using the acquisition method of accounting. Under this method of accounting, the put will be allocated to the fair value of the net assets acquired at the date of completion of the merger. The excess purchase price over to of the net assets acquired will be recorded

The financial condition and results of operations of KKR after completion of the merger will reflect KFN's balances and completion of the transaction but will not be restated retroactively to reflect the historical financial condition or results of operation of the earnings of KKR following completion of the merger will reflect acquisition accounting adjustments, including the effect of charmying value for assets and liabilities on interest expense and amortization expense. Intangible assets with indefinite useful liver goodwill will not be amortized but will be tested for impairment at least annually, and all assets including goodwill will be tested for when certain indicators are present. If, in the future, KKR determines that tangible or intangible assets (including goodwill) are impairment charge.

Regulatory Approvals and Clearances Required for

Each party's obligation to effect the merger is conditioned upon, among other things, the expiration or early termination of period under the HSR Act applicable to the merger and the other transactions contemplated by the merger agreement. See Agreement Conditions to Consummation of the Merger" beginning on page 134 of this proxy stateme

At any time before or after the effective time of the merger, the FTC, the Antitrust Division or others (including states and price could take action under the antitrust laws, including seeking to prevent the merger, to rescind the merger or to make approval conditional on the parties taking certain actions. There can be no assurance that a challenge to the merger on antitrust grounds will or, if such a challenge is made, that it would not be

Table

Under the HSR Act and the rules promulgated thereunder by the FTC, certain transactions may not be consummated unless not been given and certain information has been furnished to the FTC and the Antitrust Division and certain waiting period requirement satisfied. A transaction notifiable under the HSR Act may not be completed until the expiration of a 30-calendar-day waiting period the filing of appropriate HSR Act notification forms or the early termination of that waiting period. On February 20, 2014, KKR are filed a notification and report form under the HSR Act with the Antitrust Division and the FTC. On March 6, 2014, the FTC general termination of the waiting period under the termination of the waiting termination of the waiting

Other Government

Neither KKR nor KFN is aware of any material governmental approvals or actions that are required for completion of the merg those described above. It is presently contemplated that if any such additional material governmental approvals or actions are recommendated approvals or actions with a such additional material governmental approvals or actions with a such additional material governmental approvals or actions with a such additional material governmental approvals or actions with a such additional material governmental approvals or actions with a such additional material governmental approvals or actions with a such additional material governmental approvals or actions with a such additional material governmental approvals or actions with a such additional material governmental approvals or actions are recommendated that if any such additional material governmental approvals or actions are recommendated that if any such additional material governmental approvals or actions with a such additional material governmental approvals or actions with a such additional material governmental approvals or actions with a such additional material governmental approvals or actions with a such additional material governmental approvals or actions are recommendated that it is a such as a such a s

Dividend Policy and Share I

Subject to the provisions of applicable law, the holders of KKR common units are entitled to receive, when and as declared by directors of KKR's managing partner, distributions (whether payable in cash, property, or securities of KKR) out of the assets of available for such or availabl

Listing of KKR Common Units Issued in

It is a condition to the closing of the merger that the KKR common units to be issued in the merger be authorized for listing o subject to official notice

Delisting and Deregistration of KFN Com

If the merger is completed, KFN common shares will cease to be listed on the NYSE and will be deregistered under the Ex

KKR Unitholder Approval is N

KKR unitholders are not required to adopt the merger agreement or approve the merger or the issuance of KKR common units i

Litigation Relating to

Since December 19, 2013, fifteen putative stockholder class action lawsuits, referred to in this proxy statement/prospectus a lawsuits, were filed against KFN and certain other defendants in connection with KFN entering into the merger agreement. Each of lawsuits was filed on behalf of a putative class of KFN common shareholders against KFN, the individual members of KFN's directors, KKR, Fund Holdings, and Merger Sub. The merger lawsuits allege variously that the members of the KFN board of direct their fiduciary duties owed to KFN common shareholders by approving the proposed merger for inadequate consideration; a transaction in order to obtain benefits not equally shared by other KFN common shareholders; entering into the merger agreement preclusive deal protection devices; failing to take steps to maximize the value to be paid to the KFN common shareholders; a disclose material information necessary for KFN common shareholders to make a fully informed decision about the proposed merger lawsuits also seek to state claims against KFN, KKR, Fund Holdings.

Table

Sub for aiding and abetting these alleged breaches of fiduciary duties. In addition, certain of the complaints allege that KKR contro means of a management agreement between KFN and KKR Financial Advisors LLC, and that, as a consequence, KKR breached fiding it owed to KFN common shareholders by causing KFN to approve the merger agreement. The relief sought by the plaintiffs is lawsuits includes, among other things, declaratory and injunctive relief concerning the alleged breaches of fiduciary duties, injurprohibiting the consummation of the acquisition, rescission, an accounting by defendants, damages and attorneys' fees and costs, and

Delaware C

Nine of the merger lawsuits were filed in the Court of Chancery of the State of Delaware: (1) Parsons v. KKR Financial Holandal., C.A. No. 9210-CS, filed on December 27, 2013; (2) Bushey v. KKR Financial Holdings LLC et al., C.A. No. 9228-CS, filed on January 7, 2014-CS; (4) Pompano Beach Police & Firefighters' Ret.

Financial Holdings LLC et al., C.A. No. 9236-CS, filed on January 8, 2014; (5) Corwin v. KKR Financial Holdings LLC No. 9237-CS, filed on January 8, 2014; (6) Reiffman v. KKR Financial Holdings LLC et al., C.A. No. 9238-CS, filed on January 13, 2014; (8) Pipefitters Local Union No. 120 Pension Fund v. Fan No. 9247-CS, filed on January 14, 2014; and (9) DeMauro v. Collins et al., C.A. No. 9262-CS, filed on January 17, 2014. The Fushey complaints were voluntarily dismissed on January 29, 2014. The remaining actions in Delaware were consolidated under the KKR Financial Holdings LLC Shareholder Litigation, Consol. C.A. No. 9210-CS, on January 31, 2014, which is referred to statement/prospectus as the Delay.

On February 21, 2014, a Verified Consolidated Class Action Complaint, referred to in this proxy statement/prospectus as t complaint, was filed in the Delaware action. The Delaware complaint asserts three causes of action: (1) that the members of KF directors breached their fiduciary duties of due care and loyalty owed to KFN common shareholders, (2) that KKR is a controlling shareholders with the management agreement and that KKR breached a fiduciary duty of loyalty it allegedly owed to KFN common by causing KFN to approve the merger agreement and (3) that KKR, Fund Holdings, and Merger Sub aided and abetted the allegence breaches by the members of the KFN board

The Delaware complaint seeks declaratory relief, damages, and attorneys' fees and costs, and other relief. On March 7, 2014, a moved to dismiss the Delaware complaint. The foregoing description of the Delaware action incorporates explicitly and is qualified reference to the Delaware complaint, which is attached as Annex E to this proxy statement

California C

Five of the merger lawsuits, referred to in this proxy statement/prospectus as the California actions, were filed in Supe California, County of San Francisco: (1) Braun v. KKR Financial Holdings LLC et al., Case No. CGC-13-536281, filed on December (2) Martin v. KKR Financial Holdings LLC et al., Case No. CGC-13-536297, filed on December 19, 2013; (3) Pariselli v. Holdings L.P. et al., Case No. CGC-13-536378, filed on December 23, 2013; (4) Mancuso v. KKR Financial Holdings LLC et al. CGC-13-536445, filed on December 24, 2013; and (5) Powers v. KKR Financial Holdings LLC et al., Case No. CGC-13-536 December 27, 2013. On February 25, 2014, the California actions were consolidated under the caption In re KKR Financial Holdings LLC et al., Case No. CGC-13-536281, and the Court appointed counsel for Plaintiff Braun as Laws CGC-13-536281, and the Court appointed counsel for Plaintiff Braun as Laws CGC-13-536281.

Table

On January 27, 2014, prior to consolidation of the California actions and appointment of his counsel as Lead Counsel, Plaintif an Amended Class Action Complaint, which is referred to in this proxy statement/prospectus as the Amended Braun complaint. T Braun complaint asserts two causes of action. First, it asserts that the members of KFN's board of directors breached their fiduci loyalty, good faith, candor, due care, and independence owed to KFN common shareholders. Second, it asserts that KKR, Fund I Merger Sub aided and abetted the alleged fiduciary breaches by the members of the KFN board

The Amended Braun Complaint seeks declaratory relief, injunctive relief prohibiting consummation of the proposed transaction and attorneys' fees and costs, and other relief. A consolidated amended complaint has not been filed or designated in the California foregoing description of the California actions incorporates explicitly and is qualified entirely by reference to the Amended Brau which is attached as Annex F to this proxy statemen

Federal (

The fifteenth of the merger lawsuits was filed in the United States District Court for the District of Northern California, which in this proxy statement/prospectus as the federal action: *Bushey v. KKR Financial Holdings LLC et al*, Civ. A. No. 3:14-cv-00 January 31, 2014. On February 12, 2014, the plaintiff in this action moved for an order scheduling a preliminary injunction authorizing the plaintiff to conduct expedited discovery in support of that motion. On March 11, 2014, the plaintiff in this action motion for expedited discovery. A hearing date for the preliminary injunction motion has been set for M

The complaint in the federal action, referred to in this proxy statement/prospectus as the federal complaint, asserts several cause First, the federal complaint asserts that KFN and its board members violated Sections 14(a) of the Exchange Act and Rule 14a-9 thereunder. Second, the federal complaint alleges that the individual defendants violated Section 20(a) of the Exchange Act. Third complaint alleges breach of fiduciary duty claims against the individual KFN board members. Fourth, the federal complaint allege fiduciary duty of disclosure claims against the individual KFN board members. Fifth, the federal complaint asserts a breach of fiduciary duty of disclosure claims against the individual KFN board members. Fifth, the federal complaint asserts claims of aiding and abetting the KFN board's breaches of fiduciary duties a

The federal complaint seeks injunctive relief prohibiting consummation of the proposed transaction, rescission, an accounting b for damages, and attorneys' fees and costs, and other relief. The defendants have not been served in the federal action. The foregoing of the federal action incorporates explicitly and is qualified entirely by reference to the federal complaint, which is attached as An proxy statemen

KFN and the other defendants named in the merger lawsuits dispute certain allegations made therein and intend to vigorousl claims asserted a

<u>Table</u>

RISK FACTORS

In addition to the other information included and incorporated by reference into this proxy statement/prospectus, including addressed in the section titled "Cautionary Statement Regarding Forward-Looking Statements," you should carefully consider the for before deciding whether to vote for the adoption of the merger agreement and the merger. In addition, you should read and careful the risks associated with each of KKR and KFN and their respective businesses. These risks can be found in KKR's and KFN's respective on Form 10-K for the year ended December 31, 2013, as may be updated by subsequent Quarterly Reports on Form 10-Q, are filed with the SEC and incorporated by reference into this proxy statement/prospectus. For further information regarding the incorporated into this proxy statement/prospectus by reference, please see the section titled "Where You Can Find More Information on page 234 of this proxy statement/prospectus. Realization of any of the risks described below, any of the events described under Statement Regarding Forward-Looking Statements" or any of the risks or events described in this proxy statement/prospectus documents incorporated by reference could have a material adverse effect on KKR's, KFN's or the combined organization businesses, financial condition, cash flows and results of operations and could result in a decline in the trading prices of the common equi

Risk Factors Related to

Because the exchange ratio is fixed and because the market price of KKR common units will fluctuate prior to the consumn merger, KFN common shareholders cannot be sure of the market value of the KKR common units they will receive as merger or relative to the value of KFN common shares the

The market value of the consideration that KFN common shareholders will receive in the merger will depend on the trading p common units at the closing of the merger. The exchange ratio that determines the number of KKR common units that K shareholders will receive in the merger is fixed. This means that there is no mechanism contained in the merger agreement that wou number of KKR common units that KFN common shareholders will receive based on any decreases in the trading price of KKR co The KKR common unit price at the closing of the merger may be lower than it was on the date the merger agreement was signed, the proxy statement/prospectus or the date of the special meeting. If the KKR common unit price at the closing of the merger is less the common unit price on the date that the merger agreement was signed, then the market value of the consideration received by K shareholders will be less than the value that was contemplated at the time the merger agreement

KKR common unit price changes may result from a variety of factors, including general market and economic condition affecting KKR's industry generally or those in which KKR holds investments, changes in KKR's business, operations and price regulatory considerations. Many of these factors are beyond KKR's control. See the section entitled "Risk Factors" contained documents incorporated by reference in this proxy statement/prospectus and referred to under "Where You Can Find More beginning on page 234 of this proxy statement/prospectus. For historical and current market prices of KKR common units and K shares, see "Summary Comparative Unit Prices and Distributions" beginning on page 27 of this proxy statement/prospectus. You obtain current market quotations for KKR common units in deciding whether to vote for the adoption of the merge

The market price of KKR common units after the merger may be affected by factors different from those affecting KFN common units after the merger may be affected by factors different from those affecting KFN common units after the merger may be affected by factors different from those affecting KFN common units after the merger may be affected by factors different from those affecting KFN common units after the merger may be affected by factors different from those affecting KFN common units after the merger may be affected by factors different from those affecting KFN common units after the merger may be affected by factors different from those affecting KFN common units after the merger may be affected by factors different from those affecting KFN common units after the merger may be affected by factors different from those affecting KFN common units after the merger may be affected by factors different from the merger may be affected by the merger may be affected by the merger may be affected by the merger may be affecte

Upon completion of the merger, holders of KFN common shares will become holders of KKR common units. The businesses of from those of KFN in important to

Table

accordingly, the results of operations of KKR after the merger, as well as the market price of the KKR common units, may be affected different from those currently affecting the independent results of operations of KFN. For further information on the businesses KFN and certain factors to consider in connection with those businesses, see the documents incorporated by reference in statement/prospectus and referred to under "Where You Can Find More Information" beginning on page 234 of this proxy statemen

KKR and KFN may be unable to obtain the regulatory clearances required to complete the merger or, in order to do so, KKR and be required to comply with material restrictions or satisfy materia

The merger is subject to review by the Antitrust Division and the FTC under the HSR Act, and potentially state regulatory aut closing of the merger is subject to the condition that there be no outstanding judgment, injunction, order or decree by a government prohibiting or enjoining the merger or the other transactions contemplated by the merger agreement. KKR and KFN can provide that all required regulatory clearances will be obtained. If a governmental authority asserts objections to the merger, KKR may be divest some assets in order to obtain antitrust clearance. There can be no assurance as to the cost, scope or impact of the actions required to obtain antitrust approval. In addition, the merger agreement provides that KKR is not required to take any action of restriction if it would reasonably be expected to require that KKR, KFN or their subsidiaries or affiliates dispose of or hold separate business or assets or would reasonably be expected to result in any material limitations on KKR, KFN or their subsidiaries or affiliand operate all or a material portion of their businesses or assets. If KKR must take such actions, it could be detrimental to it or to the organization following the consummation of the merger. Furthermore, these actions could have the effect of delaying or preventing of the proposed merger or imposing additional costs on or limiting the revenues of the combined organization following the consummation.

Although the FTC granted early termination of the statutory waiting period under the HSR Act on March 6, 2014, the Antitrus the FTC could take action under the antitrust laws to prevent or rescind the merger, require the divestiture of assets or seek of Additionally, state attorneys general could seek to block or challenge the merger as they deem necessary or desirable in the public ir time, including after completion of the transaction. In addition, in some circumstances, a third party could initiate a private action unlaws challenging or seeking to enjoin the merger, before or after it is completed. KKR may not prevail and may incur significantly defending or settling any action under the antitrust laws to prevent or after it is completed.

The fairness opinion rendered to the transaction committee of the board of directors of KFN by its financial advisor was a financial analyses performed by KFN's financial advisor, which considered factors such as market and other conditions then it financial forecasts and other information made available to KFN's financial advisors, as of the date of its opinion. As a result, does not reflect changes in events or circumstances after the date of this opinion. KFN has not obtained, and does not expect updated fairness opinion from its financial advisor reflecting changes in circumstances that may have occurred since the si

The fairness opinion rendered to the transaction committee of the board of directors of KFN by Sandler O'Neill was provided in with, and at the time of, the transaction committee's and the board of directors' evaluation of the merger and the merger agreement. was based on the financial analyses performed by the financial advisor, which considered market and other conditions then if financial forecasts and other information made available to it, as of the date of its opinion, which may have changed, or may changed of the opinion. KFN has not obtained an updated opinion as of the date of this proxy statement/prospectus from its financial advisor.

Table

does not expect to obtain an updated opinion prior to the completion of the merger. Changes in the operations and prospects of K general market and economic conditions and other factors which may be beyond the control of KKR and KFN, and on which the fair was based, may have altered the value of KKR or KFN or the prices of KKR common units or KFN common shares since the opinion, or may alter the values and prices by the time the merger is completed. The opinion does not speak as of any date other that the opinion. For a description of the opinion that the transaction committee received from its financial advisor, please references. Opinion of the Financial Advisor to the KFN Transaction Committee beginning on page 42 of this proxy statement.

KFN is subject to provisions that limit its ability to pursue alternatives to the merger, which could discourage a potential compet of KFN from making a favorable alternative transaction proposal and, in specified circumstances under the merger agreement, contact the KFN to pay a termination payment of \$26.25 million to the second secon

Under the merger agreement, KFN is restricted from entering into alternative transactions. Unless and until the merger terminated, subject to specified exceptions (which are discussed in more detail in "The Merger Agreement No Solicitation by KFN Proposals" beginning on page 139 of this proxy statement/prospectus), KFN is restricted from initiating, soliciting or knowingly ence inquiries with respect to, or negotiating or providing confidential information or data relating to, any proposal or offer for acquisition proposal with any person. Under the merger agreement, in the event of a potential change by the board of directors or recommendation with respect to the proposed merger in light of a superior proposal, KFN must provide KKR with four days' not KKR to propose an adjustment to the terms of the merger agreement. These provisions could discourage a third party that may have acquiring all or a significant part of KFN from considering or proposing that acquisition, even if the third party were preconsideration with a higher per share market value than the market value proposed to be received or realized in the merger, or mig potential competing acquirer of KFN proposing to pay a lower price than it would otherwise have proposed to pay because of the ado of the termination payment that may become payable in specified circumstances (which are discussed in more detail in Agreement Termination Payment and Expenses" beginning on page 143 of this proxy statement/prospectus). For a discussion of to on KFN soliciting or entering into a takeover proposal or alternative transaction and KFN's board of directors' ability recommendation, see "The Merger Agreement No Solicitation by KFN of Alternative Proposals," and "The Merger Agreement Poorspectus, Board Recommendation" beginning on pages 137 and 138 of this proxy statement/prospectus,

Directors and officers of KFN have certain interests that are different from those of KFN common shareholde

Directors and officers of KFN are participants in arrangements that give them interests in the merger that may be different for addition to, your interests as a common shareholder of KFN. You should consider these interests in voting on the merger. The interests are described under "Special Factors" Interests of Directors and Executive Officers of KFN in the Merger beginning on proxy statemen

KKR and KFN will incur substantial transaction-related costs in connection with

KKR and KFN expect to incur a number of non-recurring transaction-related costs associated with completing the merger and the operations of the two companies. These fees and costs will be substantial. Non-recurring transaction costs include, but are not linearly paid to legal, financial and accounting advisors, filing fees and printing costs, estimates of which are seen to the companies.

Table

"Special Factors Fees and Expenses" beginning on page 98 of this proxy statement/prospectus. Additional unanticipated costs may the integration of the businesses of KK

Failure to successfully combine the businesses of KFN and KKR in the expected timeframe may adversely affect the future recombined organization, and, consequently, the value of the KKR common units that KFN common shareholders receive a

The success of the proposed merger will depend, in part, on the ability of KKR to realize the anticipated benefits from consumers that the combined organization is not able to achieve these objectives, or is not able to achieve these objectiv

Failure to complete the merger, or significant delays in completing the merger, could negatively affect the trading prices of K units and KFN common shares and the future business and financial results of KK

Completion of the merger is not assured and is subject to risks, including the risks that approval of the merger by the K shareholders or by governmental agencies is not obtained or that other closing conditions are not satisfied. If the merger is not continue are significant delays in completing the merger, the trading prices of KKR common units and KFN common shares and the future business and financial results of KKR and KFN could be negatively affected, and each of them will be subject to several risks.

the parties may be liable for damages to one another under the terms and conditions of the merger

negative reactions from the financial markets, including declines in the price of KKR common units or KFN conduction due to the fact that current prices may reflect a market assumption that the merger will be

having to pay certain significant costs relating to the merger, including, in the case of KFN in certain circ termination payment of \$26.25 million or reimbursement of KKR's costs, fees and expenses up to \$7.5 million, as described in "The Merger Agreement Termination Payment and Expenses" beginning on page 143 statement/pro

the attention of KFN's manager and management of KKR will have been diverted to the merger rather than each own operations and pursuit of other opportunities that could have been beneficial to the

Purported class action complaints have been filed against KFN, KKR, KFN's board of directors, Fund Holdings and Merger Sub the merger, and an unfavorable judgment or ruling in these lawsuits could prevent or delay the consummation of the proposed result in subst

Fifteen class action lawsuits are currently pending that challenge the merger. Each lawsuit names as defendants some or all of the individual members of KFN's board of directors, Fund Holdings and Merger Sub. Among other remedies, the plaintiffs seek proposed merger. If these lawsuits are not dismissed or otherwise resolved, they could prevent and/or delay completion of the merg in substantial costs to KFN and KKR, including any costs associated with the indemnification of directors. Additional lawsuits may connection with the proposed merger. There can be no assurance that any of the defendants will prevail in the pending litigat

Table

future litigation. The defense or settlement of any lawsuit or claim may adversely affect the combined organization's business, financ or results of operations. See "Special Factors" Litigation Relating to the Merger" beginning on page 101 of this proxy stateme

If the merger is approved by KFN common shareholders, the date that those shareholders will receive the merger consideration

As described in this proxy statement/prospectus, completing the proposed merger is subject to several conditions, not all controllable or waiveable by KKR or KFN, and KKR and KFN have agreed to use reasonable best efforts to take such actions so the will take place on the last business day of a calendar month. Accordingly, if the proposed merger is approved by KFN common share date that KFN common shareholders will receive the merger consideration depends on the completion date of the merger, which

KFN common shareholders will have a reduced ownership and voting interest after the merger and will exercise material influence over n

KFN common shareholders currently have the right to vote in the election of the KFN board of directors and certain other matt KFN. When the merger occurs, each KFN common shareholder that receives KKR common units will become a holder of KKR of with a percentage ownership of the combined organization that is much smaller than such holder's percentage ownership of KFN KKR common units are not entitled to elect the general partner, and are not entitled to elect the directors of KKR's general partner holders of KKR common units have only limited voting rights and, therefore, limited or no ability to influence management's decision KKR's business. Because of this, KFN common shareholders will have less influence on the management and policies of KKR thenow on the management and policies of KFN. See "Comparison of KKR Common Units and KFN Common Shares" and "Description Limited Partnership Agreement" beginning on pages 212 and 201 of this proxy statement/prospectus,

KKR common units to be received by KFN common shareholders as a result of the merger have materially different right

Following completion of the merger, KFN common shareholders will no longer hold KFN common shares, but will instead I KKR common units. KKR is a limited partnership, and KFN is a limited liability company. There are important differences between KFN common shareholders and the rights of holders of KKR common units. See "Comparison of KKR Common Units and KI Shares" beginning on page 212 of this proxy statement/prospectus for a discussion of the different rights associated with KFN common units. See also "Risks Related to the Ownership of KKR Common Units" beginning on page 108 statemen

Risks Related to the Ownership of KKR Con

As a limited partnership, KKR qualifies for some exemptions from the corporate governance and other requirements of

KKR is a limited partnership and, as a result, qualifies for exceptions from certain corporate governance and other requirements of the NYSE. Pursuant to these exceptions, limited partnerships may elect, and KKR has elected, not to comply with certa governance requirements of the NYSE, including the requirements: (i) that the listed company have a nominating and corporate committee that is composed entirely of independent directors; and (ii) that the listed company have a compensation committee that entirely of independent directors.

Table

In addition, as a limited partnership, KKR is not required to hold annual unitholder meetings and is not required to obtain the vote unitholders for issuances of publicly traded units regardless of whether such issuance would be in an amount equal to or greater that total number of outstanding publicly traded units. Accordingly, an investor in KKR common units does not have the same protection equity holders of entities that are subject to all of the corporate governance requirements of

KKR's founders are able to determine the outcome of any matter that may be submitted for a vote of KKR's limit

As of March 18, 2014, (i) KKR Holdings owns 393,357,457 KKR Group Partnership units, and (ii) KKR's senior en non-employee operating consultants who hold interests in KKR's business through KKR Holdings, who are referred to statement/prospectus as KKR's principals, generally have sufficient voting power to determine the outcome of those few matters submitted for a vote of the holders of KKR's common units, including a merger or consolidation of KKR's business, a sale of all or all of KKR's assets and amendments to KKR's amended and restated limited partnership agreement, which is referred to statement/prospectus as KKR's partnership agreement, that may be material to holders of KKR common units. In addition, KKR's agreement contains provisions that enable KKR to take actions that would materially and adversely affect all holders of KKR common units upon the majority vote of all outstanding voting units, and since, following the mer majority of KKR's voting units will be controlled by KKR Holdings, KKR Holdings effectively has the ability to take action materially and adversely affect the holders of KKR common units either as a whole or as a par

The voting rights of holders of KKR common units are further restricted by provisions in KKR's partnership agreement stated KKR common units held by a person that beneficially owns 20% or more of any class of KKR common units then outstanding (other managing partner or its affiliates, or a direct or subsequently approved transferee of KKR's managing partner or its affiliates) cannot any matter. KKR's partnership agreement also contains provisions limiting the ability of the holders of KKR common units to call acquire information about KKR's operations, and to influence the manner or direction of KKR's management. KKR's partnership agreement or restrict KKR's managing partner's ability to take actions that may result in KKR's partnership being treated as an entity corporation for U.S. federal (and applicable state) income tax purposes. Furthermore, holders of KKR common units would not lead to the dissenters' rights of appraisal under KKR's partnership agreement or applicable Delaware law in the event of a merger or consolidation KKR, a sale of substantially all of KKR's assets or any other transaction or event. See "Description of KKR's Limited Partnership beginning on page 201 of this proxy statemen

KKR's partnership agreement contains provisions that reduce or eliminate duties (including fiduciary duties) of KKR's manage and limit remedies available to holders of KKR common units for actions that might otherwise constitute a breach of duty. It will for holders of KKR common units to successfully challenge a resolution of a conflict of interest by KKR's managing parties.

KKR's partnership agreement contains provisions that require holders of KKR common units to waive or consent to condumnaaging partner and its affiliates that might otherwise raise issues about compliance with fiduciary duties or applicable law. I KKR's partnership agreement provides that when KKR's managing partner is acting in its individual capacity, as opposed to in it KKR's managing partner, it may act without any fiduciary obligations to holders of KKR common units, whatsoever. When KKR partner, in its capacity as K

<u>Table</u>

partner, or KKR's conflicts committee is permitted to or required to make a decision in its "sole discretion" or "discretion" or "necessary or appropriate" or "necessary or advisable," then KKR's managing partner or the conflicts committee will be entitled to a such interests and factors as it desires, including its own interests, and will have no duty or obligation (fiduciary or otherwise consideration to any interest of or factors affecting KKR or any holder of KKR common units and will not be subject to any different imposed by KKR's partnership agreement, the Delaware Revised Uniform Limited Partnership Act, which is referred to statement/prospectus as the Delaware Limited Partnership Act, or under any other law, rule or regulation or in equity. These standard obligations to which KKR's managing partner would otherwise be held. See also "KKR is a Delaware limited partnership provisions in its limited partnership agreement regarding exculpation and indemnification of its officers and directors that did Delaware General Corporation Law in a manner that may be less protective of the interests of holders of KKR common units" and of KKR's Limited Partnership Agreement" beginning on page 201 of this proxy statemen

The above modifications of fiduciary duties are expressly permitted by Delaware law. Hence, KKR and holders of KKR commonly have recourse and be able to seek remedies against KKR's managing partner if KKR's managing partner breaches its obligation KKR's partnership agreement. Unless KKR's managing partner breaches its obligations pursuant to KKR's partnership agreement holders of KKR common units will not have any recourse against KKR's managing partner even if KKR's managing partner we manner that was inconsistent with traditional fiduciary duties. Furthermore, even if there has been a breach of the obligations set for partnership agreement, KKR's partnership agreement provides that KKR's managing partner and its officers and directors will not KKR or holders of KKR common units, for errors of judgment or for any acts or omissions unless there has been a final and not judgment by a court of competent jurisdiction determining that KKR's managing partner or its officers and directors acted in bad faith in fraud or willful misconduct. These provisions are detrimental to the holders of KKR common units because they restrict the remedit to holders of KKR common units for actions that without such limitations might constitute breaches of duty including fiduciary.

Whenever a potential conflict of interest exists between KKR and its managing partner, KKR's managing partner may resolve of interest. If KKR's managing partner determines that its resolution of the conflict of interest is on terms no less favorable to KK generally being provided to or available from unrelated third parties or is fair and reasonable to KKR, taking into account the trelationships between KKR and its managing partner, then it will be presumed that in making this determination, KKR's managing in good faith. A holder of KKR common units seeking to challenge this resolution of the conflict of interest would bear to overcoming such presumption. This is different from the situation with Delaware corporations, where a conflict resolution by an interest would be presumed to be unfair and the interested party would have the burden of demonstrating that the resolution

Also, if KKR's managing partner obtains the approval of the conflicts committee of its managing partner, the resolution will be deemed to be fair and reasonable to KKR and not a breach by KKR's managing partner of any duties it may owe to KKR or hole common units. This is different from the situation with Delaware corporations, where a conflict resolution by a committee consist independent directors may, in certain circumstances, merely shift the burden of demonstrating unfairness to the plaintiff. Holders where the merger or otherwise hold a KKR common unit will be treated as having consented to the provisions set forth in KKR's agreement, including provisions regarding conflicts of interest situations that, in the absence of such provisions, might be considered fiduciary or other duties under applicable states.

Table

result, holders of KKR common units will, as a practical matter, not be able to successfully challenge an informed decision by

KKR has also agreed to indemnify KKR's managing partner and any of its affiliates and any member, partner, tax matters par director, employee agent, fiduciary or trustee of KKR's partnership, KKR's managing partner or any of KKR's affiliates and specified persons, to the fullest extent permitted by law, against any and all losses, claims, damages, liabilities, joint or sever (including legal fees and expenses), judgments, fines, penalties, interest, settlements or other amounts incurred by KKR's managing these other persons. KKR has agreed to provide this indemnification unless there has been a final and non-appealable judgment competent jurisdiction determining that these persons acted in bad faith or engaged in fraud or willful misconduct. KKR has a provide this indemnification for criminal

Any claims, suits, actions or proceedings concerning the matters described above or any other matter arising out of or relating in KKR's partnership agreement may only be brought in the Court of Chancery of the State of Delaware or, if such court does not matter jurisdiction thereof, any other court in the State of Delaware with subject matter

The market price and trading volume of KKR common units may be volatile, which could result in rapid and substantial losses for KKR co

The market price of KKR common units may be highly volatile and could be subject to wide fluctuations. In addition, the tradin KKR common units may fluctuate and cause significant price variations to occur. If the market price of KKR common usignificantly, holders of KKR common units may be unable to sell the KKR common units at an attractive price, if at all. The market price or decline significantly in the future. Some of the factors that could negatively affect the procommon units or result in fluctuations in the price or trading volume of KKR common units or result in fluctuations.

variations in KKR's quarterly operating results or distributions, which may be

KKR's policy of taking a long-term perspective on making investment, operational and strategic decisions, which to result in significant and unpredictable variations in KKR's quart

failure to meet analysts' earning

publication of research reports about KKR or the investment management industry or the failure of securities cover KKR common units

additions or departures of KKR's principals and other key management

adverse market reaction to any indebtedness KKR may incur or securities KKR may issue it

changes in market valuations of similar

speculation in the press or investment

changes or proposed changes in laws or regulations or differing interpretations thereof affecting KKR' enforcement of these laws and regulations, or announcements relating to the

a lack of liquidity in the trading of KKR co

<u>Table</u>

adverse publicity about the investment management or private equity industry generally or individual scandals,

the risks relating to the merger described in the section titled "Risks Related to the N

general market and economi

An investment in KKR common units is not an investment in any of its funds, and the assets and revenues of its funds are

KKR common units are securities of KKR & Co. L.P. only. While KKR's historical consolidated and combined financial includes financial information, including assets and revenues, of certain funds on a consolidated basis, and KKR's future financial will continue to consolidate certain of these funds, such assets and revenues are available to the fund and not to KKR except to a litthrough management fees, carried interest or other incentive income, distributions and other proceeds arising from agreement.

The price of KKR common units may decline due to the large number of common units eligible for future sale, for exchange, pursuant to KKR's equity in

The market price of KKR common units could decline as a result of sales of a large number of common units in the market or the that such sales could occur. These sales, or the possibility that these sales may occur, also might make it more difficult for Kl common units in the future at a time and at a price that it deems appropriate. As of March 18, 2014, there were 300,354,288 KKR coutstanding, which amount excludes KKR common units beneficially owned by KKR Holdings in the form of KKR Group Part discussed below, common units available for future issuance under KKR's equity incentive plans and KKR common units available issuance in connection with acquisitions may be a common units available for future issuance under KKR's equity incentive plans and KKR common units available for future issuance under KKR's equity incentive plans and KKR common units available for future issuance under KKR's equity incentive plans and KKR common units available for future issuance under KKR's equity incentive plans and KKR common units available for future issuance under KKR's equity incentive plans and KKR common units available for future issuance under KKR's equity incentive plans and KKR common units available for future issuance under KKR's equity incentive plans and KKR common units available for future issuance under KKR's equity incentive plans and KKR common units available for future issuance under KKR's equity incentive plans and KKR common units available for future issuance under KKR's equity incentive plans and KKR common units available for future issuance under KKR's equity incentive plans and KKR common units available for future issuance under KKR's equity incentive plans and KKR common units available for future issuance under KKR's equity incentive plans and KKR common units available for future issuance under KKR's equity incentive plans and KKR common units available for future issuance under KKR's equity incentive plans and KKR common units available for future issuance under KKR's equity incentive plans

As of March 18, 2014, KKR Holdings owns 393,357,457 KKR Group Partnership units that may be exchanged, on a quarte KKR common units on a one-for-one basis, subject to customary conversion rate adjustments for splits, unit distributions and recla Except for interests held by KKR's founders and certain interests held by other principals that were vested upon grant or that have the date of grant, interests in KKR Holdings that are held by KKR's principals and other persons are subject to time-base performance-based vesting and, following such vesting, additional restrictions in certain cases on exchanges for a period of one or two market price of KKR common units could decline as a result of the exchange or the perception that an exchange may occur of a larg KKR Group Partnership units for KKR common units. These exchanges, or the possibility that these exchanges may occur, also more difficult for holders of KKR common units to sell the KKR common units in the future at a time and at a price that they deem

In addition, KKR will continue to issue additional common units pursuant to KKR's equity incentive plans. The total number units which may be issued under KKR's equity incentive plans is equivalent to 15% of the number of fully diluted common units or of the beginning of the year. The amount may be increased each year to the extent that KKR issues additional equity. In add partnership agreement authorizes KKR to issue an unlimited number of additional partnership securities and options, rights, appreciation rights relating to partnership securities for the consideration and on the terms and conditions established by KKR's manain its sole discretion without the approval of holders of KKR common units. In accordance with the Delaware Limited Partnership provisions of KKR's partnership agreement, KKR may also issue additional partner interests that have designations, preferences, riand duties that are different from, and may be senior to, those applicable to KKR co

Table

KKR's structure involves complex provisions of U.S. federal income tax laws for which no clear precedent or authority may these structures also are subject to potential legislative, judicial or administrative change and differing interpretations, provides the structures also are subject to potential legislative, judicial or administrative change and differing interpretations, provides the structure of the struc

The U.S. federal income tax treatment of KKR common unitholders depends in some instances on determinations of fact and in of complex provisions of U.S. federal income tax laws for which no clear precedent or authority may be available. You should be a U.S. federal income tax rules are constantly under review by persons involved in the legislative process, the Internal Revenue Servic in this proxy statement/prospectus as the IRS, and the U.S. Department of the Treasury, referred to in this proxy statement/pros Treasury, frequently resulting in revised interpretations of established concepts, statutory changes, revisions to regulation modifications and interpretations. The present U.S. federal income tax treatment of owning KKR common units may be administrative, legislative or judicial interpretation at any time, and any such action may affect investments and commitments prev For instance, changes to the U.S. federal tax laws and interpretations thereof could make it more difficult or impossible for KKR to a partnership that is not taxable as a corporation for U.S. federal income tax purposes, affect the tax considerations of owning K units, change the character or treatment of portions of KKR's income (including, for instance, the treatment of carried interest as ordi rather than capital gain) and adversely impact your investment in KKR common units. See the discussion below under " The U.S considered legislation that would have (i) in some cases after a ten-year period, precluded KKR from qualifying as a partnershi KKR to hold carried interest through taxable subsidiary corporations and (ii) taxed certain income and gains at increased rates. I legislation were to be enacted and apply to KKR, the after-tax income and gain related to KKR's business, as well as the market p common units, could be reduced." KKR's organizational documents and agreements give KKR's managing partner broad authority t amended and restated partnership agreement from time to time as KKR's managing partner determines to be necessary or appropria the consent of KKR common unitholders, to address changes in U.S. federal, state and local income tax regulations, legislation or in In some circumstances, such revisions could have a material adverse impact on some or all KKR common unitholders. For inst managing partner could elect at some point to treat KKR as an association taxable as a corporation for U.S. federal (and applicable s tax purposes. If KKR's managing partner were to do this, the U.S. federal income tax consequences of owning KKR common ur materially different. Moreover, certain assumptions and conventions will be applied in an attempt to comply with applicable rules income, gain, deduction, loss and credit to common unitholders in a manner that reflects such common unitholders' beneficial partnership items, taking into account variation in ownership interests during each taxable year because of trading activity. Ho assumptions and conventions may not be in compliance with all aspects of applicable tax requirements. It is possible that the IR successfully that the conventions and assumptions used by KKR do not satisfy the technical requirements of the Code and regulations and could require that items of income, gain, deductions, loss or credit, including interest deductions, be adjusted, r disallowed in a manner that adversely affects KKR common

The U.S. Congress has considered legislation that would have (i) in some cases after a ten-year period, precluded KKR from que partnership or required KKR to hold carried interest through taxable subsidiary corporations and (ii) taxed certain income increased rates. If any similar legislation were to be enacted and apply to KKR, the after-tax income and gain related to KKR's well as the market price of KKR units, could

Over the past several years, a number of legislative and administrative proposals have been introduced and, in certain case passed by the U.S. House of Representatives. In May 2010, the U.S. House of Representatives passed legislation, or the "May 2010"

Table

in general, treated income and gains, including gain on sale, attributable to an interest in an investment services partnership interest, income subject to a new blended tax rate that is higher than under current law, except to the extent such ISPI would have been cons the legislation to be a qualified capital interest. Your interest in KKR, KKR's interest in KKR Fund Holdings L.P. and the interest Fund Holdings L.P. holds in entities that are entitled to receive carried interest may have been classified as ISPIs for purposes of this The U.S. Senate considered but did not pass similar legislation. On February 14, 2012, Representative Levin introduced similar legis "2012 Levin bill," that would tax carried interest at ordinary income tax rates (which would be higher than the proposed blended read that the proposed blended read to the proposed ble

Both the May 2010 House bill and the 2012 Levin bill provided that, for taxable years beginning ten years after the date of income derived with respect to an ISPI that is not a qualified capital interest and that is subject to the rules discussed above would qualifying income requirements under the publicly traded partnership rules. Therefore, if similar legislation is enacted, following se period, KKR would be precluded from qualifying as a partnership for U.S. federal income tax purposes or be required to hold a through corporations, possibly U.S. corporations. If KKR were taxed as a U.S. corporation or required to hold all ISPIs through KKR's effective tax rate would increase significantly. The federal statutory rate for corporations is currently 35%. In addition, K subject to increased state and local taxes. Furthermore, you could be subject to tax on KKR's conversion into a corporation or any required in order for KKR to hold KKR's ISPIs through a corporation. KKR's principals and other professionals could face additional consequences under the legislation, which might thereby adversely affect KKR's ability to offer attractive incentive opportunity.

On September 12, 2011, the Obama administration submitted similar legislation to Congress in the American Jobs Act the income and gain, now treated as capital gains, including gain on disposition of interests, attributable to an ISPI at rates higher that gains rate applicable to such income under current law, with an exception for certain qualified capital interests. The proposed legislation characterize certain income and gain in respect of ISPIs as non-qualifying income under the publicly traded partnership rules aft transition period from the effective date, with an exception for certain qualified capital interests. This proposed legislation follows statements by the Obama administration in support of changing the taxation of carried interest. Furthermore, in the proposed American the Obama administration proposed that current law regarding the treatment of carried interest be changed for taxable years December 31, 2012 to subject such income to ordinary income tax. In its published revenue proposal for 2013, the Obama administration's published revenue proposals for 2010, 2011 and 2012 contained similar administration's published revenue proposals for 2010, 2011 and 2012 contained similaries.

States and other jurisdictions have also considered legislation to increase taxes with respect to carried interest. For example, N periodically considered legislation under which you could be subject to New York state income tax on income in respect of KKR or as a result of certain activities of KKR's affiliates in New York, although it is unclear when or whether such legislation will

If the proposed legislation described above or any similar legislation were to be enacted and apply to KKR, the after-tax incorrelated to KKR's business, as well as the market price of KKR units, could

Table

If KKR were treated as a corporation for U.S. federal income tax or state tax purposes, then KKR's distributions to you would be reduced and the value of KKR common units could be advers

The value of your investment in KKR depends in part on KKR's being treated as a partnership for U.S. federal income tax purpose that 90% or more of KKR's gross income for every taxable year consist of qualifying income, as defined in Section 7704 and that KKR's partnership not be registered under the Investment Company Act. Qualifying income generally includes divide capital gains from the sale or other disposition of stocks and securities and certain other forms of investment income. KKR intends KKR's investments so as to satisfy these requirements, including by generally holding investments that generate non-qualifying income or more subsidiaries that are treated as corporations for U.S. federal income tax purposes. Nonetheless, KKR may not requirements, may not correctly identify investments that should be owned through corporate subsidiaries, or current law may chacause, in any of these events, KKR to be treated as a corporation for U.S. federal income tax purposes or otherwise subject KKR to income tax. KKR has not requested, and does not plan to request, a ruling from the IRS, on this or any other matter aff

If KKR were treated as a corporation for U.S. federal income tax purposes, KKR would pay U.S. federal, state and local in KKR's taxable income at the applicable tax rates. Distributions to you would generally be taxed again as corporate distributions, and gains, losses, deductions or credits would otherwise flow through to you. Because a tax would be imposed upon KKR as a corporate distributions to you would be substantially reduced which could cause a reduction in the value of KKR co

Current law may change, causing KKR to be treated as a corporation for U.S. federal or state income tax purposes or otherwise KKR to entity-level taxation. See " The U.S. Congress has considered legislation that would have (i) in some cases after a terprecluded KKR from qualifying as a partnership or required KKR to hold carried interest through taxable subsidiary corporations a certain income and gains at increased rates. If any similar legislation were to be enacted and apply to KKR, the after-tax income and to KKR's business, as well as the market price of KKR common units, could be reduced." Several states are evaluating was partnerships to entity-level taxation through the imposition of state income, franchise or other forms of taxation. If any state were to upon KKR as an entity, KKR's distributions to you would

You will be subject to U.S. federal income tax on your share of KKR's taxable income, regardless of whether you rece distributions, and may recognize income in excess of cash d

As long as 90% of KKR's gross income for each taxable year constitutes qualifying income as defined in Section 7704 of t KKR is not required to register as an investment company under the Investment Company Act on a continuing basis, and assumin change in law, KKR will be treated, for U.S. federal income tax purposes, as a partnership and not as an association or a purpartnership taxable as a corporation. As a result, a U.S. KKR common unitholder will be subject to U.S. federal, state, local and some cases, foreign income taxation on its allocable share of KKR's items of income, gain, loss, deduction and credit (including share of those items of any entity in which KKR invests that is treated as a partnership or is otherwise subject to tax on a flow-throut each of KKR's taxable years ending with or within the unitholder's taxable year, regardless of whether or when such unitholder a distributions. See "The U.S. Congress has considered legislation that would have (i) in some cases after a ten-year period, precluding qualifying as a partnership or required KKR to hold carried interest through taxable subsidiary corporations and (ii) taxed certain gains at incre

Table

any similar legislation were to be enacted and apply to KKR, the after-tax income and gain related to KKR's business, as well as the of KKR common units, could

You may not receive cash distributions equal to your allocable share of KKR's net taxable income or even the tax liability that that income. In addition, certain of KKR's holdings, including holdings, if any, in a controlled foreign corporation, or a CFC, a parainvestment company, or a PFIC, or entities treated as partnerships for U.S. federal income tax purposes, may produce taxable income receipt of cash relating to such income, and holders of KKR common units that are U.S. taxpayers may be required to take such account in determining their taxable income. In the event of an inadvertent termination of the partnership status for which the IRS limited relief, each holder of KKR common units may be obligated to make such adjustments as the IRS may require to maintain KK a partnership. Such adjustments may require the holders of KKR common units to recognize additional amounts in income during which they hold such units. In addition, because of KKR's methods of allocating income and gain among holders of KKR common may be taxed on amounts that accrued economically before you became a KKR common unitholder. For example, phantom incomportfolio or due to operational activities may arise during a month and be allocated to you, creating taxable liability that KKR would in a quarterly distribution because KKR typically considers tax distributions on an annual basis. Consequently, you may recogning without received income without received.

Although KKR expects that distributions KKR makes should be sufficient to cover a holder's tax liability in any give attributable to its investment in KKR, no assurances can be made that this will be the case. KKR will be under no obligation to make distribution and, in certain circumstances, may not be able to make any distributions or will only be able to make distributions in than a holder's tax liability attributable to its investment in KKR. In addition, KKR anticipates making quarterly distributions considers tax distributions on an annual basis but allocates taxable income on a monthly basis. As a result, if you dispose of your K units, you may be allocated taxable income during the time you held your KKR common units without receiving any cash corresponding to that period. Accordingly, each holder should ensure that it has sufficient cash flow from other sources to pay all tax

Tax gain or loss on disposition of KKR common units could be more or less th

If you sell your KKR common units, you will recognize a gain or loss equal to the difference between the amount realized and y tax basis allocated to those KKR common units. Prior distributions to you in excess of the total net taxable income allocated to y decreased the tax basis in your KKR common units. Therefore, such excess distributions will increase your taxable gain, or decrease loss, when the KKR common units are sold and may result in a taxable gain even if the sale price is less than the original cost. A p amount realized, whether or not representing gain, may be ordinary increase.

KKR common unitholders may be allocated taxable gain on the disposition of certain assets, even if they did not share in the appreciation inherent in

KKR and KKR's intermediate holding companies will be allocated taxable gains and losses recognized by the KKR Group based upon KKR's percentage ownership in each KKR Group Partnership. KKR's share of such taxable gains and losses generallocated pro rata to KKR common unitholders. In some circumstances, under the U.S. federal income tax rules affecting partnerships, the taxable gain or loss allocated to a KKR common unitholder may not correspond to that unitholder's share of tappreciation or depreciation in the particular asset. This is primarily an issue of the timing of the payment of tax, rather than a net in liability, because the gain or loss allocation would generally be expected to be offset as a KKR common unitholder sold co

Table

Non-U.S. persons face unique U.S. tax issues from owning KKR common units that may result in adverse tax consequent

KKR expects that it will be engaged in a U.S. trade or business for U.S. federal income tax purposes, including by reason of in U.S. real property holding corporations, real estate assets and natural resource and oil and gas investments, in which case some portic income would be treated as effectively connected income with respect to non-U.S. holders, or ECI. To the extent KKR's income ECI, non-U.S. KKR common unitholders generally would be subject to withholding tax on their allocable share of such income required to file a U.S. federal income tax return for such year reporting their allocable share of income effectively connected with a business and any other income treated as ECI, and would be subject to U.S. federal income tax at regular U.S. tax rates on any such it and local income taxes and filings may also apply in that event). Non-U.S. KKR common unitholders that are corporations may also a 30% branch profits tax (potentially reduced under an applicable treaty) on their actual or deemed distributions of such income distributions to non-U.S. KKR common unitholders that are attributable to profits on the sale of a U.S. real property interest may also 30% withholding tax. Also, non-U.S. KKR common unitholders may be subject to 30% withholding on allocations of KKR's income under the Code, unless an exemption from or a reduced rate of such applies (under an applicable treaty of the Code) and certain tax status information

Tax-exempt entities face unique tax issues from owning KKR common units that may result in adverse tax consequent

Generally, a tax-exempt partner of a partnership would be treated as earning unrelated business taxable income, or UBTI, if the regularly engages in a trade or business that is unrelated to the exempt function of the tax-exempt partner, if the partnership derives debt-financed property or if the partner interest itself is debt-financed. As a result of KKR's ownership of real assets and incurrence of indebtedness KKR will derive income that constitutes UBTI. Consequently, a holder of KKR common units that is a tax-exempt entire an individual retirement account, or IRA, or a 401(k) plan participant) will likely be subject to unrelated business income tax to the eallocable share of KKR's income consists of UBTI. In addition, a tax-exempt investor may be subject to unrelated business income of their KKR co

KKR cannot match transferors and transferees of KKR common units, and KKR will therefore adopt certain income tax conventions that may not conform with all aspects of applicable tax requirements. The IRS may challenge this treatment, adversely affect the value of KKR contains the conventions.

Because KKR cannot match transferors and transferees of KKR common units, KKR has adopted depreciation, amortization a accounting positions that may not conform with all aspects of existing Treasury regulations. A successful IRS challenge to those po adversely affect the amount of tax benefits available to KKR common unitholders. It also could affect the timing of these tax be amount of gain on the sale of KKR common units and could have a negative impact on the value of KKR common units or result in a adjustments to KKR common unitholders.

In addition, KKR's taxable income and losses are determined and apportioned among KKR common unitholders using convergence regards as consistent with applicable law. As a result, if you transfer your KKR common units, you may be allocated income, g deduction realized by KKR after the date of transfer. Similarly, a transferee may be allocated income, gain, loss and deduction realized prior to the date of the transferee's acquisition of KKR common units, you may be allocated income, gain, loss and deduction realized by KKR after the date of transfer.

Table

transferee may also bear the cost of withholding tax imposed with respect to income allocated to a transferor through a reductio

Holders of KKR common units may be subject to state, local and foreign taxes and return filing requirements as a result of

In addition to U.S. federal income taxes, holders of KKR common units may be subject to other taxes, including state, local taxes, unincorporated business taxes and estate, inheritance or intangible taxes that are imposed by the various jurisdictions in which business or owns property now or in the future, even if the holders of KKR common units do not reside in any of those jurisdictions. KKR common units may be required to file state and local income tax returns and pay state and local income taxes in some of jurisdictions in the U.S. and abroad. Further, holders of KKR common units may be subject to penalties for failure to complete requirements. It is the responsibility of each KKR common unitholder to file all U.S. federal, state, local and foreign tax returns required of successive to the subject to penalties for failure to complete the responsibility of each KKR common unitholder to file all U.S. federal, state, local and foreign tax returns are unitarity.

Certain U.S. holders of KKR common units are subject to additional tax on "net investme

U.S. holders that are individuals, estates or trusts are subject to a Medicare tax of 3.8% on "net investment income" (or undis investment income," in the case of estates and trusts) for each taxable year, with such tax applying to the lesser of such income or such person's adjusted gross income (with certain adjustments) over a specified amount. Net investment income includes net interest, dividends, annuities, royalties and rents and net gain attributable to the disposition of investment property. It is anticipation income and gain attributable to an investment in KKR common units will be included in a U.S. holder's "net investment income" such person's adjusted gross income (with certain adjustments) over a specified amount. Net investment income includes net interest, dividends, annuities, royalties and rents and net gain attributable to the disposition of investment property. It is anticipation income and gain attributable to an investment in KKR common units will be included in a U.S. holder's "net investment income" such person's adjusted gross income (with certain adjustments) over a specified amount. Net investment income includes net interest.

KKR may not be able to furnish to each KKR common unitholder specific tax information within 90 days after the close of each year, which means that holders of KKR common units who are U.S. taxpayers should anticipate the need to file annually a reextension of the due date of their incom

As a publicly traded partnership, KKR's operating results, including distributions of income, dividends, gains, losses or decadjustments to carrying basis, will be reported on Schedule K-1 and distributed to each KKR common unitholder annually. It may retain 90 days after the end of KKR's fiscal year to obtain the requisite information from all lower-tier entities so that Schedule prepared for the KKR common unitholders. For this reason, holders of KKR common units who are U.S. taxpayers should anticipat file annually with the IRS (and certain states) a request for an extension past April 15 or the otherwise-applicable due date of their return for the

Risks Related to KKR's Organization

Potential conflicts of interest may arise among KKR's managing partner, its affiliates and KKR. KKR's managing partner and have limited fiduciary duties to KKR and the holders of KKR Group Partnership units, which may permit them to favor their own KKR's detriment and that of the holders of KKR Group Partnership.

KKR's managing partner, which is its general partner, will manage the business and affairs of KKR's business, and will be go board of directors that is co-chaired by KKR's founders, who also serve as KKR's Co-Chief Executive Officers. Conflicts of intercamong KKR's managing partner and its affiliates, on the one hand, and KKR and holders of KKR common units, on the other hand of these conflicts, KKR's managing partner may favor its own

Table

the interests of its affiliates over KKR and holders of KKR common units. These conflicts include, among others, the

KKR's managing partner indirectly through its holding of controlling entities determines the amount and timing Group Partnership's investments and dispositions, indebtedness, issuances of additional partner interests, tax lamounts of reserves, each of which can affect the amount of cash that is available for distribution to holders of

Partne

KKR's managing partner is allowed to take into account the interests of parties other than KKR in resolving interest, which has the effect of limiting its duties, including fiduciary duties, to KKR. For example, KKR's as serve as the general partners of KKR's funds have fiduciary and contractual obligations to investors in KKR's funds obligations may cause such affiliates to regularly take actions that might adversely affect KKR's near-teroperations or cash flow. KKR's managing partner will have no obligation to intervene in, or to notify KKR of,

by suc

Because KKR's principals indirectly hold their KKR Group Partnership units through entities that are not subject income taxation and KKR holds some of its KKR Group Partnership units through one or more wholly-owned that are taxable as a corporation, conflicts may arise between KKR's principals and KKR relating to the structuring of investments, declaring distributions and of

KKR's managing partner, including its directors and officers, has limited its and their liability and reduced or eand their duties, including fiduciary duties, under KKR's partnership agreement to the fullest extent permitted by also restricting the remedies available to holders of KKR common units for actions that, without these limits constitute breaches of duty, including fiduciary duties. In addition, KKR has agreed to indemnify KKR's mana including its directors and officers, and KKR's managing partner's affiliates to the fullest extent permitted by with respect to conduct involving bad faith, fraud or willful

KKR's partnership agreement does not restrict KKR's managing partner from paying KKR or its affiliates for rendered, or from entering into additional contractual arrangements with any of these entities on KKR's behalf, s terms of any such additional contractual arrangements are fair and reasonable to KKR as determined under it agreement. The conflicts committee will be responsible for, among other things, enforcing KKR's rights and tho of KKR common units under certain agreements, against KKR Holdings and certain of its subsidiaries and general partner or limited partner of KKR Holdings, or a person who holds a partnership or equity interest in t

KKR's managing partner determines how much debt KKR may incur and that decision may adversely affecting ratings K

KKR's managing partner determines which costs incurred by it and its affiliates are reimbursal

Other than as set forth in certain confidentiality and restrictive covenant agreements, which in certain cases agreements between KKR's principals and KKR Holdings and which may not be enforceable by KKR or other modified or amended, affiliates of KKR's managing partner and existing and former personnel employ managing partner are not prohibited from engaging in other businesses or activities, including those that migh competition

Table

KKR's managing partner controls the enforcement of obligations owed to the KKR Group Partnerships by

KKR's managing partner or KKR's managing partner's conflicts committee decides whether to retain sepa accountants or others to perform service

Certain actions by KKR's managing partner's board of directors require the approval of the Class A shares of KKR's managing of which are held by KKR's

All of KKR's managing partner's outstanding Class A shares are held by KKR's principals who also hold interests in KKR partner entitling them to vote for the election of its directors. Although the affirmative vote of a majority of the directors of KKR partner is required for any action to be taken by KKR's managing partner's board of directors, certain specified actions approve managing partner's board of directors will also require the approval of a majority of the Class A shares of KKR's managing partner's actions consist of the constant of the cons

the entry into a debt financing arrangement by KKR in an amount in excess of 10% of its existing long-term (other than the entry into certain intercompany debt financing arrangement)

the issuance by KKR or its subsidiaries of any securities that would (i) represent, after such issuance, or upor exchange or exercise, as the case may be, at least 5% on a fully diluted, as converted, exchanged or exercised class of KKR's or its subsidiaries' equity securities or (ii) have designations, preferences, rights, priorities or po more favorable than those of KKR Group Partners.

the adoption by KKR of a shareholder

the amendment of KKR's partnership agreement or the limited partnership agreements of the KKR Group F

the exchange or disposition of all or substantially all of KKR's assets or the assets of any KKR Group

the merger, sale or other combination of KKR or any KKR Group Partnership with or into any o

the transfer, mortgage, pledge, hypothecation or grant of a security interest in all or substantially all of the assets

the appointment or removal of a Chief Executive Officer or a Co-Chief Executive Officer of KKR's managi

the termination of the employment of any of KKR's officers or the officers of any of KKR's subsidiaries or the termination of the employment of any of KKR's officers or the officers of any of KKR's subsidiaries, in each case, wi

the liquidation or dissolution of KKR, KKR's managing partner or any KKR Group Part

the withdrawal, removal or substitution of KKR's managing partner as its general partner or any person as the ge of a KKR Group Partnership, or the transfer of beneficial ownership of all or any part of a general partner interes

Group F

a KKR Group Partnership to any person other than one of its wholly owned

In addition, holders representing a majority of the Class A shares of KKR's managing partner have the authority to unilate KKR's managing partner's directors and also have the ability to appoint the officers of KKR's managing partner. Henry Kravis Roberts colle

Table

Class A shares representing a majority of the total voting power of the outstanding Class A shares. While neither of them acting able to control the voting of the Class A shares, they will be able to control the voting of such shares if they

Holders of KKR common units do not elect KKR's managing partner or vote on KKR's managing partner's directors and have li to influence decisions regarding KKI

Holders of KKR common units do not elect KKR's managing partner or its board of directors and, unlike the holders of comm corporation, have only limited voting rights on matters affecting KKR's business and therefore limited ability to influence decisio KKR's business. Furthermore, if holders of KKR common units are dissatisfied with the performance of KKR's managing partner, ability to remove KKR's managing partner, with or w

The control of KKR's managing partner may be transferred to a third party without KK

KKR's managing partner may transfer its general partner interest to a third party in a merger or consolidation or in a transubstantially all of its assets without KKR's consent or the consent of holders of KKR common units. Furthermore, the member managing partner may sell or transfer all or part of their limited liability company interests in KKR's managing partner without KKI subject to certain restrictions. A new general partner may not be willing or able to form new funds and could form funds that have objectives and governing terms that differ materially from those of KKR's current funds. A new owner could also have a different philosophy, employ investment professionals who are less experienced, be unsuccessful in identifying investment opportunities or record that is not as successful as KKR's track record. If any of the foregoing were to occur, KKR could experience difficulty in investments, and the value of KKR's existing investments, business, results of operations and financial condition could mate

KKR intends to pay periodic distributions to the holders of KKR common units, but its ability to do so may be limited by its holding structure and contractual

KKR intends to pay cash distributions on a quarterly basis. KKR is a holding company and has no material assets other the Group Partnership units that KKR holds through wholly-owned subsidiaries and has no independent means of generating income. A KKR intends to cause the KKR Group Partnerships to make distributions on the KKR Group Partnership units, including Partnership units that KKR directly or indirectly holds, in order to provide KKR with sufficient amounts to fund distributions KKR If the KKR Group Partnerships make such distributions, other holders of KKR Group Partnership units, including KKR Holders of KKR Group Partnership units, including KKR Group Partnerships make such distributions pro rata based on their KKR Group Partnerships.

The declaration and payment of any future distributions will be at the sole discretion of KKR's managing partner, which may check distribution policy at any time. KKR's managing partner will take into account general economic and business conditions, KKR's stand prospects, business and investment opportunities, financial condition and operating results, compensation expense, we requirements and anticipated cash needs, debt and contractual restrictions and obligations (including payment obligations pursual receivable agreement), legal, tax and regulatory restrictions, restrictions or other implications on the payment of distributions by holders of KKR Group Partnership units or by KKR's subsidiaries to KKR and such other factors as KKR's managing partner may defended the Delaware Limited Partnership Act, KKR may not make a distribution to a partner if after the distribution all of KKR's liab than liabilities to partners on account of their partnership interests and liabilities for which the recourse of creditors is limited to specific of the partnership, would exceed the fair value of KKR's assets. If KKR were to make such an impermissible distribution, any line who received a distribution all of the received a distribution and operating partners on account of their partnership interests.

Table

knew at the time of the distribution that the distribution was in violation of the Delaware Limited Partnership Act would be liable to amount of the distribution for three years. Furthermore, by paying cash distributions rather than investing that cash in KKR's busin risks slowing the pace of its growth, or not having a sufficient amount of cash to fund its operations, new investments or unantici expenditures, should the

KKR's ability to characterize such distributions as capital gains or qualified dividend income may be limited, and holders of K units should expect that some or all of such distributions may be regarded as ordin

If KKR were deemed to be an "investment company" subject to regulation under the Investment Company Act, applicable restrimake it impractical for KKR to continue its business as contemplated and could have a material adverse effect on KKI

A person will generally be deemed to be an "investment company" for purposes of the Investment Con

it is or holds itself out as being engaged primarily, or proposes to engage primarily, in the business of investing or trading in s

absent an applicable exemption, it is engaged or proposes to engage in the business of investing, reinvest holding, or trading in securities, and owns or proposes to acquire "investment securities" (within the material Investment Company Act) having a value exceeding 40% of the value of KKR's total assets (exclusive of U.S. securities and cash items) on an unconsol

KKR believes that it is engaged primarily in the business of providing investment management services and not in the business reinvesting or trading in securities. KKR regards itself as an investment management firm and does not propose to engage pribusiness of investing, reinvesting or trading in securities. Accordingly, KKR does not believe that it is an "orthodox" investment defined in Section 3(a)(1)(A) of the Investment Company Act and described in the first bullet

With regard to the provision described in the second bullet point above, KKR has no material assets other than its equit subsidiaries, which in turn have no material assets other than equity interests, directly or indirectly, in the KKR Group Partnersh these interests, KKR indirectly is the sole general partner of each of the KKR Group Partnerships and indirectly is vested with all and control over the KKR Group Partnerships. KKR does not believe its equity interests in its subsidiaries are investment securities believes that the capital interests of the general partners of its funds in their respective funds are neither securities nor investment Accordingly, based on KKR's determination, less than 40% of KKR's total assets (exclusive of U.S. government securities and cash unconsolidated basis are comprised of assets that could be considered investment securities. In this regard, as a result of the KPE KKR succeeded to a significant number of investment securities previously held by KKR & Co. (Guernsey) L.P. and now held Group Partnerships. In addition, KKR expects to make investments in other investment securities from time to time. KKR m holdings regularly to confirm its continued compliance with the 40% test described in the second bullet point above. The need to this 40% test may cause KKR to restrict its business and subsidiaries with respect to the assets in which KKR can invest and/or securities KKR may issue, sell investment securities, including on unfavorable terms, acquire assets or businesses that could change KKR's business or potentially take other actions which may be viewed as adverse by the holders of KKR common units, in ordering with exceptions provided by, and rules and regulations promulgated under, the Investment Conformity with exceptions provided by, and rules and regulations promulgated under, the Investment Conformity with exceptions provided by, and rules and regulations promulgated under, the Investment Conformity with exceptions provided by, and rules and regulations promulgated und

Table

Following the consummation of the merger, KFN will become an indirect subsidiary of KKR. KFN has conducted, and t merger KKR intends to continue to conduct, KFN's operations so that it does not meet the definition of an "investment compar Investment Company Act. KFN believes that it is not engaged, and does not propose to engage, primarily in the business of investing or trading in securities and as such does not believe that it is an "orthodox" investment company as defined in Section 3(a) Investment Company Act and described in the first bullet point above. KFN also monitors its operations such that it does not own or acquire "investment securities" having a value exceeding 40% of the value of KFN's total assets (exclusive of United States securities and cash items) on an unconsolidated basis as described in the second bullet point above. KFN is organized as a holding of conducts its operations primarily through majority owned subsidiaries, each of which are either outside of the definition of a company under Sections 3(a)(1)(A) and 3(a)(1)(C), or excepted from the definition of an investment company under the Investment Act in accordance with Rule 3a-7, Section 3(c)(1) or (7), Section 3(c)(5), or Section 3(c)(9). For purposes of the 40% test describe Investment Company Act excludes a majority owned subsidiary from the term "investment securities" unless the subsidiary is a company or relies on the exceptions from the definition of an investment company provided by Section 3(c)(1) or Section 3 Investment Company Act. Any guidance or action from the SEC or its staff, in particular changes that the SEC may ultimately proportion regarding the manner in which Rule 3a-7 applies to entities or new or modified interpretive positions under Section 3(c)(5)(C), material adverse effect on KFN's investment company status, and as a result make it more difficult for KKR to comply with its ow company status under the Investment C

The Investment Company Act and the rules thereunder contain detailed parameters for the organization and operation of companies. Among other things, the Investment Company Act and the rules thereunder limit or prohibit transactions with affilial limitations on the issuance of debt and equity securities, generally prohibit the issuance of options and impose certain governance of KKR intends to conduct its operations so that it will not be deemed to be an investment company under the Investment Company Act, imposed by the Investment Company Act, including limitations on KKR's capital structure, ability to transact business with affiliate KKR) and ability to compensate key employees, would make it impractical for KKR to continue its business as currently conducted agreements and arrangements between and among KKR, the KKR Group Partnerships and KKR Holdings, or any combination materially adversely affect KKR's business, financial condition and results of operations. In addition, KKR may be required to limit of investments that it makes as a principal, potentially divest assets acquired through KFN or in the KPE Transaction or otherwise business in a manner that does not subject it to the registration and other requirements of the Investment Company Act, including the requirements of the Investment Company Act, including the subject it to the registration and other requirements of the Investment Company Act, including the investment Company Act, including the investment Company Act, including the investment Company and investment Company Act, including the investment

KKR is a Delaware limited partnership, and there are certain provisions in its partnership agreement regarding exci indemnification of its officers and directors that differ from the Delaware General Corporation Law in a manner that may be le of the interests of holders of KKR co

KKR's partnership agreement provides that to the fullest extent permitted by applicable law, KKR's directors or officers will no KKR. However, under the Delaware General Corporation Law, which is referred to in this proxy statement/prospectus as the assuming inclusion of a provision in the certificate of incorporation exculpating directors to the fullest extent permitted under director would be liable to KKR for (i) breach of the duty of loyalty to KKR or its shareholders, (ii) intentional misconduct, knowing of the law or acts or omissions not in good faith, (iii) improper redemption of shares or declaration of dividend, or (iv) a transaction the director derived an improper personal benefit. In addition, KKR's

Table

agreement provides that KKR indemnifies its directors and officers for acts or omissions to the fullest extent provided by law. How the DGCL, a corporation can only indemnify directors and officers for acts or omissions if the director or officer acted in good faith, he reasonably believed to be in the best interests of the corporation, and, in criminal action, if the officer or director had no reasonable believe his conduct was unlawful. Accordingly, KKR's partnership agreement may be less protective of the interests of holders of K units, when compared to the DGCL, insofar as it relates to the exculpation and indemnification of KKR's officers and directors. See limited partnership agreement contains provisions that reduce or eliminate duties (including fiduciary duties) of KKR's limited part remedies available for holders of KKR common units for actions that might otherwise constitute a breach of duty. It will be difficult of KKR common units to successfully challenge a resolution of a conflict of interest by KKR's managing partner or by its conflicts and "Description of KKR's Limited Partnership Agreement" beginning on page 201 of this proxy statemen

Risks related to KKR's business and other risks related to the ownership of KKR common units and risks related organizational structure are contained in the documents incorporated by reference in this proxy statement/prospectus. See "Can Find More Information" beginning on page 234 of this proxy statement.

CAUTIONARY STATEMENT REGARDING FORWARD-LOOKING STATEMENTS

This proxy statement/prospectus and the documents incorporated herein by reference contain forward-looking statem forward-looking statements are identified as any statement that does not relate strictly to historical or current facts. They use w "anticipate," "believe," "intend," "plan," "projection," "forecast," "strategy," "position," "continue," "estimate," "expect," "may," or of those terms or other variations of them or comparable terminology. In particular, statements, express or implied, concerning fu conditions or events, future operating results, the ability to generate sales, income or cash flow, to realize cost savings or of associated with the merger, to service debt or to make distributions are forward-looking statements. Forward-looking statements are distributions may differ materially from those expressed in these forward-looking statements. Many of the factors that will determine a are beyond the ability of KKR or KFN to control or predict. Specific factors which could cause actual results to differ from forward-looking statements.

the ability to complete the merger, on a timely ba

failure to obtain, delays in obtaining or adverse conditions contained in, any required regulatory approvals or

the potential impact of the announcement or consummation of the merger on relationships, including with consultants, investors and other business re

KKR's ability to successfully integrate KFN's assets and to realize synergies from

KKR's ability to acquire new businesses and assets and integrate those operations into its existing operations, particles acquisitions in a relatively short period of time, as well as the ability to expand the same of the

the ability to successfully identify and close acquisitions and make cost-saving changes in

changes in laws or regulations, third-party relations and approvals, and decisions of courts, regulators and g bodies that may adversely affect the business or ability to compete of Kl

the timing and success of business developr

changes in accounting pronouncements that impact the measurement of results of operations, the timing of measurements are to be made and recorded, and the disclosures surrounding these disclosures are to be made and recorded, and the disclosures surrounding these disclosures are to be made and recorded, and the disclosures surrounding these disclosures are to be made and recorded.

changes in tax law, particularly as it relates to partnerships or other "pass-throu

changes in interest rates and ma

changes in prepa

changes in market conditions, particularly in the global fixed income, credit and equ

the ability to obtain insurance coverage without significant levels of self-reter

acts of nature, sabotage, terrorism (including cyber attacks) or other similar acts or accidents causing damage KKR's or KFN's insurance coverage.

possible changes in cr

capital and credit markets conditions, inflation and i

Table

national, international, regional and local economic, competitive and regulatory conditions and de-

KKR's ability to achieve cost savings and reve

unfavorable results of litigation and the fruition of contingencies referred to in the notes to the financial statement in the reports incorporated by reference into this proxy statement/ proxy

changes in law, particularly changes to Rule 3a-7 or other rules or SEC guidance concerning the determination w is an "investment company" under the Investment Co

Forward-looking statements are based on the expectations and beliefs of the respective managements of KKR and KI information currently available, concerning future events affecting KKR and KFN. Although KKR and KFN believe that these forv statements are based on reasonable assumptions, they are subject to uncertainties and factors related to KKR's and KFN's operations environments, all of which are difficult to predict and many of which are beyond KKR's and KFN's control. Any or all of the forward statements in this proxy statement/prospectus may turn out to be wrong. They can be affected by inaccurate assumptions or unknown risks and uncertainties. The foregoing list of factors should not be construed to be exhaustive. Many factors mentioned statement/prospectus, including the risks outlined under the caption "Risk Factors" in this proxy statement/prospectus and uncaptioned sections contained in KKR's and KFN's Exchange Act reports incorporated herein by reference, will be important in future results, and actual future results may vary materially. There is no assurance that the actions, events or results of the forwattements will occur, or, if any of them do, when they will occur or what effect they will have on KKR's or KFN's results of financial condition, cash flows or distributions. In view of these uncertainties, KKR and KFN caution that investors should not reliance on any forward-looking statements. Further, any forward-looking statement speaks only as of the date on which it is made as required by law, KKR and KFN undertake no obligation to update any forward-looking statement to reflect events or circumstandate on which it is made or to reflect the occurrence of anticipated or unanticipated events or circumstandate on which it is made or to reflect the occurrence of anticipated or unanticipated events or circumstandate on which it is made

<u>Table</u>

THE PARTIES

KKR

Led by Henry Kravis and George Roberts, KKR, together with its subsidiaries, is a leading global investment firm with \$90 assets under management as of December 31, 2013, and a 37-year history of leadership, innovation and investment excellence. KK offers a broad range of investment management services to its fund investors and provides capital markets services to the firm, companies and other third parties. KKR conducts its business with offices throughout the world, providing it with a global platform transactions, raising capital and carrying out capital markets.

As a global investment firm, KKR earns management, monitoring, transaction and incentive fees for providing investment r monitoring and other services to its funds, vehicles, managed accounts, specialty finance company and portfolio companies, and KR transaction-specific income from capital markets transactions. KKR earns additional investment income from investing its own capitat of its fund investors, from other principal investments and from the carried interest KKR receives from its funds and certain other

KKR is a Delaware limited partnership, and it conducts its business through its subsidiaries. KKR common units are publicly to NYSE under the syn

The principal executive offices of KKR are located at 9 West 57th Street, Suite 4200, New York, New York 10019, and number is (212) 750-8300. Additional information about KKR and its subsidiaries is included in documents incorporated by refere proxy statement/prospectus. See "Where You Can Find More Information" on page 234 of this proxy statement

KKR Fund H

Fund Holdings is an exempted limited partnership formed under the laws of the Cayman Islands and is a subsidiary of KKR Holdings GP Limited and KKR Group Holdings L.P., wholly-owned subsidiaries of KKR, are the general partners of Fund Hold business, property and affairs of Fund Holdings is managed by its general partners. Fund Holdings was formed to hold intere businesses and assets that will generate qualifying income for purposes of the qualifying income exception to the publicly tradec rules under U.S. federal tax laws. The principal executive offices of Fund Holdings are located c/o KKR, 9 West 57th Street, Suit York, New York 10019, and its telephone number is (21)

Copal Merg

Merger Sub is a Delaware limited liability company and is a direct, wholly-owned subsidiary of Fund Holdings. Merger Sub we Fund Holdings solely in contemplation of the Merger, has not conducted any business and has no assets, liabilities or other oblig nature other than as set forth in the merger agreement. The principal executive offices of Merger Sub are located at 9 Wes Suite 4200, New York, New York 10019, and its telephone number is (21)

KKR Financial Ho

KFN is a specialty finance company with expertise in a range of asset classes. Its core business strategy is to leverage the resources of its manager with the objective of generating both current income and capital appreciation by deploying capital to which include bank loans and high yield securities, natural resources, special situations, mezzanine, commercial real estate and present the company of the com

<u>Table</u>

The majority of KFN's holdings consist of corporate loans and high yield debt securities held in collateralized loan obligation in this proxy statement/prospectus as CLOs, transactions that are structured as on-balance sheet securitizations and are used financing for its investments in corporate debt. The corporate loans that KFN holds are typically purchased via assignment or partici primary or secondary market. The senior secured debt issued by the CLO transactions is primarily owned by unaffiliated third party i KFN owns the majority of the subordinated notes in the CLO transactions. KFN's other holdings primarily consist of private equity joint ventures and partnerships and working and royalty interests in oil and ga

KFN executes its core business strategy through its majority-owned subsidiaries, includi

KFN is a Delaware limited liability company and was organized on January 17, 2007. It is the successor to KKR Finar Maryland corporation. KFN's common shares are publicly traded on the NYSE under the symbol "KFN." The principal executive of are located at 555 California Street, 50th Floor, San Francisco, California 94104, and its telephone number is (41)

Table

THE KFN SPECIAL MEETING

KFN is providing this proxy statement/prospectus to its common shareholders in connection with the solicitation of proxies to the special meeting of common shareholders that KFN has called for the purpose of holding a vote upon a proposal to adopt agreement with KKR, Fund Holdings and Merger Sub and at any adjournment or postponement thereof. This proxy statement constitutes a prospectus for KKR in connection with the registration by KKR of its common units to be exchanged in connect merger. This proxy statement/prospectus is first being mailed to KFN's common shareholders on or about March 24, 2014 and promon shareholders with the information they need to know to be able to vote or instruct their vote to be cast at the special meeting common shareholders.

Date, Tin

The special meeting of KFN common shareholders will be held at The Bently Reserve, Gateway Room, 301 Battery Street, Sa California 94111, on Wednesday, April 30, 2014 at 9:00 a.m.

At the special meeting, KFN common shareholders will be asked to vote solely on the following

Proposal 1: to adopt the merger agr

Proposal 2: to approve the adjournment of the KFN special meeting, if necessary to solicit additional proxies if sufficient votes to adopt the merger agreement at the time of the spe

Recommendation of the KFN Board

Based on the unanimous recommendation of the KFN transaction committee, the board of directors of KFN recommends to shareholders of

Proposal 1: "FOR" adoption of the merger agr

Proposal 2: "FOR" any adjournment of the KFN special meeting, if necessary to solicit additional proxies if sufficient votes to adopt the merger agreement at the time of the spe

The KFN board of directors, based upon the unanimous recommendation of the KFN transaction committee, (i) determine the merger agreement and the merger are fair to and in the best interests of KFN and its common shareholders, (ii) a merger and the merger agreement and (iii) resolved to recommend adoption of the merger agreement to the KFN common slee "Special Factors" Recommendation of the KFN Board of Directors and Reasons for the Merger; Fairness of the Merger" beginning of this proxy statement.

In considering the recommendation of KFN's board of directors with respect to the merger agreement and the transactions of thereby, you should be aware that some of KFN's directors and executive officers may have interests that are different from, or in ad interests of KFN common shareholders more generally. See "Special Factors Interests of Directors and Executive Officers of KFN is beginning on page 234 of this proxy statemen

Record Date; Outstanding Shares; Shares Enti

The record date for the KFN special meeting is February 26, 2014. Only KFN common shareholders of record at the close of the record date will be entitled to rece

Table

and to vote at the special meeting or any adjournment or postponement of the special meeting. An admission ticket (or other p ownership) and some form of government issued photo identification (such as a valid driver's license or passport) will be required to the special meeting.

As of the close of business on the record date of February 26, 2014, there were 204,824,159 KFN common shares outstanding to vote at the special meeting. Each KFN common share is entitled

A complete list of KFN common shareholders entitled to vote at the KFN special meeting will be available for inspection at K located at 555 California Street, 50th Floor, San Francisco, CA 94104 during regular business hours for a period of no less than ter the special meeting and at the place of the KFN special meeting during

A quorum of common shareholders is required to adopt the merger agreement at the special meeting. At least a majority of the KFN common shares must be represented in person or by proxy at the special meeting in order to constitute a quorum. Any abstence counted in determining whether a quorum is present at the special meeting. Your broker will not be permitted to vote on the admerger agreement without instruction from you as the beneficial owner of the KFN con

Re

To adopt the merger agreement, holders of at least a majority of the outstanding KFN common shares entitled to vote thereon majority of the outstanding KFN common shares entitled to vote thereon held by common shareholders other than KKR and its affivor of adoption of the merger agreement. Because approval is based on the affirmative vote of at least a majority of the KFN common shares, a KFN common shareholder's failure to submit a proxy card or to vote in person at the special meeting or a from voting, or the failure of a KFN common shareholder who holds his or her shares in "street name" through a broker or other non voting instructions to such broker or other nominee, will have the same effect as a vote "AGAINST" adoption of the merger agree close of business on the record date for the special meeting, KKR and its affiliates beneficially owned and had the right to vote 4,6 common shares at the special meeting, which represents approximately 2.3 percent of the KFN common shares entitled to vote a

To approve the adjournment of the KFN special meeting, if necessary to solicit additional proxies if there are not sufficient very the merger agreement at the time of the special meeting, the affirmative vote of a majority of the outstanding KFN common share vote thereon present in person or represented by proxy at the special meeting is required. Abstentions will have the same eff "AGAINST" the proposal, and KFN common shares not in attendance at the special meeting will have no effect on the outcome of adjourn the special meeting provided that a quoru

Share Ownership of and Voting by KFN's Directors and Execut

At the close of business on the record date for the special meeting, KFN's directors and executive officers beneficially owned right to vote 1,662,070 KFN common shares at the special meeting, which represents approximately 0.8 percent of the KFN contential to vote at the special meeting. It is expected that KFN's directors and executive officers will vote their shares "FOR" the admerger agreement, although none of them has entered into any agreement requiring the

Table

Voting of Shares by Holder

If you are entitled to vote at the special meeting and hold your common shares in your own name, you can submit a proxy or vote the special meeting. However, KFN encourages you to submit a proxy before the special meeting even i attend the special meeting in order to ensure that your shares are voted. A proxy is a legal designation of another person to vote common shares on your behalf. If you hold common shares in your own name, you may submit a proxy for your common shares in your own name, you may submit a proxy for your common shares in your own name, you may submit a proxy for your common shares in your own name, you may submit a proxy for your common shares in your own name, you may submit a proxy for your common shares in your own name, you may submit a proxy for your common shares in your own name, you may submit a proxy for your common shares in your own name, you may submit a proxy for your common shares in your own name, you may submit a proxy for your common shares in your own name, you may submit a proxy for your common shares in your own name, you may submit a proxy for your common shares in your own name, you may submit a proxy for your common shares in your own name, you may submit a proxy for your common shares in your own name, you may submit a proxy for your common shares in your own name, you may submit a proxy for your common shares in your own name, you may submit a proxy for your common shares in your own name, you may submit a proxy for your common shares in your own name, you may submit a proxy for your common shares in your own name, you may submit a proxy for your common shares in your own name.

calling the toll-free number specified on the enclosed proxy card and following the instructions whe

accessing the Internet website specified on the enclosed proxy card and following the instructions provide

filling out, signing and dating the enclosed proxy card and mailing it in the prepaid envelope included with

When a common shareholder submits a proxy by telephone or through the Internet, his or her proxy is recorded immedencourages its common shareholders to submit their proxies using these methods whenever possible. If you submit a proxy by tele

Internet website, please do not return your proxy or

All common shares represented by each properly executed and valid proxy received before the special meeting will be voted in with the instructions given on the proxy. If a KFN common shareholder executes a proxy card without giving instructions, the K shares represented by that proxy card will be voted "FOR" approval of the proposal to adopt the merge

When a common shareholder submits a valid proxy, the proxy holder will be authorized to vote, in his or her discretion, upon business as may properly come before the meeting or any adjournment thereof provided the board of directors does not know at time before the solicitation, that such matters are to be presented at

Your vote is important. Accordingly, please submit your proxy by telephone, through the Internet or by mail, whether or no attend the special meeting in person. Proxies must be received by 11:59 p.m., Eastern Time, on Ap

Voting of Common Shares Held in S

If your common shares are held in an account at a broker or through another nominee, you must instruct the broker or other nomito vote your shares by following the instructions that the broker or other nominee provides to you with these proxy materials. Most the ability for shareholders to submit voting instructions by mail by completing a voting instruction card, by telephone and via

If you do not provide voting instructions to your broker or other nominee, your common shares will not be voted on any propo your broker or nominee does not have discretionary authority to vote. Under the current NYSE rules, a broker will not have do authority to vote your common shares at the special meeting. Accordingly, a failure to providing voting instructions to your browninee will have the same effect as a vote "AGAINST" adoption of the merger agreement. Your broker or other nominee we common shares held by it in "street name" with respect to these matters only if you provide instructions to it on how to vote. You share the directions your broker or other nomine the directions your broker or other nomine.

If you hold common shares through a broker or other nominee and wish to vote your common shares in person at the special must obtain a proxy from your broker or other nominee and present it to the inspector of election with your ballot when you vote

Table

Revocability of Proxies; Changing

You may revoke your proxy and/or change your vote at any time before your proxy is voted at the special meeting. If you as shareholder of record, you can

sending a written notice (stating that you revoke your proxy) to KFN at 555 California Street, 50th Floor, Sa California 94104, Attn: Corporate Secretary, that bears a date later than the date of the proxy you want to received prior to the spec

submitting a valid, later-dated proxy by mail, telephone or Internet that is received prior to the special

attending the special meeting and voting by ballot in person (your attendance at the special meeting will not, by any proxy that you have previous

If you hold your common shares through a broker or other nominee, you must follow the directions you receive from your broker or change or change

Solicitatio

This proxy statement/prospectus is furnished in connection with the solicitation of proxies by the KFN board of directors to be KFN special meeting. KFN will bear all costs and expenses in connection with the solicitation of proxies. KFN has engaged Inr Incorporated to assist in the solicitation of proxies for the special meeting and KFN estimates it will pay a fee of approximately \$22,00 services. KFN has also agreed to reimburse the proxy solicitor for, pay directly, or, where requested by the proxy solicitor in special advance sufficient funds to the proxy solicitor for payment of, expenses and disbursements incurred in connection with the proxy solicitor indemnify the proxy solicitor against certain losses, costs and expenses. In addition, KFN may reimburse brokerage firms and or representing beneficial owners of KFN common shares for their reasonable expenses in forwarding solicitation materials to such owners. Proxies may also be solicited by certain of KFN's directors, officers and employees by telephone, electronic mail, letter, father than the proxy solicitor and additional compensation will be present the proxy solicitor and proximately solicitation and the proxy solicitor and

Shareholders should not send share certificates with their proxies. A letter of transmittal and instructions for the surre common share certificates will be mailed to KFN common shareholders shortly after the completion o

No Otl

Under KFN's operating agreement, the business to be conducted at the special meeting will be limited to the purposes stated in KFN common shareholders provided with this proxy statement

.

Adjournments may be made for the purpose of, among other things, soliciting additional proxies. Any adjournment may be made to time by the chairman of the KFN board of directors. At any adjourned meeting, KFN may transact any business that it might have at the original meeting, provided that a quorum is present at such adjourned meeting. Proxies submitted by KFN common shareholds the special meeting will be used at any adjournment or postponement of the special meeting. References to the KFN special meeting statement/prospectus are to such special meeting as adjourned or

If you need assistance in completing your proxy card or have questions regarding the special meeting, please contact KFN's pro Innisfree M&A Incorporated, at (888) 750-5834 (banks and brokers call collect at (212

<u>Table</u>

THE MERGER AGREEMENT

The following describes the material provisions of the merger agreement, which is attached as Annex A to this proxy statement and incorporated by reference herein. The description in this section and elsewhere in this proxy statement/prospectus is qualified in by reference to the merger agreement. This summary does not purport to be complete and may not contain all of the information merger agreement that is important to you. KKR and KFN encourage you to read carefully the merger agreement in its entirety be any decisions regarding the merger as it is the legal document governing

The merger agreement and this summary of its terms have been included to provide you with information regarding the terms of agreement. Factual disclosures about KKR, KFN or any of their respective subsidiaries or affiliates contained in this proxy statement or their respective public reports filed with the SEC may supplement, update or modify the factual disclosures about KKR, respective subsidiaries or affiliates contained in the merger agreement and described in this summary. The representations, was covenants made in the merger agreement by KKR, Fund Holdings, Merger Sub and KFN were qualified and subject to important agreed to by KKR, Fund Holdings, Merger Sub and KFN in connection with negotiating the terms of the merger agreement. In paper was of the representations and warranties contained in the merger agreement and described in this summary, it is imported mind that the representations and warranties were negotiated with the principal purpose of allocating risk between the parties to agreement, rather than establishing matters as facts. The representations and warranties may also be subject to a contractual materiality different from those generally applicable to shareholders or holders of common units and reports and documents filed and in some cases were qualified by confidential disclosures that were made by each party to the other, which disclosures are not represented agreement or otherwise publicly disclosed. Moreover, information concerning the subject matter of the representations and may have changed since the date of the merger agreement and subsequent developments or new information qualifying a representation or any descriptions of those provisions should not be

Subject to the terms and conditions of the merger agreement and in accordance with Delaware law, the merger agreement pro merger of Merger Sub, an indirect subsidiary of KKR and a party to the merger agreement, with and into KFN. KFN will survive that an indirect subsidiary of KKR and the separate limited liability company existence of Merger Sub will cease. After the completion of the certificate of formation of KFN in effect immediately prior to the effective time will be the certificate of formation of the surviving and the limited liability company agreement of KFN in effect immediately effective time will be the limited liability company agreement of the surviving entity until amended in accordance with its terms are law. After the completion of the merger, the officers of KFN immediately prior to the effective time will be the officers of the surviving and the individuals designated by KKR prior to the closing of the merger will be the members of the board of directors of the surviving entity until amended in accordance with its terms are law.

Effective Ti

The effective time will be at such time that the parties file with the Secretary of State of the State of Delaware a certificate executed in accordance with the relevant provisions of the Delaware Limited Liability Company Act, or at such later date or time as by the parties to the merger agreement and specified in the certificate

<u>Table</u>

Unless the parties agree otherwise, the closing of the merger will occur at 10:00 a.m. (New York time), as soon as practicable event within three business days, following the satisfaction or waiver of the conditions to the merger provided in the merger agree than conditions that by their nature are to be satisfied at the closing of the merger, but subject to the satisfaction or waiver of those coat such other date or time as the parties to the merger agreement agree. For further discussion of the conditions to the merger, see "

Consummation of the Merger." The parties to the merger agreement have agreed to use their reasonable best efforts to coordinate as so that the closing of the merger will take place on the last business day of a calculation.

KKR and KFN currently expect to complete the merger in the first half of 2014, subject to receipt of required approvals of hol common shares and regulatory approvals and to the satisfaction or waiver of the other conditions to the transactions contemplated by agreement description.

Conditions to Consummation of

KKR and KFN may not complete the merger unless each of the following conditions is satisfie

the merger agreement must have been adopted by the affirmative vote of the holders of at least a majority of the KFN common shares entitled to vote thereon on the

the merger agreement must have been adopted by the affirmative vote of the holders of a majority of the outst common shares (other than those KFN common shares held by KKR or any affiliate of KKR) entitled to vote the

the waiting period applicable to the merger under the HSR Act, if any, must have been terminated or exconsents required under any other antitrust law must have been obtained or any applicable waiting period them have been terminated

there must be no outstanding judgment, injunction, order or decree of a competent U.S. federal or state g authority prohibiting or enjoining the completion of the merger or the other transactions contemplated b

the registration statement of which this proxy statement/prospectus forms a part must have been declared eff SEC and must not be subject to a stop order or similar restraining order by the

the KKR common units to be issued in the merger must have been approved for listing on the NYSE, subjective notices

The obligations of each of KKR, Fund Holdings and Merger Sub to effect the merger are subject to the satisfaction or value following additional following ad

(1) the representations and warranties of KFN in the merger agreement (other than those relating to the absence adverse effect) that are qualified by a "material adverse effect" qualification being true and correct both at and a of the merger agreement and at and as of the closing, (2) the representations and warranties of KFN in the merger (other than certain representations and warranties relating to the capitalization of KFN, brokerage and find takeover statutes) that are not qualified by a "material adverse effect" qualification being true and correct (we effect to any materiality qualifications therein) at and as of the date of the merger agreement and at and as of except for failures to be true and correct as would not, in the aggregate, constitute a material adverse effect with KFN, and (3) the representations and warranties.

Table

to the absence of a material adverse effect, the capitalization of KFN, brokerage and finder's fees and takeover s true and correct at and as of the date of the merger agreement and at and as of the closing, except, in the carepresentations and warranties relating to the capitalization of KFN, for any *de minimis* inaccuracies (in all carepresentations and warranties that are made as of a particular date or period must be true and correct (in the mar above) only as of such date

KFN having performed in all material respects the obligations and agreements and having complied in all material with the covenants to be performed and complied with by it under the merger agreement at or before

there not having occurred any events that, individually or in the aggregate, constitute a material adverse effect with KFN since the date of the merger

the receipt of a certificate signed by the chief executive officer or other senior officer of KFN certifying to preceding conditions have be

the receipt of a payoff letter reasonably acceptable to KKR with respect to the termination of KFN's \$150 m facility, dated as of November 30, 2012 (under which \$75.0 million of borrowings were outstanding as of D 2013), and all commitments or other extensions of credit thereunder and the satisfaction or discharge of all amou or outstanding thereunder and the satisfaction, release and discharge of all security interests, mortgages or lier and its subsidiaries' properties and assets securing such obligations, and the payoff letter requiring the administ under the credit agreement to deliver all instruments necessary or desirable to evidence or effect the for

receipt of a statement in accordance with Treasury Regulation Section 1.1445-11T(d)(2) certifying that 50% of value of the gross assets of KFN does not consist of U.S. real property interests, or that 90% or more of the value assets of the KFN does not consist of U.S. real property interests plus cash or cash

The obligations of KFN to effect the merger are subject to the satisfaction or waiver of the following additional

(1) the representations and warranties of KKR, Fund Holdings and Merger Sub (other than those relating to the material adverse effect) that are qualified by a "material adverse effect" qualification being true and correct both the date of the merger agreement and at and as of the closing, (2) the representations and warranties of KKR, Fu and Merger Sub in the merger agreement (other than certain representations and warranties relating to the cap KKR and Merger Sub and brokerage and finder's fees) that are not qualified by a "material adverse effect" being true and correct (without giving effect to any materiality qualifications therein) at and as of the date of agreement and at and as of the closing, except for failures to be true and correct as would not, in the aggregate material adverse effect with respect to KKR, (3) the representations and warranties relating to the capitalizate being true and correct in all material respects at and as of the date of the merger agreement and at and as of the closing (in all care) true and correct at and as of the date of the merger agreement and at and as of the closing (in all care) true and correct at and as of the date of the merger agreement and at orrect (in the mar above) only as of such date of such date of the merger agreement and at and correct (in the mar above) only as of such date

<u>Table</u>

each of KKR, Fund Holdings and Merger Sub having performed in all material respects the obligations and agrahaving complied in all material respects with the covenants to be performed and complied with by it under agreement at or before

there not having occurred any events that, individually or in the aggregate, constitute a material adverse effect with KKR since the date of the merger agree.

the receipt of a certificate signed by a chief executive officer or other senior officer of KKR certifying to preceding conditions have be

For purposes of the merger agreement, the term "material adverse effect" means, with respect to a party to the merger agreem circumstance, event, change, occurrence or effect that would have, or would reasonably be expected to have, a material adverse financial condition, business or results of operations of the party and its subsidiaries, taken as a whole; provided, however, that a mat effect with respect to any party will not include any fact, circumstance, event, change, occurrence or effect directly or indirectly aris attributable to: (1) any decrease in the market price of KKR common units, in the case of KKR, or KFN common shares, in the case in either case not any event underlying the decrease to the extent the event would otherwise constitute a material adverse effect of (2) conditions, events, or circumstances generally affecting the principal businesses or industries in which the party and its subsidia (3) changes in generally accepted accounting principles, referred to in this proxy statement/prospectus as GAAP, applicable law of standards, or in any interpretation of GAAP, applicable law or accounting standards; (4) changes in any analyst's recommendations, strength rating or any other recommendations or ratings as to KKR or KFN, as the case may be, or their respective subsidiaries (included) of itself, any failure to meet analyst projections); (5) the failure, in and of itself, of KKR or KFN, as the case may be, to meet any projected financial or operating performance target publicly announced or provided to the other party prior to the date of the merge as well as any change, in and of itself, by KKR or KFN, as the case may be, in any expected or projected financial or operating target as compared with any target publicly announced or provided to the other party prior to the date of the merger agreement (case, not any event underlying the failure or change to the extent the event would otherwise constitute a material adverse effect on (6) any changes or developments in United States, Europe, Asia or global economic, regulatory or political conditions in general (i outbreak or escalation of hostilities or acts of war or terrorism), or generally affecting the financial, debt, credit or securities n United States, Europe, Asia or elsewhere in the world, including changes in interest rates and commodities prices, or any nat provided, however that the exceptions in clauses (2), (3) and (6) above may be taken into account in determining whether a mat effect has occurred or is reasonably expected to occur to the extent (but only to the extent) the event has a materially disproportion the party and its subsidiaries relative to other participants in the principal businesses or industries in which the party and its subsidiaries

KFN Sharehold

KFN has agreed to hold a special meeting of KFN common shareholders as soon as is practicable after the date the registration shareholders as soon as is practicable after the date the registration of which this proxy statement/prospectus is a part) is declared effective by the SEC for the purpose of KFN common shareholders adoption of the merger agreement. The merger agreement requires KFN to submit the merger agreement to a shareholder vote (I board of directors of KFN no longer recommends adoption of the merger agreement and (2) regardless of the commencement, public disclosure or communication to KFN of any acquisition proposal (as described below). The board of directors of KFN, act unanimous recommendation of the transaction committee of the board of directors of KFN, has approved and declared advisable agreement and the

Table

contemplated thereby and recommended that holders of KFN common shares adopt the merge

For purposes of the merger agreement, the term "acquisition proposal" means any proposal or offer from any person other that subsidiaries with respect to (1) a merger, consolidation, acquisition, share exchange, business combination, reorganization, rec dissolution, liquidation or similar transaction involving KFN, (2) any purchase of an equity interest (including by means of a tender offer) representing an amount equal to or greater than a 15% voting or economic interest in KFN or (3) any purchase of assets, ownership interests representing an amount equal to or greater than 15% of the consolidated asset value (including only the portion of the equity interests of the subsidiaries of KFN that is actually owned directly or indirectly by KFN) or consolidated net reve (including only the portion of the revenues attributable to the equity interests of the subsidiaries of KFN that is actually owned indirectly by KFN); provided that the calculation of consolidated asset value and consolidated net revenues includes only the uninterests in variable interest entities that are held directly or indirectly by KFN so that the total asset value or revenues of the entitle interest entity is not included in the

KFN Prefe

The merger agreement provides that each KFN 7.375% Series A Preferred Share that is outstanding immediately prior to the remain outstanding as a preferred share of KFN, as the surviving entity in the merger, immediately following

No Solicitation by KFN of Alternativ

The merger agreement contains detailed provisions prohibiting KFN from seeking an alternative proposal to the merger. Uncondicitation" provisions, KFN has agreed that it and its subsidiaries will not, and it will use reasonable best efforts to cause its and its directors, officers, employees, agents, investment bankers, attorneys, accountants and other representatives not to, directly or indirect otherwise permitted by the merge

initiate or solicit or knowingly encourage any inquiries with respect to, or the making of an acquisition

engage in any negotiations concerning, or provide any confidential information or data to any person racquisition acquisition acquisition

approve or recommend, or propose publicly to approve or recommend, any acquisition

approve or recommend, or propose publicly to approve or recommend, or execute or enter into, any le agreement in principle, merger agreement, acquisition agreement, option agreement or other similar agreeme any acquisition

propose publicly or agree to do any of the foregoing relating to any acquisit

The merger agreement requires KFN and its subsidiaries to (1) cease and cause to be terminated any existing activities, descriptions with any person conducted prior to the date of the merger agreement with respect to an acquisition proposal and (2) each third party that executed a confidentiality agreement that relates to an acquisition proposal before the date of the merger agreement destroy all confidential information furnished to the third party by KFN or on its behalf before the date of the merger

Notwithstanding these restrictions, the merger agreement provides that, under specified circumstances at any time before KFN vote in favor of the adoption of

Table

agreement, if KFN receives a written unsolicited bona fide acquisition proposal after the date of the merger agreement that the board of KFN has determined in good faith, after consultation with its outside legal counsel and financial advisors (1) constitutes a super (as described below) or (2) could reasonably be expected to result in a superior proposal

furnish nonpublic information to a third party that makes an acquisition proposal, if, before furnishing the informations an executed confidentiality agreement with provisions no less restrictive to the third party with respect disclosure of nonpublic information than the confidentiality agreement in effect between KFN and

engage in discussions or negotiations with the third party with respect to the acquisit

For purposes of the merger agreement, the term "superior proposal" means a bona fide written acquisition proposal (except the in the definition of "acquisition proposal" to 15% are replaced by 50%) made after the date of the merger agreement by any personal (KKR or its subsidiaries on terms that the board of directors of KFN determines in good faith, after consultation with its outside legal financial advisors, and considering the factors that the board of directors of KFN considers to be appropriate (including the timing are of consummation of the proposal), are more favorable to KFN and the holders of KFN common shares than the transactions contemporate agreement, taking into account any change to the merger proposal.

KFN has also agreed in the merger agreement that it will promptly orally notify KKR of any request for information or a proposals or offers relating to an acquisition proposal indicating, in connection with the notice, the name of the person making inquiry, proposal or offer and the material terms and conditions of any proposals or offers, and that it will provide to KKR written inquiry, proposal or offer within 24 hours of the request or inquiry, proposal or offer and copies of any written or electronic corresponding any person making an acquisition proposal. KFN is required to keep KKR informed orally, as soon as reasonably practicable, of any acquisition proposal, including with respect to the status and terms of any proposal or offer and whether any proposal or withdrawn or rejected, and to provide to KKR written notice of any withdrawal or rejection and copies of any written proposals or information within 24 hours. KFN has also agreed to provide any information to KKR (not previously provided to KKR) that is another person pursuant to these provisions at substantially the same time it provides the information to the

The merger agreement provides that nothing in the merger agreement will prevent KFN, the board of directors of KFN and an thereof (including the transaction committee) from complying with its disclosure obligations under Rule 14d-9 and 14e-2 of the Exclored from issuing a "stop, look and listen" or similar communication or making any disclosure if the board of directors of KFN or an thereof (including the transaction committee) determines, in good faith, after consultation with KFN's outside legal counsel, that the so would be inconsistent with KFN's directors' fiduciary duties under applicable law of the State of Delaware; provided, however disclosure has the effect of a change of recommendation (as described below), KKR will have the right to terminate the merger agree certain circumstances, as described in "Termination of the Merge

Change in KFN Board Recor

The merger agreement provides that neither the board of directors of KFN nor any committee thereof (including the transaction will, directly or indirectly, effect a change of recommendation. For purposes of the merger agreement, the term "change of recommendation."

<u>Table</u>

to (1) withdraw, modify or qualify, or publicly propose to withdraw, modify or qualify, in a manner adverse to KKR, Fund Holding Sub, the recommendation of the board of directors of KFN that holders of KFN common shares adopt the merger agreement, (2) ap or recommend, or publicly propose to approve, adopt or recommend, any acquisition proposal, (3) in the event of the commencement offer or exchange offer for any outstanding shares of KFN's capital stock (including the KFN common shares), fail to include is statement related to the tender offer or exchange offer or the merger agreement (other than any "stop, look and listen" or similar common a statement that the recommendation of the board of directors of KFN that holders of KFN common shares adopt the merger agreement, or (4) recommend that holders of KFN common shares not adopt the merger

Notwithstanding the terms above or any other term of the merger agreement to the contrary, subject to the conditions described may, at any time prior to KFN common shareholders voting in favor of the adoption of the merger agreement, in response to the written unsolicited bona fide acquisition proposal after the date of the merger agreement that did not result from a brea non-solicitation obligations under the merger agreement, (1) terminate the merger agreement in order to enter into a definitive written with respect to the acquisition proposal or (2) effect a change of recommendation, if, prior to taking any of the foregreen terms of the described may, at any time prior to taking any of the foregreen terms are the contrary, subject to the conditions described may, at any time prior to KFN common shareholders voting in favor of the adoption of the merger agreement, in response to the merger agreement and the conditions are the conditions and the conditions are the conditions are the merger agreement and the conditions are the conditions are

the board of directors of KFN determines in good faith, after consultation with its outside legal counsel advisors, that (A) failure to terminate or change its recommendation would reasonably be expected to be inco KFN's directors' fiduciary duties under applicable law of the State of Delaware and (B) the acquisition proposal superi-

KFN provides to KKR written notice of its intention to terminate the merger agreement or changes its recomleast four business days before taking the

the board of directors of KFN continues to believe, after taking into account any modifications to the transactions contemplated by the merger agreement that are offered by KKR after its receipt of the notice reference of the preceding bullet point, that the acquisition proposal continues to constitute a super

Any material amendment to the financial terms or any other material amendment of the acquisition proposal will require that I to KKR a new written notice of its intention to terminate the merger agreement or effect a change of recommendation, and the board of KFN will be required to comply again with the foregoing requirements; provided, however, that the reference to four business deemed to be a reference to two business.

Further, notwithstanding the terms above or any other term of the merger agreement to the contrary, the board of directors of any time prior to KFN common shareholders voting in favor of the adoption of the merger agreement, effect a change of recommendation in response to the receipt of a written unsolicited bona fide acquisition proposal, which is subject to the provisions describe before effecting a change of recommendation.

the board of directors of KFN, after consultation with its outside legal counsel, determines in good faith that fail would be inconsistent with KFN's directors' fiduciary duties under applicable law of the State of

KFN provides to KKR written notice of its intention to do so at least four business days before taking the

the board of directors of KFN continues to believe, after taking into account any modifications to the transactions contemplated by the merger agreement that are offer

Table

after its receipt of the notice referred to in the preceding bullet point, that failing to do so would be reasonably inconsistent with KFN's directors' fiduciary duties under applicable law of the State of the sta

Merger Co

The merger agreement provides that, at the effective time, each KFN common share issued and outstanding immediately effective time (other than KFN common shares held by Fund Holdings or any subsidiary of Fund Holdings) will automatically be common the right to receive 0.51 KKR common units, together with cash in lieu of fractional KKR common units, if any. Any KFN common by Fund Holdings or any subsidiary of Fund Holdings will no longer be outstanding and will automatically be cancelled and retired exist, and no payment will be made with respect to them, at the effective time, each KFN common share issued and outstanding immediately effective time, each KFN common share issued and outstanding immediately effective time, each KFN common share issued and outstanding immediately effective time, each KFN common share issued and outstanding immediately effective time, each KFN common share issued and outstanding immediately effective time, each KFN common share issued and outstanding immediately effective time, each KFN common share issued and outstanding immediately effective time, each KFN common share issued and outstanding immediately effective time, each KFN common share issued and outstanding immediately effective time, each KFN common share issued and outstanding immediately effective time, each KFN common share issued and outstanding immediately effective time, each KFN common share issued and outstanding immediately effective time, each KFN common share issued and each time.

KKR will not issue any fractional KKR common units in the merger. Instead, each holder of KFN common shares that a pursuant to the merger agreement who otherwise would have received a fraction of a KKR common unit will be entitled to received agent appointed by KKR pursuant to the merger agreement, a cash payment in lieu of the fractional units representing proportionate interest in the proceeds from the sale by the exchange agent of the aggregate number of KKR common units to be del merger over the aggregate number of whole (non-fractional) KKR common units to be distributed to holders of KFN common

Treatment of Equ

Under the merger agreement, KFN equity-based awards will be treated at the effective tim

Options. As of the effective time, each option (whether vested or unvested) to purchase KFN common shares that is immediately prior to the effective time will be cancelled and converted into the right to receive an amount in cash equal to the excess value of the number of KKR common units that a holder of one KFN common unit would be entitled to in the merger (determined to the average closing price of a KKR common unit over the 10 trading day period ending on the trading day immediately preceding date) over the exercise price per KFN common share subject to the option. The option consideration will be paid in a lump su practicable after the effective time but in no event later than 10 business days following the effective time. If any option to purcommon shares has an exercise price per KFN common share that is greater than the cash value of the number of KKR common holder of one KFN common unit would be entitled to in the merger (determined by reference to the average closing price of a KKR cover the 10 trading day period ending on the trading day immediately preceding the closing date), the option will be cancelled with as of the effective time.

Restricted Shares. At the effective time, each restricted KFN common share (other than any restricted KFN common share manager) issued and outstanding under KFN's 2007 Share Incentive Plan for KKR Financial Holdings LLC, as amended from which is referred to in this proxy statement/prospectus as the 2007 Share Plan, will be converted into a number of restricted KKR of (having the same terms and conditions, including applicable vesting requirements, as applied to the restricted KFN share immediately effective time) that a KFN common share would be exchanged for in the merger (with fractional KKR common units to be agrounded to the nearest whole unit), and KKR will assume the 2007 Share Plan and any award agreement issued thereunder pursuant such restricted KFN share has been su

Table

order to provide for the foregoing. At the Effective Time, any restricted KFN common shares held by the manager will autocancelled and retired and will common shares held by the manager will auto-

Phantom Shares. Each phantom share credited to a participant's account under the KKR Financial Holdings LLC No Directors' Deferred Compensation and Share Award Plan, which is referred to in this proxy statement/prospectus as the Deferred C Plan, will, immediately upon the effective time, automatically be converted into the right to receive 0.51 validly issued KKR common fractional KKR common units to be aggregated and rounded to the nearest whole unit), but will otherwise remain outstanding the effective time and be distributed pursuant to the terms of the Deferred Compe

Adjustments to Prev

If, before the effective time, KKR or KFN declares a stock dividend or other distribution payable in KFN common shares or K units, as applicable, or securities convertible, exercisable or exchangeable into KFN common shares or KKR common units, as a effects a stock split, reclassification, combination, exchange of shares (other than any regular quarterly issuances of KKR common u KKR in accordance with its quarterly exchange process) or other similar change, the exchange ratio will be adjusted to ref appropriate effect or

KFN (or the surviving entity after the effective time), KKR, the exchange agent and any other applicable withholding agent wit to deduct and withhold from the consideration otherwise payable pursuant to the merger agreement such amounts as are required to and withheld with respect to the making of the payments under applicable tax law. To the extent that amounts are withheld and time to the applicable taxing authority, the withheld amounts will be treated as having been paid to the applicable person in respect deduction and withholding

Dividends and D

The merger agreement requires the parties to coordinate so that holders of KFN common shares do not receive in respect calendar quarter both (1) a quarterly distribution in respect of the holders' KFN common shares and (2) a quarterly distribution in r KKR common units issuable pursuant to the merger agreement. In addition, the merger agreement provides that no divide distributions declared or made after the effective time with respect to KKR common units having a record date after the effective paid to the holder of any unsurrendered certificate, and no cash payment in lieu of fractional KKR common units will be paid to any until the holder surrenders the certificate. Subject to applicable law of the State of Delaware, following surrender of any such coholder of the KKR common units issued in exchange will receive, without interest, (A) at the time of surrender, the amount of divide distributions with a record date after the effective time payable with respect to the KKR common units and not paid, less the arwithholding taxes which may be required, and (B) at the appropriate payment date subsequent to surrender, the amount of divided distributions with (i) a record date after the effective time but prior to surrender and (ii) a payment date subsequent to surrender that with respect to the KKR common units, less the amount of any withholding taxes which may

<u>Table</u>

Regulat

See "Special Factors Regulatory Approvals and Clearances Required for the Merger" beginning on page 100 statement/prospectus for a description of the material regulatory requirements for the completion of

Each of the parties to the merger agreement has agreed to use its reasonable best efforts to resolve any objections that a g authority may assert with respect to the transactions contemplated by the merger agreement under any antitrust law. Notwith foregoing, KKR, KFN and their respective subsidiaries are not required to agree, and are not permitted to agree without the other written consent, to take any action or accept any condition, restriction, obligation or requirement with respect to KKR, KFN or respective subsidiaries or affiliates or their and their respective subsidiaries' or affiliates' assets if it (1) would reasonably be expect KKR, KFN or their respective subsidiaries or affiliates to sell, license, transfer, assign, lease, dispose of or hold separate any material assets or (2) would reasonably be expected to result in any material limitations on KKR or KFN or their respective subsidiaries or own, retain, conduct or operate all or a material portion of their respective business.

Termination of the Merger

KKR and KFN may terminate the merger agreement at any time prior to the closing, whether before or after holders of KFN conhave adopted the merger agreement, by mutual wri

In addition, either KKR or KFN may terminate the merger agreement at any time prior to the

there is any law or regulation that makes completion of the merger illegal or otherwise prohibited, or if are injunction, order or decree of a competent United States federal or state governmental authority enjoining the completing the merger is entered and has become final and nonappealable; provided that the party seeking pursuant to this provision is required to have used its reasonable best efforts to render inapplicable the law or remove the judgment, injunction, order or decree as required by the merger

the merger is not completed on or before September 16, 2014; provided, however, that the right to terminate pur provision will not be available to any party whose failure to perform any covenant or obligation under the merger has been the primary cause of, or resulted in, the failure of the merger to occur on or before September.

at the special meeting (including any adjournment or postponement of the special meeting) the merger agree adopted by the affirmative vote of the holders of at least a majority of the outstanding KFN common shares en thereon on the record date, voting together as a single class, and the holders of a majority of the outstanding K shares (other than those KFN common shares held by KKR or any affiliate of KKR); provided, however, that terminate pursuant to this provision will not be available to any party whose failure to perform any covenant under the merger agreement has been the primary cause of, or resulted in, the failure of the KFN shareholder approach to the terminate pursuant to the same transfer of the terminate pursuant to this provision will not be available to any party whose failure to perform any covenant under the merger agreement has been the primary cause of, or resulted in, the failure of the KFN shareholder approach to the terminate pursuant to this provision will not be available to any party whose failure to perform any covenant to the merger agreement has been the primary cause of, or resulted in, the failure of the KFN shareholder approach to the terminate pursuant to this provision will not be available to any party whose failure to perform any covenant to the terminate pursuant to this provision will not be available to any party whose failure to perform any covenant to the terminate pursuant to this provision will not be available to any party whose failure to perform any covenant to the terminate pursuant to this provision will not be available to any party whose failure to perform any covenant to the terminate pursuant to this provision will not be available to any party whose failure to perform any covenant to the terminate pursuant to the terminate pursua

the other party has materially breached any of its representations, warranties, covenants or agreements con merger agreement, or if any fact, circumstance, event, change, occurrence or effect has occurred, which be circumstance, event, change, occurrence or effect would result in the failure of certain closing conditions to be sa prior to the September 16, 2014, and the breach or fact, circumstance, event, change, occurrence or effect would result in the failure of certain closing conditions to be satisfactorized by the control of the september 16, 2014, and the breach or fact, circumstance, event, change, occurrence or effect would result in the failure of certain closing conditions to be satisfactorized by the second of the second o

Table

effect is not capable of being cured or is not cured by the earlier of (1) 30 business days after written notice is rec party alleged to be in breach or with respect to which a fact, circumstance, event, change, occurrence or effect have occurred and (2) Septemb

In addition, KKR may terminate the merger agreement if, at any time before the adoption of the merger agreement by K shareholders, (1) a change of recommendation has occurred or (2) an acquisition proposal is publicly announced or disclosed (or publicly announces an intention (whether or not conditional) to make an acquisition proposal) after the date of the merger agreement of directors of KFN fails to affirm the recommendation of the board of directors of KFN that holders of KFN common shat merger agreement by the later of (A) five business days before the special meeting (as it may be adjourned or postponed) and (B) that the public announcement or disclosure of the acquisition proposal and the special meeting (as it may be adjourned or postponed) and the special meeting (as it may be adjourned or postponed) and (B) that the public announcement or disclosure of the acquisition proposal and the special meeting (as it may be adjourned or postponed) and the special meeting (as it may be adjourned or postponed) and (B) that the public announcement or disclosure of the acquisition proposal and the special meeting (as it may be adjourned or postponed) and (B) that the public announcement or disclosure of the acquisition proposal and the special meeting (as it may be adjourned or postponed) and (B) that the public announcement or disclosure of the acquisition proposal and the special meeting (as it may be adjourned or postponed).

In addition, KFN may terminate the merger agreement at any time before the adoption of the merger agreement by KFN shat order to enter into a definitive written agreement with respect to a superior proposal, provided that KFN has complied in all mater with its obligations under described above under "No Solicitation by KFN of Alternative Proposals "and "Change Recommendation" and pays to Merger Sub or its designee in immediately available funds the termination payment (as described by the termination payment obligation is waived by

In some cases, termination of the merger agreement may require KFN to pay a termination payment to Merger Sub or its described below under " Termination Payment a

Termination Payment an

The merger agreement provides that KFN is required to pay a termination payment of \$26.25 million, which is referred to statement/prospectus as the termination payment, to Merger Sub or its designee (unless the payment obligation is waived by Merger sub or its designee) merger agreement is te

KKR because a change of recommendation h

either KKR or KFN because the merger agreement was not adopted at the special meeting (including any adj postponement of the special meeting) at a time when KKR had the right to terminate the merger agreement change of recom

KFN in order to enter into a definitive written agreement with respect to a superi

either KKR or KFN because (1) the merger is not completed on or before September 16, 2014 or (2) the merge was not adopted at the special meeting (including any adjournment or postponement of the special meeting acquisition proposal was publicly announced or disclosed (or any person publicly announced an intention acquisition proposal) within certain time periods specified in the merger agreement and (2) within 12 months after termination, KFN enters into a letter of intent, agreement-in-principle, acquisition agreement or other similar agreement to, or publicly announces, a business combination (as described below) or completes a business combination (as described below).

KKR on the basis of a breach by KFN of any representations, warranties, covenants or agreements contained i agreement (provided that the breach did not arise from or relate to a breach of KKR's obligation to cause the machine cause KFN to fa

Table

in any material respect with the merger agreement) (1) following the public announcement or disclosure of a proposal or the intention by any person to make an acquisition proposal that was not withdrawn prior to the term (2) within 12 months after the termination, KFN enters into a letter of intent, agreement-in-principle, acquisition or other similar agreement with respect to, or publicly announces, a business combination or consummate

For purposes of the merger agreement, the term "business combination" means (1) a merger, consolidation, acquisition, sha business combination, reorganization, recapitalization, dissolution, liquidation or similar transaction involving KFN, (2) any purpose equity interest (including by means of a tender or exchange offer) representing an amount equal to or greater than a 50% voting interest in KFN or (3) any purchase of assets, securities or ownership interests representing an amount equal to or greater that consolidated asset value (including only the portion of the value of the equity interests of the subsidiaries of KFN that is actually owned directly by KFN) or consolidated net revenues of KFN (including only the portion of the revenues attributable to the equity in subsidiaries of KFN that is actually owned directly or indirectly by KFN); provided that the calculation of consolidated asset consolidated net revenues includes only the unconsolidated interests in variable interest entities that are held directly or indirectly by the total asset value or revenues of the entire variable interest entity is not included in the

Generally, all costs and expenses incurred in connection with the merger agreement and the transactions contemplated b agreement will be paid by the party incurring the expenses, except that expenses incurred in connection with filing, printing and magnetical registration statement of which this proxy statement/prospectus forms a part and this proxy statement/prospectus (including filing shared equally by Fund Holdings and KFN. However, if the merger agreement is terminated by either KKR or KFN because agreement was not adopted at the special meeting (including any adjournment or postponement of the special meeting), KFN is indemnify and hold harmless KKR, Fund Holdings and Merger Sub (unless waived by those parties) for certain of their respective out-of-pocket costs, fees and expenses not to exceed \$7.5 million in the aggregate; provided, however, that upon payment of the payment in full, KFN will no longer be required to indemnify and hold harmless KKR, Fund Holdings and Merger Sub for their context expenses. In addition, if it is judicially determined that termination of the merger agreement was by reason of a willful or intention the merger agreement, then, in addition to other remedies at law or equity for a willful or intentional breach of the merger agreement waived by the other parties) for certain of their respective reasonable out-of-pocket costs, fees a waived by the other parties) for certain of their respective reasonable out-of-pocket costs, fees and expenses incurred in connection with the merger agreement will be required to indemnify and hold harmless the other parties) for certain of their respective reasonable out-of-pocket costs, fees and expenses incurred in connection will be required to indemnify and hold harmless the other parties.

Conduct of Business Pending the Consummation of

Under the merger agreement, each of KKR and KFN has undertaken certain covenants that place restrictions on it and subsidiaries from the date of the merger agreement until the effective time, unless the other party gives its prior written consent (whi unreasonably withheld, conditioned or delayed). In addition, KFN has generally agreed that it will (1) conduct the business of subsidiaries in all material respects only in the ordinary course of business and in a manner consistent in all material respects with and (2) use commercially reasonable efforts to preserve substantially intact its business organization and maintain and preserve intact

Subject to certain exceptions set forth in the merger agreement and the disclosure schedules delivered by KFN to KKR in conthe merger agreement, unless KKR conser

Table

(which consent cannot be unreasonably withheld, conditioned or delayed), KFN has agreed to certain restrictions limiting its and i subsidiaries' ability to, among

amend or otherwise change KFN's certificate of formation or operating

issue, deliver, sell, grant, pledge, dispose of or grant a lien on, or permit a lien to exist on, any shares of any clastock of KFN or any of its subsidiaries, any other voting securities or other ownership interests, or any optio convertible securities or other rights of any kind to acquire any shares of capital stock, voting securities or equity any "phantom" share, "phantom" share rights, share appreciation rights, share-based units or other similar interes any of its subsidiaries (except for the issuance of KFN common shares issuable pursuant to the exerci outstanding on the date of the merger agreement in accordance with their terms or the crediting of additional pha (and issuance of KFN common shares in settlement thereof) under the Deferred Compensation Plan in accordance terms as a result of deferral elections previously made (and not otherwise timely revoked) in respect of 20 compensation in the ordinary course of business, and the dividend reinvestment feature under the Deferred Compensation.

sell, lease, license, pledge or dispose of, or grant a lien or permit a lien to exist on, any properties, investments or or any interests therein of KFN or any of its subsidiaries that have an aggregate value in excess of

declare, set aside, make or pay any dividend, payable in cash, equity securities, property or otherwise, with respits capital stock, except for (1) dividends by any of KFN's wholly-owned subsidiaries to KFN or an wholly-owned subsidiaries and (2) any regular quarterly distribution made by KFN in an amount up to \$0 common share for any quarterly distribution (subject to the limitations described under " Dividends and Di

adjust, reclassify, combine, split, subdivide or redeem, or purchase or otherwise acquire any of its capital securities or other ownership interests or any securities convertible into or exchangeable or exercisable for countries or other ownership interests (except for the withholding of KFN common shares in respect of of the exercise price or taxes upon the exercise of any options, the vesting of KFN common restricted shares or the securities of the exercise price or taxes upon the exercise of any options, the vesting of KFN common restricted shares or the securities of the exercise price or taxes upon the exercise of any options, the vesting of KFN common restricted shares or the securities of the exercise price or taxes upon the exercise of any options, the vesting of KFN common restricted shares or the securities of the exercise price or taxes upon the exercise of any options.

(1) acquire, including by merger, consolidation or acquisition of stock or assets or any other business combination other manner, any corporation, partnership, other business organization or any business, division or equity intere (2) incur any indebtedness for borrowed money or issue any debt securities or assume, guarantee or endorse, become responsible for, the obligations of any person, or make any loans or advances or capital contributions investment in, any person, except to employees in the ordinary course of business and in a manner consist practice or to KFN or any wholly-owned subsidiary.

modify in any material respect any accounting policies or procedures, other than as required by GAAP or app

(1) make any change to any material method of tax accounting, (2) make, change or rescind any material (3) settle or compromise any material tax liability or consent to any claim or assessment relating to a material taxes, (4) file any amended tax return, (5) file any claim for refund of a material amount of taxes, (6) enter into agreement relating to a material amount of taxes or (7) waive or extend the statute of limitations in respect of m

Table

except (1) for the acceleration of the vesting of existing phantom shares and restricted KFN common shares, (2 by the provisions of the merger agreement relating to the treatment of equity awards or (3) to ensure that any be not then out of compliance with applicable law or the terms of the benefits plan on the date of the merge (A) adopt, enter into, terminate or amend any benefits plan, (B) increase in any manner the compensation or be director, officer, employee or independent contractor, (C) grant or pay any change-in-control, retention, termination pay to, or increase in any manner the change-in-control, retention, severance or termination pay of or former director, officer, employee or independent contractor, (D) grant any equity or equity-based awards loan to any employee or independent contractor (other than the crediting of additional phantom shares (and issue common shares in settlement thereof) under the Deferred Compensation Plan in accordance with the terms there of deferral elections previously made (and not otherwise timely revoked) in respect of 2013 and 2014 compensation to fund or in any other way secure the payment of compensation or benefits under any benefits plan; of terminate any officer, employee or independent terminate any officer, employee or independent

except as required by applicable law or any judgment by a court of competent jurisdiction, (1) pay, discharge, set any claims, liabilities, obligations or litigation that are material to KFN and its subsidiaries, taken as a whole, o payment, discharge, settlement or satisfaction in the ordinary course of business and in a manner consistent with of in accordance with their terms, of liabilities disclosed, reflected or reserved against in the financial statements thereto) contained in KFN's filings with the SEC (for amounts not in excess of such reserves) or incurred since the financial statements in the ordinary course of business and in a manner consistent with past practice, (2) cancel indebtedness or (3) waive or assign any claims or rights of ma

(1) enter into, (2) terminate or cancel, except when it may be commercially reasonable to do so, (3) fail to exerc renew on terms commercially reasonable to KFN or (4) modify or amend in any material respect, any of cer contracts of KFN specified in the merger ag

agree in writing or otherwise commit to take any of the foreg

Subject to certain exceptions set forth in the merger agreement and the disclosure schedules delivered by KKR to KFN in conthe merger agreement, unless KFN consents in writing (which consent cannot be unreasonably withheld, conditioned or delaye agreed to certain restrictions limiting its and its respective subsidiaries' ability to, among

amend or otherwise change the organizational documents of KKR, Fund Holdings or Merger Sub, examendments or changes that would not (1) materially delay, materially impede or prevent the completion of the (2) adversely affect the holders of KFN common shares in any material respect differently than the holders of K

declare, set aside, make or pay any extraordinary or special dividend or distribution, payable in cash, equi property or otherwise, with respect to any KKR cor

(1) acquire or merge with any business, person or division, if entering into a definitive agreement relating to the or merger would be reasonably likely to, or (2) incur any indebtedness for borrowed money or issue any debt seed incurrence or issuance would be reasonably likely to, in either case, (A) materially delay the effectiveness of the statement of which this proxy statement/prospectus forms a part or (B) materially delay, materially impede of completion of the statement of the statement

Table

issue, deliver, sell, grant or dispose of any KKR common units or shares of any class of capital stock of KK voting securities or other ownership interests of KKR, or any options, warrants, convertible securities or other kind to acquire any KKR common units, or any "phantom" stock, "phantom" stock rights, stock apprec stock-based units or other similar interests of KKR (except, in each case, (1) in the ordinary course (including the equity-based awards, any regular quarterly issuances of KKR common units made by KKR in accordance with exchange process and sales pursuant to a certain prospectus previously filed with the SEC by KKR), (2) for the KKR common units issuable pursuant to the settlement of awards with respect to directors, employees and serving of KKR or its affiliates outstanding on the date of the merger agreement or as may be granted in the ordinary cases in accordance with customary terms) or (3) in connection with acquisitions, business combinations, join strategic alliances or similar tr

adjust, reclassify, combine, split or subdivide any KKR common units, or any capital stock, voting secur ownership interests of KKR or any securities convertible into or exchangeable or exercisable for KKR com capital stock, voting securities or other ownership interests

agree in writing or otherwise commit to take any of the foreg

Indemnification; Directors' and Officer

The merger agreement provides that, for a period of six years from and after the effective time, KKR and the surviving entity i will be required to maintain in effect (1) the current provisions (or provisions no less favorable) regarding exculpation and indem present or former officers and directors contained in the organizational documents of KFN and its subsidiaries and (2) any agreem and its subsidiaries with any of their respective present or former directors and officers that provide rights to exculpation or indemiacts or omissions occurring at or prior to the effective time (whether asserted or claimed prior to, at or after the effective time, i matters arising in connection with the transactions contemplated by the merger agreement). In addition, the merger agreement provide and after the effective time, KKR and the surviving entity will be required to indemnify, defend and hold harmless, and provide adv expenses (including advancing attorneys' fees and expenses incurred in advance of the final disposition of any actual or threatene proceeding or investigation to the fullest extent permitted under applicable law, provided that any person to whom fees and advanced provides an undertaking to repay such advances if it is ultimately determined that such person is not entitled to indemnific present and former officers and directors of KFN against all losses, claims, damages, costs, expenses, liabilities or judgments th settlement of or in connection with any action based on, relating to or arising out of the fact that he or she is or was an officer or dire or its subsidiaries prior to the effective time, whether asserted or claimed prior to, or at or after, the effective time (including acts occurring in connection with the approval of the merger agreement and the completion of the transactions contemplated b agreement) to the fullest extent provided or permitted under the organizational documents of KFN and its subsidiaries and any ind agreement entered into between KFN or its subsidiaries and the director or officer, and to the fullest extent permitted under ap

In addition, the merger agreement provides that if KKR does not elect to purchase a "tail" directors' and officers' liability insufer KFN's present and former officers and directors with coverage for six years following the effective time, and with coverage and terms and conditions no less favorable to the covered persons than the existing policies of directors' and officers' liability insurance by KFN, KFN may purchase a six year "tail" insurance policy; provided that KFN may not, without KKR's consent, make a premium the extent it exce

Table

the current annual premium paid by KFN for its directors' and officers' liability insurance (or as much comparable "tail" insurance as that amount). If KFN is unable to purchase any such "tail" insurance, KKR is required to cause Merger Sub to, at KFN's request, paraintain (and honor the obligations under), during the six-year period beginning on the date of the effective time, a "tail" insurance respect to directors' and officers' liability insurance of comparable coverage and amounts and containing terms and conditions no le in the aggregate, to the officers and directors of KFN as KFN's existing policies with a claims period of six years from the effort provided, however, that Merger Sub is not required to pay more than 300% of the current annual premium paid by KFN for its officers' liability insurance as the aggregate premium for the "tail" insurance policies for its entire period (in which case Merger Suas much comparable "tail" insurance as possible for the same current annual premium paid by KFN for the "tail" insurance as possible for the same current annual premium paid by KFN for its conficers' liability insurance as the aggregate premium for the "tail" insurance policies for its entire period (in which case Merger Suas much comparable "tail" insurance as possible for the same current annual premium paid by KFN for its conficers' liability insurance as possible for the same current annual premium paid by KFN for its entire period (in which case Merger Suas much comparable "tail" insurance as possible for the same current annual premium paid by KFN for its entire period (in which case Merger Suas much comparable "tail" insurance as possible for the same current annual premium paid by KFN for its entire period (in which case Merger Suas much comparable "tail" insurance as possible for the same current annual premium paid by KFN for its entire period (in which case Merger Suas much comparable "tail" insurance as possible for the same current annual premium paid by KFN for its entire period (in which cas

Amendment

The merger agreement provides that it may be amended by the parties at any time before or after the adoption of the merger a KFN shareholders, but after any such approval, no amendment is permitted that by applicable law requires further approval or authorized the holders of KFN common shares without such further approval or a

At any time prior to the effective time, KKR (with respect to KKR, Fund Holdings and Merger Sub) and KFN (by ap transaction committee of the board of directors of KFN) may, to the extent legally allowed, (1) extend the time for the performance obligations or other acts of the other parties, (2) waive any inaccuracies in the representations and warranties contained in the merger or in any certificate delivered pursuant to the merger agreement and (3) waive compliance with any of the agreements or conditions the merger agreement or in any document delivered pursuant to the merge

Remedies; Specific F

The merger agreement provides that, in the event KFN pays the termination payment (described under "Termination Expenses") to Merger Sub or its designee when required, KFN will have no further liability to KKR, Fund Holdings or Merger respective affiliates, except in the case of fraud or any willful or intention breach of any provision of the merger agreement. In merger agreement provides that nothing in the merger agreement will relieve any party of any liability for fraud or any willful or breach of any provision of the merger agreement. The merger agreement also provides that the parties are entitled to an injunction or threatened violations of the merger agreement and to specifically enforce the merger

Representations and

The merger agreement contains representations and warranties made by KKR, Fund Holdings and Merger Sub to KFN, and repaid warranties made by KFN to KKR, Fund Holdings and Merger Sub. These representations and warranties have been made as benefit of the other parties to the merger agreement.

may be intended not as statements of fact or of the condition of the parties to the merger agreement or the subsidiaries, but rather as a way of allocating the risk to one of the parties if those statements prove to be

have been qualified by disclosures that were made to the other party in connection with the negotiation of agreement, which disclosures may not be reflected in the merge

may apply standards of materiality in a way that is different from what may be viewed as material to you or oth

The representations and warranties made by both KKR, Fund Holdings and Merger Sub, on the one hand, and KFN, on the

were made only as of the date of the merger agreement or such other date or dates as may be specified i



<u>Table</u>

relate to, among

agreement and are subject to more recent de

organization, standing, power and sim

Lugar Filling. SARRS FACILIO INC - FORM 10-Q
title t
b
certain contra
the opinion received from the financial advisor to the transaction committee of the board of direc
takeover
insur Additional representations and warranties made only by KKR, Fund Holdings and Merger Sub relate to, among
operations of
149

Table

status under provisions of KFN's operating agreement relating to ownership of KFN's capita

no knowledge of certain persons that any representation or warranty of KFN made in the merger agreemen inaccurate as of the date of the merge

Additional

It is a condition to the closing of the merger that the KKR common units to be issued in connection with the merger be approved on the NYSE, subject to official notice

In addition, the merger agreement provides that until the closing of the merger, the manager will not, and KKR will cause the to, in its capacity as manager of KFN, take any action (unless acting at the direction of the board of directors of KFN) that would c fail to comply in any material respect with any of the provisions of the merge

The merger agreement also contains covenants relating to, among other things, cooperation in the preparation statement/prospectus and the holding of the special meeting, access to information and confidentiality, the tax treatment of the mannouncements, dispositions of equity securities in the merger by persons who may be subject to the reporting requirements of Security the Exchange Act, litigation related to the merger, notice of certain matters, activities of Merger Sub and

UNAUDITED PRO FORMA CONDENSED COMBINED FINANCIAL STATEMENTS

The unaudited pro forma condensed combined statements of operations for the year ended December 31, 2013 combined consolidated statements of operations of KKR and KFN, giving effect to the merger as if it had occurred on January 1, 2013. To consolidated financial information has been adjusted in the unaudited pro forma condensed combined financial statements to give forma events that are (1) directly attributable to the merger, (2) factually supportable, and (3) with respect to the statements of expected to have a continuing impact on the combined results. The unaudited pro forma condensed combined financial information with the accompanying notes to the unaudited pro forma condensed combined financial statements. In addition, the proforma condensed combined financial information was based on and should be read in conjunct

separate historical audited financial statements of KKR as of and for the year ended December 31, 2013 and the included in KKR's Annual Report on Form 10-K for the year ended December 31, 2013, which is incorporated into this proxy statement/pro

separate historical audited financial statements of KFN as of and for the year ended December 31, 2013 and the included in KFN's Annual Report on Form 10-K for the year ended December 31, 2013, which is incorporated into this proxy statemen

The unaudited pro forma condensed combined financial information has been presented for informational purposes only and is most current information available. The pro forma information is not necessarily indicative of what the combined company's financial results of operations actually would have been had the merger been completed as of the dates indicated. In addition, the unaudite condensed combined financial information does not purport to project the future financial position or operating results of the combined financial information does not purport to project the future financial position or operating results of the combined financial purposes only and is most current information available.

The unaudited pro forma condensed combined financial information has been prepared using the acquisition method of acco existing U.S. generally accepted accounting principles, or GAAP standards, which are subject to change and interpretation. K treated as the acquirer in the merger for accounting purposes. The acquisition accounting is dependent upon certain valuations and that have yet to commence or progress to a stage where there is sufficient information for a definitive measurement. Accordingly, the adjustments are preliminary and have been made solely for the purpose of providing unaudited pro forma condensed combininformation. Differences between these preliminary estimates and the final acquisition accounting may occur and these differences material impact on the accompanying unaudited pro forma condensed combined financial statements and the combined company's formations and financial statements.

The unaudited pro forma condensed combined financial information does not reflect any cost savings, operating synergies enhancements that the combined company may achieve as a result of the merger or the costs to integrate the operations of KKR and costs necessary to achieve these cost savings, operating synergies and revenue en

Unaudited Pro Forma Condensed Combined Statement of Financial Condition

As of December 31, 2013

(Amounts in Thousands)

	KKR	KFN*	Pro Forma ljustments**		Pro Fori Combin
Assets					
Cash and Cash Equivalents	\$ 1,306,383	\$ 157,167	\$		\$ 1,463
Cash and Cash Equivalents Held at Consolidated					
Entities	440,808	319,121			759
Restricted Cash and Cash Equivalents	57,775	31,264			89
Investments	47,383,697	7,617,543	(12,302)	(a)	54,988
Due From Affiliates	143,908		(5,574)	(b)	138
Other Assets	2,094,630	592,103	342,064	(c)	3,028
Total Assets	\$ 51,427,201	\$ 8,717,198	\$ 324,188		\$ 60,468
Liabilities & Equity					
Debt Obligations	\$ 1,908,606	\$ 6,020,465	\$ (74,781)	(d)	\$ 7,854
Due to Affiliates	93,851	5,574	(5,574)	(e)	93
Accounts Payable, Accrued Expenses and Other					
Liabilities	2,839,926	163,425	32,802	(f)	3,036
Total Liabilities	4,842,383	6,189,464	(47,553)		10,984
Redeemable Noncontrolling Interests	627,807				627
Equity					
KKR & Co. L.P. Partners' Capital	2,722,010	2,166,112	491,563	(g)	5,379
Noncontrolling Interests	43,235,001	361,622	(119,822)	(h)	43,476
Total Equity	45,957,011	2,527,734	371,741		48,856
	, ,	, , , -	,		, -
Total Liabilities & Equity	\$ 51,427,201	\$ 8,717,198	\$ 324,188		\$ 60,468
<u> </u>	, ,	, ,	, -		,

The accompanying notes to the unaudited pro forma condensed combined financial statements are an integral part of thes

For purposes of the unaudited pro forma condensed combined financial statements, certain reclassifications were mathematical financial statements to conform to those classifications used by KKR. The reclassifications are explained *Accounting Policies and Financial Statement Classifications* beginning on page 155 of this proxy statement

The pro forma adjustments are explained in Note 6. Pro Forma Adjustments beginning on page 160 of this proxy statemen

Unaudited Pro Forma Condensed Combined Statement of Operations

For the year ended December 31, 2013

(Amounts in Thousands, Except Unit Data)

	KKR	KFN*	Pro Forma Adjustments**		Pro Fort Combin
Revenues					
Fees	\$ 762,546	\$ 119,178	\$ (78,409)	(i)	\$ 803
Expenses					
Compensation and Benefits	1,266,592	6,131	(6,131)		1,266
Occupancy and Related Charges	61,720	1,060	(1,060)		63
General, Administrative and Other	438,826	172,972	(71,218)	(1)	540
Total Expenses	1,767,138	180,163	(78,409)	l	1,868
Investment Income (Loss)					
Net Gains (Losses) from Investment Activities	7,826,082	97,409	(3,196)	(m)	7,920
Dividend Income	695,521	14,772	(3,190)	(111)	7,920
Interest Income	474,759	432,660	(12,148)	(n)	89:
Interest Expense	(99,616)	(190,159)		(n)	(27)
Total Investment Income (Loss)	8,896,746	354,682	(3,196)	ı	9,248
Income (Loss) Before Taxes	7,892,154	293,697	(3,196)	ı	8,182
Income Taxes	37,926	467	14,877	(o)	53
Net Income (Loss)	7,854,228	293,230	(18,073)		8,129
Less: Net Income (Loss) Attributable to	7,034,220	293,230	(10,073)		0,123
Redeemable Noncontrolling Interests	62,255				62
Less: Net Income (Loss) Attributable to	,				
Noncontrolling Interests	7,100,747		(61)	(p)	7,100
Less: Preferred Share Distributions		27,411		•	27
Net Income (Loss) Attributable to KKR & Co. L.P.	\$ 691,226	\$ 265,819	\$ (18,012)		\$ 939
Net Income (Loss) Attributable to					
KKR & Co. L.P. Per Common Unit					
Basic	\$ 2.51	1.31			\$
Diluted	\$ 2.30	\$ 1.31			\$
Weighted Average Common Units Outstanding					

Basic	274,910,628	202,411,445	(97,951,124) (q)	379,370
Diluted	300,254,090	202,411,445	(97,951,124) (q)	404,714

The accompanying notes to the unaudited pro forma condensed combined financial statements are an integral part of thes

For purposes of the unaudited pro forma condensed combined financial statements, certain reclassifications were man historical financial statements to conform to those classifications used by KKR. The reclassifications are explained *Accounting Policies and Financial Statement Classifications* beginning on page 155 of this proxy statement

The pro forma adjustments are explained in Note 6. Pro Forma Adjustments beginning on page 160 of this proxy statemen

<u>Table</u>

Notes to Unaudited Pro Forma Condensed Combined Financial Statements

(in thousands, except per share data)

1. Description of

On December 16, 2013, KFN, KKR, Fund Holdings and Merger Sub entered into a merger agreement, pursuant to which, sterms and conditions set forth in the merger agreement, KFN will be acquired by Fund Holdings, a subsidiary of KKR. If the merger by KFN common shareholders and the other conditions to completion of the merger that are described in this proxy statement/prostatisfied or waived, KFN will merge with and into Merger Sub, KFN continuing as the surviving entity as an indirect subsidiary of effective time of the merger, each issued and outstanding KFN common share will be converted into the right to receive 0.51 K units. Fractional shares shall be entitled to receive a cash payment in lieu of any fractional KKR common units. Each KFN coutstanding and unexercised immediately prior to the effective time of the merger will be cancelled and converted into the right amount in cash equal to the excess of (1) the cash value of the number of KKR common units that a holder of a KFN common share entitled to receive, over (2) the exercise price per KFN common share subject to such option immediately prior to the merge outstanding phantom share under KFN's Non-Employee Directors' Deferred Compensation and Share Award Plan shall autoconverted into the right to receive 0.51 KKR common units. Each preferred share of KFN shall remain an outstanding preferred share the surviving entity following

2. Basis of I

The unaudited pro forma condensed combined financial information was prepared using the acquisition method of account based on the historical financial statements of KKR and KFN. The acquisition method of accounting is based in accordance with Standards Codification Topic 805, *Business Combinations*, which is referred in this proxy statement/prospectus as ASC 805, and value concepts defined in Accounting Standards Codification Topic 820, *Fair Value Measurements*, which KKR has adopted as required existing U.S. GAAP standards are subject to change and in

ASC 805 requires, among other things, that the assets acquired and liabilities assumed be recognized at their estimated fair val acquisition date. In addition, it establishes that any equity-based consideration transferred be measured at the closing date of the rethen-current market price. This particular requirement will likely result in a per common unit equity component that is different from assumed in these unaudited pro forma condensed combined financia

Under the acquisition method of accounting, the assets acquired and liabilities assumed will be recorded as of the completion of at their respective fair values and added to those of KKR. Financial statements and reported results of operations of KKR issued after of the merger will reflect these values, but will not be retroactively restated to reflect the historical financial position or results of KFN. In addition, under ASC 805, acquisition-related transaction costs (i.e., advisory, legal, valuation, other professional fees) are as a component of consideration transferred but are accounted for as expenses in the periods in which the costs are incurred. The unformated combined statement of operations does not reflect any acquisition-related transaction costs. However, total acquisit transaction costs expected to be incurred by KKR and KFN, which are estimated to be approximately \$12.9 million and \$1.00 million are respectively, are reflected in the unaudited proformated combined statement of financial condition as an increase to account accrued expenses and oth

Notes to Unaudited Pro Forma Condensed Combined Financial Statements (Continued)

(in thousands, except per share data)

2. Basis of Presentation

Accounting Standards Codification Topic 820, *Fair Value Measurements* defines the term "fair value" and sets forth to requirements for any asset or liability measured at fair value, expands related disclosure requirements and specifies a hierarchy techniques based on the nature of the inputs used to develop the fair value measures. Fair value is defined as "the price that would be sell an asset or paid to transfer a liability in an orderly transaction between market participants at the measurement date." This is concept for the valuation of the asset or liability. In addition, market participants are assumed to be buyers and sellers in the prire most advantageous) market for the asset or liability. Fair value measurements for an asset assume the highest and best use by participants. Many of these fair value measurements can be highly subjective and it is also possible that other professionals, applying judgment to the same facts and circumstances, could develop and support a range of alternative estimation.

KFN has historically been externally managed and advised by an indirect subsidiary of KKR pursuant to the management agree that agreement, KKR manages KFN's day-to-day operations and performs all services and activities relating to the management of F liabilities and operations, subject to the direction and oversight of KFN's board of directors. While the management agreement will c the merger, the fee income and operating expenses and related balance sheet assets and liabilities previously recognized between KF will be eliminated in consolidation and therefore the pro forma effect of those eliminations are reflected in these unaudited pro form combined financial statements. This accounting treatment is referred to as the "effective settlement" of the management

There are certain entities that have historically been consolidated by KKR that hold investments in structured financing veh consolidated by KFN. As such, these investments and related income statement amounts will be eliminated in consolidation and therefore a effects of those eliminations are reflected in these unaudited pro forma condensed combined financial

If the merger were consummated on January 1, 2013, it is expected that KKR's total reportable segment results would be follows for the year ended December 31, 2013: (i) segment fees would decrease by approximately \$69 million, reflecting the elimin related to the aforementioned management agreement, (ii) segment expenses would increase by approximately \$29 million, reflecting total other expenses of \$98 million reduced by \$69 million for the elimination of related party management compensation and investment income would increase by approximately \$364 million. The net effect of these items is expected to result in an increase segment economic net income of approximately \$266 million. The \$18 million of net pro forma adjustments to the unaudite condensed combined statement of operations relate to the impact on noncontrolling interests associated with KKR Holdings a income taxes, each of which are excluded from KKR's segment economic

3. Accounting Policies and Financial Statement Cl

Upon consummation of the merger, KKR will review KFN's accounting policies. As a result of that review, KKR may identify between the accounting policies of the two companies that, when conformed, could have a material impact on the consolidar statements. The unaudited pro forma condensed combined financial statements do not assume any differences in accounting

For purposes of the unaudited pro forma condensed combined financial statements, certain reclassifications were made to KFI financial statements to conform to those cl

VEN CI---:

Accumulated Other Comprehensive Loss

Notes to Unaudited Pro Forma Condensed Combined Financial Statements (Continued)

(in thousands, except per share data)

3. Accounting Policies and Financial Statement Classifications

used by KKR. Upon completion of the merger, KKR will review KFN's accounting policies and financial statement classifications. A that review, it may become necessary to make additional reclassifications to the consolidated financial

Financial statement reclassifications are refl

WWD Classicians

Classification of KFN's Audited Consolidated Balance Sheet When Combined with KKR's Audited Consolidated Statement of Financial Condition As of December 31, 2013 (Amounts in Thousands)

KFN Classification			KKR Classification
Assets	Φ.	155.165	
Cash and Cash Equivalents	\$		Cash and Cash Equivalents
Restricted Cash and Cash Equivalents		350,385	Cash and Cash Equivalents Held at Consolidated Ent \$319,121 / Restricted Cash and Cash Equivalents \$3
Securities		573,312	Investments
Corporate Loans, Net		6,466,720	Investments
Equity Investments, at Estimated Fair Value		181,212	Investments
Oil and Gas Properties, Net		400,369	Other Assets
Interests in Joint Ventures and Partnerships		436,241	Investments \$385,802 / Other Assets \$50,439
Derivative Assets		30,224	Other Assets
Interest and Principal Receivable		33,570	Other Assets
Other Assets		87,998	Other Assets \$77,501 / Investments \$10,497
Total Assets	\$	8,717,198	
Liabilities			
Collateralized Loan Obligation Secured Notes	\$	5,249,383	Debt Obligations
Credit Facilities		125,289	Debt Obligations
Senior Notes		362,276	Debt Obligations
Junior Subordinated Notes		283,517	Debt Obligations
Accounts Payable, Accrued Expenses, and Other Liabilities		58,215	Accounts Payable, Accrued Expenses, and Other Lia
Accrued Interest Payable		23,575	Accounts Payable, Accrued Expenses, and Other Lia
Related Party Payable		5,574	Due to Affiliates
Derivative Liabilities		81,635	Accounts Payable, Accrued Expenses, and Other Lia
Total Liabilities		6,189,464	
Total Liabilities Shareholders' Equity		6,189,464	

Noncontrolling Interests \$361,622

(15,652) KKR & Co. L.P. Partners' Capital

Accumulated Deficit	(771,731) KKR & Co. L.P. Partners' Capital
Total Shareholders' Equity	2,527,734
Total Liabilities and Shareholders' Equity	\$ 8,717,198
	156

Notes to Unaudited Pro Forma Condensed Combined Financial Statements (Continued)

(in thousands, except per share data)

3. Accounting Policies and Financial Statement Classifications

Classification of KFN's Audited Consolidated Statement of Operations When Combined with KKR's Audited Consolidated Statement of Operations For the Year Ended December 31, 2013 (Amounts in Thousands)

KFN Classification		KKR Classification
Revenues		
Loan Interest Income	\$ 360,287	Interest Income
Securities Interest Income	49,518	Interest Income
Oil and Gas Revenue	119,178	Fees
Other	16,923	Dividend Income \$14,772/ Interest Income \$2,151
Total Revenues	545,906	
Investment Costs and Expenses		
Interest Expense	163,216	Interest Expense
Interest Expense to Affiliates	26,943	Interest Expense
Provision for Loan Losses		Net Gains (Losses) from Investment Activities
Oil and Gas Production Costs		
Oil and Gas Depreciation, Depletion and Amortization	44,061	General, Administrative and Other
Other	2,859	General, Administrative and Other
Total Investment Costs and Expenses Other Income	305,705	
Net Realized and Unrealized Gain on Investments	157,074	Net Gains (Losses) from Investment Activities
Net Realized and Unrealized Cost on Derivatives and Foreign	157,071	Net Gallis (Losses) from investment retivities
Exchange	(6.838)	Net Gains (Losses) from Investment Activities
Net Loss on Restructuring and Extinguishment of Debt		Net Gains (Losses) from Investment Activities
Other Income		Interest Income
Total Other Income	150,925	
Other Expenses		
Related Party Management Compensation	68,576	General, Administrative and Other \$65,052 / Comper and Benefits \$3,524
General, Administrative and Directors Expenses	19,524	General, Administrative and Other \$15,857 / Comper and Benefits \$2,607 / Occupancy and Related Charge
Professional Services	9,329	General, Administrative and Other
Total Other Expenses	97,429	
•	· ·	

Income Before Income Taxes		293,697	
Income Tax Expense (Benefit)		467	Income Taxes (Benefit)
Preferred Share Distributions		27,411	Preferred Share Distributions
Net Income Available to Common Shareholders	\$	265,819	
Net income Available to Common Shareholders	Ф	205,619	
		157	

Notes to Unaudited Pro Forma Condensed Combined Financial Statements (Continued)

(in thousands, except per share data)

4. Preliminary Estimated Acquisition Co

The following is a preliminary estimate of consideration expected to be transferred to effect the acquisition of KFN pursuant t

	`	nts in Thousands, er Share/Unit Data)
Number of KFN common shares outstanding at December 31, 2013	-	204,824
Exchange ratio		0.51
Number of KKR common units issued		104,460
KKR common unit price on March 7, 2014(a)	\$	24.28
Estimated fair value of issued KKR common units	\$	2,536,295
Plus: Estimated incremental value of Management Agreement(b)		145,000
Total preliminary estimated acquisition consideration	\$	2,681,295

The restricted KFN common shares are included in the KFN common shares outstanding. No additional consideration is a outstanding KFN options as the exercise price is above the current trading price of KFN common shares. Additionally, no cor attributable to KFN phantom shares, which is already reflected as a liability in the net ass

The estimated equity consideration expected to be transferred reflected in these unaudited pro forma condensed combin statements does not purport to represent what the actual consideration transferred will be when the merger is cons accordance with ASC 805, the fair value of equity securities issued as part of the consideration transferred will be meat closing date of the merger at the then-current market price. This requirement will likely result in a per common unit equity different from the \$24.28 assumed in these unaudited pro forma condensed combined financial statements and that differ material. An increase or decrease of \$1.00 in the KKR common unit price on the closing date of the merger compared to the assumed in these unaudited pro forma condensed combined financial statements would increase or decrease the considerat to be transferred by approximately \$105 million, which would result in a corresponding increase or decrease

Represents incremental consideration paid by KKR stemming from the effective settlement of the management agreem KKR and KFN upon consummation of the merger. The management agreement represents a preexisting contractual between the parties to the business combination that in effect becomes settled from an accounting perspective upon the connection with this effective settlement, KKR is anticipated to recognize as gain for accounting purposes the different between the management agreement and comparable third party arrangements under current market conditions, not to contractual termination fee provisions of the management agreement. The effective settlement of the management accounted for separately from the business combination and the deemed payment by KKR results in incremental consideration for the transaction. The analysis of the fair value of the management agreement has not been completed, and in its preliminary stages. Based on the limited analysis that has been conducted to date, we believe the reasonable range outcomes for the accounting gain that may be recorded is between approximately \$60 million and \$230 million, with the the range representation.

Notes to Unaudited Pro Forma Condensed Combined Financial Statements (Continued)

(in thousands, except per share data)

4. Preliminary Estimated Acquisition Consideration

contractual termination payment that would be due to be paid by KFN to KKR to terminate the management agreember 31, 2013. Because the analysis of the fair value of the management agreement is in its preliminary stages reflected in the unaudited pro forma condensed combined financial statements of \$145 million represents the midpoint described above. As such, we believe this amount could reasonably increase or decrease by \$85 million from the amoun the unaudited pro forma condensed combined financial statements, which would increase or decrease goodwill accordingly above represents management's best current estimate of the reasonably possible accounting gain that may be record currently available information and estimates and assumptions that KKR's management believe are reasonable as of the proxy statement/prospectus. Any of the factors underlying these estimates and assumptions may change or prove to different, and the estimates and assumptions may not be representative of facts existing at the closing date of the mer possible that no accounting gain is ultimately recorded on the effective settlement of the management agreement if the comparable third party arrangements under current market conditions is determined to be consistent with the management the date of closing or

5. Preliminary Estimated Purchase Pric

The following is a preliminary estimate of the assets to be acquired and the liabilities to be assumed by KKR in reconciled to the estimated consideration expected to be

	(Amoun	its in Thousands)
Book value of net assets acquired at December 31, 2013	\$	2,527,734
Adjusted for:		
Preferred shares		(361,622)
Adjusted book value of net assets acquired	\$	2,166,112
Adjustments to:		
Investments(a)		107,520
Other assets(a)		10,066
Debt obligations(a)		74,781
Accounts payable, accrued expenses and other liabilities(a)		(102)
Goodwill(b)		322,918
Preliminary estimated acquisition consideration	\$	2,681,295

To adjust the carrying value of certain assets and liabilities

Goodwill is calculated as the difference between the acquisition date fair value of the consideration expected to be transfervalues assigned to the assets acquired and liabilities assumed. Goodwill is no

In the case of this transaction, there were no known intangible asse

The allocation of the estimated acquisition consideration for KFN is preliminary because the merger with KFN has not yet been The preliminary allocation for KFN is based on estimates, assumptions, valuations and other activities which have not yet been final

Notes to Unaudited Pro Forma Condensed Combined Financial Statements (Continued)

(in thousands, except per share data)

5. Preliminary Estimated Purchase Price Allocation

make a definitive allocation. Accordingly, the pro forma adjustments to allocate the acquisition consideration will remain prelicious KKR's management determines the final acquisition consideration and the fair values of assets acquired, net of liabilities assum determination of the acquisition consideration allocation is anticipated to be completed as soon as practicable after the closing of the final amounts allocated to assets acquired and liabilities assumed, including intangible assets, could differ materially from the amount in the unaudited pro forma condensed consolidated financia

6. Pro Forma A

The adjustments included in the column under the heading "Pro Forma Adjustments" on the unaudited pro forma condense financial statements represent the following (amounts in

Unaudited Pro Forma Condensed Combined Statement of Financi

	De	cember 31, 2013
To adjust certain investments to fair value	\$	107,520
To eliminate certain investments held at consolidated entities*		(119,822)
Total adjustments to Investments	\$	(12,302)

There is an investment vehicle that has historically been consolidated by KKR where KFN holds an economic interest. A investments and corresponding interests will be eliminated in consolidation and therefore the pro forma effects of those are reflected in these unaudited pro forma condensed combined financial

Due fro

	ember 31, 2013
To eliminate amounts due from KFN to KKR	\$ (5,574)

	Dec	cember 31, 2013
To adjust amounts to fair value	\$	10,066
To record goodwill in connection with acquisition		322,918
To record deferred tax asset in connection with acquisition*		9,080
	\$	342,064

This amount represents a net deferred tax asset anticipated to be recorded in connection with the merger associated with a KKR Management Holdings (

<u>Table</u>

Notes to Unaudited Pro Forma Condensed Combined Financial Statements (Continued)

(in thousands, except per share data)

6. Pro Forma Adjustments

of the tax basis of the tangible and intangible assets of KKR Management Holdings L.P. as a result of the issuance of K units. The impact of this adjustment is reflected as an offset to equity in the unaudited pro forma condensed combined financial financial condensed combined condensed condensed

Debt

	December 31, 2013	
To adjust debt obligations to fair value	\$	(74,781)

Due

December 31, 2013	
\$	(5,574)

Accounts Payable, Accrued Expenses and Othe

	December 31, 2013	
To adjust amounts to fair value	\$	102
To record acquisition-related transaction costs expected to be incurred by KKR*		9,900
To record acquisition-related transaction costs expected to be incurred by KFN		22,800
	\$	32,802

The acquisition-related transaction costs incurred by KKR is expected to total \$12.9 million, of which \$3 million was accrued on KKR's consolidated statement of financial condition as of December 1.0 million was accrued to total \$12.9 million, of which \$3 million was accrued to total \$12.9 million, of which \$3 million was accrued to total \$12.9 million, of which \$3 million was accrued to total \$12.9 million, of which \$3 million was accrued to total \$12.9 million, of which \$3 million was accrued to total \$12.9 million, of which \$3 million was accrued to total \$12.9 million, of which \$3 million was accrued to total \$12.9 million, of which \$3 million was accrued to total \$12.9 million, of which \$3 million was account to the statement of the statem

The adjustment to KKR & Co. L.P. Partners' Capital is the effect of cumulative adjustments made t liabilities as descr

The adjustment to noncontrolling interests is the effect of an adjustment in (a) as described as the effect of an adjustment in (b) as described as the effect of an adjustment in (b) as described as the effect of an adjustment in (c) as described as the effect of an adjustment in (c) as described as the effect of an adjustment in (d) as described as the effect of an adjustment in (e) as described as the effect of an adjustment in (e) as described as the effect of an adjustment in (e) as described as the effect of an adjustment in (e) as described as the effect of an adjustment in (e) as described as the effect of an adjustment in (e) as described as the effect of an adjustment in (e) as described as the effect of an adjustment in (e) as described as the effect of an adjustment in (e) as described as the effect of an adjustment in (e) as described as the effect of an effect of a constant and the effect of a constant as the effect of a constant and the effect of a constant and the effect of a constant as the effect of a constant and the effect of

<u>Table</u>

Notes to Unaudited Pro Forma Condensed Combined Financial Statements (Continued)

(in thousands, except per share data)

6. Pro Forma Adjustments

Unaudited Pro Forma Condensed Combined Statements of

	 ar Ended tember 31, 2013
Elimination of management fees from KFN and related parties*	\$ (42,311)
Elimination of incentive fees from KFN*	(22,741)
Elimination of reimbursable expenses from KFN and related parties**	(13,357)
Total adjustment to Fees	\$ (78,409)

KKR and KFN are parties to the management agreement pursuant to which KKR manages the day-to-day operations of I to the direction and oversight of KFN's board of directors. This adjustment represents the removal of the historical management agreement by KKR and incurred by KFN for the year ended December 1.

Certain of KFN's operating costs are initially incurred by KKR. Such costs are then reimbursed to KKR pursuant to the management agreement, provided that such costs and reimbursements are no greater than those which would be particle professionals, consultants or other third parties on an arm's length basis. These costs generally consist of the following employee compensation and benefits costs since KFN does not have any employees, (ii) certain occupancy and related of KFN's administrative and principal executive offices are leased by KKR and (iii) certain general and administrative adjustment represents the removal of the historical expense reimbursements received by KKR for the year ended December 1.

Compensation a

	Year Ended December 31, 2013	
Elimination of employee compensation and benefits expense from KFN*	\$	(6,131)

KFN does not have any employees. KFN is managed by KKR Financial Advisors LLC, an indirect subsidiary of KKR, put management agreement. Certain employee compensation and benefits costs incurred by KKR are allocated to KFN put terms of the management agreement. This adjustment represents the removal of the historical compensation and benefits costs incurred by KFN for the year ended December 1.

<u>Table</u>

Notes to Unaudited Pro Forma Condensed Combined Financial Statements (Continued)

(in thousands, except per share data)

6. Pro Forma Adjustments

Occupancy and Rela

	Year Ended December 31, 2013	
Elimination of occupancy and related charges from KFN*	\$	(1,060)

KFN's administrative and principal executive offices are leased by KKR. Certain occupancy and related charges incurred allocated to KFN pursuant to the terms of the management agreement. This adjustment represents the removal of occupancy and related charges reimbursements paid by KFN for the year ended December 1.

General, Administrativ

	Decer	r Ended mber 31, 2013
Elimination of management fee expenses paid to KKR by KFN and related parties*	\$	(42,311)
Elimination of incentive fee expenses paid to KKR by KFN*		(22,741)
Elimination of reimbursable expenses paid to KKR by KFN and related parties**		(6,166)
Total adjustment to General, Administrative and Other	\$	(71,218)

KKR and KFN are parties to the management agreement pursuant to which KKR, through its indirect subsidiary, mana to-day operations of KFN subject to the direction and oversight of KFN's board of directors. This adjustment represents the historical management fees and incentive fees incurred by KFN and earned by KKR for the year ended December 1.

KKR has historically performed certain general and administrative services on behalf of KFN pursuant to the terms of the agreement. This adjustment represents the removal of the historical general, administrative and other charges paid by KFN ended December 1.

Net Gains (Losses) from Investme

	Year Ended December 31, 2013	
Elimination of net gains from consolidated entities*	\$	(3,196)

There is an investment vehicle that has historically been consolidated by KKR where KFN holds an economic interest. related income statement amounts will be eliminated in consolidation and therefore the pro forma effects of those eliminated in these unaudited pro forma condensed combined financial

<u>Table</u>

Notes to Unaudited Pro Forma Condensed Combined Financial Statements (Continued)

(in thousands, except per share data)

6. Pro Forma Adjustments

Interest Income/Inter

	ar Ended ember 31, 2013
Elimination of interest income from consolidated entities*	\$ (12,148)
Elimination of interest expense from consolidated entities*	12,148
	\$

There are certain entities that have historically been consolidated by KKR that hold investments in structured financing are consolidated by KFN. As such, the related income statement amounts will be eliminated in consolidation and thereform effects of those eliminations are reflected in these unaudited pro forma condensed combined financial

Income Tax

	Dece	r Ended ember 31, 2013
To record estimated incremental corporate income taxes*	\$	14,877

This adjustment represents the incremental corporate income taxes assumed to be incurred subsequent to the merger reincome that is expected to be generated by KKR Management Holdings L.P. as a result of the merger. This amou management's best estimate of incremental corporate income taxes based on currently available information and e assumptions that KKR's management believe are reasonable as of the date of this proxy statement/prospectus. Any cunderlying these estimates and assumptions may change or prove to be materially different, and the estimates and assumptions be representative of facts existing at the closing date of

Net Income (Loss) Attributable to Noncontrolli

Year Ended December 31, 2013

\$	3,135
	(3,196)
Φ	((1)
\$	(61)
	\$

Noncontrolling interests held by KKR Holdings L.P. include economic interests held by KKR's principals and other p KKR Group Partnerships. This adjustment allocates a portion of the incremental income in the combined business, as ref pro forma adjustments, to KKR Holdings L.P. based on KKR Holdings L.P.'s pro forma ownership percentage in the Partnerships for the

Notes to Unaudited Pro Forma Condensed Combined Financial Statements (Continued)

(in thousands, except per share data)

6. Pro Forma Adjustments

December 31, 2013. Such ownership percentages, which are reflective of the KKR common units assumed to be issued in with the merger, amounted to approximately 52.2% for the year ended December 31, 2013.

There is an investment vehicle that has historically been consolidated by KKR where KFN holds an economic interest. related income statement amounts will be eliminated in consolidation and therefore the pro forma effects of those eliminated in these unaudited pro forma condensed combined financia

Shares 6

	Year Ended December 31, 2013
Basic	
KKR weighted average units outstanding	274,910,628
Number of KKR common units issued*	104,460,321
Pro forma weighted average KKR common units outstanding	379,370,949
Diluted	
KKR weighted average units outstanding	300,254,090
Number of KKR common units issued*	104,460,321
Pro forma weighted average KKR common units outstanding	404,714,411

This amount assumes that there are no fractional units issued in connection with the merger. Fractional shares will repayment in lieu of any fractional KKR co

TRANSACTIONS IN KFN COMMON SHARES

Purcha

The following table shows purchases of KFN common shares during the past two years effected by KFN. The purchases v solely to satisfy withholding tax obligations incurred by certain employees of KFN as a result of the vesting of certain rest previously granted to such employees, who did not receive any cash proceeds from the

Quarter Ended	Total Number of Shares Purchased	Pr	lange of ices Paid er Share	ighted Average rice Paid per Share
March 31, 2012	10,264	\$	9.35	\$ 9.35
March 31, 2013	25,204	\$	11.09	\$ 11.09
Total	35,468			

Purchases by the KKR

The KKR Participants have not effected any purchases of KFN common shares during the pa

Transactions During the Pas

The following table shows grants of KFN phantom shares made on January 15, 2014 pursuant to the Deferred Compe

	Number of Shares	
Non-Employee Director	Granted	Implied Price
Vincent Paul Finigan	1,966.88	\$ 9.75
Paul M. Hazen	578.41	9.75
R. Glenn Hubbard	8,136.97	9.93
Deborah H. McAneny	1,544.44	9.75
Willy R. Strothotte	6.757.86	9.93

The following table shows transfers of KFN common shares made on March 10, 2014 by certain executive officers of Financial Advisors LLC. The transfers were effected in the ordinary course of business solely to satisfy withholding tax obligations such officers as a result of the vesting of certain restricted shares previously granted to such officers, who did not receive any cash previously granted to such officers.

	Number of Shares		
Officer	Transferred	Im	plied Price
Nicole J. Macarchuk	7,219	\$	12.22
Michael R. McFerran	10,859	\$	12.22

SECURITY OWNERSHIP OF CERTAIN BENEFICIAL OWNERS AND MANAGEMENT(1)

The following table shows, as of March 18, 2014 certain information about stock ownership of all persons known to KFN to or beneficially more than 5% of KFN's outstanding common shares based on KFN's review of SEC filings. This table also present information regarding each of KFN's named executive officers and directors based upon information furnished to KFN by these number and percentage of shares beneficially owned is determined under SEC rules, and the information is not necessarily beneficial ownership for any other purpose. Unless otherwise indicated in the footnotes, each person has sole voting and investment shares such powers with his or her spouse) with respect to the shares shown as beneficially of the shares shown as the sha

	Number of KFN Common Shares	Percentage of KFN Common Shares
Name and Address of Beneficial Owner(1)	Beneficially Owned	Beneficially Owned(2)
FMR LLC(3)	30,723,622	15.0%
Leon G. Cooperman(4)	19,564,131	9.6%
Morgan Stanley(5)	16,203,116	7.9%
Thornburg Investment Management, Inc.(6)	12,180,609	6.0%
Paul M. Hazen(7)	707,756	*
Craig J. Farr	0	*
Michael R. McFerran(8)	142,136	*
Nicole J. Macarchuk(9)	47,208	*
Tracy L. Collins(10)	68,043	*
Robert L. Edwards(11)	19,869	*
Vincent Paul Finigan(12)	12,607	*
R. Glenn Hubbard(13)	55,341	*
Ross J. Kari(14)	57,766	*
Ely L. Licht(15)	85,943	*
Deborah H. McAneny(16)	33,616	*
Scott C. Nuttall(17)	381,019	*
Scott A. Ryles(18)	32,688	*
Willy R. Strothotte(19)	0	*
All officers and directors as a group (14 persons)	1,643,992	*

Holdings represent less than 1% of all shares

The address for all officers and directors of KFN is c/o KKR Financial Advisors LLC, 555 California Street, 50th Floor, Sa

Based on 204,824,159 KFN common shares outstanding as of Mar

This beneficial ownership information is based upon the Schedule 13G/A filed with the SEC on February 14, 2014 by FN Edward C. Johnson 3d, chairman of FMR LLC. Fidelity Management & Research Company ("Fidelity"), a wholly-owned of FMR LLC, may be deemed to be the beneficial owner of 29,769,406 common shares as a result of acting as investmen various investment companies. Pyramis Global Advisors Trust Company ("PGATC"), an indirect wholly-owned FMR LLC, may be deemed to be the beneficial owner of 954,216 common shares as a result of its serving as investmen institutional accounts owning such shares. Mr. Johnson and FMR LLC, through their control of Fidelity and PGATC, each voting power on 954,216 common shares, (b) shared voting power on no common shares, (c) sole dispositive power or common shares and (d) shared dispositive power on no common shares. The address for FMR LLC and Mr. Johnson is Street, Boston, Massachusetts 02210. The address of PGATC is 900 Salem Street, Smithfield, Rhode Is

<u>Table</u>

This beneficial ownership information is based upon the Schedule 13G/A filed with the SEC on February 5, 201-Cooperman. Mr. Cooperman is the Managing Member of Omega Associates, L.L.C. ("Associates"). Associates is the ge of Omega Capital Partners, L.P. ("Capital LP"), Omega Capital Investors, L.P. ("Investors LP"), Omega Equity In ("Equity LP"), and Omega Charitable Partnership L.P. ("Charitable LP"). Mr. Cooperman is the President and majority st Omega Advisors, Inc. ("Advisors"), which serves as the investment manager to Omega Overseas Partners, Ltd. ("Overseas also serves as a discretionary investment advisor to a limited number of institutional clients (the "Managed Mr. Cooperman has investment authority over the Michael S. Cooperman WRA Trust (the "WRA Trust"), and is a trusted and Toby Cooperman Family Foundation (the "Foundation") and of The Cooperman Family Fund for a Jewish Fut Fund"). On the basis of these relationships, Mr. Cooperman may be deemed to have (a) sole voting power on 13,299, shares, (b) shared voting power on 6,264,531 common shares, (c) sole dispositive power on 13,299,600 common shares at dispositive power on 6,264,531 common shares. This consists of 2,949,030 common shares owned by Capital LP; 1,074, shares owned by Investors LP; 1,271,195 common shares owned by Equity LP; 3,398,687 common shares owned 100,000 common shares owned by Charitable LP, 6,264,531 common shares owned by the Managed Accounts; 2,700, shares owned by the Foundation; 1,000,000 common shares owned by Mr. Cooperman; 55,000 common shares owned Fund, 51,058 common shares owned by Mr. Cooperman's son, Michael S. Cooperman; and 700,000 common shares of WRA Trust. The address for Leon G. Cooperman is 11431 W. Palmetto Park Road, Boca Raton, Fl

This beneficial ownership information is based upon the Schedule 13G filed with the SEC on January 28, 2014 by Mor Morgan Stanley has (a) sole voting power on 15,775,540 shares, (b) shared voting power on 359,825 shares, (c) sole disposed on 16,203,116 shares and shared dispositive power on no shares. The address for Morgan Stanley is 1585 Broadway, New

This beneficial ownership information is based upon the Schedule 13G/A filed with the SEC on January 21, 2014 be Investment Management Inc. Thornburg Investment Management, Inc. has (a) sole voting power on 12,180,609 shares voting power on no shares, (c) sole dispositive power on 12,180,609 shares and (d) shared dispositive power on no shares. for Thornburg Investment Management, Inc. is 2300 North Ridgetop Road, Santa Fe, New M

Includes an aggregate of 61,567 common shares directly held by trusts for which Mr. Hazen is an investment advisor a may be deemed to have beneficial ownership, but has disclaimed any beneficial ownership. Includes 5,057 unvested restrict shares and excludes 26,212 phantom shares, which may only be distributed to Mr. Hazen on the first day of the year fremoval or separation from the KFN board of directors or, if earlier, upon a change in control of KFN. The merger will not change in control of KFN for

Includes 41,972 unvested restricted con

Includes 26,264 unvested restricted con

Includes 14,538 unvested restricted con

Includes 15,429 unvested restricted con

Includes 6,548 unvested restricted common shares and excludes 89,135 phantom shares, which may only be distributed to on the first day of the year following his removal or separation from the KFN board of directors or, if earlier, upon a chan of KFN. The merger will not constitute a change in control of KFN for

<u>Table</u>

Excludes 240,668 phantom shares, which may only be distributed to Mr. Hubbard on the first day of the year following his separation from the KFN board of directors or, if earlier, upon a change in control of KFN. The merger will not constitute control of KFN for

Includes 14,538 unvested restricted con

Includes 14,538 unvested restricted con

Excludes 69,991 phantom shares, which may only be distributed to Ms. McAneny on the first day of the year following he separation from the KFN board of directors or, if earlier, upon a change in control of KFN. The merger will not constitute control of KFN for

Includes 5,455 shares owned by an investment vehicle for which Mr. Nuttall's spouse exercises investi

Includes 14,538 unvested restricted con

Excludes 200,327 phantom shares, which may only be distributed to Mr. Strothotte on the first day of the year following his separation from the KFN board of directors or, if earlier, upon a change in control of KFN. The merger will not constitute control of KFN for

Table

SECURITY OWNERSHIP OF THE KKR PARTICIPANTS AND CERTAIN AFFILIATES

Other than as set forth in the following table, as of March 18, 2014, the KKR Participants and their respective directors, execu controlling persons and subsidiaries, as applicable, did not beneficially own any KFN common shares. The number and percen common shares beneficially owned is determined under SEC rules, and the information is not necessarily indicative of beneficial o

Name and Address of Beneficial Owner(1)	Number of KFN Common Shares Beneficially Owned	Percentage of KFN Common Shares Beneficially Owned(2)(3)
KKR Financial Advisors LLC(4)(5)	154,161	*
Henry R. Kravis(5)(6)	2,477,494	1.2%
George R. Roberts(5)(7)	1,998,476	1.0%
Todd A. Fisher	197,892	*
William J. Janetschek(8)	188,513	*
David J. Sorkin	25,558	*

Holdings represent less than 1% of all KFN common shares

Except as otherwise noted, the address for the KKR Participants and their respective directors, executive officers, control and subsidiaries, as applicable, is c/o KKR, 9 West 57th Street, Suite 4200, New York, New

Based on 204,824,159 KFN common shares outstanding as of Mar

As described in "The Merger Agreement Conditions to Consummation of the Merger" beginning on page 134 statement/prospectus, KKR and KFN may not complete the merger unless the merger agreement has been adopted by the vote of the holders of at least a majority of the outstanding KFN common shares entitled to vote thereon on the record dat the holders of a majority of the outstanding KFN common shares other than those KFN common shares held by KKR or an KKR. As of the close of business on the record date for the special meeting, affiliates of KKR beneficially owned and have vote 4,658,021 KFN common shares at the special meeting, representing approximately 2.3% of the KFN common share vote at the special meeting.

Represents 65,011 unvested restricted KFN common shares and 89,150 vested KFN common shares held by KI Advisors LLC. The address for KKR Financial Advisors LLC is 555 California Street, 50th Floor, San Francisco, Calif

The sole member of KKR Financial Advisors LLC is KKR Asset Management LLC. The sole member o Management LLC is Kohlberg Kravis Roberts & Co. L.P. The sole general partner of KKR Management Holdings Management Holdings Corp. The sole shareholder of KKR Management Holdings Corp. is KKR Group Holdings I general partner of KKR Group Holdings L.P. is KKR Group Limited. The sole shareholder of KKR Group Limited is KI general partner of KKR is KKR Management LLC. The designated members of KKR Management LLC are Henry F Georg

Represents KFN common shares held by Mr. Kravis, his spouse, an entity controlled by Mr. Kravis and certain trusts which may be deemed to beneficially own by virtue of Mr. Kravis being a trustee of such trusts as well as 65,011 unvested recommon shares and 89,150 vested KFN common shares held by KKR Financial Action of the common shares and 89,150 vested KFN common shares held by KKR Financial Action of the common shares and 89,150 vested KFN common shares held by KKR Financial Action of the common shares and 89,150 vested KFN common shares held by KKR Financial Action of the common shares and 89,150 vested KFN common shares held by KKR Financial Action of the common shares are common shares and 89,150 vested KFN common shares held by KKR Financial Action of the common shares are common shares and 89,150 vested KFN common shares held by KKR Financial Action of the common shares are common shares and 89,150 vested KFN common shares held by KKR Financial Action of the common shares are common shares and 89,150 vested KFN common shares held by KKR Financial Action of the common shares are common shares and 89,150 vested KFN common shares held by KKR Financial Action of the common shares are common shares are common shares and 89,150 vested KFN common shares are common shar

Represents KFN common shares held by entities controlled by Mr. Roberts or over which he may be deemed to exercis decisions, and a trust for the benefit of Mr. Ro

Table

which he is a trustee as well as 65,011 unvested restricted KFN common shares and 89,150 vested KFN common shares he Financial Advisors LLC. The address for Mr. Roberts is 2800 Sand Hill Road, Menlo Park, Calif

Represents KFN common shares held by Mr. Janetschek and certain trusts which Mr. Janetschek may be deemed to bene by virtue of Mr. Janetschek being a trustee of such trusts, including trusts where neither Mr. Janetschek nor his family

MATERIAL U.S. FEDERAL INCOME TAX CONSEQUENCES OF THE MERGER

This summary discusses the material U.S. federal income tax consequences of the merger to KFN shareholders. This summary provisions of the Code on the regulations promulgated thereunder and on published administrative rulings and judicial decisions, are subject to change at any time, possibly with retroactive effect. This discussion is necessarily general and may not apply to all a investors, some of which, such as banks, thrifts, insurance companies, persons liable for the alternative minimum tax, deale investors that do not hold their KFN common shares as capital assets, may be subject to s

The U.S. federal income tax consequences of the merger to a partner in an entity or arrangement treated as a partnership for income tax purposes that holds KFN common shares generally will depend on the status of the partner and the activities of the Partners in a partnership holding KFN common shares should consult their own

For purposes of this discussion, a "U.S. Holder" is a beneficial owner of KFN common shares that is for U.S. federal income to (i) an individual citizen or resident of the United States; (ii) a corporation (or other entity treated as a corporation for U.S. federal purposes) created or organized in or under the laws of the United States, any state thereof or the District of Columbia; (iii) an estat of which is subject to U.S. federal income taxation regardless of its source; or (iv) a trust which either (A) is subject to the primary of a court within the United States and one or more United States persons have the authority to control all substantial decisions of (B) has a valid election in effect under applicable Treasury regulations to be treated as a U.S. person. A "Non-U.S. Holder" is a beneficial owner of KFN common shares (other than a partnership) that is not a

KFN common shareholders should consult their own tax advisors concerning the U.S. federal, state and local income tax co in their particular situations of the merger, as well as any consequences under the laws of any other taxing jurisdiction. This disc addresses the material U.S. federal income tax considerations of the merger and does not address the tax considerations under any tax jurisdiction other than the United States. Non-U.S. Holders, therefore, should consult their own tax advisors regar consequences to them of the merger under the laws of their own taxing

Material U.S. Federal Income Tax Consequences of the Merger to U

Tax characterization of

The merger will be a taxable transaction to U.S. Holders for U.S. federal income tax purposes. In general, the merger will be taxable sale of a U.S. Holder's KFN common shares in exchange for the KKR common units and cash in lieu of fractional units re-

Amount and character of gain or los

Your receipt of KKR common units and cash in lieu of fractional units in exchange for your KFN common shares in the merg will cause you to recognize gain or loss measured by the difference, if any, between (i) the sum of (A) the fair market value common units received, (B) the amount of cash received and (C) your share of KFN's nonrecourse debt immediately prior to the (ii) your adjusted tax basis in your KFN common shares. Any such gain or loss recognized generally will be treated as capital gai will be long-term capital gain or loss if your holding period for your KFN common shares exceeds one year. However, a portion of a will be treated as ordinary income to the extent attributable to your allocable share of unrealized gain or loss in KFN's assets described in Section 751 of the Code. This would include (x) ordinary income KFN would recognize if it sold its debt instruments market discount, (y) unremitted earnings of any controlled foreign corporation, or "CFC," held by KFN, although in the case of a who is an individual, the amount treated as ordinary income may be a supplied to the code.

Table

pursuant to Section 1248 of the Code and (z) ordinary income attributable to unrealized receivables, inventory items and depletion, and intangible drilling and development cost, or "IDC," recapture. Such ordinary income could exceed the net taxable gain realized the merger and may be recognized even if there is a net taxable loss realized as a result of the merger. Thus, a holder of KFN cormay recognize both ordinary income and a capital loss as a result of the merger. If KFN holds stock in a passive foreign investment "PFIC," that is not a qualified electing fund, or "QEF," at the time of the merger you would be treated as disposing of an interest in the extent of your pro rata share of such PFIC stock held by KFN with any gain you are deemed to recognize as a result of such treated as ordinary income and subject to an interest in the stock of the property of the p

Capital gain recognized by a U.S. Holder will generally be long-term capital gain subject to tax at preferential rates if such U.S. individual who has held his or her KFN common shares for more than twelve months on the date of the merger. The deductibil losses is subject to

KFN items of income, gain, loss and deduction for the taxable period ending on the date of

U.S. Holders of KFN common shares will be allocated their share of KFN's items of income, gain, loss and deduction for the ta of KFN ending on the date of the merger in accordance with the terms of KFN's operating agreement. A U.S. Holder will be su federal income taxes on any such allocated income and gain even though such U.S. Holder will not receive any additional cash districe KFN attributable to such allocated income and gain. Any such income and gain allocated to a U.S. Holder will increase the U.S. basis in the KFN common shares held and, therefore, will reduce the gain (or increase the loss) recognized by such U.S. Holder return the merger. Any losses or deductions allocated to a U.S. Holder will decrease the U.S. Holder's tax basis in the KFN common shares the gain (or reduce the loss) recognized by such U.S. Holder resulting from

Tax basis in KKR common units received in

A U.S. Holder's tax basis in the KKR common units received in the merger will equal the fair market value of such KKR co

Holding period in KKR common units received in

A U.S. Holder's holding period in the KKR common units received in the merger will begin on the day after the date o

Material U.S. Federal Income Tax Consequences of the Merger to Non-V

If you are a Non-U.S. Holder of KFN common shares, a portion of any gain recognized by you in the merger (which will be the same manner described above for a U.S. Holder) may be treated for U.S. federal income tax purposes as effectively connected hence you may be subject to U.S. federal income tax on such portion. Moreover, if the fair market value of KFN's investments in "Ureal property interests," which include KFN's investments in natural resources, real estate and REIT subsidiaries that invest princestate, represent 50% or more of the total fair market value of KFN's assets, KFN common units could be treated as "United States interests." In such case, gain recognized by a Non-U.S. Holder on the disposition of KFN common shares would be treated for income tax purposes as effectively connected income (unless the KFN common shares are regularly traded on a securities may non-U.S. Holder did not own more than 5% of all KFN common shares at any time during the shorter of the five-year period precedent of disposition of such Non-U.S. Holder's holding period). K

Table

believe that the fair market value of KFN's investments in "United States real property interests" represents 50% or more of the total value of KFN's assets at this time. If gain from the disposition of KFN common units is treated as effectively connected income, the Holder may be subject to U.S. federal income and/or withholding tax on the

In general, different rules from those described above apply in the case of Non-U.S. Holders subject to special treatment under income tax law, including a Non-U.S. Holder (i) who has an office or fixed place of business in the United States or is otherwise of U.S. trade or business; (ii) who is an individual present in the United States for 183 or more days or has a "tax home" in the United States for 183 or more days or has a "t

If you are a Non-U.S. Holder, you are urged to consult your tax advisors with regard to the U.S. federal income tax consequence owning and disposing of KFN common units, as well as the effects of state, local and non-U

Backup Withholding and Informatio

U.S. federal backup withholding tax may apply to cash payments to a U.S. Holder under the merger agreement, unless the
(i) furnishes an accurate tax identification number and otherwise complies with applicable U.S. information reporting or
requirements (typically, by submitting a completed and signed IRS Form W-9) or (ii) is a corporation (other than certain S corporation
exempt recipient and, when required, demonstrates such fact. Cash payments made pursuant to the merger will also be subject to
reporting unless an exemp

In general, a Non-U.S. Holder will not be subject to U.S. federal backup withholding and information reporting with re payments to the Non-U.S. Holder under the merger agreement if the Non-U.S. Holder has provided an IRS Form W-8BE Form W-8ECI if the Non-U.S. Holder's gain is effectively connected with the conduct of a U.S. trade or business). If shares are he foreign partnership or other flow-through entity, certain documentation requirements also apply to the partnership or other flow-th

The backup withholding tax is not an additional tax. Any amounts withheld under the backup withholding rules may be allowed against a holder's U.S. federal income tax liability and may entitle such holder to a refund, so long as the required information is time

The U.S. federal income tax consequences described above are not intended to constitute a complete descripti consequences relating to the merger. Because individual circumstances may differ, each holder should consult its own regarding the applicability of the rules discussed above to such holder and the particular U.S. federal income tax conseque holder of the merger in light of such holder's particular circumstances and the application of state, local and forei

Table

DESCRIPTION OF KKR COMMON UNITS

KKR Con

KKR common units represent limited partner interests in KKR, which is a limited partnership. KKR common unitholders a participate in distributions and exercise the rights or privileges available to limited partners under KKR's partnership agreen dependent upon the KKR Group Partnerships to fund any distributions that KKR may make to its common unitholders. For a descriptive rights and preferences of KKR common unitholders in and to distributions, see "Management's Discussion and Analysis Condition and Results of Operations Liquidity Liquidity Needs Distributions," in KKR's Annual Report on Form 10-K for December 31, 2013 filed with the SEC on February 24, 2014 and incorporated by reference in this proxy statemen

Unless KKR's managing partner determines otherwise, KKR issues all its common units in uncerti

Number of Common Units and Furth

As of March 18, 2014, KKR had approximately 300,354,288 common units outstanding. KKR's partnership agreement author issue an unlimited number of additional partnership securities and options, rights, warrants and appreciation rights relating to securities for the consideration and on the terms and conditions established by KKR's managing partner in its sole discretion without of its common unitholders. In accordance with the Delaware Limited Partnership Act and the provisions of KKR's partnership agree may also issue additional partnership interests that have designations, preferences, rights, powers and duties that are different from senior to, those applicable to its co

Where KKR Common Units

KKR's outstanding common units are listed on the NYSE under the symbol "KKR." It is a condition to the closing of the me KKR common units to be issued in connection with the merger be authorized for listing on the NYSE, subject to official notice

Transfer of KKR Co

By acceptance of the transfer of KKR common units in accordance with KKR's partnership agreement, each transferee of K units will be admitted as a unitholder with respect to the common units transferred when such transfer and admission is reflected in I and records. Additionally, each transferee of KKR co

will represent that the transferee has the capacity, power and authority to enter into the KKR partnership

will become bound by the terms of, and will be deemed to have agreed to be bound by, the KKR partnership agr

will give the consents, approvals, acknowledgements and waivers set forth in the KKR partnershi

A transferee will become a substituted limited partner of the KKR limited partnership for the transferred KKR common units a upon the recording of the transfer on KKR's books and records. KKR's managing partner may cause any transfers to be recorded on I and records no less frequently the

<u>Table</u>

KKR common units are securities and are transferable according to the laws governing transfers of securities. In addition to acquired upon transfer, the transferor gives the transferee the right to become a substituted limited partner in the KKR limited partner transferred KKR co

Until a KKR common unit has been transferred on KKR's books, KKR and the transfer agent, notwithstanding any notice to may treat the record holder of the KKR common unit as the absolute owner for all purposes, except as otherwise required by exchange regulations. A beneficial holder's rights are limited solely to those that it has against the record holder as a result of ar between the beneficial owner and the record holder.

Transfer Agent ar

American Stock Transfer & Trust Company, LLC serves as registrar and transfer agent for the KKR co

<u>Table</u>

CONFLICTS OF INTEREST AND FIDUCIARY RESPONSIBILITIES

Conflicts

Conflicts of interest exist and may arise in the future as a result of the relationships between KKR's managing partner and including each party's respective owners, on the one hand, and KKR and its limited partners, on the other hand. KKR's managing partner of Management LLC, is the general partner of KKR that manages KKR. Whenever a potential conflict arises between KKR's managing its affiliates, on the one hand, and KKR or any limited partner, on the other hand, KKR's managing partner will resolve that cor partnership agreement contains provisions that reduce and eliminate KKR's managing partner's duties, including fiduciary duti unitholders. KKR's partnership agreement also restricts the remedies available to KKR's unitholders for actions taken that we limitations might constitute breaches of duty, including fidu

Under KKR's partnership agreement, KKR's managing partner will not be in breach of its obligations under KKR's partnership a its duties to KKR or KKR's unitholders if the resolution of the

approved by KKR's conflicts committee, although KKR's managing partner is not obligated to seek suc

approved by the vote of a majority of the outstanding common units, excluding any common units own managing partner or any of its affiliates, although KKR's managing partner is not obligated to seek such

on terms which are, in the aggregate, no less favorable to KKR than those generally being provided to or av unrelated thir

fair and reasonable to KKR, taking into account the totality of the relationships among the parties involved, inc transactions that may be particularly favorable or advantage

KKR's managing partner may, but is not required to, seek the approval of such resolution from the conflicts committe unitholders. If KKR's managing partner does not seek approval from the conflicts committee or KKR's unitholders and the board of KKR's managing partner determines that the resolution or course of action taken with respect to the conflict of interest satisfies standards set forth in the third and fourth bullet points above, then it will be presumed that in making its decision the board of direct good faith, and in any proceeding brought by or on behalf of any limited partner or KKR or any other person bound by KKR's agreement, the person bringing or prosecuting such proceeding will have the burden of overcoming such presumption. Unless the reconflict is specifically provided for in KKR's partnership agreement, KKR's managing partner or the conflicts committee may factors it determines in its sole discretion to consider when resolving a conflict. KKR's partnership agreement provides that KKF partner will be conclusively presumed to be acting in good faith if KKR's managing partner subjectively believes that the determina not made is in the best interesting the conflicts committee in the partner will be conclusively presumed to be acting in good faith if KKR's managing partner subjectively believes that the determination of the partner will be conclusively presumed to be acting in good faith if KKR's managing partner subjectively believes that the determination of the partner will be conclusively presumed to be acting in good faith if KKR's managing partner subjectively believes that the determination of the partner will be conclusively presumed to be acting in good faith if KKR's managing partner subjectively believes that the determination of the partner will be conclusively presumed to be acting in good faith if KKR's managing partner subjectively believes that the determination of the partner will be conclusively presumed to be acting the partner will be conclusively presumed to be acting the pa

Covered

The conflicts committee is responsible for enforcing KKR's rights under certain agreements against KKR Holdings and subsidiaries and designees, a general partner or limited partner of KKR Holdings, or a person who holds a partnership or equity is foregoing entities. The conflicts committee is also authorized to take any action pursuant to any authority or rights granted to succunder those agreements or with respect to any amendment, supplement, modification or waiver to any those agreement that wou modify such authority or rights

Table

the conflicts committee shall approve any amendment to those agreements that in the reasonable judgment of KKR's managing partr directors creates or will result in a conflic

Potent

Conflicts of interest could arise in the situations described below, as

Actions taken by KKR's managing partner may affect the amount of cash flow from operations to KKR's common

The amount of cash flow from operations that is available for distribution to KKR's common unitholders is affected by decision managing partner regarding suc

the amount and timing of cash expenditures, including those relating to co

the amount and timing of investments and o

levels of in

levels of re

issuances of additional partnersh

In addition, borrowings by KKR and KKR's affiliates from KKR's managing partner and its affiliates do not constitute a breach owed by KKR's managing partner to KKR's common unitholders. KKR's partnership agreement provides that KKR and its substrow funds from KKR's managing partner and its affiliates on terms that are fair and reasonable to KKR. Under KKR's partnership those borrowings will be deemed to be fair and reasonable if: (1) they are approved in accordance with the terms of the partnership (2) the terms are no less favorable to KKR than those generally being provided to or available from unrelated third parties or (3) fair and reasonable to KKR, taking into account the totality of the relationships between the parties involved, including other transport may be or have been particularly favorable or advantage

KKR will reimburse its managing partner and its affiliates f

KKR will reimburse its managing partner and its affiliates for costs incurred in managing and operating KKR and KKR's be example, KKR does not elect, appoint or employ any directors, officers or other employees. All of those persons are elected, employed by KKR's managing partner on KKR's behalf. KKR's partnership agreement provides that its managing partner will dexpenses that are alloc

KKR's managing partner has limited its liability regarding KKR's

KKR's managing partner has limited its liability under contractual arrangements so that the other party has recourse only to ke and not against KKR's managing partner, its assets or its owners. KKR's partnership agreement provides that any action taken by a partner to limit its liability or KKR's liability is not a breach of KKR's managing partner's fiduciary duties, even if KKR could he more favorable terms without the limitation on liability. The limitation on KKR's managing partner's liability does not constitute compliance with U.S. federal securities laws that would be void under Section 14 of the Section 14 of

<u>Table</u>

KKR's common unitholders will have no right to enforce obligations of KKR's managing partner and its affiliates under agre

Any agreements between KKR on the one hand, and KKR's managing partner and its affiliates on the other, will not grant KK unitholders, separate and apart from KKR, the right to enforce the obligations of KKR's managing partner and its affiliates in I

Contracts between KKR, on the one hand, and KKR's managing partner and its affiliates, on the other, will not be the result of a

KKR's partnership agreement allows KKR's managing partner to determine in its sole discretion any amounts to pay itself or its any services rendered to KKR. KKR's managing partner may also enter into additional contractual arrangements with any of its KKR's behalf. Neither KKR's partnership agreement nor any of the other agreements, contracts and arrangements between KKR on t and KKR's managing partner and its affiliates on the other, are or will be the result of arm's-length negotiations. KKR's managing determine the terms of these transactions so long as such arrangements are fair and reasonable to KKR as determined under KKR's agreement. KKR's managing partner and its affiliates will have no obligation to permit KKR to use any facilities or assets of KKF partner and its affiliates, except as may be provided in contracts entered into specifically dealing with such use. There will not be are of KKR's managing partner and its affiliates to enter into any contracts

KKR's common units are subject to KKR's managing partner's limit

KKR's managing partner may exercise its right to call and purchase common units as provided in KKR's partnership agreem this right to one of its affiliates or to KKR. KKR's managing partner may use its own discretion, free of fiduciary duty restrictions, in whether to exercise this right. As a result, a KKR common unitholder may have such holder's common units purchased from such undesirable time or price. See "Description of KKR's Limited Partnership Agreement Limited Call Right" beginning on page 208 statemen

KKR may choose not to retain separate counsel for KKR or for the holders of its co-

Attorneys, independent accountants and others who will perform services for KKR are selected by KKR's managing partner or committee, and may perform services for KKR's managing partner and its affiliates. KKR may retain separate counsel for KK common unitholders in the event of a conflict of interest between KKR's managing partner and its affiliates on the one hand, and unitholders on the other, depending on the nature of the conflict, but is not requi

KKR's managing partner's affiliates may compete

KKR's partnership agreement provides that KKR's managing partner will be restricted from engaging in any business activitic activities incidental to its ownership of interests in KKR. Except as provided in the non-competition, non-solicitation and coagreements to which KKR's principals will be subject, affiliates of KKR's managing partner, including its owners, are not provided in the non-competition, non-solicitation and coagreements to which KKR's principals will be subject, affiliates of KKR's managing partner, including its owners, are not provided in the non-competition, non-solicitation and coagreements to which KKR's principals will be subject, affiliates of KKR's managing partner, including its owners, are not provided in the non-competition, non-solicitation and coagreements to which KKR's principals will be subject, affiliates of KKR's managing partner, including its owners, are not provided in the non-competition, non-solicitation and coagreements to which KKR's principals will be subject, affiliates of KKR's managing partner, including its owners, are not provided in the non-competition, non-solicitation and coagreements to which KKR's principals will be subject, affiliates of KKR's managing partner, including its owners, are not provided in the non-competition, non-solicitation and coagreements to which KKR's principals will be subject, affiliates of KKR's managing partner, including its owners, are not provided in the non-competition.

Certain of KKR's subsidiaries have obligations to investors in KKR's investment funds and may have obligations to other third may conflict with you

KKR's subsidiaries that serve as the investment advisors or general partners of KKR's investment funds have fiduciary and obligations to the investors in those funds and so

Table

subsidiaries, including its broker-dealers, may have contractual duties to other third parties. As a result, KKR expects to regularly with respect to the allocation of investments among its investment funds (including funds that have different fee structures), the pur of investments in its investment funds, the structuring of investment transactions for those funds, the advice and services KKF otherwise that comply with these fiduciary and contractual obligations. In addition, KKR's principals have made personal investigation of its investment funds, which may result in conflicts of interest among investors in its funds or its unitholders regarding decisions for these funds. Some of these actions might at the same time adversely affect KKR's near-term results of operations of these funds.

U.S. federal income tax considerations of KKR's principals may conflict with yo

Because KKR's principals will hold a portion of their KKR Group Partnership units directly or through entities that are recorporate income taxation and KKR holds its units in one of the KKR Group Partnerships through a subsidiary that is subject to corporation in the United States, conflicts may arise between KKR's principals and KKR relating to the selection and structuring of or transactions. KKR's common unitholders will be deemed to expressly acknowledge that KKR's managing partner is under no consider the separate interests of such holders, including among other things the tax consequences to KKR's common unitholders whether to cause KKR to take or decline to take

Fidu

KKR's managing partner is accountable to KKR and KKR's common unitholders as a fiduciary. Fiduciary duties owed to KKR's by KKR's managing partner are prescribed by law and KKR's partnership agreement. The Delaware Limited Partnership Act pelaware limited partnerships may in their partnership agreements expand, restrict or eliminate the duties, including fiduciary duties owed by a general partner to limited partners and the

KKR's partnership agreement contains various provisions modifying, restricting and eliminating the duties, including fiduciar might otherwise be owed by KKR's managing partner. KKR has adopted these restrictions to allow KKR's managing partner or its engage in transactions with KKR that would otherwise be prohibited by state-law fiduciary duty standards and to take into account of other parties in addition to KKR's interests when resolving conflicts of interest. Without these modifications, KKR's managing part to make decisions involving conflicts of interest would be restricted. These modifications are detrimental to KKR's common unithol they restrict the remedies available to KKR's common unitholders for actions that without those limitations might constitute break including a fiduciary duty, as described below, and they permit KKR's managing partner to take into account the interests of the addition to KKR's interests when resolving conflicts.

<u>Table</u>

The following is a summary of the material restrictions on the fiduciary duties owed by KKR's managing partner to KKR's

State Law Fiduciary Duty Standards

General

Partnership Agreement Modified Standards

Fiduciary duties are generally considered to include an obligation to act in good fait due care and loyalty. In the absence of a provision in a partnership agreement provide otherwise, the duty of care would generally require a general partner to act for the partnership agreement providing otherwise, the duty of loyalty would generally general partner of a Delaware limited partnership from taking any action or engaging transaction that is not in the best interests of the partnership where a conflict of interpresent.

KKR's partnership agreement contains provisions that waive duties of or consent to KKR's managing partner and its affiliates that might otherwise raise issues about co with fiduciary duties or applicable law. For example, KKR's partnership agreement that when KKR's managing partner, in its capacity as KKR's managing partner, is proor required to make a decision in its "sole discretion" or "discretion" or that it deems or appropriate" or "necessary or advisable" then KKR's managing partner will be enconsider only such interests and factors as it desires, including its own interests, and no duty or obligation (fiduciary or otherwise) to give any consideration to any facto KKR or any limited partners, including KKR's common unitholders, and will not be any different standards imposed by KKR's partnership agreement, the Delaware Lin Partnership Act or under any other law, rule or regulation or in equity. In addition, wair managing partner is acting in its individual capacity, as opposed to in its capacity as managing partner, it may act without any fiduciary obligation to KKR or the common unitholders whatsoever. These standards reduce the obligations to which KKR's mapartner would otherwise be held.

In addition to the other more specific provisions limiting the obligations of KKR's n partner, KKR's partnership agreement further provides that KKR's managing partner officers and directors will not be liable to KKR, KKR's limited partners, including K common unitholders, or assignees for errors of judgment or for any acts or omission there has been a final and non-appealable judgment by a court of competent jurisdic determining that KKR's managing partner or its officers and directors acted in bad for engaged in fraud or willful misconduct.

General

Special Provisions Regarding Affiliated Transactions

KKR's partnership agreement generally provides that affiliated transactions and reconflicts of interest not involving a vote of unitholders and that are not approved by conflicts committee of the board of directors of KKR's managing partner or by KKR unitholders must be:

on terms no less favorable to KKR than those generally being provided to or availab unrelated third parties; or

fair and reasonable to KKR, taking into account the totality of the relationships between parties involved (including other transactions that may be particularly favorable or advantageous to KKR).

If KKR's managing partner does not seek approval from the conflicts committee or unitholders and the board of directors of KKR's managing partner determines that the or course of action taken with respect to the conflict of interest satisfies either of the set forth in the bullet points above, then it will be presumed that in making its decision board of directors acted in good faith, and in any proceeding brought by or on behall limited partner, including KKR's common unitholders, or KKR or any other person KKR's partnership agreement, the person bringing or prosecuting such proceeding where the obligations to KKR's managing partner would otherwise be held.

Rights and Remedies of Unitholders

The Delaware Limited Partnership Act generally provides that a limited partner may legal action on behalf of the partnership to recover damages from a third-party when partner has refused to institute the action or where an effort to cause a general partner not likely to succeed. In addition, the statutory or case law of some jurisdictions may limited partner to institute legal action on behalf of himself and all other similarly sillimited partners to recover damages from a general partner for violations of its fiduct to the limited partners. KKR's partnership agreement provides that legal action may instituted in Delaware.

By holding KKR common units, each common unitholder will automatically agree to be bound by the provisions in KKR's agreement, including the provisions described above and as described in "Description of KKR's Limited Partnership Agreement" It page 201 of this proxy statement/prospectus. This is in accordance with the policy of the Delaware Limited Partnership Act principle of freedom of contract and the enforceability of partnership agreements. The failure of a common unitholder to sign KKR's agreement does not render KKR's partnership agreement unenforceable against

Table

KKR has agreed to indemnify KKR's managing partner and any of its affiliates and any member, partner, tax matters partirector, employee, agent, fiduciary or trustee of KKR, KKR's managing partner or any of KKR's affiliates and certain other specifies the fullest extent permitted by law, against any and all losses, claims, damages, liabilities, joint or several, expenses (including lexpenses), judgments, fines, penalties, interest, settlements or other amounts incurred by KKR's managing partner or these other partner or the full loss of the full loss of the partner or the full loss of the full l

MATERIAL U.S. FEDERAL TAX CONSEQUENCES OF KKR COMMON UNIT OWNERSHIP

This summary discusses the material U.S. federal tax considerations related to the ownership and disposition of KKR commothed the date hereof. This summary is based on provisions of the U.S. Code on the regulations promulgated thereunder and administrative rulings and judicial decisions, all of which are subject to change at any time, possibly with retroactive effect. This necessarily general and may not apply to all categories of investors, some of which, such as banks, thrifts, insurance companies, possibly general and may not apply to all categories of investors, some of which, such as banks, thrifts, insurance companies, possibly general and may not apply to all categories of investors, some of which, such as banks, thrifts, insurance companies, possibly general and may not apply to all categories of the alternative minimum tax, dealers, investors who were deemed to own 10% or more of any foreign corporation owned by KKR account the investor's interest in such foreign corporation as a result of their ownership interest in KKR or otherwise), and other is do not own their KKR common units as capital assets, may be subject to special rules. Tax-exempt organizations and mutual funds a separately below. The actual tax consequences of the ownership of KKR common units will vary depending on your circumstances.

For purposes of this discussion, a "U.S. Holder" is a beneficial owner of KKR common units that is for U.S. federal income to (i) an individual citizen or resident of the United States; (ii) a corporation (or other entity treated as a corporation for U.S. federal purposes) created or organized in or under the laws of the United States, any state thereof or the District of Columbia; (iii) an estat of which is subject to U.S. federal income taxation regardless of its source; or (iv) a trust which either (A) is subject to the primary of a court within the United States and one or more United States persons have the authority to control all substantial decisions of (B) has a valid election in effect under applicable Treasury regulations to be treated as a U.S. person. A "Non-U.S. Holder" is a benefit of KKR common units (other than a partnership) that is not a

If a partnership holds KKR common units, the tax treatment of a partner in the partnership will depend upon the status of the activities of the partnership. If you are a partner of a partnership that holds KKR common units, you should consult your tax a discussion does not constitute tax advice and is not intended to be a substitute for t

Potential KKR common unitholders should consult their own tax advisors concerning the U.S. federal, state and local inc estate tax consequences in their particular situations of the ownership and disposition of KKR common units, as well as any counter the laws of any other taxing jurisdiction. This discussion only addresses the material U.S. federal tax considerations of the and disposition of KKR common units and does not address the tax considerations under the laws of any tax jurisdiction of United States. Non-U.S. Holders, therefore, should consult their own tax advisors regarding the tax consequences to them of the and disposition of KKR common units under the laws of their own taxing.

Taxat

Subject to the discussion set forth in the next paragraph, an entity that is treated as a partnership for U.S. federal income tax purposes and incurs no U.S. federal income tax liabilities. Each partner of a partnership in take into account its allocable share of items of income, gain, loss and deduction of the partnership in computing its U.S. federal liability, regardless of the extent to which, or whether, it receives cash distributions from the partnership, and thus may inculiabilities unrelated to (and in excess of) any distributions from the partnership. Distributions of cash by a partnership to a partner are unless the amount of cash distributed to a partner is in excess of the partner's adjusted basis in its partner.

An entity that would otherwise be classified as a partnership for U.S. federal income tax purposes may nonetheless be corporation if it is a "publicly traded partnership," unless an exception applies. An entity that would otherwise be classified as a partnership if (i) interests in the partnership are traded on an established securities market or (ii) in

Table

partnership are readily tradable on a secondary market or the substantial equivalent thereof. KKR is a publicly traded

However, an exception to taxation as a corporation, referred to as the "Qualifying Income Exception," exists if at least partnership's gross income for every taxable year consists of "qualifying income" and the partnership is not required to regist Investment Company Act. Qualifying income includes certain interest income, dividends, real property rents, gains from the disposition of real property, and any gain from the sale or disposition of a capital asset or other property held for the production of otherwise constitutes qualify

KKR's managing partner has adopted a set of investment policies and procedures that govern the types of investments KKR ca income KKR can earn), including structuring certain investments through entities, such as KKR's intermediate holding company, corporations for U.S. federal income tax purposes (as discussed further below), to ensure that KKR will meet the Qualifying Incom in each taxable year. Except as otherwise noted, the remainder of this discussion assumes that KKR will be treated as a partnership corporation for U.S. federal income to

If KKR fails to meet the Qualifying Income Exception, other than a failure that is determined by the IRS to be inadvertent and within a reasonable time after discovery, or if KKR is required to register under the Investment Company Act, KKR will be treate transferred all of its assets, subject to liabilities, to a newly formed corporation, on the first day of the year in which it fails to meet the Income Exception, in return for stock in that corporation, and then distributed the stock to the KKR common unitholders in liquidation terests in KKR. Based on current law, this deemed contribution and liquidation would be tax-free to KKR common unitholders so I does not have liabilities in excess of the tax basis of its assets at that time. Thereafter, KKR would be treated as a corporation for income.

If KKR were treated as a corporation in any taxable year, either as a result of a failure to meet the Qualifying Income otherwise, its items of income, gain, loss and deduction would be reflected only on its tax return rather than being passed through common unitholders, and it would be subject to U.S. corporate income tax on its taxable income. Distributions made to K unitholders would be treated as either taxable dividend income, which may be eligible for reduced rates of taxation, to the extra current or accumulated earnings and profits, or in the absence of earnings and profits, as a nontaxable return of capital, to the holder's tax basis in the KKR common units, or as taxable capital gain, after the holder's basis is reduced to zero. In addition, i Non-U.S. Holders, distributions treated as dividends would be subject to withholding tax. Accordingly, treatment as a corpo materially reduce a holder's after-tax return and thus could result in a reduction of the value of the KKR co

If at the end of any taxable year KKR fails to meet the Qualifying Income Exception, it may still qualify as a partnership if it relief under the Code for an inadvertent termination of partnership status. This relief will be available if: (i) the failure is cureasonable time after discovery; (ii) the failure is determined by the IRS to be inadvertent; and (iii) KKR agrees to make such (including adjustments with respect to its partners) or to pay such amounts as are required by the IRS. It is not possible to state w would be entitled to this relief in any or all circumstances. If this relief provision is inapplicable to a particular set of circumstance KKR, it will not qualify as a partnership for federal income tax purposes. Even if this relief provision applies and KKR retains its status, it or its unitholders (during the failure period) will be required to pay such amounts as are determined

Table

Taxation of KKR's Intermediate Holdin

The income derived by KKR from KKR's fund management services and certain other investments likely will not be qualifying purposes of the Qualifying Income Exception. Therefore, in order to meet the Qualifying Income Exception, KKR holds interests Group Partnership that holds such fund management companies and other investments that may not generate qualifying income for the Qualifying Income Exception, indirectly through KKR's intermediate holding company, KKR Management Holdings Corp., what is a corporation for U.S. federal income to

As the holder of KKR Management Holdings Corp. common stock, KKR is not taxed directly on the earnings of KKR I Holdings Corp. or the earnings of entities held through KKR Management Holdings Corp. Rather, as a partner of KKR I Holdings L.P., KKR Management Holdings Corp. incurs U.S. federal income taxes on its proportionate share of any net taxable incomangement Holdings L.P. KKR Management Holdings Corp.'s liability for U.S. federal income taxes and applicable state, local and could be increased if the IRS were to successfully reallocate income or deductions of the related entities conducting KK

Distributions of cash or other property that KKR receives from KKR Management Holdings Corp. will constitute dividends for income tax purposes to the extent paid from KKR Management Holdings Corp.'s current or accumulated earnings and profits (a under U.S. federal income tax principles). If the amount of a distribution by KKR Management Holdings Corp. exceeds its accumulated earnings and profits, such excess will be treated as a tax-free return of capital to the extent of KKR's tax basis Management Holdings Corp. common stock, and thereafter will be treated as a

If KKR forms, for other purposes, a U.S. corporation or other entity treated as a U.S. corporation for U.S. federal income tax p corporation would be subject to U.S. federal income tax or

Personal Holding

KKR Management Holdings Corp. could be subject to additional U.S. federal income tax on a portion of its income if it is dete a personal holding company, or PHC, for U.S. federal income tax purposes. Subject to certain exceptions, a U.S. corporation will be a PHC for U.S. federal income tax purposes in a given taxable year if (i) at any time during the last half of such taxable year, individuals (without regard to their citizenship or residency and including as individuals for this purpose certain entities su tax-exempt organizations and pension funds) own or are deemed to own (pursuant to certain constructive ownership rules) more that stock of the corporation by value and (ii) at least 60% of the corporation's adjusted ordinary gross income, as determined for U.S. fee tax purposes, for such taxable year consists of PHC income (which includes, among other things, dividends, interest, royalties, a under certain circumstate.

Due to applicable attribution rules, it is likely that five or fewer individuals or tax-exempt organizations will be treated as own or constructively more than 50% of the value of KKR Management Holdings Corp. common stock. Consequently, KKR Management Corp. could be or become a PHC, depending on whether it fails the PHC gross income test. If, as a factual matter, the incompanient Holdings Corp. fails the PHC gross income test, it will be a PHC. Certain aspects of the gross income test cannot be precentainty. Thus, no assurance can be given that KKR Management Holdings Corp. will not become a PHC following this offer

Table

If KKR Management Holdings Corp. is or were to become a PHC in a given taxable year, it would be subject to an additional 2 on its undistributed PHC income, which generally includes the company's taxable income, subject to certain adjustments. If KKR Holdings Corp. were to become a PHC and had significant amounts of undistributed PHC income, the amount of PHC tax could however, distributions of such income reduce the PHC income s

Certain State, Local and Non-U.S.

KKR and its subsidiaries may be subject to state, local or non-U.S. taxation in various jurisdictions, including those in wh transact business, own property or reside. For example, KKR and its subsidiaries may be subject to New York City unincorporated KKR may be required to file tax returns in some or all of those jurisdictions. The state, local or non-U.S. tax treatment of KKR and unitholders may not conform to the U.S. federal income tax treatment discussed herein. KKR will pay non-U.S. taxes, and dispositio property or operations involving, or investments in, foreign property may give rise to non-U.S. income or other tax liability in amount be substantial. Any non-U.S. taxes incurred by KKR may not pass through to KKR common unitholders as a credit against their

Consequences to U.S. Holders of KKR Con

The following is a summary of the material U.S. federal income tax consequences that will apply to you as a U.S. Holder of K

For U.S. federal income tax purposes, your allocable share of KKR's items of income, gain, loss, deduction or credit will be KKR's partnership agreement if such allocations have "substantial economic effect" or are determined to be in accordance with yo KKR. KKR believes that for U.S. federal income tax purposes, such allocations will have substantial economic effect or be in accordance with your interest in KKR, and KKR's managing partner intends to prepare tax returns based on such allocations. If the IRS successfull the allocations made pursuant to KKR's partnership agreement, the resulting allocations for U.S. federal income tax purposes in favorable than the allocations set forth in KKR's partnership.

The characterization of an item of KKR's income, gain, loss, deduction or credit will be determined at the KKR (rather than a Similarly, the characterization of an item of Fund Holdings' income, gain, loss deduction or credit will be determined at the I Holdings or the level of any subsidiary partnership in which Fund Holdings owns an interest rather than at the KKR level. Distrib receives from KKR Management Holdings Corp. will be taxable as dividend income to the extent of KKR Management Holdings Co and accumulated earnings and profits and, to the extent allocable to individual holders of KKR common units, they will be eligible for rate of tax of 20%, provided that certain holding period requirements are satisfied. Also, a U.S. Holder that is a corporation limitations, may be entitled to a dividends received deduction with respect to its shares of dividends paid to KKR by KKR I

KKR may derive taxable income from an investment that is not matched by a corresponding distribution of cash. In add provisions of the Code may be applicable to certain of KKR's investments, and may affect the timing of KKR's income, requiring consequently, you) to recognize taxable income before KKR (or you) receive cash, if any, attributable to such income. Accordingly, that your allocable share of KKR's income for a particular taxable year could exceed any cash distribution you receive for the year rise to an out-of-pocket tax liab

<u>Table</u>

You will have an initial tax basis in your KKR common units equal to the amount paid for your KKR common units. Your increased by your share of KKR's income and by increases in your share of KKR's liabilities, if any. Your basis will be decreased, be zero, by distributions from KKR, by your share of KKR's losses and by any decrease in your share of KKR.

If you acquire KKR common units in separate transactions you must combine the basis of those units and maintain a single basis for all those units. Upon a sale or other disposition of less than all of the KKR common units, a portion of that tax basis must to the KKR common units.

Limits on Deductions for Losses a

Your deduction of your share of KKR's losses will be limited to your tax basis in your KKR common units and, if you are an in corporate holder that is subject to the "at risk" rules, to the amount for which you are considered to be "at risk" with respect to KKR's that is less than your tax basis. In general, you will be at risk to the extent of your tax basis in your KKR common units, reduct portion of that basis attributable to your share of KKR's liabilities for which you will not be personally liable and (2) any amount of borrow to acquire or hold your KKR common units, if the lender of those borrowed funds owns an interest in KKR, is related to you only to the KKR common units for repayment. Your at risk amount will generally increase by your allocable share of KKR's increased by cash distributions to you and your allocable share of losses and deductions. You must recapture losses deducted years to the extent that distributions cause your at risk amount to be less than zero at the end of any taxable year. Losses disallowed as a result of these limitations will carry forward and will be allowable to the extent that your tax basis or at risk amount, whill limiting factor, subsequently increases. Any excess loss above that gain previously suspended by the at risk or basis limitations may

KKR does not expect to generate income or losses from "passive activities" for purposes of Section 469 of the Code. According allocated to you by KKR may not be offset by your Section 469 passive losses and losses allocated to you may not be used to Section 469 passive income. In addition, other provisions of the Code may limit or disallow any deduction for losses by you can associated with certain assets of KKR in certain cases. You should consult with your tax advisors regarding the limitations on the deal losses that you may be subject to under applicable sections

Limitations on Deductibility of Organizational Expenses and Synd

Neither KKR nor any U.S. Holder may deduct organizational or syndication expenses. Syndication fees (which would include placement fees or commissions or underwriting discount payable to third parties) must be capitalized and cannot be amortized

Limitations on Interest

Your share of KKR's interest expense is likely to be treated as "investment interest" expense. If you are a non-corporate U.S deductibility of "investment interest" expense is limited to the amount of your "net investment income." Your share of KKR's of interest income will be treated as investment income, although "qualified dividend income" subject to reduced rates of tax in the individual will only be treated as investment income if you elect to treat such dividend as ordinary income not subject to reduced rate addition, U.S. state and local tax laws may disallow deductions for your share of KKR's interest.

Table

The computation of your investment interest expense will take into account interest on any margin account borrowing or other I to purchase a common unit. Net investment income includes gross income from property held for investment and amounts treated income under the passive loss rules less deductible expenses, other than interest, directly connected with the production of investment does not include long-term capital gains attributable to the disposition of property held for investment. For this purpose, at capital gain or qualifying dividend income that is taxable at long-term capital gain rates is excluded from net investment income, un Holder elects to pay tax on such gain or dividend income at ordinary in the production of the production of investment income, un the production of property held for investment.

Deductibility of Partnership Investment Expenditures by Individual Partners and by Trusts

Subject to certain exceptions, all miscellaneous itemized deductions of an individual taxpayer, and certain of such deductions of trust, are deductible only to the extent that such deductions exceed 2% of the taxpayer's adjusted gross income. Moreover, t allowable itemized deductions of individuals whose gross income exceeds an applicable threshold amount are subject to reduction be equal to the lesser of (1) 3% of the excess of the individual's adjusted gross income over the threshold amount, or (2) 80% of the an itemized

The operating expenses of Fund Holdings, including any management fees paid, may be treated as miscellaneous itemize subject to the foregoing rule. Accordingly, if you are a non-corporate U.S. Holder, you should consult your tax advisors with rapplication of these

Treatment of I

Distributions of cash by KKR will not be taxable to you to the extent of your adjusted tax basis (described above) in your K units. Any cash distributions in excess of your adjusted tax basis will be considered to be gain from the sale or exchange of your K units (described below). Under current laws, such gain would be treated as capital gain and would be long-term capital gain if period for your KKR common units exceeds one year, subject to certain exceptions (described below). A reduction in your alloc KKR's liabilities, and certain distributions of marketable securities by KKR, are treated similar to cash distributions for U.S. federal

Sale or Exchange of KKR Co

You will recognize gain or loss on a sale of KKR common units equal to the difference, if any, between the amount realizadjusted tax basis in the KKR common units sold. Your amount realized will be measured by the sum of the cash or the fair ma other property received plus your share of KKR's liabilities, if any, at the time of such sale

Subject to the exceptions discussed in this paragraph, gain or loss recognized by you on the sale or exchange of a KKR common taxable as capital gain or loss and will be long-term capital gain or loss if your holding period in your KKR common units is gree year on the date of such sale or exchange. If KKR has not made a qualifying electing fund election, or QEF election, to treat KKR's passive foreign investment company, or PFIC, as a qualified electing fund, or QEF, gain attributable to such an interest would ordinary income and would be subject to an interest charge. In addition, certain gain attributable to KKR's investment in a controcorporation, or CFC, may be ordinary income and certain gain attributable to "unrealized receivables" or "inventory iten characterized as ordinary income rather than capital gain. For example, if KKR holds debt acquired at a market discount, accordinate to the deductibility of capital losses is subject to

Table

Holders who acquire units at different times and intend to sell all or a portion of the units within a year of their most recent urged to consult their tax advisors regarding the application of certain "split holding period" rules to them and the treatment of any goal long-term or short-term capital

Foreign Tax Credit

Subject to certain exceptions and limitations, you will be entitled to a foreign tax credit with respect to your allocable share foreign taxes paid on KKR's income and gains (other than the income and gains of KKR's intermediate holding company). Comple depending on your particular circumstances, limit the availability or use of foreign tax credits. Gains from the sale of foreign investm treated as U.S. source gains. Consequently, you may not be able to use the foreign tax credit arising from any foreign taxes important sunless such credit can be applied (subject to applicable limitations) against tax due on other income treated as derived sources. Certain losses that KKR incurs may be treated as foreign source losses, which could reduce the amount of foreign tax credits.

Section

KKR has an election in place pursuant to Section 754 of the Code. The election is irrevocable without the consent of the legenerally require KKR to adjust the tax basis in its assets, or "inside basis," attributable to a transferee of KKR common Section 743(b) of the Code to reflect the purchase price of the KKR common units paid by the transferee. In addition, KKR Holdings L.P. has made a Section 754 election. Therefore, similar adjustments will be made upon the transfer of interests in KKR.

Even though KKR will have a Section 754 election in effect, because there is no Section 754 election in effect for Fund Holding will not make an election for it, it is unlikely that KKR's Section 754 election will provide any substantial benefit or detriment to a KKR co

The calculations involved in the Section 754 election are complex. KKR will make them on the basis of assumptions as to KKR's assets and o

Uniformity of KKR Common Units, Transferor/Transfered

Because KKR cannot match transferors and transferees of KKR common units, it will adopt depreciation, amortization accounting positions that may not conform with all aspects of existing Treasury regulations. A successful IRS challenge to those po adversely affect the amount of tax benefits available to you. It also could affect the timing of these tax benefits or the amount of gai of KKR common units and could have a negative impact on the value of KKR common units or result in audits of and adjustment common unitholders.

In addition, generally KKR's taxable income and losses will be determined and apportioned among investors using convergered as consistent with applicable law. As a result, if you transfer your KKR common units, you may be allocated income, g deduction realized by KKR after the date of transfer. Similarly, a transferee may be allocated income, gain, loss and deduction realized prior to the date of the transferee's acquisition of KKR co

Although Section 706 of the Code generally provides guidelines for allocations of items of partnership income and deducti transferors and transferees of partner interests, it is not clear that KKR's allocation method complies with its requirements. If KKR' were not permitted, the IRS might contend that KKR's taxable income or losses must be reallocated among the investors. If such were sustained, your respective tax liabilities would be adjute.

Table

possible detriment. KKR's managing partner is authorized to revise its method of allocation between transferors and transfered among investors whose interests otherwise vary during a tax

Foreign Currency

KKR's functional currency will be the U.S. dollar, and its income or loss will be calculated in U.S. dollars. It is likely the recognize "foreign currency" gain or loss with respect to transactions involving non-U.S. dollar currencies. In general, foreign currency loss is treated as ordinary income or loss. You should consult your tax advisor with respect to the tax treatment of foreign currency

Passive Foreign Investmen

KKR may own directly or indirectly interests in foreign entities that are treated as corporations for U.S. federal income tax put may be subject to special rules as a result of your indirect investments in such foreign corporations, including the rules apprinted investment in a passive foreign investment company, or PFIC. KKR Management Holdings Corp. will be subject to similar rules apprinted below with respect to any PFICs owned directly or including the rules apprinted below with respect to any PFICs owned directly or including the rules apprinted below with respect to any PFICs owned directly or including the rules apprinted below with respect to any PFICs owned directly or including the rules apprinted below with respect to any PFICs owned directly or including the rules apprinted below with respect to any PFICs owned directly or including the rules apprinted below with respect to any PFICs owned directly or including the rules apprinted below with respect to any PFICs owned directly or including the rules apprinted below with respect to any PFICs owned directly or including the rules apprinted below with respect to any PFICs owned directly or including the rules apprinted below with respect to any PFICs owned directly or including the rules apprinted below with respect to any PFICs owned directly or including the rules apprinted by the rules apprinted

A PFIC is defined as any foreign corporation with respect to which either (1) 75% or more of the gross income for a taxable year income" or (2) 50% or more of its assets in any taxable year (generally based on the quarterly average of the value of its assets) prod income." There are no minimum stock ownership requirements for shareholders in PFICs. Once a corporation qualifies as a PFIC it certain exceptions, always treated as a PFIC, regardless of whether it satisfies either of the qualification tests in subsequent years. disposition of stock of a PFIC, as well as income realized on certain "excess distributions" by the PFIC, is treated as though realized the shorter of your holding period in KKR common units or KKR's holding period in the PFIC. Such gain or income is taxable income and dividends paid by a PFIC to an individual will not be eligible for the reduced rates of taxation that are available for certain dividends. In addition, an interest charge would be imposed on you based on the tax deferred from

Although it may not always be possible, KKR expects to make a QEF election under the Code where possible with respect to treated as a PFIC to treat such non-U.S. entity as a QEF in the first year KKR holds shares in such entity. A QEF election is effet taxable year for which the election is made and all subsequent taxable years and may not be revoked without the consent of the makes a QEF election with respect to its interest in a PFIC, in lieu of the foregoing treatment, it would be required to include in incora a portion of the ordinary earnings and net capital gains of the QEF called "QEF Inclusions," even if not distributed to it. Thus, ho required to report taxable income as a result of QEF Inclusions without corresponding receipts of cash. However, a holder may e until the occurrence of certain events, payment of the U.S. federal income tax attributable to QEF Inclusions for which no current are received, but will be required to pay interest on the deferred tax computed by using the statutory rate of interest applicable to an time for payment of tax. KKR's tax basis in the shares of such non-U.S. entities, and a holder's basis in KKR common units, will be reflect QEF Inclusions. No portion of the QEF Inclusion attributable to ordinary income will be eligible for reduced rates of taxatic included as QEF Inclusions with respect to direct and indirect investments generally will not be taxed again when actually dist should consult your tax advisors as to the manner in which QEF Inclusions affect your allocable share of KKR's income and your KKR co

Alternatively, in the case of a PFIC that is a publicly traded foreign company, KKR may make an election to "mark to market" such foreign company on an annual basis. Purs

Table

an election, you would include in each year as ordinary income the excess, if any, of the fair market value of such stock over its adjuttee end of the taxable year. You may treat as ordinary loss any excess of the adjusted basis of the stock over its fair market value at the year, but only to the extent of the net amount previously included in income as a result of the election in

KKR may make certain investments, including for instance investments in specialized investment funds or investments in furthrough non-U.S. corporate subsidiaries of the KKR Group Partnerships or through other non-U.S. corporations. Such entities may U.S. federal income tax purposes. In addition, certain of KKR's investments could be in PFICs. Thus, KKR can make no assurance its investments will not be treated as held through a PFIC or as interests in PFICs or that such PFICs will be eligible for the "mar election, or that as to any such PFICs KKR will be able to make QI

If KKR does not make a QEF election with respect to a PFIC, Section 1291 of the Code will treat all gain on a disposition by KI of such entity, gain on the disposition of KKR common units by a holder at a time when KKR owns shares of such entity, as well as defined "excess distributions," as if the gain or excess distribution were ordinary income earned ratably over the shorter of the p which the holder held its KKR common units or the period during which KKR held shares in such entity. For gain and excess allocated to prior years, (i) the tax rate will be the highest in effect for that taxable year and (ii) the tax will be payable generally we to offsets from deductions, losses and expenses. Holders will also be subject to an interest charge for any deferred tax. No poordinary income will be eligible for the favorable tax rate applicable to "qualified dividend income" for individual U

Controlled Foreign (

A non-U.S. entity will be treated as a controlled foreign corporation, or CFC, if it is treated as a corporation for U.S. federa purposes and if more than 50% of (i) the total combined voting power of all classes of stock of the non-U.S. entity entitled to vote or value of the stock of the non-U.S. entity is owned by U.S. Shareholders on any day during the taxable year of such non-U.S. er purpose, a "U.S. Shareholder" with respect to a non-U.S. entity means a U.S. person (including a U.S. partnership like KKR) that of more of the total combined voting power of all classes of stock of the non-U.S. entity entity

When making investment or other decisions, KKR will consider whether an investment will be a CFC and the consequences rel If KKR is a U.S. Shareholder in a non-U.S. entity that is treated as a CFC, each KKR common unitholder may be required to includits allocable share of the CFC's "Subpart F" income reported by KKR. Subpart F income generally includes dividends, interest, net goale or disposition of securities, non-actively managed rents and certain other generally passive types of income. The aggregate Subpinclusions in any taxable year relating to a particular CFC are limited to such entity's current earnings and profits. These inclusions a ordinary income (whether or not such inclusions are attributable to net capital gains). Thus, an investor may be required to report income its allocable share of the CFC's Subpart F income reported by KKR without corresponding receipts of cash and may not capital gain treatment with respect to the portion of KKR's earnings (if any) attributable to net capital gains

The tax basis of KKR's shares of such non-U.S. entity, and your tax basis in your KKR common units, will be increased t required Subpart F income inclusions. Such income will be treated as income from sources within the United States, for certain forei purposes, to the extent derived by the CFC from U.S. sources. Such income will not be eligible for the reduced rate of tax applicable dividend income" for individual U.S. persons. See

Table

" Limitations on Interest Deductions." Amounts included as such income with respect to direct and indirect investments general taxable again when actually

Regardless of whether any CFC has Subpart F income, any gain allocated to you from KKR's disposition of stock in a CFC w as dividend income to the extent of your allocable share of the current and/or accumulated earnings and profits of the CFC, w eligible for the reduced rates of taxation applicable to certain qualified dividends. In this regard, earnings would not include previously taxed pursuant to the CFC rules. However, net losses (if any) of a non-U.S. entity owned by KKR that is treated as a pass through to you. Moreover, a portion of your gain from the sale or exchange of your KKR common units may be treated as ordin Any portion of any gain from the sale or exchange of a common unit that is attributable to a CFC may be treated as an "unrealized taxable as ordinary income. See " Sale or Exchange of KKR Co

If a non-U.S. entity held by KKR is classified as both a CFC and a PFIC during the time KKR is a U.S. Shareholder of such non-you will be required to include amounts in income with respect to such non-U.S. entity pursuant to this subheading, and the c described under " Passive Foreign Investment Companies" above will not apply. If KKR's ownership percentage in a non-U.S. such that KKR is not a U.S. Shareholder with respect to such non-U.S. entity, then you may be subject to the PFIC rules. The interact rules is complex, and prospective holders are urged to consult their tax advisors in

Investme

To manage its affairs so as to meet the Qualifying Income Exception for the publicly traded partnership rules (discussed above) with certain requirements in its partnership agreement, KKR may need to structure certain investments through entities classified as for U.S. federal income tax purposes. However, because KKR common unitholders will be located in numerous taxing juri assurances can be given that any such investment structure will be beneficial to all KKR common unitholders to the same extent, a impose additional tax burdens on some KKR common unitholders. As discussed above, if the entity were a non-U.S. corporati considered a CFC or PFIC. If the entity were a U.S. corporation, it would be subject to U.S. federal income tax on its opera including any gain recognized on its disposal of its investments. In addition, if the investment involves U.S. real estate, gain re disposition of the real estate would generally be subject to U.S. federal income tax, whether the corporation is a U.S. or a non-U.S.

Taxes in Other State, Local, and Non-U.S.

In addition to U.S. federal income tax consequences, you may be subject to potential U.S. state and local taxes because of an in KKR in the U.S. state or locality in which you are a resident for tax purposes or in which KKR has investments or activitic jurisdictions in which KKR holds certain real estate, oil, gas or similar natural resource-related investments. You may also be streturn filing obligations and income, franchise or other taxes, including withholding taxes, in state, local or non-U.S. jurisdictions in invests, or in which entities in which KKR owns interests conduct activities or derive income. Income or gains from investments have be subject to withholding or other taxes in jurisdictions outside the United States, subject to the possibility of reduction undincome tax treaties. If you wish to claim the benefit of an applicable income tax treaty, you may be required to submit inform authorities in such jurisdictions. You should consult your own tax advisors regarding the U.S. state, local and non-U.S. tax consequences.

<u>Table</u>

U.S. Federal

KKR common units will be included in the gross estate of a U.S. citizen or resident for U.S. federal estate tax purposes. Ther federal estate tax may be payable in connection with the death of a holder of KKR common units. Prospective individual U.S. Ho consult their own tax advisors concerning the potential U.S. federal estate tax consequences with respect to KKR co

M

U.S. Holders that are individuals, estates or trusts are subject to a Medicare tax of 3.8% on "net investment income" (or undis investment income," in the case of estates and trusts) for each taxable year, with such tax applying to the lesser of such income or such person's adjusted gross income (with certain adjustments) over a specified amount. Net investment income includes net interest, dividends, annuities, royalties and rents and net gain attributable to the disposition of investment property. It is anticip income and gain attributable to your ownership of KKR common units will be included in your "net investment income" subject to the

U.S. Taxation of Tax-Exempt U.S. Holders of KKR Con

A holder of KKR common units that is a tax-exempt organization for U.S. federal income tax purposes and therefore generally U.S. federal income taxation will nevertheless be subject to unrelated business taxable income, or UBTI, to the extent, if any, that share of KKR's income consists of UBTI. A tax-exempt partner of a partnership that regularly engages in a trade or business which to the exempt function of the tax-exempt partner must include in computing its UBTI its pro rata share (whether or not distributed partnership's gross income and deductions derived from such unrelated trade or business. Moreover, a tax-exempt partner of a partner treated as earning UBTI to the extent that such partnership derives income from "debt-financed property," or if the partner interest financed. Debt-financed property means property held to produce income with respect to which there is "acquisition indebted indebtedness incurred in acquiring or holding."

As a result of incurring acquisition indebtedness and certain investments in natural resource assets, such as oil and gas propertic derive income that constitutes UBTI. Consequently, a holder of KKR common units that is a tax-exempt organization (including a retirement account or 401(k) plan participant) will likely be subject to unrelated business income tax to the extent that its alloc KKR's income consists of UBTI. In addition, a tax-exempt partner may be subject to unrelated business income tax on a sale common units. Tax exempt U.S. Holders of KKR common units should consult their own tax advisors regarding all aspe

Investments by U.S. M

U.S. mutual funds that are treated as regulated investment companies, or RICs, for U.S. federal income tax purposes are requother things, to meet an annual 90% gross income and a quarterly 50% asset value test under Section 851(b) of the Code to make favorable U.S. federal income tax status. The 90% gross income test requires that, for a corporation to qualify as a RIC, at least such corporation's annual income must be "qualifying income," which is generally limited to investment income of various types. The value test requires that, for a corporation to qualify as a RIC, at the close of each quarter of the taxable year, at least 50 percent of such corporation's total assets must be represented by cash and cash items (including receivables), government securities, securities, and other securities limited

Table

any one issuer to an amount not greater in value than 5 percent of the value of the total assets of the corporation and to not more that of the outstanding voting securities of

The treatment of an investment by a RIC in KKR common units for purposes of these tests will depend on whether KKR is "qualifying publicly traded partnership." If KKR is so treated, then the KKR common units themselves are the relevant assets for purpose so the saset value test and the net income from the KKR common units is the relevant gross income for purposes of the 90% gross RICs may not invest greater than 25 percent of their assets in one or more qualifying publicly traded partnerships. All income dequalifying publicly traded partnership is considered qualifying income for purposes of the RIC 90% gross income test above. However, is not treated as a qualifying publicly traded partnership for purposes of the RIC rules, then the relevant assets for the RIC asset to RIC's allocable share of the underlying assets held by KKR and the relevant gross income for the RIC income test will be the RI share of the underlying gross income earned by KKR, including assets held in connection with and income derived with respect to in natural resources assets, such as oil and gas properties, which may not be qualifying assets or income for the RIC qualifying asset tests above. Whether KKR will qualify as a "qualifying publicly traded partnership" depends on the exact nature of its future investing likely that KKR will not be treated as a "qualifying publicly traded partnership." In addition, as discussed above under "Conseq Holders of KKR Common Units," KKR may derive taxable income from an investment that is not matched by a corresponding cash Accordingly, a RIC investing in KKR common units may recognize income for U.S. federal income tax purposes without receive which to make distributions in amounts necessary to satisfy the distribution requirements under Sections 852 and 4982 of the Code income and excise taxes. RICs should consult their own tax advisors about the U.S. tax consequences of an investment in KKR co

Consequences to Non-U.S. Holders of KKR Con

U.S. Income Tax Co

KKR expects that it will be engaged in a U.S. trade or business for U.S. federal income tax purposes, including by reason of its in U.S. real property, corporations that own significant amounts of U.S. real property, and oil and gas properties, in which case son its income would be treated as effectively connected income with respect to Non-U.S. Holders, or ECI. If a Non-U.S. Holder we being engaged in a U.S. trade or business in any year because of an investment in KKR common units in such year, such Nongenerally would be: (1) subject to withholding by KKR on such Non-U.S. Holder's distributions of ECI; (2) required to file a U.S. fed tax return for such year reporting its allocable share, if any, of income or loss effectively connected with such trade or business, inclusions from U.S. sources not related to KKR.; and (3) required to pay U.S. federal income tax at regular U.S. federal income tax such income. Moreover, a corporate Non-U.S. Holder might be subject to a U.S. branch profits tax on its allocable share of its ECI. withheld would be creditable against such Non-U.S. Holder's U.S. federal income tax liability, and such Non-U.S. Holder could clair the extent that the amount withheld exceeded such Non-U.S. Holder's U.S. federal income tax liability for the taxable year. Finally, it treated as being engaged in a U.S. trade or business, a portion of any gain recognized by a holder who is a Non-U.S. Holder exchange of its KKR common units could be treated for U.S. federal income tax purposes as ECI, and hence such Non-U.S. Holder to U.S. federal income tax on the sale or exchange of its KKR common units could be treated for U.S. federal income tax on the sale or exchange of its KKR common units could be treated for U.S. federal income tax on the sale or exchange of its KKR common units could be treated for U.S. federal income tax on the sale or exchange of its KKR common units could be treated for U.S. federal income tax on the sale or exchange of its KKR common units could be treated for U.S. federal incom

Distributions to you may also be subject to U.S. withholding tax to the extent such distribution is attributable to the sale of property interest. Also, you may be subject to U.S. withholding tax on allocations of KKR's income that are fixed or determinal periodic income under the Code, unless an exemption from or a reduced rate of such withholding applies and certain

Table

information is provided. Although each Non-U.S. Holder is required to provide an IRS Form W-8, KKR may not be able to provide information related to the tax status of its investors to the KKR Group Partnerships or KKR Management Holdings Corp. for obtaining reduced rates of withholding on behalf of its investors. If such information is not provided, to the extent KKR receives dis KKR Management Holdings Corp. or from a U.S. corporation through Fund Holdings and its investment vehicles, your alloc distributions of such income will be subject to U.S. withholding tax. Therefore, if you would not be subject to U.S. tax based on your are eligible for a reduced rate of U.S. withholding, you may need to take additional steps to receive a credit or refund to withholding tax paid on your account. This may include the filing of a non-resident U.S. income tax return with the IRS. A limitations, if you reside in a treaty jurisdiction which does not treat KKR as a pass-through entity, you may not be eligible to receive credit of excess U.S. withholding taxes paid on your account. You should consult your tax advisors regarding the treatment of U.S.

Special rules may apply in the case of a Non-U.S. Holder that: (1) has an office or fixed place of business in the United States; (a in the United States for 183 days or more in a taxable year; or (3) is a former citizen of the United States, a foreign insurance contreated as holding a partner interest in KKR in connection with their U.S. business, a PFIC or a corporation that accumulates earn U.S. federal income tax. You should consult your tax advisors regarding the application of these

U.S. Federal Estate Tax Co

The U.S. federal estate tax treatment of KKR common units with regard to the estate of a non-citizen who is not a resident of States is not entirely clear. If KKR common units are includable in the U.S. gross estate of such person, then a U.S. federal estate payable in connection with the death of such person. Non-U.S. Holders who are non-citizens and not residents of the United Scientific consult their own tax advisors concerning the potential U.S. federal estate tax consequences of owning KKR co

Administra

KKR currently uses the calendar year as its taxable year for U.S. federal income tax purposes. Under certain circumstances currently believes are unlikely to apply, a taxable year other than the calendar year may be required for su

Tax Ma

KKR's managing partner will act as its "tax matters partner." As the tax matters partner, KKR's managing partner will have t subject to certain restrictions, to act on KKR's behalf in connection with any administrative or judicial review of KKR's items of it loss, deduct

Informa

KKR has agreed to furnish to you, as soon as reasonably practicable after the close of each calendar year, tax informatic Schedule K-1), which describes on a U.S. dollar basis your share of KKR's income, gain, loss and deduction for its preceding tax may require longer than 90 days after the end of KKR's fiscal year to obtain the requisite information from all lower-tier entities may be prepared for KKR. Consequently, common unitholders who are U.S. taxpayers should anticipate the need to file annually (and certain states) a request for an extension past April 15 or the otherwise applicable due date of their income tax return for the tax addition, each partner will be required to report for all tax purposes consistently with the information provided by KKR for the

Table

In preparing this information, KKR will use various accounting and reporting conventions, some of which have been men previous discussion, to determine your share of income, gain, loss and deduction. The IRS may successfully contend that cer reporting conventions are impermissible, which could result in an adjustment to your income.

KKR may be audited by the IRS. Adjustments resulting from an IRS audit may require you to adjust a prior year's tax liability may result in an audit of your own tax return. Any audit of your tax return could result in adjustments not related to KKR's tax return those related to KKR's tax return.

Tax Shelter

If KKR were to engage in a "reportable transaction," KKR (and possibly you and others) would be required to make a detailed the transaction to the IRS in accordance with regulations governing tax shelters and other potentially tax-motivated transactions. A may be a reportable transaction based upon any of several factors, including the fact that it is a type of tax avoidance transactiontified by the IRS as a "listed transaction" or that it produces certain kinds of losses in excess of \$2 million. An investment in I considered a "reportable transaction" if, for example, KKR recognizes certain significant losses in the future. In certain circumstance unitholder who disposes of KKR common units in a transaction resulting in the recognition by such holder of significant losses certain threshold amounts may be obligated to disclose its participation in such transaction. KKR's participation in a reportable transaction dincrease the likelihood that its U.S. federal income tax information return (and possibly your tax return) would be audited. Certain of these rules are currently unclear and it is possible that they may be applicable in situations other than significant loss. Moreover, if KKR were to participate in a reportable transaction with a significant purpose to avoid or evade tax, or in any listed transaction with a broad scope; (ii) for those persons otherwise entitled to deduct federal tax deficiencies, nondeductibility of interest on any resulting tax liability; and (iii) in the case of a listed transaction, an extension of limitations. Common unitholders should consult their tax advisors concerning any possible disclosure obligation under the governing tax shelters with respect to the dispositions of their interest.

Constructive

Subject to the electing large partnership rules described below, KKR will be considered to have been terminated for U.S. federa purposes if there is a sale or exchange of 50% or more of the total interests in KKR's capital and profits within a 12-month per termination would result in the close of its taxable year for all of KKR's common unitholders. In the case of a holder reporting on a other than a fiscal year ending on KKR's year-end, the closing of KKR's taxable year may result in more than 12 months of K income or loss being includable in the holder's taxable income for the year of termination. KKR would be required to make new after a termination. A termination could also result in penalties if KKR were unable to determine that the termination had occurred. termination might either accelerate the application of, or subject KKR to, any tax legislation enacted before the

Elective Procedures for Large

The Code allows large partnerships to elect streamlined procedures for income tax reporting. This election would reduce the items that must be separately stated on the Schedules K-1 that are issued to KKR common unitholders, and such Schedules K-1 would provided to KKR common unitholders on or before the first March 15 following the close of each taxable year. In addition, this election would reduce the provided to KKR from suffering a "technical termination" (which

Table

KKR's taxable year) if within a 12-month period there is a sale or exchange of 50 percent or more of KKR's total interests. It is put might make such an election, if eligible. If KKR makes such election, IRS audit adjustments will flow through to KKR common until the years in which the adjustments take effect, rather than the year to which the adjustment relates. In addition, KKR, rather than unitholders individually, generally will be liable for any interest and penalties that result from an audit

Withholding and Backup

For each calendar year, KKR will report to you and the IRS the amount of distributions KKR made to you and the amount of income tax (if any) that KKR withheld on those distributions. The proper application to KKR of rules for withholding under Section Code (applicable to certain dividends, interest and similar items) is unclear. Because the documentation KKR receives may not properly allocable to a particular holder of KKR common units. For example, KKR may impose withholding, remit that amount to the reduce the amount of a distribution paid to a Non-U.S. Holder. It may turn out, however, the corresponding amount of KKR's incorporal allocable to such holder, and the withholding should have been less than the actual withholding. Such holder would be credit against the holder's U.S. federal income tax liability for all withholding, including any such excess withholding, but if the exceeded the holder's U.S. federal income tax liability, the holder would have to apply for a refund to obtain the benefit of withholding. Similarly, KKR may fail to withhold on a distribution, and it may turn out the corresponding income was properly a Non-U.S. Holder and withholding should have been imposed. In that event, KKR intends to pay the underwithheld amount to the IR may treat such under-withholding as an expense that will be borne by all partners on a pro rata basis (since KKR may be unable to such excess withholding tax cost to the relevant Non-U.S.)

Under the backup withholding rules, you may be subject to backup withholding tax (at the applicable rate, currently 28%) we distributions paid unless: (i) you are an exempt recipient and demonstrate this fact when required; or (ii) you provide a taxpayer in number, certify as to no loss of exemption from backup withholding tax and otherwise comply with the applicable requirements of withholding tax rules. If you are an exempt holder, you should indicate your exempt status on a properly completed IRS Form W-9. Holder may qualify as an exempt recipient by submitting a properly completed applicable IRS Form W-8. Backup withhold additional tax. The amount of any backup withholding from a payment to you will be allowed as a credit against your U.S. federal liability and may entitle you

If you do not timely provide KKR (or the clearing agent or other intermediary, as appropriate) with IRS Form W-8 or W-9, as a such form is not properly completed, you may become subject to U.S. backup withholding taxes in excess of what would have been KKR received certifications from all investors. Such excess U.S. backup withholding taxes may be treated by KKR as an expense borne by all investors on a pro rata basis (since KKR may be unable to allocate any such excess withholding tax cost to the holders timely provide the proper U.S. tax ce

Additional Withholding R

Under recently enacted legislation and administrative guidance, the relevant withholding agent may be required to withhold interest, dividends and other fixed or determinable annual or periodical gains, profits and income from sources within the United State June 30, 2014 or gross proceeds from the sale of any property of a type that can produce interest or dividends from sources within States occurring after December 31, 2016 to (1) a foreign financi

<u>Table</u>

(which for this purpose includes foreign broker-dealers, clearing organizations, investment companies, hedge funds and certain other entities) unless such foreign financial institution enters into an agreement with the Treasury pursuant to which it agrees to verify disclose its U.S. accountholders to the IRS and complies with certain other specified requirements or (2) a non-financial foreign entities to the payment unless such entity (a) certifies that it does not have any substantial U.S. owners, (b) provides the natural and taxpayer identification number of each of its substantial U.S. owners and meets certain other specified requirements or (4) qualifies for an exemption from this withholding. Non-U.S. and U.S. Holders are encouraged to consult their own tax advisors in possible implications of this proposed legislation on their investment in KKR contributions.

Nomine

Persons who hold an interest in KKR as a nominee for another person are required to furn

the name, address and taxpayer identification number of the beneficial owner and

whether the beneficial owner is: (i) a person that is not a U.S. person; (ii) a foreign government, an international or any wholly owned agency or instrumentality of either of the foregoing; or (iii) a tax-ex-

the amount and description of KKR common units held, acquired or transferred for the beneficia

specific information including the dates of acquisitions and transfers, means of acquisitions and transfers and acc for purchases, as well as the amount of net proceed

Brokers and financial institutions are required to furnish additional information, including whether they are U.S. persons information on KKR common units they acquire, hold or transfer for their own account. A penalty of \$50 per failure, up to a \$100,000 per calendar year, is imposed by the Code for failure to report that information to KKR. The nominee is required beneficial owner of the KKR common units with the information furnis

New Legislation or Administrative or Jud

The rules dealing with U.S. federal income taxation are constantly under review by persons involved in the legislative process, the Treasury, frequently resulting in revised interpretations of established concepts, statutory changes, revisions to regulation modifications and interpretations. No assurance can be given as to whether, or in what form, any proposals affecting KKR or KK unitholders will be enacted. The present U.S. federal income tax treatment of an investment in KKR common units may be administrative, legislative or judicial interpretation at any time, and any such action may affect investments and commitments prevently considerations of an investment tax laws and interpretations thereof could make it more difficult or impossible to be treated as a that is not taxable as a corporation for U.S. federal income tax purposes, affect or cause KKR to change its investments and committed the tax considerations of an investment in KKR, change the character or treatment of portions of KKR's income (including, for treatment of carried interest as ordinary income rather than capital gain) and adversely affect an investment in KKR common undescribed in the sections entitled "Risk Factors" Risks Related to the Ownership of KKR Common Units KKR's structure in provisions of U.S. federal income tax laws for which no clear precedent or authority may be available. These structures also a potential potential potential contents and committed in the sections and the potential contents are constantly and adversely affect an investment in KKR common undescribed in the sections entitled "Risk Factors" Risks Related to the Ownership of KKR Common Units KKR's structure in provisions of U.S. federal income tax laws for which no clear precedent or authority may be available. These structures also a potential content and the content and t

Table

judicial or administrative change and differing interpretations, possibly on a retroactive basis," and "Risk Factors Risks Related to to of KKR Common Units The U.S. Congress has considered legislation that would have (i) in some cases after a ten-year period, put from qualifying as a partnership or required KKR to hold carried interest through taxable subsidiary corporations and (ii) taxed ce and gains at increased rates. If any similar legislation were to be enacted and apply to KKR, the after tax income and gain related business, as well as the market price of KKR common units, could be reduced." Additionally, on February 26, 2014, Representative Chairman of the House Ways and Means Committee, released a discussion draft summarizing proposed legislation that would, things (1) generally treat publicly traded partnerships (other than those deriving 90 percent of their income from activities relating to natural resources) as taxable corporations for tax years beginning after 2016 and (2) recharacterize a portion of capital gain partnership interests held in connection with the performance of services as ordinary income for tax years beginning after 2014. KK common unitholders could be adversely affected by any such change in, or any new, tax law, regulation or interpretation. KKR's or documents and agreements permit the board of directors to modify the amended and restated operating agreement from time to time consent of the common unitholders, in order to address certain changes in U.S. federal income tax regulations, legislation or interpretation some circumstances, such revisions could have a material adverse impact on some or all of KKR's common

THE FOREGOING DISCUSSION IS NOT INTENDED AS A SUBSTITUTE FOR CAREFUL TAX PLANNING. THE TAX RELATING TO KKR AND ITS COMMON UNITHOLDERS ARE COMPLEX AND ARE SUBJECT TO VARYING INTERPR. MOREOVER, THE MEANING AND IMPACT OF TAX LAWS AND OF PROPOSED CHANGES WILL VARY WITH THE PACIFIC CONSTRUCT OF EACH KKR COMMON UNITHOLDER. KKR COMMON UNITHOLDERS SHOULD CONSULT ADVISORS WITH RESPECT TO THE U.S. FEDERAL, STATE, LOCAL, FOREIGN AND OTHER TAX CONSEQUENCES TO OWNING KKR COMMON UNITS. THIS FOREGOING DISCUSSION ONLY ADDRESSES THE MATERIAL U.S. FEDERAL CONSIDERATIONS OF THE OWNERSHIP AND DISPOSITION OF KKR COMMON UNITS AND DOES NOT ADDRESSED CONSEQUENCES UNDER THE LAWS OF ANY TAX JURISDICTION OTHER THAN THE UNITED STATES. NON-U.S. THEREFORE, SHOULD CONSULT THEIR OWN TAX ADVISORS REGARDING THE TAX CONSIDERATIONS TO THIS U.S. OWNERSHIP AND DISPOSITION OF KKR COMMON UNITS UNDER THE LAWS OF THEIR OWN TAXING JURISDICTION OF THE LAWS OF THE

<u>Table</u>

DESCRIPTION OF KKR'S LIMITED PARTNERSHIP AGREEMENT

The following is a description of the material terms of KKR's partnership agreement and is qualified in its entirety by reference provisions of KKR's partnership agreement, which has been filed as an exhibit to the registration statement of which statement/prospectus forms a part. Because this description is only a summary of the terms of KKR's partnership agreement, it may all of the information that you may find important. For additional information, you should read "Description of KKR Common Units U.S. Federal Income Tax Consequences of the Merger" and "Material U.S. Federal Income Tax Consequences of KKR Common Units U.S. Federal Income Tax Consequences of KKR Common Units U.S. Federal Income Tax Consequences of KKR Common Units U.S. Federal Income Tax Consequences of KKR Common Units U.S. Federal Income Tax Consequences of KKR Common Units U.S. Federal Income Tax Consequences of KKR Common Units U.S. Federal Income Tax Consequences of KKR Common Units U.S. Federal Income Tax Consequences of KKR Common Units U.S. Federal Income Tax Consequences of KKR Common Units U.S. Federal Income Tax Consequences Of KKR Common Units U.S. Federal Income Tax Consequences Of KKR Common Units U.S. Federal Income Tax Consequences Of KKR Common Units U.S. Federal Income Tax Consequences Of KKR Common Units U.S. Federal U.S. Federal Income Tax Consequences Of KKR Common Units U.S. Federal U.S. Federal

KKR's Manag

KKR's managing partner, KKR Management LLC, is the general partner of KKR that manages all of KKR's operations a KKR's managing partner is authorized in general to perform all acts that it determines to be necessary or appropriate to carry out KK and to conduct KKR's business. KKR's managing partner is wholly owned by KKR's principals and controlled by KKR's for common unitholders have only limited voting rights relating to certain matters and, therefore, will have limited or no ability management's decisions regarding KK

Under KKR's partnership agreement, KKR is permitted to engage, directly or indirectly, in any business activity that is approve managing partner and that lawfully may be conducted by a limited partnership organized under D

Power

Each limited partner of KKR, and each person who acquires a limited partner interest in accordance with KKR's partnership grants to KKR's managing partner and, if appointed, a liquidator, a power of attorney to, among other things, execute and fil required for KKR's qualification, continuance, dissolution or termination. The power of attorney also grants KKR's managin authority to amend, and to make consents and waivers under, KKR's partnership agreement and certificate of limited partnership, in accordance with KKR's partnership

Capital Co

KKR's common unitholders are not obligated to make additional capital contributions, except as described below under "Limitation Limitation Lim

Limi

Assuming that a limited partner of KKR does not participate in the control of KKR's business within the meaning of the Delaw Partnership Act and that person otherwise acts in conformity with the provisions of KKR's partnership agreement, such person's li the Delaware Limited Partnership Act would be limited, subject to possible exceptions, to the amount of capital such person is contribute to KKR for such person's common units plus such person's share of any undistributed profits and assets. If it were determine that the right, or exercise of the right, by the limited partner

to approve some amendments to KKR's partnership ag

to take other action under KKR's partnership agreement, constituted "participation in the control" of KKR's bus purposes of the Delaware Limited Partnersh

Table

KKR's limited partners could be held personally liable for KKR's obligations under the laws of Delaware to the as KKR's managing partner. This liability would extend to persons who transact business with KKR who reason that the limited partner is a general partner. Neither KKR's partnership agreement nor the Delaware Limited Part specifically will provide for legal recourse against KKR's managing partner if a limited partner were to lose lim through any fault of KKR's managing partner. While this does not mean that a limited partner could not seek legal KKR knows of no precedent for this type of a claim in Delaware case law. The limitation on KKR's managiliability does not constitute a waiver of compliance with U.S. federal securities laws that would be void under State of the securities and the securities laws that would be void under State of t

Under the Delaware Limited Partnership Act, a limited partnership may not make a distribution to a partner if, after the dis liabilities of the limited partnership, other than liabilities to partners on account of their partner interests and liabilities for which the creditors is limited to specific property of the partnership, would exceed the fair value of the assets of the limited partnership. For the determining the fair value of the assets of a limited partnership, the Delaware Limited Partnership Act provides that the fair value subject to liability for which recourse of creditors is limited will be included in the assets of the limited partnership only to the extensivalue of that property exceeds the non-recourse liability. The Delaware Limited Partnership Act provides that a limited partner who distribution and knew at the time of the distribution that the distribution was in violation of the Delaware Limited Partnership Act, limited partnership for the amount of the distribution for three years. Under the Delaware Limited Partnership, excepts on is not obligated for liabilities unknown to him at the time he became a limited partner and that could not be ascertained from partnership.

Moreover, if it were determined that KKR were conducting business in any state without compliance with the applicable limited statute, or that the right or exercise of the right by the limited partners as a group to approve some amendments to the limited agreement or to take other action under the limited partnership agreement constituted "participation in the control" of KKR's purposes of the statutes of any relevant jurisdiction, then the limited partners could be held personally liable for KKR's obligations u of that jurisdiction to the same extent as KKR's managing partner. KKR intends to operate in a manner that KKR's managing partner reasonable and necessary or appropriate to preserve the limited liability of the limited

Issuance of Addition

KKR's partnership agreement authorizes KKR to issue an unlimited number of additional partnership securities and options, rig and appreciation rights relating to partnership securities for the consideration and on the terms and conditions established by KKB partner in its sole discretion without the approval of any lim

In accordance with the Delaware Limited Partnership Act and the provisions of KKR's partnership agreement, KKR counadditional partnership interests that have designations, preferences, rights, powers and duties that are different from, and may be seen applicable to its co

202

Distributions will be made to the partners pro rata according to the percentages of their respective partner interests. See "M

Discussion and Analysis of Financial Condition at

Table

Operations Liquidity Liquidity Needs Distributions," in KKR's Quarterly Report on Form 10-Q for the nine-month period endo 2013 filed with the SEC on November 1, 2013 and incorporated by reference in this proxy statemen

Amendment of the Limited Partnership

Amendments to KKR's partnership agreement may be proposed only by KKR's managing partner. To adopt a proposed amen than the amendments that do not require limited partner approval discussed below, KKR's managing partner must seek approval of the a majority of the outstanding voting units (as defined below) in order to approve the amendment or call a meeting of the limite consider and vote upon the proposed amendment. On any matter that may be submitted for a vote of KKR common unitholders, the KKR Group Partnership units hold special voting units in KKR's partnership that provide them with a number of votes that is aggregate number of KKR Group Partnership units that they then hold and entitle them to participate in the vote on the same basis unitholders of KKR's partnership. See "Meetings; Voting." The KKR Group Partnership units, other than the KKR Group Partnership by KKR, are owned by KKR Holdings, which is owned by KKR's principals and other persons and is controlled by KKR

Prohibited A

No amendment may be made

enlarge the obligations of any limited partner without its consent, except that any amendment that would have adverse effect on the rights or preferences of any class of partner interests in relation to other classes of partner is be approved by the holders of at least a majority of the type or class of partner interests so

enlarge the obligations of, restrict in any way any action by or rights of, or reduce in any way the amounts or reimbursable or otherwise payable by KKR, to KKR's managing partner or any of its affiliates without the consequence of the managing partner, which may be given or withheld in its solution.

The provision of the KKR partnership agreement preventing the amendments having the effects described in clauses (1) or (2) amended upon the approval of the holders of at least 90% of the outstanding

No Limited Partn

KKR's managing partner may generally make amendments to KKR's partnership agreement or certificate of limited partnership approval of any limited partnership.

a change in the name of the partnership, the location of the partnership's principal place of business, the registered agent or its regis

the admission, substitution, withdrawal or removal of partners in accordance with KKR's partnershi

a change that KKR's managing partner determines is necessary or appropriate for KKR to qualify or to qualification as a limited partnership or a partnership in which the limited partners have limited liability unde any state or other jurisdiction or to ensure that KKR will not be treated as an association taxable as a corporation taxed as an entity for U.S. federal income to

Table

an amendment that KKR's managing partner determines to be necessary or appropriate to address certain cha federal, state and local income tax regulations, legislation or in

an amendment that is necessary, in the opinion of KKR's counsel, to prevent KKR or KKR's managing directors, officers, employees, agents or trustees, from having a material risk of being in any manner sub provisions of the Investment Company Act, the Investment Advisers Act of 1940, as amended, or "plan asset adopted under Employee Retirement Income Security Act of 1974, as amended, whether or not substantially si asset regulations currently applied or proposed by the U.S. Department.

a change in KKR's fiscal year or taxable year and rela

an amendment that KKR's managing partner determines in its sole discretion to be necessary or appropriate for authorization or issuance of any class or series of partnership securities or options, rights, warrants or appreciating to partnersh

any amendment expressly permitted in KKR's partnership agreement to be made by KKR's managing partner

an amendment effected, necessitated or contemplated by an agreement of merger, consolidation or o combination agreement that has been approved under the terms of KKR's partnershi

an amendment effected, necessitated or contemplated by an amendment to the partnership agreement of a Partnership that requires unitholders of the KKR Group Partnership to provide a statement, certification or o evidence regarding whether such unitholder is subject to U.S. federal income taxation on the income generated

any amendment that in the sole discretion of KKR's managing partner is necessary or appropriate to reflect and the formation by KKR of, or its investment in, any corporation, partnership, joint venture, limited liability compentity, as otherwise permitted by KKR's partnership

a merger, conversion or conveyance to another limited liability entity that is newly formed and has no assets, operations at the time of the merger, conversion or conveyance other than those it receives by way of the merger

any amendment that KKR's managing partner determines to be necessary or appropriate to cure any ambigui mistake, defect or inco

any other amendments substantially similar to any of the matters described in (1) through

In addition, KKR's managing partner could make amendments to KKR's partnership agreement without the approval of any line if those amendments, in the discretion of KKR's mana

do not adversely affect KKR's limited partners considered as a whole (or adversely affect any particular class of partner interests) in any mat

are necessary or appropriate to satisfy any requirements, conditions or guidelines contained in any opinion, directly ruling or regulation of any federal, state, local

Table

agency or judicial authority or contained in any federal, state, local or non-U.S. statute (including the Delay

are necessary or appropriate to facilitate the trading of limited partner interests or to comply with any rule guideline or requirement of any securities exchange on which the limited partner interests are or will be listed

are necessary or appropriate for any action taken by KKR's managing partner relating to splits or combinations of the provisions of KKR's partnership a

are required to effect the intent of the provisions of KKR's partnership agreement or are otherwise contemplat partnershi

Opinion of Counsel and Limited Partn

KKR's managing partner will not be required to obtain an opinion of counsel that an amendment will not result in a loss of lim to the limited partners if one of the amendments described above under "No Limited Partner Approval" should occur. No other a KKR's partnership agreement (other than an amendment pursuant to a merger, sale or other disposition of assets effected in accordary provisions described under "Merger, Sale or Other Disposition of Assets" or an amendment described in the following paragraphs effective without the approval of holders of at least 90% of the outstanding voting units, unless KKR obtains an opinion of counsel that the amendment will not affect the limited liability under the Delaware Limited Partnership Act of any of the limited

In addition to the above restrictions, any amendment that would have a material adverse effect on the rights or preferences o class of partner interests in relation to other classes of partner interests will also require the approval of the holders of at least a material adverse effect on the rights or preferences o class of partner interests in relation to other classes of partner interests will also require the approval of the holders of at least a material adverse effect on the rights or preferences o class of partner interests in relation to other classes of partner interests will also require the approval of the holders of at least a material adverse effect on the rights or preferences or class of partner interests will also require the approval of the holders of at least a material adverse effect on the rights or preferences or class of partner interests will also require the approval of the holders of at least a material adverse effect on the rights or preferences or class of partner interests will also require the approval of the holders of at least a material adverse effect on the rights of the class of the cl

In addition, any amendment that reduces the voting percentage required to take any action must be approved by the affirm limited partners whose aggregate outstanding voting units constitute not less than the voting requirement sought to

Merger, Sale or Other Dispositi

KKR's partnership agreement provides that KKR's managing partner may, with the approval of the holders of at least a magnetic outstanding voting units, sell, exchange or otherwise dispose of all or substantially all of KKR's assets in a single transaction or a seri transactions, including by way of merger, consolidation or other combination, or approve the sale, exchange or other disposis substantially all of the assets of KKR's subsidiaries. KKR's managing partner in its sole discretion may mortgage, pledge, hypotheca security interest in all or substantially all of KKR's assets (including for the benefit of persons other than KKR or its subsidiaries prior approval of the holders of KKR's outstanding voting units. KKR's managing partner could also sell all or substantially all of lunder any forced sale of any or all of KKR's assets pursuant to the foreclosure or other realization upon those encumbrances with approval of the holders of KKR's outstanding

If conditions specified in KKR's partnership agreement are satisfied, KKR's managing partner may in its sole discretion conv KKR or any of its subsidiaries into, or convey some or all of its assets to, a newly formed entity if the sole purpose of that merger or is to effect a mere change in its legal form into another limited liability entity. KKR's unitholders will not be entitled to dissent appraisal under KKR's partnership agreement or the Delaware Limited Partnership Act in the event of a merger or consolidat substantially all of KKR's assets or any other similar transact

Table

Election to be Treated as a

If KKR's managing partner, in its sole discretion, determines that it is no longer in KKR's interests to continue as a partner federal income tax purposes, KKR's managing partner may elect to treat KKR as an association or as a publicly traded partnership corporation for U.S. federal (and applicable state) income tax purposes or may choose to effect such change by merger, conversion of the conversion of t

KKR will di

the election of KKR's managing partner to dissolve KKR, if approved by the holders of a majority of the vot KKR's outstanding

there being no limited partners, unless KKR is continued without dissolution in accordance with the Delay

the entry of a decree of judicial dissolution of KKR pursuant to the Delaware Limited Partne

the withdrawal of KKR's managing partner or any other event that results in its ceasing to be KKR's managing than by reason of a transfer of general partner interests or withdrawal of KKR's managing partner following admission of a successor, in each case in accordance with KKR's partnershi

Upon a dissolution under clause (4), the holders of a majority of the voting power of KKR's outstanding voting units could also specific time limitations, to continue KKR's business without dissolution on the same terms and conditions described in KKR's agreement by appointing as a successor managing partner an individual or entity approved by the holders of a majority of the voting outstanding voting units, subject to KKR's receipt of an opinion of counsel to the effect that (i) the action would not result in the lo liability of any limited partner and (ii) neither KKR nor any of its subsidiaries (excluding those formed or existing as corporation treated as an association taxable as a corporation or otherwise be taxable as an entity for U.S. federal income tax purposes upon the that right

Liquidation and Distribution

Upon KKR's dissolution, KKR's managing partner shall act, or select one or more persons to act, as liquidator. Unless KKR is a limited partnership, the liquidator authorized to wind up KKR's affairs will, acting with all of the powers of KKR's managing pa liquidator deems necessary or appropriate in its judgment, liquidate KKR's assets and apply the proceeds of the liquidation first, KKR's liabilities as provided in KKR's partnership agreement and by law, and thereafter, to the limited partners pro rata acc percentages of their respective partner interests as of a record date selected by the liquidator. The liquidator may defer liquidati assets for a reasonable period of time or distribute assets to partners in kind if it determines that an immediate sale or distribution of KKR's assets would be impractical or would cause undue loss to

Withdrawal of KKR's Manag

Except as described below, KKR's managing partner will agree not to withdraw voluntarily as KKR's managing pa December 31, 2020 without obtaining the approval of the holders of at least a majority of the outstanding voting units, excluding held by KKR's managing partner and its affiliates, and furnishing an opinion of counsel regarding tax and lim

Table

matters. On or after December 31, 2020, KKR's managing partner may withdraw as managing partner without first obtaining approximation common unitholder by giving 90 days' advance notice, and that withdrawal will not constitute a violation of KKR's partnership Notwithstanding the foregoing, KKR's managing partner could withdraw at any time without unitholder approval upon 90 days' ad to the limited partners if at least 50% of the outstanding KKR common units are beneficially owned, owned of record or otherwise one person and its affiliates other than KKR's managing partner and

Upon the withdrawal of KKR's managing partner under any circumstances, the holders of a majority of the voting pow outstanding voting units may elect a successor to that withdrawing managing partner. If a successor is not elected, or is elected but a counsel regarding limited liability and tax matters cannot be obtained, KKR will be dissolved, wound up and liquidated, unless with time limitations after that withdrawal, the holders of a majority of the voting power of KKR's outstanding voting units agree continue KKR's business and to appoint a successor managing partner. See "Disso

KKR's managing partner may not be removed or expelled, with or without cause, by

In the event of withdrawal of a managing partner, the departing managing partner will have the option to require the success partner to purchase the general partner interest of the departing managing partner for a cash payment equal to its fair market value will be determined by agreement between the departing managing partner and the successor managing partner. If no reached within 30 days of a managing partner's departure, an independent investment banking firm or other independent expert, we may rely on other experts, selected by the departing managing partner and the successor managing partner will determine the fair managing managing partner and the successor managing partner cannot agree upon an expert within 45 days of such managing partner, then an expert chosen by agreement of the experts selected by each of them will determine the fair no

If the option described above is not exercised by either the departing managing partner or the successor managing partner, to managing partner's general partner interest will automatically convert into KKR common units pursuant to a valuation of those determined by an investment banking firm or other independent expert selected in the manner described in the preceding

In addition, KKR will be required to reimburse the departing managing partner for all amounts due the departing mana including without limitation all employee-related liabilities, including severance liabilities, incurred for the termination of an employed by the departing managing partner or its affiliates for KI

Transfer of General Partn

Except for transfer by KKR's managing partner of all, but not less than all, of its general partner interests in KKR to an affilia managing partner, or to another entity as part of the merger or consolidation of KKR's managing partner with or into another entity of by KKR's managing partner of all or substantially all of its assets to another entity, KKR's managing partner may not transfer all or a general partner interest in KKR to another person prior to December 31, 2018 without the approval of the holders of at least a material power of KKR's outstanding voting units, excluding voting units held by KKR's managing partner and its affiliatest December 31, 2018, KKR's managing partner may transfer all or any part of its general partner interest without first obtaining appropriate to a condition of this transfer, the transferee must assume the rights and duties of KKR's managing partner to whose transferee has succeeded, agree to be bound by the provisions of KKR's partnership agreement and furnish an opinion of counse limited liability matters. At any time, the members of KKR's managing partner may sell or transfer all or part of their limited liability interests in KKR's managing partner without the approval of the KKR common

<u>Table</u>

Limite

If

less than 10% of the then issued and outstanding limited partner interests of any class (other than special vincluding KKR's limited partnership units, are held by persons other than KKR's managing partner and its

KKR is subjected to registration under the provisions of the Investment C

KKR's managing partner will have the right, which it may assign in whole or in part to any of its affiliates or to KKR, to acquir less than all, of the remaining limited partner interests of the class held by unaffiliated persons as of a record date to be select managing partner, on at least ten but not more than 60 days' notice. The purchase price in the event of this purchase is the

the current market price as of the date three days before the date the notice is

the highest cash price paid by KKR's managing partner or any of its affiliates acting in concert with KKR fo partner interests of the class purchased within the 90 days preceding the date on which KKR's managing partn notice of its election to purchase those limited part

As a result of KKR's managing partner's right to purchase outstanding limited partner interests, a holder of limited partner in have such interests purchased at an undesirable time or price. The U.S. tax consequences to a common unitholder of the exercise of are the same as a sale by that unitholder of its limited partnership units in the market. See "Material U.S. Federal Income Tax Con the Merger" beginning on page 172 of this proxy statemen

Sinking Fund; Preem

KKR will not establish a sinking fund and will not grant any preemptive rights with respect to KKR's limited part

Meet

Except as described below regarding a person or group owning 20% or more of KKR's limited partnership units then outstar holders of limited partnership units or of the special voting units to be issued to holders of KKR Group Partnership units on the rec be entitled to notice of, and to vote at, meetings of KKR's limited partners and to act upon matters as to which holders of limited partners to which holders of limited partners are to which holders of limited partnership units or the received pa

Except as described below regarding a person or group owning 20% or more of KKR's limited partnership units then outst record holder of a KKR common unit will be entitled to a number of votes equal to the number of limited partnership units held KKR issued special voting units to each holder of KKR Group Partnership units that provide them with a number of votes that is aggregate number of KKR Group Partnership units that they hold and entitle them to participate in the vote on the same basis as KKR refers to its common units and special voting units as "voting units." If the ratio at which KKR Group Partnership units are e for KKR's common units changes from one-for-one, the number of votes to which the holders of the special voting units are en adjusted accordingly. Additional limited partner interests having special voting rights could also be issued. See " Issuance Security of the special voting that they have a support of the special voting units are entitle them to participate in the vote on the same basis as the partner of the special voting units are entitle them to participate in the vote on the same basis as the partner of the special voting units are entitle them to partner of the partner of the special voting units are entitle them to partner of the partner of the

In the case of KKR common units held by KKR's managing partner on behalf of non-citizen assignees, KKR's managing distribute the votes on those units in the same ratios as the votes of partners in respect of other limited partner interests are cast. KKF

Table

not anticipate that any meeting of unitholders will be called in the foreseeable future. Any action that is required or permitted to be limited partners may be taken either at a meeting of the limited partners or without a meeting, without a vote and without prior notice in writing describing the action so taken are signed by limited partners owning not less than the minimum percentage of the voting outstanding limited partner interests that would be necessary to authorize or take that action at a meeting. Meetings of the limited be called by KKR's managing partner or by limited partners owning at least 50% or more of the voting power of the outstanding lirinterests of the class for which a meeting is proposed. Unitholders may vote either in person or by proxy at meetings. The holders of the voting power of the outstanding limited partner interests of the class for which a meeting has been called, represented in proxy, will constitute a quorum unless any action by the limited partners requires approval by holders of a greater percentage of partner interests, in which case the quorum will be the greater

However, if at any time any person or group (other than KKR's managing partner and its affiliates, or a direct or subsequent transferee of KKR's managing partner or its affiliates) acquires, in the aggregate, beneficial ownership of 20% or more of any clausits then outstanding, that person or group will lose voting rights on all of its units and the units may not be voted on any matter and considered to be outstanding when sending notices of a meeting of unitholders, calculating required votes, determining the presence or for other similar purposes. KKR's units held in nominee or street name account will be voted by the broker or other nominee in with the instruction of the beneficial owner unless the arrangement between the beneficial owner and his nominee provides

Status as Lim

By transfer of KKR's common units in accordance with KKR's partnership agreement, each transferee of common units will be a limited partner with respect to the common units transferred when such transfer and admission is reflected in KKR's books and recast described under "Limited Liability" above, in KKR's partnership agreement or pursuant to Section 17-804 of the Dela Partnership Act (which relates to the liability of a limited partner who receives a distribution of assets upon the winding uppartnership and who knew at the time of such distribution that it was in violation of this provision) the common units will be fully

Non-Citizen Assignees;

If KKR is or becomes subject to U.S. federal, state, local, foreign or other laws or regulations that in the determination of KKF partner create a substantial risk of cancellation or forfeiture of any property in which KKR has an interest because of the nationality or other related status of any limited partner, KKR may redeem the common units held by that limited partner at their current mar avoid any cancellation or forfeiture, KKR's managing partner may require each limited partner to furnish information about his citizenship or related status. If a limited partner fails to furnish information about his nationality, citizenship or other related status with after a request for the information or KKR's managing partner determines, with the advice of counsel, after receipt of the information determined partner is not an eligible citizen, the limited partner may be treated as a non-citizen assignee. A non-citizen assignee does right to direct the voting of his limited partnership units and may not receive distributions in kind upon KKR's partnership's

<u>Table</u>

Inde

Under KKR's partnership agreement, in most circumstances KKR would indemnify the following persons, to the fullest extent law, from and against all losses, claims, damages, liabilities, joint or several, expenses (including legal fees and expenses), judg penalties, interest, settlements or other contents of the fullest extent law, from and against all losses, claims, damages, liabilities, joint or several, expenses (including legal fees and expenses), judg penalties, interest, settlements or other contents of the full settlement in the full sett

its manag

any departing manag

any person who is or was an affiliate of a managing partner or any departing manag

any person who is or was a member, partner, tax matters partner, officer, director, employee, agent, fiduciary KKR or its subsidiaries, KKR's managing partner or any departing managing partner or any affiliate of subsidiaries, KKR's managing partner or any departing managing partner or any departing managing partner.

any person who is or was serving at the request of KKR's managing partner or any departing managing p affiliate of KKR's managing partner or any departing managing partner as an officer, director, employee, men agent, fiduciary or trustee of another

any person designated by KKR's mana

KKR has agreed to provide this indemnification unless there has been a final and non-appealable judgment by a court of jurisdiction determining that these persons acted in bad faith or engaged in fraud or willful misconduct. KKR has also agreed to indemnification for criminal proceedings. Any indemnification under these provisions will only be out of KKR's assets. Unless agrees, KKR's managing partner will not be personally liable for, or have any obligation to contribute or loan funds or assets to KIKKR to effectuate indemnification. The indemnification of the persons described above shall be secondary to any indemnification su entitled from another person or the relevant KKR fund to the extent applicable. KKR may purchase insurance against liabilities ass and expenses incurred by persons for KKR's activities, regardless of whether KKR would have the power to indemnify the persons disabilities under its partnership

Exclusive Delaware

KKR's partnership agreement provides that each of its limited partners and KKR's managing partner and each person holding as interest in KKR, to the fullest extent permitted by law, (1) irrevocably agrees that any claims, suits, actions or proceedings aris relating in any way to KKR's partnership agreement shall be exclusively brought in the Court of Chancery of the State of Delawa court does not have subject matter jurisdiction thereof, any other court in the State of Delaware with subject matter jurisdiction; (2 submits to the exclusive jurisdiction of such courts in connection with any such claim, suit, action or proceeding; (3) irrevocably a and waives any right to, assert in any such claim, suit, action or proceeding that (A) it is not personally subject to the jurisdiction or any other court to which proceedings in such courts may be appealed, (B) such claim, suit, action or proceeding is brought in an inforum, or (C) the venue of such claim, suit, action or proceeding is improper; (4) expressly waives any requirement for the posting of party bringing such claim, suit, action or proceeding; (5) consents to process being served in any such claim, suit, action or proceeding, certified mail, return receipt requested, a copy thereof to such party at the address in effect for notices hereunder, and agreement shall a any right to service of process and notice thereof; provided, that nothing in clause (5) hereof shall a any right to service shall constitute good and sufficient service of process and notice thereof; provided, that nothing in clause (5) hereof shall a any right to service of process.

<u>Table</u>

in any other manner permitted by law; and (6) irrevocably waives any and all right to trial by jury in any such claim, suit, action or

Books

KKR's managing partner is required to keep appropriate books of KKR's business at its principal offices or any other place d KKR's managing partner. The books would be maintained for both tax and financial reporting purposes on an accrual basis. For tax a reporting purposes, KKR's year ends on December 31. See "Material U.S. Federal Tax Consequences of KKR Common Unit beginning on page 184 of this proxy statemen

Right to Inspect KKR's Books a

KKR's partnership agreement provides that a limited partner can, for a purpose reasonably related to such person's interes partner, upon reasonable written demand and at such person's own expense, have furnished to

promptly after becoming available, a copy of KKR's U.S. federal, state and local income tax

copies of KKR's partnership agreement, its certificate of limited partnership of, related amendments and power under which they have be

KKR's managing partner may, and intends to, keep confidential from the limited partners trade secrets or other information the which KKR's managing partner believes is not in KKR's best interests or which KKR is required by law or by agreements with the keep

COMPARISON OF KKR COMMON UNITS AND KFN COMMON SHARES

The following section of this proxy statement/prospectus describes the differences between the rights of KKR common unithol rights of KFN common shareholders, to the extent such differences are material. It does not purport to be a complete statement of KKR common unitholders under applicable Delaware law and KKR's partnership agreement, or the rights of KFN common shareh applicable Delaware law and KFN's operating

We encourage you to read carefully the relevant provisions of the Delaware Limited Partnership Act and the Delaware Limited Partnership Act and the Delaware Limited Partnership agreement. The following section of this proxy statement is qualified in its entirety by reference to KKR's partnership agreement and KFN's operating

	KKR	KFN
Type of Legal Entity	KKR is a Delaware limited partnership with a general partner.	KFN is a Delaware limited liability compar
Voting Rights of Holders	KKR's common unitholders have only limited voting and consent rights and have no right to elect or remove KKR's managing partner or the directors on the board of KKR's managing partner. As such, KKR's common unitholders have limited or no ability to influence management's decisions regarding KKR's business.	KFN common shareholders are entitled one per share as provided in KFN's operating agreement. Members of KFN's board of directly a plurality vote of the holder KFN's common shares and holders of KFN common shares are entitled nominate indivitor election to KFN's board of directors.
Management and Control	KKR's managing partner is the general partner of KKR and manages all of KKR's operations and activities. KKR's managing partner is authorized in general to perform all acts that it determines to be necessary or appropriate to carry out KKR's purposes and to conduct KKR's business. KKR's managing partner is wholly owned by its principals and controlled by its founders.	Subject to limited specified exceptions, the business and affairs of KFN are managed board of directors. However, for so long as management agreement is in effect, but sub all times to the oversight of the KFN board directors, the manager shall manage the bu of KFN and provide services to KFN in accordance with the terms of the managem agreement.
	KKR's partnership agreement and the Delaware Limited Partnership Act prohibit limited partners from participating in the operation, management or control of KKR's business.	

NYSE Governance Protections

KKR

KFN

As a limited liability company, KFN does r qualify for the exceptions available to KKR

the corporate governance and other requires

of the rules of the NYSE. Accordingly, KF.

As a limited partnership, KKR qualifies for exceptions from certain corporate governance and other requirements of the rules of the NYSE. Pursuant to these exceptions, limited partnerships may elect, and KKR has elected, not to comply with certain corporate governance requirements of the NYSE, including the requirements: (1) that the listed company have a nominating and corporate governance committee that is composed entirely of independent directors; and (2) that the listed company have a compensation committee that is composed entirely of independent directors. The limited liability company agreement of KKR's managing partner requires it to maintain a board of directors, not less than a majority of whom are independent pursuant to NYSE rules relating to corporate governance matters. The board of

common shareholders enjoy greater corpora governance protection under the NYSE rule KKR common unitholders.

Board Structure

directors of KKR's managing partner is required to maintain an audit committee (each member of which is an independent director and financially literate), a conflicts committee (each member of which is an independent director), a nominating and corporate governance committee (of which at least one member must be an independent director) and an executive committee (comprised of the chairman or co-chairman of the board and any other directors selected by the chairman or co-chairman). The board of directors of KKR's managing partner does not maintain a compensation committee and, as is permitted under the exceptions to the NYSE rules noted above, the nominating and corporate governance committee of the board of directors of KKR's managing partner is not composed entirely of independent directors. The board of directors of KKR's managing partner is empowered to appoint additional committees as it may deem appropriate. 213

KFN's operating agreement provides that a majority of its entire board of directors may time increase or decrease the number of dir on the board. Unless KFN's operating agree is amended, the number of directors may no less than five or more than thirteen, with a majority of the board consisting of indepen directors. Except as may be provided by the of directors in setting the terms of any class series of shares, any vacancies on the board directors may only be filled by the affirmat vote of a majority of the remaining director office, even if the remaining directors do no constitute a quorum, and any director elected fill a vacancy will serve for the remainder of full term of the directorship in which the va occurred and until a successor is duly elected qualifies.

KKR

The holders of a majority of the Class A shares of KKR's managing partner, all of which are held by KKR principals, have the power, in their sole discretion, to (a) determine the number of directors and such directors' term of office, (b) appoint directors, (c) remove and replace directors at any time, with or without cause and for any reason or no reason, and (d) fill any positions created by the board of directors as a result of an increase in the size of the board of directors or vacancies. These holders also have certain approval rights over actions that KKR may take. At any time there is one or more "Designated Members" (defined in the managing partner's limited liability company agreement as Henry R. Kravis and George R. Roberts, who may designate additional and successor "Designated Members"), the then-serving Designated Members are deemed to represent a majority interest of holders of Class A shares of KKR's managing partner for the above purposes. As a result, Henry R. Kravis and George R. Roberts can elect the board of directors of KKR's managing partner with no other vote of holders of membership interests in KKR's general partner or KKR common unitholders. KKR's managing partner's limited liability company agreement does not provide for the classification of directors.

KFN

The board of directors is empowered to apprexecutive committee, an audit committee, a compensation committee, a nominating and governance committee and such other committee to be board of directors may deem approprious As a limited liability company, KFN is subjicted NYSE rules that require KFN's compensation committee and nominating and corporate governance committee to be composed sole independent directors. KFN's operating agreedoes not provide for classification of directors.

Removal of Managing Member / Directors

KKR's partnership agreement provides that its managing partner may not be removed or expelled, with or without cause, by KKR's unitholders.

Subject to the rights of holders or more or a series of shares, any director or the entire be directors may be removed from office at an but only for cause and then only by the affit vote of the members of KFN holding at least two-thirds of the votes entitled to be cast in election of directors. For purposes of remove "cause" means with respect to any particular director, conviction of a felony or a final jut of a court of competent jurisdiction holding such director caused demonstrable, materiat to KFN through bad faith or active and delidishonesty.

Table

KKR

As noted above, the holders of a majority of the Class A shares of KKR's managing partner have the power, in their sole discretion, to remove one or all of the directors on the board of directors of KKR's managing partner. In addition, by virtue of their status as Designated Members of KKR's managing partner, Henry R. Kravis and George R. Roberts can exercise that power with no other vote of holders of membership interests in KKR's general partner or KKR common unitholders. meetings of its unitholders. KKR's partnership agreement provides that meetings of its unitholders may be called by KKR's managing of the voting power of the outstanding limited partner interests of the class for which a meeting

KFN

Unitholder / Shareholder Meetings

KKR is not required to, and does not, hold regular partner or by limited partners owning at least 50% has been called.

KFN is required to hold an annual meeting common shareholders for the election of diand for the transaction of any other busines may be raised at such annual meeting. Spec meetings of KFN's common shareholders n only be called by the chairman of the board directors, the president of KFN, the chief executive officer of KFN or the board of di or by the secretary of KFN upon the written request of the members entitled to cast not than a majority of the votes entitled to be ca such a meeting.

Action Through Writing

Any action that is required or permitted to be taken by KKR's limited partners, if authorized by KKR's managing partner, may be taken without a meeting, without a vote and without prior notice if an approval in writing setting forth the action so taken is signed by limited partners of KKR owning not less than the minimum percentage of the voting power of the outstanding limited partner interests that would be necessary to authorize or take that action at a meeting at which all the KKR limited partners were present and voted. As of the date of this proxy statement/prospectus, KKR Holdings L.P., whose general partner is currently controlled by Henry R. Kravis and George R. Roberts, controls such a vote.

Any action that is required or permitted to l taken by KFN's members may be taken wit meeting, without a vote and without prior n a unanimous written consent setting forth th action so taken is signed by all of KFN's me

Distributions

Subject to applicable law, KKR's partnership agreement provides that KKR's managing partner may determine the amount and timing of any distributions to KKR's unitholders in its sole discretion.

KKR

KKR's managing partner has determined to make distributions to KKR's limited partners pro rata according to such limited partners' percentage interests in KKR.

Amendments to Governing Instruments

KKR's managing partner may amend KKR's partnership agreement without the approval of any partner, any unitholder or any other person for various specified reasons; provided, that no provision of KKR's partnership agreement that requires the vote or consent of unitholders holding a percentage of the voting power of outstanding voting units to take any action will be amended, altered, changed, repealed or rescinded in any respect that would have the effect of reducing such voting percentage unless such amendment is approved by the written consent or the affirmative vote of unitholders whose aggregate outstanding voting units constitute not less than the voting or consent requirement sought to be reduced; provided further, that no amendment to KKR's partnership agreement may (a) subject to limited exceptions, enlarge the obligations of any limited partner without its consent or (b) enlarge the obligations of, restrict in any way any action by or rights of, or reduce in any way the amounts distributable, reimbursable or otherwise payable to KKR's managing partner or any of its affiliates without KKR's managing partner's consent (which may be given or withheld in its sole discretion); provided further, that subject to limited exceptions, any amendment that would have a material adverse effect on the rights or

216

KFN

Subject to applicable law, the board of dire KFN may, in its sole discretion, determine whether to declare and pay, and the amount timing of, any distributions to its members. Subject to any share designations, distribution be paid in proportion to each KFN members respective percentage ownership.

The KFN operating agreement may be ame only by a majority vote of the board of dire KFN, except that any amendment to the fol provisions requires an affirmative vote of h of at least a majority of the outstanding KF. common shares: the authorization of an inc the number of authorized KFN common sha the voting rights of the holders of KFN con shares; the hiring of a replacement manager following the termination of the manageme agreement; the rights of holders of KFN co shares to vote on the merger or consolidation KFN, the sale, lease or exchange of all or substantially all of KFN's property or assets certain other business combinations or transactions; the rights of holders of KFN common shares to vote on the dissolution, up and liquidation of KFN; and the provision the operating agreement governing amenda thereof.

KFN

KKR

preferences of any class of partnership interests in relation to other classes of partnership interests must be approved by the holders of not less than a majority of the outstanding partnership interests of the class affected; provided further, that subject to certain exceptions, no amendment to KKR's partnership agreement shall become effective without the approval of unitholders holding at least 90% of the voting power of the outstanding voting units unless KKR obtains an opinion of counsel to the effect that such amendment will not affect the limited liability of any limited partner under the Delaware Limited Partnership Act. Any amendment of KKR's partnership agreement is required to be approved by the holders of a majority of the Class A shares of KKR's managing partner. As noted above, by virtue of their status as Designated Members of KKR's managing partner, Henry R. Kravis and George R. Roberts can exercise that power with no other vote of holders of membership interests in KKR's general partner.

Issuance of Additional Securities

KKR's partnership agreement provides that KKR may issue additional partnership securities and options, rights, warrants and appreciation rights relating to partnership securities at any time for such consideration and on such terms and conditions as KKR's managing partner, KKR Management LLC, shall determine in its sole discretion without the approval of any limited partners. KKR's managing partner may fix the designations, preferences, rights, powers and duties of any class or series of such securities in its sole discretion.

KFN is authorized, pursuant to action by its of directors, to issue up to 500,000,000 con shares and 50,000,000 preferred shares in comore series. As of March 18, 2014, KFN has 204,824,159 common shares and 49,500,000 preferred shares outstanding.

Asset Sales, Mergers and Consolidations

KKR

KKR's partnership agreement generally prohibits KKR's managing partner, without the prior approval of the holders of a majority of the voting power of KKR's outstanding voting units, from causing KKR to sell, exchange or otherwise dispose of all or substantially all of its and its subsidiaries' assets, taken as a whole, in a single transaction or a series of related transactions. However, KKR's managing partner in its sole discretion may mortgage, pledge, hypothecate or grant a security interest in all or substantially all of KKR's and its subsidiaries' assets (including for the benefit of persons other than KKR or its subsidiaries, including KKR's affiliates) without that approval. KKR's managing partner may also sell all or substantially all of KKR or its subsidiaries' assets under any forced sale of any or all of KKR or its subsidiaries' assets pursuant to the foreclosure of, or other realization upon, those encumbrances without that approval. KKR's partnership agreement also prohibits its managing partner from merging, consolidating or combining KKR with one or more other business entities without the approval of the holders of a majority of the voting power of KKR's outstanding voting units, except to convert KKR to another limited liability entity if certain conditions are met. Any merger, sale or other combination of KKR with or into any other person is required to be approved by the holders of a majority of the Class A shares of KKR's managing partner. As noted above, by virtue of their status as Designated Members of KKR's managing partner, Henry R. Kravis and George R. Roberts can exercise that power with no other vote of holders of membership interests in KKR's general partner or KKR common unitholders.

KFN

KFN's operating agreement prohibits the m or consolidation of KFN with or into another business entity, or the sale, lease or exchange all or substantially all of KFN's property or unless, in each case, KFN's board of director adopts a resolution by a majority vote appro such action and unless such action is approthe affirmative vote of the holders of a maje the outstanding shares entitled to vote there KFN's operating agreement contains provis based on Section 203 of the DGCL which p KFN from engaging in a business combinat with an interested shareholder for a period years following the time that the holder of s became an interested shareholder unless such business combination is approved by the affirmative vote of the holders of 66²/₃% of outstanding KFN common shares, excluding shares held by the interested shareholder or affiliate or associate of the interested sharel An interested shareholder is (a) a person wh directly or indirectly, controls 15% or more outstanding voting shares of KFN at any tir within the prior three-year period or (b) a p who is an assignee of shares owned by an interested shareholder in a transaction not involving a public offering at any time with prior three-year period. A business combina includes (1) a merger or consolidation of K any subsidiary of KFN with or caused by an interested shareholder or any affiliate of an interested shareholder, (2) a sale or other disposition of property or assets, or issuance transfer of any securities of KFN or any subsidiary, with or caused by an interested shareholder or any affiliate of an interested shareholder having an aggregate market val equal to 10% or more of the aggregate mark value of the outstanding shares of KFN and

KKR

KFN

(3) certain transactions that would increase interested shareholder's proportionate share ownership in KFN. This provision does not

Fiduciary Duties

KKR's partnership agreement contains provisions that reduce and eliminate its managing partner's duties (including fiduciary duties) to KKR's common unitholders. KKR's partnership agreement also restricts the remedies available to KKR common unitholders for actions taken that without those limitations might constitute breaches of duty (including fiduciary duties). In addition, the limited liability company agreement of KKR's managing partner has provisions that reduce and eliminate duties (including fiduciary duties) that KKR's managing partner's members or directors on the board of directors of KKR's managing partner may have otherwise had absent such provisions.

where the business combination or the trans that resulted in the holder of shares becomi interested shareholder is approved by KFN of directors prior to the time the interested shareholder acquired its 15% interest. KFN's operating agreement provides that the duties of the directors of KFN will generall consistent with the duties of a director of a Delaware corporation. A director of a Dela corporation owes fiduciary duties of good f due care and loyalty to the corporation and stockholders. To the fullest extent permitted law, no director of KFN has any duties (fid or otherwise) with respect to any action or of the manager, and any actions or inaction KFN directors that cause KFN to act in

compliance or in accordance with the mana

agreement are deemed consistent and comp

with the fiduciary duties of such directors a

not constitute a breach of any duty under the operating agreement or existing in law, in e

or otherwise.

Indemnification

KKR

KKR's partnership agreement provides, in most circumstances, for the indemnification of the following persons, to the fullest extent permitted by law, from and against all losses, claims, damages, liabilities, joint or several, expenses (including legal fees and expenses), judgments, fines, penalties, interest, settlements or other amounts arising from any and all threatened, pending or completed claims, demands, actions, suits or proceedings, in which such person may be involved or is threatened to be involved by reason of his status as such: KKR's managing partner; any departing managing partner; any person who is or was an affiliate of KKR's managing partner or any departing managing partner; any person who is or was a member, partner, tax matters partner, officer, director, employee, agent, fiduciary or trustee of KKR or KKR's subsidiaries, KKR's managing partner or any departing managing partner or any affiliate of KKR, KKR's subsidiaries, KKR's managing partner or any departing managing partner; any person who is or was serving at the request of KKR's managing partner or any departing managing partner or any affiliate of KKR's managing partner or departing managing partner as an officer, director, employee, member, partner, tax matters partner, agent, fiduciary or trustee of another person; or any person designated by KKR's managing partner in its sole discretion.

KFN

Section 18-108 of the Delaware Limited Li Company Act allows a limited liability con to indemnify and hold harmless any member manager or other person from and against a all claims and demands whatsoever. KFN's operating agreement provides that, to the fu extent permitted by law, KFN will indemni directors and officers or any person who wa a party or is threatened to be made a party t threatened, pending or completed action, su proceeding, whether civil, criminal, admini or investigative (other than an action by or right of KFN) by reason of the fact that the is or was a director, officer, employee, tax i member or agent of KFN, or is or was servi the request of KFN as a director, officer, en or agent of another company, against exper (including attorneys' fees), judgments, fines amounts paid in settlement actually and reasonably incurred by the person in conne with such action, suit or proceeding if the p acted in good faith and in a manner the pers reasonably believed to be in or not opposed best interests of KFN and, with respect to a criminal action or proceeding, had no reaso cause to believe the person's conduct was unlawful.

Limitations on Liability o	f
Directors and Officers	

KKR

KKR's partnership agreement provides that KKR's managing partner and its affiliates are not liable to KKR or KKR's unitholders for any losses, claims, damages, liabilities, joint or several, expenses (including legal fees and expenses), judgments, fines, penalties, interest, settlements or other amounts arising as a result of any act or omission of a person for whom indemnification is provided for under the partnership agreement, or for any breach of contract or any breach of duties (including breach of fiduciary duties) whether arising at law, in equity or otherwise, unless there has been a final and non-appealable judgment entered by a court of competent jurisdiction determining that, in respect of the matter in question, the applicable person acted in bad faith or engaged in fraud or willful misconduct. KKR's partnership agreement is governed by and will be construed in accordance with the laws of the State of Delaware, without regard to the

principles of conflicts of laws.

KKR's partnership agreement provides that its unitholders submit to the exclusive jurisdiction of the Court of Chancery in the State of Delaware or, if such court does not have subject matter jurisdiction, any other court in the State of Delaware with subject matter jurisdiction.

KFN

KFN's operating agreement provides that a director of KFN will not be liable to it, any subsidiaries or any holder of KFN common for monetary damages for acts or omissions arising from the performance of any such director's obligations or duties in connectio KFN, including breach of fiduciary duty, exfollows: for breach of the directors' duty of to KFN or the holders of KFN common sha acts or omissions not in good faith or which involve intentional misconduct or a knowin violation of law, or for any transaction from the director derived an improper personal by

KFN's operating agreement is governed by will be construed in accordance with the latthe State of Delaware, without regard to the principles of conflicts of laws.

KFN's operating agreement provides that it members irrevocably submit to the non-excipurisdiction and venue of any Delaware stat or United States federal court sitting in Wilmington, Delaware in disputes arising of KFN's operating agreement.

Governing Law

Jurisdiction

OTHER IMPORTANT INFORMATION REGARDING THE PARTIES

The current business address of KFN and each of its executive officers and directors, is 555 California Street, 50th Floor, Sa California 94104, and the business telephone number of each such entity or person is (415) 315-3620. The beneficial ownership, if such entity or person of KFN common shares is described in greater detail in the section titled "Security Ownership of Certain Benef and Management" beginning on page 167 of this proxy statemen

During the past five years none of KFN or any of its directors and executive officers have been (i) convicted in a crimina (excluding traffic violations or similar misdemeanors) or (ii) party to any judicial or administrative proceeding (except for matt dismissed without sanction or settlement) that resulted in a judgment, decree or final order enjoining the person from future vio prohibiting activities subject to, federal or state securities laws, or a finding of any violation of federal or state securities.

The names and material occupations, positions, offices or employment during the past five years of the directors and executiv

Tracy L. Collins Director. Ms. Collins has been a director of KFN since August 2006. During her career in financial service worked as a Senior Managing Director and Head of Asset-Backed Securities Research at Bear Stearns & Co., Inc. from March 1 December 2000 and prior to that as a Managing Director and Head of Asset-Backed Securities Research at Credit Suisse Group (form as Credit Suisse First Boston) for approximately nine years. Ms. Collins is a citizen of the U

Robert L. Edwards Director. Mr. Edwards has been a director of KFN since November 2011. He became Chief Execut Safeway Inc. in May 2013, and has been President of Safeway since April 2012. Prior to his roles as Chief Executive Officer and I was Chief Financial Officer and Executive Vice President, respectively, of Safeway since March 2004. He was Executive Vice F Chief Financial Officer of Maxtor Corporation from September 2003 to March 2004. Prior to joining Maxtor, Mr. Edwards was President, Chief Financial Officer and Chief Administrative Officer at Imation Corporation, a developer, manufacturer and supplier and optical data storage media, from 1998 to 2003. Before joining Imation, Mr. Edwards had a successful 20-year career at San Corporation, and held positions of increasing responsibility in the areas of finance, administration and corporate development. Mr. citizen of the U

Craig J. Farr Director, President and Chief Executive Officer. Mr. Farr has been a director, and has served as President and Cl Officer, of KFN since July 2013. Mr. Farr, a Member of KKR Management LLC, the general partner of KKR, and an employee Group, joined the KKR Group in 2006 and serves as the Head of KKR Asset Management LLC and Global Head of the KKR Group, joined the KKR Group in 2006 and serves as the Head of KKR Asset Management LLC and Global Head of the KKR Group, Mr. Farr spent 12 years at Citig Markets and Origination business as well as on its Risk Committee. Prior to joining the KKR Group, Mr. Farr spent 12 years at Citig Markets Inc. where he was promoted to Managing Director in 2001 and served as Co-Head of North American Equity Cap Mr. Farr's previous responsibilities at Citigroup included Head of U.S. Convertible and Corporate Equity Derivative Origination. Mr. Farr's previous Brothers within the investment banking division. Mr. Farr graduated with a Bachelor of Commerce fundamental University in Kingston, Canada. Mr. Farr is a citize

Vincent Paul Finigan Director. Mr. Finigan has been a director of KFN since February 2006. He is currently Of Counsel with t Snyder Miller & Orton LLP. He was a partn

Table

firm of Morgan, Lewis & Bockius LLP from 2003 to 2007 and was with the law firm of Brobeck, Phleger & Harrison LLP from 1

Mr. Finigan is a citizen of the U

Paul M. Hazen Director. Mr. Hazen has been the Chairman of the KFN Board of Directors since July 2004. Mr. Hazen was Fargo Bank for 31 years, retiring in 2001. With Wells Fargo, he served as Vice Chairman from 1978 to 1984, President and COO of 1995, Chairman and President and CEO from 1995 to 1998 and Chairman from 1998 to 2001. Mr. Hazen was also a senior advisor Group from 2001 to 2011. Mr. Hazen also brings considerable directorial and board committee experience to the Board. Mr. director of Safeway for 21 years, including Lead Independent Director beginning in May 2004, before stepping off in May 2012 2000 to May 2010, he served as a director of Xstrata AG, listed on the London stock exchange, where Mr. Hazen served on the R Committee. Mr. Hazen is also Chairman of Accel-KKR, which is not an investment fund managed by the KKR Group independently from it. From June 1999 to July 2006, Mr. Hazen served as Deputy Chairman and Lead Independent Director Group PLC, where he also served as Chair of the Audit Committee. From January 2001 to June 2005, Mr. Hazen served as a director of the Year by the Outstanding Directors Exchange (ODX). Mr. Hazen holds a B.S. in Finance from the University of Ar MBA from the University of California at Berkeley. Mr. Hazen is a citizen of the University of California at Berkeley. Mr. Hazen is a citizen of the University of California at Berkeley.

R. Glenn Hubbard Director. Mr. Hubbard has been a director of KFN since October 2004. He was named the Dean of Columbia Graduate School of Business in 2004. From 1994 to the present time, Mr. Hubbard has been the Russell L. Carson Professor of Economics in the Graduate School of Business of Columbia University, where he is a professor of Economics. From February 2001 2003, he took a leave of absence from his teaching post to serve as Chairman of the U.S. Council of Economic Advisers. He Assistant Secretary of the U.S. Treasury Department for Tax Policy from 1991 to 1993. Mr. Hubbard is a research associate at Bureau of Economic Research, and a visiting scholar at the America Enterprise Institute in Washington, D.C. He is currently ADP, Inc., BlackRock Closed-End Funds and MetLife, Inc. He is also a past Chairman and trustee of the Economic Club of New trustee of the Fifth Avenue Presbyterian Church. He was previously a board member of Capmark Financial, Duke Realty Ripplewood Holdings, RH Donnelly and Information Services Group, Inc. Mr. Hubbard serves as KFN's Lead Independ Mr. Hubbard is a citizen of the U.S.

Ross J. Kari Director. Mr. Kari has been a director of KFN since August 2004. He was the Chief Financial Officer of the Loan Mortgage Corporation (Freddie Mac) from September 2009 to November 2013. Until September 2009, he was the Chief Financial Fifth Third Bancorp. Prior to that he was Executive Vice President and CFO of Safeco Corporation from June 2006 to November Executive Vice President and Chief Operating Officer of the Federal Home Loan Bank of San Francisco from March 2002 to Mr. Kari was Chief Financial Officer of MyCFO, a wealth management firm, from 2001 to 2002, and prior to that, was employed at for 18 years, including as Chief Financial Officer from 1998 to 2001. Mr. Kari is a citizen of the U

Ely L. Licht Director. Mr. Licht has been a director of KFN since April 2005. From March 2007 to February 2009 he was E President and Chairman of Credit Policy for Capmark Financial, and from March 2009 to August 2009 he served as a consultant Financial. Prior to joining Capmark Financial, Mr. Licht was in retirement. From January 2002 to December 2002, Mr. Licht was serving as a consultant to Wells Fargo on a part-time basis. Prior to his semi-retirement, Mr. Licht was employed at Wells Fargo including as Chief C

Table

from November 1998 to December 2001, and Executive Vice President for Credit Administration from February 1990 to Nov Mr. Licht is a citizen of the U

Deborah H. McAneny Director. Ms. McAneny has been a director of KFN since April 2005. Ms. McAneny was the Chief Ope of Benchmark Assisted Living, LLC from April 2007 to May 2009. Prior to that she was employed at John Hancock Financial 20 years, including as Executive Vice President for Structured and Alternative Investments of John Hancock Financial Services and of its Policy Committee from 2002 to 2004, Senior Vice President for John Hancock's Real Estate Investment Group from 2000 to 20 Vice President of the Real Estate Investment Group from 1997 to 2000. She is currently the lead independent director of HFF, Inc. at the Chairman of its Nominating and Governance Committee and a member of its Audit Committee. She is also a director of RRE Trust, Inc. and a member of its audit committee. In addition she is a director of Benchmark Senior Living, LLC and RREEF Ame and a Trustee of the University of Vermont, the Rivers School and is a past President of the Commercial Mortgage Securities Ms. McAneny holds and an Advanced Professional Director Certification from the American College of Corporate Directors, a na company director education and credentialing organization. Mr. McAneny is a citizen of the University of Vermont, the Rivers Certification from the American College of Corporate Directors, a na company director education and credentialing organization.

Nicole J. Macarchuk General Counsel and Secretary. Ms. Macarchuk has been KFN's General Counsel and Secretary since No. Ms. Macarchuk has been employed by the KKR Group since July 2010 and is a Managing Director and General Counsel to KFI Prior to joining the KKR Group, Ms. Macarchuk was Co-General Counsel of Och-Ziff Capital Management Group LLC, a global asset management firm from 2005 to April 2010. At Och-Ziff, Ms. Macarchuk was head counsel responsible for supporting the private equity, energy and transactional business groups and initiatives, as well as provided coverage to the U.S. public distress structured products business units and public equity business units with respect to transactional matters. Ms. Macarchuk was O'Melveny & Myers LLP and O'Sullivan LLP from 1996 to 2005 where she focused on private equity transactions and fund form 1993 to 1996, Ms. Macarchuk was an associate at White & Case where she focused on banking and corporate credit transactions. Ms received a B.A. in Economics from Fordham University and a J.D., cum laude, from Fordham University Law School. Ms. Macarchuk citizen of the U

Michael R. McFerran Chief Operating Officer and Chief Financial Officer. Mr. McFerran has been KFN's Chief Operating December 2008 and Chief Financial Officer since November 2010. Mr. McFerran has been employed by the KKR Group since June Chief Financial Officer of KFN's manager. Prior to joining the KKR Group, Mr. McFerran spent the majority of his career at Ernst their financial services practice where he specialized in providing assurance and advisory services to a diverse mix of financial se Additionally, prior to joining the KKR Group, Mr. McFerran held Vice President positions at both XL Capital Ltd. and Ameri Mr. McFerran holds an M.B.A. from the Haas School of Business at U.C. Berkeley and is a Certified Public Accountant. Mr. Notice of the U.C. Berkeley and is a Certified Public Accountant. Mr. Notice of the U.C. Berkeley and is a Certified Public Accountant. Mr. Notice of the U.C. Berkeley and is a Certified Public Accountant. Mr. Notice of the U.C. Berkeley and is a Certified Public Accountant. Mr. Notice of the U.C. Berkeley and is a Certified Public Accountant. Mr. Notice of the U.C. Berkeley and is a Certified Public Accountant. Mr. Notice of the U.C. Berkeley and is a Certified Public Accountant. Mr. Notice of the U.C. Berkeley and is a Certified Public Accountant. Mr. Notice of the U.C. Berkeley and is a Certified Public Accountant. Mr. Notice of the U.C. Berkeley and is a Certified Public Accountant.

Scott C. Nuttall Director. Mr. Nuttall has been a director of KFN since July 2004 and is a Member of KKR's managing employee of the KKR Group. Mr. Nuttall joined the KKR Group in 1996 and heads the KKR Group's Global Capital and Asset I Group which includes the Client and Partner Group, KKR Capital Markets, KKR Asset Management and Prisma Capital Partners played a significant role in several of the KKR Group's private equity investments. Mr. Nuttall is currently a member of the board of First Data. He is actively involved in funds affiliated with the KKR Group and is a member of the KKR Group's management commi joining the KKR Group, he was with the Blackstone Group where he was involved in numerous merchant banking and merger and transactions. He received a B.S., summa cum laude, from the University of Pennsylvania. Mr. Nuttall is a citizen of the University of Pennsylvania.

<u>Table</u>

Scott A. Ryles Director. Mr. Ryles has a been a director of KFN since November 2008. He is currently the Managing Member Capital Strategies, LLC, which manages funds with an investment focus in consumer and small business loans. From January 2011 to 2012 he was the Chairman and CEO of Home Value Protection, Inc., an insurance holding company and parent of Home Value In He was previously Vice Chairman of Cowen and Company from February 2007 to December 2010. Prior to joining Cowen and Mr. Ryles served as the Chief Executive Officer of Procinea Management LLC and was the founder, Chief Executive Officer and Epoch Partners, Inc. Mr. Ryles also spent 18 years at Merrill Lynch in various investment banking positions including Co-Head of Investment Banking, Global Head of Natural Resources, Global Head of Technology and was a member of the firm's Operating Mr. Ryles was previously a Director of ArcSight, Inc., Fortify Software, Inc. and The Gymboree Corporation. We have been ac 2012, Home Value Insurance, Inc. was dissolved under the direction of the Ohio Superintendent of Insurance, all creditors and plan policyholders were paid in full and all premiums paid by policyholders were refunded. Mr. Ryles is a citizen of the User Insurance of the Ohio Superintendent of Insurance of the User Insurance of the Us

Willy R. Strothotte Director. Mr. Strothotte has a been a director of KFN since January 2007. Mr. Strothotte was Chairma International AG until 2011. From 1961 to 1978 Mr. Strothotte held various positions with responsibility for international trading i minerals in Germany, Belgium and the USA. In 1978, Mr. Strothotte joined Glencore International, taking up the position of Head of Minerals in 1984. Mr. Strothotte was appointed Chief Executive Officer of Glencore in 1993 and held the combined positions of Chief Executive Officer from 1994 until 2002, when the roles of Chairman and Chief Executive were split. Mr. Strothotte was Xstrata AG from 1994 to 2011, and Chairman of Xstrata plc from 2002 until 2011 and was previously a director of Centur Corporation and Minara Resources Limited. Mr. Strothotte is a citizen

The KKR

KKR, Fund Holdings, Merger Sub, KKR Group Holdings L.P., KKR Fund Holdings GP Limited, KKR Group Limited Management LLC are referred to collectively in this proxy statement/prospectus as the KKR Participants. Merger Sub's sole mer Holdings. The general partners of Fund Holdings are (1) KKR Group Holdings L.P., an exempted company formed under the Cayman Islands whose primary business is holding equity interests in each of KKR Management Holdings Corp., KKl Partnership L.P., Fund Holdings and KKR Fund Holdings GP Limited, and (2) KKR Fund Holdings GP Limited, an exempted com under the laws of the Cayman Islands whose primary business consists of performing the functions of, and serving as, the non-economator partner of Fund Holdings. KKR Group Holdings L.P. is the sole shareholder of KKR Fund Holdings GP Limited, and Holdings L.P.'s general partner is KKR Group Limited, an exempted company formed under the laws of the Cayman Islands whose business consists of performing the functions of, and serving as, the sole general partner of KKR Group Holdings L.P. The sole shareholder is KKR. KKR's general partner is KKR Management LLC, a Delaware limited liability company whose primal consists of performing the functions of, and serving as, the sole general partner is KKR Management LLC, and serving as, the sole general partner is KKR Management LLC, and serving as, the sole general partner is KKR Management LLC, and serving as, the sole general partner is KKR Management LLC, and serving as, the sole general partner is KKR Management LLC, and serving as, the sole general partner is KKR Management LLC, and serving as, the sole general partner is KKR Management LLC, and serving as, the sole general partner is KKR Management LLC, and serving as, the sole general partner is KKR Management LLC, and serving as, the sole general partner is KKR Management LLC, and serving as, the sole general partner is KKR Management LLC, and serving as, the sole general partner is KKR Management LLC, and serving as the serving as the serving a

The current business address of each of the KKR Participants and their executive officers, directors or controlling persons, as a c/o KKR, 9 West 57th Street, Suite 4200, New York, New York 10019, and the business telephone number of each such entity (212) 750-8300. The beneficial ownership, if any, of each such entity or person of KFN common shares is described in greater section titled "Security Ownership of the KKR Participants and Certain Affiliates" beginning on page 170 of this proxy statemen

During the past five years none of the KKR Participants or any of their directors, executive officers or controlling persons, a have been (i) convicted in a crimina

Table

(excluding traffic violations or similar misdemeanors) or (ii) party to any judicial or administrative proceeding (except for matt dismissed without sanction or settlement) that resulted in a judgment, decree or final order enjoining the person from future vio prohibiting activities subject to, federal or state securities laws, or a finding of any violation of federal or state sec

Board of Directors and Executive Officers of KKR Mana

The names and material occupations, positions, offices or employment during the past five years of the directors and executive KKR Management LLC, the general partner of KKR, are

Henry R. Kravis Co-Chairman and Co-Chief Executive Officer. Mr. Kravis co-founded KKR's business in 1976 and is Co-Co-Chief Executive Officer of KKR Management LLC. He is actively involved in managing the firm and serves on each of the reg Equity Investment and Portfolio Management Committees. Mr. Kravis currently serves on the boards of First Data Corporation International Capital Corporation Limited. He also serves as a director, chairman emeritus or trustee of several cultural, profe educational institutions, including The Business Council, Claremont McKenna College, Columbia Business School, Mount Sinai Partnership Fund for New York City, Partnership for New York City, Rockefeller University, Tsinghua University School of Ec Management and Sponsors for Educational Opportunity. He earned a B.A. from Claremont McKenna College in 1967 and an M.B. Columbia Business School in 1969. Mr. Kravis is a citizen of the University School of

George R. Roberts Co-Chairman and Co-Chief Executive Officer. Mr. Roberts co-founded KKR's business in 1976 and is Co-Co-Chief Executive Officer of KKR Management LLC. He is actively involved in managing the firm and serves on each of the reg Equity Investment and Portfolio Management Committees. Mr. Roberts serves as a director or trustee of several cultural and institutions, including Claremont McKenna College. He is also founder and chairman of the board of directors of REDF, a S nonprofit organization. He earned a B.A. from Claremont McKenna College in 1966 and a J.D. from the University of California (H. School in 1969. Mr. Roberts is a citizen of the U

David C. Drummond Director. Mr. Drummond has been a member of the board of directors since March 14, 2014. Mr. Diserved as the senior vice president, corporate development of Google Inc. since January 2006 and as its chief legal officer since Dece Previously, he served as Google Inc.'s vice president, corporate development and general counsel since February 2002. Pri Google Inc., from July 1999 to February 2002, Mr. Drummond served as chief financial officer of SmartForce, an education applications company. Prior to that, Mr. Drummond was a partner at the law firm of Wilson Sonsini Goodrich & Rosati. Mr. Drummond J.D. from Stanford University and a B.A. in history from Santa Clara University. Mr. Drummond is a citizen of the University.

Joseph A. Grundfest Director. Mr. Grundfest has been a member of the board of directors since July 15, 2010. Mr. Grundfest member of the faculty of Stanford Law School since 1990, where he is the William A. Franke Professor of Law and Business. He is faculty of the Arthur and Toni Rembe Rock Center for Corporate Governance at Stanford University; founder and director of Direct a venue for the professional education of directors of publicly traded corporations; and co-founder of Financial Engines, Inc., a services and advice to participants in employer-sponsored retirement plans, where he has served as a director since its incep Mr. Grundfest was a Commissioner of the SEC from 1985 to 1990. He holds a B.A. in Economics from Yale University and a J.D. fr Law School. Mr. Grundfest is a citizen of the University and the Un

Table

John B. Hess Director. Mr. Hess has been a member of the board of directors since July 28, 2011. Mr. Hess has been the cofficer of Hess Corporation since 1995 and a director since 1978. He was also director of Dow Chemical Co. from 2006 to 2013. He director on the Business Council, the Trilateral Commission, the Council on Foreign Relations and the Center for Strategic and Studies. Mr. Hess is a member of the board of trustees at the New York Public Library, Mount Sinai Hospital, the Deerfield Acad Dean's Advisors of Harvard Business School. Mr. Hess also serves as a member of the board of directors of Lincoln Center for the Arts. Mr. Hess earned a B.A. from Harvard College and an M.B.A. from Harvard Business School. Mr. Hess is a citizen of the University of the Council Center for the Council Center for the Council Center for the Cen

Dieter Rampl Director. Mr. Rampl has been a member of the board of directors since July 15, 2010. Mr. Rampl was chairman Group from 2006 to April 2012. Previously, Mr. Rampl was a member of the board for corporate business and corporate finance as Vereinsbank (Munich) from 1995 to January 1, 2003 when he was appointed as spokesman of the board of managing directors of 2006. In addition, Mr. Rampl has been a managing director of Charterhouse, London, manager of the corporate business of BHF and general manager of BHF North America, and he also operated in the foreign trade financing area at Société de Banque currently the vice chairman of Mediobanca S.p.A., chairman of the supervisory board of Koenig & Bauer AG and a member of the board of FC Bayern München AG, and was the chairman of the supervisory board of Bayerische Börse AG until June 2010 Mr. Rampl previously served as a director and chairman of the audit committee of KKR Guernsey GP Limited, the general KKR & Co. (Guernsey), L.P., formerly known as KKR Private Equity Investors, L.P. Mr. Rampl is a citize

Patricia F. Russo Director. Ms. Russo has been a member of the board of directors since April 15, 2011. Ms. Russo sexecutive officer of Alcatel-Lucent from 2006 to 2008. Prior to the merger of Alcatel and Lucent in 2006, she served as chairm executive officer of Lucent Technologies, Inc. from 2003 to 2006, and as president and chief executive officer from 2002 to 2006, rejoining Lucent in 2002, Ms. Russo was president and chief operating officer of Eastman Kodak Company from March 2001 to Dec Ms. Russo has been a director of Alcoa Inc. since 2008, Merck & Co., Inc. since 2009, General Motors Company sin Hewlett-Packard Company since 2011. Prior to its merger with Merck in 2009, Ms. Russo served as a director of Schering-Plough Ms. Russo is a citizen of the U

Thomas M. Schoewe Director. Mr. Shoewe has been a member of the board of directors since March 14, 2011. Mr. Schoewe vice president and chief financial officer for Wal-Mart Stores, Inc., a position he held from 2000 to 2010, and was employed by transitional capacity to January 2011. Prior to his employment at Walmart, Mr. Schoewe served as senior vice president and cheeficer for Black and Decker Corp., a position he held from 1993 to 1999. Prior to that, he served for four years as Black and I president of finance. He previously held the position of vice president of business planning and analysis. He joined Black and Deafter serving at Chicago-based Beatrice Companies, where he was chief financial officer and controller of Beatrice Consumer Dura has served on the board of directors of Northrop Grumman Corporation and General Motors Company since 2011. From 2001 to Mark Schoewe graduated from Loyola University of Chicago with a bachelor's of business administration degree in finance. Mr. Schoewe graduated from Loyola University of Chicago with a bachelor's of business administration degree in finance. Mr. citizen of the University of Chicago with a bachelor's of business administration degree in finance.

Robert W. Scully Director. Mr. Scully has been a member of the board of directors since July 15, 2010. Mr. Scully was a resolvent of the Chairman of Morgan Stanley from 2007 until his retirement in 2009, where he had previously been co-president, global capital markets and vice chairman of investment banking. Prior to joining Morgan Stanley, he served as a managing director Brothers and at Salomon Brothers. Mr. Scully has served as a director of Zoe

Table

June 2013. Previously, he was a director of Bank of America Corporation until May 2013, and has also served as a director of GMA Services and MSCI Inc. He holds an A.B. from Princeton University and an MBA from Harvard Business School. Mr. Scully is a director of GMA and the services are served as a director of GMA and the services and MSCI Inc. He holds an A.B. from Princeton University and an MBA from Harvard Business School. Mr. Scully is a director of GMA and the services are served as a director of GMA and the services and MSCI Inc. He holds an A.B. from Princeton University and an MBA from Harvard Business School. Mr. Scully is a director of GMA and the services are served as a director of GMA and the services are served as a director of GMA and the services are served as a director of GMA and the services are served as a director of GMA and the services are served as a director of GMA and the services are served as a director of GMA and the services are served as a director of GMA and the services are served as a director of GMA and the services are served as a director of GMA and the s

Todd A. Fisher Chief Administrative Officer. Mr. Fisher joined the KKR Group in 1993 and is Chief Administrative Officer. Mr. Fisher is responsible for overseeing the finance, legal, information human resources, public affairs and office operations functions, coordinating with the various businesses and geographies of the KKI overseeing the firm's efforts in real estate investments. He is a member of KKR's Real Estate Investment Committee. Mr. Fisher is Maxeda B.V. Previously, he served as a director of Rockwood Holdings, Inc. until January 2013, Northgate Information Solutions pl ALEA Group Holdings AG until 2007, and Bristol West Insurance Group until 2007. Prior to joining the KKR Group, Mr. Fisher Goldman, Sachs & Co. in New York and for Drexel Burnham Lambert in Los Angeles. Mr. Fisher holds a B.A. from Brown University in International Affairs from Johns Hopkins University, and an M.B.A. from the Wharton School of the University of Pennsy currently a member of the Board of Trustees of Brown University, the Board of Advisors for The Johns Hopkins University Advanced International Studies, the Advisory Board of the Clinton Health Access Initiative, the United States Holocaust Memorial the Council on Foreign Relations. Mr. Fisher is a citizen of the University and the Council on Foreign Relations.

William J. Janetschek Chief Financial Officer. Mr. Janetschek joined the KKR Group in 1997 and is Chief Financial Officer. Mr. Janetschek joined the KKR Group in 1997 and is Chief Financial Officer. Mr. Janetschek is Total 13 years. He holds a B.S. from St. John's University and an M.S., Taxation from Pace University. Mr. Janetschek is actively in community, serving as a sponsor and member of a variety of non-profit organizations including Student Sponsor Partners a University. Mr. Janetschek is a citizen of the University. Mr. Janetschek is a citizen of the University.

David J. Sorkin General Counsel. Mr. Sorkin joined the KKR Group in 2007 and is General Counsel and Section Management LLC and an employee of the KKR Group. Prior to joining the KKR Group, Mr. Sorkin was with Simpson Thacher & for 22 years. He served as a partner at the law firm and also served on the executive committee and was one of KKR's principal outsing the received a B.A., summa cum laude, from Williams College and a J.D., cum laude, from Harvard Law School. Mr. Sorkin is a contract of the contract

Board of Directors and Executive Officers of KKR Fund Holdings

The names and material occupations, positions, offices or employment during the past five years of the directors and executive KKR Fund Holdings GP Limited, a general partner of Fund Holdings, are

Henry R. Kravis Co-Chief Executive Officer and President. Please refer to "Board of Directors and Executive Officer and President LLC" for more information regarding Mr. Kravis' employed

George R. Roberts Co-Chief Executive Officer and President. Please refer to "Board of Directors and Executive Officer and President LLC" for more information regarding Mr. Roberts' employed

Todd A. Fisher Director. Please refer to " Board of Directors and Executive Officers of KKR Management LLC" for m regarding Mr. Fisher's employed

William J. Janetschek Director. Please refer to "Board of Directors and Executive Officers of KKR Management LLC" for m regarding Mr. Janetschek's employi

Table

David J. Sorkin Director and Secretary. Please refer to "Board of Directors and Executive Officers of KKR Management information regarding Mr. Sorkin's employing the control of the contr

Board of Directors and Executive Officers of KKR Gr

The names and material occupations, positions, offices or employment during the past five years of the directors and executive KKR Group Limited, the general partner KKR Group Holdings L.P., are

Henry R. Kravis Co-Chief Executive Officer and President. Please refer to "Board of Directors and Executive Co-Chief Executive Officer and President. Please refer to "Board of Directors and Executive Co-Chief Executive Officer and President. Please refer to "Board of Directors and Executive Co-Chief Executive Officer and President. Please refer to "Board of Directors and Executive Officer and President. Please refer to "Board of Directors and Executive Officer and President. Please refer to "Board of Directors and Executive Officer and President. Please refer to "Board of Directors and Executive Officer and President. Please refer to "Board of Directors and Executive Officer and President. Please refer to "Board of Directors and Executive Officer and President. Please refer to "Board of Directors and Executive Officer and President. Please refer to "Board of Directors and Executive Officer and President. Please refer to "Board of Directors and Executive Officer and President. Please refer to "Board of Directors and President. Please refer to "Board of Directors

George R. Roberts Co-Chief Executive Officer and President. Please refer to "Board of Directors and Executive Officer and President LLC" for more information regarding Mr. Roberts' employers.

Todd A. Fisher Director. Please refer to " Board of Directors and Executive Officers of KKR Management LLC" for m regarding Mr. Fisher's employe

William J. Janetschek Director and Secretary. Please refer to "Board of Directors and Executive Officers of KKR Managemore information regarding Mr. Janetschek's employi

David J. Sorkin Director and Secretary. Please refer to "Board of Directors and Executive Officers of KKR Management information regarding Mr. Sorkin's employing the control of the contr

Executive Officers of

The names and material occupations, positions, offices or employment during the past five years of the executive officers of Mo

Todd A. Fisher Chief Administrative Officer and Vice President. Please refer to "Board of Directors and Executive Communication Management LLC" for more information regarding Mr. Fisher's employer

William J. Janetschek Chief Financial Officer and Vice President. Please refer to "Board of Directors and Executive C Management LLC" for more information regarding Mr. Janetschek's employe

David J. Sorkin General Counsel, Vice President and Secretary. Please refer to "Board of Directors and Executive Counsel, Vice President and Secretary. Please refer to "Board of Directors and Executive Counsel, Vice President and Secretary. Please refer to "Board of Directors and Executive Counsel, Vice President and Secretary. Please refer to "Board of Directors and Executive Counsel, Vice President and Secretary. Please refer to "Board of Directors and Executive Counsel, Vice President and Secretary. Please refer to "Board of Directors and Executive Counsel, Vice President and Secretary. Please refer to "Board of Directors and Executive Counsel, Vice President and Secretary. Please refer to "Board of Directors and Executive Counsel, Vice President and Secretary. Please refer to "Board of Directors and Executive Counsel, Vice President and Secretary. Please refer to "Board of Directors and Executive Counsel, Vice President and Secretary. Please refer to "Board of Directors and Executive Counsel, Vice President and Preside

PROVISIONS FOR UNAFFILIATED SECURITY HOLDERS

No provision has been made (1) to grant unaffiliated KFN common shareholders access to the corporate files of KFN, any other proposed merger or any of their respective affiliates or (2) to obtain counsel or appraisal services at the expense of KFN or any other

APPROVAL OF ADJOURNMENT OF THE KFN SPECIAL MEETING

KFN common shareholders are being asked to approve a proposal that will give KFN's board of directors authority to adjourn special meeting, if necessary to solicit additional proxies if there are not sufficient votes to adopt the merger agreement at the time special meeting. If this proposal is approved, the KFN special meeting could be adjourned to any date. If the KFN special meeting KFN common shareholders who have already submitted their proxies will be able to revoke them at any time prior to their use. If proxy and do not indicate how you wish to vote on any proposal, or if you indicate that you wish to vote in favor of the adjournment but do not indicate a choice on the adjournment proposal, your common shares will be voted in favor of the adjournment but if you indicate that you wish to vote against the adoption of the merger agreement, your common shares will only be voted in adjournment proposal if you indicate that you wish to vote in favor of that proposal. The affirmative vote of holders of at least a moutstanding KFN common shares entitled to vote thereon present in person or represented by proxy at the KFN special meeting will to approve the adjournment of the KFN special meeting will

The KFN board of directors recommends that you vote "FOR" the adjournment of the KFN special meeting, if necessary additional proxies if there are not sufficient votes to adopt the merger agreement at the time of the KFN special meeting, if necessary additional proxies if there are not sufficient votes to adopt the merger agreement at the time of the KFN special meeting, if necessary additional proxies is the proximal prox

KFN SHAREHOLDER PROPOSALS

KFN will not hold an annual meeting of its shareholders in 2014 if the merger has been completed. If, however, KFN does annual meeting, under the rules of the SEC, in order to be considered for possible inclusion in the proxy statement for KFN's meeting, all KFN shareholder proposals must be received at KFN's principal executive offices, Attn: Corporate Secretary, 555 Calif 50th Floor, San Francisco, CA, 94104. KFN must have received all such shareholder proposals on or before November 19, 201 consider only proposals meeting the requirements of the applicable rules

For any proposal that is not submitted for inclusion in KFN's proxy material for the 2014 annual meeting but is instead presented directly at that meeting, Rule 14a-4(c) under the Exchange Act permits KFN management to exercise discretionary vot under proxies it solicits unless KFN receives timely notice of the proposal in accordance with the procedures set forth in Section KFN operating agreement. For a shareholder proposal or director nomination to be properly submitted for presentation at KFN's meeting, KFN's Corporate Secretary must have received written notice of the proposal or nomination at KFN's principal executive o the period beginning on October 20, 2013 and ending at 5:00 p.m., Eastern time, on November 19, 2013; provided, however, tha that the date of the annual meeting is advanced or delayed by more than 30 days from the first anniversary of the date of the preannual meeting, notice by the holder of shares to be timely must be so delivered not earlier than the 150th day prior to the date of meeting and not later than 5:00 p.m., Eastern time, on the later of the 120th day prior to the date of such annual meeting or following the day on which public announcement of the date of such meeting is first made. However, in the event the KFN board increases the number of directors in accordance with KFN's operating agreement and there is no public announcement of such a 130 days prior to the first anniversary of the date of mailing of the notice of the preceding year's annual meeting, a holder of share also be considered timely, but only with respect to nominees for any new positions created by such increase, if it shall be delive Corporate Secretary at KFN's principal executive office not later than 5:00 p.m., Eastern time, on the tenth day following the day of public announcement is first made by KFN. The public announcement of a postponement or an adjournment of an annual mee commence a new time period for the giving of a holder of shares' notice as desc

Shareholder proposals or director nominations must contain information specified in the KFN operating agreement and must all matters for shareholder action under KFN's operating agreement and under Delaware law. In addition, if a shareholder delivers a notice with respect to a proposal or director nomination, in order for such proposal or nomination to be properly brought before meeting, the shareholder must have actually delivered a proxy statement and form of proxy to holders of at least the percentar required to carry the proposal or elect the shareholder's nominee, as applicable. If no solicitation notice is delivered with respect to a nomination, in order for such proposal or nomination to be properly brought before the annual meeting, the shareholder or benefits the proposal or director nomination must not have solicited a number of proxies sufficient to have required the delivery of a number of proxies sufficient to have required the delivery of a number of proxies sufficient to have required the delivery of a number of proxies sufficient to have required the delivery of a number of proxies sufficient to have required the delivery of a number of proxies sufficient to have required the delivery of a number of proxies sufficient to have required the delivery of a number of proxies sufficient to have required the delivery of a number of proxies sufficient to have required the delivery of a number of proxies sufficient to have required the delivery of a number of proxies sufficient to have required the delivery of a number of proxies sufficient to have required the delivery of a number of proxies sufficient to have required the delivery of a number of proxies sufficient to have required the delivery of a number of proxies sufficient to have required the delivery of a number of proxies sufficient to have required the delivery of a number of proxies sufficient to have required the delivery of a number of proxies sufficient to have required the number of proxies and number of proxies and number of proxies and n

LEGAL MATTERS

The validity of the KKR common units to be issued in connection with the merger will be passed upon for KKR by Simpson Bartlett LLP, New York, New York, and Simpson Thacher & Bartlett LLP has opined as to certain U.S. federal income tax consequences ownership and disposition of KKR common units. Certain partners of Simpson Thacher & Bartlett LLP, members of their families persons have an interest representing less than 1% of the capital commitments of investment funds that KI

EXPERTS

The consolidated financial statements of KKR & Co. L.P. and subsidiaries incorporated in this proxy statement/prospectus from the Annual Report on Form 10-K for the year ended December 31, 2013 of KKR & Co. L.P. and subsidiaries, and the effective internal control over financial reporting of KKR & Co. L.P. and subsidiaries have been audited by Deloitte & Touche LLP, an registered public accounting firm, as stated in their report which is also incorporated herein by reference. Such financial statements incorporated in reliance upon the report of such firm given upon their authority as experts in accounting a

The consolidated financial statements of KKR Financial Holdings LLC and subsidiaries incorporated in this proxy statement/p reference from the Annual Report on Form 10-K for the year ended December 31, 2013 of KKR Financial Holdings LLC and subsidiaries of the internal control over financial reporting of KKR Financial Holdings LLC and subsidiaries have bee Deloitte & Touche LLP, an independent registered public accounting firm, as stated in their reports which are also incorporate reference. Such financial statements have been so incorporated in reliance upon the report of such firm given upon their authority accounting a

Table

WHERE YOU CAN FIND MORE INFORMATION

KKR has filed with the SEC a registration statement under the Securities Act of which this proxy statement/prospectus forms a registers the KKR common units to be issued to KFN shareholders in connection with the merger. The registration statement, i exhibits and schedules attached to the registration statement, contains additional relevant information about KKR and its common rules and regulations of the SEC allow KKR and KFN to omit certain information that is included in the registration statement from the statement f

KKR and KFN file annual, quarterly and special reports and other information with the SEC. KFN also files proxy statements were the SEC allows KKR and KFN to "incorporate by reference" into this proxy statement/prospectus the information they file with the means that they can disclose important information to you by referring you to those documents. The information incorporated by reimportant part of this proxy statement/ prospectus, and information that KKR or KFN files later with the SEC will automaticall supersede such information as well as the information included in this proxy statement/ prospectus. Some documents or information called for by Items 2.02 and 7.01 of Form 8-K, or the exhibits related thereto under Item 9.01 of Form 8-K, are deemed furnished and accordance with SEC rules. None of those documents and none of that information is incorporated by reference in statement/prospectus. This proxy statement/prospectus also contains summaries of certain provisions contained in some of the described herein, but reference is made to the actual documents for complete information. All of the summaries are qualified in their reference to the actual

KKR and KFN incorporate by reference the documents listed below and any future filings they make with the SEC under Se 13(c), 14 or 15(d) of the Securities Exchange Act of 1934 until the termination of the securities Exchange and 1934 until the termination of the securities Exchange Act of 1934 until the termination of the securities Ex

KKR's Filings (SEC File I

Annual Report on Form 10-K for the year ended Decemb

Current Reports on Form 8-K filed on February 6, 2014, February 19, 2014 and March 1'

Registration Statement on Form 8-A12B filed on Ju

KKR will provide a copy of any document incorporated by reference in this proxy statement/prospectus and any exhibit incorporated by reference in the documents it incorporates by reference, without charge, by written or oral request directed to following address and teleph

Attention: Inves

KK

9 West 57th Stree New York, New (87

Email: Investor-Relatio

KFN's Filings (SEC File I

Annual Report on Form 10-K for the year ended Decemb

Current Report on Form 8-K filed on January 3

Registration Statement on Form 8-A12B filed on Janua

Table

KFN will provide a copy of any document incorporated by reference in this proxy statement/ prospectus and any exhibit incorporated by reference in the documents it incorporates by reference, without charge, by written or oral request directed to following address and teleph

Attention: Inves 555 California Stre San Francisc (85

KKR Financial H

Email: KFN.

KKR and KFN also make available free of charge on their internet websites at www. http://ir.kkr.com/kfn_ir/kfn_overview.cfm, respectively, the reports and other information filed by KKR and KFN with the SE reasonably practicable after such material is electronically filed or furnished to the SEC. Neither KKR's nor KFN's websites, nor the contained on their websites, is part of this proxy statement/prospectus or the documents incorporated

The SEC maintains an Internet website that contains reports, proxy and information statements and other material that are filed SEC's Electronic Data Gathering, Analysis and Retrieval (EDGAR) System. This system can be accessed at www.sec.gov. Information that KKR and KFN file with the SEC by reference to their names or to their SEC file numbers. You also may read a document KKR or KFN files with the SEC at the SEC's public reference room.

100 F

Washington

Please call the SEC at 1-800-SEC-0330 for further information about the public reference room and its copy charges. KKR SE also available to the public through the New York Stock Exchange at 20 Broad Street, New York, New

The information concerning KKR contained in this proxy statement/prospectus or incorporated by reference has been provide and the information concerning KFN contained in this proxy statement/prospectus or incorporated by reference has been provided by the provided by the

In order to receive timely delivery of requested documents in advance of the KFN special meeting, your request should no later than April 23, 2014. If you request any documents, KKR or KFN will mail them to you by first class mail, or ano prompt means, within one business day after receipt of you

Neither KKR nor KFN has authorized anyone to give any information or make any representation about the merger, KKR or different from, or in addition to, that contained in this proxy statement/prospectus or in any of the materials that have been incomplete. Therefore, if any one distributes this type of information, you should not rely on it. If you are in a jurisdiction where exchange or sell, or solicitations of offers to exchange or purchase, the securities offered by this proxy statement/prospectus or the securities is unlawful, or you are a person to whom it is unlawful to direct these types or activities, then the offer presented statement/prospectus does not extend to you. The information contained in this proxy statement/prospectus speaks only as of its does of information in a document incorporated by reference, as of the date of such document, unless the information specifically in the date of such document, unless the information specifically in the date of such document.

Ехеси

AGREEMENT AND PLAN OF MERGER

by and among

KKR & CO. L.P.

KKR FUND HOLDINGS L.P.,

COPAL MERGER SUB LLC

and

KKR FINANCIAL HOLDINGS LLC

Dated as of December 16, 2013

TABLE OF CONTENTS

ARTICLE I	THE MERGER	Page
1.1	The Merger	<u>A-2</u>
1.2	Effective Time; Closing Date	<u>A-2</u>
<u>1.3</u>	Effects of the Merger	<u>A-2</u>
<u>1.4</u>	Organizational Documents and Directors and Officers of the Company	<u>A-2</u>
_		<u>A-2</u>
ARTICLE I	I EFFECT OF THE MERGER	<u>A-3</u>
2.1	Effect on Shares	
<u>2.2</u>	Exchange Ratio; Fractional Shares; Adjustments	<u>A-3</u>
<u>2.3</u>	Exchange of Shares	<u>A-4</u>
2.4	Treatment of Stock Options and Other Equity Based Awards	<u>A-5</u>
<u>2.5</u>	Withholding	<u>A-7</u>
		<u>A-8</u>
	II REPRESENTATIONS AND WARRANTIES OF THE PARENT PARTIES	<u>A-8</u>
3.1	Organization and Standing	<u>A-8</u>
3.2	Power and Authority	<u>A-8</u>
3.3	Capitalization of Parent and Merger Sub	<u>A-9</u>
<u>3.4</u>	Conflicts; Consents and Approvals	<u>A-9</u>
3.5	No Material Adverse Effect	<u>A-10</u>
3.6	Parent SEC Documents; Controls	<u>A-10</u>
<u>3.7</u>	Registration Statement; Proxy Statement	A-11
3.8	Litigation	<u>A-11</u>
<u>3.9</u>	Brokerage and Finder's Fees	<u>A-11</u>
3.10	Operations of Merger Sub	<u>A-12</u>
3.11	Compliance with Law	A-12
3.12	<u>Undisclosed Liabilities</u>	<u>A-12</u>
3.13	Operation of Parent's Business	
3.14	Investment Company Act	<u>A-12</u>
<u>3.15</u>	Status under Company Operating Agreement	<u>A-12</u>
		<u>A-12</u>

3.16	Knowledge of Parent Parties	<u>A-12</u>
ARTICLE IV	REPRESENTATIONS AND WARRANTIES OF THE COMPANY	
4.1		<u>A-12</u>
<u>4.1</u>	Organization and Standing	A-13
<u>4.2</u>	Subsidiaries	
4.3	Power and Authority	<u>A-13</u>
<u> 11.5</u>	1 cut and radion,	<u>A-13</u>
<u>4.4</u>	Capitalization of the Company	<u>A-14</u>
<u>4.5</u>	Conflicts; Consents and Approvals	<u>A-14</u>
		<u>A-15</u>

<u>4.6</u>	No Material Adverse Effect	Page <u>A-16</u>
4.7	Company SEC Documents; Controls	<u>A-16</u>
4.8	Compliance with Law	<u>A-17</u>
4.9	<u>Taxes</u>	<u>A-17</u>
4.10	<u>Title to Properties</u>	A-18
4.11	Registration Statement; Proxy Statement	A-18
4.12	Litigation	<u>A-18</u>
4.13	Brokerage and Finder's Fees	A-18
4.14	Benefit Plans	A-19
4.15	Contracts	
4.16	<u>Undisclosed Liabilities</u>	<u>A-19</u>
4.17	Operation of the Company's Business	<u>A-19</u>
4.18	Opinion of Financial Advisors	<u>A-19</u>
4.19	Takeover Statutes	<u>A-20</u>
4.20	Insurance	<u>A-20</u>
4.21	Investment Company Act	<u>A-20</u>
		<u>A-20</u>
ARTICLE V	MUTUAL COVENANTS OF THE PARTIES	<u>A-20</u>
<u>5.1</u>	Preparation of Proxy Statement; Shareholder Meeting	A-20 A-20
<u>5.2</u>	Access to Information; Confidentiality	
<u>5.3</u>	Filings: Reasonable Best Efforts; Notification	<u>A-21</u>
<u>5.4</u>	Tax Treatment	<u>A-22</u>
<u>5.5</u>	Public Announcements	<u>A-24</u>
<u>5.6</u>	Section 16 Matters	<u>A-24</u>
<u>5.7</u>	Transaction Litigation	<u>A-24</u>
<u>5.8</u>	Notification of Certain Matters	<u>A-24</u>
		<u>A-24</u>
ARTICLE VI	ADDITIONAL COVENANTS OF THE PARTIES	<u>A-25</u>
<u>6.1</u>	Conduct of Parent's Operations	A-25
<u>6.2</u>	Conduct of the Company's Operations	
<u>6.3</u>	Indemnification; Directors' and Officers' Insurance	A-26
<u>6.4</u>	Activities of Merger Sub	<u>A-28</u>

<u>6.5</u>	NYSE Listing		<u>A-29</u>
			<u>A-29</u>
<u>6.6</u>	Company Acquisition Proposals		<u>A-29</u>
<u>6.7</u>	Approvals		<u>A-32</u>
		ii	

<u>6.8</u>	Compliance with Agreement		Page <u>A-32</u>	
ARTICLE	ARTICLE VII CONDITIONS TO THE MERGER			
<u>7.1</u>	Conditions to the Obligations of Each Party		<u>A-32</u>	
<u>7.2</u>	Conditions to the Obligations of the Company		<u>A-32</u>	
<u>7.3</u>	Conditions to the Obligations of Parent Parties		<u>A-33</u>	
			<u>A-33</u>	
ARTICLE	VIII TERMINATION AND AMENDMENT		<u>A-34</u>	
<u>8.1</u>	Termination		<u>A-34</u>	
<u>8.2</u>	Effect of Termination		<u>A-35</u>	
<u>8.3</u>	Amendment		<u>A-37</u>	
<u>8.4</u>	Extension; Waiver		<u>A-37</u>	
ARTICLE	X MISCELLANEOUS			
9.1	Survival of Representations and Warranties		<u>A-38</u>	
9.2	Notices		<u>A-38</u>	
9.3	Interpretation		<u>A-38</u>	
9.4	<u>Counterparts</u>		<u>A-38</u>	
9.5	Entire Agreement		<u>A-41</u>	
9.6	<u>Severability</u>		<u>A-41</u>	
9.7	Third-Party Beneficiaries		<u>A-41</u>	
9.8	Governing Law		<u>A-41</u>	
9.9	Consent to Jurisdiction; Venue		<u>A-41</u>	
9.10	Specific Performance		<u>A-41</u>	
	WAIVER OF JURY TRIAL		<u>A-42</u>	
9.11			<u>A-42</u>	
9.12	Assignment		<u>A-42</u>	
9.13	Expenses :		<u>A-42</u>	
	1	iii		

INDEX OF DEFINED TERMS

	D
2007 Share Plan	Page A-7
424(b)(3) Prospectus	A-39
Action	A-11
affiliate	A-39
Agreement	A-1
Antitrust Laws	A-23
Applicable Law	A-10
Board Recommendation Notice	A-28
Book-Entry Share	A-5
Business Combination	A-37
business day	A-39
Cash Value	A-7
Certificate	A-5
Certificate of Merger	A-2
Closing Closing Data	A-2 A-2
Closing Date Code	A-39
Commission	A-39 A-10
Common Unit Trust	A-10 A-4
Company	A-1
Company Acquisition Agreement	A-30
Company Acquisition Proposal	A-32
Company Board	A-1
Company Board Recommendation	A-14
Company Certificate of Formation	A-13
Company Change of Recommendation	A-30
Company Common Shares	A-14
Company Disclosure Schedule	A-12
Company Operating Agreement	A-1
Company Option	A-7
Company Plan	A-19
Company SEC Documents	A-16
Company Securities	A-15
Company Shareholder Approval	A-14
Company Shareholder Meeting	A-21 A-32
Company Superior Proposal	A-32 A-22
Confidentiality Agreement Contract	A-22 A-10
Costs	A-10 A-36
DE LLC Act	A-1
Deferred Compensation Plan	A-3
Effective Time	A-2
End Date	A-35
ERISA	A-39
Event	A-40
Excess Units	A-4
Exchange Act	A-10
Exchange Agent	A-5
Exchange Fund	A-5
Exchange Ratio	A-4
Existing Company Credit Agreement	A-39
Fund Holdings	A-1
GAAP	A-39
Governmental Authority	A-10
Grant Date Group Partnerships	A-14
Group Partnerships HSR Act	A-22 A-10
KKR Financial Advisors	A-10 A-12
TANA T HIGHERAL AUVISOLS	A-12

knowledge	A-40	
Lien	A-40	
Management Agreement	A-40	
Management Holdings	A-9	
Material Adverse Effect	A-40	
Material Company Contract	A-19	
Maximum Premium	A-29	
Measurement Date	A-9	
Merger	A-1	
Merger Consideration	A-3	
Merger Sub	A-1	
Merger Sub Common Units	A-3	
NYSE	A-4	
Option Consideration	A-7	
Order	A-23	
Parent	A-1	
Parent Board	A-1	
Parent Common Units	A-1	
Parent Disclosure Schedule	A-8	
Parent Parties	A-1	
Parent SEC Documents	A-10	
Parent Securities	A-9	
parties	A-1	
pdf	A-41	
Per Share Amount	A-7	
Person	A-40	
Phantom Shares	A-3	
Preferred Shares	A-14	
Prospectus	A-11	
Proxy Statement	A-11	
Registration Statement	A-11	
Representatives	A-29	
Securities Act	A-10	
subsidiary	A-40	
Superior Proposal Notice	A-31	
Surviving Company Share	A-3	
Surviving Entity	A-2	
Tax Returns	A-18	
Taxes	A-18	
Termination Payment	A-36	
Transaction Committee	A-1	
Transaction Committee Financial Advisor	A-20	

AGREEMENT AND PLAN OF MERGER

This Agreement and Plan of Merger (this "Agreement") is made and entered into as of December 16, 2013, by and among KKR a Delaware limited partnership ("Parent"), KKR Fund Holdings L.P., an exempted limited partnership formed under the laws of Islands ("Fund Holdings"), Copal Merger Sub LLC, a Delaware limited liability company ("Merger Sub" and, taken togethe Holdings and Parent, the "Parent Parties"), and KKR Financial Holdings LLC, a Delaware limited liability company (the "Co together with the Parent Parties, the "Parent Parties").

WITNESSETH:

WHEREAS, the parties intend that Merger Sub be merged with and into the Company (the "Merger"), with the Company states as an indirect subsidiary of Parent, upon the terms and subject to the conditions set forth in this Agreement and the provided Liability Company Act of the State of Delaware (the "Delaware (the "Delawa

WHEREAS, the board of directors of the Company (the "Company Board") has appointed a committee of the Company Board only of independent and disinterested directors of the Company (the "Transaction Committee") to review and negotiate recommendation to the Company Board with respect to, the transactions contemplated by this Agreement, including

WHEREAS, the Company Board, acting upon the unanimous recommendation of the Transaction Committee, unanimousl Scott Nuttall and Craig Farr, who were not present at the meeting of the Company Board) has (i) determined that the transactions of by this Agreement, including the Merger, are fair to, and in the best interests of, the holders of the Company's Common Shares (as defined in the Company's Amended and Restated Operating Agreement, dated as of May 3, 2007 and as amended from time to time date hereof, the "Company Operating Agreement"), (ii) approved and declared advisable the execution, delivery and perform Agreement and the transactions contemplated herein, including the Merger, and (iii) resolved to recommend that the holders of the Common Shares approve

WHEREAS, the board of directors of the general partner of Parent (the "Parent Board"), the board of directors of Fund Hold sole member of Merger Sub have, on the terms and subject to the conditions set forth herein, approved and declared advisable the delivery and performance of this Agreement and the transactions contemplated herein, including

WHEREAS, by virtue of the Merger, the holders of the Company's Common Shares will receive a continuing equity in Company through the ownership of common units representing the limited partnership interests of Parent (such common units Comm

WHEREAS, immediately prior to the Closing, Parent will issue to Fund Holdings a number of Parent Common Unit distributed to the holders of the Company's Common Shares pursuant to Section 2.3 in connection with

WHEREAS, each Preferred Share that is outstanding immediately prior to the Merger shall remain outstanding as a Preferred Surviving Entity immediately following

WHEREAS, the Company's 7.500% Senior Notes due March 20, 2042 and the Company's 8.375% Senior Notes due Novem that are outstanding immediately prior to the Merger shall remain outstanding as indebtedness of the Surviving Entity immediately f

WHEREAS, the Company and the Parent Parties desire to make certain representations, warranties, covenants and a connection with this

NOW, THEREFORE, in consideration of these premises and the representations, warranties, covenants and agreements comes Agreement, the parties agreement, the parties agreement.

ARTICLE I

THE MERGER

1.1 The Merger. Upon the terms and subject to the conditions of this Agreement, and in accordance with the provide DE LLC Act, at the Effective Time, Merger Sub shall be merged with and into the Company and the separate limited liability company of Merger Sub shall cease, and the Company shall be the surviving limited liability company in the Merger (the "Surviving Entity become an indirect subsidial action of this Agreement, and in accordance with the provide of the provide subsidial action of this Agreement, and in accordance with the provide subsidial action of this Agreement, and in accordance with the provide subsidial action of this Agreement, and in accordance with the provide subsidial action of the Effective Time, Merger Sub shall be merged with and into the Company and the separate limited liability company of Merger Sub shall cease, and the Company shall be the surviving limited liability company in the Merger (the "Surviving Entity become an indirect subsidial action of the Company shall be the surviving limited liability company in the Merger (the "Surviving Entity become an indirect subsidial action of the Company shall be the surviving limited liability company in the Merger (the "Surviving Entity become an indirect subsidial action of the Company shall be the surviving limited liability company in the Merger (the "Surviving Entity become an indirect subsidial action of the Company shall be the surviving limited liability company in the Merger (the "Surviving Entity become an indirect subsidial action of the Company shall be the surviving limited liability company in the Merger (the "Surviving Entity become action of the Company shall be the surviving limited liability company in the Merger (the "Surviving Entity become action of the Company shall be the surviving limited liability company in the Merger (the "Surviving Entity become action of the Company shall be the surviving limited liability company in the Merger (the "Surviving Entity become action of the Company shall be the surviving limite

1.2 Effective Time; Clos

- (a) Subject to the provisions of this Agreement, as promptly as practicable on the Closing Date, the parties shall file be filed) with the Secretary of State of the State of Delaware a certificate of merger for the Merger, executed in accordar relevant provisions of the DE LLC Act (the "Certificate of Merger"), and shall make all other filings or recordings requir DE LLC Act in connection with the Merger. The Certificate of Merger will provide that the Merger shall become effective as the Certificate of Merger shall have been duly filed with the Secretary of State of the State of Delaware or at such late time as may be agreed by the parties and specified in the Certificate of Merger (the time at which the Merger become referred to herein as the "Effe."
- (b) Prior to the filing of the Certificate of Merger, a closing (the "Closing") shall be held at 10:00 a.m., New Yorl offices of Simpson Thacher & Bartlett LLP, 425 Lexington Avenue, New York, New York 10017, or at such other place a may agree in writing, (i) as soon as practicable (but in any event within three business days) following the date upon wh conditions set forth in Article VII have been satisfied or waived (other than those conditions that by their nature are to be waived at the Closing, but subject to the satisfaction or waiver of those conditions), or (ii) on such other date and/or at such as the parties may agree in writing so long as all of the conditions set forth in Article VII (other than those conditions nature are to be satisfied or waived at the Closing, but subject to the satisfaction or waiver of those conditions) have been waived at or prior to such date and/or time. The date on which the Closing takes place is referred to herein as the "Cl Subject to the satisfaction or waiver of all of the conditions set forth in Article VII, the parties agree to use reasonable be coordinate and cooperate in taking such actions so that the Closing will take place on the last business day of a calc
- 1.3 Effects of the Merger. The Merger shall have the effects set forth in the DE LLC Act. Without limiting the genforegoing, at the Effective Time, all the property, rights, privileges, powers and franchises of the Company and Merger Sub shall become the debts, liabilities and Surviving Entity, and all debts, liabilities and duties of the Company and Merger Sub shall become the debts, liabilities and
- 1.4 Organizational Documents and Directors and Officers of the Company. At the Effective Time, the certificate of form Company as in effect immediately prior to the Merger shall be the certificate of formation of the Surviving Entity, and the lim company agreement of the Company in effect immediately prior to the Merger shall be the limited liability company agreement of the Entity; in each case, until such organizational document is amended in accordance with Applicable Law and as provided in such or document. The officers of the Company immediately prior to the Effective Time shall be the officers of the Surviving Entity. Each of shall take all necessary action to cause those individuals designated in writing by Pare

prior to the Closing Date to be the members of the board of directors of the Surviving Entity immediately following the Effective their respective successors are duly elected, appointed or qualified or until their earlier death, resignation or removal in accordance limited liability company agreement of the Surviving Entity and App

ARTICLE II

EFFECT OF THE MERGER

2.1 Effect or

- (a) At the Effective Time, by virtue of the Merger and without any further action on the part of the Company, any Pa the holder of any securities of the Company or any
 - (i) Treatment of Merger Sub Common Units. Each common unit representing membership interests of (the "Merger Sub Common Units") issued and outstanding immediately prior to the Effective Time shall be one validly issued, fully paid and nonassessable Common Share of the Surviving Entity (each such Common with each Common Share of the Surviving Entity issued pursuant to Section 2.1(a)(ii), a "Surviving Company S that immediately following the Effective Time, such newly issued Surviving Company Shares shall constit issued and outstanding capital stock of the Surviving Entity other than the Preference
 - (ii) Treatment of Certain Company Common Shares. Each Company Common Share issued and immediately prior to the Effective Time that is held by Fund Holdings or any subsidiary of Fund Holdings shall outstanding and shall automatically be cancelled and retired and shall cease to exist, and no payment shall be respect thereto. Each Company Common Share issued and outstanding immediately prior to the Effective Tim by any wholly-owned subsidiary of the Company shall remain outstanding and be converted into Common Surviving Entity representing the same percentage ownership of the Surviving Entity as such Company Corepresented in the Company immediately prior to the Effective Times of the Surviving Entity as such Company Corepresented in the Company immediately prior to the Effective Times of the Surviving Entity as such Company Corepresented in the Company immediately prior to the Effective Times of the Surviving Entity as such Company Corepresented in the Company immediately prior to the Effective Times of the Surviving Entity as such Company Corepresented in the Company immediately prior to the Effective Times of the Surviving Entity as such Company Corepresented in the Company immediately prior to the Effective Times of the Surviving Entity as such Company Corespondent to the Effective Times of the Surviving Entity as such Company Corepresented in the Company immediately prior to the Effective Times of the Surviving Entity as such Company Corepresented to the Effective Times of the Surviving Entity as such Company Corepresented to the Effective Times of the Surviving Entity as such Company Corepresented to the Effective Times of the Surviving Entity as such Company Corepresented to the Effective Times of the Surviving Entity as such Company Corepresented to the Effective Times of the Surviving Entity as such Corepresented to the Effective Times of the Surviving Entity as such Corepresented to the Effective Times of the Surviving Entity as such Corepresented to the Effective Times of the Surviving Entity as such Cor
 - (iii) Conversion of Company Common Shares. Each Company Common Share issued and outstanding prior to the Effective Time (other than the Company Common Shares described in Section 2.1(a)(ii)) shall autoconverted into the right to receive a number of validly issued Parent Common Units equal to the Exchange Rawith cash in lieu of fractional Parent Common Units, if any, payable pursuant to Section 2.2(b) (collectively, Consideration"). Each Company Common Share, when so converted pursuant to this Section 2.1(a)(iii), shall outstanding and shall be cancelled and shall or
- (b) No Effect on Phantom Shares. Each phantom share (the "Phantom Shares") credited to a participant's according to the "Deferred Compensation and Share Award Plan (the "Deferred Compensation") shall, immediately upon the Effective Time, automatically be converted into the right to receive a number of volume Parent Common Units equal to the Exchange Ratio (with fractional Parent Common Units to be aggregated and rounded to whole unit), but shall otherwise remain outstanding following the Effective Time and be distributed pursuant to the Deferred Compensation of the Parent Compensation Parent Compensation and Share Award Plan (the "Deferred Compensation") shall, immediately upon the Effective Time and be aggregated and rounded to whole unit), but shall otherwise remain outstanding following the Effective Time and be distributed pursuant to the Deferred Compensation and Share Award Plan (the "Deferred Compensation") shall, immediately upon the Effective Time, automatically be converted into the right to receive a number of volume and the Effective Time and be distributed pursuant to the Deferred Compensation and Share Award Plan (the "Deferred Compensation") shall, immediately upon the Effective Time and be distributed pursuant to the Parent Compensation and Share Award Plan (the "Deferred Compensation") shall, immediately upon the Effective Time and be distributed pursuant to the Parent Compensation and Share Award Plan (the "Deferred Compensation") shall be presented by the Parent Compensation and Share Award Plan (the "Deferred Compensation") shall be presented by the Parent Compensation and Share Award Plan (the "Deferred Compensation") shall be presented by the Parent Compensation and Share Award Plan (the "Deferred Compensation") shall be presented by the Parent Compensation and Parent Compensation and Parent Compensation and Parent Compensation and Parent Com
 - (c) No Effect on Preferred Shares. Each Preferred Share issued and outstanding immediately prior to the Effective remain an outstanding Preferred Share of the Surviving Entity following the Effective remains an outstanding Preferred Share of the Surviving Entity following the Effective remains an outstanding Preferred Share of the Surviving Entity following the Effective remains an outstanding Preferred Share of the Surviving Entity following the Effective remains an outstanding Preferred Share of the Surviving Entity following the Effective remains an outstanding Preferred Share of the Surviving Entity following the Effective remains an outstanding Preferred Share of the Surviving Entity following the Effective remains an outstanding Preferred Share of the Surviving Entity following the Effective remains an outstanding Preferred Share of the Surviving Entity following the Effective remains an outstanding Preferred Share of the Surviving Entity following the Effective remains an outstanding Preferred Share of the Surviving Entity following the Effective remains an outstanding Preferred Share of the Surviving Entity following the Effective remains an outstanding Preferred Share of the Surviving Entity following the Effective remains an outstanding Preferred Share of the Surviving Entity following the Effective remains an outstanding Preferred Share of the Surviving Entity following the Effective Remains and Preferred Share of the Surviving Entity following the Effective Remains and Preferred Share of the Surviving Entity following the Effective Remains and Preferred Share of the Surviving Entity following the Effective Remains and Preferred Share of the Surviving Entity following the Effective Remains and Preferred Share of the Effective Remains a

- 2.2 Exchange Ratio; Fractional Shares; Adj
- (a) The "Exchange Ratio" shall be equal to 0.51, subject to adjustment in accordance with Se
- (b) No fractional Parent Common Unit shall be issued in the Merger as a result of the conversion processions 2.1(a)(iii) and 2.1(b), but in lieu thereof each Person otherwise entitled to a fractional Parent Common Unit shall be receive, from the Exchange Agent in accordance with the provisions of this Section 2.2(b), a cash payment in lieu of surparent Common Unit representing such Person's proportionate interest, if any, in the proceeds from the sale by the Exc (reduced by any fees of the Exchange Agent attributable to such sale) in one or more transactions of Parent Common Units excess of (i) the aggregate number of Parent Common Units to be delivered to the Exchange Agent by Fund Holding Section 2.3(a) over (ii) the aggregate number of whole (non-fractional) Parent Common Units to be distributed Section 2.3(b) (such excess being the "Excess Units"), which such proportionate interest shall be paid as soon as reasonably following such transaction or transactions. The parties acknowledge that payment of the cash consideration in lieu of issuit Parent Common Units was not separately bargained-for consideration but merely represents a mechanical rounding off for avoiding the expense and inconvenience that would otherwise be caused by the issuance of fractional Parent Common Units, shall sell the Excess Units at then prevailing market prices on the New York Stock Exchange ("NYSE") in provided in Section Section 2.3(a) and 2.1(b), but in lieu of issuit provided in Section 2.2(b), a cash payment in lieu of such that the prevailing market prices on the New York Stock Exchange ("NYSE") in provided in Section 2.2(b), a cash payment in lieu of such that the provided in Section 2.2(b), a cash payment in lieu of such that the provided in Section 2.2(b), a cash payment in lieu of such that the provided in Section 2.2(b), a cash payment in lieu of such that the provided in Section 2.2(b), a cash payment in lieu of such that the provided in Section 2.2(b), a cash payment in lieu of such that the provided in Section 2.2(b), a
- (c) The sale of the Excess Units by the Exchange Agent shall be executed on the NYSE at then-prevailing mark shall be executed in round lots to the extent practicable. The Exchange Agent shall hold such proceeds in trust for the would have otherwise received fractional Parent Common Units (the "Common Unit Trust"). The Exchange Agent shall be portion of the Common Unit Trust to which each such Person shall be entitled by multiplying the amount of the aggreg comprising the Common Unit Trust by a fraction, the numerator of which is the amount of the fractional Parent Common to which such Person would otherwise be entitled and the denominator of which is the aggregate amount of fractional Parent Unit interests to which all such Persons would otherwise
- (d) In the event that, prior to the Effective Time, the Company or Parent shall declare a stock dividend or other payable in Company Common Shares or Parent Common Units, as applicable, or securities convertible, exercisable or e into Company Common Shares or Parent Common Units, as applicable, or effect a stock split, reclassification, combination of shares (other than, for the avoidance of doubt, any regular quarterly issuances of Parent Common Units made accordance with its quarterly exchange process) or other similar change with respect to the Company Common Shares (other than, as applicable, the Exchange Ratio set forth in Section 2.2(a) shall be adjusted to reflect fully the approprisched dividend, distribution, stock split, reclassification, combination, exchange of shares (other than, for the avoidance of regular quarterly issuances of Parent Common Units made by Parent in accordance with its quarterly exchange process.

2.3 Exchange of

(a) Exchange Agent. Prior to the Effective Time, American Stock Transfer and Trust Company, LLC (commercial bank, trust company or transfer agent that is mutually selected by Fund Holdings and the Company) shall act agent (the "Exchange Agent") for the delivery of the Merger Consideration pursuant to Section 2.1(a)(iii) and the Park Units pursuant to Section 2.1(b). At or prior to the Effective Time, Fund Holdings shall deposit with the Exchange Agent to feach Person entitled to payment pursuant to Sections 2.1(a)(iii) and 2.1(b), for exchange in accordance with the and in accordance with an exchange agent agreement with Fund Holdings in form and substance reasonably approved by the prior to the Effective Time, (i) the Parent Common Units to be delivered pursuant to Sections 2.1(a)(iii) and 2.1(b) and (in of cash in U.S. dollars sufficient to pay unpaid dividends and distributions, if any, payable pursuant to Section 2.3(c)

Common Units, together with any cash for unpaid dividends or distributions, if any, with respect thereto, are reference.

(b) Exchange Procedures. As soon as reasonably practicable after the Effective Time, the Surviving Entity sh Exchange Agent to mail to each holder, if any, of a certificate (a "Certificate") which immediately prior to the Ef represented outstanding Company Common Shares that were converted into the right to receive the Merger Consideration Section 2.1(a)(iii): (i) a letter of transmittal (the form and substance of which shall have been reasonably approved by t prior to the Effective Time and which shall specify that delivery shall be effected, and risk of loss and title to the Certificat only upon delivery of the Certificates to the Exchange Agent and shall be in such form and have such other customary Fund Holdings may reasonably specify) and (ii) instructions for effecting the surrender of the Certificates in exchange for Consideration and unpaid dividends and distributions, if any, payable pursuant to Section 2.3(c). Upon proper surrender of for cancellation to the Exchange Agent, together with a letter of transmittal, duly completed and validly executed in acc the instructions thereto, and such other documents as may reasonably be required by the Exchange Agent, the holder of suc shall be entitled to receive in exchange therefor: (A) that whole number of Parent Common Units which such holder ha receive pursuant to Section 2.1(a)(iii) in such denominations and registered in such names as such holder may reasonably (B) a check representing the amount of cash in lieu of fractional Parent Common Units, if any, and unpaid dividends and if any, which such holder has the right to receive pursuant to the provisions of this Article II, after giving effect to withholding tax. Any Certificate so surrendered shall immediately be cancelled. Until surrendered as contemplated by this each Certificate shall be deemed at any time after the Effective Time to represent only the right to receive upon surrende Consideration and unpaid dividends and distributions, if any, as provided in this Article II, after giving effect to withholding tax. The Surviving Entity, without any action on the part of any such Person, shall cause the Exchange Agen as of the Effective Time, to each holder of uncertificated Company Common Shares (any such Company Com 'Book-Entry Share") that whole number of Parent Common Units that each such Person is entitled to receiv Section 2.1(a)(iii) and (y) mail to each such Person, as soon as reasonably practicable after the Effective Time, a check rep amount of cash in lieu of fractional Parent Common Units, if any, and unpaid dividends and distributions, if any, which has the right to receive pursuant to the provisions of this Article II, after giving effect to any required withholding tax. No be paid or accrued on the cash in lieu of fractional Parent Common Units, if any, and unpaid dividends and distribu payable to holders of the Company Shares. In the event of a transfer of ownership of Company Common Shares that is n on the transfer records of the Company, the proper whole number of Parent Common Units, together with a check for t

fractional Parent Common Units, if any, and unpaid dividends and distributions, if any, may be issued to such tradocuments required to evidence and effect such transfer and to evidence that any applicable transfer Taxes have been prespect to Company Common Shares represented by a Certificate, the Certificate held by such transferee, are presented by a Certificate held by such transferee, are presented by a Certificate held by such transferee.

- (c) Distributions with Respect to Unexchanged Shares. The parties shall coordinate so that holders of the Compa Shares will not receive in respect of the same calendar quarter both (i) a quarterly distribution in respect of such holde Common Shares and (ii) a quarterly distribution in respect of the Parent Common Units issuable pursuant to this Agreement the immediately foregoing sentence, whenever a dividend or distribution is declared or made after the date of this Agi respect to Parent Common Units with a record date after the Effective Time, such declaration shall include a dividend or d respect of all Parent Common Units issuable pursuant to this Agreement. Notwithstanding any other provisions of this A dividends or other distributions declared or made after the Effective Time with respect to Parent Common Units having after the Effective Time shall be paid to the holder of any unsurrendered Certificate, and no cash payment in lieu of frac Common Units shall be paid to any such holder, until such holder shall surrender such Certificate as provided in this Subject to the effect of Applicable Law of the State of Delaware, following surrender of any such Certificate, there shall be holder of the Parent Common Units issued in exchange therefor, without interest, (i) at the time of such surrender, the dividends or other distributions with a record date after the Effective Time theretofore payable with respect to such Par Units and not paid, less the amount of any withholding taxes which may be required thereon, and (ii) at the appropriate part of the control o subsequent to surrender, the amount of dividends or other distributions with (x) a record date after the Effective Timsurrender and (y) a payment date subsequent to surrender that are payable with respect to such Parent Common Units, les of any withholding taxes which may be requ
- (d) No Further Ownership Rights in Company Common Shares. All Parent Common Units issued upon Certificates or cancellation of Book-Entry Shares in accordance with the terms of this Agreement (including any cash pair this Article II) shall be deemed to have been issued in full satisfaction of all rights pertaining to the Company Correpresented by such Certificates and Book-Entry Shares, and there shall be no further registration of transfers on the transfer the Company of Company Common Shares outstanding immediately prior to the Effective Time. If, after the Effective Time are presented to the Surviving Entity for any reason, they shall be cancelled and exchanged as provided in the
- (e) Lost Certificates. In the event that any Certificate shall have been lost, stolen or destroyed, upon the holder's with the replacement requirements established by the Exchange Agent, including the making of an affidavit to that necessary, the posting by the holder of a bond in customary amount as indemnity against any claim that may be mad Exchange Agent or the Surviving Entity with respect to such Certificate, the Exchange Agent shall deliver in exchange stolen or destroyed Certificate, the Merger Consideration and any dividends and distributions which such holder has the rig pursuant to the
 - (f) Termination of Exchange Fund. Any portion of the Exchange Fund (including any interest or other amounts r respect thereto) that remains undistributed six months after the date of the mailing required by Section 2.3(b) shall be del Surviving Entity, upon demand, and any holder of Certificates who has not theretofore complied with this Section 2.3 sh look only to the Surviving Entity for satisfaction of its clair

Consideration and any dividends and distributions which such holder has the right to receive pursuant to the

- (g) No Liability. None of Parent, Fund Holdings, the Surviving Entity or the Exchange Agent shall be liable to a respect of any portion of the Exchange Fund or the Merger Consideration delivered to a public official pursuant to ar abandoned property, escheat or similar Applicable Law. If any Certificates shall not have been surrendered prior to seve the Effective Time (or immediately prior to such earlier date on which any cash, any cash in lieu of fractional Parent Commany dividends or distributions with respect to whole Parent Common Units in respect of such Certificate would otherwise become the property of any Governmental Authority), any such cash, dividends or distributions in respect of such Certificate would clear of all claims or in Person previously ent
- (h) Investment of Exchange Fund. The Exchange Agent shall invest any cash included in the Exchange Fund directed by Fund Holdings, on a daily basis; provided, that no losses on such investment will affect the cash payable to for of Company Common Shares pursuant to the provisions of this Article II. Any interest and other income resulting investments shall be paid to Fund Holdings upon termination of the Exchange Fund pursuant to St.

2.4 Treatment of Stock Options and Other Equity Based

- (a) As of the Effective Time, by virtue of the Merger and without any action on the part of the holders thereof (whether vested or unvested) to purchase Company Common Shares that is outstanding immediately prior to the Effective a "Company Option") shall be cancelled, and the holder thereof shall be entitled to receive an amount in cash equal to any, of (1) the Per Share Amount over (2) the exercise price per Company Common Share subject to such Company Option immediately prior to the Effective Time (the "Option Consideration"). The Option Consideration shall be paid in a lump as practicable after the Effective Time but in no event later than 10 business days following the Effective Time. For pur paragraph, "Per Share Amount" means the Cash Value of the number of Parent Common Units (including any frac Common Units) that a holder of one Company Common Share would be entitled to receive pursuant to Section 2.1(a Exchange Ratio described thereunder, and "Cash Value" means, with respect to a Parent Common Unit, the average closi Parent Common Unit over the 10 trading day period ending on the trading day immediately preceding the Closing Date. that any Company Option has an exercise price per Company Common Share that is greater than the Per Share A Company Option shall be cancelled without payment in respect thereof as of the Effective Time, without liability to the Parent or any of the Parent o
- (b) At the Effective Time, each restricted Company Common Share (other than any restricted Company Common Share (Star Financial Advisors) issued and outstanding under the 2007 Share Incentive Plan for KKR Financial Holdings LLC, from time to time (the "2007 Share Plan") shall, by virtue of the Merger and without any action on the part of the holder converted into a number of restricted Parent Common Units (having the same terms and conditions, including applied requirements, as applied to such restricted Company Common Share immediately prior to the Effective Time) that Common Share would be exchanged for in accordance with Section 2.1(a)(iii) (with fractional Parent Common Units to be and rounded to the nearest whole unit), and Parent shall assume the 2007 Share Plan and any award agreement issue pursuant to which any such restricted Company Common Share has been granted in order to provide for the forest

Effective Time, any restricted Company Common Shares held by KKR Financial Advisors shall automatically be cancelle

- (c) Prior to the Effective Time, the Company and the Company Board and the compensation committee of the Compon the one hand, and Parent and the Parent Board, on the other hand, shall take all actions necessary to effectuate the prov
- 2.5 Withholding. The Company (or the Surviving Entity after the Effective Time), Parent, the Exchange Agent and any other withholding agent shall be entitled to deduct and withhold from the consideration otherwise payable pursuant to this Agreement suclare required to be deducted and withheld with respect to the making of such payment under applicable Tax law. To the extent that an withheld and timely paid over to the applicable taxing authority, such withheld amounts shall be treated for all purposes of this A having been paid to the applicable Person in respect of which such deduction and withholding

ARTICLE III

REPRESENTATIONS AND WARRANTIES OF THE PARENT PARTIES

In order to induce the Company to enter into this Agreement, subject to the exceptions set forth in the disclosure schedule Parent to the Company and dated as of the date of this Agreement (the "Parent Disclosure Schedule") (which exceptions shall identify a Section to which such exception relates, it being understood and agreed that each such exception shall be deemed to be di under such Section and any other Section of this Agreement (other than Section 3.5) to which it is reasonably apparent on its face sucrelates) and subject to the information disclosed by Parent in any Parent SEC Document filed during the period from December 31, 2 the business day prior to the date of this Agreement (other than in any risk factor or other cautionary or forward-looking disclosure such Parent SEC Document), the Parent Parties represent and warrant to the Compan

- 3.1 Organization and Standing. Each Parent Party is a legal entity duly organized, validly existing and in good standing ur of its respective jurisdiction of organization with all requisite power and authority to own, lease, use and operate its properties and to business as and where now owned, leased, used, operated and conducted. Each Parent Party is duly qualified to do business and in go in each jurisdiction in which the nature of the business conducted by it or the property it owns, leases or operates, makes such necessary, except where the failure to be so qualified or in good standing in such jurisdiction would not, individually or in the constitute a Material Adverse Effect with respect to Parent. Except as would not have an adverse effect on the ability of any Paconsummate any of the transactions contemplated by this Agreement, no Parent Party is in default of the performance, observance of any provision of its organizational documents. Parent has furnished to the Company prior to the date hereof a complete and correct organizational documents of each beginning to the party of the date hereof and correct organizational documents of each beginning to the party of the date hereof and correct organizational documents.
- 3.2 Power and Authority. Each Parent Party has all requisite power and authority to enter into and deliver this Agreement to the adoption of this Agreement by Fund Holdings in its capacity as the sole member of Merger Sub with respect to the Merger, to obligations under this Agreement and to consummate the transactions contemplated by this Agreement. The execution and de Agreement and the consummation of the transactions contemplated hereby by the Parent Parties have been duly authorized by all reconsumer of each Parent Party, subject to the adoption of this Agreement by Fund Holdings in its capacity as the sole member of with respect to the Merger. This Agreement has been duly executed and delivered by each Parent Party, and, assuming due a execution and delivery by the Company, constitutes the legal, valid and binding obligation of each Parent Party enforceable against in accordance with its terms except that the enforcement hereof may be

(i) bankruptcy, insolvency, reorganization, moratorium or other similar laws, now or hereafter in effect, relating to creditors' rights g
 (ii) general principles of equity (regardless of whether enforceability is considered in a proceeding in equity or at law). Other than the this Agreement by Fund Holdings in its capacity as the sole member of Merger Sub with respect to the Merger, no vote of the hoclass or series of capital stock (or other securities) of any Parent Party is necessary to approve the transactions contemplated by this

3.3 Capitalization of Parent and Me

- (a) The number of Parent Common Units that may be issued by Parent is not limited by Parent's organizationa
- (b) As of December 12, 2013 (the "Measurement Date"), 288,143,327 Parent Common Units were outstandi Measurement Date to the date of this Agreement, there have been no issuances of Parent Common Units except upon th exercise of awards to employees and service providers of Parent or its affiliates. The outstanding Parent Common Units ha authorized and validly issued and are free of any preemptive rights and were issued in compliance in all material Applicable Law. Except as set forth in this Section 3.3(b) or in Section 3.3(b) of the Parent Disclosure Schedule, as of th Agreement, there are no (i) outstanding equity securities of Parent, (ii) securities of Parent or its subsidiaries conve exchangeable for equity securities of Parent, (iii) options, warrants, calls, phantom stock or other rights to acquire from subsidiaries, or obligation of Parent or its subsidiaries to issue, any equity securities or securities convertible into or exch equity securities of Parent (the items in clauses (i), (ii) and (iii) are referred to collectively as "Parent Securities"), or (iv) obligations of Parent or any subsidiary of Parent to repurchase, redeem or otherwise acquire any Parent Securities. Exc forth in Section 3.3(b) of the Parent Disclosure Schedule, or (y) in connection with the repurchase or acquisition of Par Units pursuant to the terms of any equity incentive plan of Parent, neither Parent nor any of its subsidiaries is a party to that (i) obligates Parent or any of its subsidiaries to repurchase, redeem or otherwise acquire any Parent Securities, (ii) voting or transfer of, requires registration of, or grants any preemptive rights, anti-dilutive rights, rights of first refusal or rights with respect to, any Parent Securities, or (iii) otherwise relates to, creates, establishes or defines the terms and condi-Parent Securities. Since the Measurement Date, Parent has not granted, committed to grant or otherwise created or obligation with respect to any Parent Securities, other than as permitted by Section 6.1 or as set forth in Section 3.3(b) Disclosu
 - (c) All outstanding Merger Sub Common Units are duly authorized, validly issued and free of preemptive rights an record by Fund Holdings (except for any outstanding Merger Sub Common Units assigned to KKR Management Holdings") by Fund Holdings pursuant to Section 9.12) free and clear of any Liens. The Merger Sub Corepresent all of Merger Sub's members
 - 3.4 *Conflicts; Consents and Approvals.* Neither the execution and delivery of this Agreement by any Parent consummation by the Parent Parties of the Merger and the other transactions contemplated
 - (a) conflict with, or result in a breach of any provision of, the organizational documents of any Parent Party, subject of this Agreement by Fund Holdings in its capacity as the sole member of Merger Sub with respect to
- (b) violate, or conflict with, or result in a breach of any provision of, or constitute a default (or an event which, with notice, the passage of time or otherwise, would constitute a default) under, or entitle any party (with the giving of notice, the time or otherwise).

terminate, accelerate, adversely modify or call a default under, or result in the creation of any Lien upon any of the proper of Parent or any of its subsidiaries under, any of the terms, conditions or provisions of any note, bond, mortgage, inden trust, license, contract, undertaking, agreement, lease or other instrument (each, a "Contract") to which Parent or any of its

- (c) assuming compliance with the matters referred to in Section 3.4(d), violate any applicable laws, statutes, rules, legally binding policies or guidelines promulgated, or judgments, decisions or orders entered by any Governmental Autho laws, statutes, orders, rules, regulations, policies, guidelines, judgments, decisions and orders, collectively, "Apploapplicable to Parent or any of its subsidiaries or any of their respective properties
- (d) require any action or consent or approval of, or review by, or registration or filing by Parent or any of its subsitional third party or any local, domestic, foreign or multi-national court, arbitral tribunal, mediator, administrative agency or or other governmental or regulatory body, agency, instrumentality or authority (a "*Governmental Authority*" (i) authorization for inclusion of the Parent Common Units to be delivered pursuant to this Agreement on the NYSE, subject notice of issuance, (ii) any notifications required by the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amend with the rules and regulations thereunder, the "*HSR Act*"), (iii) registrations or other actions required under foreign, for Antitrust Laws or securities or blue-sky laws, (iv) the filing of the Certificate of Merger with the Secretary of State of Delaware, and (v) consents or approvals of, or notifications to, any Governmental Authority set forth in Section 3.4(d)

except in the case of clauses (b), (c) and (d) above for any of the foregoing that would not, individually or in the aggregate, constituted Adverse Effect with respect to Parent or a material adverse effect on the ability of any Parent Party to consummate the transactions of the by this

3.5 No Material Adverse Effect. Since December 31, 2012 through the date of this Agreement, there have been no individually or in the aggregate, constitute a Material Adverse Effect with respect to the control of the

3.6 Parent SEC Documents;

(a) Parent has timely filed or received the appropriate extension of time within which to file with the United States S Exchange Commission (the "Commission") all forms, reports, schedules, statements and other documents required to since January 1, 2010 under the Securities Exchange Act of 1934, as amended (together with the rules and regulations the "Exchange Act") or the Securities Act of 1933, as amended (together with the rules and regulations thereunder, the "Sec (such documents, as supplemented and amended since the time of filing, collectively, the "Parent SEC Documents"). The Documents, including any financial statements or schedules included therein, at the time filed (and, in the case or statements, on the dates of effectiveness) (i) did not contain any untrue statement of a material fact or omit to state a required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under whimade, not misleading and (ii) complied as to form in all material respects with the applicable requirements of the Exchange Securities Act, as the case may be. The financial statements of Parent included in the Parent SEC Documents were accordance with GAAP during the periods involved (except as may be indicated in the notes thereto or, in the case statements, as permitted by Form 10-Q of the Commission), and fairly present in all material respects (subject in the case statements to normal, recurring audit adjustments) the consolidated financial position of Parent and its consolidated substatements to normal, recurring audit adjustments) the consolidated financial position of Parent and its consolidated substatements to normal, recurring audit adjustments) the consolidated financial position of Parent and its consolidated substatements to normal, recurring audit adjustments) the consolidated financial position of Parent and its consolidated substatements.

the dates thereof and the consolidated results of their operations and cash flows for the periods then ended. No subsidiary subject to the periodic reporting requirements of the Exchange Act other than as part of Parent's consolidated results of the Exchange and the Exchange are the periodic reporting requirements of the Exchange are the periodic reporting reporting

- (b) Parent maintains a system of internal control over financial reporting (within the meaning of Rules 13a-15(f) a promulgated under the Exchange Act) designed to provide reasonable assurances regarding the reliability of financial reporting preparation of financial statements for external purposes in accordance with GAAP. Parent (i) maintains disclosure procedures (within the meaning of Rules 13a-15(e) and 15d-15(e) promulgated under the Exchange Act) designed to information required to be disclosed by Parent in the reports that it files and submits under the Exchange Act is recorded summarized and reported within the time periods specified in the Commission's rules and forms, including that information be disclosed by Parent in the reports that it files and submits under the Exchange Act is accumulated and commanagement of Parent as appropriate to allow timely decisions regarding required disclosure, and (ii) has disclosed, bath most recent evaluation by the Co-Chief Executive Officers and Chief Financial Officer of the general partner of Parent internal control over financial reporting, to its auditors and the audit committee of the Parent Board (A) all significant definition adversely affect its ability to record, process, summarize and report financial data and (B) any fraud, whether or not involves management or other employees who have a significant role in Parent's internal control over financial control over financ
- 3.7 Registration Statement; Proxy Statement. None of the information supplied or to be supplied by any Parent Party for incorporation by reference in the registration statement on Form S-4 (such registration statement as it may be amended or supplied including any such amendments or supplements, the "Registration Statement") to be filed with the Commission by Parent under the Act, including the prospectus relating to the Parent Common Units to be delivered pursuant to this Agreement (as it may be supplemented and including any such amendments or supplements, the "Prospectus") and the proxy statement and form of proxy reaction Company Shareholder Meeting (as it may be amended, supplemented or modified and including any such amendments or supplements of the Company Statement, at the date of meeting the date of the Company Shareholder Meeting, will contain any untrue statement of a material fact or omit to state a material fact restated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not The Registration Statement, except for such portions thereof that relate only to the Company and its subsidiaries, will comply as to material respects with the provisions of the Securities Act and the Experiment of the Securities Act and the Experiment of the Securities and the Securities Act and the Experiment of the Securities Act and the Securities Act a
- 3.8 Litigation. There is no suit, claim, action, proceeding, litigation, arbitration, mediation or investigation (an "Action") p the knowledge of Parent, threatened against Parent or any of its subsidiaries or their respective officers or directors that would, individually the aggregate, constitute a Material Adverse Effect with respect to Parent. Neither Parent nor any of its subsidiaries nor any ass properties of such Persons is subject to any outstanding order, writ, injunction or decree that would, individually or in the aggregate Material Adverse Effect with respective.
 - 3.9 Brokerage and Finder's Fees. Except for Parent's obligations to Goldman, Sachs & Co. and Lazard Frères & Co. Parent, any of its subsidiaries nor any of their respective equityholders, directors, officers or employees has incurred or will incur Parent or its subsidiaries any brokerage, finder's, financial advisor's or similar fee or commission in connection with the contemplated by this

- 3.10 Operations of Merger Sub. Merger Sub is a direct, wholly-owned subsidiary of Fund Holdings, was formed solely for of engaging in the transactions contemplated by this Agreement, has engaged in no other business activities and has conducted i only as contemplated by this
- 3.11 Compliance with Law. Parent and its subsidiaries are in compliance with, and at all times since January 1, 2010 compliance with, all Applicable Law relating to Parent, its subsidiaries or their respective business or properties, except where the fa compliance with such Applicable Law would not, individually or in the aggregate, constitute a Material Adverse Effect with respective No investigation or review by any Governmental Authority with respect to Parent or its subsidiaries is pending or, to the knowled threatened, nor has any Governmental Authority indicated in writing an intention to conduct the same, in each case other than those of which would not, individually or in the aggregate, constitute a Material Adverse Effect with respective No. 11.
- 3.12 Undisclosed Liabilities. Except (a) as and to the extent disclosed or reserved against on the unaudited consolidated by of Parent as of September 30, 2013 included in the Parent SEC Documents, (b) as incurred after the date thereof in the ordina business consistent with past practice, or (c) as set forth in Section 3.12 to the Parent Disclosure Schedule, Parent, together with its does not have any liabilities or obligations of any nature, whether known or unknown, absolute, accrued, contingent or otherwise due or to become due in each case required by GAAP to be reflected or reserved against in the consolidated balance sheet of F subsidiaries (or disclosed in the notes to such balance sheet), that would, individually or in the aggregate, constitute a Material Account of the parent SEC Documents, (b) as incurred after the date thereof in the ordinate business consistent with past practice, or (c) as set forth in Section 3.12 to the Parent Disclosure Schedule, Parent, together with its does not have any liabilities or obligations of any nature, whether known or unknown, absolute, accrued, contingent or otherwise due or to become due in each case required by GAAP to be reflected or reserved against in the consolidated balance sheet of F subsidiaries (or disclosed in the notes to such balance sheet), that would, individually or in the aggregate, constitute a Material Account of the parent of
 - 3.13 Operation of Parent's Business. Since September 30, 2013 through the date of this Agreement, neither Parent r subsidiaries has engaged in any transaction that, if done after execution of this Agreement, would violate Section 6.1(b) or (c) in
 - 3.14 *Investment Company Act.* Parent is not required to register as an "investment company" as that term is defined to otherwise subject to regulation under, the Investment Company Act of 1940,
- 3.15 Status under Company Operating Agreement. No Parent Party, and no subsidiary of any Parent Party, is an Interested such term is defined in the Company Operating Agreement) or is, or after the consummation of the transactions contemplated by thi would be, an Affiliate or Associate of an Interested Member (in each case, as such term is defined in the Company Operating
 - 3.16 Knowledge of Parent Parties. As of the date hereof, none of the Persons listed in Section 3.16 of the Parent Discloss has actual knowledge that any representation or warranty of the Company made in this Agreement is untrue or inaccurate. Each Parties hereby acknowledges and agrees that it and its affiliates, including KKR Financial Advisors LLC (together with its su assigns, "KKR Financial Advisors"), shall have no right or remedy hereunder based on the inaccuracy of any representation contained in Article IV which any of the Persons listed in Section 3.16 had actual knowledge as of the

ARTICLE IV

REPRESENTATIONS AND WARRANTIES OF THE COMPANY

In order to induce the Parent Parties to enter into this Agreement, subject to the exceptions set forth in the disclosure schedule the Company to Parent and dated as of the date of this Agreement (the "Company Disclosure Schedule") (which exceptions shall identify a Section to which such exception relates, it being understood and agreed that each such exception relates.)

deemed to be disclosed both under such Section and any other Section of this Agreement (other than Section 4.6) to which it is apparent on its face such disclosure relates) and subject to the information disclosed by the Company in any Company SEC Doduring the period from December 31, 2011 through the business day prior to the date of this Agreement (other than in any risk facautionary or forward-looking disclosure contained in such Company SEC Document), the Company represents and warrants

4.1 *Organization and Standing.* The Company is a limited liability company duly organized, validly existing and in gunder the laws of the State of Delaware with all requisite limited liability company power and authority to own, lease, use an properties and to conduct its business as and where now owned, leased, used, operated and conducted. The Company is duly qubusiness and in good standing in each jurisdiction in which the nature of the business conducted by it or the property it owns, leases makes such qualification necessary, except where the failure to be so qualified or in good standing in such jurisdiction would not, in in the aggregate, constitute a Material Adverse Effect with respect to the Company. The Company is not in default of the posservance or fulfillment of any provision of its certificate of formation as in effect on the date of this Agreement (the "Company Formation"). Except as would not have an adverse effect on the ability of the Company to consummate the transactions contemp Agreement, the Company is not in default of the performance, observance or fulfillment of any provision of Company Operating The Company has furnished to Parent prior to the date hereof a complete and correct copy of each of the Company Certificate of Formation (Company Operating Company Ope

4.2 Subsidiaries. All equity interests of the Company's subsidiaries are owned by the Company or another wholly-owned the Company free and clear of all Liens (other than Liens arising under applicable securities laws). Except as provided in Secti Company Disclosure Schedule, as of the date hereof, except for the Company's subsidiaries, the Company does not own any capita other equity interest in, or any interest convertible into or exercisable or exchangeable for any capital stock of or other equity into other Person. Each of the outstanding equity interests of each of the Company's subsidiaries is duly authorized, validly issued, fundamentary in the company of the Company or its wholly-owned subsidiaries. Section 4.2 of the Company Disclosure Schedule sets forth each subsidiaries of the date hereof. There are no outstanding (i) securities of the Company or its subsidiaries convertible into or exchangeable for capital stock or voting securities or securities on the subsidiaries, and no obligation of the Company's subsidiaries to issue, any capital stock, voting securities or securities convertible for capital stock or voting securities of the Company's subsidiaries to issue, any capital stock or voting securities or securities convertible for capital stock or voting securities of the Company's subsidiaries.

4.3 Power and Authority. The Company has all requisite limited liability company power and authority to enter into and Agreement and, subject to receipt of the Company Shareholder Approval, to perform its obligations under this Agreement and to the transactions contemplated hereby. The execution and delivery of this Agreement and the consummation of the transactions contemplated by the Company has been duly authorized by all necessary limited liability company action on the part of the Company to receipt of the Company Shareholder Approval. This Agreement has been duly executed and delivered by the Company and, a authorization, execution and delivery by each of the Parent Parties, constitutes the legal, valid and binding obligation of t enforceable against the Company in accordance with its terms except that the enforcement hereof may be limited by (i) bankruptcy reorganization, moratorium or other similar laws, now or hereafter in effect, relating to creditors' rights generally and (ii) general equity (regardless of whether enforceability is considered in a proceeding in equity or at law). The Company Board, acting upon the

recommendation of the Transaction Committee, at a duly held meeting, unanimously (other than Scott Nuttall and Craig Farr, of present at the meeting) has (i) determined that the transactions contemplated by this Agreement, including the Merger, are fair to, an interests of, the holders of the Company Common Shares, (ii) approved and declared advisable the execution, delivery and perform Agreement and the consummation of the transactions contemplated herein, including the Merger, and (iii) subject to Section 6.60 recommend that the holders of the Company Common Shares approve the Merger (the "Company Board Recommendation") and such matter be submitted for consideration of the holders of the Company Common Shares at the Company Shareholder Meeting. A representations and warranties of the Parent Parties set forth in Section 3.15 are true and correct, the affirmative vote of holders of the outstanding Company Common Shares entitled to vote thereon on the record date for the Company Shareholder Meeting, voting single class, with respect to approval of the Merger is the only vote of the holders of any class or series of capital stock (or other sect Company necessary to approve the Merger and the other transactions contemplated hereby (the "Company Shareholder Approval" and authority of the Transaction Committee to take the actions contemplated by this Agreement to be taken by the Transaction Combeen duly authorized by all necessary limited liability company action on the part of the company action on the part of the

4.4 Capitalization of the

- (a) The authorized capital stock of the Company consists of 500,000,000 Company Common Shares (such term as shall have the meaning given to "Common Shares" in the Company Operating Agreement) and 50,000,000 Preferred Share as used herein shall have the meaning given to it in the Company Operating
- (b) As of the Measurement Date, (i) 204,824,159 Company Common Shares (including 669,820 restricted Compan Shares) were issued and outstanding and (ii) 14,950,000 Preferred Shares were issued and outstanding (each such Prebeing a 7.375% Series A LLC Preferred Share issued by the Company). Since the Measurement Date to the date of this there have been no issuances of capital stock of the Company except upon the exercise of Company Options or settlement Shares. The outstanding Company Common Shares and Preferred Shares have been duly authorized and validly issued paid and nonassessable and free of any preemptive rights and were issued in compliance in all material respects with App Section 4.4(b) of the Company Disclosure Schedule sets forth as of the Measurement Date, a complete and corr outstanding Company Securities other than Company Common Shares and Preferred Shares (including Company Phantom Shares), the number of Company Common Shares issuable thereunder or with respect thereto and the exercise p No subsidiary of the Company owns any Company Common Shares or any Preferred Shares. As of the Measurement Date shares of Company Common Shares were reserved for issuance under the 2007 Share Plan in connection with the outstanding Company Options. Since the Measurement Date, the Company has not granted, committed to grant or otherwick assumed any obligation with respect to any Company Securities, other than as permitted by
- (c) With respect to the Company Options, (i) each grant of a Company Option was duly authorized no later that which the grant of such Company Option was by its terms to be effective (the "Grant Date") by all necessary limited liabil action, including, as applicable, approval by the Company Board (or a duly constituted and authorized committee thereof authorized delegate thereof, and any required shareholder approval by the necessary number of votes or written consents, (grant was made in accordance with the terms of the 2007 Share Plan, the Exchange Act and all other Applicable Law, i rules of the NYSE, and (iii) the per share exercise price of each Company Option was not less than the fair market value of Common Share on the applicable Grant Common Share On the Applicable Gran

Company has not granted, and there is no and has been no Company policy or practice to grant, Company Option otherwise coordinate the grant of Company Options with, the release or other public announcement of material informati the Company or any of its subsidiaries or their financial results

(d) Except as set forth in Section 4.4(b) or Section 4.4(b) or 4.4(d) of the Company Disclosure Schedule, as of th Agreement, except with respect to Company Common Shares issued upon the exercise of Company Options or iss settlement of Phantom Shares, in each case, subsequent to the Measurement Date, there were no (i) outstanding shares of or other voting securities of the Company, (ii) securities of the Company or its subsidiaries convertible into or exchangeal of capital stock or voting securities of the Company, (iii) options, warrants, calls, phantom stock or other rights to acq Company or its subsidiaries, or obligation of the Company or its subsidiaries to issue, any capital stock, voting securities convertible into or exchangeable for capital stock or voting securities of the Company (the items in clauses (i), (ii) and (iii to collectively as "Company Securities"), or (iv) outstanding obligations of the Company or any subsidiary of the repurchase, redeem or otherwise acquire any Company Securities (except with respect to the withholding of Company Cor in respect of the payment of the exercise price or Taxes upon the exercise of any Company Option, vesting of restric Company Common Stock or settlement of any Phantom Unit). Except (x) as set forth in Section 4.4(d) of the Compan Schedule or (y) in connection with the repurchase or acquisition of Company Common Shares pursuant to the terms of Co (and the withholding of Company Common Shares in respect of the payment of the exercise price or Taxes upon the ex Company Option, vesting of a restricted share of Company Common Stock or settlement of any Phantom Unit), neither t nor any of its subsidiaries is a party to any Contract that (i) obligates the Company or any of its subsidiaries to repurchas otherwise acquire any Company Securities, (ii) relates to the voting or transfer of, requires registration of, or grants an rights, anti-dilutive rights, rights of first refusal or other similar rights with respect to, any Company Securities or (iii) other to, creates, establishes or defines the terms and conditions of, any Company Securities. There are no director, independent or employee share incentive plans or arrangements of the Company, other than the Company Plans, under which a Securities are

- 4.5 Conflicts; Consents and Approvals. Except as set forth in Section 4.5 of the Company Disclosure Schedule, neither t and delivery of this Agreement by the Company nor the consummation by the Company of the Merger and the other transactions of
 - (a) conflict with, or result in a breach of any provision of, the Company Certificate of Formation, the Company Agreement or the Share Designation of the Company's outstanding Preferred Shares, subject to the receipt of the Company
 - (b) violate, or conflict with, or result in a breach of any provision of, or constitute a default (or an event which, with notice, the passage of time or otherwise, would constitute a default) under, or entitle any party (with the giving of notice, the time or otherwise) to terminate, accelerate, adversely modify or call a default under, or result in the creation of any Lien the properties or assets of the Company or any of its subsidiaries under, any of the terms, conditions or provisions of any which the Company or any of its subsidiaries
 - (c) assuming compliance with the matters referred to in Section 4.5(d), violate any Applicable Law applicable to the any of its subsidiaries or any of their respective properties

(d) require any action or consent or approval of, or review by, or registration or filing by the Company or any of with, any third party or any Governmental Authority, other than (i) receipt of the Company Shareholder Approval, (ii) any required by the HSR Act, (iii) registrations or other actions required under foreign, federal or state Antitrust Laws or blue-sky laws, (iv) the filing of the Certificate of Merger with the Secretary of State of the State of Delaware, and (v approvals of, or notifications to, any Governmental Authority set forth in Section 4.5(d) of the Company Disclosure.

except in the case of clauses (b), (c) and (d) above for any of the foregoing that would not, individually or in the aggregate, constitu Adverse Effect with respect to the Company or a material adverse effect on the ability of the Company to consummate any of the contemplated by this

4.6 *No Material Adverse Effect.* Since December 31, 2012 through the date of this Agreement, there have been no Event individually or in the aggregate, constitute a Material Adverse Effect with respect to the second control of the control of the second control of the control

4.7 Company SEC Documents;

(a) The Company has timely filed or received the appropriate extension of time within which to file with the Co forms, reports, schedules, statements and other documents required to be filed by it since January 1, 2010 under the Exc the Securities Act (such documents, as supplemented and amended since the time of filing, collectively, the "Co Documents"). The Company SEC Documents, including any financial statements or schedules included therein, at the tim in the case of registration statements and proxy statements, on the dates of effectiveness and the dates of mailing, respect not contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessar make the statements therein, in light of the circumstances under which they were made, not misleading and (ii) complied all material respects with the applicable requirements of the Exchange Act and the Securities Act, as the case may be. Statements of the Company included in the Company SEC Documents were prepared in accordance with GAAP durin involved (except as may be indicated in the notes thereto or, in the case of unaudited statements, as permitted by Form Commission), and fairly present in all material respects (subject in the case of unaudited statements to normal, re adjustments) the consolidated financial position of the Company and its consolidated subsidiaries as at the dates the consolidated results of their operations and cash flows for the periods then ended. No subsidiary of the Company is superiodic reporting requirements of the Exchange Act other than as part of the Company's consolidated reporting requirements of the Exchange Act other than as part of the Company's consolidated reporting requirements of the Exchange Act other than as part of the Company's consolidated reporting requirements of the Exchange Act other than as part of the Company's consolidated reporting requirements of the Exchange Act other than as part of the Company's consolidated reporting requirements of the Exchange Act o

(b) The Company maintains a system of internal control over financial reporting (within the meaning of Rules 1 15d-15(f) promulgated under the Exchange Act) designed to provide reasonable assurances regarding the reliability reporting and the preparation of financial statements for external purposes in accordance with GAAP. The Company disclosure controls and procedures (within the meaning of Rules 13a-15(e) and 15d-15(e) promulgated under the Exdesigned to ensure that information required to be disclosed by the Company in the reports that it files and submits under that information required to be disclosed by the Company in the reports that it files and submits under the Excential transfer of the Company in the reports that it files and submits under the Excential transfer of the Company as appropriate to allow timely decisions regard disclosure, and (ii) has disclosed, based upon the most recent evaluation by the Chief Executive Officer and Chief Financial the Company of the Company's internal

financial reporting, to its auditors and the audit committee of the Company Board (A) all significant deficiencies weaknesses in the design or operation of the Company's internal control over financial reporting which are reasonal adversely affect its ability to record, process, summarize and report financial data and (B) any fraud, whether or not involves management or other employees who have a significant role in the Company's internal control over financial data.

- 4.8 Compliance with Law. The Company and its subsidiaries are in compliance with, and at all times since January 1, 201 in compliance with, all Applicable Law relating to the Company, its subsidiaries or their respective business or properties, exce failure to be in compliance with such Applicable Law would not, individually or in the aggregate, constitute a Material Adverse respect to the Company. No investigation or review by any Governmental Authority with respect to the Company or its subsidiaries or, to the knowledge of the Company, threatened, nor has any Governmental Authority indicated in writing an intention to conduct each case other than those the outcome of which would not, individually or in the aggregate, constitute a Material Adverse Effect with
 - 4.9 Taxes. Except for such matters that would not, individually or in the aggregate, constitute a Material Adverse Effect w
 - (a) The Company and its subsidiaries (i) have timely filed all Tax Returns required to have been filed by or with r Company or any of its subsidiaries, and all such Tax Returns are true, correct and complete; (ii) have within the time prescribed by Applicable Law paid all Taxes required to be paid by them; (iii) have adequate accruals and reserves on statements included in the Company SEC Documents as of the date of this Agreement for Taxes in accordance with GAA not requested or filed any document having the effect of causing any extension of time within which to file any Tax Return of any fiscal year which have not since been filed; and (v) have not received written notice of any deficiencies for any Taxing authority, against the Company or any of its subsidiaries. Neither the Company nor any of its subsidiaries is the sucurrently ongoing Tax audit, Action or other proceeding with respect to Taxes nor has any Tax audit, Action or other proceeding with respect to Taxes been proposed against any of them in writing. Neither the Company nor any of its subsidiaries has request or has waived any statute of limitations in respect of Taxes or agreed to any extension of time with respect to a Tax a deficiency. There are no Liens with respect to Taxes upon any of the properties or assets, real or personal, tangible or intation Company or any of its subsidiaries (other than Liens for Taxes not yet due). No claim has ever been made in writing authority of a jurisdiction where the Company or one of its subsidiaries has not filed Tax Returns claiming that the Comsubsidiary is or may be subject to taxation by that
 - (b) The Company and its subsidiaries have withheld and paid all Taxes required to have been withheld and paid in with amounts paid or owing to any employee, independent contractor, creditor, equityholder or other
 - (c) Neither the Company nor any of its subsidiaries was a "distributing corporation" or "controlled corporation" in intended to qualify under Section 355 of the Code within the past two years or otherwise as part of a plan that includes Neither the Company nor any of its subsidiaries are parties to any contractual obligation relating to Tax sharing, Tax independent allocation. Neither the Company nor any of its subsidiaries has any liability for the Taxes of any person under Treasury section 1.1502-6 (or any similar provision of state, local or foreign law), as a transferee or successor, by Contract of
 - (d) Neither the Company nor any of its subsidiaries has participated in any "listed transaction" within the meaning Regulation Secti

- (e) The Company has at all time since its inception been treated as a partnership, and not as an association or a "pu partnership" (within the meaning of Section 7704(b) of the Code) taxable as a corporation, for United States federa purposes. KKR Financial Holdings II, LLC has at all times been treated as a real estate investment trust (within the Section 856 of the Code) for United States federal income t
- (f) The Company has in effect a valid election under Section 754 of the Code (and any equivalent election for apparent and local income Tax purposes), which will remain in effect for the taxable year including the Code (and any equivalent election for apparent to the code (and any equivalent election for apparent to the code (and any equivalent election for apparent to the code (and any equivalent election for apparent to the code (and any equivalent election for apparent to the code (and any equivalent election for apparent to the code (and any equivalent election for apparent to the code (and any equivalent election for apparent to the code (and any equivalent election for apparent to the code (and any equivalent election for apparent to the code (and any equivalent election for apparent to the code (and any equivalent election for apparent to the code (and any equivalent election for apparent to the code (and any equivalent election for apparent to the code (and any equivalent election for apparent election el
 - (g) For purposes of this
 - (i) "Tax Returns" means returns, reports and forms (including schedules and attachments thereto) filed to be filed with respect to Taxes with any Governmental Authority responsible for the imposition or collections.
 - (ii) "Taxes" means all taxes, fees or similar governmental charges (whether U.S. or non-U.S. federal provincial, territorial or municipal), including income, gross receipts, profits, sales, use, occupation, value added, transfer, franchise, withholding, payroll, employment, capital, goods and services, gross income, business, en severance, service, service use, unemployment, social security, stamp, custom, escheat, excise, or real or persociatives, together with any interest, penalties, additions to tax or additional amounts imposed or assessed with reimposed by any taxing authority with res
- 4.10 *Title to Properties.* Other than in respect of investments of the Company and its subsidiaries that are included in the "Natural Resources" and "Other" business segments as reported in its financial statements that are included in the Company SEC (a) no real property is leased by the Company or its subsidiaries and (b) no real property is owned by the Company or its
- 4.11 Registration Statement; Proxy Statement. None of the information supplied or to be supplied by the Company for incorporation by reference in the Registration Statement, including the Prospectus and the Proxy Statement, at the time the Statement becomes effective or, in the case of the Proxy Statement, at the date of mailing and at the date of the Company Sharehol will contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary in order statements therein, in light of the circumstances under which they were made, not misleading. The Proxy Statement, except for statement that relate only to Parent and its subsidiaries, will comply as to form in all material respects with the provisions of the Security
- 4.12 Litigation. There is no Action pending or, to the knowledge of the Company, threatened against the Company subsidiaries or their respective officers or directors that would, individually or in the aggregate, constitute a Material Adverse Effect to the Company. Neither the Company nor any of its subsidiaries nor any assets, rights or properties of such Persons is subject to any order, writ, injunction or decree that would, individually or in the aggregate, constitute a Material Adverse Effect with respect to the
- 4.13 *Brokerage and Finder's Fees.* Except for the Company's obligations to the Transaction Committee Financial Adviso Company, any of its subsidiaries nor any of their respective equityholders, directors, officers or employees has incurred or will incur the Company or its subsidiaries any brokerage, finder's, financial advisor's or similar fee or commission in connection with the contemplated by this Agreement. The Company has furnished to Parent a complete and correct copy of the engagement letter of the Committee Financial Advisor's properties.

4.14 *Benefit Plans*. Except for the 2007 Share Plan and Deferred Compensation Plan, none of the Company or its subsidiar sponsor, or contribute to, or have at any time prior to the date of this Agreement maintained, sponsored or contributed to, any be programs, policies, agreements or other arrangements, including any employee welfare plan within the meaning of Section 3(1) of employee pension benefit plan within the meaning of Section 3(2) of ERISA (in each case whether or not such plan is subject to employment, individual consulting or other compensation agreements and any bonus, incentive, equity or equity-based compensation, vacation, stock purchase, stock option, severance, employment, change of control, salary continuation, health or life fringe benefit plan, program or agreement (any of the foregoing, a "Commentation of the company or its subsidiar sponsor.

Each Material Company Contract is (assuming due authorization, execution and delivery by each other pa valid and binding obligation of the Company or its subsidiaries, as applicable, and, to the knowledge of the Company, the valid obligation of each other party thereto, except where the failure to be valid and binding would not, individually or in the aggregate Material Adverse Effect with respect to the Company. Neither the Company nor its subsidiaries nor, to the knowledge of the Company party thereto is in violation of or in default in respect of, nor has there occurred an event or condition which with the passage of time notice (or both) would constitute a default under or permit the termination of, any Material Company Contract except for such defaults under or terminations which would not, individually or in the aggregate, constitute a Material Adverse Effect with 1 Company. For purposes of this Agreement, "Material Company Contract" means any Contract to which the Company or any of its si a party and that falls within any of the following categories: (a) any Contract that is a material contract (as defined in Item 6 Regulation S-K of the Commission but without giving effect to the provisions of clause (i) thereof relating to the exclusion of Cont into more than two years before the filing of a registration statement) in effect on the date hereof, (b) any Contract with respec venture or partnership that is material to the Company and its subsidiaries, taken as a whole, and (c) indentures, mortgages, prom loan agreements or guarantees of borrowed money made by the Company or its subsidiaries in excess of \$5,000,000, letters of cre behalf of the Company or its subsidiaries or commitments for the borrowing or the lending of amounts by the Company or its su excess of \$5,000,000, Contracts providing for the creation of any Lien upon any of the assets of the Company or its subsidi aggregate value in excess of \$5,000,000, and Contracts governing any trust preferred securities issued by the Company or any of its Section 4.15 of the Company Disclosure Schedule lists, as of the date hereof, all Contracts specified in clause (c) of the definition Company Contract" and Parent has been provided complete and correct copies of suc

4.16 Undisclosed Liabilities. Except (a) as and to the extent disclosed or reserved against on the unaudited consolidated to fithe Company as of September 30, 2013 included in the Company SEC Documents, (b) as incurred after the date thereof in the ord of business consistent with past practice, or (c) as set forth in Section 4.16 to the Company Disclosure Schedule, the Company, togo subsidiaries, does not have any liabilities or obligations of any nature, whether known or unknown, absolute, accrued, contingent and whether due or to become due in each case required by GAAP to be reflected or reserved against in the consolidated balance Company and its subsidiaries (or disclosed in the notes to such balance sheet), that would, individually or in the aggregate, constituent of the consolidated balance and the subsidiaries (or disclosed in the notes to such balance sheet), that would, individually or in the aggregate, constituent of the consolidated balance and the consolidated balance and the consolidated balance sheet).

4.17 Operation of the Company's Business. Since September 30, 2013 through the date of this Agreement, neither the Compositis subsidiaries has engaged in any transaction that, if done after execution of this Agreement, would violate Section 6.2(c), (d), (j) in any material content of the Company's Business. Since September 30, 2013 through the date of this Agreement, neither the Composition of its subsidiaries has engaged in any transaction that, if done after execution of this Agreement, would violate Section 6.2(c), (d), (ii) in any material content of the Company's Business.

- 4.18 Opinion of Financial Advisors. The Transaction Committee has received the opinion of Sandler O'Neill & Partner "Transaction Committee Financial Advisor") to the effect that, as of the date of the opinion and based upon and subject to the limitations set forth therein, the Merger Consideration to be received is fair to the holders of Company Common Shares from a financial view and, as of the date hereof, such opinion has not been withdrawn, revoked or modified. A signed copy of such opinion available to Parent as soon as practicable after the date of this
- 4.19 Takeover Statutes. Assuming the representations and warranties of the Parent Parties set forth in Section 3.15 are true prior to the date of this Agreement, the Company Board has taken all action necessary to exempt under or make not subject to the real any state takeover law or state law that purports to limit or restrict business combinations or the ability to acquire or vote sleeper and (c) the other transactions contemplated by
- 4.20 *Insurance*. Except as would not, individually or in the aggregate, constitute a Material Adverse Effect with r Company: (a) all directors' and officers' liability insurance policies of the Company and its subsidiaries are in full force and effect insurance in such amounts and against such risks as is customary for the industries in which the Company and its subsidiaries (b) neither the Company nor any of its subsidiaries is in breach or default, and neither the Company nor any of its subsidiaries h action or failed to take any action which, with notice or the lapse of time, would constitute such a breach or default, or permit te modification, of any of such insurance policies. None of the Company nor any of its subsidiaries has received any written notice of of any such insurance policies as to which the Company or such subsidiary has not obtained replacement insurance of similar scope Set forth in Section 4.20 of the Company Disclosure Schedule is the amount of the annual premium currently paid by the Comsubsidiaries for its directors' and officers' liability insurance as of the date of this
 - 4.21 *Investment Company Act.* Neither the Company nor any of its subsidiaries is required to register as an "investment that term is defined in, and is not otherwise subject to regulation under, the Investment Company Act of 1940,

ARTICLE V

MUTUAL COVENANTS OF THE PARTIES

- 5.1 Preparation of Proxy Statement; Shareholder
- (a) As promptly as practicable following the date of this Agreement, Parent and the Company shall cooperate in p shall prepare (i) the Proxy Statement to be sent to the holders of the Company Shares relating to the Company Shareholder and (ii) the Registration Statement, and Parent shall file with the Commission the Registration Statement, in which the Proxishall be included as a prospectus, in connection with the registration under the Securities Act of the Parent Common delivered pursuant to this Agreement. Parent and the Company shall use their reasonable best efforts to have the Proxicleared by the Commission and the Registration Statement declared effective by the Commission as promptly as practical the Registration Statement effective through the Eff
 - (b) Each of Parent and the Company shall promptly furnish all information concerning it or its equityholders to the provide such other assistance, as may be reasonably requested in connection with the preparation, filing and district Registration Statement and the Proxy Statement. The Proxy Statement and the Registration Statement shall include all

reasonably requested by such other party to be included therein. Each of Parent and the Company shall, as promptly as practice the receipt thereof, provide the other party with copies of any written comments and advise the other party of any oral concepts to the Proxy Statement and the Registration Statement received from the Commission, including any requestion for amendments or supplements to the Registration Statement or Proxy Statement, and shall provide the other of all correspondence between it and its Representatives, on the one hand, and the Commission, on the other hand. Notwith foregoing, prior to filing the Registration Statement or mailing the Proxy Statement or responding to any comments of the with respect thereto, each of Parent and the Company (i) shall provide the other with a reasonable opportunity to review a on such document or response (including the proposed final version of such document or response) and (ii) shall give due to including in such document or response any comments reasonably proposed by the other. Each of Parent and the Company shall use the other, promptly after receipt of notice thereof, of the time of effectiveness of the Registration Statement, the iss stop order relating thereto or the suspension of the qualification of Parent Common Units for offering or sale in any jurity each of Parent and the Company shall use its reasonable best efforts to have any such stop order or suspension lifted otherwise.

- (c) If, at any time prior to the Effective Time, either party obtains knowledge of any information pertaining to it of provided by it for inclusion in the Registration Statement or the Proxy Statement that would require any amendment or state Registration Statement or the Proxy Statement so that any of such documents would not include any untrue statement fact or omit to state any material fact necessary in order to make the statements therein, in the light of the circumstances they were made, not misleading, such party shall promptly advise the other party and Parent and the Company shall cooprompt filing with the Commission of any necessary amendment or supplement to the Proxy Statement and the Registration and, as required by Applicable Law, in disseminating the information contained in such amendment or supplement to the
- (d) The Company shall, as soon as practicable following the date the Registration Statement is declared efficient Commission, (i) duly call, give notice of, convene and hold a special meeting of the holders of Company Common "Company Shareholder Meeting") for purposes of obtaining the Company Shareholder Approval and (ii) in furtherance the reasonable best efforts to cause the Proxy Statement to be distributed to the holders of Company Common Shares. With written consent of Parent, no proposals other than the Company Shareholder Approval and routine proposals required in with such approval shall be included in the Proxy Statement or transacted at the Company Shareholder Meeting. Unless the Board or the Transaction Committee shall have made a Company Change of Recommendation as permitted by Sect Section 6.6(f), the Company shall use its reasonable best efforts to solicit the Company Shareholder Approval and shall Company Board Recommendation in the Proxy Statement. The Company agrees that, unless this Agreement has been the accordance with Section 8.1, its obligations pursuant to this Section 5.1 to convene and hold the Company Shareholder Meeting. Acquisition Proposal or by the effecting of a Company Change of Recommendation by the Company of a Company Change of Recommendation by the Company of a Company Change of Recommendation by the Company
- 5.2 Access to Information; Confidentiality. Subject to Applicable Law, each of the Company and Parent shall, and shall c its respective subsidiaries to, permit the other and its Representatives during the period prior to the Effective Time to have reasonable purposes related to the consummation of the Merger and the other transactions contempts.

Agreement, during normal business times and upon reasonable advance written notice, to the other party's and its subsidiari properties, books, records, Contracts and documents; *provided* that the foregoing shall not require any party to (i) provide access to or documents that, in the reasonable judgment of such party, would (x) constitute a waiver of the attorney-client privilege held by (y) violate any Applicable Law or any agreement to which such party is a party. Information exchanged pursuant to this Section subject to the confidentiality agreement, dated as of November 18, 2013, by and between Parent and the Company (the "Confidentiality agreement"). No investigation conducted, or information provided, pursuant to this Section 5.2 shall affect or be deemed to representation or warranty made in this

5.3 Filings; Reasonable Best Efforts; No

(a) Upon the terms and subject to the conditions of this Agreement, and except where a different standard of effor for in this Agreement, each of the parties shall use its reasonable best efforts (subject to, and in accordance with, Applic take, or cause to be taken, all actions, and to do, or cause to be done, and to assist and cooperate with the other parties things necessary, proper or advisable to consummate and make effective, in the most expeditious manner practicable and prior to the End Date, the Merger and the other transactions contemplated by this Agreement, including (i) the obnecessary actions or nonactions, waivers, consents, licenses, permits, authorizations, orders and approvals from G Authorities and the making of all other necessary registrations and filings, (ii) the obtaining of all consents, approvals or withird parties related to or required in connection with the Merger that are necessary or desirable in connection with the Method other transactions contemplated by this Agreement and material to the business of the Company or Parent, as the case may preparation of the Proxy Statement and the Registration Statement, including the Prospectus, (iv) the execution and deadditional instruments necessary to consummate any of the transactions contemplated by, and to fully carry out the purp Agreement, and (v) the providing of all such information concerning such party, its subsidiaries, its affiliates and its substantial officers, directors, employees and partners as may reasonably be requested in connection with any of the matter.

(b) Each of the Company and Parent shall (A) make or cause to be made, as promptly as practicable after the Agreement, any filings required of such party or any of its subsidiaries or affiliates under the HSR Act with respect to the contemplated by this Agreement, including any "secondary acquisition" filings or any filings that may be required if the contemplated by this Agreement will result in a change in the "ultimate parent entity" (as defined under the HSR Holdings or Management Holdings (together with Fund Holdings, the "Group Partnerships"), (B) make or cause to be made filings as are required under other Antitrust Laws with respect to the transactions contemplated by this Agreemed reasonably practicable after the date of this Agreement, (C) substantially comply at the earliest practicable date with an additional information, documents or other materials received by such party or any of its subsidiaries from the FC Commission, the Antitrust Division of the Department of Justice or any other Governmental Authority under the HSR Act Antitrust Laws, and (D) cooperate in good faith with the other party in obtaining all approvals required under applicated Laws and in connection with any such filing (including, with respect to the party making a filing, providing copied documents to the non-filing party and its advisors prior to filing and, if requested, to accept all reasonable additions, changes suggested in connection therewith) and in connection with resolving any investigation or other inquiry of any such the Governmental Authority under any Antitrust Laws with respect to any such filing or any such transaction. Each party reasonable best efforts to furnish to each party.

information required for any application or other filing to be made pursuant to any Applicable Law in connection with the the other transactions contemplated by this Agreement. Each party shall give the other party to this Agreement reasonable of any communication with, and any proposed understanding, undertaking or agreement with, any Governmental Author any such filings or any such transaction. Neither party shall independently participate in any formal meeting, or e substantive conversation, with any Governmental Authority in respect of any such filings, investigation or other inquiry w the other party reasonable prior notice of the meeting or conversation and, to the extent permitted by such Governmental A opportunity to attend any such meeting or participate in any such conversation. The parties will consult and cooper another, in connection with any analyses, appearances, presentations, memoranda, briefs, arguments, opinions and propo submitted by or on behalf of any party in connection with proceedings under or relating to the HSR Act or other An Neither party will directly or indirectly extend any waiting period under the HSR Act or enter into any agreement with a G Authority related to this Agreement or the transactions contemplated by this Agreement except with the prior written c other party (such consent not to be unreasonably withheld, conditioned or delayed). Each party shall promptly furnish th with copies of all correspondence, filings and written communications between them and their affiliates and the representatives on one hand, and any such Governmental Authority or its respective staff on the other hand, with re Agreement and the transactions contemplated by this Agreement, except that any materials concerning one party's val other party or the transactions contemplated by this Agreement, or any party's internal financial information, may be red shall be responsible for all filing fees under the HSR Act or other An

(c) Subject to Section 5.3(d), each of Parent and the Company shall use its reasonable best efforts to resolve such any, as may be asserted by any Governmental Authority with respect to the transactions contemplated by this Agreement HSR Act, the Sherman Act, as amended, the Clayton Act, as amended, the Federal Trade Commission Act, as amended, a federal, state or foreign statutes, rules, regulations, orders, decrees, administrative or judicial doctrines or other laws that to prohibit, restrict or regulate actions having the purpose or effect of monopolization or restraint of trade (collective Laws"). In connection therewith and subject to Section 5.3(d), if any Action is instituted (or threatened to be instituted) charansaction contemplated by this Agreement as violative of any Antitrust Law, each of the Company and Parent shall contest and resist any such Action (through negotiation, litigation or otherwise), including any legislative, administrative action, and to have vacated, lifted, reversed or overturned any decree, judgment, injunction or other order whether preliminary or permanent (each an "Order"), that is in effect and that prohibits, prevents, delays or restricts consumm Merger or any other transactions contemplated by this Agreement, including by pursuing all available avenues of admining judicial appeal and all available legislative action. Each of the Company and Parent shall use its reasonable best efforts action as may be required to cause the expiration or termination of the waiting periods under the HSR Act or other Antitrus respect to such transactions as promptly as possible after the execution of this

(d) Notwithstanding anything to the contrary contained in this Agreement, none of Parent, the Company, or respective subsidiaries shall be obligated to agree, and none of Parent, the Company, or any of their respective subsidiaries without the other party's prior written consent, to take any action or accept any condition, restriction, obligation or requirespect to Parent, the Company, or any of their respective subsidiaries or affiliates or their and their respective subsidiaries' assets if such action, condition, restriction, obligation or requirements (i) would reasonably be expected to require Parent, to or their respective subsidiaries or affiliates to sell, license, transfer, assign, le

of or hold separate any material business or assets or (ii) would reasonably be expected to result in any material limitation or the Company or their respective subsidiaries or affiliates to own, retain, conduct or operate all or a material por respective business.

- 5.4 *Tax Treatment.* The parties agree that, for U.S. federal income tax purposes, the Merger shall be treated as a taxable Company Common Shares (other than the Company Common Shares described in Section 2.1(a)(ii)) for the Merger Company Common Shares (other than the Company Common Shares described in Section 2.1(a)(ii)) for the Merger Company Common Shares (other than the Company Common Shares described in Section 2.1(a)(iii)) for the Merger Shares (other than the Company Common Shares described in Section 2.1(a)(iii)) for the Merger Shares (other than the Company Common Shares described in Section 2.1(a)(iii)) for the Merger Shares (other than the Company Common Shares described in Section 2.1(a)(iii)) for the Merger Shares (other than the Company Common Shares described in Section 2.1(a)(iii)) for the Merger Shares (other than the Company Common Shares described in Section 2.1(a)(iii)) for the Merger Shares (other than the Company Common Shares described in Section 2.1(a)(iii)) for the Merger Shares (other than the Company Common Shares described in Section 2.1(a)(iii)) for the Merger Shares (other than the Company Common Shares described in Section 2.1(a)(iii)) for the Merger Shares (other than the Company Common Shares described in Section 2.1(a)(iii)) for the Merger Shares (other than the Company Common Shares (other than the Compan
- 5.5 Public Announcements. The initial press release concerning the Merger and the other transactions contemplated by thi shall be a joint press release. Except (a) as required by Applicable Law or the requirements of the NYSE (and in that event only if t permit) or (b) with respect to any Company Change of Recommendation or any action taken by the Company, the Company Transaction Committee pursuant to and in accordance with Section 6.6, at all times prior to the earlier of the Closing or termin Agreement pursuant to Section 8.1, Parent and the Company shall consult with each other before issuing any press release or statement or comment with respect to the Merger or any of the other transactions contemplated by this Agreement and shall not is press release or make such other public statement or comment prior to such
- 5.6 Section 16 Matters. Prior to the Closing, each of Parent and the Company shall use all reasonable efforts to approve i accordance with the procedures set forth in Rule 16b-3 promulgated under the Exchange Act and the Skadden, Arps, Slate Flom LLP SEC No-Action Letter (January 12, 1999) any dispositions of Company Common Shares (including derivative securities to Company Common Shares) or acquisitions of Parent Common Units (including derivative securities with respect to Parent Corresulting from the transactions contemplated by this Agreement by each officer or director of Parent or the Company who is subject to fine Exchange Act (or who will become subject to Section 16 of the Exchange Act as a result of the transactions contemplated respect to securities of Parent or the
 - 5.7 Transaction Litigation. The Company shall give Parent the opportunity to participate in the defense or settlement of any litigation against the Company or its directors relating to the Merger and the other transactions contemplated by this Agreement, settlement shall be agreed to without the prior written consent of Parent (which consent shall not be unreasonably withheld, co delayed). For purposes of this paragraph, "participate" means that the non-litigating party will be kept apprised of proposed strate significant decisions with respect to the litigation by the litigating party (to the extent the attorney-client privilege between the lit and its counsel is not undermined), and the non-litigating party may offer comments or suggestions with respect to the litigation by afforded any decision making power or other authority over the litigation except for the settlement consent set forth above. Not anything to the contrary in this Section 5.7, no consent of any Company director shall be required for the settlement of any Action such settlement (i) does not include any admission of wrongdoing by such Company director and (ii) does not require any payr Company director for which Parent (or any of its affiliates) has not agreed to provide comple

5.8 Notification of Certain

(a) Parent shall give prompt notice to the Company of (i) the occurrence or non-occurrence of any event the one-occurrence of which would cause a breach of a representation or warranty contained in Article III, (ii) any failure of a to comply with or satisfy any covenant, condition or agreement to be complied with or satisfied by it under this Agricii) any Events that, individually or in the aggregate, would prevent the Merger from being completed on or prior to the the case of clauses (i) and (ii) above solely if such breach or failure would result in the failure to satisfy one or more of the set forth in Section 7.2(a) or (b); provided, however, that a failure to comply with this Section 5.

constitute the failure of any condition set forth in Section 7.2(a) or (b) to be satisfied unless the underlying inaccuracy or be independently result in the failure of a condition set forth in Section 7.2(a) or (b) to be satisfied; *provided*, *further*, that the any notice pursuant to this Section 5.8(a) shall not limit or otherwise affect the remedies available under this Agree

(b) The Company shall give prompt notice to Parent of (i) the occurrence or non-occurrence of any event the one-occurrence of which would cause a breach of a representation or warranty contained in Article IV, (ii) any failure of to comply with or satisfy any covenant, condition or agreement to be complied with or satisfied by it under this Agrician (iii) any Events that, individually or in the aggregate, would prevent the Merger from being completed on or prior to the the case of clauses (i) and (ii) above solely if such breach or failure would result in the failure to satisfy one or more of the set forth in Section 7.3(a) or (b); *provided, however*, that a failure to comply with this Section 5.8(b) will not constitute any condition set forth in Section 7.3(a) or (b) to be satisfied unless the underlying inaccuracy or breach would independe the failure of a condition set forth in Section 7.3(a) or (b) to be satisfied; *provided, further*, that the delivery of any notice this Section 5.8(b) shall not limit or otherwise affect the remedies available under this Agreement.

ARTICLE VI

ADDITIONAL COVENANTS OF THE PARTIES

- 6.1 Conduct of Parent's Operations. During the period from the date of this Agreement to the Effective Time, Parent stable and the period of the Company action of the Company action, or with the consent, of the Transaction Committee (which consent shall not be unreasonably withheld, conditioned
 - (a) amend or otherwise change the organizational documents of any Parent Party, except for any amendments or would not (i) materially delay, materially impede or prevent the consummation of the Merger and (ii) adversely affect the Company Common Shares in any material respect differently than the holders of Parent Company Common Shares in any material respect differently than the holders of Parent Company Common Shares in any material respect differently than the holders of Parent Company Common Shares in any material respect differently than the holders of Parent Company Common Shares in any material respect differently than the holders of Parent Company Common Shares in any material respect differently than the holders of Parent Company Common Shares in any material respect differently than the holders of Parent Company Common Shares in any material respect differently than the holders of Parent Company Common Shares in any material respect differently than the holders of Parent Company Common Shares in any material respect differently than the holders of Parent Company Common Shares in any material respect differently than the holders of Parent Company Common Shares in any material respect differently than the holders of Parent Company Common Shares in any material respect differently than the holders of Parent Company Common Shares in any material respect differently than the holders of Parent Company Common Shares in any material respect differently than the holders of Parent Company Common Shares in any material respect differently than the holders of Parent Company Common Shares in any material respect differently than the holders of Parent Company Common Shares in any material respect differently than the holders of Parent Company Common Shares in any material respect differently than the holders of Parent Company Common Shares in any material respect differently than the holders of Parent Company Common Shares in any material respect differently than the holders of the holders of the holders of the hol
 - (b) declare, set aside, make or pay any extraordinary or special dividend or distribution, payable in cash, equi property or otherwise, with respect to any of the Parent Common Units (other than, for the avoidance of doubt, any regulation distributions made by Parent in accordance with its distributions
 - (c) (i) acquire or merge with any business, Person or division thereof, if entering into a definitive agreement rel acquisition or merger would be reasonably likely to, or (ii) incur any indebtedness for borrowed money or issue any debt such incurrence or issuance would be reasonably likely to, in either case, (x) materially delay the effectiveness of the Statement or (y) materially delay, materially impede or prevent, the consummation of
 - (d) issue, deliver, sell, grant or dispose of any Parent Common Units or shares of any class of capital stock of Pare voting securities or other ownership interests of Parent, or any options, warrants, convertible securities or other rights o acquire any such Parent Common Units, or any "phantom" stock, "phantom" stock rights, stock appreciation rights, stock or other similar interests of Parent (except, in each case, (i) in the ordinary course (including the issuance of equity-based regular quarterly issuances of Parent

Units made by Parent in accordance with its quarterly exchange process and sales pursuant to the 424(b)(3) Prospectus issuance of Parent Common Units issuable pursuant to the settlement of awards with respect to directors, employees providers of Parent or its affiliates outstanding on the date hereof or as may otherwise be granted in the ordinary course, is accordance with customary terms) or (iii) in connection with acquisitions, business combinations, joint ventures or strate or similar t

- (e) adjust, reclassify, combine, split or subdivide any Parent Common Units, or any capital stock, voting secur ownership interests of Parent or any securities convertible into or exchangeable or exercisable for Parent Common Unstock, voting securities or other ownership interests
 - (f) agree in writing or otherwise commit to take any of the foreg
- 6.2 Conduct of the Company's Operations. Except (i) as expressly required or permitted by this Agreement, (ii) as Applicable Law, (iii) as expressly permitted under this Section 6.2 or as set forth in Section 6.2 of the Company Disclosure Sche actions taken by KKR Financial Advisors acting pursuant to the terms of the Management Agreement, or (v) with the prior writter Parent (which consent shall not be unreasonably withheld, delayed or conditioned), during the period from the date of this Agree Effective Time, the businesses of the Company and its subsidiaries shall in all material respects be conducted only, and the Company subsidiaries shall not take any action except in all material respects, in the ordinary course of business and in a manner consistent in respects with past practice, and the Company shall (and shall cause each of its subsidiaries to) use commercially reasonable effort substantially intact their business organization and maintain and preserve intact their current business relationships. Notwith foregoing, during the period from the date of this Agreement to the Effective Time, the Company shall not, and shall not permit or its subsidiaries to, except (i) as expressly required or permitted by this Agreement, (ii) as required by Applicable Law, (iii) a Section 6.2 of the Company Disclosure Schedule, or (iv) for actions taken by KKR Financial Advisors acting pursuant to the Management Agreement, without the prior written consent of Parent (which consent shall not be unreasonably withheld, conditioned
 - (a) amend or otherwise change the Company Certificate of Formation or the Company Operating
 - (b) issue, deliver, sell, grant, pledge, dispose of or grant a Lien on, or permit a Lien to exist on, any shares of any clastock of the Company or any of its subsidiaries, any other voting securities or other ownership interests, or any optio convertible securities or other rights of any kind to acquire any shares of such capital stock, voting securities or equity into "phantom" stock, "phantom" stock rights, stock appreciation rights, stock-based units or other similar interests of the Composition of its subsidiaries (except for the issuance of Company Common Shares issuable pursuant to the exercise of Composition on the date hereof in accordance with their terms on the date hereof or the crediting of additional Phantom issuance of Company Common Shares in settlement thereof) under the Deferred Compensation Plan in accordance with thereof as a result of deferral elections previously made (and not otherwise timely revoked) in respect of 2013 and 2014 control in the ordinary course of business, and the dividend reinvestment feature under the Deferred Compensation.
 - (c) (i) sell, lease, license, pledge or dispose of or (ii) grant a Lien on, or permit a Lien to exist on, any properties, in other assets or any interests therein of the Company or any of its subsidiaries that have an aggregate value in excess of

- (d) declare, set aside, make or pay any dividend, payable in cash, equity securities, property or otherwise, with resp its capital stock, except for (i) dividends by any of the Company's direct or indirect wholly-owned subsidiaries to the Con of its other wholly-owned subsidiaries and (ii), subject to the first sentence of Section 2.3(c), any regular quarterly distribut the Company in accordance with its distribution policy in an amount up to \$0.22 per Company Common Share for any su
- (e) adjust, reclassify, combine, split, subdivide or redeem, or purchase or otherwise acquire, directly or indirect capital stock, voting securities or other ownership interests or any securities convertible into or exchangeable or exercisab stock, voting securities or other ownership interests (except for the withholding of Company Common Shares in respect of of the exercise price or Taxes upon the exercise of any Company Option, the vesting of restricted shares of Company Co or the settlement of any Ph.
- (f) (i) acquire, including by merger, consolidation or acquisition of stock or assets or any other business combination other manner, any corporation, partnership, other business organization or any business, division or equity interest thereof any indebtedness for borrowed money or issue any debt securities or assume, guarantee or endorse, or otherwise become for, the obligations of any Person, or make any loans or advances or capital contribution to, or investment in, any Person employees in the ordinary course of business and in a manner consistent with past practice or to the Company or any we subsidiary of the
 - (g) modify in any material respect any accounting policies or procedures, other than as required by GAAP or App
- (h) (i) make any change (or file any such change) to any material method of Tax accounting, (ii) make, change or material Tax election; (iii) settle or compromise any material Tax liability or consent to any claim or assessment relating amount of Taxes; (iv) file any amended Tax Return; (v) file any claim for refund of a material amount of Taxes; (vi) e closing agreement relating to a material amount of Taxes; or (vii) waive or extend the statute of limitations in respec
- (i) except (1) for the acceleration of the vesting of existing Phantom Shares and restricted Company Common Strequired under Section 2.4 or (3) to ensure that any Company Plan is not then out of compliance with Applicable Law or such Company Plan on the date hereof, (i) adopt, enter into, terminate or amend any Company Plan; (ii) increase in any compensation or benefits of any director, officer, employee or independent contractor; (iii) grant or pay any chang retention, severance or termination pay to, or increase in any manner the change-in-control, retention, severance or termination to any employee or independent contractor; (iv) grant any equity or equity-based awards loan to any employee or independent contractor (other than the crediting of additional Phantom Shares (and issuance Common Shares in settlement thereof) under the Deferred Compensation Plan in accordance with the terms thereof deferral elections previously made (and not otherwise timely revoked) in respect of 2013 and 2014 compensation in course of business, and the dividend reinvestment feature under the Deferred Compensation Plan); (v) take any action to foother way secure the payment of compensation or benefits under any Company Plan; or (vi) hire or terminate any officer, independent
- (j) except as required by Applicable Law or any judgment by a court of competent jurisdiction and subject to
 (i) pay, discharge, settle or satisfy any claims, liabilities, obligations or litigation (absolute, accrued, asserted or unasserted or otherwise) that are material to the Company and its subsidiaries, taken as a whole, other than the company are discovered.

discharge, settlement or satisfaction in the ordinary course of business and in a manner consistent with past practice of in with their terms, of liabilities disclosed, reflected or reserved against in the financial statements (or the notes thereto) con Company SEC Documents (for amounts not in excess of such reserves) or incurred since the date of such financial state ordinary course of business and in a manner consistent with past practice; (ii) cancel any material indebtedness; or (iii) was any claims or rights of many claims or rights of many claims.

- (k) (i) enter into, (ii) terminate or cancel, except when it may be commercially reasonable to do so, (iii) fail to exerc renew on terms commercially reasonable to the Company, or (iv) modify or amend in any material respect, any Mater.
 - (l) agree in writing or otherwise commit to take any of the foreg
 - 6.3 Indemnification; Directors' and Officers' Is
- (a) For a period of six years from and after the Effective Time, Parent and Surviving Entity shall maintain in effect (provisions (or provisions no less favorable) regarding exculpation and indemnification of present or former officers contained in the organizational documents of each of the Company and its subsidiaries and (ii) any agreements of the Con subsidiaries with any of their respective present or former directors and officers existing as on the date hereof that Persons rights to exculpation or indemnification for acts or omissions occurring at or prior to the Effective Time (whether claimed prior to, at or after the Effective Time, including any matters arising in connection with the transactions contemp Agreement). From and after the Effective Time, each of Parent and the Surviving Entity shall indemnify, defend and he and provide advancement of expenses (including advancing attorneys' fees and expenses incurred in advance of the final d any actual or threatened claim, suit, proceeding or investigation to the fullest extent permitted under Applicable Law; prov Person to whom fees and expenses are advanced provides an undertaking to repay such advances if it is ultimately det such Person is not entitled to indemnification) to, the present and former officers and directors of the Company again claims, damages, costs, expenses, liabilities or judgments that are paid in settlement of or in connection with any Ac whole or in part on, relating in whole or in part to, or arising in whole or in part out of the fact that such Person is or was director of the Company or any of its subsidiaries prior to the Effective Time, whether asserted or claimed prior to, or a Effective Time (including acts or omissions occurring in connection with the approval of this Agreement and the consumi transactions contemplated hereby) to the fullest extent provided or permitted under the organizational documents of the C its subsidiaries and any indemnification agreement entered into between the Company or any of its subsidiaries and (representative forms of which have been made available to Parent prior to the date of this Agreement), in each case as i the date of this Agreement, and to the fullest extent permitted under App
- (b) If Parent does not elect to purchase a "tail" directors' and officers' liability insurance policy for the Company's former officers and directors who are covered prior to the Effective Time by the directors' and officers' liability insurance maintained by the Company with coverage for six years following the Effective Time, and with coverage and amounts at conditions no less favorable to the covered persons than the existing policies of directors' and officers' liability insurance me the Company, the Company may, at its option prior to the Effective Time, purchase such a six year "tail" insurance policitate the Company shall not, without Parent' consent, make a premium payment for such insurance to the extent such premium 200% of the current annual premium paid by the Company shall not, without Parent' consent, make a premium payment for such insurance to the extent such premium paid by the Company shall not, without Parent' consent, make a premium payment for such insurance to the extent such premium paid by the Company shall not, without Parent' consent, make a premium payment for such insurance to the extent such premium paid by the Company shall not, without Parent' consent, make a premium payment for such insurance to the extent such premium paid by the Company shall not premium paid by the Com

"tail" insurance described in the first sentence of this Section 6.3(b) for an amount equal to or less than the Maximum F. Company shall be entitled to obtain as much comparable "tail" insurance as possible for an amount equal to the Maximum the Company is unable to purchase any such "tail" insurance contemplated in the two preceding sentences, Parent shall c Sub to, at the Company's request, purchase, and maintain in full force and effect (and honor the obligations under), during period beginning on the date of the Effective Time, a "tail" insurance policy from one or more insurance carriers believed and reputable with respect to directors' and officers' liability insurance of comparable coverage and amounts and containing conditions no less favorable, in the aggregate, to the officers and directors of the Company as the Company's existing policy for the benefit of the current and former officers and directors with a claims period of six years from the Effective Time however, that in no event shall Merger Sub be required to pay more than the Maximum Premium as the aggregate preme "tail" insurance policies for its entire period, in which case Merger Sub will obtain as much comparable "tail" insurance as

its directors' and officers' liability insurance (such amount being the "Maximum Premium"). If the Company is unable

- (c) If Parent or the Surviving Entity or any of their respective successors or assigns (i) shall consolidate with or more other Person and shall not be the continuing or surviving corporation or entity of such consolidation or merger or (ii) shall or substantially all of its properties or assets to any Person, then, in each case, Parent, the Surviving Entity or any of the successors and assigns, as applicable, shall take such action as may be necessary so that such Person shall assume all of the obligations set forth in this
- (d) The provisions of this Section 6.3 are intended to be for the benefit of, and shall be enforceable by, each present officer and director referred to in this Section 6.3 and his or her heirs and representatives, and are in addition to, and not in for or to the exclusion of, any other rights to indemnification or contribution that any such Person may have or be a Contract, Applicable Law of Contract, Applicable Law o
- 6.4 Activities of Merger Sub. Prior to the Effective Time, Merger Sub shall not conduct any business or make any invest than as specifically contemplated by this Agreement and will not have any assets (other than, if applicable, a *de minimis* amount of the issuance of membership interests in connection with its initial formation) or any material
 - 6.5 NYSE Listing. Parent shall use its reasonable best efforts to cause the Parent Common Units issuable pursuant to this A be approved for listing on the NYSE, subject to official notice of issuance, prior to

6.6 Company Acquisition F

an amount equal to the Maximu

(a) Subject to Section 6.6(c) through Section 6.6(f), the Company agrees that, from the date of this Agreement unti or, if earlier, the termination of this Agreement in accordance with Article VIII, neither it nor any of its subsidiaries sha shall use its reasonable best efforts to cause its and its subsidiaries' directors, officers, employees, agents, investm attorneys, accountants and other representatives (collectively, "Representatives") not to, directly or indirectly, (i) initiate knowingly encourage any inquiries with respect to, or the making of a Company Acquisition Proposal, (ii) engage in any concerning, or provide any confidential information or data to any Person relating to, a Company Acquisition Proposal, (iii) recommend, or propose publicly to approve or recommend, any Company Acquisition Proposal, (iv) approve or recommend, any letter of intent, agreement in principle, merge acquisition agreement, option agreement or other similar agreement relating to any Company

Proposal (each a "Company Acquisition Agreement"), or (v) propose publicly or agree to do any of the foregoing re Company Acquisiti

- (b) Subject to Section 6.6(c) through Section 6.6(f), prior to the Closing, neither the Company Board nor any comm (including the Transaction Committee) shall, directly or indirectly, (i) withdraw, modify or qualify, or publicly propose modify or qualify, in a manner adverse to any Parent Party, the Company Board Recommendation, (ii) approve, adopt or or publicly propose to approve, adopt or recommend, any Company Acquisition Proposal, (iii) in the event of the comme tender offer or exchange offer for any outstanding shares of the Company's capital stock (including the Company Comfail to include in any public statement related to such tender offer or exchange offer or this Agreement (other than any "statement" or similar communication) a statement that the Company Board Recommendation has not changed or referring Company Board Recommendation, or (iv) recommend that the holders of the Company Shares not approve the Merger described in clauses (i)-(iv) above being referred to as a "Company Change of Recommendation or the company Cha
- (c) Nothing contained in this Agreement shall prevent the Company, the Company Board, the Transaction Composite committee of the Company Board from complying with its disclosure obligations under Rule 14d-9 and 14e-2 promute the Exchange Act or from issuing a "stop, look and listen" or similar communication or making any disclosure if the Composite Transaction Committee or any other committee of the Company Board determines, in good faith, after consulta Company's outside legal counsel, that the failure to do so would be inconsistent with the Company's directors' fiduciary Applicable Law of the State of Delaware; provided, however, that if such disclosure has the effect of a Company Recommendation, Parent shall have the right to terminate this Agreement to the extent set forth in Sect.
- (d) Notwithstanding the limitations set forth in this Section 6.6, until the earlier of receipt of the Company Sharehold and any termination of this Agreement pursuant to Section 8.1, if after the date of this Agreement the Company recei unsolicited bona fide Company Acquisition Proposal that the Company Board has determined in good faith, after consultation outside legal counsel and financial advisors (i) constitutes a Company Superior Proposal or (ii) could reasonably be expering a Company Superior Proposal, then the Company may take the following actions: (A) furnish nonpublic information party making such Company Acquisition Proposal, if, and only if, prior to furnishing such information, the Company receing third party an executed confidentiality agreement with provisions no less restrictive to such third party with respect disclosure of nonpublic information than the Confidentiality Agreement and (B) engage in discussions or negotiations we party with respect to such Company Acquisition
- (e) Notwithstanding anything in this Agreement to the contrary, nothing contained in this Agreement shall prevent t from, at any time prior to, but not after, the time of the receipt of the Company Shareholder Approval, in response to the written unsolicited bona fide Company Acquisition Proposal after the date of this Agreement that did not result from Sections 6.6(a)and (g), (i) terminating this Agreement pursuant to Section 8.1(e) in order to enter into a definitive written with respect to such Company Acquisition Proposal or (ii) effecting a Company Change of Recommendation, if, prior to the actions described in clauses (i) or (ii), (A) the Company Board determines in good faith, after consultation with its counsel and financial advisors, that (x) failure to take such action would reasonably be expected to be inconsistent with the directors' fiduciary duties under Applicable Law of the State of Delaware and (y) such Company Acquisition Proposal Company Superior Proposal, (B) Parent shall have received write

"Superior Proposal Notice") of the Company's intention to take such action at least four business days prior to the ta action, and (C) the Company Board continues to believe, after taking into account any modifications to the terms of the contemplated by this Agreement that are offered by Parent after its receipt of the Superior Proposal Notice, that su Acquisition Proposal continues to constitute a Company Superior Proposal. Any material amendment to the financial other material amendment of such Company Acquisition Proposal shall require a new Superior Proposal Notice and the Company Board shall be required to comply again with the requirements of this Section 6.6(e); provided, however the company acquisition Proposal Stall be deemed to be a reference to two business days shall be deemed to be a reference to two business days shall be deemed to be a reference to two business days.

- (f) Notwithstanding anything in this Agreement to the contrary, nothing contained in this Agreement shall prevent to Board from, at any time prior to, but not after, the time of the receipt of the Company Shareholder Approval, effecting Change of Recommendation (other than in response to the receipt of a written unsolicited bona fide Company Acquisition which shall be subject to Section 6.6(e)) if, prior to taking such action, the Company Board, after consultation with its counsel, determines in good faith that failure to take such action would be inconsistent with the Company's directors' fide under Applicable Law of the State of Delaware; provided, however that (i) Parent shall have received written note Recommendation Notice") of the Company's intention to take any such action at least four business days prior to the tax action and (ii) the Company Board continues to believe, after taking into account any modifications to the terms of the contemplated by this Agreement that are offered by Parent after its receipt of such Board Recommendation Notice that such action would be reasonably likely to be inconsistent with the Company's directors' fiduciary duties under Applicable Contemplated by the contemplated by the contemplated by the contemplated by the company's directors' fiduciary duties under Applicable and action would be reasonably likely to be inconsistent with the Company's directors' fiduciary duties under Applicable Contemplated by the contemplated by
- (g) The Company agrees that it and its subsidiaries will (i) immediately cease and cause to be terminated any existi discussions or negotiations with any Person (other than the parties) conducted prior to the date of this Agreement with re Company Acquisition Proposal and (ii) request that each third party that has heretofore executed a confidentiality ag relates to a Company Acquisition Proposal (other than Parent) return or destroy all confidential information heretofore such third party by the Company or on its behalf. The Company agrees that it and its subsidiaries will take the necess promptly inform its and its subsidiaries' Representatives of the obligations undertaken in this
- (h) From and after the date of this Agreement, the Company shall promptly orally notify Parent of any request for in any inquiries, proposals or offers relating to a Company Acquisition Proposal indicating, in connection with such notice, such Person making such request, inquiry, proposal or offer and the material terms and conditions of any proposals or of Company shall provide to Parent written notice of any such inquiry, proposal or offer within 24 hours of such event and of written or electronic correspondence to or from any Person making an Company Acquisition Proposal. The Company shall informed orally, as soon as is reasonably practicable, of the status of any Company Acquisition Proposal, including with restatus and terms of any such proposal or offer and whether any such proposal or offer has been withdrawn or rejected and to shall provide to Parent written notice of any such withdrawal or rejection and copies of any written proposals or information within 24 hours. The Company also agrees to provide any information to Parent (not previously provided to Figure 1 is providing to another Person pursuant to this Section 6.6 at substantially the same time it provides such information Person. All information provided to Parent under this Section 6.6 shall be kept confidential by Parent in accordance with the Confidentiality

(i) For purposes of this

"Company Acquisition Proposal" means any proposal or offer made by any Person other than Parent or its substrespect to (A) a merger, consolidation, acquisition, share exchange, business combination, reorganization, recapitalization liquidation or similar transaction involving the Company, (B) any purchase of an equity interest (including by means of exchange offer) representing an amount equal to or greater than a 15% voting or economic interest in the Company purchase of assets, securities or ownership interests representing an amount equal to or greater than 15% of the consolvalue (including only the portion of the value of the equity interests of the subsidiaries of the Company that is actually ow or indirectly by the Company) or consolidated net revenues of the Company (including only the portion of the revenues at the equity interests of the subsidiaries of the Company that is actually owned directly or indirectly by the Company); prove the avoidance of doubt, the calculation of consolidated asset value and consolidated net revenues shall include only the uninterests in variable interest entities that are held directly or indirectly by the Company so that the total asset value or revenues are not revenues and the total asset value or revenues interest entity is not included in such

"Company Superior Proposal" means a bona fide written Company Acquisition Proposal (except that references in to of "Company Acquisition Proposal" to 15% shall be replaced by 50%) made after the date of this Agreement by any Personal Parent or its subsidiaries on terms that the Company Board determines in good faith, after consultation with its outside and financial advisors, and considering such factors as the Company Board considers to be appropriate (including the likelihood of consummation of such proposal), are more favorable to the Company and the holders of the Company Shall transactions contemplated by this Agreement, taking into account any change to the transaction proposal.

6.7 Approvals. Other than the Company Shareholder Approval, each of the parties agree to obtain all requisite board shareholder and member approvals, to the extent not obtained prior to the date of this Agreement, required to be obtained to con

6.8 Compliance with Agreement. Parent agrees that from and after the date of this Agreement until the Closing, KI Advisors shall not, and Parent shall cause KKR Financial Advisors not to, in its capacity as manager of the Company put Management Agreement and the Company Operating Agreement, take any action (unless acting at the direction of the Company would cause the Company to fail to comply in any material respect with any of the provisions of this

ARTICLE VII

CONDITIONS TO THE MERGER

- 7.1 Conditions to the Obligations of Each Party. The respective obligations of each of the parties to consummate the Mer subject to the fulfillment (or waiver by all parties) at or prior to the Effective Time of the following
 - (a) The Company Shareholder Approval shall have been obtained in accordance with the Company's organizations and Apr
 - (b) At the Company Shareholder Meeting, the holders of a majority of the outstanding Company Common Share those Company Common Shares held by Parent or any affiliate of Parent) shall have approved the Merger and the other contemp.

- (c) (i) Each waiting period applicable to the Merger and the other transactions contemplated by this Agreement un Act, if any, including with respect to the Group Partnerships, if applicable, shall have expired or been earlier terminate consents required under any other Antitrust Law shall have been obtained or any applicable waiting period thereund
 - (d) No outstanding judgment, injunction, order or decree of a competent United States federal or state Governmen shall prohibit or enjoin the consummation of the Merger or the other transactions contemplated by this
 - (e) The Commission shall have declared the Registration Statement effective under the Securities Act, and no similar restraining order by the Commission suspending the effectiveness of the Registration Statement shall
 - (f) The Parent Common Units to be issued in connection with the Merger shall have been approved for listing o subject to official notice
- 7.2 Conditions to the Obligations of the Company. The obligations of the Company to consummate the Merger shall be s fulfillment of the following conditions unless waived by the
 - (a) (i) The representations and warranties of the Parent Parties set forth in Article III (other than Section 3.5) that are a "Material Adverse Effect" qualification shall be true and correct as so qualified at and as of the date of this Agreement of the Closing as though made at and as of such times, (ii) the representations and warranties of the Parent Partie Article III (other than Section 3.3 and Section 3.9) that are not qualified by a "Material Adverse Effect" qualification shall correct (without giving effect to any materiality qualifications therein) at and as of the date of this Agreement and at Closing as though made at and as of such times, except for such failures to be true and correct as would not, in the aggrega a Material Adverse Effect with respect to Parent, (iii) the representations and warranties set forth in Section 3.3 shall correct in all material respects at and as of the date of this Agreement and at and as of the Closing as though made at an of this Agreement and at and as of the Closing as though made at an of this Agreement and at and as of the Closing as though made at and as of such times; provided, however, that we clauses (i), (ii), (iii) and (iv) above, representations and warranties that are made as of a particular date or period shall correct (in the manner set forth in clause (i), (ii), (iii) or (iv) above, as applicable) only as of such date or period shall correct (in the manner set forth in clause (i), (ii), (iii) or (iv) above, as applicable) only as of such date or period shall correct (in the manner set forth in clause (i), (ii), (iii) or (iv) above, as applicable) only as of such date or period shall correct (in the manner set forth in clause (i), (ii), (iii) or (iv) above, as applicable) only as of such date or period shall correct (in the manner set forth in clause (i), (iii) or (iv) above, as applicable) only as of such date or period shall correct (in the manner set forth in clause (i), (iii) or (iv) above, as applicable)
 - (b) Each Parent Party shall have performed in all material respects the obligations and agreements and shall have co material respects with the covenants to be performed and complied with by it under this Agreement at or prior to
 - (c) Parent shall have furnished the Company with a certificate dated the Closing Date signed on its behalf by a Chi Officer or other senior officer of the general partner of Parent to the effect that the conditions set forth in Sections 7.2(a) have be
 - (d) Since the date of this Agreement, there shall not have occurred any Events that, individually or in the aggregate

 Material Adverse Effect with respec
- 7.3 Conditions to the Obligations of Parent Parties. The obligations of the Parent Parties to consummate the Merger shall the fulfillment of the following conditions unless waive
 - (a) (i) The representations and warranties of the Company set forth in Article IV (other than Section 4.6) that are q "Material Adverse Effect" qualification shall

correct as so qualified at and as of the date of this Agreement and at and as of the Closing as though made at and as of (ii) the representations and warranties of the Company set forth in Article IV (other than Section 4.4 (other than the last Section 4.4(b)), Section 4.13 and Section 4.19) that are not qualified by a "Material Adverse Effect" qualification shall correct (without giving effect to any materiality qualifications therein) at and as of the date of this Agreement and at Closing as though made at and as of such times, except for such failures to be true and correct as would not, in the aggregate a Material Adverse Effect with respect to the Company, and (iii) the representation and warranties set forth in Section 4.4 (bi), Section 4.6, Section 4.13 and Section 4.19 shall be true and correct at and as of the Agreement and at and as of the Closing as though made at and as of such times, except, in the case of Section 4.4 (other sentence of Section 4.4(b)), for any *de minimis* inaccuracies; *provided*, *however*, that with respect to clauses (i), (ii), and representations and warranties that are made as of a particular date or period shall be true and correct (in the manner clause (i), (ii), or (iii) above, as applicable) only as of such date or period shall be true and correct (in the manner clause (i), (ii), or (iii) above, as applicable) only as of such date or period shall be true and correct (in the manner clause (i), (iii), or (iiii) above, as applicable) only as of such date or period shall be true and correct (in the manner clause (i), (iii), or (iiii) above, as applicable) only as of such date or period shall be true and correct (in the manner clause (i), (iii), or (iiii) above, as applicable) only as of such date or period shall be true and correct (in the manner clause (i), (iii), or (iiii) above, as applicable)

- (b) The Company shall have performed in all material respects the obligations and agreements and shall have co material respects with the covenants to be performed and complied with by it under this Agreement at or prior to
- (c) The Company shall have furnished Parent with a certificate dated the Closing Date signed on its behalf Executive Officer or other senior officer of the Company to the effect that the conditions set forth in Sections 7.3(a), 7.3(b) have be
- (d) Parent shall have received a payoff letter reasonably acceptable to it with respect to the termination of the Existi Credit Agreement and all commitments or other extensions of credit thereunder and the satisfaction and discharge of premium, if any, interest, fees and other amounts then due or outstanding thereunder and the satisfaction, release and dissecurity interests, mortgages and Liens over the Company and its subsidiaries' properties and assets securing such oblasuch payoff letter shall require the administrative agent under the Existing Company Credit Agreement to deliver al necessary or desirable to evidence or effect the
 - (e) Since the date of this Agreement, there shall not have occurred any Events that, individually or in the aggregate

 Material Adverse Effect with respect to the
- (f) The Company shall have delivered to Parent a statement in accordance with Treasury Regulation Section 1.14-certifying that 50% or more of the value of the gross assets of the Company does not consist of U.S. real property interests or more of the value of the gross assets of the Company does not consist of U.S. real property interests plus cash or cash

ARTICLE VIII

TERMINATION AND AMENDMENT

- 8.1 Termination. This Agreement may be terminated and the Merger may be abandoned at any time prior to
 - (a) by mutual written consent of Parent and the
- (b) by either Parent or the Company if there shall be any law or regulation that makes consummation of the Mer otherwise prohibited, or if any judgment, injunction, order or decree of a competent United States federal or state G Authority enjoining the parties from consummating the Merger shall have been entered and such judgmen

order or decree shall have become final and nonappealable; *provided* that the party seeking to terminate this Agreemen this Section 8.1(b) shall have used its reasonable best efforts to render inapplicable such law or regulation or remove such injunction, order or decree as required by

- (c) by either Parent or the Company if the Merger shall not have been consummated on or before September 16, 20 *Date*"); *provided, however*, that the right to terminate this Agreement under this Section 8.1(c) shall not be available whose failure to perform any covenant or obligation under this Agreement has been the primary cause of, or resulted in, the Merger to occur on or before the performance of the primary cause of the primary cause of the Merger to occur on or before the performance of the primary cause of the primary cause of the primary cause of the performance of the primary cause of the primary ca
- (d) by Parent if prior to obtaining the Company Shareholder Approval (i) a Company Change of Recommendation occurred or (ii) if after the date hereof a Company Acquisition Proposal was publicly announced or disclosed (or any Person publicly announced an intention (whether or not conditional) to make such Company Acquisition Proposal) and the Companish to affirm the Company Board Recommendation by the later to occur of (x) five business days prior to the date of the Shareholder Meeting (as such date may have been adjourned or postponed) and (y) five business days after receipt of a windle business days as may exist between the date announcement or disclosure of the Company Acquisition Proposal and the date of the Company Sharehold
- (e) by the Company, at any time prior to obtaining the Company Shareholder Approval, in order to enter into a define agreement with respect to a Company Superior Proposal it received, *provided* that the Company has complied in all mater with its obligations under Section 6.6 and, in connection with the termination of this Agreement, the Company pays to M its designee in immediately available funds the Termination Payment required to be paid by Section 8.2(b) unless superior obligation is waived by
- (f) by Parent or the Company if at the Company Shareholder Meeting (including any adjournment or postponement Company Shareholder Approval shall not have been obtained; *provided*, *however*, that the right to terminate this Agreeme Section 8.1(f) shall not be available to any party whose failure to perform any covenant or obligation under this Agreeme the primary cause of, or resulted in, the failure of the Company Shareholder Approval to be
- (g) by Parent or the Company if there shall have been a material breach by the other of any of its representations covenants or agreements contained in this Agreement, or if any Event shall have occurred, which breach or Event would failure of one or more of the conditions set forth in Section 7.2(a), 7.2(b) or 7.2(d) (in the case of a breach by, or Event with Parent) or Section 7.3(a), 7.3(b) or 7.3(e) (in the case of a breach by, or Event with respect to, the Company) to be satisfied to the End Date, and such breach or Event shall not be capable of being cured or shall not have been cured by the ear business days after detailed written notice thereof shall have been received by the party alleged to be in breach or with responsible and Event is alleged to have occurred and (ii) the company of the company of the company of the case of a breach by the case of

8.2 Effect of Ter

(a) In the event of the termination of this Agreement pursuant to Section 8.1, this Agreement, except for the provise second sentence of Section 5.2, this Section 8.2 and Sections 9.2 through 9.13, shall become void and have no effect, liability on the part of any party or its directors, officers, general partners or equityholders with respect thereto. Notwith foregoing, nothing in this Section 8.2 shall relieve any party to this Agreement of liability for fraud or any willful or intention of any provision of this Agreement.

it shall be judicially determined that termination of this Agreement was by reason of a willful or intentional breach of this then, in addition to other remedies at law or equity for a willful or intentional breach of this Agreement, the party so for willfully or intentionally breached this Agreement shall indemnify and hold harmless the other parties (unless waived be parties) for their respective reasonable out-of-pocket costs, fees and expenses of their counsel, accountants, financial advisor experts and advisors as well as fees and expenses incident to negotiation, preparation and execution of this Agreement documentation (collectively, "Costs"). No termination of this Agreement shall affect the obligations of the parties confidentiality Agreement, all of which obligations shall survive termination of this Agreement in accordance with the

- (b) If this Agreement is terminated pursuant to Section 8.1(d)(i), or is terminated by the Company or Paren Section 8.1(f) at a time when this Agreement was terminable pursuant to Section 8.1(d)(i), then the Company will, business days following any such termination, pay Merger Sub or its designee in cash by wire transfer of immediately average to an account designated by Merger Sub a termination payment in an amount equal to \$26,250,000 (the "Termination Payment obligation is waived by
 - (c) If this Agreement is terminated pursuant to Section 8.1(e), then the Company shall, concurrently with such term to Merger Sub or its designee in cash by wire transfer in immediately available funds to an account designated by Me Termination Payment unless such payment obligation is waived by
- (d) If this Agreement is terminated pursuant to Section 8.1(c) or 8.1(f), and (A) (x) in the case of a termination Section 8.1(c), prior to the termination of this Agreement and (y) in the case of a termination pursuant to Section 8.1(f) Company Shareholder Meeting (including any adjournment or postponement thereof) at which the Company Shareholde was not obtained, a Company Acquisition Proposal was publicly announced or disclosed (or any Person shall have publicle an intention to make a Company Acquisition Proposal) and (B) within 12 months after the date of such termination, the enters into a letter of intent, agreement-in-principle, acquisition agreement or other similar agreement with respect to announces, a Business Combination or consummates a Business Combination, then the Company will, prior to the consummation of a Business Combination or execution of a definitive agreement with respect thereto, pay to Merg designee in cash by wire transfer in immediately available funds to an account designated by Merger Sub the Termination unless such payment obligation is waived by
- (e) If this Agreement is terminated by Parent for any reason pursuant to Section 8.1(g) on the basis of a be representations, warranties, covenants or agreements contained in this Agreement (provided that such breach did not a relate to a breach of Section 6.8) (A) following the public announcement or disclosure of a Company Acquisition Prointention by any Person to make a Company Acquisition Proposal that was not withdrawn prior to such termination of thi and (B) within 12 months after the date of such termination pursuant to Section 8.1(g), the Company enters into a let agreement-in-principle, acquisition agreement or other similar agreement with respect to, or publicly announce. Combination or consummates a Business Combination, then the Company will, upon the earlier of the consummation of Combination or execution of a definitive agreement with respect thereto, pay to Merger Sub or its designee in cash by wi immediately available funds to an account designated by Merger Sub the Termination Payment unless such payment waived by
 - (f) If this Agreement is terminated by Parent or the Company pursuant to Section 8.1(f), then the Company shall in hold harmless the Parent Parties (unless w

Parent Parties) for their Costs not to exceed \$7,500,000 in the aggregate; *provided*, *however*, that upon payment by the Cor Termination Payment in full, the Company shall no longer be required to indemnify and hold harmless the Parent Par respective Costs pursuant to this So

- (g) For the purposes of this Section 8.2, "Business Combination" means (i) a merger, consolidation, acquisition, sha business combination, reorganization, recapitalization, dissolution, liquidation or similar transaction involving the Compurchase of an equity interest (including by means of a tender or exchange offer) representing an amount equal to or go 50% voting or economic interest in the Company, or (C) any purchase of assets, securities or ownership interests reparamount equal to or greater than 50% of the consolidated asset value (including only the portion of the value of the equity the subsidiaries of the Company that is actually owned directly or indirectly by the Company) or consolidated net revacually owned directly or indirectly by the portion of the revenues attributable to the equity interests of the subsidiaries of the Coractually owned directly or indirectly by the Company); provided that, for the avoidance of doubt, the calculation of consolidated and consolidated net revenues shall include only the unconsolidated interests in variable interest entities that are held indirectly by the Company so that the total asset value or revenues of the entire variable interest entity is not include only the unconsolidated interests and the original provided interest entity is not included to the equity interest entity is not included to the equity interest explanation.
- (h) The parties acknowledge and agree that in no event shall the Company be required to pay the Termination Payment than one occasion. In the event the Termination Payment is paid to a party in accordance with this Section 8.2, such payr the sole and exclusive remedy of such party and its subsidiaries, equityholders and Representatives against the other party subsidiaries, equityholders and Representatives with respect to the termination, event or breach giving rise to that payr Company and shall have no further liability with respect to this Agreement or the transactions contemplated hereby Parties or their respective affiliates, except in the case of fraud or any willful or intentional breach of any provision of this
- 8.3 Amendment. This Agreement may be amended by the parties at any time before or after receipt of the Company Approval, but after any such approval, no amendment shall be made which by Applicable Law requires further approval or authorization. Notwithstanding the foregoing, this Agreen be amended except by an instrument or instruments in writing signed and delivered by an authorized representative of each or approval.
- 8.4 Extension; Waiver. At any time prior to the Effective Time, Parent (with respect to Parent, Fund Holdings and Merger Company (with respect to the Company by approval of the Transaction Committee) may, to the extent legally allowed, (a) extend the performance of any of the obligations or other acts of such other parties, (b) waive any inaccuracies in the representations an contained in this Agreement or in any certificate delivered pursuant to this Agreement, and (c) waive compliance with any of the acconditions contained in this Agreement or in any document delivered pursuant to this Agreement. For the avoidance of doubt, each permitted to waive any rights it may have under Sections 8.1 and 8.2. Any agreement on the part of a party to any such extension or be valid only if set forth in a written instrument signed on behalf of such party. No delay on the part of any party in exercis hereunder shall operate as a waiver of such right, nor shall any waiver on the part of any party of any such right nor any single or party of any such right preclude any further exercise of such right or the exercise of any other

ARTICLE IX

MISCELLANEOUS

9.1 Survival of Representations and Warranties. The representations and warranties made in this Agreement or in an instrument or other document delivered pursuant to this Agreement by the parties shall not survive the Closing. This Section 9.1 sl any covenant or agreement of the parties, which by its terms contemplates performance after the Closing or after the terminal states.

9.2 Notices. All notices, requests, claims, demands and other communications under this Agreement shall be in writing given and shall be deemed given upon receipt if delivered personally, telecopied (delivery of which is confirmed) or dispatched by recognized overnight courier service to the parties (delivery of which is confirmed) or by registered or certified mail (postage paid, requested) to the respective parties at the following addresses (or at such other address for a party as shall be specified by

(a) if to the

KKR Financial H 555 California Stree San Francisc Attention: Gen

Facsimile: (41

with a copy to (which shall not const

Wachtell, Lipton, Rosen 51 Wes

New Yor Attention: Edwar

Attention:

Facsimile: (21

Davi

(b) if to Parent, Fund Holdings or

KKI 9 West 57th Stree

Attention: Gen Facsimile: (21

with a copy to (which shall not const

Simpson Thacher & 425 Lexin New Yor Attention: Gary

Facsimile: (21

9.3 Interp

(a) When a reference is made in this Agreement to an Article or Section, such reference shall be to an Article or Section, such reference shall be to an Article or Section, such reference shall be to an Article or Section, such reference shall be to an Article or Section, such reference shall be to an Article or Section, such reference shall be to an Article or Section, such reference shall be to an Article or Section, such reference shall be to an Article or Section, such reference shall be to an Article or Section, such reference shall be to an Article or Section, such reference shall be to an Article or Section, such reference shall be to an Article or Section, such reference shall be to an Article or Section, such reference shall be to an Article or Section, such reference shall be to an Article or Section, such reference shall be to an Article or Section shall be to an

this Agreement unless otherwise defined in such certificate or other document. The definitions contained in this A applicable to the singular as well as the plural forms of such terms and to the masculine as well as to the feminine and no of such term. Where a word or phrase is defined in this Agreement, each of its other grammatical forms shall have a co meaning. Any statute defined or referred to in this Agreement or in any agreement or instrument that is referred to in this means such statute as from time to time amended, modified or supplemented, including by succession of comparal statutes. The parties have participated jointly in the drafting and negotiation of this Agreement. If an ambiguity or questio interpretation arises, this Agreement shall be construed as if it is drafted jointly by all the parties, and no presumption proof shall arise favoring or disfavoring any party by virtue of authorship of any of the provisions of this Agreement. V words "include," "includes," or "including" are used in this Agreement, they shall be deemed to be followed by the wo limitation." The words "hereof," "herein," "herewith" and "hereby" and words of similar import shall, unless otherw construed to refer to this Agreement as a whole and not to any particular provision of this Agreement. The phrases "th Agreement," "the date hereof" and words of similar import shall be deemed to refer to the date set forth on the cover Agreement. Whenever the last day for the exercise of any right or the discharge of any duty under this Agreement falls or business day, the party having such right or duty shall have until the next business day to exercise such right or discharge Unless otherwise indicated, the word "day" shall be interpreted as a calendar day. No summary of this Agreement prepared behalf of any party shall affect the meaning or interpretation of this Agreement. References to "dollars" or "\$" are to V

- (b) For purposes of this Agreement, "424(b)(3) Prospectus" means that certain prospectus with Registration No. filed by Parent pursuant to Ru
- (c) For purposes of this Agreement, "affiliate" means, with respect to any Person, any other Person that, directly controls, or is controlled by, or is under common control with, such Person; provided that references to an "affiliate" of a not be deemed to include (i) any investment fund or other investment advisory vehicle or (ii) any portfolio comp investment of any investment fund or other investment advisory vehicle. As used in this definition, "control" (include correlative meanings, "controlled by" and "under common control with") shall mean the possession, directly or indirectly, or more intermediaries, of the power to direct or cause the direction of management or policies of a Person, whether ownership of securities or partnership or other ownership interests, by contract of
 - (d) For purposes of this Agreement, a "business day" means a Monday, Tuesday, Wednesday, Thursday or Frid banking institutions in New York City in the State of New York are not authorized or obligated by Applicable I
 - (e) For purposes of this Agreement, "Code" means the Internal Revenue Code of 1986,
 - (f) For purposes of this Agreement, "ERISA" means the Employee Retirement Income Security Act of 1974 and th
- (g) For purposes of this Agreement, "Existing Company Credit Agreement" means that certain credit agreemen November 30, 2012, among the Company, each lender from time to time party thereto, Citibank, N.A., as swingline lende bank, and Citibank, N.A., as administ
 - (h) For purposes of this Agreement, "GAAP" means United States generally accepted accounting principles

- (i) For purposes of this Agreement, "knowledge" means (x) with respect to Parent, the actual knowledge of the Pers Section 9.3(i) of the Parent Disclosure Schedule and (y) with respect to the Company, the actual knowledge of Pers Section 9.3(i) of the Company Disclosure Schedule and (y) with respect to the Company the actual knowledge of Pers Section 9.3(i) of the Company Disclosure Schedule and (y) with respect to Parent, the actual knowledge of the Pers Section 9.3(i) of the Parent Disclosure Schedule and (y) with respect to the Company, the actual knowledge of Pers Section 9.3(i) of the Parent Disclosure Schedule and (y) with respect to the Company the actual knowledge of Pers Section 9.3(i) of the Parent Disclosure Schedule and (y) with respect to the Company the actual knowledge of Pers Section 9.3(i) of the Parent Disclosure Schedule and (y) with respect to the Company Disclosure Schedule and (y) with respe
 - (j) For purposes of this Agreement, "Lien" means any security interest, pledge, hypothecation, mortgage, lien or e
- (k) For purposes of this Agreement, "Management Agreement" means that certain Amended and Restated 1 Agreement, dated as of May 4, 2007 (as amended), by and among the Company, KKR Financial Corp. and KKR Financial
- (1) For the purposes of this Agreement, a "Material Adverse Effect" with respect to any party shall mean any fact, c event, change, occurrence or effect (an "Event") that would have, or would reasonably be expected to have, a material adve the financial condition, business or results of operations of such party and its subsidiaries, taken as a whole; provided, ho Material Adverse Effect with respect to any party shall not include any Event directly or indirectly arising out of or at (i) any decrease in the market price of Parent Common Units, in the case of Parent, or the Company Common Shares, in the Company (but in either case not any Event underlying such decrease to the extent such Event would otherwise constitu Adverse Effect on such party); (ii) conditions, events, or circumstances generally affecting the principal businesses or which such party and its subsidiaries operate; (iii) changes in GAAP, Applicable Law or accounting standards, or in any i of GAAP, Applicable Law or accounting standards; (iv) changes in any analyst's recommendations, any financial strength other recommendations or ratings as to Parent or the Company, as the case may be, or their respective subsidiaries (includi itself, any failure to meet analyst projections); (v) the failure, in and of itself, of Parent or the Company, as the case ma any expected or projected financial or operating performance target publicly announced or provided to the other party price of this Agreement, as well as any change, in and of itself, by Parent or the Company, as the case may be, in any expected financial or operating performance target as compared with any target publicly announced or provided to the other part date of this Agreement (but, in either case, not any Event underlying such failure or change to the extent such Event wou constitute a Material Adverse Effect on such party); or (vi) any changes or developments in United States, Europe, A economic, regulatory or political conditions in general (including the outbreak or escalation of hostilities or acts of war or t generally affecting the financial, debt, credit or securities markets in the United States, Europe, Asia or elsewhere including changes in interest rates and commodities prices, or any natural disaster; provided, however that the exception subclauses (ii), (iii) and (vi) may be taken into account in determining whether a Material Adverse Effect has occurred or expected to occur to the extent (but only to the extent) such Event has a materially disproportionate impact on such subsidiaries relative to other participants in the principal businesses or industries in which such party and its subsidiaries
 - (m) For purposes of this Agreement, "Person" means an individual, a corporation, a limited liability company, a pa association, a trust or any other entity, group (as such term is used in Section 13 of the Exchange Act) or organization Governmental Authority, and any permitted successors and assigns of
 - (n) For purposes of this Agreement, a "subsidiary" of any Person means another Person an amount of the voting other voting interests of which is sufficient to elect at least a majority of its board of directors, managers, general partner governing body (or, if there are no such voting securities or voting interests, 50% or more of the equity

other equity interests of which) is owned directly or indirectly by such first Person; *provided* that a subsidiary does not incinvestment fund or other investment advisory vehicle or (y) any portfolio company or other investment of any investment advisory vehicle or (y) any portfolio company or other investment of any investment advi

- 9.4 *Counterparts.* This Agreement may be executed in counterparts, which together shall constitute one and the same Agreement parties may execute more than one copy of this Agreement, each of which shall constitute an original. Signatures to this Agreement by facsimile transmission, by electronic mail in "portable document format" ("pdf") form, or by any other electronic means intended the original graphic and pictorial appearance of a document, will have the same effect as physical delivery of the paper document original graphic and pictorial appearance of a document, will have the same effect as physical delivery of the paper document original graphic and pictorial appearance of a document, will have the same effect as physical delivery of the paper document original graphic and pictorial appearance of a document, will have the same effect as physical delivery of the paper document original graphic and pictorial appearance of a document original graphic and pictorial appearance or a document or graphic and pictorial appearance or graphic appearanc
- 9.5 Entire Agreement. This Agreement (including any exhibits and schedules hereto, the Parent Disclosure Schedule and t Disclosure Schedule) and the Confidentiality Agreement constitute the entire agreement among the parties and supersede all prior and understandings or representations by or among the parties whether written and oral with respect to the subject matter hereof a being understood that the Confidentiality Agreement shall continue in full force and effect until the Closing and shall survive any te
- 9.6 Severability. Any term or provision of this Agreement that is invalid or unenforceable in any jurisdiction sh jurisdiction, be ineffective to the extent of such invalidity or unenforceability without rendering invalid or unenforceable the remaining provisions of this Agreement or affecting the validity or enforceability of any of the terms or provisions of this Agreement jurisdiction. If any provision of this Agreement is so broad as to be unenforceable, the provision shall be interpreted to be only so broad as to be unenforceable.
 - 9.7 *Third-Party Beneficiaries.* Except for the agreement set forth in Section 6.3, nothing in this Agreement, express of intended or shall be construed to create any third-party beneficiaries or confer upon any Person other than the parties any rights remedies of any nature whatsoever under or by reason of this
 - 9.8 Governing Law. This agreement shall be governed by, and construed in accordance with, the laws of the State regardless of the laws of another jurisdiction that might otherwise govern under applicable principles of conflicts of
 - 9.9 Consent to Jurisdictio
 - (a) Each of the parties irrevocably and unconditionally agrees that any Action arising out of or relating to this Agree rights and obligations arising under this Agreement, or for recognition and enforcement of any judgment in respect of thi and the rights and obligations arising under this Agreement brought by another party or its successors or assigns, shall be determined exclusively in the Court of Chancery of the State of Delaware, or in the event (but only in the event) that successors are subject matter jurisdiction over such Action, in the United States District Court for the District of Delaware parties irrevocably submits with regard to any Action for itself and in respect of its property, generally and uncondition personal jurisdiction of the aforesaid courts and agrees that it will not bring any Action relating to this Agreement transactions contemplated by this Agreement in any court other than the aforesaid courts. Each of the parties irreunconditionally waives, and agrees not to assert, by way of motion, as a defense, counterclaim or otherwise, in any Action to this Agreement, (i) any claim that it is not personally subject to the jurisdiction of the above named courts for any reason the failure to serve in accordance with this Section 9.9, (ii) any claim that it or its property is exempt or immune from jutice and such court or from any legal process commenced in such courts (whether through service of notice, attachment prior attachment in aid of execution of judgment, execution of judgment or otherwise) and (iii) to the fullest extent permitted by Law, and the failure to serve in accordance with the secution of judgment or otherwise) and (iii) to the fullest extent permitted by Law, and the failure to serve in accordance with the secution of judgment or otherwise) and (iii) to the fullest extent permitted by Law, and the failure to serve in accordance with the secution of judgment or otherwise) and (iii) to the fullest extent permitted by Law, and the failure to serve in accordance with the fail

- (A) the Action in such court is brought in an inconvenient forum, (B) the venue of such Action is improper or (C) this A the subject matter of this Agreement, may not be enforced in or by such courts. Each of the parties also agrees the non-appealable judgment against a party in connection with any Action shall be conclusive and binding on such party a award or judgment may be enforced in any court of competent jurisdiction, either within or outside of the United States. A exemplified copy of such award or judgment shall be conclusive evidence of the fact and amount of such award or judgment shall be conclusive evidence.
- (b) Each of the parties consents to service being made through the notice procedures set forth in Section 9.2 an service of any process, summons, notice or document by registered mail (return receipt requested and first-class postage prespective addresses set forth in Section 9.2 shall be effective service of process for any Action in connection with this Athe transactions contemplated by this Agreement. Nothing in this Section 9.9 shall affect the right of any party to serve leg
- 9.10 Specific Performance. The parties recognize and agree that if for any reason any of the provisions of this Agree performed in accordance with their specific terms or are otherwise breached or violated, immediate and irreparable harm or injucaused for which money damages would not be an adequate remedy under Applicable Law. Accordingly, each party agrees that, in a other remedies to which it may be entitled, each of the parties is entitled to a decree of specific performance, and each of the parties be entitled to an injunction restraining any violation or threatened violation of any of the provisions of this Agreement without the posting a bond or other form of security. In the event that any Action should be brought in equity to enforce any of the provisions of this Agreement without the posting a bond or other form of security. In the event that any Action should be brought in equity to enforce any of the provisions of the prov
 - 9.11 WAIVER OF JURY TRIAL. EACH OF THE PARTIES HERETO IRREVOCABLY WAIVES ANY AND ALI TRIAL BY JURY IN ANY LEGAL PROCEEDING ARISING OUT OF OR RELATING TO THIS AGREEME TRANSACTIONS CONTEMPLATE
- 9.12 Assignment. Neither this Agreement nor any of the rights, interests or obligations arising under this Agreement shall be indirectly assigned, delegated sublicensed or transferred by any of the parties (whether by operation of law or otherwise), in whole any other Person (including any bankruptcy trustee) without the prior written consent of the other parties; provided, however, that Fu shall have the right to assign any of its rights or obligations hereunder to Management Holdings and shall be permitted to assign any Merger Sub Common Units to Management Holdings, in each case, without the need for consent from any other party (provided Holdings shall remain liable for all obligations hereunder to the extent not performed by Management Holdings). Subject to the sentence, this Agreement shall be binding upon, inure to the benefit of and be enforceable by the parties and their respective su
- 9.13 Expenses. Subject to the provisions of Section 8.2, all costs and expenses incurred in connection with this Agreet transactions contemplated by this Agreement and thereby shall be paid by the party incurring such expenses, except that those exper in connection with filing, printing and mailing the Registration Statement and the Proxy Statement (including filing fees related the shared equally by Fund Holdings and the proxy Statement (including filing fees related the shared equally by Fund Holdings).

[Remainder of Page Intentionally Left Blank; Signature Page Follows]

IN WITNESS WHEREOF, the parties have executed this Agreement as of the date first w

KKR & CO. L.P.

By: KKR Management LLC, its general partner

By: /s/ DAVID SORKIN

Name: David Sorkin

Title: General Counsel & Secretary

KKR FUND HOLDINGS L.P.

By: KKR Fund Holdings GP Limited,

a general partner

By: /s/ DAVID SORKIN

Name: David Sorkin Title: Director

COPAL MERGER SUB LLC

By: /s/ DAVID SORKIN

Name: David Sorkin

Title: General Counsel & Secretary

KKR FINANCIAL HOLDINGS LLC

By: /s/ NICOLE J. MACARCHUK

Name: Nicole J. Macarchuk Title: General Counsel

[Signature Page to Agreement and Plan of Merger]

List of Schedules Omitted from the Merge

Pursuant to Item 601(b)(2) of Regulation S-K, the schedules to the merger agreement included in this Annex A have been of briefly identifying the contents of the omitted schedules is set forth below. The registrant agrees to furnish supplementally a copy of schedules to the Securities and Exchange Commission of the Securit

Parent Disclosu

Section 3.2	Power and Authority	
Section 3.3(b)	Capitalization of Parent and Merger Sub	
Section 3.4(d)	Conflicts; Consents and Approvals	
Section 3.8	Litigation	
Section 3.12	Undisclosed Liabilities	
Section 3.16	Knowledge of the Parent Parties.	
Section 6.1	Conduct of Parent's Operations	
Section 9.3(i)	Knowledge	
		Company Disclos
Section 4.2	Subsidiaries	
Section 4.4	Capitalization of the Company	
Section 4.15	Contracts	
Section 4.20	Insurance	
Section 6.2	Conduct of the Company's Operations	
Section 9.3(i)	Knowledge	
	A-44	

Table

[LETTERHEAD OF SANDLER O'NEILL & PARTNERS, L.P.]

Decem

Transaction Committee of the Board KKR Financial H 555 California Stree San Francisc

Ladies and

KKR Financial Holdings LLC (the "Company"), KKR & Co L.P. ("Parent"), KKR Fund Holdings L.P. ("Funds") and I Sub LLC ("Merger Sub" and collectively with Parent and Funds, the "Parent Parties") will enter into an Agreement and Plan of Me of December 16, 2013 (the "Agreement") pursuant to which Merger Sub will be merged with and into the Company (the "Merger Company surviving the Merger as a subsidiary of Funds. Pursuant to the terms of the Agreement, each share of Company common and outstanding prior to the effective time shall automatically be converted into the right to receive a number of validly issued fundamental to the Exchange Ratio (the "Merger Consideration"). The Excisequal to 0.510. Cash will be paid in lieu of any fractional Parent Common Units. Capitalized terms used herein without definition the meanings assigned to them in the Agreement. The other terms and conditions of the Merger are more fully set forth in the Agreement as to whether the Merger Consideration is fair, from a financial point of view, to the holders of Company Compa

Sandler O'Neill & Partners, L.P., as part of its investment banking business, is regularly engaged in the valuation of financia and their securities in connection with mergers and acquisitions and other corporate transactions. In connection with this opin reviewed, among other things: (i) the Agreement; (ii) certain publicly available financial statements and other historical financial in the Company that we deemed relevant; (iii) certain publicly available financial statements and other historical financial informat that we deemed relevant; (iv) publicly available mean analyst earnings estimates for the Company for the years ending December December 31, 2014 and a publicly available mean analyst growth estimate for the years thereafter and in each case as discussed w management of the Company; (v) publicly available mean analyst economic net income estimates for Parent for the years ending I 2013 and December 31, 2014 and a publicly available mean analyst growth estimate for the years thereafter and in each case as di the senior management of Parent; (vi) the pro forma financial impact of the Merger on Parent, based on assumptions relating t expenses, purchase accounting adjustments and cost savings determined by the senior management of the Company and Parent; (vii) reported historical price and trading activity for Company's and Parent's common stock, including a comparison of certain finance market information for Company and Parent and similar publicly available information for certain other similar companies the which are publicly traded; (viii) the financial terms of certain recent business combinations among other asset vehicles and af transactions in the financial services industry, to the extent publicly available; (ix) certain of the terms of the management agreement which, KKR Financial Advisors LLC, an indirect, wholly-owned subsidiary of Parent manages the Company; (x) the cu environment generally and the financial services environment in particular; and (xi) such other information, financial studies, investigations and financial, economic and market criteria as we considered relevant. We also discussed with certain members are considered relevant. management of the Company the business, financial condition, results of operations and prospects of the Company and held simila with certain members of senior management of Parent regarding the business, financial condition, results of operations and prospec

In performing our review, we have relied upon the accuracy and completeness of all of the financial and other informa available to us from public sources, that was provi

Table

the Company or Parent or their respective representatives or that was otherwise reviewed by us and have assumed such a completeness for purposes of rendering this opinion. We have further relied on the assurances of the respective managements of t and Parent that they are not aware of any facts or circumstances that would make any of such information inaccurate or misleading. been asked to and have not undertaken an independent verification of any of such information. We did not make an independent appraisal of the specific assets, the collateral securing assets or the liabilities (contingent or otherwise) of the Company, Parent or respective subsidiaries, or the collectability of any such assets, nor have we been furnished with any such evaluations of

We have also assumed that there has been no material change in the Company's assets, financial condition, results of operations prospects since the date of the most recent financial statements made available to us. We have assumed in all respects material to that the Company will remain as a going concern for all periods relevant to our analyses. We express no opinion as to any accounting and tax matters relating to the Merger or any other related transactions contemplated by the

Sandler O'Neill used publicly available earnings estimates and long-term growth rates for the Company and Parent in its a respective managements of the Company and Parent confirmed to us that they reflected the best currently available estimates and judgments of the respective future financial performances of the Company and Parent, respectively, and we assumed that such performance achieved. With respect to the projections of transaction expenses, purchase accounting adjustments and cost savings determined by a with the senior management of Parent, the management of Parent confirmed to us that they reflected the best currently available expressions of such management and we assumed that such performances would be achieved. We express no opinion as to suppose the projections or the assumptions on which they are based. We have also assumed that there has been no material change in the Companent's assets, financial condition, results of operations, business or prospects since the date of the most recent financial state available to us. We have assumed in all respects material to our analysis that Company and Parent will remain as going concerns for relevant to our analyses, that all of the representations and warranties contained in the Agreement and all related agreements are true that each party to the agreements will perform all of the covenants required to be performed by such party under the agreements are conditions precedent in the agreements are

Our opinion is necessarily based on financial, economic, market and other conditions as in effect on, and the information made us as of, the date hereof. Events occurring after the date hereof could materially affect this opinion. We have not undertaken to up reaffirm or withdraw this opinion or otherwise comment upon events occurring after the

We have acted as financial advisor to the Transaction Committee of the Board of Directors of the Company in connection with and will receive a fee for our services, a substantial portion of which is contingent upon consummation of the Merger. We will also for this fairness opinion. The Company has also agreed to indemnify us against certain liabilities arising out of our engagement. O directed to the Transaction Committee of the board of directors of the Company, in connection with its consideration of the Mer addresses the fairness of the Merger Consideration to the holders of the Company common shares, from a financial point of view. does not address the underlying business decision of the Company to engage in the Merger, the relative merits of the Merger as com other alternative business strategies that might exist for the Company or the effect of any other transaction in which the Company of the Company of the Merger by any Company officer, director or employee, if any, relative to the amount of compensation to be received shar

<u>Table</u>

opinion is not to be quoted or referred to, in whole or in part, in a registration statement, prospectus or in any other document, proposed or in part, in a registration statement, prospectus or in any other document, proposed or in part, in a registration statement, prospectus or in any other document, proposed or in part, in a registration statement, prospectus or in any other document, proposed or in part, in a registration statement, prospectus or in any other document, proposed or in part, in a registration statement, prospectus or in any other document, proposed or in part, in a registration statement, prospectus or in any other document, proposed or in part, in a registration statement, prospectus or in any other document, proposed or in part, in a registration statement, prospectus or in any other purposed or in part, in a registration statement, proposed or in a registration statement, proposed or in part, in a registration

Based upon and subject to the foregoing, it is our opinion that the consideration to be exchanged in the Merger is fair to the h Company common stock from a financial p

Very truly yours,

/s/ Sandler O'Neill & Partners, L.P. B-3

200 West Street -- New York, New York Tel: 212-902-1000 -- Fax: 2

PERSONAL AND CONF

Decem

Board of Directors of KKR Manaş as General Partner of KKI 9 West 57th Stree New York, New

Ladies and

You have requested our opinion as to the fairness from a financial point of view to KKR & Co. L.P. (the "Company") of the ex (the "Exchange Ratio") of 0.51 common units, representing the limited partnership interests (the "Company Common Units") of the be issued in exchange for each Common Share (as defined in the Amended and Restated Operating Agreement of KKR Financial H ("KFN)) (the "KFN Common Stock") of KFN pursuant to the Agreement and Plan of Merger, dated as of December 16, 2013 (the "Aby and among the Company, KKR Fund Holdings L.P. ("KKR Holdings"), Copal Merger Sub LLC, a wholly-owned subsides the Holdings of the Park of th

Goldman, Sachs & Co. and its affiliates are engaged in advisory, underwriting and financing, principal investing, sales research, investment management and other financial and non-financial activities and services for various persons and entitie Sachs & Co. and its affiliates and employees, and funds or other entities in which they invest or have other economic interests or wit co-invest, may at any time purchase, sell, hold or vote long or short positions and investments in securities, derivatives, loans, or currencies, credit default swaps and other financial instruments of the Company and its portfolio companies, KFN and any of the affiliates and third parties, or any currency or commodity that may be involved in the transaction contemplated by the Ag "Transaction") for the accounts of Goldman, Sachs & Co. and its affiliates and employees and their customers. We have acted advisor to the Company in connection with, and have participated in certain of the negotiations leading to, the Transaction. We expe fees for our services in connection with the Transaction, all of which are contingent upon consummation of the Transaction, and t has agreed to reimburse our expenses arising, and indemnify us against certain liabilities that may arise, out of our engagement. provided certain investment banking services to the Company and its affiliates and portfolio companies from time to time f Investment Banking Division has received, and may receive, compensation, including having acted as joint book-running manager to a public offering of 30,000,000 shares of common stock of Dollar General Corporation ("Dollar General"), a portfolio con Company, in June 2012; as joint book-running manager with respect to a public offering of 36,000,000 shares of common sto General in September 2012; as joint book-running manager with respect to an offering of 6.5% Senior Notes due 2020 (aggregation of the control of the contro amount \$825,000,000) and 6.5% Senior Subordinated Notes due 2020 (aggregate principal amount \$800,000,000) of Biomet, Inc. company of the Company, in September 2012; as joint book-running manager with respect to a public offering of 38,500,000 share: stock of Nielsen Holdings N.V. ("Nielsen"), a portfolio company of the Company, in February 2013; as joint book-running manager to a public offering of 30,000,000 shares of common stock of Dollar General in March 2013; as joint book-running manager with public offering of Dollar General's 1.875% Senior Notes due 2018 (aggregate principal amount \$400,000,000

Securities and Investment Services Provided by Goldman, Sachs & Co.

General's 3.250% Senior Notes due 2023 (aggregate principal amount \$900,000,000) in April 2013; and as joint book-running necessect to a public offering of 35,000,000 shares of common stock of Nielsen in May 2013. We may also in the future provid banking services to the Company, KFN and their respective affiliates and portfolio companies for which our Investment Banking I receive compensation. Affiliates of Goldman, Sachs & Co. also may have co-invested with the Company and its affiliates from time may have invested in limited partnership units of affiliates of the Company from time to time and may do so

In connection with this opinion, we have reviewed, among other things, the Agreement; the Company's Registration Statement including the prospectus contained therein dated September 16, 2010 relating to the initial public offering of Company Common V reports to unitholders or shareholders and Annual Reports on Form 10-K of the Company for the three years ended December 31. KFN for the five years ended December 31, 2012; certain interim reports to unitholders or shareholders and Quarterly Reports on F the Company and KFN; certain other communications from the Company and KFN to their respective unitholders or shareho publicly available research analyst reports for KKR and KFN; certain financial analyses and forecasts for KFN prepared by its r certain internal financial analysis and forecasts for the Company prepared by its management for the fourth quarter of 2013, certain analyses and forecasts for the Company prepared by research analysts and certain financial analyses and forecasts for KFN pre management of the Company, in each case, as approved for our use by the Company (the "Forecasts"); and certain cost savings a synergies projected by the management of the Company to result from the Transaction, as approved for our use by the C "Synergies"). We have also held discussions with members of the senior managements of the Company and KFN regarding their a the past and current business operations, financial condition and future prospects of KFN and with members of senior manage Company regarding their assessment of the past and current business operations, financial condition and future prospects of the Com strategic rationale for, and the potential benefits of, the Transaction; reviewed the reported price and trading activity for the Compa Units and the shares of KFN Common Stock; compared certain financial and stock market information for the Company and KFN information for certain other companies the securities of which are publicly traded; reviewed the financial terms of certain rec combinations in the permanent capital industry and in other industries; and performed such other studies and analyses, and considered factors, as we deemed

For purposes of rendering this opinion, we have, with your consent, relied upon and assumed the accuracy and completeness financial, legal, regulatory, tax, accounting and other information provided to, discussed with or reviewed by, us, without a responsibility for independent verification thereof. In that regard, we have assumed with your consent that the Forecasts and the S reasonable and reflect the best currently available estimates and judgments of the management of the Company. We have independent evaluation (except for certain valuations of the Collateralized Loan Obligation assets of KFN) or appraisal of the subsidiaries (including any contingent, derivative or other off-balance-sheet assets and liabilities) of the Company or KFN or any of the subsidiaries and we have not been furnished with any such evaluation or appraisal. We have assumed that all governmental, regula consents and approvals necessary for the consummation of the Transaction will be obtained without any adverse effect on the Compor on the expected benefits of the Transaction in any way meaningful to our analysis. We also have assumed that the Transactonsummated on the terms set forth in the Agreement, without the waiver or modification of any term or condition the effect of which in any way meaningful to

Our opinion does not address the underlying business decision of the Company to engage in the Transaction, or the relative Transaction as compared to any strategic alternati

C-2

be available to the Company; nor does it address any legal, regulatory, tax or accounting matters. This opinion addresses only the fair financial point of view to the Company, as of the date hereof, of the Exchange Ratio pursuant to the Agreement. We do not express and our opinion does not address, any other term or aspect of the Agreement or Transaction or any term or aspect of any other a instrument contemplated by the Agreement or entered into or amended in connection with the Transaction, including, the fairness of the Company; nor as to the fairness of the amount or nature of any compensation to be paid or payable to any of the officers employees of the Company or KFN, or any class of such persons, in connection with the Transaction, whether relative to the Exception pursuant to the Agreement or otherwise. We are not expressing any opinion as to the prices at which Company Common Units will time or as to the impact of the Transaction on the solvency or viability of the Company or KFN or the ability of the Company or their respective obligations when they come due. Our opinion is necessarily based on economic, monetary, market and other conference of the information made available to us as of, the date hereof and we assume no responsibility for updating, revising of this opinion based on circumstances, developments or events occurring after the date hereof. Our advisory services and the opinion herein are provided for the information and assistance of the Board of Directors of KKR Management LLC, as General Partner of the inconnection with its consideration of the Transaction. This opinion has been approved by a fairness committee of Goldman,

Based upon and subject to the foregoing, it is our opinion that, as of the date hereof, the Exchange Ratio pursuant to the Agree from a financial point of view to the

Ve

(GOLDMAN, SA

C-3

CON

Decem

9 We

The Independent Directors Constituting the Conflicts Committee of Directors of KKR Mana c/o KKI

New York, New

Dear Independe

We understand that KKR & Co. L.P., a Delaware limited partnership ("Buyer"), KKR Fund Holdings L.P., an exempted limited formed under the laws of the Cayman Islands and an indirect controlled affiliate of Buyer, Copal Merger Sub LLC, a Delaware limited company and an indirect subsidiary of Buyer ("Merger Sub"), and KKR Financial Holdings LLC, a Delaware limited liabil ("Target"), propose to enter into an Agreement and Plan of Merger, dated as of December 16, 2013 (the "Agreement"), pursuant to will acquire Target (the "Transaction"). Pursuant to the Agreement, Merger Sub will be merged with and into Target and each common share of Target ("Target Common Shares"), will be converted into the right to receive 0.51 (the "Exchange Ratio") of a common share partnership interests of Buyer ("Buyer Common Units"). The terms and conditions of the Transaction are in forth in the

You have requested our opinion as of the date hereof as to the fairness, from a financial point of view, to Buyer of the Exc provided for in the

In connection with this opini

Reviewed the financial terms and conditions of the

Reviewed certain publicly available historical business and financial information relating to Targe

Reviewed various financial forecasts and other data provided to us by Target and Buyer relating to the business of Business of Business of Business of Business of Target and Suyer relating to the business of Business of Target and Buyer relating to the business of Business of Business of Business of Target and Buyer relating to the business of Business of Business of Business of Target and Buyer relating to the business of Business of

Held discussions with members of the senior managements of Target and the general partner of Buyer with a businesses and prospects of Target and Buyer, respectively, and reviewed certain synergies and other benefits a the management of the general partner of Buyer to be realized from the

Reviewed public information with respect to certain other companies in lines of business we believe to be gener in evaluating the businesses of Target and Buyer,

Reviewed the financial terms of certain business combinations involving companies in lines of business we generally relevant in evaluating the businesses of Target and Buyer,

Reviewed historical prices and trading volumes of Target Common Shares and Buyer Co

Reviewed the potential pro forma financial impact of the Transaction on Buyer based on the financial forecas above relating to Target and

Conducted such other financial studies, analyses and investigations as we deemed

We have assumed and relied upon the accuracy and completeness of the foregoing information, without independent verifical information. We have not conducted any independent valuation or appraisal of any of the assets or liabilities (contingent or otherwing or Buyer or concerning the solvency or fair value of Target or Buyer, and we have not been furnished with any such valuation Management of the general partner of Buyer has advised us that (i) financial forecasts prepared by Buyer with respect to the fut performance of Target reflect the best currently available estimates and judgments of the anticipated future financial performance of (ii) publicly available forecasts with respect to Buyer reflect the best currently available estimates and judgments of the anticipated future financial performance of Buyer. Accordingly, with your consent, for purposes of our analysis of Target and Buyer, we have financial forecasts. We have assumed, with your consent, that the financial forecasts utilized in our analyses, including those related and other benefits, have been reasonably prepared on bases reflecting the best currently available estimates and judgments as financial performance of Target and Buyer,

Further, our opinion is necessarily based on economic, monetary, market and other conditions as in effect on, and the information available to us as of, the date hereof. We assume no responsibility for updating or revising our opinion based on circumstant occurring after the date hereof. We do not express any opinion as to the prices at which Buyer Common Units or Target Common trade at any time subsequent to the announcement of the Transaction. As you know, we have been retained only to provide this opinion does not address the relative merits of the Transaction as compared transaction or business strategy in which Buyer might engage or the merits of the underlying decision by Buyer to engage in the

In rendering our opinion, we have assumed, with your consent, that the Transaction will be consummated on the terms des Agreement, without any waiver or modification of any material terms or conditions. We also have assumed, with your consent, that of necessary governmental, regulatory or third party approvals and consents for the Transaction will not have an adverse effect on Buy the Transaction. We do not express any opinion as to any tax or other consequences that might result from the Transaction, nor does address any legal, tax, regulatory or accounting matters, as to which we understand that you obtained such advice as you deemed not qualified professionals. We express no view or opinion as to any terms or other aspects (other than the Exchange Ratio to the extension of the Transaction, including, without limitation, the form or structure of the Transaction or any agreements or an entered into in connection with, or contemplated by, the Transaction. In addition, we express no view or opinion as to the fairness or an auture of, or any other aspects relating to, the compensation to any officers, directors or employees of any parties to the Transaction of such persons, relative to the Exchange Ratio of such persons, relative to the Exchange Ratio of such persons.

Lazard Fréres & Co. LLC ("Lazard") is acting as financial advisor to the Independent Directors constituting the Conflicts Co "Independent Directors") of the Board of Directors of KKR Management LLC, the general partner of Buyer, in connection with the and will receive a fee for such services, payable upon the rendering of this opinion. We in the past have provided, currently are provide future may provide certain investment banking services to Buyer and certain of its affiliates, for which we have received and compensation, including

two years, advising Buyer or an affiliate thereof with respect to its proposed acquisition of SBB/Telemach Group and advisin company of Buyer or an affiliate thereof with respect to a potential transaction, and having advised an affiliate of Buyer in connect provision of debt financing to Uralita, Capmark Financial Group in connection with its restructurings and related transactions, and other affiliates or portfolio companies with repsect to transactions that were not consummated. In addition, in the ordinary course, La Holdings LLC (an entity indirectly owned in large part by current and former managing directors of Lazard) and our and the affiliates and employees may trade securities of Buyer, Target and certain of their respective affiliates for their own account accounts of their customers and, accordingly, may at any time hold a long or short position in such securities, and may also transactions that were not consummated. In addition, in the ordinary course, La Holdings LLC (an entity indirectly owned in large part by current and former managing directors of Lazard) and our and the affiliates and employees may trade securities of Buyer, Target and certain of their respective affiliates. The issuance of this opinion was approved by Committee.

Our engagement and the opinion expressed herein are for the benefit of the Independent Directors (in their capacity as such) and is rendered to the Independent Directors in connection with their evaluation of the Transaction. Our opinion is not intended to constitute a recommendation to any shareholder as to how such shareholder should vote or act with respect to the Transaction of the Tran

Based on and subject to the foregoing, we are of the opinion that, as of the date hereof, the Exchange Ratio provided for in the is fair, from a financial point of vie

Very truly yours,

LAZARD FRERES & CO. LLC

By:

Joseph R. Cassanelli Managing Director

D-3

IN THE COURT OF CHANCERY OF THE STATE OF DELAWARE

In re KKR FINANCIAL HOLDINGS LLC

SHAREHOLDER LITIGATION : C.A. No. 9210-CS

:

VERIFIED CONSOLIDATED CLASS ACTION COMPLAINT

1. Pompano Beach Police & Firefighters' Retirement System, Robert A. Corwin, Eric Greene, Margaret DeMauro and Pipe Union No. 120 Pension Fund (collectively, "Plaintiffs"), by and through their undersigned counsel, bring this stockholder class action of themselves and the other public stockholders of KKR Financial Holdings LLC ("KFN" or the "Company"), a Delaware limit company, against KFN, members of KFN's board of directors (the "Board"), KKR & Co. L.P. ("KKR"), a Delaware limited partnership, and Holdings L.P., a Cayman Islands limited partnership, and Copal Merger Sub LLC, a Delaware limited liability company, alleg of fiduciary duties and aiding and abetting thereof in connection with KKR's attempt to acquire KFN for inadequate consideration and December 16, 2013, (the "Proposed Transaction"), pursuant to an Agreement and Plan of Merger (the "Merger Agreement"), to to of the Company's public stockholders. Plaintiffs allege the following upon knowledge as to themselves and upon information and be other matters.

NATURE OF THE ACTION

- 2. KFN is a publicly-traded specialty finance limited liability company whose business is generating income and capital primarily through investing in sub-investment grade corporate debt securities. KFN's primary asset and reason for existence is subordinated notes in collateralized loan transactions ("CLOs") that finance the leveraged buyout activities of KKR the entity that
- 3. KFN itself has no employees or operations. Instead, KKR, through its affiliates, provides those personnel and services p Amended and Restated Management Agreement entered into as of May 4, 2007, by and among KFN, KKR Financial Corp., and KI Advisors LLC, as amended (the "Management Agreement"). The Management Agreement grants KKR and its affiliates the absolute select, acquire and manage KFN's investment portfolio as well as supporting administrative functions, for which KFN pay management and incentive in the select of the s
- 4. The Management Agreement gives KKR complete control over every aspect of KFN's business. As KFN's most recent makes abundantly clear, KFN has "no separate facilities and [is] *completely reliant* on [its] Manager (*i.e.*, KKR), which has significate as to the implementation and execution of our business and investment strategies and our risk management practices." (Empletely reliant)
- 5. The Management Agreement renews automatically each year and, as explained in greater detail below, can be terminate certain circumstances with nearly six months advance notice to KKR and the payment of a termination fee (the "Termination Fee") to *four times* the average annual base management and incentives fees from the prior two years. This Termination Fee runs into the millions of dollars and, in light of KFN's size, is completely preclusive of any alternative bidder accordingly.
- 6. Pursuant to the terms of the Proposed Transaction, KKR will acquire all of KFN's outstanding stock by converting each K share into the right to receive 0.51 common units of KKR (the "Exchange Ratio"). Based on KKR's common unit price December 16, 2013, prior to the announcement of the Proposed Transaction, this implies a price of only \$12.79 per KFN share in a d approximately \$2.6 billion. Upon completion of the Proposed Transaction, KFN will become a wholly-owned subsidiary of

Table

- 7. The Proposed Transaction is subject to entire fairness review because KKR (the acquirer) controls KFN (the target (i) KKR provides all of KFN's management and operations pursuant to the Management Agreement; (ii) all of KFN's officers are e KKR and its affiliates; and (iii) the Termination Fee due to KKR under the Management Agreement is a prohibitive barrier to the a KFN by any potential purchaser other.
- 8. Even if KKR did not control KFN as matter of Delaware law, which it does, the Proposed Transaction would still be sub fairness review because a majority of the KFN Board is not disinterested or independent. Six of the current twelve members of the were originally installed by KKR to the board of directors of KFN's predecessor when that entity was still a privately held company least eight of KFN's twelve directors have longstanding relationships with KKR and/or one another that rendered them incapable of assessing the merits of the Proposed Transaction. Thus, even if KKR did not completely dominate the affairs of the Company, the lack of independence nonetheless rendered the Proposed Transaction a *fait accompli* and would call into question the impartiality of
- 9. The Proposed Transaction does not come close to satisfying that demanding standard of review to prove entire fairness. leading to the KFN Board's entry into the Merger Agreement with KKR was seriously and materially flawed and the Proposed Transaction to fairly compensate KFN's stockholders. At the time the Proposed Transaction was announced, KFN's common shares were one-year low, while KKR common units traded at a one-year high. Moreover, KKR, as KFN's manager, has unique access information concerning the value of KFN's assets (information that, in turn, is largely based on KKR modeling because there are n market prices for the vast majority of KFN's holdings), and understood that KFN's share price was temporarily depressed. Indeed, a October 2013, analysts at Sandler O'Neill + Partners, L.P. ("Sandler O'Neill"), the Transaction Committee's (defined below) or advisor, increased their share price target for KFN to \$25.00 per share, 139.23% higher than the Company's then-current according to an October 15, 2013 article on TickerReport.com. The same article notes that in October 2013, analysts at Credit Su KFN shares at \$21.00 per share, also significantly higher than that agreed to in the Proposed Transaction. Meanwhile, KFN did not to explore alternatives to a sale to KKR at a price that K
 - 10. The Board created a transaction committee of purportedly independent directors (the "Transaction Committee") for evaluating the Proposed Transaction a transparent attempt to insulate the Proposed Transaction from close judicial scrutiny described herein, at least half of the directors appointed to the committee suffered from disabling conflicts of interest. Further, condirectors who were not members of the Transaction Committee repeatedly took over key aspects of the negotiations with KKR and themselves into the committee's deliberations, shattering any semblance of a fair and independent directors.
- 11. To make matters worse, the Transaction Committee retained a financial advisor, Sandler O'Neill, on terms that gave Sandle unreasonable incentive to recommend the Proposed Transaction. Specifically, \$15 million of Sandler O'Neill's \$17 million fee (at 88.2%) was wholly contingent on consummation of the Proposed Transaction. Given that Sandler O'Neill's engagement was limited only KKR's proposal, and Sandler O'Neill was never given the chance to explore alternatives (thus obviating the usual justification bankers on a full or nearly-full contingency basis), it was no surprise that Sandler O'Neill delivered a hopelessly flawed and sla opinion in support of the Proposed
 - 12. Moreover, the Proposed Transaction will leave KFN's stockholders with a minority share of KKR's outstanding common KFN's operating agreement provides that the fiduciary duties of KFN directors are "generally consistent with those of a director of corporation," the KKR partnership agreement expressly excludes "any duties or liabilities, including any fid-

Table

to the Partnership, any Limited Partner or any other Person bound by the [Partnership] Agreement." Thus, after consummation of transaction, current KFN stockholders will be left with a minority interest in a limited partnership in the form of unit interests with e protection against director misconduct an inferior set of rights compared to those held by KFN stockholders today further upurported fairness of the Proposed

13. Accordingly, Plaintiffs bring this action individually and as a class action on behalf of the public stockholders of KFN to for the breaches of fiduciary duties al

PARTIES

- 14. Plaintiff Pompano Beach Police & Firefighters' Retirement System is a stockholder of KFN and has owned shares of KFI the relevant time period al
- 15. Plaintiff Robert A. Corwin is a stockholder of KFN and has owned shares of KFN throughout the relevant time period al
 - 16. Plaintiff Eric Greene is a stockholder of KFN and has owned shares of KFN throughout the relevant time period al
- 17. Plaintiff Margaret DeMauro is a stockholder of KFN and has owned shares of KFN throughout the relevant time period al
- 18. Plaintiff Pipefitters Local Union No. 120 Pension Fund is a stockholder of KFN and has owned shares of KFN throughout time period al
- 19. Defendant KFN is a Delaware limited liability company. Its principal executive offices are located at 555 California Street San Francisco, California 94104. KFN is the successor to KKR Financial Corp. ("KKR Financial"), a real estate investment tr incorporated in Maryland. KKR Financial completed an initial private placement of shares of its common stock in August 2004, an its initial public offering of common stock in June 2005. KKR Financial underwent a restructuring on May 4, 2007, pursuant to whi a subsidiary of KFN, which had been formed on January 17, 2007, and each outstanding share of KKR Financial common stock w into one common share of KFN common stock. KFN's shares are publicly traded on the New York Stock Exchange ("NYSE") under
 - 20. Defendant Tracy Collins ("Collins") has been a KFN director since 2007. Defendant Collins served as a director of KF from August 2006 throug
 - 21. Defendant Robert L. Edwards ("Edwards") has served as a member of the KFN Board since November 2011. He is curren Executive Officer ("CEO") of Safeway Inc. ("Safeway"), an American supermarket chain operator. KKR acquired Safeway leveraged buyout, and took the company public again in 1990. KKR co-founder George Roberts ("Roberts") served on the Safeway February 2001, and co-founder Henry Kravis ("Kravis") served on the Safeway board until
- 22. Defendant Craig J. Farr ("Farr") is the President and CEO of KFN and a member of its Board. He was appointed to the July 24, 2013, following the retirement of former CEO William Sonneborn. Farr joined KKR in 2006. He also serves as Head of Management LLC, a subsidiary of KKR, which shares its principal executive offices with KFN. Farr is also a Member of KKR, over Capital Markets and Origination business, and sits on its Risk
 - 23. Defendant Vincent Paul Finigan, Jr. ("Finigan") has been a KFN director since 2007. Defendant Finigan served as a dire Financial from February 2006 throug

<u>Table</u>

- 24. Defendant Paul M. Hazen ("Hazen") is the Chairman of the KFN Board, and has been a member of the KFN Board Defendant Hazen served as Chairman of KKR Financial from July 2004 through May 2007. Hazen is also the Chairman of Accel-Ki ("Accel-KKR"), a private equity firm specializing in technology that was co-founded by a KKR affiliate. In 2005 Hazen joined Alias, a provider of 3D graphics technology and services, after the company was acquired by Accel-KKR. Hazen is a Special Adv and sits on the board of directors of KSL Recreation Group Inc. with KKR founders Kravis and Roberts. Additionally, Hazen also board of Safeway, and served on the board of Xstrata AG ("Xstrata"). Hazen was employed with Wells Fargo Bank ("Wells Fargo") where he held several high-ranking positions, including Preside
- 25. Defendant R. Glenn Hubbard ("Hubbard") has been a KFN director since 2007, and is KFN's Lead Independent Director Hubbard served as a director of KKR Financial from October 2004 through May 2007. Since 1994, Hubbard has been a professor of at the Graduate School of Business of Columbia University. In 2010, KKR co-founder Kravis pledged a \$100 million donation to business school. Kravis is a co-chair of the business school's board of overseers, and one of the business school's buildings is nam Hubbard also previously served on the board of Capmark Financial Group Inc. (
- 26. Defendant Ross J. Kari ("Kari") has been a KFN director since 2007, and is Chair of the Audit Committee. Defendant Kardirector of KKR Financial from August 2004 through May 2007. In addition, Kari was employed at Wells Fargo for 18 years, included Financial Officer from 1998 the
- 27. Defendant Ely L. Licht ("Licht") has been a KFN director since 2007. Defendant Licht served as a director of KKR Fi April 2005 through May 2007. From March 2007 to February 2009 he was Executive Vice President and Chairman of Cred Capmark, and from March 2009 to August 2009 he was a consultant to the company. Licht was employed at Wells Fargo for 18 years as Chief Credit Officer from November 1998 to December 2001, and Executive Vice President for Credit Administration from February 2009.
 - 28. Defendant Deborah H. McAneny ("McAneny") has been a KFN director since 2007. Defendant McAneny served as a dire Financial from April 2005 throug
- 29. Defendant Scott C. Nuttall ("Nuttall") has been a KFN director since 2007. Defendant Nutall served as a director of KFR from July 2004 through May 2007. Nuttall is a Member of KKR and sits on its Management Committee. He joined KKR in 1995 KKR's Global Capital and Asset Management Group. He is actively involved in funds affiliated with KKR, and has played a significant number of KKR's private equity investments. He is also a member of the board of directors of First Data Corp., a company which K in 2007. Nutall also serves on the board of directors of Kindercare Learning Centers LLC with KKR founders Kravis
 - 30. Defendant Scott Ryles ("Ryles") has been a KFN director since Nov
 - 31. Defendant Willy Strothotte ("Strothotte") has been a KFN director since 2007. Defendant Strothotte served as a direction January 2007 through May 2007. Strothotte was also Chairman of Xstrata, a position which he held from 1
 - 32. Defendants Collins, Edwards, Farr, Finigan, Hazen, Hubbard, Kari, Licht, McAneny, Nuttall, Ryles, and Strothotte at collectively hereinafter as the "Individual I

Table

- 33. Defendant KKR is a Delaware limited partnership. Its principal executive offices are located at 9 West 57th Street, Suit York, New
- 34. Defendant KKR Fund Holdings L.P. is an exempted limited partnership formed under the laws of the Cayman Islands. executive offices are located at 9 West 57th Street, Suite 4200, New York, New
 - 35. Defendant Copal Merger Sub LLC is a Delaware limited liability company. Its principal executive offices are locat 57th Street, Suite 4200, New York, New

JURISDICTION AND VENUE

- 36. This Court has jurisdiction over this action and each Defendant named herein. KFN is a Delaware limited liability com Individual Defendants are its directors. In addition, all Defendants have minimum contacts sufficient to render the exercise of jurisd courts of this State proper. Section 9.9 of the Merger Agreement specifies that "[e]ach of the parties irrevocably and unconditionall any Action arising out of or relating to this Agreement ... shall be brought and determined exclusively in the Court of Chancery or
 - 37. As KFN is a Delaware limited liability company, this Court has jurisdiction over KFN pursuant to 6 Del.
 - 38. As each of the Individual Defendants are managers of a Delaware limited liability company, this Court has jurisdiction over Individual Defendants pursuant to 6 Del.
 - 39. As KKR is a Delaware limited partnership, this Court has jurisdiction over KKR pursuant to 6 Del. C. § 17-105. In addit KKR participates materially in the management of KFN, this Court has jurisdiction over KKR pursuant to 6 Del.
 - 40. This Court has jurisdiction over KKR Fund Holdings L.P. pursuant to 10 De
 - 41. This Court has jurisdiction over Copal Merger Sub LLC, a Delaware limited liability company, pursuant to 6 Del.

SUBSTANTIVE ALLEGATIONS

I. BACKGROUND OF KKR

- 42. Defendant KKR is a private equity firm based in New York that specializes in leveraged buyouts. The company was foun and today it is a multinational investment firm with approximately \$90.2 billion in assets under management as of September completed an initial public offering on July 15, 2010, and its common units trade under the ticker symbol "KKR" o
- 43. KFN was spun-out from Defendant KKR in 2004, but did not have its initial public offering until June 2005. Current publicly-traded specialty finance company whose business is generating income and capital appreciation. Although it was initially on REIT, the Company accomplishes those goals primarily through investing in sub-investment grade securities, primarily corpor subordinated notes in CLOs (though KFN has also invested in other areas, including natural resources and real estate). The vas KFN's holdings and liabilities are so-called "Level 2" and "Level 3" assets, meaning that there are no observable market prices for th and liabilities. KFN values those assets and liabilities by applying a valuation model, which relies on input, estimations and judgment itself supplies a
 - 44. According to KKR's Form 10-K filed with the U.S. Securities and Exchange Commission ("SEC") on February 22 December 31, 2012, KKR owned only 0.1% of KFN's outstanding shares (0.35% if KKR were to exercise all of its vested options

Table

45. Despite KFN's status as a publicly-traded entity, KFN is completely controlled by KKR. KFN has no employees of According to the Company's Form 10-K filed on February 28, 2013, all of KFN's operations are provided pursuant to the Agreement by KKR Financial Advisors LLC ("KFA"), a wholly-owned subsidiary of KKR Asset Management LLC, who wholly-owned subsidiary of KKR. KFN's February 28, 2013 Form 10-

Our Manager is responsible for our operations and performs all services and activities relating to the management of liabilities and operations. Pursuant to the terms of the Management Agreement, our Manager provides us with our management along with appropriate support personnel. All of our executive officers are employees or members of KKR or one of the Management Agreement, our Manager provides us with our management of the Management Agreement, our Manager provides us with our management of the Management Agreement, our Manager provides us with our management of the Management Agreement, our Manager provides us with our management of the Management Agreement, our Manager provides us with our management of the Management Agreement, our Manager provides us with our management of the Management Agreement, our Manager provides us with our management along with appropriate support personnel.

* * :

We are highly dependent on our Manager and may not find a suitable replacement if our Manager terminates the M

We have no employees. Our Manager, and its officers and employees, allocate a portion of their time to businesses and a are not related to, or affiliated with, us and, accordingly, its officers and employees do not spend all of their time in activities and our investment portfolio We have no separate facilities and are completely reliant on our Manager significant discretion as to the implementation and execution of our business and investment strategies and our risk in practices. (Employees.)

- 46. Under the Management Agreement, KFA is responsible for: (i) selecting, purchasing and selling KFN's investment financing and risk management; and (iii) providing investment advisory services to KFN. As noted above, KKR also provides critic the models that KFN uses to value its assets an
- 47. For its services, KFN is required to pay KKR and its affiliates a "Base Management Fee" of one-twelfth of the Comp multiplied by 1.75%. The Base Management Fee for the year ended December 31, 2012, was \$28.2 million, and \$26.3 million for th December 31, 2011. The Management Agreement also provides for incentive compensation to KKR and its affiliates on a quarterly legistrate complex formula described in detail in the Company's February 28, 2013, Form 10-K. For the year ended December 31, 201 \$37.6 million in incentive fees to KKR and its affiliates, and \$34.2 million for the year ended December 31, 2011. KKR and its affiliates for \$10.0 million in the top of the year ended December 31, 2012. KFN also maintains the same executive offices as KKR Asset Management Agreement, and KFN reimbursed KKR and its affiliates for \$10.0 million in the year ended December 31, 2012. KFN also maintains the same executive offices as KKR Asset Management Agreement, and KFN reimbursed KKR and its affiliates for \$10.0 million in the year ended December 31, 2012. KFN also maintains the same executive offices as KKR Asset Management Agreement and KFN reimbursed KKR and its affiliates for \$10.0 million in the year ended December 31, 2012. KFN also maintains the same executive offices as KKR Asset Management Agreement and KFN reimbursed KKR and its affiliates for \$10.0 million in the year ended December 31, 2012. KFN also maintains the same executive offices as KKR Asset Management Agreement and KFN reimbursed KKR and its affiliates and the Year ended December 31, 2012. KFN also maintains the same executive offices as KKR Asset Management Agreement and KFN reimbursed KKR and its affiliates and the Year ended December 31, 2012. KFN also maintains the same executive offices as KKR Asset Management Agreement and the Year ended December 31, 2012.
 - 48. The Management Agreement renews automatically every year by its own terms. While KFN and its stockholders have a under the Management Agreement to terminate the agreement, they can do so only under certain conditions, generally with nearl advance notice, and with a Termination Fee of four times the sum of the average annual base management and incentive fees for years preceding the termination. The Company's February 28, 2013 Form 10-

Our independent directors review our Manager's performance annually and the Management Agreement may be termina (upon 180 day prior written notice) upon the affirmative vote of at least two-thirds of our independent directors, or by holders of a majority of our outstanding common shares, based upon (1) unsatisfactory performance by the Manager that detrimental to us or (2) a determination that the management fees payable to our Manager are not fair, subject to our Mana prevent such a termination under this clause (2) by accepting a mutually acceptable reduction of management fees. We mu

180 day prior written notice of any such termination and our Manager will be paid a termination fee equal to four time the average annual base management fee and the average annual incentive fee for the two 12-month periods preceding termination, calculated as of the end of the most recently completed fiscal quarter prior to the date of termination. (Empl

- 49. For example, if the Management Agreement had been terminated by KFN on December 31, 2012, KFN would have bee pay KKR approximately \$252.6 million. As of the Company's November 6, 2013 Form 10-Q, the Company had only at \$221.9 million in cash and cash equivalents on its balance sheet. Thus, even in the limited circumstances where KFN may to Management Agreement, the terms of the agreement would make doing so prohibitively expensive, tightly securing KKR's contrained its operations (and making it impossible for anyone other than KKR to acquire KFN absent KKR's consent to waive its Term
- 50. KFN's portfolio of investments (selected by KKR) consists largely of debt securities used to finance private equity leveral of other companies, primarily led by KKR. According to a December 17, 2013 article in the FINANCIAL TIMES, KFN has investigated acquisitions including Energy Future Holdings, Biomet, Del Monte, First Data, HCA and Toys "R" Us. Over 90% of KFN's portfoliating rate debt indexed to the 3-month LIBOR, according to the FINANCIAL TIMES and to the Company's Form 10-K filed with
- 51. In July 2013, long time CEO of the Company, William Sonneborn ("Sonneborn"), was replaced by Defendant Farr, wheld any executive management position with comparable responsibilities to those of a CEO of a publicly traded company. While prior affiliation with KFN, he had worked for KKR since 2007. In less than three months and immediately after being informed that make an offer for the Company, Farr announced that KFN was dramatically altering its corporation.

II. DEFENDANTS ANNOUNCE THE PROPOSED TRAI

- 52. On December 16, 2013, KFN filed a Form 8-K with the SEC stating that it had entered into an Agreement and Plan of M KFN, KKR, KKR Fund Holdings L.P. and Copal Merger Sub LLC, dated as of December 16, 2013, pursuant to which KFN wow wholly-owned subsidiary of KKR Fund Holdings L.P. and Copal Merger Sub LLC, dated as of December 16, 2013, pursuant to which KFN wow wholly-owned subsidiary of KKR Fund Holdings L.P. and Copal Merger Sub LLC, dated as of December 16, 2013, pursuant to which KFN wow wholly-owned subsidiary of KKR Fund Holdings L.P. and Copal Merger Sub LLC, dated as of December 16, 2013, pursuant to which KFN wow wholly-owned subsidiary of KKR Fund Holdings L.P. and Copal Merger Sub LLC, dated as of December 16, 2013, pursuant to which KFN wow wholly-owned subsidiary of KKR Fund Holdings L.P. and Copal Merger Sub LLC, dated as of December 16, 2013, pursuant to which KFN wow wholly-owned subsidiary of KKR Fund Holdings L.P. and Copal Merger Sub LLC, dated as of December 16, 2013, pursuant to which KFN wow wholly-owned subsidiary of KKR Fund Holdings L.P. and Copal Merger Sub LLC, dated as of December 16, 2013, pursuant to which KFN wow wholly-owned subsidiary of KKR Fund Holdings L.P. and Copal Merger Sub LLC, dated as of December 16, 2013, pursuant to which KFN wow wholly-owned subsidiary of KKR Fund Holdings L.P. and Copal Merger Sub LLC, dated as of December 16, 2013, pursuant to which KFN wow wholly-owned subsidiary of KKR Fund Holdings L.P. and Copal Merger Sub LLC, dated as of December 16, 2013, pursuant to which KFN wow wholly-owned subsidiary of KKR Fund Holdings L.P. and Copal Merger Sub LLC, dated as of December 16, 2013, pursuant to which KFN was also because the copal Merger Sub LLC, dated as of December 16, 2013, pursuant to which KFN was also because the copal Merger Sub LLC, dated as of December 16, 2013, pursuant to which KFN was also because the copal Merger Sub LLC, dated as of December 16, 2013, pursuant to which KFN was also because the copal Merger Sub LLC, dated as of December
- 53. Pursuant to the Merger Agreement, at the effective time of the merger, each KFN common share will be converted into receive 0.51 common units of KKR. KKR common units closed at \$25.08 on December 16, 2013, prior to the announcement of the Transaction, implying a per share value of \$12.79 for each share of KFN under the Exchange Ratio, representing an approx premium to KFN's closing price of \$9.45 on December 16, 2013 a time when KFN's shares were artificially depressed and trading no low, as explained in greater detail below. The value of the Proposed Transaction at the time of announcement was approximately Since the announcement of the Proposed Transaction, KKR shares have fallen, closing at \$23.83 per share on February
- 54. KKR issued a press release on December 16, 2013 touting the benefits of the Proposed Transaction. The press release not will gain access to a \$2.9 billion investment portfolio that it already manages and thus knows well, while preserving KFN's "attrac structure" including over \$1 billion in long-term fixed rate debt and perpetual preferred securities. KKR also stated that the Proposed will give it a substantially increased capital base, increasing on a book value basis as of September 30, 2013, to \$9.3 billion from \$7 support its investment activities and "accelerate [the] growth of KKR's new
 - 55. The Proposed Transaction is expected to close in the first

Table

III. THE PROPOSED TRANSACTION IS NOT ENTIRELY FAIR TO KFN'S PUBLIC STOCK

THE KFN BOARD, DOMINATED AND CONTROLLED BY KKR, PLACED ITS OWN INTER THOSE OF KKR AHEAD OF THE INTERESTS OF KFN'S PUBLIC STOCK

56. In approving the Proposed Transaction, the KFN Board breached its fiduciary duties of loyalty, due care and good faith ow public stockholders by placing Board members' own financial interests and KKR's financial interests ahead of those of I

KKR's Unwillingness To Waive The Termination Fee Eliminated The Possibility Of A Compo

- 57. KKR's refusal to waive the Termination Fee built into the Management Agreement between KKR and KFN is powerful & KKR would not, and did not, allow KFN to conduct a fair sales process. Indeed, any "bidding process" that might have existed
- 58. For KKR to demonstrate that it did not control KFN throughout the negotiations and sales process, it necessarily would lead allowed KFN to explore alternatives such as an alternative transaction or moving forward as a standalone entity. Because the Fee would have imposed a cost of hundreds of millions of dollars on KFN more cash than KFN had on hand there was no chance the bidder would emerge. Accordingly, unless KKR had been willing to waive, or at the very least modify, the Transaction Fee, there been no bona fide search for
- 59. Instead, and not surprisingly, the Termination Fee served its obvious purpose: to prevent any other bidder from e ultimately acquiring KFN. Although KKR was not obligated to waive the Transaction Fee, its refusal to do so exposes the Proposed to the exacting entire fairness standard, which it can
- 60. The Termination Fee, along with the deal protections discussed below, eliminated any possibility that a competing emerge for KFN. Any potential third-party acquirer interested in KFN would have to make an unsolicited bid without access to diligence information, knowing that: (i) more than 25% of the outstanding shares will likely be voted in support of a deal we explained below); (ii) KKR has access to nonpublic information about the true value of KFN's assets and liabilities (which were KKR models); (iii) KKR has unlimited matching rights; and (iv) the bidder will have to wait nearly six months and pay hundreds of dollars on top of the cost of acquiring KFN in order to acquire the Company a virtually insurmountable burden to any potential third And, during that six months, KKR would be able to use its control over KFN to threaten to impair the Company in order to thwart and bids. Indeed, on December 17, 2013, Deutsche Bank downgraded KFN from "buy" to "hold," noting the unlikelihood of any emerging, according to a December 17, 2013 article on Street

KKR Stacked The KFN Board With Conflict

61. KFN's predecessor, KKR Financial, was founded by KKR in July 2004 and completed its initial public offering of common stock in June 2005. Defendants Hazen and Nuttall sat on the KKR Financial board since its inception. In August and O respectively, defendants Kari and Hubbard were appointed to the KKR Financial board. In April 2005, defendants Licht and McAner KKR Financial board. As described above, KKR Financial did not commence its initial public offering until June 2005. Thus, on current members of the KFN Board defendants Hazen, Nuttall, Kari, Hubbard, Licht, and McAneny are individuals who were KKR to manage and oversee a 100% owned K

<u>Table</u>

- 62. Defendant Farr is a KKR Member and the Head of KKR Asset Management LLC, a KKR affiliate and the parent compa manager, KFA. KKR describes Defendant Nuttall as Member & Head of its Global Capital and Asset Management Group. Nu director of First Data Corp., a company KKR acquired in 2007, and serves on the board of directors of Kindercare Learning Center KKR founders Kravis and Roberts. Given their conflicting interests, defendants Farr and Nuttall are incapable of negotiating independent.
 - 63. Defendant Hazen similarly has longstanding ties with KKR. Hazen currently serves as a Special Advisor to KKR, as served on KFA's Investment Committee alongside KKR co-founders Kravis and Roberts. Hazen is also the Chairman of Accel affiliate, of which Roberts and Kravis serve as Spec
- 64. Hazen has previously demonstrated his inability or unwillingness to negotiate at arms-length with KKR, and KKR has exploited that weakness. Hazen was a director of Safeway from 1990 through 2012, and, for a portion of that time, served together founding partner Roberts, as well as James H. Greene, Jr. ("Greene"), an Advisory Partner of KKR, and Robert I. MacDonnell ("M. Roberts' brother-in-law and a retired KKR Partner. KKR acquired Safeway in 1986, and took it public again in 1990. Under Hazen's Safeway engaged in numerous related-party transactions with KKR. I
 - (a) Greene and Roberts, and KKR founding partner Kravis, served together on the board of Randall's F ("Randall's"). Safeway merged with Randall's in 1999, while Greene and Roberts sat on the Safeway board, paying total c of \$1.3 billion to acquire all the outstanding shares of Randall's. At the time, entities affiliated with KKR owned approxim Randall's and 9% of Safeway. RFM Acquisition LLC, a KKR affiliate, received \$475.7 million and 8,027,964 shares of Saferits stake in Randall's. KKR also received \$8.6 million for advisory services in connection with the merger. In 2002 an KKR had liquidated its last Safeway shares, Safeway took a series of write-downs totaling \$1.3 billion, equal to \$1.3 billion investment
 - (b) Safeway engaged in multi-million dollar real estate transactions with partnerships owned in part by KKR or KI directors. Between 1990 and 2001, Safeway transferred or sold 172 properties valued at over \$130 million to a partnersh by a KKR affiliate, Pacific Resources Associates ("PRA"), and a partnership formed in 1987 between Safeway and PI Development Associates ("PDA"). From 1990 through 2000, Safeway paid PDA \$19 million in real estate management f PDA was dissolved in 2000, with the remaining properties apportioned between Safeway paid PDA was dissolved in 2000.
 - (c) In 2002, Safeway sold 8 properties to another limited partnership, Pacific Realty Associates ("PacTrust"), for and paid PacTrust (or PacTrust partnerships) over \$2 million for rent and related expenses. KKR-affiliated directors Green and MacDonnell held an ownership interest
 - (d) Between 1993 and 2002, Safeway also paid Carmel Valley Partners over \$2 million in rent. PRA and Pac 80 percent of Carmel Val
 - (e) Safeway had a strategic alliance with, and 50 percent voting interest in, GroceryWorks Holdings, Inc. ("Groc which served as Safeway's exclusive on-line grocery channel. Accel-KKR, of which Hazen is Chairman, owned approxim GroceryWorks. Roberts, Greene, and MacDonnell had indirect interests in Accel-KKR which totaled approximately 2%
 - (f) Between 1990 and 1998, Safeway paid KKR approximately \$12 million in consulting fees. Initially, these co were an extension of arrangements made when Safeway was privately held by KKR, whereby consulting fees automatica

<u>Table</u>

- 65. The transactions described above all of which occurred under Hazen's directorship with Safeway caused Gove International ("GMI"), an independent corporate governance monitoring firm, to issue a report stating that "[d]ue to the relations Safeway and KKR including significant related party transactions and board representation on behalf of KKR executives, GMI has company's Board Accountability section as a significant governance of the company's Board Accountability section as a significant governance of the company's Board Accountability section as a significant governance of the company's Board Accountability section as a significant governance of the company's Board Accountability section as a significant governance of the company's Board Accountability section as a significant governance of the company's Board Accountability section as a significant governance of the company's Board Accountability section as a significant governance of the company's Board Accountability section as a significant governance of the company's Board Accountability section as a significant governance of the company's Board Accountability section as a significant governance of the company's Board Accountability section as a significant governance of the company's Board Accountability section as a significant governance of the company's Board Accountability section as a significant governance of the company's Board Accountability section as a significant governance of the company's Board Accountability section as a significant governance of the company's Board Accountability section as a significant governance of the company's Board Accountability section as a significant governance of the company of the
- 66. Similarly, Glass Lewis, an independent proxy advisory firm, warned prior to Safeway's 2003 annual meeting that Sa number of affiliated directors "raises serious concerns about the objectivity and independence of the board and its ability to perform
- 67. Defendant Edwards similarly has ties to Safeway, and therefore KKR, having joined Safeway as its Executive Vice F Chief Financial Officer in March 2004 when there were four KKR-affiliated directors on its board, including KKR co-founder F overwhelming KKR influence caused a public complaint to be made, and pressure to be exerted, by numerous pension fur shareholders, including the California Public Employees' Retirement System, Connecticut Retirement Plans and Trust Funds, New Common Retirement Fund, New York City Employees' Retirement System and the Illinois Board of Investment. Hazen's apport specifically challenged, as the pension funds pointed out that "[h]is position as chairman of a KKR affiliate and as a senior I disqualifies him as an independent director." Edwards specifically advocated for Hazen's appointment to the Safe
- 68. Defendant Kari was employed at Wells Fargo for 18 years, including as Chief Financial Officer from 1998 to 2001. Def was also employed at Wells Fargo for 18 years, including as Chief Credit Officer from November 1998 to December 2001, and Ex President for Credit Administration from February 1990 to November 1998. Defendant Hazen, who, as described above, independence from KKR, was employed with Wells Fargo for 31 years, including as Vice Chairman from 1978 to 1984, Preside from 1984 to 1995, Chairman and President and CEO from 1995 to 1998, and Chairman from 1998 to 2001. Accordingly, the positic corresponding income earned by Kari and Licht during 17 years of their 18-year careers at Wells Fargo were under Hazen's tenure CEO, and/or Chairman of Wells Fargo. As such, Kari and Licht were beholden to Hazen, owed their advancement and success to he years, and therefore are unable to act independent.
 - 69. Further, Defendant Licht was the Executive Vice President and Chairman of Credit Policy of Capmark from March 2007 2009, and served as a consultant to Capmark from March 2009 to August 2009. As of March 31, 2006, an investor group led by 75% of Capmark. Thus, Licht is also indet
 - 70. Defendant Strothotte is also connected to Hazen. Strothotte was the Chairman of Xstrata from 1994 to 2011. Hazen sat o board from May 2000 throug
- 71. Defendant Collins was a managing director with Credit Suisse, which has an ongoing profitable relationship with KKR. I Credit Suisse has been the financial advisor to KKR acquisitions, including the 2013 strategic transactions involving PRA International Brickman Group
- 72. Defendant Hubbard one of the initial KKR-appointed members of KFN's predecessor's board previously served Capmark. Additionally, since 1994, Hubbard has been a professor of economics at the Graduate School of Business of Columbia University Business School. KKR recently made the single largest donation in the history University Business School \$100,000,000. KKR co-founder, co-CEO and board member Kravis is the co-chair of Columbia's board and one of the school's buildings is named after him. The two have such a close relationship that Hubbard has

1	Kari and Light both joined Wells Fargo in 1983, while Hazen was Vic

Table

Kravis as "the one who's kept me most focused on how teaching and research space happens", and the donation was so important to prospects that it eliminated what Hubbard termed "a constraint" on the school's growth by funding a n

73. Thus, the KFN Board consists primarily of members either directly affiliated with KKR or with interrelationships that tie KKR, preventing them from disinterestedly considering the Proposed Transaction, and setting the stage for KKR's opportunistic

The Transaction Committee Is Fraught With Conflicts And Did Not Engage In A Fair Or Reason

- 74. According to KFN's charter, the Affiliated Transactions Committee (consisting of Finigan, McAneny, Ryles and supposed to "represent and assist the independent directors on the Board in their oversight and review of transactions ... by and Company and [KKR] and entities that are controlled by or under the common control of KKR." Rather than empower the Affiliated Committee, the Board created the Transaction Committee, consisting of defendants Collins, Edwards (chair), Finigan, Kari, M
- 75. As described immediately above, three of those six directors Kari, Edwards, and McAneny have significant ties to interested KFN directors. Unsurprisingly, the conflicted Transaction Committee failed to discharge its fiduciary duties and adequate on behalf of KFN's public s
- 76. Moreover, the legal advisor to the Transaction Committee, Wachtell, Lipton, Rosen & Katz ("Wachtell"), has extensive ties to KKR as well. For example, Wachtell advises KKR with relation to its interest in Energy Future Holdings Corp. (formerly which at the time of its \$45 billion buyout was the largest leveraged buyout in history. Despite this ongoing relationship with KKR, been charged with advising the Transaction Committee in its negotiations with KKR. This clear conflict of interest calls into independence of this

The KFN Board Permitted An Unfair Process In Which KKR Used The Management Agreeme Through The Proposed Transaction And Preclude Alternative T

- 77. The Form S-4 Registration Statement that KKR filed with the SEC on January 14, 2014 (the "Proxy") demonstrates that to Transaction is the product of a fundamentally un
 - 78. First, as noted above, the Transaction Committee's makeup was undoubtedly flawed, leaving the committee independently or exert any leverage in negotiation
- 79. Second, the Transaction Committee's creation and operation evokes more of a "box-checking" exercise than it does a gent permit and achieve a true arms-length bargaining process. Indeed, while the Transaction Committee allowed itself to be created and to the final outcome of the negotiations, it did not actually take any steps to protect the interests of KFN's public states.
 - 80. The Transaction Committee allowed a plainly conflicted director, Hazen, to lead and control KFN's role in the deal nego Transaction Committee failed to explore alternatives and, further, allowed a tainted process in which KKR enforced with contractual protections like the Termination Fee, which effectively undermined a fair process. In addition, the Transaction Commi impose defensive measures, such as a poison pill, that could have slowed down the sales process and enabled KI concessions including additional consideration.
 - 81. Indeed, while KKR and KFN created the Transaction Committee to argue for the burden-shifting benefit of having such a the Proxy itself shows that the Transaction Cor

Table

not have any of the hallmarks of the well-functioning and independent special committee that has garnered respect under Delay discussed below, Defendant Hazen was primarily (and almost exclusively) responsible for representing KFN in talks with KKF Special Advisor, in addition to his other KKR-related roles, Hazen was not independent of KKR, and so the fact that he was negotiate against KKR on behalf of KFN exposes the inherent unfairness in the sales process. Indeed, although he was not a m Transaction Committee, Hazen steered the sales process at all times. Thus, KKR controlled the process entirely while the Transaction

- 82. Third, the Transaction Committee did not even meet until after Hazen had already conducted extensive and key discussion including agreeing to let KKR leave the Termination
- 83. According to the Proxy, KKR first expressed its interest in acquiring KFN to Hazen in October 2013. Proxy at 58 expressed its interest in acquiring KFN to Hazen, Hazen raised KKR's interest during a KFN Board meeting on October 22, 2013. During that meeting, the Board granted KKR permission to use confidential information about KFN that KKR gained solely in it KFN's manager, in order to make an acquisition
- 84. The Board did not put in place any meaningful protections, such as a poison pill, that might provide KFN leverage in nego KKR or that would allow KFN to slow down the process in order to give the Company time to explore alternative strategic options, extract more consideration out of KKR. Moreover, there is no indication that the Board considered either refusing to allow KK confidential information, conditioning that use on concessions from KKR, or making that confidential information available to ot
- 85. After the October 22 Board meeting, the conflicted Hazen raised with Defendant Farr who serves as President, CEO, a KFN, and is a senior KKR executive whether KKR would consider modifying or eliminating the Termination Fee in the Manageme *Id.* at 59. Farr then contacted KKR, and was informed that KKR was unwilling to modify or eliminate the Termination Fee. *Id.* at 59 contains no explanation as to why the admittedly and clearly conflicted Farr spearheaded the negotiation of this critical issue betwe KFN. But the Proxy also makes clear that at no time did the Transaction Committee itself actually raise this issue with KKR for the KFN's or the contact of the contact of
- 86. Although the KFN Board was first informed of KKR's interest on October 22, it was not until October 31 nine days in negotiations over the Termination Fee and use of confidential information had already occurred that KFN formed the Transaction C
- 87. On November 21, 2013, Hazen attended a meeting of the Transaction Committee for the purpose of updating the commit on KFN's prospects as a stand-alone entity in the event that a transaction with KKR was not pursued. *Id.* The Proxy does not defend that the Transaction Committee, but the committee then moved forward with its negotiations with KKR. There is not that the Transaction Committee actually considered moving forward as a stand-alone entity, or other alternatives, such as seeking a
- 88. By this point KFN, through the Transaction Committee, had given up any opportunity to extract concessions from KKR secure a more valuable offer for KFN's public shareholders. KFN obtained only meager improvements to KKR's proposal, wi knowing that a deal was inevitable. For example, the Proxy notes that the Transaction Committee recognized that KFN's shares were their one-year low while KKR's units traded near a one-year high, making an all-stock deal potentially disadvantageous to KFN. The Committee apparently raised the possibility of a cash deal, but this idea appears to have been summarily dismissed by KKR, we incentive to give in on this point given its control of

<u>Table</u>

- 89. On December 9, 2013, at the request of the Transaction Committee, Hubbard, whom KFN holds out as its lead independent met with Kravis and Roberts (co-heads and founders of KKR) in a purported attempt to negotiate a higher price for KFN's shareh 61. Notably, Hubbard was not on the Transaction Committee charged with negotiating the deal and in fact has significant conflicts (in detail above). To permit Hubbard, who was beholden to KKR, to lead this negotiation against the top decision-makers at KKR rev Board was more interested in creating the appearance of a fair process than in actually obtaining a higher price for KFN's stockhown meeting with Hubbard, Kravis and Roberts refused to increase the merger consideration above 0.50 KKR units for each KFN shared.
- 90. The next day, December 10, 2013, the KFN Board met with Kravis and Roberts. After that meeting, KKR made a token in offer, to a "best and final offer" of 0.51 KKR units per KFN share. *Id.* At the request of Defendant Edwards, chair of the Transaction Hazen contacted KKR and asked for consideration of 0.52 KKR units per KFN share. *Id.* KKR stood firm at 0.51 KKR units per which the Transaction Committee duly
- 91. On December 13, 2013, the *entire* KFN board (excluding Farr and Nuttall, who are employed by KKR)² met to discuss to Transaction. Only *after* this discussion did the Transaction Committee meet, at which point the Transaction Committee voted to recovered Transaction to the Board. *Id.* at 61-62. The full Board (again excluding Farr and Nuttall) then reconvened and approved the
 - 92. On December 16, 2013, KKR and KFN executed the Merger Agreement, and together issued a joint press release and Proposed Transaction. *Id.* at 63. The deal was announced less than two months after Hazen informed the Board of KKR's interesting Crucially, KFN did not explore *any* alternative deals outside of a potential KFN share repurchase, and did not otherwise take any strength of the competitive bidding for the c

The KFN Board Further Permitted KKR To Lock Up The Proposed Transaction With Unreas Preclusive Deal

- 93. Even with the Management Agreement (including the Termination Fee) effectively barring any reasonable chance of an all for KFN, KKR and the KFN Board determined to take no chances with their low-priced deal. Especially when considered in combine Management Agreement, the Merger Agreement contains a host of deal protection devices intended to, and having the effect of, I reasonable chance of an alternative suitor offering more valuable consideration to KFN's public shareholders, in breach of the Defendants' fiduciary duties to those s
 - 94. The Merger Agreement contains a break-up fee of \$26.25 million significantly increasing the cost of a competing transition interested third-party bidder. While this fee represents an impediment on its own, when coupled with the massive Termination Management Agreement, there can be no doubt that there was never the slightest prospect of anyone but KKR
 - 95. The Merger Agreement also includes a restrictive "No-Shop" provision barring the Company from soliciting alternative protein third-parties or entering into discussions or negotiations or providing non-public information in connection with any alternative Moreover, the limited "fiduciary out" applies only after the Board receives an unsolicited bid for at least

Although the Proxy does not specifically identify which KFN directors the phrase "directors who were employ encompasses, the Proxy lists only Farr and Nuttall, but not Special Advisor to KKR Hazen, as KKR employees. As appears that the conflicted Hazen participated in this crucial Bo

Table

Company that could reasonably be expected to lead to a competing bid. Unless and until a potential alternative suitor makes a bli least 15% of the Company that the Board determines could lead to a superior proposal, KFN is prohibited from furnishing and diligence information, including nonpublic information concerning the valuation of KFN's assets as

96. The Merger Agreement further includes an unlimited "Matching Rights" provision requiring the KFN Board to submit an proposal to KKR within 24 hours of receiving a competing proposal and giving KKR four business days' notice prior to exercisin "fiduciary out" noted above, while giving KKR the right to propose changes to the Merger Agreement to match any alter discouraging competing bidders from incurring the time and expense of preparing and pursuing a co

97. Together, these provisions unreasonably restrain competition for control of KFN and ensure that KFN's public stockhol have a chance of receiving a topping bid to KKR's undervalued and unfair Proposed

The KFN Shareholder Vote On The Proposed Transaction Is Illusory And There Is Zero Likeli

98. Although the closing of the Proposed Transaction is subject to the approval of the Merger Agreement by an affirmative vo of at least a majority of KFN's common shares, including a majority of shares held by persons other than KKR and its affiliates, KKF ensured that its undervalued and opportunistic buyout is a fat

99. The majority of the minority provision in the Merger Agreement expressly carves out the following from the defini "affiliates" whose votes will not count towards approval: investment funds, other investment advisory vehicles, and portfolio c

100. KKR is party to a strategic partnership agreement with Fidelity Investments, which directly and indirectly holds approxim KFN's outstanding shares and is KFN's single largest shareholder. KFN's CEO, Defendant Farr, personally negotiated this strategic with Fidelity on behalf of KKR. Under the terms of the KKR-Fidelity partnership, Fidelity has exclusive access to all retail secur allocated to KKR in all U.S. public offerings in which KKR participates as an underwriter, including all IPOs and follow-on offer Fidelity has ample reason to vote in favor of the Proposed Transaction favoring its partner KKR, even if it fails to maximize share

101. In addition, as noted above, the Termination Fee of potentially hundreds of millions of dollars provided in the Managemen makes any competing bid economically unfeasible, particularly when coupled with the \$26.25 million termination fee in the Merger Moreover, KKR's unique access to nonpublic KFN information has enabled KKR to evaluate KFN's portfolio, and take advanta factors, in a way that other potential suitors cannot. Indeed, although the market temporarily depressed KFN's stock price becau general economic concerns, KKR enjoyed its unique insight into KFN's asset base to opportunistically time its squeeze out of KFN as noted above, KKR, through several affiliates, provides all of KFN's management and operations, including selecting, acquiring a investments. Thus, KKR was in a position to know, and did know, that KFN's portfolio of investments was comprised of over 90° rate debt instruments. Indeed, as KKR Member and KFN director Nutall explained during KKR's December 16, 2013 investor [a] critical item to understand is that KFN itself does not have any employees. KKR is the manager of KFN's balance sheet, and, as underwritten every asset and manage the portfolio today. So, we are exceedingly familiar with the balance sheet, as we built managed it since 2004." (Emphasis added). Nuttall continued, "these aren't just any assets. They're assets that we know. We like

E-14

underwritten all of them." (Empl

Table

102. Given the opaque nature of KFN's assets, which have no readily observable market prices and rely on KKR mode valuation, only KKR has access to the information necessary to understand the fair market value of KFN's assets, while I stockholders are left with no meaningful way to determine whether they are receiving fair value for their shares under the Proposed

THE PROPOSED TRANSACTION GROSSLY UNDERVA

KKR Timed The Proposed Transaction To Take Advantage Of KFN's Temporarily Depressed

103. KKR timed the Proposed Transaction to take advantage of a huge disparity between the market price of its common unit share price. Indeed, KKR's timing could not have been bette

104. The FINANCIAL TIMES noted on December 17, 2013 that KFN's shares are down more than 10% in 2013, and in December 17, 2013 that KFN's shares are down more than 10% in 2013, and in December 17, 2013 that KFN's shares are down more than 10% in 2013, and in December 17, 2013 that KFN's shares are down more than 10% in 2013, and in December 17, 2013 that KFN's shares are down more than 10% in 2013, and in December 17, 2013 that KFN's shares are down more than 10% in 2013, and in December 17, 2013 that KFN's shares are down more than 10% in 2013, and in December 17, 2013 that KFN's shares are down more than 10% in 2013, and in December 17, 2013 that KFN's shares are down more than 10% in 2013, and in December 17, 2013 that KFN's shares are down more than 10% in 2013, and in December 17, 2013 that KFN's shares are down more than 10% in 2013, and in December 17, 2013 that KFN's shares are down more than 10% in 2013, and in December 17, 2013 that KFN's shares are down more than 10% in 2013, and in December 17, 2013 that KFN's shares are down more than 10% in 2013, and in December 17, 2013 that KFN's shares are down more than 10% in 2013, and in December 17, 2013 that KFN's shares are down more than 10% in 2013, and in December 17, 2013 that KFN's shares are down more than 10% in 2013, and in December 17, 2013 that KFN's shares are down more than 10% in 2013, and in December 17, 2013 that KFN's shares are down more than 10% in 2013, and in December 17, 2013 that KFN's shares are down more than 10% in 2013, and in December 17, 2013 that KFN's shares are down more than 10% in 2013, and in December 17, 2013 that KFN's shares are down more than 10% in 2013, and in December 17, 2013 that KFN's shares are down more than 10% in 2013, and in December 17, 2013 that KFN's shares are down more than 10% in 2013, and in December 17, 2013 that KFN's shares are down more than 10% in 2013, and in 2013, a

105. The following chart illustrates the advantageous timing of the Proposed Transaction's Exchange Ratio which will pay units of KKR for each share of F

106. KFN's stock price drop, particularly from October through the announcement of the Proposed Transaction, was direct KKR's control of the Company. KKR was able to temporarily dampen KFN's performance long enough to enter into the Proposed T a bargain base.

107. Specifically, despite the fact that the Company had invested in an increasing number of real estate projects and was well perpendicular experience massive profits in the oil sector (discussed *infra*), the Company's newly appointed CEO Farr, a KKR employed announced during an October 23, 2013 conference call, that the Company was "going to narrow our new capital commitments to strategies instead of six ... the three areas we are focused on for new capital commitments are CLO [collateralized loan obligated opportunistic credits and then a new, a related area, specialty financed lending." Real estate ventures, an emerging source of percentage of per

108. Not surprisingly, analysts were quick to downgrade the Company on this unexpected news that, moving forward, it wou pursue these profit sources. For example, the very next day, FBR Capital downgraded the Company

Table

109. The market reaction to the unforeseen change in strategy was swift and immediate and mirrored the analyst down Company's stock price, which had been trending upward for the previous month and was approaching its peak for the year, plum stock dropped nearly ten percent in just one week's time, and continued to fall until the days prior to the Proposed Transaction's an hitting a low of less than \$9 per share, a drop in value of almost twenty percent. In fact, the Company was trading below its book value of the unexpected change in management.

110. The volume traded the day following Farr's announcement of the Company's sudden departure from two of its most profit clearly accomplished KKR's goal of inducing investors to sell their shares of KFN *en masse* and thereby drive down the stock Company: well over 8,000,000 KFN shares traded October 24, 2013, more than twice as many shares changed hands than on any strading in more than

111. Because it was able to exert direct downward pressure on the Company's share price, KKR then structured the Proposed T a stock-for-stock merger, allowing it to acquire undervalued KFN shares with its own overv

112. Under Delaware's Limited Liability Company Act, KFN's shareholders are not entitled to appraisal rights in connect Proposed Transaction unless such rights are granted to them under KFN's operating agreement or the Merger Agreement. See 6 Del.

The Amended and Restated Operating Agreement of KKR Financial Holdings LLC, dated as of May 3, 2007, as amended however, states that KFN's shareholders "are not entitled to dissenters' rights of appraisal in the event of a merger, consolidation or contain or any other similar transaction or event." The Merger Agreement likewise does not grant appraisal rights to KFN stockholders stockholders have no alternative to seek fair value for

113. In addition, the KFN Board failed to negotiate for the inclusion of a price collar mechanism in the Merger Agreement to potential decreases in the value of KKR's units subsequent to the announcement of the Proposed Transaction but prior to the February 20, 2014, KKR's units were down approximately 4.9% to \$23.83, reducing the implied value under the Exchapproximately \$12.15 per share. A collar would have provided for adjustments to the Exchange Ratio to protect against potential KKR's unit price price.

114. Events that have transpired since the announcement of the Proposed Transaction have only served to confirm KKR's of timing with respect to the Proposed Transaction. While the price of KKR's units has continued to fall, further negatively impost shareholders because of the lack of a price collar mechanism as noted above, KFN reported net income available to common share \$62 million or \$0.30 per diluted common share for the fourth quarter of 2013, compared to \$33 million or \$0.16 per diluted common the third quarter of 2013. On KFN's fourth quarter earnings call of January 31, 2014, KFN management attributed the increase to "a of gains on investments in a quarter where we saw rallies in both credit and equity markets." Specifically, KFN management of

[KFN] generated \$62 million of net income to common shareholders for the fourth quarter, which implies an annualize equity of over 11% for the quarter. This compares with \$33 million, or 6% return on equity, for the third quarter-over-quarter improvement in earnings was primarily driven by a \$27 million increase in other income, nearly investigated.

115. While the benefits from KFN's increased earnings will now inure to the benefit of KKR rather than KFN shareholders, further insult to injury is the fact that distributions to the former KFN shareholders will actually *diminish* as a result of the Proposed Transcription (2013, KFN stockholders received total distributions of \$0.90 per common share. As previously

shareholders will receive 0.51 common units of KKR for each common share of KFN. Following the merger, KKR's distributions at to be \$1.46 per KKR unit. Thus, following the closing of the Proposed Transaction, KFN shareholders will receive \$0.74 per interest, *or an 18% drop in yield*. And even that diminished return for KFN stockholders is illusory, in that the declaration and pay distributions are subject to the discretion of the board of directors of the general partner of KKR, which does not owe fiduciary dut unitholders, as a

116. Much like analysts widely panned the change in Company strategy, analysts did not believe that KFN received a fair dea For example, Wells Fargo analyst Joel Houck downgraded the Company's stock immediately upon the announcement of Transaction, stating that it failed to represent an adequate premium based on the Company

KKR Opportunistically Timed The Proposed Transaction To Take Advantage Of A Foreca Interest Rates Which Will Drive KFN's Va

117. KKR, knowing that interest rate increases in the near future would dramatically increase the value of KFN, opportunistical Proposed Transaction so that it could acquire KFN at a discount before interest rates increased. KKR understood that an increase in would increase the value of KFN's floating rate debt investments and, in turn, increase the value of KFN. These circumstances promact quickly to force an acquisition of KFN before a forecasted increase in market interest rates in order to capture the increase in KF.

118. In its Form 10-Q filed with the SEC on November 6, 2013, KFN displays a chart showing the increase in its net income increase in its benchmark

Interest rate risk impacts our interest income, interest expense, prepayments, as well as the fair value of our investments, derivatives and liabilities. We generally fund our variable rate investments with variable rate borrowings with similar interfrequencies. Based on our variable rate investments and related variable rate borrowings as of September 30, 2013, we expense in interest rates would impact net income by approximately (amounts in

Change in interest rates	Ann	ual Impact
Increase of 1.0%	\$	(14,929)
Increase of 2.0%	\$	5,370
Increase of 3.0%	\$	25,670
Increase of 4.0%	\$	45,970
Increase of 5.0%	\$	66,269

119. KKR understood all of this and positioned itself to capture the lion's share of any benefits to KFN from rising i

120. KKR itself has publicly acknowledged the likely shift towards higher interest rates signaled by recent Federal Restatements. KKR's Form 10-Q filed with the SEC on November 1, 2013, notes that "[t]he U.S. Federal Reserve has also signatupcoming shift in monetary policy and tapering of quantitative easing [that] could cause interest rates to rise substantially which could be a substantially which

121. KKR decided to take no chances. It spotted the opportunity to use its control over KFN to realize a windfall by buyir public stockholders now, at a time of historically low interest rates, and then await a windfall after the expected increase in intercorresponding boom in l

KFN's Future Ren

122. Although KKR exerted its influence over KFN in order to acquire the Company cheaply, KFN was dripping with a profitability would, absent the Proposed Transaction, have sent the Company's share price skyrocketing. Oil and natural gas revenues doubled as a percentage of KFN's revenues compared to previous year results, as announced in a May 1, 2013 press release d Company's first quarter 2013 results. This trend continued through the second quarter 2013, which showed that oil and gas comprise of the Company's revenues, and by the third quarter of 2013, oil and gas projects constituted 25% of the Company's total revenue poised to realize significantly increased revenues from oil, above these already rapidly growing returns, as a result of key initiatives announced by the

123. On July 31, 2013, EXCO Resources ("EXCO") and the Company entered into a participation agreement (the "Agreement") whereby in exchange for a cash payment of approximately \$131 million, plus the obligation going forward to jointly developments, KFN received the right to 75% of each well's returns. These wells are in the lucrative Eagle Ford Shale range of Texa been widely referred to as "the most profitable field in the world" by natural resources titans, including J. Michael Yeager, CEO of I

124. Beyond merely being found in the world's most profitable oil field, these wells are also located in what is referred to subfield, possibly the highest producing and easiest to drill subsection of the Eagle Ford Shale range. These leases are near natural greatly minimizing the cost and time involved in drilling and completing oil wells. The first well drilled in the Buda area by the company paid back its drilling costs, pumping pure profit back to its investors, in less than four months. A more recent well returned just six weeks. Between the five most active oil companies in this subfield, there have been 13 Buda wells drilled: the first nine we their costs in less than six months and the other four have only been producing for a few weeks but already appear extremel

125. KKR also has a share in the EXCO fields, and it is clear that, by acquiring KFN, it wants to monopolize control over the stream of profitability. The terms of the Participation Agreement allow EXCO to buyout the 75% interest held by KFN in late 2014 at rates. KFN was either likely to take a monumental, one-time jolt to its revenue stream or poised to provide lasting, drama shareholders. Given the Company's impending success with its oil investments and the importance to its revenues that oil and gas also is no surprise that the market reacted negatively to the October announcement by KKR member and KFN CEO Farr of the Company abandonment of this valuable revenue stream, although the true purpose of the announcements was to cause the share price of KFN to allow KKR's acquisition of the Company to occur at a price dramatically below KFN

The Transaction Committee's Financial Advisor, Sandler O'Neill, Was Hopelessly Conflic Fairness Opinion Is Sever

126. The Transaction Committee retained Sandler O'Neill as its financial advisor in connection with the Proposed Transactio and scope of Sandler O'Neill's engagement confirm that the bank's intended role was simply to rubber stamp KKR's advantageou and create the façade of a fair and independent confirmation.

127. The Proxy states that Sandler O'Neill was to receive a fee of \$17 million, purportedly for serving as the financial a Transaction Committee. The bulk of this fee, however, was not contingent upon the performance of advisory duties. Instead, a grow \$15 million of the fee was contingent upon the consummation of the Proposed Transaction. Sandler O'Neill therefore had an unincentive to recommend in favor of the Proposed Transaction, regardless of its econ

Table

128. In the usual corporate sale situation, target boards justify a heavily incentive laden fee to their banker by saying it rewards to obtain the highest price available, from whatever source. But here, there was no reason to award Sandler O'Neill an almost entirel fee, since it was not retained to look for any alternative other than KKR. Rather, the high contingent fee was simply a means to Sandler would bless whatever price KKR was willing to pay, regardless whether that price was fair to KR.

129. Motivated by the prospect of such a hefty payday, Sandler O'Neill delivered a thinly-supported fairness opinion to the Committee, concluding that, as of December 16, 2013, the Exchange Ratio was fair, from a financial point of view, to KF stockholders. Glaring weaknesses in Sandler O'Neill's self-interested fairness opinion, however, only serve to undermine its

130. First, the scope of Sandler O'Neill's retention was unreasonably narrow. The bank was not retained to, and did not, conside potential alternatives to the Proposed Transaction, including whether remaining as a standalone business might provide a higher retroublic shareholders. The Proxy makes no mention of any efforts by the Transaction Committee or Sandler O'Neill to conduct any market check or to actively shop the Company and, as discussed above, the Merger Agreement contains a restrictive "No-Shop barring the Transaction Committee from undertaking any such efforts post-signing. Tellingly, Sandler O'Neill's fairness opinical address the underlying business decision of the Company to engage in the [Proposed Transaction], the relative merits of the Transaction] as compared to any other alternative business strategies that might exist for the Company or the effect of any other to which the Company might exist for the Company or the effect of any other to which the Company might exist for the Company or the effect of any other to which the Company might exist for the Company or the effect of any other to which the Company might exist for the Company or the effect of any other to which the Company might exist for the Company or the effect of any other to which the Company might exist for the Company or the effect of any other to which the Company might exist for the Company migh

131. Second, Sandler O'Neill does not appear to have been provided with KFN management's financial projections, such a 3-year or 5-year earnings before interest, taxes, depreciation and amortization ("EBITDA") or free cash flow projections commonly mergers and acquisitions to estimate the value of the target company. Management projections of future financial performance considered to be critical because management best understands the business and is therefore in the best position to estimate performance. Yet the Proxy does not list management projections among the sources of information that Sandler O'Neill rearnings estimates" for 2013 and 2014, "publicly available mean analyst growth estimates for years thereafter" and "discuss[ions senior management. The Proxy provides no explanation why KFN, a sophisticated investment vehicle controlled by a global private did not prepare basic management financial projections to assist Sandler O'Neill in its valuates.

132. In fact, the Proxy goes on to state that KFN management had prepared financial projections only for the year 2015 available to Sandler O'Neill, who elected not to consider them. *See* Proxy at 78-79. Sandler O'Neill's valuation analysis relied available analyst estimates rather than a complete set of projections created by the group best positioned to forecast KFN's fut performance its own management team. Sandler O'Neill's valuation analysis and fairness opinion therefore did not and could meaningful bases upon which the Transaction Committee could evaluate the fairness of the Exc

133. Third, it appears that Sandler O'Neill was not given access to internal due diligence information from KKR in order to financial condition, as one would expect in any arms-length transaction involving a stock-for-stock deal. Any disinterested final would have insisted on access to confidential internal due diligence from KKR, subject to an appropriate and customary or agreement if necessary, in order to ensure that the value of KKR's stock implied by the Exchange Ratio was adequately supported be look at its financial condition based on all relevant information, and that there were no other material issues at KK

Table

negatively impact KFN's public stockholders post-closing. That Sandler O'Neill failed to do so confirms that its chief concern w justifying the price that KKR had alread

134. Fourth, Sandler O'Neill improperly evaluated the Proposed Transaction by comparing it to cherry-picked example involving "companies in the financial services industry where the buyer and the target had commercial relationships similar to the KFN and KKR" precisely the type of transactions likely to lead to the target company being undervalued. Because the Propose outperformed these "comparative deals," Sandler O'Neill incorrectly determined the

Wall Street Analyst Reports, Including A Report By The KFN Board's Own Financial Advis Analysis Of Comparable Companies, Confirm That The Proposed Transaction Undervalues Th

135. Several Wall Street analysts have noted that KFN is strategically poised for strong financial performance. As recently as O the Transaction Committee's own financial advisor, Sandler O'Neill, increased its share price target for KFN to \$25.00 per share, acc October 15, 2013 article on TickerReport.com. In October 2013, analysts at Credit Suisse reportedly placed a price target of \$21.00 KFN shares. And a December 18, 2013 article on TickerReport.com notes that analysts at Morgan Stanley raised their price target for \$24.00 to \$27.00 per share in a November 14, 2013 research note. The Board, however, approved a sale of KFN to KKR for con only \$12.79 per share nearlyhalf of what its own financial advisor believed the shares to be worth just two me

136. Sandler O'Neill and Credit Suisse were not outliers. Compass Point Research & Trading, LLC noted in an analyst report 2013 that it "believe[s] KFN shares remain undervalued, especially for a business that is capable of producing a high teens ROE dur cycle. In addition, KFN is well positioned for structurally higher interest rates, especially if driven by higher inflation expectate initially the benefit will be muted due to floors on lending rates." Similarly, Wells Fargo Securities issued an analyst report on October noting that KFN was in a "transitional period" and that Wells Fargo Securities "remain[ed] constructive on the long-term probusiness and its ability to generate low-mid teen returns, and [reiterated its] Outperform rating

137. Further, at the time it was announced, the Proposed Transaction's price represented a P/E multiple of 9.3x. By contrast, the ratio for the top 20 asset management companies is 17.6x, nearly double the multiple represented by the Proposed Transaction. This with other asset management companies and the severe discount in the P/E ratio represented by the Proposed Transaction further of the Proposed Transaction substantially under

THE PROPOSED TRANSACTION LEAVES KFN SHAREHOLDERS WITH A MINORITY INT INFERIOR COMMON UNITS IN A CONTROLLED

138. KKR is a limited partnership that is controlled by its managing partner, which is in turn controlled by KKR's founders. filings make clear that "[c]ommon unit holders have only limited voting rights relating to certain matters and, therefore, will have *lability to influence management's decisions* regarding [KKR's] business." (Empl

139. Moreover, KKR's partnership agreement expressly excludes "any duties or liabilities, including any fiduciary duties to the any Limited Partner or any other Person bound by the [Partnership] Agreement." By contrast, KFN's operating agreement prov fiduciary duties of KFN directors are "generally consistent with those of a director of a Delaware of the contract of the contract

Table

140. Thus, after consummation of the Proposed Transaction, current KFN shareholders will be left with a minority interest in limited partnership in the form of unit interests with inferior protection against fiduciary misconduct than KFN's cu

CLASS ACTION ALLEGATIONS

141. Plaintiffs bring this action on behalf of themselves and as a class action on behalf of the public stockholders of KF Defendants and their affiliates (the "Class") pursuant to Court of Chancery Rule 23. Excluded from the Class are Defendants he person, firm, trust, corporation, or other entity related to or affiliated with any of the

142. This action is properly maintainable as a

143. The Class is so numerous that joinder of all members is impracticable. As of October 30, 2013, there were 204,824,159 should be common stock

144. There are questions of law and fact which are common to the Class including, inter alia, the

Whether the Proposed Transaction is entirely fair to KFN's public s

Whether the Individual Defendants have breached their fiduciary duties to Plaintiffs and the other members of connection with the Proposed Tran

Whether Plaintiffs and the Class are entitled

145. Plaintiffs are committed to prosecuting this action and have retained competent counsel experienced in litigation of Plaintiffs' claims are typical of claims of the other members of the Class and Plaintiffs have the same interests as the other members Accordingly, Plaintiffs are adequate representative of the Class and will fairly and adequately protect the interests

146. The prosecution of separate actions by individual Class members would create the risk of inconsistent or varying adjudrespect to individual Class members that would establish incompatible standards of conduct for Defendants, or adjudications with individual Class members that would as a practical matter be dispositive of the interests of the other members not party to the adjustentially impair or impede their ability to protect the substantially impair or impede their ability to protect the substantially impair or impede their ability to protect the substantially impair or impede their ability to protect the substantially impair or impede their ability to protect the substantially impair or impede their ability to protect the substantially impair or impede their ability to protect the substantially impair or impede their ability to protect the substantially impair or impede their ability to protect the substantially impair or impede their ability to protect the substantially impair or impede their ability to protect the substantially impair or impede their ability to protect the substantially impair or impede their ability to protect the substantially impair or impede their ability to protect the substantially impair or impede their ability to protect the substantial transfer to the substantial transfer transfer to the substantial transfer transfe

IV. THE ENTIRE FAIRNESS STANDARD APPLIES TO THE PROPOSED TRAI

147. Where corporate fiduciaries such as KKR in this case stand on both sides of a transaction, Delaware law applies the standard and imposes heightened fiduciary obligations on directors to ensure that the public stockholders are treated fair *Acquisition Co. v. Macmillan, Inc.*, 559 A.2d 1261, 1280 (Del. 1988) ("[D]irectors are required to demonstrate both their utmost gother most scrupulous inherent fairness of transactions in which they possess a financial, business or other personal interest which does upon the corporation or all stockholders

148. This exacting standard involves a two part inquiry into whether a corporate transaction is the result of a fair process and respective. The initial burden of establishing a fair process and fair price rests with the directors defending the

149. Here, the entire fairness standard applies to the Proposed Transaction because KKR controls KFN. Under the Management KKR provides all of KFN's management, investment activities and administrative services and thus exercises complete control business. The terms of the Management Agreement, which renews automatically each year, only permits KFN to terminate it under of circumstances. Unless KKR has breached the

Table

Agreement, KFN can only terminate it by a vote of at least two-thirds of its independent directors or by a majority vote of its common shares *if* KKR's performance is deemed materially detrimental to the Company or the management fees are deemed to (subject to KKR's right to negotiate a mutually- acceptable reduction in fees). Thus, KFN could not terminate the Management Agreement require over six-months' advance notice to KKR and the payment of a prohibitively expensive Termination Fee in the hundreds of dollars. Thus, KKR is able to effectively wield complete control over KFN by virtue of the Management

150. Even if KKR did not control KFN, KFN's Board still breached its duty of loyalty to KFN's public shareholders, and the Transaction does not satisfy entire fair

151. Because the Board did not constitute a special committee that was empowered in any meaningful way to negotiate w protect KFN's interests, and instead allowed the highly conflicted Hazen to dominate negotiations, the Board acted disloyally to be a second or constitute a special committee that was empowered in any meaningful way to negotiate w

COUNT I

AGAINST THE INDIVIDUAL DEFENDANTS FOR BREACH OF FIDUCIARY DUTY

152. Plaintiffs repeat and reallege the foregoing paragraphs as if fully set

153. The Individual Defendants have violated their fiduciary duties of due care and loyalty owed to KFN's stockholders by ag Proposed Transaction, which does not provide a fair or value maximizing price to the Company's public stockholders and which we following an ur

154. As a result of the actions of the Individual Defendants, Plaintiffs and the Class have been and will be damaged in that forced to receive unfair consideration for their shares of KFN common stock under the Merger Agreement's Exchange Ratio follow

155. Plaintiffs and the Class have no adequate re

COUNT II

AGAINST DEFENDANT KKR FOR BREACH OF FIDUCIARY DUTY

156. Plaintiffs repeat and reallege the foregoing paragraphs as if fully set

157. Defendant KKR has breached its fiduciary duty of loyalty in its capacity as a controlling stockholder of KFN. KKR, w certain of its affiliates is a stockholder of KFN, dominates and controls KFN by virtue of the Management Agreement through whic all of KFN's management and operations. In addition, given the oppressive cost to KFN of terminating the Management Agreement, reasonably take any action contrary to KK

158. By causing KFN to enter into the Merger Agreement at an unfair price and following an unfair process, KKR has failed to fiduciary obligations to Plaintiffs a

159. Plaintiffs and the Class have no adequate re

COUNT III

AGAINST DEFENDANTS KKR, KKR FUND HOLDINGS L.P., AND COPAL MERGER SUB LLC FOR AIDING AND ABETTING BREACHES OF FIDUCIARY DUTY

160. Plaintiffs repeat and reallege the foregoing paragraphs as if fully set

Table

161. As alleged in detail herein, the Individual Defendants have breached their fiduciary duties to Plaintiffs and the other me

162. KKR, KKR Fund Holdings L.P. and Copal Merger Sub LLC (the "Aiding and Abetting Defendants") have aided an Individual Defendants in their breaches of fiduciary duty. As parties to the Merger Agreement, the Aiding and Abetting Defendants of the Individual Defendants' breaches of fiduciary duties and in fact actively and knowingly encouraged and participated in said order to obtain substantial financial benefits for themselves, to the detriment of KFN's common s

163. Plaintiffs and the Class have no adequate re

WHEREFORE, Plaintiffs demand judgment in their favor and in favor of the Class and against Defendants

Declaring that this action is properly maintainable as a

Declaring that the Individual Defendants have breached their fiduciary duties to KFN's public s

Declaring that KKR has breached its fiduciary duties to KFN's public s

Declaring that the Aiding and Abetting Defendants aided and abetted the breaches of fiduciary duty by the

Awarding compensatory damages to Plaintiffs a

Awarding Plaintiffs the costs and disbursements of this action, including reasonable attorneys' and expe

G. Granting such other and further relief as this Court deems jus

<u>Table</u>

Dated: Febru

GRANT & EISENHOFER P.A.

OF COUNSEL:

Mark Lebovitch
Jeroen van Kwawegen
Adam Hollander
BERNSTEIN LITOWITZ BERGER &
GROSSMANN LLP
1285 Avenue of the Americas
New York, New York 10019
(212) 554-1400

Co-Lead Counsel for Plaintiffs

Christopher J. Keller Eric J. Belfi Michael W. Stocker LABATON SUCHAROW LLP 140 Broadway New York, New York 10005 (212) 907-0700

Co-Lead Counsel for Plaintiffs

/s/ MARY S. THOMAS

Stuart M. Grant (Del. Bar No. 2526) Mary S. Thomas (Del. Bar No. 5072) Nathan A. Cook (Del. Bar No. 4841) Bernard C. Devieux (Del. Bar No. 5689) Oderah C. Nwaeze (Del. Bar No. 5697) 123 Justison Street Wilmington, Delaware 19801 (302) 622-7000

Co-Lead Counsel for Plaintiffs

LABATON SUCHAROW LLP

Christine S. Azar (Del. Bar No. 4170) Peter C. Wood, Jr. (Del. Bar No. 5249) Ned C. Weinberger (Del. Bar No. 5256) 300 Delaware Avenue, Suite 1225 Wilmington, Delaware 19801 (302) 573-2530

Co-Lead Counsel for Plaintiffs

<u>Table</u>

ROBBINS GELLEI
&
RANDALL J. BARG
A. RICK ATWOOD,
DAVID T. WISSBROECK
EDWARD M. GERGOSI
DAVID A. KNOT
655 West Broadway
San Dieg

Attorneys

Telephone: 6 619/23

[Additional counsel appear on sign

SUPERIOR COURT OF THE STATE OF CALIFORNIA COUNTY OF SAN FRANCISCO

and on Behalf of All Others Similarly Situated, Plaintiffs, Plaintiffs, OLASS ACTION OLAS	IRVING S. AND JUDITH BRAUN, Joint Account, Individually) <u>VIA FAX</u>
vs.	and on Benan of An Others Similarly Situated,) Case No. CGC-13-536281
KKR FINANCIAL HOLDINGS LLC, KKR & CO. L.P., KKR FUND HOLDINGS L.P., COPAL MERGER SUB LLC, CRAIG J. FARR, PAUL M. HAZEN. TRACY LYNN COLLINS, ROBERT L. EDWARDS, VINCENT PAUL FINIGAN, R. GLENN HUBBARD, PH.D. ROSS J. KARI, ELY L. LICHT, DEBORAH H. MCANENY, SCOTT C. NUTTALL, SCOTT RYLES, WILLY STROTHOTTE and DOES 1-25, inclusive,	Plaintiffs,) <u>CLASS ACTION</u>
KKR & CO. L.P., KKR FUND HOLDINGS L.P., COPAL MERGER SUB LLC, CRAIG J. FARR, PAUL M. HAZEN. TRACY LYNN COLLINS, ROBERT L. EDWARDS, VINCENT PAUL FINIGAN, R. GLENN HUBBARD, PH.D. ROSS J. KARI, ELY L. LICHT, DEBORAH H. MCANENY, SCOTT C. NUTTALL, SCOTT RYLES, WILLY STROTHOTTE and DOES 1-25, inclusive,	VS.	
KKR FUND HOLDINGS L.P., COPAL MERGER SUB LLC, CRAIG J. FARR, PAUL M. HAZEN. TRACY LYNN COLLINS, ROBERT L. EDWARDS, VINCENT PAUL FINIGAN, R. GLENN HUBBARD, PH.D. ROSS J. KARI, ELY L. LICHT, DEBORAH H. MCANENY, SCOTT C. NUTTALL, SCOTT RYLES, WILLY STROTHOTTE and DOES 1-25, inclusive,	•	
COPAL MERGER SUB LLC, CRAIG J. FARR, PAUL M. HAZEN. TRACY LYNN COLLINS, ROBERT L. EDWARDS, VINCENT PAUL FINIGAN, R. GLENN HUBBARD, PH.D. ROSS J. KARI, ELY L. LICHT, DEBORAH H. MCANENY, SCOTT C. NUTTALL, SCOTT RYLES, WILLY STROTHOTTE and DOES 1-25, inclusive,)	,	
PAUL M. HAZEN. TRACY LYNN COLLINS, ROBERT L. EDWARDS, VINCENT PAUL FINIGAN, R. GLENN HUBBARD, PH.D. ROSS J. KARI, ELY L. LICHT, DEBORAH H. MCANENY, SCOTT C. NUTTALL, SCOTT RYLES, WILLY STROTHOTTE and DOES 1-25, inclusive,		
TRACY LYNN COLLINS, ROBERT L. EDWARDS, VINCENT PAUL FINIGAN, R. GLENN HUBBARD, PH.D. ROSS J. KARI, ELY L. LICHT, DEBORAH H. MCANENY, SCOTT C. NUTTALL, SCOTT RYLES, WILLY STROTHOTTE and DOES 1-25, inclusive,)	, , , , , , , , , , , , , , , , , , ,)
ROBERT L. EDWARDS, VINCENT PAUL FINIGAN, R. GLENN HUBBARD, PH.D. ROSS J. KARI, ELY L. LICHT, DEBORAH H. MCANENY, SCOTT C. NUTTALL, SCOTT RYLES, WILLY STROTHOTTE and DOES 1-25, inclusive,)	PAUL M. HAZEN.)
VINCENT PAUL FINIGAN, R. GLENN HUBBARD, PH.D. ROSS J. KARI, ELY L. LICHT, DEBORAH H. MCANENY, SCOTT C. NUTTALL, SCOTT RYLES, WILLY STROTHOTTE and DOES 1-25, inclusive,))
R. GLENN HUBBARD, PH.D. ROSS J. KARI, ELY L. LICHT, DEBORAH H. MCANENY, SCOTT C. NUTTALL, SCOTT RYLES, WILLY STROTHOTTE and DOES 1-25, inclusive,)		
ROSS J. KARI, ELY L. LICHT, DEBORAH H. MCANENY, SCOTT C. NUTTALL, SCOTT RYLES, WILLY STROTHOTTE and DOES 1-25, inclusive,)	· · · · · · · · · · · · · · · · · · ·	
ELY L. LICHT, DEBORAH H. MCANENY, SCOTT C. NUTTALL, SCOTT RYLES, WILLY STROTHOTTE and DOES 1-25, inclusive,)		
SCOTT C. NUTTALL, SCOTT RYLES, WILLY STROTHOTTE and DOES 1-25, inclusive,)))
SCOTT RYLES, WILLY STROTHOTTE and DOES 1-25, inclusive,)	DEBORAH H. MCANENY,)
WILLY STROTHOTTE and) DOES 1-25, inclusive,)		
DOES 1-25, inclusive,	<i>'</i>	
)		
Defendants.)	DOES 1-23, Inclusive,	
	Defendants.)
•)

AMENDED COMPLAINT FOR BREACHES OF FIDUCIARY DUTY AND AIDING AND ABETTING

Plaintiffs, by their attorneys, alleg

SUMMARY OF THE ACTION

- 1. This is a stockholder class action brought on behalf of the holders of KKR Financial Holdings LLC ("KFN" or the common stock against KFN, the members of KFN's Board of Directors (the "Board"), KKR & Co. L.P., a Delaware limited ("KKR LP"), KKR Fund Holdings L.P., an exempted limited partnership formed under the laws of the Cayman islands ("Fund Holdings Copal Merger Sub LLC, a Delaware limited liability company ("Merger Sub," and with KKR LP and Fund Holdings, "KKR") for fiduciary duty and/or other violations of state law arising out of defendants' efforts to complete the sale of the Company to KKR punfair process and for an unfair price (the "Proposed Acquisition"). In pursuing the unlawful plan to sell KFN to KKR, each of the violated applicable law by directly breaching and/or aiding the other defendants' breaches of their fiduciary duties of loyalty, due independence, good faith and
 - 2. KFN is a specialty finance company with expertise in a range of asset classes. KFN is managed and advised by KR Advisors LLC ("Manager"), a wholly-owned subsidiary of KKR ("Manager" and "KKR" often collectively referred to herein
- 3. On December 16, 2013. KKR and KFN announced that they had entered into an Agreement and Plan of Merger Agreement") pursuant to which KKR will acquire KFN for a mere 0.51 common units of KKR LP for each common share of KFN. I closing price of the KKR LP common units as of December 16, 2013, the exchange ratio equates to a value of \$12.79 per common share of KFN with the Merger Agreement provides that, upon the terms and subject to the conditions set forth in the Merger Agreement. KFN with the Werger Agreement provides that the wholly-owned subsidi
- 4. KKR created KFN and continues to control KFN by virtue of a management agreement between Manager and KFN (the "I Agreement"). KFN has no facilities, no offices, no employees, and no independent management. In its most recent annual report ("I admitted that "[w]e have no separate facilities and are completely reliant on our Manager, which has significant discretimplementation and execution of our business and investment strategies and our risk management practices." The Management Agguarantees that the KFN Board has no real supervision or oversight over KKR, and indeed. KKR is solely responsible for monitor performance then providing periodic reports regarding that own performance to the KFN Board. In addition, in light of the egregious fee provisions in the Management Agreement, KFN is effectively unable to terminate KKR as
- 5. While KFN's valuable assets would ordinarily attract the interest of other buyers in a sale process, the Proposed Acquis Management Agreement combine to erect an insurmountable barrier to any and all other potential purchasers. The Management contains a one-sided termination provision (the "Management Termination Fee") that requires KFN to pay an excessive penalty if it terminate while essentially allowing KKR to terminate the agreement without cost. Given the control that KKR yields over KFN, it any other buyer would seek to dissolve the Management Agreement in order to effectuate its control over the entity it had just purcanother buyer attempts to do so, it must provide 180 days' notice and pay an over \$252.6 million termination fee to KKR. This figure more than KFN's current cash and cash equivalents and represents an effective 10.3% termination fee on the Proposed Acquisition common equity value of
 - 6. Despite the significant impact of a 10.3% termination fee to the Proposed Merger, the Board did not attempt to en provision was negotiated by unconflicted fiduciaries. Rather, Craig J. Farr ("Farr"), CEO of KFN and perhaps the most conflicted member, engaged in a single discussion with his colleagues at KKR, wherein KKR informed Farr that KKR "was unwilling."

Table

or eliminate the termination fee if there were a change of control of KFN." The KFN Board never again discussed the issue an attempted to unwind the damage that Farr had a

- 7. Farr was grossly conflicted and should not have been KFN's sole representative in this crucial conversation. In addition to as CEO of KFN, Farr is also the Head of KKR Asset Management, is a member of KKR's Risk Committee, and oversees KKR's Cap and origination business. Farr is essentially responsible for Manager's financial performance and it was not in Farr's interests to no material change to an agreement by a company for which he is
 - 8. After Farr and his colleagues at KKR agreed not to amend the Management Termination Fee in the event of a change of Board's "Transaction Committee" conducted a short process marked by repeated concessions and a minimal increase in merger of KKR dictated the terms of the Proposed Acquisition and in light of KKR's control over KFN's operations, the KFN Board yielded leverage in the negotiations. Throughout the negotiations, KKR already knew what it would ultimately pay for KFN, KKR wield power over KFN through the Management Agreement, and the KFN Board's limited process did nothing to change KKR's
- 9. As a result of that flawed process, plaintiff stockholders of KFN are receiving units with far less protections than their common shares and results in an even lower value for the consideration received. For example, defendants admit in the S-4 that K shareholders will "not have a voting interest in most matters after the merger and will exercise materially less or no influence over n and that they will have "no control over KKR's management." This is because KKR common unitholders are not entitled to elect partner and are not entitled to elect the directors of KKR's general partner. As additionally described herein, the Proposed consideration significantly undervalues both KFN's inherent value and its value to KKR and defendants improperly manipulated the Proposed Acquisition's an
- 10. Compounding these problems, on January 15, 2014, defendants filed with the SEC a materially false and misleadi Registration Statement Under the Securities Act of 1933 that omitted several crucial facts regarding the Proposed Acquisition as de-
- 11. In sum, acting on their own self-interest, defendants utilized a defective sales process that was not designed to maximize value or protect the interests of KFN shareholders, but rather was designed to divert the Company's valuable assets to KKR defendants has breached their fiduciary duties of loyalty, due care, independence, candor, good faith and fair dealing, and/or habetted such breaches. Rather than acting in the best interests of the Company's shareholders, defendants spent substantial effort structural terms of the Proposed Acquisition to aggrandize their own personal interests and to meet the specific needs of KKR. whice eliminate the majority of the equity interest of KFN's shareholders. Plaintiffs seek to enjoin the Proposed

JURISDICTION AND VENUE

- 12. This Court has jurisdiction over the cause of action asserted herein pursuant to the California Constitution, art. VI, §10, case is a cause not given by statute to other
- 13. This Court has jurisdiction over KFN because KFN is a citizen of California and Delaware as it is incorporated in Delaware principal place of business at 555 California Street, 50th Floor, San Francisco, California. This action is no
 - 14. Venue is proper in this Court because the conduct at issue took place and had an effect in

PARTIES

- 15. Plaintiffs Irving S. and Judith Braun, Joint Account, are, and at all times relevant hereto were, sharehold
- 16. Defendant KFN is a Delaware corporation, headquartered in San Francisco, California. KFN is sued as an aider and al
 - 17. Defendant KKR LP is a Delaware limited partnership. KKR LP is sued as an aider and all
- 18. Defendant Fund Holdings is an exempted limited partnership formed under the laws of the Cayman islands. Fund Holdings an aider and all
 - 19. Defendant Merger Sub is a Delaware limited liability company. Merger Sub is sued as an aider and al
 - 20. Defendant Farr is and at all relevant times has been the President and CEO of KFN and a member of the Board. Farr a Head of KKR Asset Management LLC and oversees KKR's Capital Markets and Origination business. Farr is a member of the Board.
- 21. Defendant Paul M. Hazen ("Hazen") is and at all relevant times has been Chairman and a member of the Board. Hazen al board KKR subsidiary Accel-KKR and serves as a senior advisor to KKR. Additionally, Hazen serves on the board of KSL I company founded with the support of KKR Co-Founder Henry Kravis. Hazen served on the Phelps Dodge Board shortly after Phelps its stake in Accuride to the Accuride management and a KKR affiliate. Hazen has served on the Airtouch and Vodaphone Boards with Arun Sarin, now a senior KKR advisor and CEO at Accel-KKR Telecom. Hazen served on the Safeway Inc. Board after KK the company in 1986. Hazen remained on the Safeway board until 2012 long after KKR sold the company in 1999. Hazen sits on the Willis Group Holdings Plc, a company chaired Corporation with KKR and KFN director S
 - 22. Defendant Tracy Lynn Collins ("Collins") is and at all relevant times has been a member of the KFN Board. Collins jo August 2006 and serves on KFN's Audit and Compensation
 - 23. Defendant Robert L. Edwards ("Edwards") is and at all relevant times has been a member of the KFN Board. Edwards director since November 2011 and serves on the audit committee. Edwards serves on the Flextronics Inc. board, a software composition KKR purchased a controlling stake in April 2006. Edwards served as Safeway's CFO from March 2004 until stepping into the F CEO position in May 2013, shortly after Hazen stepped down from Safeway's Chairman of Nominating and Governance Co Member of Compensation Commi
 - 24. Defendant Vincent Paul Finigan ("Finigan") is and at all relevant times has been a member of the KFN Board since 2006 a serves on KFN's Compensation, nominating & Corporate Governance, and Affiliated Transactions
- 25. Defendant R. Glenn Hubbard Ph.D ("Hubbard") is and at all relevant times has been a member of the Board since Octol serves on KFN's Audit and Compensation Committees. Hubbard is the Dean of Columbia Business School, to which KKR Co-Fo Kravis donated \$100 million in 2010, the largest donation in the school's history. Hubbard served on the Board of Capmark Financia from 2006 to 2008 after KKR purchased a significant stake in Capmark in 2006. Capmark went bankrupt in 2009 when KKR a Sachs & Co. ("Goldman Sachs") loaded the company with debt in a leveraged buyout. KKR maintains a majority stake

Table

which emerged from Bankruptcy in 2011. Hubbard also serves with KKR independent director Patricia Russo at the Economic

- 26. Defendant Ross J. Kari ("Kari") is and at all relevant times has been a member of the Board. Kari joined the KFN boa 2004 and serves on the Audit Committee. Kari spent a large portion of his career at Wells Fargo from 1983 to 2001, during which from senior financial analyst to executive vice president and chief financial officer. Kari left Wells Fargo in 2001, around the same left his position as Chairman and CEO of Wells Fargo to join KKR affiliate
- 27. Defendant Ely L. Licht ("Licht") is and at all relevant times has been a member of the Board. Licht formerly served as KKR. Licht joined Capmark Financial Group Inc. in 2007. From March 2007 to February 2009 he was Executive Vice President at of Credit Policy for Capmark Financial, and from March 2009 to August 2009 he served as a consultant to Capmark Financial. Lie employed at Wells Fargo for 18 years, including as Chief Credit Officer from November 1998 to December 2001, and Executive Vice for Credit Administration from February 1990 to November 1998. Licht left Wells Fargo in 2001, around the same time Hazen left as Chairman and CEO of Wells Fargo to join KKR affiliate
 - 28. Defendant Deborah H. McAneny ("McAneny") is and at all relevant times has been a member of the Board. McAne. Board in April 2005 and serves on the Compensation. Nominating & Governance, and Affiliated Transactions
- 29. Defendant Scott C. Nuttall ("Nuttall") is and at all relevant times has been a member of the Board. Nuttall joined KKR in head of KKR's Global Capital and Asset Management Group. He is also actively involved in other companies and funds affiliated we has played a significant role in KKR's private equity investments in Alea Group Holdings, Amphenol, Bristol West Holding Financial, First Data Corporation, KinderCare Learning Centers, Legg Mason, Masonite International, Walter Industries, and Venter and Corporation of the Board. Nuttall joined KKR in head of KKR's Global Capital and Asset Management Group. He is also actively involved in other companies and funds affiliated we have played a significant role in KKR's private equity investments in Alea Group Holdings, Amphenol, Bristol West Holding Financial, First Data Corporation, KinderCare Learning Centers, Legg Mason, Masonite International, Walter Industries, and Venter Industries Industr
 - 30. Defendant Scott Ryles ("Ryles") is and at all relevant times has been a member of the Board. Ryles joined the KFN board i 2008 and serves on KFN's Audit and Affilitated Transactions
- 31. Defendant Willy Strothotte ("Strothotte") is and at all relevant times has been a member of the Board. Strothotte joined the in November 2008 and serves on the Compensation and Affiliated Transaction Committees. Strothotte was appointed Chief Executive Glencore in 1993 and held the combined positions of Chairman and Chief Executive Officer from 1994 until 2001, when the roles and Chief Executive Officer were split. Strothotte has been Chairman of Xstrata AG since 1994, and Chairman of Xstrata plc sin 2002. Strothotte sat on the Glencore Xstata International Board with KFN director Hazen and former KKR Partner Robert MacDeadviser John Mack now serves on Glencore Xstata International Board with KFN director Hazen and former KKR Partner Robert MacDeadviser John Mack now serves on Glencore Xstata International Board with KFN director Hazen and former KKR Partner Robert MacDeadviser John Mack now serves on Glencore Xstata International Board with KFN director Hazen and former KKR Partner Robert MacDeadviser John Mack now serves on Glencore Xstata International Board with KFN director Hazen and former KKR Partner Robert MacDeadviser John Mack now serves on Glencore Xstata International Board with KFN director Hazen and former KKR Partner Robert MacDeadviser John Mack now serves on Glencore Xstata International Board with KFN director Hazen and former KKR Partner Robert MacDeadviser John Mack now serves on Glencore Xstata International Board with KFN director Hazen and former KKR Partner Robert MacDeadviser John Mack now serves on Glencore Xstata International Board with KFN director Hazen and former KKR Partner Robert MacDeadviser John Mack now serves on Glencore Xstata International Board with KFN director Hazen and Grant Machael Robert MacAdeadviser John Mack now serves on Glencore Xstata International Board with KFN director Hazen Robert MacAdeadviser John Mack Now Serves MacAdeadviser Machael Robert MacAdeadviser MacAdeadviser MacAdeadviser MacAdeadviser MacAdeadviser MacAdeadviser MacAdeadviser MacAdeadviser
 - 32. The defendants named above in ¶20-31 are sometimes collectively referred to herein as the "Individual"
 - 33. The true names and capacities of defendants sued herein under California Code of Civil Procedure §474 as Does I inclusive, are presently not known to plaintiffs, who therefore sues these defendants by such fictitious names. Plaintiffs will seek to Complaint and include these Doe defendants' true names and capacities when they are ascertained. Each of the fictitiously named or responsible in some manner for the conduct alleged herein and for the injuries suffered in the conduct alleged herein and for the injuries suffered in the conduct alleged herein and for the injuries suffered in the conduct alleged herein and for the injuries suffered in the conduct alleged herein and for the injuries suffered in the conduct alleged herein and for the injuries suffered in the conduct alleged herein and for the injuries suffered in the conduct alleged herein and for the injuries suffered in the conduct alleged herein and for the injuries suffered in the conduct alleged herein and for the injuries suffered in the conduct alleged herein and for the injuries suffered in the conduct alleged herein and for the injuries suffered in the conduct alleged herein and for the injuries suffered in the conduct alleged herein and for the injuries suffered in the conduct alleged herein and for the injuries suffered in the conduct alleged herein and the conduct alleged herein and for the injuries suffered in the conduct alleged herein and for the conduct

CLASS ACTION ALLEGATIONS

- 34. Plaintiffs bring this action individually and as a class action pursuant to California Code of Civil Procedure §382 on holders of KFN stock who are being and will be harmed by defendants' actions described below (the "Class"). Excluded from defendants herein and any person, firm, trust, corporation, or other entity related to or affiliated with any
 - 35. This action is properly maintainable as a
 - 36. The Class is so numerous that joinder of all members is impracticable. According to the Merger Agreement. K
 204.8 million shares issued and outstanding as of Decembers.
 - 37. There are questions of law and fact which are common to the Class and which predominate over questions affecting ar Class member. The common questions include, *inter alia*, the
 - (a) whether the Individual Defendants have breached their fiduciary duties of undivided loyalty, independence, or consecution with the Proposed
 - (b) whether defendants are engaging in self-dealing in connection with the Proposed
 - (c) whether the Individual Defendants have breached their fiduciary duty to secure and obtain the best value reasonal circumstances for the benefit of plaintiffs and the other members of the Class in connection with the Proposed
 - (d) whether defendants are unjustly enriching themselves and other insiders or affiliates of KFN
 - (e) whether the Individual Defendants have breached any of their other fiduciary duties to plaintiffs and the other me Class in connection with the Proposed Acquisition. including the duties of good faith, diligence, candor and
 - (f) whether the defendants. in bad faith and for improper motives, have impeded or erected barriers to discourage for the Company
 - (g) whether the Proposed Acquisition compensation payable to plaintiffs and the Class is unfair and ina
 - (h) whether plaintiffs and the other members of the Class would be irreparably harmed were the transactions of herein of
 - 38. Plaintiffs' claims are typical of the claims of the other members of the Class and plaintiffs do not have any interests a
- 39. Plaintiffs are adequate representatives of the Class, have retained competent counsel experienced in litigation of this nat fairly and adequately protect the interests
- 40. The prosecution of separate actions by individual members of the Class would create a risk of inconsistent or varying a with respect to individual members of the Class which would establish incompatible standards of conduct for the party opposi
 - 41. Plaintiffs anticipate that there will be no difficulty in the management of this litigation. A class action is superior to ot methods for the fair and efficient adjudication of this

42. Defendants have acted on grounds generally applicable to the Class with respect to the matters complained of herein, the appropriate the relief sought herein with respect to the Class

THE PROPOSED ACQUISITION

43. On December 16, 2013. KKR and KFN announced that they had entered into the Merger Agreement, pursuant to whi acquire KFN for a mere 0.51 common units of KKR LP for each common share of KFN. Based on the closing price of the KKR units as of December 16, 2013, the exchange ratio equates to a value of \$12.79 per common share of KFN. The Merger Agreement pupon the terms and subject to the conditions set forth in the Merger Agreement, KFN will become a wholly-owned subsidiary of Fundamental Processing States and States are supported by the conditions of the Merger Agreement, KFN will become a wholly-owned subsidiary of Fundamental Processing States are supported by the conditions of the Merger Agreement, KFN will become a wholly-owned subsidiary of Fundamental Processing States are supported by the Condition of the Merger Agreement, KFN will become a wholly-owned subsidiary of Fundamental Processing States are supported by the Condition of the Merger Agreement, KFN will become a wholly-owned subsidiary of Fundamental Processing States are supported by the Condition of the Merger Agreement, KFN will become a wholly-owned subsidiary of Fundamental Processing States are supported by the Condition of the Merger Agreement of the Merger Agre

KKR Created KFN and Continue KFN Through the Management

- 44. KKR previously owned KFN and spun it off nine years ago. KFN is managed by a wholly-owned subsidiary of KKR. above, several members of the management of KKR and the Company have significant longstanding ties to one another. Since July 2 CEO of KFN has been defendant Farr, who is also a member of KKR, and the head of the KKR affiliate that wholly owns the M members of KKR, including Farr, sit of
- 45. Under the management agreement between KKR and KFN. KKR manages the business of KFN. KFN admitted in its most that "[w]e do not employ personnel and therefore rely on the resources and personnel of our [KKR] to conduct our operations." KF that "[w]e do not have any employees. We are managed by KKR Financial Advisors LLC, our Manager, pursuant to the Agreement. Our Manager is a wholly-owned subsidiary of KAM and all of our executive officers are members or employees of KI more of its affiliates." KEN described this relationship in the 10-1

We have no separate facilities and are completely reliant on our Manager, which has significant discret implementation and execution of our business and investment strategies and our risk management practices. We are all the risk that our Manager will terminate the Management Agreement and that no suitable replacement will be found. We our success depends to a significant extent upon the experience of our Manager's executive officers, whose continued s guaranteed.... We depend on the diligence, skill and network of business contacts of the senior management and professionals of our Manager.... Our future success will depend on the continued service of the senior management investment professionals of our

46. The KFN Board provides only limited oversight of KKR in its r

Our board of directors has approved very broad investment policies for our Manager and does not approve individual decisions made by our Manager except in limited circumstances. Our Manager is authorized to follow very broad investment (our "Investment Policies") and, in connection with the conversion transaction, these Investment Policies were revised to greater latitude to our Manager with respect to certain matters relating to transactions with our affiliates. Our directors review and approve our Investment Policies. Our board of directors generally does not approve any individual investment approving a limited set of transactions with affiliates that require the prior approval of the Affiliated Transactions Comboard of directors. Furthermore, transactions entered into by our Manager may be difficult or impossible to terminate or a Manager has significant latitude within the broad parameters of the Investment Policies in determining the types of decide are proper investment.

47. The Management Agreement guarantees that KFN has no real supervision or oversight over KKR. The Management between Manager/KKR and KFN provides that that KKR shall manage the "assets and the day-to-day operations of the Compan supervision of KFN's Board, but also that KKR will "counsel[] the Company in connection with policy decisions to be made by Directors." Under §2(b)(xviii) of the Management Agreement, KKR is solely responsible for monitoring its own performance as periodic reports to KFN's Board regarding the same. KKR is thus responsible for counseling KFN's Board. monitoring its own performance providing reports to KFN's Board regarding how to supervision of the Management Agreement, KKR is thus responsible for counseling KFN's Board regarding how to supervision of KFN's Board regarding how

Management Tern

- 48. While KFN's valuable assets would ordinarily attract the interest of other buyers in a sale process, the Proposed Acc Management Agreement combine to erect an insurmountable barrier to any and all other potential purchasers. Given the control that over KFN, it is likely that any other buyer would seek to dissolve the Management Agreement in order to effectuate its actual contentity it had just purchased. But the Management Termination Fee provides for a windfall payment to KKR in the event KFI terminate. In effect, the Merger Agreement and Management Agreement combine to create a \$252.6 million termination fee competing acquired.
- 49. The Management Termination Fee provisions operate as follows. The Management Agreement is automatically renewed for term, indefinitely, on December 31 of each year, If KFN elects not to renew the Management Agreement with KKR. KFN must purinot less than 180 days prior to the expiration of the then existing term." KKR is then provided an opportunity to re-negotiate its parand if a revised compensation agreement is not reached, the Management Agreement will terminate. On the date of the terminate required to pay the Management Termination Fee to KKR. The Management Termination Fee is "equal to the amount of four time the average annual Base Management Fee and the average annual Incentive Compensation earned by the Manager during the typeriods immediately preceding the date of such termination, calculated as of the end of the most recently completed fiscal quarter date of termination." In contrast, KKR can terminate the Management Agreement at any point without payment or pen
- 50. Calculation of the Management Termination Fee requires ascertainment of the relevant Base Management Fees a Compensation Fees paid by KFN to KKR. The monthly Base Management Fee paid by KFN to KKR is equal to 1/12 of KFN's equit in the Management Agreement, multiplied by 1.75%. KFN paid the following Base Management Fees to KKR over the past

2012	\$ 28.2 million
2011	\$ 26.3 million
2010	\$ 19.1 million

51. The monthly Incentive Compensation Fees paid by KFN to KKR in an amount equal to the product of (i) 25% of the dollar which (a) KFN's Net Income, before incentive compensation, per weighted average share of KFN's common shares for such qual (b) an amount equal to (A) the weighted average of the price per share of the common stock of KKR Financial Corp. in its August placement and the prices per share of the common stock of KKR Financial Corp. in its initial public offering and any subsequent KKR Financial Holdings LLC multiplied by (B) the greater of (1) 2.00% and (2) 0.50% plus one-fourth of the Ten Year Treasury I quarter, multiplied by (ii) the weighted average number of

Table

shares outstanding in such quarter. KFN paid the following Incentive Compensation Fees to KKR over the pas

2012	\$ 37.6 million
2011	\$ 34.2 million
2010	\$ 29.1 million

- 52. The Base Management Fees and Incentive Compensation Fees for 2013 are currently unknown, but based on currently average financial performance metrics, it is unlikely the fees will be less than 2012 or 2011. Thus, "four times the sum of the average Management Fee and the average annual Incentive Compensation Fee" earned during 2012 and 2011 is \$252.6 million.(1) Given to claimed the Proposed Acquisition involves a total common equity value of \$2.6 billion, this figure amounts to an effective 10.3%
- 53. KFN's business would be crippled if it were forced to pay KKR the Management Termination Fee. KFN held \$221.9 m and cash equivalents as of September 30, 2012 and would have certain difficulty gathering over \$252.6 million in cash. KFN's lates that "[1]iquidity is essential to our business." Thus, if KKR were forced to pay the Management Termination Fee to KKR. KFN wou to meet other obligations and, according to the 10-K, KFN "would likely need to liquidate unencumbered assets, such as our invaling portfolios, to meet maturing liabilities. We may be unable to sell some of our assets, or we may have to sell assets at a d market value, either of which could adversely affect our results of operations and may have a negative impact on the market common shares and any other securities we
- 54. KKR effectively acts as a CEO (and indeed, an entire management team) holding a golden parachute worth over 10% value of KFN. As a result, the Board cannot simply fire KKR defendants have ensured that KKR will control the operations of KFN Board can do not
 - 55. In full, the "Termination" term of the Management Agreement state
 - (a) Until this Agreement is terminated in accordance with its terms, this Agreement shall be in effect until Decem (the "Initial Term") and shall be automatically renewed for a one-year term each anniversary date thereafter (a "Renewal T at least two-thirds of the Independent Directors or the holders of a majority of the outstanding Shares agree that (i) the unsatisfactory performance by the Manager that is materially detrimental to the Company or (ii) the compensation p Manager hereunder is unfair; provided that the Company shall not have the right to terminate this Agreement under clause the Manager agrees to continue to provide the services under this Agreement at a fee that at least two-thirds of the Directors determines to be fair pursuant to the procedure set forth below. If the Company elects not to renew this Agree expiration of the Initial Term or any such one-year extension term as set forth above, the Company shall deliver to the M written notice (the "Termination Notice") of the Company's intention not to renew this Agreement based upon the term this Section 13(a) not less than 180 days prior to the expiration of the then existing term. If the Company so elects not a Agreement, the Company shall designate the date (the "Effective Termination Date"), not less than 180 days from the notice, on which the Manager shall cease to provide services under this Agreement and this Agreement shall terminate of provided, however, that in the event that such Termination Notice is given in connection with a determination that the companyable to the Manager is unfair, the Manager shall have the right to renegotiate such companyable to the Manager is unfair, the Manager shall have the right to renegotiate such companyable to the Manager is unfair, the Manager shall have the right to renegotiate such companyable to the Manager is unfair, the Manager shall have the right to renegotiate such companyable to the Manager is unfair, the Manager shall have the right to renegotiate such companyable to the Manager is unfair,

The average Base Management Fees and Incentive Compensation Fees paid to KKR in 2012 and 2011 were \$32.9 \$30.25 million. The sum of these figures, \$63.15 million, multiplied by four, is \$2.9

Table

delivering to the Company, no fewer than forty-five (45) days prior to the prospective Effective Termination Date, writter such notice, a "Notice of Proposal to Negotiate") of its intention to renegotiate its compensation under this Agreement. The Company (represented by the Independent Directors) and the Manager shall endeavor to negotiate in good fait compensation payable to the Manager under this Agreement. Provided that the Manager and at least two-thirds of the Directors agree to the terms of the revised compensation to be payable to the Manager within 45 days following the revised of Proposal to Negotiate, the Termination Notice shall be deemed of no force and effect and this Agreement shall full force and effect on the terms stated in this Agreement, except that the compensation payable to the Manager hereunder revised compensation then agreed upon by the parties to this Agreement. The Company and the Manager agree to execute an amendment to this Agreement setting forth such revised compensation promptly upon reaching an agreement regarding event that the Company and the Manager are unable to agree to the terms of the revised compensation to be payable to during such 45 day period, this Agreement shall terminate, such termination to be effective on the date which is the late (10) days following the end of such 45 day period and (B) the Effective Termination Date originally set forth in the

- (b) In the event that this Agreement is terminated in accordance with the provisions of Section 13(a) of this Ag Company shall pay to the Manager, on the date on which such termination is effective, a termination fee (the "Termination to the amount of four times the sum of the average annual Base Management Fee and the average annual Incentive C earned by the Manager during the two 12- month periods immediately preceding the date of such termination, calculated of the most recently completed fiscal quarter prior to the date of termination. The obligation of the Company to pay the Fee shall survive the termination of this
- (c) No later than 180 days prior to the anniversary date of this Agreement of any year during the Initial Term or Re the Manager may deliver written notice to the Company informing it of the Manager's intention to decline to renew this whereupon this Agreement shall not be renewed and extended and this Agreement shall terminate effective on the annive this Agreement next following the delivery of
- (d) If this Agreement is terminated pursuant to this Section 13, such termination shall be without any further obligation of either party to the other, except as provided in Sections 6, 9, 10, 13(b) and 16 of this Agreement. In addition of this Agreement shall survive termination of this

The Board Did Not Properly Consider or Negotiate the Management Term

56. Given the drastic impact of a 10.3% termination fee on the willingness of another bidder to come forward, one would be Board's "Transaction Committee" would have diligently considered and negotiated that provision's effect on the Proposed Acquinconflicted process. They did not. The S-4 identifies just a single discussion of the issue, between the most conflicted Board memb before the Transaction Committee had even be

Additionally, prior to the formation of the transaction committee, Mr. Hazen discussed with Craig Fair. CEO of KFN, the prodifying or eliminating the termination fee in the management agreement if there were a change of control of KF contacted a representative of KKR and was informed that KKR was unwilling to modify or eliminate the termination were a change of control of KFN.

- 57. With that, negotiations regarding the Management Termination Fee were over. When the Transaction Committee was formed, it never again discussed the issue and never once attempted to unwind the damage that Farr had a
- 58. Farr, as described herein, is grossly conflicted and should not have been KFN's sole representative in this crucial con addition to his position as CEO of KFN. Farr is also the Head of KKR Asset Management, is a member of KKR's Risk Committee, a KKR's Capital Markets and origination business, which has accounted for over 30% of KKR's fee-related earnings in the parameter (KKR Financial Advisors LLC) is a wholly-owned subsidiary of KKR Asset Management LLC (of which Farr is the Head) reports to Kravis and Roberts, Farr is essentially responsible for Manager's financial performance. It was obviously not in Farr' negotiate hard for a material change to an agreement by a company for which he is responsible. This sort of divided loyalty, and stan sides of a transaction, is precisely the sort of conflict that the Board should have carefully guarded against. Defendants, however ignored these protections in designing the Proposed
 - 59. Consequently, as the result of a single conversation between the most conflicted individual, the additional 10.3% effective fee remains in place and will block any other poten

The Board Conducted a Flawed, Conflic

- 60. After Farr and his colleagues at KKR agreed not to amend the Management Termination Fee in the event of a change of Board's "Transaction Committee" conducted a short process marked by repeated concessions and a minimal increase in merger consist October 31, 2013, KFN formed the Transaction Committee, consisting of Collins, Edwards, Finigan, Kari, McAneny, and Ryles. At time, KKR's offer existed at .46 KKR common units per KFN common share. By December 13, when the KFN Board voted in Proposed Acquisition, the ratio had increased just .51 KKR common units per KFN common share, a paltry increase particularly who obtained no other concessions.
- 61. KKR dictated the terms of the Proposed Acquisition and in light of KKR's stranglehold over KFN's operations, the KFN B little actual leverage in the negotiations. After KKR initially proposed an exchange ratio for common units, the KFN Transaction determined that its "preference in any transaction was for merger consideration consisting entirely of cash." The Transaction informed KKR of that preference on November 13, 2013 and the very same day, KKR responded that it would not consider a transaction Committee did not attempt to re-open discussions regarding the Management Termination Fee. Nor did the Transaction attempt to seek an all cash transaction from other buyers, or at least use that attempt as leverage against KKR. Nor did the Committee seek to continue the negotiation regarding cash consideration. The Transaction Committee capitulated and never again re consideration from KKR, or any
- 62. KKR and the Transaction Committee's portrayal of back-and-forth negotiation in the S-4 is a charade. According to November 19, 2013 the Transaction Committee informed KKR it would cancel a due diligence session if KKR did not increase exchange ratio up from .46. The same day, KKR continued the kabuki dance and increased the proposed ratio to .48, so the "continued to be place and the negotiations continued. Cancellation of a "due diligence session" would have done nothing to impact KK proceed KKR already controls every aspects of KFN's day-to-day operations and, under the Management Agreement, it is KKR the KFN Board about KFN's operations, not the other way around. KKR would have already obtained, and indeed, generated, and diligence" of KFN. Throughout the negotiations, KKR already knew what it would ultimately pay for KFN, KKR wielded immense KFN through the Management Agreement, and the Transaction Committee's toothless negotiations did nothing to change KKR's

If the Proposed Acquisition is Consummated, Plaintiffs Will Be Deprived of Voting Rights and Additional Shareholder Proposed Acquisition is Consummated, Plaintiffs Will Be Deprived of Voting Rights and Additional Shareholder Proposed Acquisition is Consummated, Plaintiffs Will Be Deprived of Voting Rights and Additional Shareholder Proposed Acquisition is Consummated, Plaintiffs Will Be Deprived of Voting Rights and Additional Shareholder Proposed Acquisition is Consummated, Plaintiffs Will Be Deprived of Voting Rights and Additional Shareholder Proposed Acquisition is Consummated, Plaintiffs Will Be Deprived of Voting Rights and Additional Shareholder Proposed Acquisition is Consummated, Plaintiffs Will Be Deprived of Voting Rights and Additional Shareholder Proposed Acquisition is Consummated, Plaintiffs Will Be Deprived of Voting Rights and Additional Shareholder Proposed Acquisition is Consummated, Plaintiffs Will Be Deprived of Voting Rights and Additional Shareholder Proposed Acquisition is Consummated and Proposed Acquisition is Consummated Acquisition is Consummated Acquisition is Consummated Acquisit

- 63. Plaintiff stockholders of KFN will receive substantially lesser protections and rights in the KKR units pledged as considerample, defendants admit in the S-4 that KFN common shareholders will "not have a voting interest in most matters after the metexercise materially less or no influence over management" and that they will have "no control over KKR's management." This is be common unitholders are not entitled to elect the general partner and are not entitled to elect the directors of KKR's general partner.
- 64. Additionally, because KKR is a limited partnership, it need not comply with various corporate governance requirements of including that it have nominating, corporate governance, and compensation committees composed entirely of independent directors. not required to hold annual unitholder meetings. Defendants thus admit that "an investor in KKR common units does not he protections afforded to equity holders of entities that are subject to all of the corporate governance requirements of the NYS.

KKR's principals, generally have sufficient voting power to determine the outcome of those few matters that may be sufvote of the holders of KKR's common units, including a merger or consolidation of KKR's business, a sale of all or substated KKR's assets and amendments to KKR's amended and restated limited partnership agreement, which is referred to statement/prospectus as KKR's partnership agreement, that may be material to holders of KKR common units. In add partnership agreement contains provisions that enable KKR to take actions that would materially and adversely affect a KKR common units or a particular class of holders of common units upon the majority vote of all outstanding voting unit following the merger, nearly a majority of KKR's voting units will be controlled by KKR Holdings, KKR Holdings effect the ability to take actions that could materially and adversely affect the holders of KKR common units either as a wear to the sufficient to the sufficient to the sufficient voting units will be controlled by KKR common units either as a wear to the sufficient voting units will be controlled by KKR common units either as a wear to the sufficient voting units will be controlled by KKR common units either as a wear to the sufficient voting units will be controlled by KKR common units either as a wear to the sufficient voting units will be controlled by KKR common units either as a wear to the sufficient voting units will be controlled by KKR common units either as a wear to the sufficient voting units will be controlled by KKR to take actions that could materially and adversely affect the holders of the sufficient voting units will be controlled by KKR to take actions that could materially and adversely affect the holders of the sufficient voting units will be controlled by KKR to take actions that could materially and adversely affect the holders of the sufficient voting units will be controlled by KKR to take actions the sufficient voting units will be controlled by KKR to take actions the sufficient vot

The voting rights of holders of KKR common units are further restricted by provisions in KKR's partnership agreement sta KKR common units held by a person that beneficially owns 20% or more of any class of KKR common units then outsta than KKR's managing partner or its affiliates, or a direct or subsequently approved transferee of KKR's managing

affiliates) cannot be voted on any matter. KKR's partnership agreement also contains provisions limiting the ability of the KKR common units to call meetings, to acquire information about KKR's operations, and to influence the manner of KKR's management. KKR's partnership agreement does not restrict KKR's managing partner's ability to take actions that KKR's partnership being treated as an entity taxable as a corporation for U.S. federal (and applicable state) income to Furthermore, holders of KKR common units would not be entitled to dissenters' rights of appraisal under KKR's partnersh or applicable Delaware law in the event of a merger or consolidation involving KKR, a sale of substantially all of KKR's other transactions.

65. Additionally, according to defendants, "KKR's partnership agreement contains provisions that reduce or eliminate dution fiduciary duties) of KKR's managing partner and limit remedies available to holders of KKR common units for actions that mig constitute a breach of duty. It will be difficult for holders of KKR common units to successfully challenge a resolution of a conflict of KKR's managing partner or by its conflicts committee." In sum, plaintiffs are receiving units with far less protections than their common shares and results in an even lower value for the consideration.

Defendants Manipulated the Timing of the Proposed

66. KKR, along with the Individual Defendants, used their inside knowledge of KFN's expected financial performance, a publicly available stock prices, to manipulate the timing of the Proposed Acquisition. When the Proposed Acquisition was anno common units were trading at a price near their 52-week high, while KFN common shares were trading at a price near their 52-we following chart compares the trading prices of KKR common units with KFN common shares over trading year prior to announce Proposed Acquisition, demonstrating the overbought nature of KKR common units, relative to the oversold KFN common units, relative to the oversold KFN common units.

The Proposed Acquisition Undervalues the Sha

67. The \$12.79 per share price significantly undervalues both KFN's inherent value and its value to KKR. KFN traded as hi per share as recently as July 18, 2013, and the Proposed Acquisition price represents a meager 9% premium to that high. The Acquisition appears timed to allow KKR to take advantage of a temporary decline in KFN's business, and at a price that does not reprospects. KFN has fallen more than 10% this year as investors expressed disappointment in its strategy. William Sonneborn step July as CEO of the business and was replaced by defendant Farr. Following Farr's appointment, KFN announced it would simplify by reducing investments in natural resources, commercial real estate, private equity and mezzanine financing. KFN's remaining strate collateralized loan obligations, opportunistic credit and specialty lending. Henry Kravis and George Roberts. Co-Chairmen and KKR, stated: "Through this transaction, we are acquiring a business with a fully invested, complementary portfolio of assets while it scale and diversity of KKR's balance sheet. Furthermore, through the distribution of KFN's realized earnings, the transaction is provide a meaningfully greater recurring component to KKR's distribution and also be immediately accretive on a total distribution

68. It was a further breach of defendants' duties to fail to include a collar in the Merger Agreement that would protect KFN from a dwindling of the value of the deal in the event of a decline in the market price of KKR LP units. Moreover, defendants a Proposed Acquisition in breach of their fiduciary duties to KFN's public shareholders, which they brought about through an unfair s Rather than undertake a full and fair sales process designed to maximize shareholder value as their fiduciary duties require, the Box catered to their liquidity goals, as well as to the interest.

The Materially Mi

- 69. On January 15, 2014, defendants filed with the SEC a materially false and misleading S-4 that omitted several crucial factor the Proposed Acquisition. These omitted f
 - (a) with respect to KFN's financial projections, the Downside Case, Base Case, and Aggressive Case provimanagement for calendar years 2014 and 2015 for the following items: (i) revenue; (ii) net income; (iii) book value; and
 - (b) Sandler O'Neill & Partners L.P.'s ("Sandler O'Neill") rationale for not utilizing the financial projections prepared Manager) in any of
 - (c) are the financial projections relied upon by Sandler O'Neill in its analyses, for both KKR and KFN, for all years a the following items: (i) revenue; (ii) economic net income (KKR only); and (iii) assets under management
 - (d) with respect to Sandler O'Neill's comparable company analysis, the following individual multiples and metrics each of the selected public companies comparable to KFN: (i) Price/ Net Asset Value; (ii) Price/ 2013E EPS; (iii) Price/ (iv) Price/ 2015E EPS; and (v) Current Div
 - (e) with respect to Sandler O'Neill's comparable company analysis, whether Sandler O'Neill conducted benchmarking analysis for KFN, in relation to the selected comparable public
 - (f) with respect to Sandler O'Neill's comparable company analysis: the following individual multiples and metrics each of the selected public companies comparable to KKR: (i) 2015E Dividend Yield; (ii) Price/ 2012 Adjusted Economic (iii) Price/ 2013E Economic Net Income; (iv) Price/ 2014E Economic Net Income; and (v) Price/ 2015E Economic
 - (g) with respect to Sandler O'Neill's net present value analysis of KFN: (i) the basis for Sandler O'Neill's selected di range of 7.0% to 12.0%; (ii) Sandler O'Neill's rationale for using a discount rate range of 7.0% to 12.0% for its NPV analysis dividend yields and a discount rate range of 7.0% to 11.0% for its NPV analysis based on tangible book value multiples: O'Neill's rationale for the differing discount range across dividend yields and tangible book value multiples; and (iv) Sand rationale for not using symmetric discount rates around the calculated discount range.
 - (h) with respect to Sandler O'Neill's net present value analysis of KKR: (i) Sandler O'Neill's basis for selecting multiple range of 8.0x to 13.0x; and (ii) Sandler O'Neill's rationale for not using symmetric discount rates around the discount rates.
 - (i) with respect to Sandler O'Neill's analysis of selected merger transactions: (i) the specific Transaction Value/ Boo Share multiples for each of the precedent merger transactions selected by Sandler O'Neill; (ii) whether Sandler O'Neill form of benchmarking analysis for KFN in relation to the selected precedent transactions; and (iii) Sandler O'Neill's including the Annaly Capital Management / CreXus Investment Corp. transaction in both lists when the two lists appear to
 - (j) the specific services, Goldman Sachs has provided to any of the parties involved in the transaction, or any of the in the last two years and how much compensation was received for services.
 - (k) the specific services, Lazard Frères & Co. has provided to any of the parties involved in the transaction, or affiliates, in the last two years and how much compensation was received for services.

Table

- (1) the amount of compensation that Lazard Frères & Co. received for acting as financial advisor to the conflicts con
- (m) the specific services, Sandler O'Neill has provided to any of the parties involved in the transaction, or any of the in the last two years and how much compensation was received for services.

Defendants Agreed to Unreasonable Deal Protec

- 70. As explained herein, the Management Termination Fee amounts to at least \$256.2 million, based on 2012 and 2011 results an effective 10.3% termination fee. On this basis alone, any and all other potential buyers are entirely precluded from bringing
- 71. In addition to the already preclusive Management Termination Fee, defendants implemented further deal protectic guarantee that KKR will not lose its preferred position. In the Merger Agreement, the Board created a playing field that is tilted in fa by agreeing to at least three provisions in derogation of their fiduciary duties to KFN's shareholder
 - a "No Shop" provision that requires KFN to discontinue any discussions with all other potential acquirers and from entering into new discussions, thus precluding the Board from engaging in a fair process to sell the seeking out the best possible price for KFN's shareholders as their fiduciary du
 - a "Matching Rights" provision that requires KFN to give KKR full information about competing acquisition p then allows KKR to match any competing proposal, thus discouraging competing pro
 - a "Termination Fee" provision whereby KFN agreed to pay KKR a \$26,500,000 penalty in the event KFN rece offer for its shareholders, despite the no sho
- 72. Acting on their own self-interest, defendants utilized a defective sales process that was not designed to maximize shareho protect the interests of KFN shareholders, but rather was designed to divert the Company's valuable assets to KKR. Each of the de breached their fiduciary duties of loyalty, due care, independence, candor, good faith and fair dealing, and/or has aided and breaches. Rather than acting in the best interests of the Company's shareholders, defendants spent substantial effort tailoring the structure of the Proposed Acquisition to aggrandize their own personal interests and to meet the specific needs of KKR, which efforts will equity interest of KFN's s

THE INDIVIDUAL DEFENDANTS' FIDUCIARY DUTIES AND THE "ENTIRE FAIRNESS" STANDARD

73. While KFN is an LLC. KFN's operating agreement provides that, except as otherwise provided therein or otherwise recompleted Liability Company Act, the fiduciary duties of KFN's directors will generally be consistent with those of a Delaware corporation. Under Delaware law, in any situation where the directors of a publicly traded corporation undertake a transact result in either (i) a change in corporate control. or (ii) a break-up of the corporation's assets, the directors have an affirmat obligation to obtain the highest value reasonably available for the corporation's shareholders, and if such transaction will result in corporate control, the shareholders are entitled to receive a significant premium. To diligently comply with these duties, the direct take any action that: (a) adversely affects the value provided to the corporation's shareholders; (b) will discourage or inhibit alternate purchase control of the corporation or its assets; (c) contractually prohibits them from complying with their fiduciary duties; (d) we adversely affect their duty to search and secure the

Table

reasonably available under the circumstances for the corporation's shareholders: and/or (e) will provide the directors with preferent at the expense of, or separate from, the public s

- 74. In accordance with their duties of loyalty and good faith, the Individual Defendants, as directors and/or officers of KFN, a to refrain from: (a) participating in any transaction where the directors' or officers' loyalties are divided; (b) participating in any transaction the directors or officers receive or are entitled to receive a personal financial benefit not equally shared by the public shareh corporation; and/or (c) unjustly enriching themselves at the expense or to the detriment of the public states.
 - 75. Plaintiffs allege herein that the Individual Defendants, separately and together, in connection with the Proposed Acquisit fiduciary duties owed to plaintiffs and the other public shareholders of KFN, including their duties of loyalty, good faith. candor, independence. insofar as they stood on both sides of the transaction and engaged in self-dealing and obtained for themselves personal financial benefits, not shared equally by plaintiffs or the Class. As a result of the Individual Defendants' self divided loyalties, neither plaintiffs nor the Class will receive adequate or fair value for their KFN common stock in the Proposed
 - 76. Because the Individual Defendants have breached their duties of due care, loyalty and good faith in connection with Acquisition, the burden of proving the inherent or entire fairness of the Proposed Acquisition, including all aspects of its negotiation price and terms. is placed upon the Individual Defendants as a n

CAUSE OF ACTION

Claim for Breach of Fiduciary Duties and Aiding and Abetting

- 77. Plaintiffs repeat and reallege each allegation set
- 78. The Individual Defendants, aided and abetted by the Company, KKR LP, Fund Holdings and Merger Sub, have violated the duties of care, loyalty, candor, good faith and independence owed to the public shareholders of KFN and have acted to put their personance of the interests of KFN's second of the interest of KFN's s
 - 79. By the acts, transactions and courses of conduct alleged herein, defendants, individually and acting as a part of a commattempting to unfairly deprive plaintiffs and other members of the Class of the true value of their investment.
- 80. The Individual Defendants have violated their fiduciary duties by entering into the Proposed Acquisition without regard to of the transaction to KFN's shareholders. KFN, KKR LP, Fund Holdings and Merger Sub aided and abetted the Individual Defenda of fiduciary duties owed to plaintiffs and the other holders of
- 81. As demonstrated by the allegations above, the Individual Defendants failed to exercise the care required, and breached the loyalty, good faith, candor and independence owed to the shareholders of KFN because, among other reasons, they failed to ensure a and maximization of sharely
- 82. Because the Individual Defendants dominate and control the business and corporate affairs of KFN, and are in possessi corporate information concerning KFN's assets, business and future prospects, there exists an imbalance and disparity of kn economic power between them and the public shareholders of KFN which makes it inherently unfair for them to pursue any propose wherein they will reap disproportionate benefits to the exclusion of maximizing stockl

Table

- 83. By reason of the foregoing acts, practices and course of conduct, the defendants have failed to exercise ordinary care and the exercise of their fiduciary obligations toward plaintiffs and the other members
 - 84. As a result of the actions of defendants, plaintiffs and the Class have been and will be irrepart
- 85. Unless the Proposed Acquisition is enjoined by the Court, defendants will continue to breach their fiduciary duties owed and the members of the Class, will not engage in arm's- length negotiations on the Proposed Acquisition's terms, and will not sup shareholders sufficient information to enable them to make informed decisions regarding the tender of their shares in connect Proposed Acquisition, and may consummate the Proposed Acquisition, all to the irreparable harm of the members
 - 86. Plaintiffs and the members of the Class have no adequate remedy at law. Only through the exercise of this Court's equitable plaintiffs and the Class be fully protected from the immediate and irreparable injury which defendants' actions threat

PRAYER FOR RELIEF

WHEREFORE. plaintiffs demand injunctive relief, in plaintiffs' favor and in favor of the Class and against defendants

- A. Declaring that this action is properly maintainable as a
- B. Declaring and decreeing that the Merger Agreement was entered into in breach of the fiduciary duties of the defendants and unlawful and un
- C. Enjoining defendants, their agents, counsel, employees and all persons acting in concert with them from consummating to Acquisition, unless and until the Company adopts and implements a procedure or process to obtain the highest possible value for s
 - D. Directing the Individual Defendants to exercise their fiduciary duties to obtain a transaction that is in the best interes shareholders and to refrain from entering into any transaction until the process for the sale or merger of the Company is completely highest possible value.
 - E. Rescinding, to the extent already implemented. the Merger Agreement or any of the to
 - F. Awarding plaintiffs the costs and disbursements of this action, including reasonable attorneys' and expe

<u>Table</u>

G. Granting such other and further equitable relief as this Court may deem just

DATED: January 27, 2014

ROBBINS GELLER RUDMAN & DOWD LLP RANDALL J. BARON A. RICK ATWOOD. JR. DAVID T. WISSBROECKER EDWARD M. GERGOSIAN DAVID A. KNOTTS

DAVID A. KNOTTS

655 West Broadway, Suite 1900 San Diego, CA 92101 Telephone: 619/231-1058 619/231-7423 (fax)

LAW OFFICES OF CURTIS V. TRINKO, LLP CURTIS V. TRINKO 16 West 46th Street, 7th Floor New York, NY 10036 Telephone: 212/490-9550 212/986-0158 (fax)

Attorneys for Plaintiffs F-17

<u>Table</u>

Robert S. Green (State Bar James Robert Noblin (State Bar Lesley E. Weaver (State Bar

GREEN & NO

700 Larkspur Landing Circ Larkspu Telephone: (41

> Facsimile: (41 Email: gnecf@class

Attorneys

IN THE UNITED STATES DISTRICT COURT

FOR THE NORTHERN DISTRICT OF CALIFORNIA

GARY BUSHEY, Individually and On Behalf of All Others

Similarly Situated,

Case No.:

Plaintiff,

CLASS ACTION COMPLAINT FOR BREACH OF FIDUCIARY DUTIES AND VIOLATIONS OF FEDER

SECURITIES LAWS

vs.

KKR FINANCIAL HOLDINGS LLC, PAUL M. HAZEN, CRAIG J. DEMAND FOR JURY TRIAL FARR, TRACY COLLINS, ROBERT L. EDWARDS, VINCENT PAUL FINIGAN, R. GLENN HUBBARD, ROSS J. KARI, ELY L. LICHT, DEBORAH H. MCANENY, SCOTT C. NUTTALL, SCOTT RYLES, WILLY STROTHOTTE, KKR & CO. L.P., KKR FUND HOLDINGS L.P., and COPAL MERGER SUB LLC,

Defendants.

CLASS ACTION COMPLAINT FOR BREACH OF FIDUCIARY DUTIES AND VIOLATIONS OF FEDERAL SECURIT

Plaintiff, by his undersigned attorneys, for this class action complaint against defendants, alleges upon personal knowledge w himself and upon information and belief based upon, *inter alia*, the investigation of counsel as to all other allegations herein

NATURE OF THE ACTION

- 1. This is a class action on behalf of the public stockholders of KKR Financial Holdings LLC ("KFN" or the "Company") and its Board of Directors (the "Board" or the "Individual Defendants"), to enjoin a proposed transaction announced on December 1 "Proposed Transaction"), pursuant to which KFN will be acquired by KKR & Co. L.P. ("KKR"), KKR Fund Holdings L.P., an exen partnership ("Fund Holdings"), and its wholly-owned subsidiary, Copal Merger Sub LLC ("Moreover 1 to 1) and its wholly-owned subsidiary.
- 2. On December 16, 2013, the Board caused KFN to enter into a definitive agreement and plan of merger (the "Merger Apursuant to which KKR will acquire all of the outstanding shares of Company common stock in a stock-for-stock transaction stockholders receiving only 0.51 common units of KKR for each common share of KFN, which is an implied value for KFN of \$12 based on the closing prices of KFN and KKR the day the Merger Agreement was executed and announced. At the time of the execution of the Merger Agreement, KKR common units were trading at a price near their 52-week high, while KFN common trading at a price near their 52-week low, thus distorting the fairness of the consideration offered in the Proposed
- 3. The Proposed Transaction is the product of an unfair process and deprives KFN's public stockholders of the ability to particle Company's long-term prospects. Furthermore, in approving the Merger Agreement, the Individual Defendants breached their fiducing plaintiff and the Class (defined herein). Moreover, as alleged herein, KFN aided and abetted the Individual Defendants' breaches duties. Plaintiff also brings claims against the defendants for their violations of Sections 14(a) and 20(a) of the Securities Exchange (the "Exchange Act") and Rule 14a-9 promulgated thereunder ("Foundation of Sections 14a).
- 4. KKR indirectly, externally manages KFN's day-to-day operations, including performing all services relating to the ma KFN's assets, liabilities and operations, and providing KFN with its management team and support personnel. Moreover, KFN has no rather, all of KFN's executive officers and two of its directors are employees of KKR or one or more of its subsidiaries. Further placed six of KFN's current directors on the board of KFN's predecessor, KKR Financial. As such, KKR controls KFN and the transaction must be entirely fair as to process and price. As described herein, the Proposed Transaction is not
 - 5. Compounding the unfairness of the Proposed Transaction, defendants issued materially incomplete and misleading discl Registration Statement filed on Form S-4 with the United States Securities and Exchange Commission ("SEC") on January 1 "Registration Statement"). The Registration Statement is deficient and misleading in that it fails to provide adequate disclosure o information related to the Proposed
 - 6. Plaintiff seeks enjoinment of the Proposed Transaction or, alternatively, rescission of the Proposed Transaction in the ever are able to con

JURISDICTION AND VENUE

7. This Court has subject matter jurisdiction over this action under 28 U.S.C. § 1331 (federal question jurisdiction), as pla violations of Sections 14(a) and 20(a) of the Exchange Act and Rule 14a-9 promulgated thereunder. The Court has supplementa over any claims arising under state law pursuant to 28 U.S.C.

G-1

- 8. This Court has personal jurisdiction over each of the defendants because each either is organized under the laws of, conduin, and maintains operations in this District, or is an individual who either is present in this District for jurisdictional purposes or has minimum contacts with this District as to render the exercise of jurisdiction by this Court permissible under traditional notions of
 - 9. Venue is proper in this District pursuant to 28 U.S.C. § 1391 because: (a) one or more of the defendants either resides in executive offices here; (b) a substantial portion of the transactions and wrongs complained of herein occurred here; and (c) deference substantial compensation and other transfers of money here by doing business here and engaging in activities having an

PARTIES

- 10. Plaintiff is, and has been continuously throughout all times relevant hereto, the owner of KFN co
- 11. Defendant KFN is a Delaware corporation and maintains its principal executive offices at 555 California Street, 50t Francisco, California 94104. The Company, together with its subsidiaries, operates as a specialty finance company with expertise asset classes. KFN's credit segment primarily invests in financial assets, such as below investment grade corporate debt comp secured and unsecured loans, mezzanine loans, private and public equity investments, high yield bonds, distressed and stressed de and marketable equity securities. KFN's common stock is traded on the NASDAQ under the ticker syr
 - 12. Defendant Paul M. Hazen ("Hazen") currently serves as Chairman of the Board. Hazen is also Chairman of Accel-KKR and
 - 13. Defendant Craig J. Farr ("Farr") has served as a KFN director since 2006. According to the Company's website, Farr is Executive Officer ("CEO"). Farr is also the head of KKR Asset Management LLC, an affil
 - 14. Defendant Tracy Collins ("Collins") has served as a KFN director since August 2006. Collins is a member of t

 Compensation
 - 15. Defendant Robert L. Edwards ("Edwards") has served as a KFN director since November 2011. Edwards is a member
 - 16. Defendant Vincent Paul Finigan ("Finigan") has served as a KFN director since February 2006. Finigan is Chair of t Transactions Committee and is a member of the Compensation and Nominating & Corporate Governance
 - 17. Defendant R. Glenn Hubbard ("Hubbard") has served as a KFN director since October 2004. Hubbard is Chair of the N

 Corporate Governance Committee and is a member of the Audit
 - 18. Defendant Ross J. Kari ("Kari") has served as a KFN director since August 2004. Kari is Chair of the Audi
 - 19. Defendant Ely L. Licht ("Licht") has served as a KFN director since
 - 20. Defendant Deborah H. McAneny ("McAneny") has served as a KFN director since April 2005. McAneny is Chair of the Committee and is a member of the Nominating & Corporate Governance and Affiliated Transactions

G-2

Table

- 21. Defendant Scott C. Nuttall ("Nuttall") has served as a KFN director since 1996. Nuttall is also the head of KKR's Global Asset Management Group, and is actively involved in other companies and funds associated.
- 22. Defendant Scott Ryles ("Ryles") has served as a KFN director since November 2008. Ryles is a member of the Audit a Transactions
 - 23. Defendant Willy Strothotte ("Strothotte") has served as a KFN director since January 2007. Strothotte is a m

 Compensation and Affiliated Transactions
- 24. The defendants identified in paragraphs 12 through 23 are collectively referred to herein as the "Individual Defendants." their positions as directors and/or officers of KFN, the Individual Defendants are in a fiduciary relationship with plaintiff and the stockhold
 - 25. Each of the Individual Defendants at all relevant times had the power to control and direct KFN to engage in the miscon herein. The Individual Defendants' fiduciary obligations required them to act in the best interest of plaintiff and all KFN s
- 26. Each of the Individual Defendants owes fiduciary duties of loyalty, good faith, due care, and full and fair disclosure to plate other members of the Class. The Individual Defendants are acting in concert with one another in violating their fiduciary duties as all and, specifically, in connection with the Proposed
 - 27. Plaintiff alleges herein that the Individual Defendants, separately and together, in connection with the Proposed Transaction and are continuing to violate, the fiduciary duties they owe to plaintiff and the Company's other public stockholders, due to the have engaged in all or part of the unlawful acts, plans, schemes, or transactions complain
 - 28. Defendant KKR is a Delaware corporation with its corporate headquarters located at 9 West 57th Street, Suite 4200, New
 - 29. Defendant Fund Holdings is an exempted limited partnership formed under the laws of the Cay
 - 30. Defendant Merger Sub is a Delaware corporation and a wholly-owned subsidiary of Fu

CLASS ACTION ALLEGATIONS

- 31. Plaintiff brings this action as a class action, pursuant to Federal Rule of Civil Procedure 23, on behalf of himself at stockholders of KFN (the "Class"). Excluded from the Class are defendants herein and any person, firm, trust, corporation, or other to or affiliated with an
 - 32. This action is properly maintainable as a
 - 33. The Class is so numerous that joinder of all members is impracticable. As of October 30, 2013, there were approximately shares of KFN common stock outstanding, held by hundreds, if not thousands, of individuals and entities scattered throughout
 - 34. Questions of law and fact are common to the Class, including, as
 - a. Whether defendants have breached their fiduciary duties owed to plaintiff and the
 - b. Whether defendants will irreparably harm plaintiff and the other members of the Class if defendants' conduct conduc

G-3

Table

- 35. Plaintiff is committed to prosecuting this action and has retained competent counsel experienced in litigation of this natural claims are typical of the claims of the other members of the Class and plaintiff has the same interests as the other members Accordingly, plaintiff is an adequate representative of the Class and will fairly and adequately protect the interests
- 36. The prosecution of separate actions by individual members of the Class would create the risk of inconsistent or varying a that would establish incompatible standards of conduct for defendants, or adjudications that would, as a practical matter, be dispositely interests of individual members of the Class who are not parties to the adjudications or would substantially impair or impede the Class members' ability to protect the
 - 37. Defendants have acted, or refused to act, on grounds generally applicable to the Class as a whole and are causing injury Class. Therefore, final injunctive relief on behalf of the Class is

SUBSTANTIVE ALLEGATIONS

Background of th

- 38. According to its website, KFN is a specialty finance company with expertise in a range of asset classes that generates incits majority-owned subsidiaries. Those subsidiaries hold financial assets primarily consisting of corporate loans and securities, incits secured and unsecured loans, mezzanine loans, high-yield corporate bonds, distressed and stressed debt securities, and man non-marketable equity securities. In addition, those subsidiaries hold working and royalty interests in oil and natural gas properties interests in joint ventures and partnerships focused on commercial real estate and special properties.
 - 39. KFN's core business strategy is to leverage the proprietary resources of its manager with the objective of generating income and capital a

KKR C

- 40. KKR controls KFN in a number of respects. For example, KFN is externally managed and advised by KKR Financial A (the "Manager"), pursuant to a management agreement between KFN and the Manager (the "Management Agreement"). The subsidiary of KKR Asset Management LLC, and an indirect subsidiary of Kohlberg Kravis Roberts & Co. L.P., which is a subsidiary of KFN's operations and performs all services relating to the of KFN's assets, liabilities and operations, and provides KFN with its management team and suppo
 - 41. As the Company stated in its Annual Report, filed on Form 10-K with the SEC on Februa

We are highly dependent on our Manager and may not find a suitable replacement if our Manager terminates the Manag

We have no employees. Our Manager, and its officers and employees, allocate a portion of their time to businesses and a are not related to, or affiliated with, us and, accordingly, its officers and employees do not spend all of their time n activities and our investment portfolio. . . . We have no separate facilities and are completely reliant on our Manage significant discretion as to the implementation and execution of our business and investment strategies and our risk practices. We are also subject to the risk that our Manager will terminate the Management Agreement and that no suitable will be found. We believe that our success depends to a significant extent upon the experience of our Manager's execution whose continued service is not

Table

- 42. In addition, according to the Registration Statement, "[a]ll of KFN's executive officers and Scott Nuttall and Craig Farr, t directors, are employees of KKR or one or more of its subsidiaries." The Registration Statement further states that "[d]ue to the created by the management agreement, members of the KFN board of directors and the KFN management team routinely representatives of KKR in the course of their normal dealings with resp
- 43. Furthermore, KKR founded KFN's predecessor, KFN Financial, in July 2004. Since KKR Financial's founding commenced its initial public offering, KKR appointed six of KKR Financial's board members, which all currently serve on the two Board. Accordingly, KKR controls the management and day-to-day operations of KFN, and the Board is behol

KFN's Stockholders' Interests Will Be Drastically

- 44. Should the Proposed Transaction be consummated, KFN's stockholders' ownership and voting interests in KKR will be diminished compared to their interests in KFN. For example, according to the Registration Statement, the KFN stockholders that a common units will own a "much smaller" percentage of KKR than the ownership percentage they currently own in KFN. Moreove to the KFN's stockholders' current rights, holders of KKR common units are not entitled to elect KKR's general partner, and are n elect the directors of KKR's general partner. In addition, holders of KKR common units have only limited voting rights and, therefor no ability to influence management's decisions regarding KKR's business. Because of this, KFN common stockholders will have less influence on the management and policies of KKR than they now have on the management and policies.
- 45. In addition, KKR's managing and general partner, KKR Management LLC, manages the business and affairs of KKR's bus governed by a board of directors that is co-chaired by KKR's founders, Henry Kravis ("Kravis") and George Roberts ("Roberts"), where as KKR's co-CEOs. As the Registration Statement states: "Conflicts of interest may arise among KKR's managing partner and its the one hand, and KKR and holders of KKR common units, on the other hand. As a result of these conflicts, KKR's managing partner its own interests and the interests of its affiliates over KKR and holders of KKR common units." This is so because KKR's managing including its directors and officers, have limited their liability and reduced or eliminated their duties, including fiduciary duties, upartnership agreement, while also restricting the remedies available to holders of KKR common units for actions that, without these might constitute breaches of fiduciary duties.

The Company is Poised for Future Growth

- 46. On July 23, 2013, KFN issued a press release announcing its financial results for the second quarter ended June 30, 2013. A things, the Company reported that net income available to common shareholders totaled \$79.2 million, an increase of 11% from the of 2012, and net income per diluted common share totaled \$0.39. Moreover, a quarterly cash distribution of \$0.21 per common declared for the second quarter of 2013, and book value per common share was \$10.41 as of June 2013.
- 47. With respect to these results, William C. Sonneborn ("Sonneborn"), former CEO of KFN, stated: "Market volatility durin quarter provided us with an attractive window to deploy some of our sizable cash balance. For example, we took the opportunity to CLO, capitalizing on a less crowded market to accumulate a more favorably priced loan portfolio than was possible in the f
 - 48. On September 26, 2013, the Company issued a press release announcing that the Board declared a cash distribution of \$6 share on its outstanding 7.375% Series A LI

Table

Shares. The distribution was made payable on October 15, 2013 to the holders of record of the Preferred Shares as of the close of

- 49. On October 23, 2013, KFN reported that a quarterly cash distribution of \$0.22 per common share was declared for the thi 2013. Individual Defendant Farr, who replaced Sonneborn as KFN's CEO after he stepped down in July 2013, stated in the Octorelease that "we found a number of attractive opportunities for capital deployment during the quarter, and we're focusing the cash-yielding, credit-oriented strategies in order to drive ongoing shareh
- 50. In October 2013, KFN's own financial advisor in connection with the Proposed Transaction, Sandler O'Neill & Partners, L O'Neill"), indicated that the Company's price target was \$25 per share. Moreover, Credit Suisse set a price target at \$21 per share. I price targets, the Company entered into the Merger Agreement for an implied value of only \$12.79 per share for the state of the company entered into the Merger Agreement for an implied value of only \$12.79 per share for the company entered into the Merger Agreement for an implied value of only \$12.79 per share for the company entered into the Merger Agreement for an implied value of only \$12.79 per share for the company entered into the Merger Agreement for an implied value of only \$12.79 per share for the company entered into the Merger Agreement for an implied value of only \$12.79 per share for the company entered into the Merger Agreement for an implied value of only \$12.79 per share for the company entered into the Merger Agreement for an implied value of only \$12.79 per share for the company entered into the Merger Agreement for an implied value of only \$12.79 per share for the company entered into the Merger Agreement for an implied value of only \$12.79 per share for the company entered into the Merger Agreement for an implied value of only \$12.79 per share for the company entered into the
 - 51. Moreover, at the time of the approval and execution of the Merger Agreement, KFN's common shares were trading at a pri 52-week low, but were on the verge of significantly increasing to account for the Company's improved financi
- 52. Given KKR's control over the Company's management and the Board, KKR was aware of KFN's poise for growth and s near future. Attempting to take advantage of the Company's then-depressed trading price, KKR caused KFN to enter into the Merge following a wholly-inadequate process, as det

The Unfair Process Leading Up to the Proposed

- 53. The Proposed Transaction is the result of an unfair process that was tilted in favor of KKR. After receiving an acquisit from KKR, the Board did not conduct a market check or even consider seeking alternative bidders. Moreover, despite the fact the established a committee to consider the Proposed Transaction, the members of the committee are behol
- 54. In October 2013, KKR informed Individual Defendant Hazen, Chairman of the Board and senior advisor to KKR, the considering making an offer to acquire KFN. On October 22, 2013, the Board held a regularly scheduled meeting, during which Haze the Board of KKR's acquisition overtures. The Board discussed the possibility of such an offer, including that the confidentiality rethe Management Agreement could be read to require that KKR obtain authorization from KFN prior to completing the analysis make any acquisition proposal. After discussion, the Board (including Individual Defendants Nuttall and Farr, who are employ granted KKR permission to use the information it had about KFN, as well as to access additional information about KFN (although what additional information), to make an acquisition proposal. The Board allowed this without first engaging and consulting with outside financial and legal advisors. The Board also determined that if KKR made an acquisition proposal, the proposal would be retransaction committee, but the Board did not establish a committee at this time to negotiate the terms of a proposal
 - 55. On October 30, 2013, KKR submitted a letter to KFN, proposing to acquire KFN in a stock-for-stock transaction for common units in exchange for each KFN common units in exchange for
- 56. On October 31, 2013, the Board held a telephonic meeting, during which it designated a transaction committee of purportedly independent directors, Individual Defendants Collins, Edwards, Finigan, Kari, McAneny, and Ryles (the "Transaction Collins, Edwards, Finigan, Kari, McAneny, and Ryles (the "Transaction Collins, Edwards, Finigan, Kari, McAneny, and Ryles (the "Transaction Collins, Edwards, Finigan, Kari, McAneny, and Ryles (the "Transaction Collins, Edwards, Finigan, Kari, McAneny, and Ryles (the "Transaction Collins, Edwards, Finigan, Kari, McAneny, and Ryles (the "Transaction Collins, Edwards, Finigan, Kari, McAneny, and Ryles (the "Transaction Collins, Edwards, Finigan, Kari, McAneny, and Ryles (the "Transaction Collins, Edwards, Finigan, Kari, McAneny, and Ryles (the "Transaction Collins, Edwards, Finigan, Kari, McAneny, and Ryles (the "Transaction Collins, Edwards, Finigan, Kari, McAneny, and Ryles (the "Transaction Collins, Edwards, Finigan, Kari, McAneny, and Ryles (the "Transaction Collins, Edwards, Finigan, Kari, McAneny, and Ryles (the "Transaction Collins, Edwards, Finigan, Kari, McAneny, and Ryles (the "Transaction Collins, Edwards, Finigan, Kari, McAneny, Edwards, Finigan, Kari, McAneny, Edwards, Finigan, Finigan,
 - 57. On November 6, 2013, the Transaction Committee held a telephonic meeting and appointed Edwards as its Chair and reta O'Neill as its financial advisor and Wacl

Table

- Rosen & Katz ("Wachtell Lipton") as its legal advisor. The Registration Statement does not disclose the reasons the Transaction determined to retain these advisors and whether the Transaction Committee considered of
 - 58. On November 12, 2013, the Transaction Committee met again to consider the terms of KKR's proposal and ultimately d inform KKR that the proposed exchange ratio of 0.46 KKR common units per KFN common share was too low and that the Committee preferred a transaction consisting entirely of cash or
- 59. On November 19, 2013, representatives of KKR contacted representatives of Sandler O'Neill to communicate a revised KKR to acquire KFN for consideration consisting entirely of KKR common units, at an exchange ratio of 0.48 KKR common uKFN common share. KKR informed Sandler O'Neill that KKR was unwilling to consider a transaction in which any portion of the common share.
- 60. On November 21, 2013, the Transaction Committee held a telephonic meeting, during which it discussed the terms of acquisition proposal made by KKR with its advisors. The Transaction Committee determined to inform KKR that the proposed exch 0.48 KKR common units per KFN common share still did not offer sufficient value to KFN's stockholders and that the Transaction continued to view a transaction including cash consideration as more preferable than an all-equity
- 61. On November 26, 2013, KKR submitted a revised proposal to acquire KFN for consideration consisting entirely of K units, at an exchange ratio of 0.50 KKR common units per KFN common share. KKR also indicated that KKR remained unwilling transaction in which the consideration included any portion of cash, and that the current offer was KKR's best an
- 62. On December 9, 2013, Individual Defendant Hubbard, who is not a member of the Transaction Committee, met wit Roberts, the co-CEOs of KKR, in order to discuss the Proposed Transaction. At this meeting, Hubbard requested that KKR increase consideration above the most recently proposed exchange ratio of 0.50 KKR common units per KFN common share. Kravis informed Hubbard that KKR was unwill
 - 63. On December 10, 2013, the Board and the Transaction Committee held meetings. Kravis, Roberts, and other KKR reattended a portion of the Board meeting to discuss KKR's business wi
 - 64. KKR ultimately informed Sandler O'Neill that KKR was willing to increase the proposed consideration to only 0.51 K units per KFN common share, with consideration consisting entirely of KKR common units. KKR further stated that it would n circumstances further incre
- 65. Later in the day on December 10, 2013, Hazen, at Edwards' request, contacted representatives of KKR to request that K the proposed exchange ratio to 0.52 KKR common units per KFN common share, but was informed by KKR that the current exchange ratio of 0.51 KKR common units per KFN common share was KKR's best and final offer. This exchange ratio had an imple the Company of only \$12.79 per share based on the closing prices of KFN and KKR the day the Merger Agreement was 6
- 66. On December 13, 2013, the Board held a telephonic meeting to discuss the Proposed Transaction. After a discussic meeting was adjourned, and a meeting of the Transaction Committee was convened. At this meeting, representatives of Sandler O'N an opinion to the Transaction Committee that the consideration offered by KKR was fair, from a financial point of view, to the hol common shares, despite the fact that Sandler O'Neill set a pr

Table

- the Company at \$25 per share only two months prior. Following discussion, the members of the Transaction Committee voted to record the Board approve the Merger Agreement. Following the conclusion of the Transaction Committee meeting, the Board meeting was and the Board approved and declared advisable the Merger Agreement and the Proposed
 - 67. On December 15 and 16, 2013, the board of directors of KKR's managing partner, KKR Management LLC, met to discuss the Proposed
 - 68. On December 16, 2013, KKR, Fund Holdings, Merger Sub, and KFN executed the Merger Agreement, and the parties in press release announcing the Proposed

The Proposed

- 69. The Company entered into the Merger Agreement on December 16, 2013, pursuant to which KKR will acquire all of the shares of common stock of the Company in a stock-for-stock transaction, with KFN's stockholders receiving only 0.51 common units each common shares.
 - 70. On December 16, 2013, KFN filed a Form 8-K with the SEC, attaching as Exhibit 2.1 the Merger
 - 71. To the detriment of the Company's stockholders, the terms of the Merger Agreement substantially favor KKR and are unreasonably dissuade potential suitors from making company to the detriment of the Company's stockholders, the terms of the Merger Agreement substantially favor KKR and are unreasonably dissuade potential suitors from making company.
 - 72. For example, the Individual Defendants have all but ensured that another entity will not emerge with a competing proposal to a "No Solicitation" provision in Section 6.6 of the Merger Agreement that prohibits the Individual Defendants from solicitin proposals and severely constrains their ability to communicate and negotiate with potential buyers who wish to submit or ha unsolicited alternative proposals. Section 6.6(a) of the Merger Agree
 - (a) Subject to Section 6.6(c) through Section 6.6(f), the Company agrees that, from the date of this Agreement until the Company agrees that, from the date of this Agreement until the Company agrees that, from the date of this Agreement until the Company agrees that, from the date of this Agreement until the Company agrees that, from the date of this Agreement until the Company agrees that the Company Agreement in propose its subsidiaries shall, and use its reasonable best efforts to cause its and its subsidiaries' directors, officers, employees, agents, investment banker accountants and other representatives (collectively, "Representatives") not to, directly or indirectly, (i) initiate or solicit of encourage any inquiries with respect to, or the making of a Company Acquisition Proposal, (ii) engage in any negotiations or provide any confidential information or data to any Person relating to, a Company Acquisition Proposal, (iii) approve or recommend, or propose publicly to approve or recommend, any Company Acquisition Proposal, (iv) approve or recommend, agreement, option agreement or other similar agreement relating to any Company Acquisition Proposal (each Acquisition Agreement"), or (v) propose publicly or agree to do any of the foregoing relating to any Company Acquisition
 - 73. Moreover, Section 6.6(g) of the Merger Agreement
 - (g) The Company agrees that it and its subsidiaries will (i) immediately cease and cause to be terminated any existi discussions or negotiations with any Person (other than the parties) conducted prior to the date of this Agreement with re Company Acquisition Proposal and (ii) request that each third party that has heretofore executed a confidentiality ag relates to a Company Acquisition Proposal (other than Parent) return or destroy all confidential information heretofore such third party by the Comp

Table

behalf. The Company agrees that it and its subsidiaries will take the necessary steps to promptly inform its and its Representatives of the obligations undertaken in this

- 74. Further, pursuant to Section 6.6(h) of the Merger Agreement, the Company must advise KKR, within twenty-four laproposals or inquiries received from other parties, including, *inter alia*, the identity of the party making the proposal and the materic conditions of the proposal. Moreover, the Company is required to "keep [KKR] informed orally, as soon as is reasonably practical status of any Company Acquisition Proposal." Section 6.6(h) of the Merger Agree
 - (h) From and after the date of this Agreement, the Company shall promptly orally notify Parent of any request for inform inquiries, proposals or offers relating to a Company Acquisition Proposal indicating, in connection with such notice, the reperson making such request, inquiry, proposal or offer and the material terms and conditions of any proposals or of Company shall provide to Parent written notice of any such inquiry, proposal or offer within 24 hours of such event and converted or electronic correspondence to or from any Person making an Company Acquisition Proposal. The Company shall informed orally, as soon as is reasonably practicable, of the status of any Company Acquisition Proposal, including with restatus and terms of any such proposal or offer and whether any such proposal or offer has been withdrawn or rejected and to shall provide to Parent written notice of any such withdrawal or rejection and copies of any written proposals or information within 24 hours. The Company also agrees to provide any information to Parent (not previously provided to F is providing to another Person pursuant to this Section 6.6 at substantially the same time it provides such information Person. All information provided to Parent under this Section 6.6 shall be kept confidential by Parent in accordance with the Confidentiality
- 75. Moreover, the Merger Agreement contains a highly restrictive "fiduciary out" provision permitting the Board to withdraw of the Proposed Transaction under extremely limited circumstances, and grants KKR a "matching right" with respect to any "Superimade to the Company. Section 6.6(e) of the Merger Agreement of the Company. Section 6.6(e) of the Merger Agreement contains a highly restrictive "fiduciary out" provision permitting the Board to withdraw of the Proposed Transaction under extremely limited circumstances, and grants KKR a "matching right" with respect to any "Superiment Company. Section 6.6(e) of the Merger Agreement contains a highly restrictive "fiduciary out" provision permitting the Board to withdraw of the Proposed Transaction under extremely limited circumstances, and grants KKR a "matching right" with respect to any "Superiment Company. Section 6.6(e) of the Merger Agreement contains a superiment circumstance of the Company.
 - (e) Notwithstanding anything in this Agreement to the contrary, nothing contained in this Agreement shall prevent the Contrary at any time prior to, but not after, the time of the receipt of the Company Shareholder Approval, in response to the receipt unsolicited bona fide Company Acquisition Proposal after the date of this Agreement that did not result from Sections 6.6(a) and (g), (i) terminating this Agreement pursuant to Section 8.1(e) in order to enter into a definitive writtee with respect to such Company Acquisition Proposal or (ii) effecting a Company Change of Recommendation, if, prior to the actions described in clauses (i) or (ii), (A) the Company Board determines in good faith, after consultation with its counsel and financial advisors, that (x) failure to take such action would reasonably be expected to be inconsistent with the directors' fiduciary duties under Applicable Law of the State of Delaware and (y) such Company Acquisition Proposal Company Superior Proposal, (B) Parent shall have received written notice (a "Superior Proposal Notice") of the Company to take such action at least four business days prior to the taking of such action, and (C) the Company Board continues to taking into account any modifications to the terms of the transactions contemplated by this Agreement that are offered by its receipt of the Superior Proposal Notice, that such Company Acquisition Proposal continues to constitute a Comp. Proposal. Any material amendment to the financial terms or any other material amendment of such Company Acquisit shall require a new Superior Proposal Notice and the Company and the Company Board shall be required to comply agreed requirements of this Section 6.6(e); provide

Table

that the reference herein to four business days shall be deemed to be a reference to two business

- 76. Further locking up control of the Company in favor of KKR is Section 8.2(b) of the Merger Agreement, which contains pr "Termination Fee" of \$26.25 million, payable by the Company to KKR if the Individual Defendants cause the Company to terminate Agreement pursuant to the lawful exercise of their fiduciary duties. KFN may also be required to reimburse KKR for its expression of the company to the second second
 - 77. By agreeing to all of the deal protection devices, the Individual Defendants have locked up the Proposed Transact precluded other bidders from making successful competing offers for the
 - 78. The consideration to be paid to plaintiff and the Class in the Proposed Transaction is unfair and inadequate because, things, the intrinsic value of KFN is materially in excess of the amount offered in the Proposed
 - 79. At the time of the approval and execution of the Merger Agreement, KKR common units were trading at a price near the high, while KFN common shares were trading at a price near their 52-week low, thus distorting the fairness of the consideration of Proposed
 - 80. In addition, in October 2013, Sandler O'Neill indicated that the Company's price target was \$25 per share. Moreover, Cree a price target of \$
 - 81. The Proposed Transaction will deny Class members their right to share proportionately and equitably in the true Company's valuable and profitable business, and future growth in profits a
 - 82. As a result, defendants have breached their fiduciary duties they owe to the Company's public stockholders because the will not receive adequate or fair value for their KFN common stock in the Proposed

Defendants' Interests in the Proposed

- 83. As reflected in the Registration Statement, certain of the Company's directors and officers stand to receive substantial result of the Proposed
- 84. For example, as a result of the Proposed Transaction, the directors and officers of the Company stand to receive substantia their currently illiquid shares of Compa

reflected in the table below, the directors and officers own approximately 1.6 million shares in the

	Number of KFN Common Shares
Name and Address of Beneficial Owner	Beneficially Owned
Paul M. Hazen	707,756
Craig J. Farr	0
Michael R. McFerran	152,995
Nicole J. Macarchuk	54,427
Tracy L. Collins	68,043
Robert L. Edwards	19,869
Vincent Paul Finigan	12,607
R. Glenn Hubbard	55,341
Ross J. Kari	57,766
Ely L. Licht	85,943
Deborah H. McAneny	33,616
Scott C. Nuttall	375,564
Scott A. Ryles	32,688
Willy R. Strothotte	0
All officers and directors as a group (14 persons)	1,656,615

85. In addition, the restricted KFN common shares held by KKR's directors and executive officers (who are employed by F converted into 0.51 restricted KKR common units upon consummation of the Proposed Transaction, as reflected in the following

	Restricted
Director/Executive Officer	Shares
Craig Farr	0
Michael R. McFerran	70,353
Nicole J. Macarchuk	44,874
Tracy L.Collins	14,538
Robert L. Edwards	15,429
Vincent Paul Finigan	6,548
Paul M. Hazen	5,057
R. Glenn Hubbard	0
Ross J. Kari	14,538
Ely L. Licht	14,538
Deborah H. McAneny	0
Scott C. Nuttall	0
Scott A. Ryles	14,538
Willy R. Strohotte	0

Table

86. Moreover, each KFN phantom share owned by the directors of KFN will be converted into a fully-vested phantom share reflected by the

Director/Executive Officer	Phantom Shares	Restricted Phantom Shares
Craig Farr	0	0
Michael R. McFerran	0	0
Nicole J. Macarchuk	0	0
Tracy L.Collins	0	0
Robert L. Edwards	0	0
Vincent Paul Finigan	79,179	7,990
Paul M. Hazen	5,770	19,864
R. Glenn Hubbard	211,763	20,768
Ross J. Kari	0	0
Ely L. Licht	0	0
Deborah H. McAneny	53,909	14,538
Scott C. Nuttall	0	0
Scott A. Ryles	0	0
Willy R. Strohotte	179,032	14,538

87. Accordingly, the Company's directors and officers stand to receive significant benefits and thus have reasons to support Transaction, which is otherwise against the best interests of KFN's

The Materially Misleading Registration

- 88. Defendants filed the Registration Statement with the SEC on January 15, 2014 in connection with the Proposed Tra alleged below and elsewhere herein, the Registration Statement omits material information that must be disclosed to KFN's storenable them to render an informed decision with respect to the Proposed
- 89. The Registration Statement omits material information with respect to the process and events leading up to the Proposed as well as the opinions and analyses of KFN's financial advisor, Sandler O'Neill. This omitted information, if disclosed, would signite the total mix of information available to KFN's s
 - 90. The Registration Statement is materially misleading in that it fails to disclose what authority and/or limitations the Board the Transaction Committee with respect to considering any proposal from KKR or seeking alternative and the Transaction Committee with respect to considering any proposal from KKR or seeking alternative and the Transaction Committee with respect to considering any proposal from KKR or seeking alternative and the Transaction Committee with respect to considering any proposal from KKR or seeking alternative and the Transaction Committee with respect to considering any proposal from KKR or seeking alternative and the Transaction Committee with respect to considering any proposal from KKR or seeking alternative and the Transaction Committee with respect to considering any proposal from KKR or seeking alternative and the Transaction Committee with respect to considering any proposal from KKR or seeking alternative and the Transaction Committee with respect to considering any proposal from the Transaction Committee with respect to considering any proposal from the Transaction Committee with respect to considering and the Transaction Committee with respect to considering any proposal from the Transaction Committee with respect to considering and the Transaction Committee with the Transactio
 - 91. The Registration Statement is materially misleading in that it fails to disclose whether KKR's financial advisors in connec Proposed Transaction, Goldman, Sachs & Co. ("Goldman Sachs") and Lazard Frères & Co. LLC ("Lazard"), represented KFN
 - 92. The Registration Statement fails to disclose the reason for selecting Sandler O'Neill and whether the Transaction considered other finance.
 - 93. The Registration Statement is materially misleading in that it fails to disclose the closing price of KFN shares and KKR un KKR submitted an offer to acquire the Company at a certain stock-for-stock exception.
 - 94. The Registration Statement fails to disclose the reason KKR deemed a taxable transaction preferable to a non-taxable given the fact that a stock-for-stock transaction could be structured as a non-taxable

Table

- 95. The Registration Statement does not indicate the reason the Transaction Committee preferred all-cash consideration are refused to consider paying any portion of the consideration.
- 96. The Registration Statement is materially misleading in that it fails to disclose the Company's projections used by Sandler
 - 97. The Registration Statement fails to disclose the source of analyst consensus estimates that was considered by Sar
 - 98. The Registration Statement fails to disclose the projected cost savings and other synergies considered by Sar
- 99. With respect to the Transaction Multiples analysis performed by Sandler O'Neill, the Registration Statement fails to disc there were any dilutive securities reflected in the Company's 204,824,159 outstanding share count, and if so, the fully diluted share correct calculation. In addition, the Registration Statement fails to disclose KFN's net asset value and how it was
- 100. With respect to the Comparable Company Analysis for KFN, the Registration Statement fails to disclose: (i) the company-pricing multiples and metrics and their observed ranges; (ii) the price-to-book multiples, distinguishing between "book value" and "ta value" as necessary; and (iii) how Sandler O'Neill used the analysis in assessing the fairness of the offer, given that no value concreached and no implied exchange ratio was
 - 101. With respect to the Comparable Company Analysis for KKR, the Registration Statement fails to disclose: (i) the company-pricing multiples and metrics and their observed ranges; (ii) the definition of "economic net income" ("ENI") and distinguish E income"; (iii) how "implied ENI growth" was calculated; and (iv) how Sandler O'Neill used the analysis in assessing the fairness given that no value conclusions were reached and no implied exchange ratio was
 - 102. With respect to the Research Analyst Estimates and Price Targets analysis performed by Sandler O'Neill, the Registration fails to disclose the range of price targets rather than just the mean and me
- 103. With respect to the Net Present Value Analysis of KFN performed by Sandler O'Neill, the Registration Statement fails (i) what Sandler O'Neill performed its analysis of, and if cash flow, whether they were levered or unlevered (and the definition (ii) what "circumstances" Sandler O'Neill performed its analysis under; and (iii) the source of the range of P/TBV multiples assume O'Neill's sensitivity table. The Registration Statement also indicates that Sandler O'Neill sensitized three variables (net income 25%], dividend yields [7.0%-12.0%], and dividend amounts [\$0.165-\$0.275]). However, the table on page 74 of the Registration Statement also income variance and dividend yield). This inconsistency must be sensitized to the Net Present Value of the Net Value of the Net Present Value of the Net Value of t
- 104. With respect to the Net Present Value Analysis of KKR performed by Sandler O'Neill, the Registration Statement fail
 (i) what Sandler O'Neill performed its analysis of, and if cash flow, whether they were levered or unlevered (and the definition or f
 (ii) what "circumstances" Sandler O'Neill performed its an
- 105. With respect to the Analysis of Selected Merger Transactions performed by Sandler O'Neill, the Registration State disclose: (i) the transaction-by-transaction values and pricing multiples and the observed ranges; and (ii) how Sandler O'Neill used of Selected Merger Transactions in assessing the fairness of the offer, given that no value conclusions were reached and no implies ratio we

Table

106. With respect to Sandler O'Neill's Pro Forma Results Analysis, the Registration Statement fails to disclose the c "distributable earnings" and "income distribution," and fails to distinguish these from "distributions," "net income," and "Economic

107. Lastly, the Registration Statement is materially misleading in that it fails to disclose the fees received by Sandler O'Neill friends investment banking services and the nature of the

108. As such, plaintiff seeks enjoinment of the Proposed

COUNT I

(Violations of Section 14(a) of the Exchange Act and Rule 14a-9 Against the Company and the Individual Defendants)

109. Plaintiff repeats and realleges the preceding allegations as if fully set

- 110. Defendants have issued the Registration Statement with the intention of soliciting stockholder support of the Proposed
- 111. Rule 14a-9, promulgated by the SEC pursuant to Section 14(a) of the Exchange Act, provides that a proxy statement shal "any statement which, at the time and in the light of the circumstances under which it is made, is false or misleading with respect to fact, or which omits to state any material fact necessary in order to make the statement therein not false or misleading." *S*
 - 112. Specifically, the Registration Statement violates Section 14(a) and Rule 14a-9 because it is materially misleading in numer and omits material facts, including those set forth above. Moreover, in the exercise of reasonable care, defendants should have kn Registration Statement is materially misleading and omits material facts that are necessary to render them non
 - 113. The Company and the Individual Defendants had actual knowledge of the misrepresentations and omissions of material f herein, or acted with reckless disregard
- 114. The misrepresentations and omission in the Registration Statement are material to plaintiff, and plaintiff will be de entitlement to cast a fully informed vote if such misrepresentations and omissions are not corrected prior to the vote on the Proposed

COUNT II

(Violations of Section 20(a) of the Exchange Act Against the Individual Defendants)

115. Plaintiff repeats and realleges the preceding allegations as if fully set

- 116. The Individual Defendants acted as controlling persons of KFN within the meaning of Section 20(a) of the Exchange A herein. By virtue of their positions as directors and/or officers of KFN, and participation in the Company's operations and/or intimat of the false statements contained in the Registration Statement filed with the SEC, they had the power to influence and control, and and control, the decision making of the Company, including the content and dissemination of the various statements that plaintiff false and
 - 117. Each of the Individual Defendants was provided with, or had unlimited access to copies of, the Registration Statements alleged by plaintiff to be misleading prior to and/or shortly after these statements were issued and had the ability to issuance of the statements or cause the statements to
- 118. In particular, each of the Individual Defendants had direct and supervisory involvement in the day-to-day operations of the and, therefore, is presumed to have had the power to control or influence the particular transactions giving rise to the securities viola

Table

exercised the same. The Registration Statement at issue contains the unanimous recommendation of each of the Individual Defendant the Proposed Transaction. They were, thus, directly involved in the making of the

119. In addition, as the Registration Statement sets forth at length, and as described herein, the Individual Defendants were each negotiating, reviewing, and approving the Proposed Transaction. The Registration Statement purports to describe the various information that the Individual Defendants reviewed and considered. The Individual Defendants participated in drafting and/or gave on the content of those

120. By virtue of the foregoing, the Individual Defendants have violated Section 20(a) of the Ex

121. As set forth above, the Individual Defendants have each violated Section 14(a) and Rule 14a-9, by their acts and omissio herein. By virtue of their positions as controlling persons, these defendants are liable pursuant to Section 20(a) of the Exchange Ac and proximate result of Individual Defendants' conduct, plaintiff and the Class will be irrepart

COUNT III (Breach of Fiduciary Duties Against the Individual Defendants)

122. Plaintiff repeats and realleges the preceding allegations as if fully set

123. As members of the Company's Board, the Individual Defendants have fiduciary obligations to: (a) undertake an appropria of KFN's net worth as a merger/acquisition candidate; (b) take all appropriate steps to enhance KFN's value and attract merger/acquisition candidate; (c) act independently to protect the interests of the Company's public stockholders; (d) adequately e conflicts of interest exist between the Individual Defendants' own interests and their fiduciary obligations, and, if such conflicts ex that all conflicts are resolved in the best interests of KFN's public stockholders; (e) actively evaluate the Proposed Transaction and meaningful auction with third parties in an attempt to obtain the best value on any sale of KFN; and (f) disclose all material information in the company's second company second comp

124. The Individual Defendants have breached their fiduciary duties to plaintiff a

125. As alleged herein, the Individual Defendants have initiated a process to sell KFN that undervalues the Company. In agreeing to the Proposed Transaction, the Individual Defendants have capped the price of KFN at a price that does not adequate Company's true value. The Individual Defendants also failed to sufficiently inform themselves of KFN's value, or disregarded the the Company. Furthermore, any alternate acquiror will be faced with engaging in discussions with a management team and B committed to the Proposed

126. As such, unless the Individual Defendants' conduct is enjoined by the Court, they will continue to breach their fiduci

127. Plaintiff and the members of the Class have no adequate re

COUNT IV (Breach of Fiduciary Duty of Disclosure Against the Individual Defendants)

128. Plaintiff repeats and realleges the preceding allegations as if fully set

129. The Individual Defendants have caused materially misleading and incomplete information to be disseminated to the Compstockholders. The Individual Defendants have an obligation to be complete and accurate in their

Table

130. The Registration Statement fails to disclose material information, including financial information and information necessa the statements contained therein from being

131. The misleading omissions and disclosures by defendants concerning information and analyses presented to and consi Board and its advisors affirm the inadequacy of disclosures to the Company's stockholders. Because of defendants' failure to provide disclosure, plaintiff and the Class will be stripped of their ability to make an informed decision with respect to the Proposed Transtatus are dama

132. Plaintiff and the members of the Class have no adequate re

COUNT V (Breach of Fiduciary Duty Against KKR)

133. Plaintiffs repeat and reallege the preceding allegations as if fully set

134. As set forth herein, KKR effectively controls KFN, including running the day-to- day operations of KFN. As such, KK owes fiduciary duties to the Company's s

135. As a product of the negotiated Merger Agreement between KFN and KKR, the Proposed Transaction is subject to end review. The Proposed Transaction provides grossly inadequate consideration to KFN's public stockholders for their Company stock and the Transaction Committee lacked independence from KKR and were unable to truly seek out reasonable alternatives or maximing result of KKR's likely refusal to permit the Board or Transaction Committee to seek strategic alternatives to its proposal. As such, to Transaction is unfair as to price

136. Plaintiffs and the Class have no adequate remedy at law. Only through the exercise of this Court's equitable powers can put the Class be fully protected from the immediate and irreparable injury that KKR's actions threat

COUNT VI (Aiding and Abetting the Board's Breaches of Fiduciary Duties Against KFN)

137. Plaintiff repeats and realleges the preceding allegations as if fully set

- 138. KFN, by reason of its status as a party to the Merger Agreement and/or possession of nonpublic information, has aided an aforementioned breaches of fidu
- 139. Such breaches of fiduciary duties could not and would not have occurred but for the conduct of KFN which, therefore, have abetted such breaches in KKR's attempted acquisi
- 140. As a result of the unlawful actions of KFN, plaintiff and the Class will be irreparably harmed in that they will not receive t KFN's assets and business. Unless the actions of KFN are enjoined by the Court, they will continue to aid and abet the breaches duties owed to plaintiff a

141. Plaintiff and the Class have no adequate re

PRAYER FOR RELIEF

WHEREFORE, plaintiff prays for judgment and relie

A. Ordering that this action may be maintained as a class action and certifying plaintiff as the Class representative and plain

<u>Table</u>

- B. Preliminarily and permanently enjoining defendants and all persons acting in concert with them from proceeding with, co or closing the Proposed
 - C. In the event defendants consummate the Proposed Transaction, rescinding it and setting it aside or awarding rescissory plaintiff at
 - D. Directing defendants to account to plaintiff and the Class for their damages sustained because of the wrongs complain
 - E. Awarding plaintiff the costs of this action, including reasonable allowance for plaintiff's attorneys' and expe

F. Granting such other and further relief as this Court may deem jus

DATED: January 31, 2014

GREEN & NOBLIN, P.C.

By: /s/ ROBERT S. GREEN

Robert S. Green

700 Larkspur Landing Circle, Suite 275 Larkspur, CA 94939

Telephone: (415) 477-6700 Facsimile: (415) 477-6710

OF COUNSEL:

Seth D. Rigrodsky Brian D. Long Gina M. Serra

RIGRODSKY & LONG, P.A.

2 Righter Parkway, Suite 120 Wilmington, DE 19803 Telephone: (302) 295-5310

Attorneys for Plaintiff G-17

Form of Preliminary Proxy